

Some pages of this thesis may have been removed for copyright restrictions.

If you have discovered material in Aston Research Explorer which is unlawful e.g. breaches copyright, (either yours or that of a third party) or any other law, including but not limited to those relating to patent, trademark, confidentiality, data protection, obscenity, defamation, libel, then please read our <u>Takedown policy</u> and contact the service immediately (openaccess@aston.ac.uk)

A Holistic Supply Chain Integration theoretical framework within the Maltese Manufacturing SMEs

A Grounded Theory Approach

Ronald Cuschieri

Doctor of Philosophy

ASTON UNIVERSITY

September 2015

©Ronald Cuschieri, 2015

Ronald Cuschieri asserts his moral right to be identified as the author of this thesis.

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and that no quotation from the thesis and no information derived from it may be published without proper acknowledgement.

ASTON UNIVERSITY A Holistic Supply Chain Integration theoretical framework within the Maltese Manufacturing SMEs Ronald Cuschieri Doctor of Philosophy

2015

Purpose: The thesis focus is to generate a theory on supply chain integration (SCI) within the Maltese manufacturing SMEs. Due to the current literature gaps there is lack of scholarly work associated with the theory of how to achieve SCI across the supply chain (SC) (Zhao et al., 2008; Flynn et al., 2010) and on holistic SCI frameworks (Holmberg, 2000; Mentzer et al., 2001; Flynn et al., 2010; Garetti & Taisch, 2012). The thesis seeks to address these gaps by trying to establish two main objectives: the SCI substantive theory; and the SCI holistic conceptual framework, needed to be adopted both within focal firm and beyond the dyad relationships across the SC.

Method: The Straussian Grounded Theory methodology (GTM) was used to inductively derive the SCI theory. The research data collection and analysis was performed on 22 in-depth interviews from 17 firms within various manufacturing sector, using both primary and secondary data. The grounded theory data analysis was based on both Computer Assisted Qualitative Data Analysis Software (CAQDAS) and storyline narrative approaches to serve as data analytic triangulation tools so as to validate the emerged theory. Analytic intricacies and iterative sense-making was pivotal in all theorising stages to arrive at the final substantive theory with its conceptual framework.

Findings: The research findings inductively established that the substantive theory is based on a crossdisciplinary approach mainly consisting of the following disciplines: management, leadership, business, manufacturing and SC strategies respectively and technology, both within information systems and manufacturing automated processes. The research established a substantive theory for SMEs to achieve SCI, which revolves around the integrative management and leadership approach (IMLA). The IMLA is considered to be an original conceptual term within the SCI theory, and it is defined to be both a driver and a binding force of all SC actors, based on the integration of a dual concept of management and leadership. The management approach maintains all the necessary quality standards and sustainable measures in place and at the same time streamline and optimise all activities to meet competitively all operational and business targets. The leadership approach focuses ideally on a shared leadership style, to promote a decentralised management approach based on teamwork initiatives, but it is not excluded that firms may adopt an individual leadership style, based on a centralised management approach. The research also established a set of propositions to contribute to the discipline scholarly work, which may also be tested by quantitative future research to increase the research robustness.

Limitations: The findings are based on a relatively small sample of 17 firms involved in various positions within the SC.

Practical & Methodological implications: The conceptual model with its underpinning theory and the recommended road map are to serve as a guide for all managers across the command chain in various SC positions to deploy the right IMLA, as the core theme within the conceptual model, to achieve outstanding competitive performance. This research also extends the applicability of GTM within the management discipline.

Originality: This research outlines an original concept within SCI literature, which is a core driver, represented by the term IMLA. Such an IMLA is able to achieve a holistic SCI management approach beyond the SC dyadic relationship by adopting four leadership traits, situated within the strategic perspective of the firm based on the business, manufacturing and SC strategies with the role of technology as an enabler and even a driver to achieve competitive capabilities.

Keywords: Supply chain management, Technology, Leadership, Management and Grounded Theory.

Acknowledgements

I would like to sincerely thank my supervisors, Professor Prasanta Dey and Dr Judy Scully for their continuous supervision during this project. In particular, Professor Prasanta for his academic and practical expertise on the subject discipline and Dr Scully for her commitment to guide me through the Grounded Theory Methodology and for her dedication and patience, who always added words of encouragement to improve, exchanged several mails with both feedback and appreciation to recognise my work and also for all the support given whenever needed, for which I am truly grateful. I would also like to thank the Malta Government Scholarship Scheme (MGSS) who sponsored me throughout this PhD.

I would also like to thank Dr Joseph Vancell, Dr Francis Debono, Dr Nicola Rathbone, Dr Clive Zammit, Dr Vimal Kumar Eswarlal, Dr Alex Rizzo and especially my father, Alfred Cuschieri, for their willing ear and constructive comments during all my PhD stages.

I would also like to thank all the participants who were interviewed and who were ready to share their experiences and knowledge on the subject considering their busy schedules, especially other University of Malta colleagues who supported me and offered all their help such as Dr Joseph Agius and several others who are being left out not to miss anyone from the list. Special thanks go to Mr Alex Borg for his commitment to help me access key people within the Maltese industry.

Finally, I would like to thank my family and friends for all their understanding, enthusiasm and encouragement during the PhD, especially my daughters Laura and Michela, and last but not least my wife Marcelle, who had to put up with the long hours during the last five years, which I had to dedicate to my PhD to complete it successfully.

List of Contents

Page Number

Thesi	is Summary	/	
Ackn	owledgeme	ent	
List o	of Contents.		
List o	of Tables		
List o	of Figures		
List o	of abbreviat	ions	
Chap	oter 1: Intr	oduction	
1.1	Research g	eneral backg	round 14
1.2	The Malte	se manufactu	ring scenario
1.3	Research p	ourpose	
1.4	Research r	broblem state	ment
1.5	Research s	significance.	
1.6	Research of	objectives and	d research question
1.7	Summary	of the aim of	the study within the manufacturing SC 2
1.8	Thesis stru	icture	
Chap	oter 2: The	A Priori an	d Post Literature Reviews 23
2.1	Introduct	tion	
2.2	The a pri	iori literature	review
2.3	The outli	ine of the a p	riori systematic literature review
2.4	Overview	v of the supp	ly chain integration theory within the a priori literature review 20
2.5 The foundational themes within the supply chain integration literature			nes within the supply chain integration literature
	2.5.1	The manufa	cturing supply chain theoretical outline
	2.5.2	Global virtu	al networks within the manufacturing supply chain
	2.5.3	The High V	alue Manufacturing supply chain
	2.5.4	The manufa	cturing SCM challenging perspective
	2.5.5	Supply chai	n integration within SCM
	2.5.6	Supply chai	n integration strategic review
	2.5.7	Supply chai	n integration theoretical underpinning foundations
	2.5.8	Technology	deployment: SC information systems and manufacturing processes 30
	2.5.9	Supply chai	n integration literature gaps
	2.5.10	Information	technology literature gaps within the supply chain context
	2.5.11	The tentativ	e research question
	2.5.12	Summary of	f the a priori literature review
2.6	The post	literature rev	view associated with the emerged theory
	2.6.1	The outline	of the post systematic literature review
	2.6.2	Overview of	f all the themes within the emerged theory
	2.6.3	The core car	tegory: Integrative Management and Leadership Approach 44
		(IMLA)	
	2.6.3.1	The manage	ement concept synthesis within the IMLA core category
		2.6.3.1.1	The management timeline
		2.6.3.1.2	The management role
		2.6.3.1.3	SCM perspective
		2.6.3.1.4	The summary of the management concept from both the focal 4'
			firm and SC perspectives.
	2.6.3.2	The leaders	hip concept within the IMLA from the focal firm perspective 4
		2.6.3.2.1	The leadership timeline
		2.6.3.2.2	Strategic leadership/shared leadership against individual leadership. 44

2.6.3.2.3	Transformation leadership trait
2.6.3.2.4	Servant leadership trait.
2.6.3.2.5	Transactional leadership trait
2.6.3.2.6	Situational leadership trait
2.6.3.2.7	The role of power and control in leadership: Empowerment and trust
Leadership	theories from the SC perspective
2.6.3.3.1	The strategic leadership approach across the SC
2.6.3.3.2	The Leadership trait within SCM: A stakeholder theory perspective
2.6.3.3.3	SC actors' power based relationships: The power matrix
2.6.3.3.4	SC actors' collaborative relationships: Buyer-Suppliers
	relationship type framework
2.6.3.3.5	The aims of collaborative relationships
2.6.3.3.6	SC actors' value-adding roles
2.6.3.3.7	SC actors' alignment for performance improvement
Summary o	f the leadership concept from both the focal firm and SC perspective
2.6.3.4.1	The leadership traits: transformational, transactional, servant and
	situational
2.6.3.4.2	The integrated leadership approach
2.6.3.4.3	Effective leadership with the right strategic power approach
26344	A holistic and integrated management and leadership approach
Business s	trategy theme
2641	Corporate and business strategies
2642	The husiness operations strategy
Manufactu	iring strategy theme
2651	Manufacturing strategy contribution to competitiveness
2.6.5.2	The manufacturing strategy and manufacturing leadership practices
2.6.5.2	The manufacturing strategy and SCI
2.0.5.5 2654	The manufacturing strategy and SCI
Supply Ch	ain (SC) Strategy theme
2661	SC strategy and supply chain integration to achieve strategic alignment
2.6.6.2	SC strategy from a manufacturing and marketing perspective
2.0.0.2	Strategic supply chain integration with all SC actors
2.0.0.3	SC strategic decision based on the manufacturing 'make' or
2.0.0.4	'buy' strategic decision
2665	The SC strategy as a function of the dominant players
2.0.0.5	The SC strategy through process integration: The SCOP
2.0.0.0	practitioner's model
Holistic su	unly chain integration (SCI) management theme
2671	The holistic SCI approach within its contextual conditions
2.0.7.1	Holistic SCI approach: Agile and optimised SC and
2.0.7.2	manufacturing operations
Technolog	ry deployment (in information systems and automation) theme
2681	Information Technology (IT) as an anablar of SC and
2.0.6.1	monufacturing processos
2682	IT promotes SCI
2.0.0.2	IT promotes SCI
2.0.8.3	norsportive
Compatiti	perspective
	Sustainable compatitive capabilities within the facel firm and
2.0.9.1	Sustainable competitive capabilities within the local firm and
	across the SC from a DDV parapative
2602	across the SC from a RBV perspective
	2.6.3.2.3 2.6.3.2.4 2.6.3.2.5 2.6.3.2.6 2.6.3.2.7 Leadershij 2.6.3.3.1 2.6.3.3.2 2.6.3.3.3 2.6.3.3.3 2.6.3.3.4 2.6.3.3.5 2.6.3.3.6 2.6.3.3.7 Summary of 2.6.3.4.1 2.6.3.4.2 2.6.3.4.3 2.6.3.4.3 2.6.3.4.4 Business s 2.6.4.1 2.6.5.2 2.6.5.3 2.6.5.4 Supply Ch 2.6.6.1 2.6.6.2 2.6.6.3 2.6.6.4 2.6.6.5 2.6.6.6 Holistic su 2.6.7.1 2.6.7.2 Technolog 2.6.8.1 2.6.8.2 2.6.8.3 Competitiv

		2.6.9.3	Competitive capabilities through quality management: A RBV perspective	80
		2.6.9.4	Competitive capabilities from an individual assigned priority or from a cumulative approach	81
		2.6.9.5	Competitive capabilities direct/indirect relationship to SCI operational performance outcomes	81
	2.6.10	Literature	gaps from the post literature review	82
		2.6.10.1	Leadership literature gaps	82
		2.6.10.2	Supply chain integration literature gaps	83
		2.6.10.3	Sustainability literature gaps	84
		2.6.10.4	Manufacturing literature gaps	84
	2.6.11	Summary	of the a priori and post literature reviews	85
Chap	ter 3: Res	search Meth	odology	87
3.1	Introduc	tion		87
3.2	Research	h methodolo	gy justification	87
3.3	Methode	ology selecti	on	90
3.4	The Gro	ounded Theo	ry approach: Straussian methodology overview	90
3.5	Seminal	works assoc	ciated with the Grounded Theory Method (GTM)	92
3.6	Researc	h philosophi	cal approach and research quality	93
	3.6.1	Epistemolo	ogy	93
	3.6.2	Ontology.		94
	3.6.3	Research c	Juality	94
3.7	Research Temporal Design, Sampling Strategies, Sample Size and Unit of Analysis 98			
	3.7.1	Research t	emporal design	98
	3.7.2	Sampling s	strategies	98
		3.7.2.1	First stage sampling strategy	99
		3.7.2.2	Second stage sampling strategy	100
	3.7.3	Sample siz	e	101
	3.7.4	Unit of ana	alysis	102
3.8	Researc	h ethics		102
3.9	Groundir	ng the emerge	d theory in the data and the role of the a priori and post Literature reviews	103
3.10	The GT	M research o	lesign three coding techniques	106
	3.10.1	Open Codi	ing stage	106
	3.10.2	Axial Cod	ing stage	107
	3.10.3	Selective C	Coding stage	108
3.11	Data col	llection and	analysis data management tools and techniques	109
	3.11.1	Computer-	assisted qualitative data analysis software (CAQDAS)	109
	3.11.2	Qualitative	e storyline technique supported by displays	11(
		3.11.2.1	Qualitative storyline technique	11(
		3.11.2.2	Displays or mind maps technique	111
		3.11.2.3	Triangulation of data analysis techniques and data sources	111
3.12	Impleme	entation outl	ine of the overall research data collection and analysis phases	112
	3.12.1	Research r	nethod phase 1: Open sampling with Open Coding analysis	112
		technique	assisted by CAQDAS.	
	3.12.2	Research r	nethod phase 2: Open sampling with Axial Coding analysis	110
		technique	based on the storyline analysis approach	
	3.12.3	Research r	nethod phase 3: Theoretical sampling with Selective Coding based vline analysis approach under the GTM	118
3.13	Chapter	summary		120
		•		

Chapt	ter 4: Dat	a Collection	and Analysis based on the Open and Axial Coding	123	
	nques			100	
4.1	Introduc	ction to the o	verall data collection and analysis three phases	123	
4.2	Phase 1: Data collection and analysis based on the Open Coding technique			123	
4.3	Phase 2:	Data collect	tion and analysis based on the Axial Coding technique	124	
	4.3.1	Introductio	in to the Axial Coding stage	124	
	4.3.2	Findings fr	om the open sampling data collection under the Axial Coding	125	
		analytic ap	proach		
		4.3.2.1	Managing technology deployment tentative category	126	
		4.3.2.2	Leading effectively tentative category	130	
		4.3.2.3	Managing process within the SC and manufacturing tentative	134	
		4224	Category	140	
		4.3.2.4	Managing supply chain integration tentative category	140	
		4.3.2.5	Managing the SC strategy category	145	
		4.3.2.6	Managing the manufacturing strategy tentative category	152	
		4.3.2.7	Managing the business strategy tentative category	159	
4.4	Overvie	w of the cha	pter	165	
Chapt	ter 5: Dat	a Collection	and Analysis based on the Selective Coding Technique	167	
5.1	Introduc	tion to Selec	tive Coding stage	167	
5.2	Finding	s from the th	eoretical sampling data collection under the Selective Coding	167	
	analytic	approach			
5.3	The Coding Paradigm to represent the emerged theory				
	5.3.1	Actions Ca	tegories of the theory: The core category with the five main	168	
	01011	categories		100	
		5.3.1.1	The sixteen sub-categories within the holistic SCI management	170	
	532	Outcomes	Categories of the theory: Competitive canabilities	172	
	533	Context ca	tagories of the theory. Contextual conditions	172	
	524	Context ca	m the analysis of the secondary data sources	170	
	5.5.4		Desticinent Observation	170	
		5.5.4.1		1/8	
5 4		5.5.4.2	Analysis of Documentary Evidence	1/9	
5.4	The fina	ll emerged th	leory	180	
5.5	Overvie	w of the cha	pter	181	
Chapt	ter 6: The	eory Genera	ition	183	
6.1	Introduc	tion to the su	ubstantive theory generation	183	
6.2	Summar	y of the eme	rged grounded theory	183	
6.3	The rese	The research theory based on the core category around all of its categories			
	6.3.1	The Core C	Category: The Integrative Management and Leadership approach	184	
		(IMLA)			
	6.3.2	The busine	ss strategy.	186	
	6.3.3	The manuf	acturing strategy	187	
	634	The supply	v chain strategy	188	
	635	The IMI A	in-conjunction with the business manufacturing and supply chain	189	
	0.5.5	strategies	in conjunction with the business, manufacturing and suppry chain	109	
	6.3.6	The holisti	c SCI management approach	190	
	6.3.7	The techno	ology deployment.	195	
	6.3.8	Competitiv	e capabilities	196	
6.4	Conclus	ion	*	198	

Chapt	ter 7: Disc	cussion - The	eoretical Integration	199
7.1	Introduc	tion		199
7.2	The eme	nergent theory within the extant literature		
7.3 A high level overview of the key findings: The IMLA explanation, originality,		v of the key findings: The IMLA explanation, originality,	200	
	significa	nce, definitio	on and challenges	
	7.3.1	The IMLA of	explanation and its originality	200
	7.3.2	The study sig	gnificance and literature gaps from a multiple theoretical perspective	201
	7.3.3	Overview of	f the challenges to achieve SCI within the literature relative to the	202
		research cor	re theme	
	7.3.4	Comparison	of the key contemporary theoretical models from the literature	203
		with the em	erged theory	
		7.3.4.1	Integrated chain management study	203
		7.3.4.2	SC leadership and SC followership study	204
		7.3.4.3	Quality management practices study	205
		7.3.4.4	Supplier integration mechanisms study	206
7.4	The eme	rgent themes	within the extant literature	207
	7.4.1	The core the	eme: Comparison of the integrative management and leadership	207
		approach (II	MLA) within the extant literature	
		7.4.1.1	The Management concept within the IMLA based on the focal	207
			firm and across the SC	
		7.4.1.2	The Leadership concept within the IMLA based on the focal	209
		/////_	firm and across the SC: Strategic leadership style against	_0/
			individual leadership style	
		7.4.1.3	The IMLA Four Leadership Traits: Transformational.	211
		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Transactional Servant and Situational leadership styles	
		7414	The IMLA's collaborative types (close collaborative and arm's	213
		,	length) & processes	210
		7415	The IMLA vis a vis SCI and performance	215
		7.4.1.5	IMI A's collaborative initiatives: cooperative/non-coercive and	215
		7.4.1.0	aggressive/coercive strategies	215
		7417	The IML A's human element from a social sustainable	217
		//	nerspective within the SCM literature	217
	742	Rusiness str	rategy and the IML A within the supply chain context	220
	7.4.2	Comparison	of the manufacturing strategy emerged theory within the extent	220
	7.4.5	literature	for the manufacturing strategy emerged theory within the extant	221
		interature		
		 7 / 3 1	High value manufacturing and sustainability within SMEs	221
		7.4.3.1	Manufacturing Stratagy and Supply Chain Stratagy for the SC	221
		7.4.3.2	competitiveness	
		7133	The IMLA and the manufacturing strategy for a competitive	222
		7.4.3.3	strategy	223
		7131	The manufacturing strategy and the operational capabilities	222
		7.4.3.4	The convitigation of monufacturing	223
	7 4 4	7.4.3.3 Commoniaen	The servicization of manufacturing	224
	1.4.4	Comparison	To the SC strategy emerged theory within the extant literature	224
		/.4.4.1	I he IMLA within the SC Strategy with its virtual enterprise	224
		7 4 4 2	approach: vertical or virtual integration	225
		1.4.4.2	SU strategy design to fit different manufacturing sectors	226
		1.4.4.3	The sustainable SU strategy	227
		1.4.4.4	The SC Strategy enabled by technology deployment for	227
			performance improvements	- - ·
		1.4.4.5	The SC strategy long term effects	228
	7.4.5	Holistic SC	I management approach	229

	7.4.5.1The SC strategy from a holistic SCI management approach229
	enabled by technology
_	7.4.5.2 Holistic SCI management within manufacturing
7.4	6 Technology deployment in information systems and automation
7.4	7 The competitive capabilities
7.5 Su	nmary of the discussion chapter
Chapter 8	Conclusion
8.1 Int	oduction
8.2 Re	earch scope and the substantive theory main issues and challenges
8.3 Re	earch significance
8.4 Th	research emerged theory outline based on the core theme: The IMLA 246
8.5 Th	oretical, methodological and practitioners' contributions
8.5	1 Theoretical contributions 248
8.5	2 Methodological contributions 254
8.5	3 Contributions to practice and to policy
8.6 Re	earch limitations
8.7 Su	gestions for future research
8.8 Su	nmary of the conclusion
List of Re	erences
Appendic	s
Appendix	: Overview of the Maltese manufacturing scenario
A1.1 The	Ialtese Business Scenario vis a vis the EU and Global Economy
A1.2 The	conomic Statistics of the Maltese Manufacturing SMEs
A1.3 Malt	's infrastructure within the manufacturing SC
Appendix	A Priori Literature Review Significant Concepts
Appendix	: Interview guide for convenient/open sampling, interview letter & consent form 311
Appendix	: Holistic SCI management theme with its 16 sub-themes
A4.1 The	xteen sub-themes within the holistic SCI management approach
A4.1.1 I	ean Management theme 317
A412 N	anaging Sustainability theme 317
A413 N	anaging Performance Measurement theme 320
A414 N	anaging Auditing Operations 320
$\Delta 1 15$	anaging Supply Management theme
A416 N	anaging suppry Wanagement theme
A4.1.0 N	anaging class-now
A4.1.7 N	anaging change and hinovation.
A4.1.0 N	anaging conadoration
A4.1.9 N	anaging culture
A4.1.10 N	anaging knowledge
A4.1.11 N	anaging customer service
A4.1.12 N	anaging information
A4.1.13 N	anaging logistics
A4.1.14 F	anning and forecasting
A4.1.15 N	anaging quality
A4.1.16 N	anaging risks
A4.1.17 N	anaging trust
Appendix	A breakdown of all the participants under the both open sampling and 337
	theoretical sampling phases
Appendix	: Interview guide for theoretical sampling phase

Appendix 7:	Interview guide for theoretical sampling for researcher with Probe List	342
Appendix 8:	Laundry list of all Open Codes from MAXQDA 2007 code segment	345
Appendix 9:	CAQDAS Outcome of all Emerged Categories for all the Eight Interviews	348
Appendix 10:	CAQDAS Outcome of a Filtered Set (based on a frequency count > 10) of	355
	Emerged Categories for all the Eight Interviews	
Appendix 11:	Axial Coding of the 15 non-key tentative categories	358
	A11.1 Auditing Operations tentative category	358
	A11.2 Managing Cash Flow tentative category	360
	A11.3 Managing Collaboration tentative category	363
	A11.4 Managing Change tentative category	367
	A11.5 Managing Information Sharing/Communication tentative category	369
	A11.6 Managing Customer Service tentative category	372
	A11.7 Managing Knowledge tentative category	374
	A11.8 Lean Management and Quality Standards Compliance category	376
	A11.9 Managing Logistics tentative category.	379
	A11.10 Managing Performance Measurements tentative category	382
	A11.11 Planning and forecasting tentative category	385
	A11.12 Managing Risks tentative category	388
	A11.13 Supply Management tentative category	392
	A11.14 Managing Trust tentative category	397
	A11.15 Managing Culture tentative category	400
Appendix 12:	The IMLA supporting strategic actions to achieve competitive capabilities	403
11	A12.1 The business strategy	403
	A12.2 The Manufacturing Strategy	403
	A12.3 The Supply Chain Strategy.	403
	A12.4 The holistic SCI management approach	404
	A12.4.1 Auditing Operations.	404
	A12.4.2 Supply Management.	404
	A12.4.3 Managing Knowledge	404
	A12.4.4 Managing Cash Flow	404
	A12.4.5 Managing Change/Innovation	404
	A12.4.6 Managing Collaboration	404
	A12.4.7 Managing Culture.	404
	A12.4.8 Managing Customer Service.	405
	A12.4.9 Managing Information	405
	A12.4.10 Managing Lean Operations	405
	A12.4.11 Managing Logistics	405
	A12.4.12 Managing Performance Measurements	405
	A12.4.13 Managing Planning and Forecasting	405
	A12.4.14 Managing Quality and Sustainability	406
	A12.4.15 Managing Risks	406
	A12.4.16 Managing Trust	406
	A12.4 5 The Technology Deployment	406
		100

List of Tables

Table 3.1:	Research Methodology Outline	91
Table 3.2:	Research Methodology Phases, Coding Stages & Sampling Strategies	101
Table 3.3:	A demo showing the audit trail of five Open Codes	115
Table 4.1:	Breakdown list of all the data collection and analysis phases	123
Table 4.2:	List of tentative categories	125
Table 4.3:	Managing technology deployment tentative category	129

Table 4.4:	Leading effectively tentative category	133
Table 4.5:	Managing process within the SC and manufacturing tentative category	138
Table 4.6:	Managing SCI tentative category	143
Table 4.7:	Managing the SC strategy tentative category	150
Table 4.8:	Managing the manufacturing strategy tentative category	157
Table 4.9:	Managing the business strategy tentative category	163
Table 4.10:	Open Coding to Axial Coding Code Matrix	166
Table 5.1:	Code Matrix for the categories within the research phases	181
Table 7.1:	Summary of the Conceptual model categories within the literature	241
Table 8.1:	Propositions from all the emerged substantive theory	254
Table A1.1:	GDP at current market prices	302
Table A1.2:	Manufacturing by NACE code with imports and exports levels	307
Table A1.3:	Gross domestic product at market prices	307
Table A11.1:	Auditing operations tentative category	359
Table A11.2:	Managing Cash Flow tentative category	362
Table A11.3:	Managing Collaboration tentative category	366
Table A11.4:	Managing Change tentative category	368
Table A11.5:	Managing Information Sharing/Communication tentative category	371
Table A11.6:	Managing Customer Service tentative category	373
Table A11.7:	Managing Knowledge tentative category	375
Table A11.8:	Lean Management and Quality Standards Compliance tentative category	378
Table A11.9:	Managing Logistics tentative category	381
Table A11.10	: Managing Performance Measurements tentative category	384
Table A11.11	: Planning and forecasting tentative category	387
Table A11.12	: Managing Risks tentative category	391
Table A11.13	: Supply management tentative category	389
Table A11.14	: Managing trust tentative category	398
Table A11.15	: Managing culture tentative category	401

List of figures

0		
Figure 1.1:	Thesis structure (Author)	24
Figure 2.1:	SCM processes two key components	46
Figure 2.2:	The buyer and supplier power matrix	56
Figure 2.3:	Buyer–supplier relationship types	57
Figure 3.1:	The Grounded Theory research strategy	104
Figure: 3.2	Screenshot of MAXQDA 2007 used for the Open Coding stage data	113
collection and	d analysis	
Figure 3.3:	Code memo sample from MAXQDA 2007	114
Figure 3.4:	Snapshot of collaboration Open Code from MAXQDA 2007	116
Figure 4.1:	Managing technology deployment category tentative category	130
Figure 4.2:	Leading effectively tentative category	134
Figure 4.3:	Managing Process within the SC and manufacturing tentative category	139
Figure 4.4:	Managing SCI tentative category	145
Figure 4.5:	Managing the SC strategy tentative category	152
Figure 4.6:	Managing the manufacturing strategy tentative category	159
Figure 4.7:	Managing the business strategy tentative category	165
Figure 5.1:	Outline of the overall GTM coding data analytic approach	168
Figure 6.1:	The emerged SCI theoretical model	197
Figure 7.1:	The SCI theoretical model based on the emerged main categories	201
Figure 7.2:	The final SCI theoretical model	235
Figure A1.1:	Industrial Production Index for EU and G-20 members	301

List of abbreviations

Abbreviation	Meaning
3BL	Triple Bottom Line sustainable measures
BOM	Bill of Material
BSI	British Standard Institution
CAD	Computer-Aided Design
CAM	Computer-Aided Manufacturing
CAQDAS	Computer-Assisted Qualitative Data Analysis Software
CIM	Computer-Integrated Manufacture
CEG	Collaborative Enterprise Concept
CPFR	Collaborative Planning Forecast and Replenishment
CRM	Customer Relationship Management
CSF	Critical Success factors
CSR	Corporate Social Responsibility
DE	Defunct Enterprise
DREG	Dynamic Enterprise Reference Grid
DSS	Decision Support Systems
EAI	Enterprise Application Integration
EDI	Electronic Data Interchange
EE	Extended Enterprise
EFQM	European Foundation for Quality Management
EHS	Environmental, Health & Safety
EIS	Executive Information Systems
EPC	Electronic Product Code
ERDF	European Regional Development Fund
ERP	Enterprise Resource Planning system
EU	European Union
Eurostat	European Statistical System
FMS	Flexible Manufacturing Systems

FP	Framework Protocol
GDP	Gross Domestic Product
GMP	Good Manufacturing Practices
GSCM	Green Supply Chain Management
GTM	Grounded Theory Method
GVA	Gross Value Added
HVM	High Value Manufacturing
ICT	Information Communication Technology
IEA	International Energy Agency
IMLA	Integrative Management and Leadership Approach
IOS	Inter-Organizational Systems
ЮТ	Internet of Things
IS	Information Systems
ISA	International Society of Automation
ISIC	International Standard Industrial Classification of all Economic
	Activities of the United Nations
ISO	International Standard Organisation
IT	Information Technology
JIT	Just-in-time
MBNOA	Malcolm Baldrige National Quality Award
MESA	Manufacturing Enterprise Solutions Association
MFSA	Malta Financial Services Authority
NACE	European Classification of Economic Activities
NSO	National Statistics Office (Malta)
OECD	Organisation for Economic Cooperation and Development
OM	Operations Management
PESTEL	Political, Economic, Social, Environmental & Legal
PL	Party Logistics (e.g. 3PL & 4 PL)
PPS	Purchasing Power Standard
RBV	Resource Based View
RFID	Radio Frequency Identification Code
RoHS	The Restriction on the Use of Certain Hazardous Substances in
	Electrical and Electronic Equipment Directive
SaaS	Software as a Service
SC	Supply Chain
SCI	Supply Chain Integration
SCM	Supply Chain Management
SCOR	Supply Chain Operations Reference
SIS	Small Island State
SMEs	Small & Medium Enterprises
SOA	Service-Oriented Architecture
SSCM	Sustainable Supply Chain Management
SWOT	Strength, Weaknesses, Opportunities & Threats
ТСА	Transaction Cost Analysis
TUV	Technical Inspection Association
UL	Underwriters Laboratories standard
VIE	Vertical Integrated Enterprise
VE	Virtual Enterprise
VMI	Vendor Management Inventory
WEEE	Waste of electrical and electronic equipment (amend)
WEF	World Economic Forum

Chapter 1: Introduction

1.1 Research general background: The role of Supply Chain Integration within manufacturing based SMEs enabled by technology to meet sustainability and competitiveness

Maltese SMEs, as all other SMEs across the globe, are considered the engine of the economy since they possess: the potential to pursue innovative investments in niche markets; the ability to generate opportunities for employment; and the capabilities to provide a pivotal contribution to the nation's economy (Curran & Blackburn, 2001; Demirbag et al., 2006; Ernst & Young, 2013). In spite of the SMEs potential within the worldwide economy, there exists limited studies covering research based on the supply chain operations within the SMEs (Curran & Blackburn, 2001; Bayraktar et al., 2009), especially in the Maltese Melitensia.

In general, SMEs in different sectors, lack the resources relative to large firms (e.g. capital, labour and equipment), where such resources are also function of the SME type, such as either being a Multinational based SME or else as a standalone contractor or else a family owned business (Curran & Blackburn, 2001; Bayraktar et al., 2009; Ernst & Young 2013). Such a lack of both resources and purchasing power is a well established phenomenon in SMEs and unless they are supported by Government initiatives and programmes or larger firms, they need to carry a greater burden, relative to large firms, to meet the demanding environmental, social and economical sustainable measures (Luken & Stares, 2005; Lee, 2008; Lee & Klassen, 2008; Glover et al., 2014; Hsueh, 2015). SMEs play a key role within the supply chain (SC) context, by taking various roles, such as suppliers, distributors, producers and customers (Hong & Jeong, 2006).

Within the manufacturing sector it is a well conceived requirement, that nowadays for firms to stay in business, there is the need for an integrated strategic relationship between various firms and not to operate as autonomous entities (Dyer, 1994; Cox, 1999; Lambert & Cooper, 2000; Fawcett & Magnan, 2004; Bagchi et al., 2005; Ketchen & Hult, 2007; Koufteros et al., 2007). Though, it cannot be excluded that a well organised SC set-up may still stifle or hinder a certain level of change and innovation due to the established adopted systems, traditional mindset, legacy applications and established procedures (Neuman & Samuels 1996; Sorenson, 2003). On the other hand, one has to consider that with SCI there is more to gain than to lose, as a result of the synergies and teamwork efforts with well committed and non-adversarial relationships with all SC actors (Gunasekaran et al., 2008; Cheung et al., 2012; Huo, 2012; Schoenherr & Swink, 2012).

Current manufacturing practices are focussing on high value manufacturing (HVM), which consist from the application of leading-edge technical knowledge and expertise for the creation of products, production processes and associated services, which have strong potential to bring sustainable growth and high economic value (Technology Strategy Board, 2012). Such a manufacturing context requires the effective management of supply chains to support the contemporary HVM manufacturing initiatives, having a distributed and fragmented SC stakeholders across the globe (Livesey, 2006; Martinez et al., 2008; Bayraktar et al., 2009). This manufacturing paradigm shift incorporates a change from vertical integration to virtual integration (Prahalad et al., 1994; Power, 2005; Wang & Wei, 2007; Gunasekaran et al., 2008). SCI integration initiatives are pivotal in such a challenging business scenario, since they are able to create a sound platform for optimizing and streamlining all manufacturing processes across all SC tiers up to the end customer (Cohen & Lee, 1988; Gunasekaran et al., 2008; Koufteros et al., 2007; Lambert, 2008; Bayraktar et al., 2009; Huo, 2012). Such a stance is achieved, since with SCI, the SMEs are in a position to gain from joint-commitments and synergistic collaborative effects, in all strategic, tactical and operational management levels, both from staff within the focal firm and also from teamwork efforts with all of its stakeholders.

The role of information technology (IT), within the above SC context, is considered part and parcel of such a manufacturing scenario, since IT has the capabilities to provide a platform for all interconnections of all SC actors, both locally and globally. In fact, from the SC literature, it has been established that one of the key drivers of this change, is technology based on the internet platform to promote open technology based systems (Metcalfe et al., 2002; Muffatto & Payaro, 2004; White et al., 2005; Sanders, 2007). Furthermore, to meet such a design and development of new products and services, within this competitive manufacturing environment, SC actors need a technological-based infrastructure, as a source of information integration, such as enterprise resource planning systems (ERPs), for streamlining of all operations and for leveraging organizational competitiveness (Simatupang et al., 2004; Shi & Gregory, 2005; Cassivi, 2006; Gunasekaran et al., 2008; Sanders, 2008; Hoch & Dulebohn, 2013). A SC actor can either employ an in-house ERP system, covering all operations from customer relationship management to all daily operations' needs, by deploying an enterprise application integration (EAI) system to have interoperability between the multiple disparate systems across the SC (Loh et al., 2006). Else, a SC actor can exploit the cloud-based supply chain management application, through outsourced enterprise class solution at an attractive cost (Metcalfe et al., 2002). Such an ERP platform, with its interconnection of all SC actors, provides information visibility and supply chain flexibility, being the major advantages provided by IT support to achieve SCI (Fawcett et al., 2007; Wang & Wei, 2007; Gunasekaran et al., 2008; Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014).

Furthermore, with today's current demands for sustainable operations, from the triple bottom-line perspective, it is much easier for the focal firm in-conjunction with all SC actors to employ an integrated approach so as to work in unison and share common standards and procedures to meet sustainability across the overall SC, so as to produce the expected sustainable product. As a result, this business scenario is creating a challenging strategic horizon, which can be transformed to success, once all opportunities in the market are exploited, but at the same needs an effective management approach, to manage the overall SC from a sustainable perspective, so as to promote competitive capabilities to survive and win in the market. SCI provides such an effective management solution.

1.2 The Maltese manufacturing scenario

Malta has a GDP of $\in 6.4$ billion and a population of around 400,000 living in an area of 316 sq. km, and is strategically located at the heart of the Mediterranean (Eurostat, 2012). Malta is one of the smallest and most densely populated countries in the world and is considered as a Small Island State. Malta is the smallest euro area economy, highly dependent on trade and tourism. Foreign-owned non-financial corporations dominate the domestic market and the economy relies on foreign direct investment (FDI) flows and European Union (EU) funds for investment. Over the last decade, the Maltese economy underwent a gradual shift from the manufacturing towards the services sector, but still manufacturing has still a pivotal role but with a declining contribution to the economy.

Whereas in the year 2000, manufacturing accounted for 22.4% of Gross Value Added (GVA), in 2010, it only accounted for 13.6%. On the other hand, the services sector contributed to 59% of GVA in 2010, up from 53% in 2000 (Malta Council of Science and Technology, 2011). In their 2010 report on the subject, Angelou Economics (2010) recommended that the future economic development efforts must continue to support the transition to higher value added opportunities, founded on a knowledge-based economy. In the manufacturing sector, the majority of enterprises (92% in 2010) were engaged in medium to low and low-technological based activities, with only a minority (8% in 2010) of manufacturing enterprises falling within the high or moderate to high technological categories.

Small Island states (SISs), such as Malta, have always been labelled as non-viable and with a higher level of vulnerability than resilience (Bertram, 2004). Though, past wealth statistics, especially since Independence in 1964 and with the subsequent EU membership in 2004, depict Malta as a resilient country which performed competitively well, even when compared to larger countries performance statistics, such as UK and other European countries, as referred in the World Economic Forum (WEF, 2014a) performance statistics, throughout the last years. In fact, Malta has been scoring in economic wealth reasonably well in the last decade, especially in its main sectors, such as Tourism, Financial

Markets and Services, Information Technology (such as gaming and betfair), Healthcare and Value Added Manufacturing. A profile of the GDP outline for the last seven years' timeline by Eurostat (2014) for EU member states; the Maltese track record performance statistics relative to other EU countries; the Maltese economy assessment with respect to the 12 pillars of competitiveness by the WEF (2014a) in its Global Competitiveness Report (2014–2015); the Maltese outline of SMEs within manufacturing with the overall GDP contribution together with an overview of the different types of manufacturing sectors; and the manufacturing supply chain infrastructure, are all referred in Appendix 1.

1.3 Research purpose

The research aims to explore the role of SCI for Small and Medium Sized Enterprises (SMEs) within the Maltese manufacturing. In particular, the thesis looks at the interaction of all SC actors within the SC. The scope is to empirically explore what is happening within SCI initiatives taken by SMEs in the Maltese manufacturing sector, so as to generate a substantive theory to represent the reality of this phenomenon from practice. As a result the purpose of this study is to establish the required characteristics to achieve supply chain integration across the SC.

The research strategy is based on a Grounded Theory methodology, so as explore what the data are saying from 22 in-depth interviews based on two sampling phases. Various data analytic management tools are used to support all the data analysis during three main stages (i.e. Open, Axial and Selective Coding), to promote research scientific rigour, such as the CAQDAS and the storyline method supported by narrative based-memos and visual aids, in particular, logic diagrams as tables, code matrices as tree diagrams and conceptual frameworks as mind maps. The established SCI reality is subjective and with a multiple perspective, since it is socially constructed through a constructivist epistemological approach, but the generated theory still meets the credibility, plausibility and trustworthiness research quality criteria (Glaser & Strauss, 1967). Such research validity and reliability criteria are met, since the grounded theory method is employed with scientific rigour, where the generated substantive theory is generalizable to the theory grounded in data (Bryman, 2008), through the constant comparison process of verification (Strauss & Corbin, 1994), but it has contextual uniqueness and significance to the sample of the study (Bryman, 2008). Such a context is function of the sample of participants from the various firms with their different roles, which span different SC tiers within the manufacturing sector, such as suppliers, production plants, freight forwarders and IT contractors. All SC actors have their individual micro contextual conditions and a common Maltese macro context, which are all operating with an international scope, which is function of their supply chain operations within their destination market players (e.g. customers and manufacturers).

1.4 Research problem statement

In the SCM arena, SC practices show that innovative and holistic managerial and technological practices are still limited on how to achieve SCI (Holmberg, 2000; Mentzer et al., 2001; Flynn et al., 2010; Garetti & Taisch, 2012). It was also inferred that, 'Managing the entire supply chain is a very difficult and challenging task' (Lambert & Cooper, 2000, p.68). The SC strategy needs to decide on how to gain from the optimum performance from each SC members so as to eliminate the risks associated with sub-optimisation effect that may result from friction between SC actors optimal (Cox et al., 2004a; Narayandas & Rangan, 2004). There exist firms which are hesitating and are finding it difficult in integrating operations and are not convinced with the potential benefits and the competitive edge attained through the deployment of SCI initiatives (Braunscheidel & Suresh, 2009; Thakkar et al., 2009; Flynn et al., 2010; Zhao et al., 2011). Similarly, some firms are neither convinced by the benefits that can be attained from the investment in technology based applications (Dos Santos & Sussman, 2000; Devaraj et al., 2007). Furthermore, holistic SCI frameworks are lacking within the extant literature (Vickery et al., 2003; Pero et al., 2010; Schoenherr & Swink, 2012). The literature on the social element focused on people, regarding the labour, skills and the forming of relationships, are a neglected area of research (McCarter et al., 2005; Shub & Stonebraker 2009; Ashby et al., 2012; Lengnick-Hall et al., 2013). Finally, managing through a strategic leadership approach based on a distributed or shared leadership style is very challenging to achieve when one considers the pervasive influence of the single centric individual leadership approach due to the leading role that top management takes on all the people across all management levels (Tsui et al., 2006; Ling et al., 2008).

The Maltese Government, together with the EU established programmes, such as the new 2014-2020 'SME assistance' Programming Period (Ernst & Young 2013, p.8), are undertaking various incentives to encourage and support further all SMEs investments, so as to meet the contemporary competitive manufacturing global scenario. In fact, the Malta Enterprise is providing several supporting measures, such as soft loans, tax credits and loan interest subsidy, among other schemes (Malta Enterprise, 2012). In spite of such a positive business climate for all SMEs, a number of firms are still finding it difficult to invest and meet the challenges offered within their respective sectors.

One cannot exclude the fact, that Malta, as a Small Island State (SIS), all firms engaged within the supply chain cannot exploit the economies of scales advantages, due to the relative small sized orders, which are leading to higher costs per unit of production, which makes it more challenging to compete on a global scale. Such a low level of throughput has a domino effect on all associated SC actors, such as suppliers, distributors and carriers (i.e. freight forwarders). Furthermore, being a SIS, all SC actors also have to face the insularity and peripherality challenging characteristics, since they need to carry

higher associated logistics costs, both with respect to the lack containers' volume efficient usage and for the greater transport distances that need to be covered to reach most of the destination markets.

Furthermore, all the above must be seen within the context of sustainable operations, both within all manufacturing design and operations and also for all logistics companies, who need to manage effectively, through dedicated processes, so as: to economize their operations to meet profitability; to manage all the human resources base to be social responsible; and to meet all environmental sustainable challenges through all the necessary investments and effective running of all sustainable processes. From all the three sustainable measures (i.e. 3BL), the environmental performance is determining all the other two sustainable measures, with today's stringent environmental standards and corporate social responsibility requirements (Elkington, 2002; Manderson, 2006; Bojarski et al., 2009). As a result, the customer, which can be a SC player or an end-user, is nowadays expecting that all their products are to be environmental sustainable from all aspects of materials, transport and manufacturing processes, otherwise they do not buy (Vachon & Klassen, 2006; Zhu et al., 2008; Caniels et al., 2013).

1.5 Research significance

The research significance is based on the fact that manufacturing is the source of all goods for a society and that SCM is undergoing a major paradigm shift, with great deal of changes and complexities, which are transforming the way the processes are being implemented. Such a paradigm shift is being attributed to several reasons such as: changing business environment (i.e. political, economical and legal), increased globalization, information technology developments, corporate social responsibility and sustainable green operations requirements and internet-based 24x7 management systems. This SCM dynamic environment is substantiated by various scholars such as Handfield & Melnyk (1998); Devaraj et al. (2007) and Flynn et al. (2010).

Furthermore, it was claimed that SCM is still in its embryonic/infancy stage (Pagell, 2004; Zhao et al., 2008) and as a result research is quite limited to explain how to achieve SCI (Zhao et al., 2008) and how to determine the influencing factors of SCI (Flynn et al., 2010). The study validity is further reinforced by assertion that the 'theory surrounding supply chain integration is still under developed' (Schoenherr & Swink, 2012, p.110). Finally, it was stated that the social element, 'in terms of labour, skills and the forming of relationships' area is a neglected research area (Ashby et al., 2012, p.506) and that 'downstream integration has received little attention in supply chain research' (Guan & Rehme, 2012, p.188). From the leadership literature, it was also asserted that within the 'management of integrated supply chain' is a 'largely ignored area' (Defee at al., 2010, p.764). There are few

scholarly works which explicitly consider network based leadership in the macro context of interorganisational networks (Sydow et al., 2011). Finally, the manufacturing sector needs further research in the area of 'holistic research approach integrating technological, managerial and business-model solutions together with a change in the customer behaviour, with the perspective of a more sustainable manufacturing for the production of more sustainable products and services' (Garetti & Taisch, 2012, p.88).

Such contextual complex requirements and the nature of SCM dynamic changes with the lack of theories and explanations on how to achieve SCI within manufacturing, makes the research exploratory and challenging to promote a significant contribution.

1.6 Research objectives and research question

The research aims to identify and analyse the supply chain integration (SCI) initiatives taken by all SMEs, as SC actors, within the manufacturing sector. The intention is to establish all the characteristics needed to achieve a holistic SCI approach within the contemporary contextual conditions, such as the economic challenges (e.g. recessions), the competitive environment and the demanding sustainable practices. Such characteristics are to be represented in a theoretical model, to outline the best practices which are to be adopted by SMEs within the Maltese manufacturing to achieve competitive capabilities.

Therefore, this research focuses on the holistic SCI practices within SMEs in the Maltese manufacturing. As a result, the two research objectives are:

• Research Objective 1 (RO1): To identify the characteristics needed to achieve SCI within SMEs in the Maltese manufacturing; and

• Research Objective 2 (RO2): To develop a SCI theoretical framework for SMEs in the Maltese manufacturing.

The research question that guides the study and supporting objectives is the following two-part question. First, which are the characteristics needed to achieve SCI within SMEs in the Maltese manufacturing? Second, which are the key factors that constitute SCI that promotes competitive performance within SMEs in the Maltese manufacturing?

1.7 Summary of the aim of the study within the manufacturing SC

From Malta's manufacturing profile with its assessment (Section 1.2), one can clearly refer, that Malta as Small Island State has a very strong logistics network, backed by an excellent port infrastructure and a knowledge-based economy with substantial EU support. With such a context, this research scope is to revamp the supply chain operations related to all SMEs within the manufacturing sector, both for local and international owned firms, since the SC is strong as its weakest link. The local scene based SC is able to gain from various advantages that promotes HVM, through easier proactive commitment and lean operations based on more sustainable practices, in view of the small size of the island. The focus of the research is to determine the SCI improvement initiatives needed on high technological and high value added manufacturing. Such a research is also in line with the global manufacturing SC perspective, which is also undergoing a major paradigm shift with great deal of change and complexity, and is transforming the way the processes are being implemented in line with HVM principles.

In summary it is a common understanding and a key current concern of all firms engaged in the manufacturing sector with their stakeholders across the SC, that for them to remain competitive, there is the pivotal need to work together in an integrated approach. Such a stance needs the right management approach supported by an investment in the latest technology manufacturing equipment and in the latest ICT applications to promote real-time flow of information with all of its stakeholders, backed by an efficient logistics backbone for all products movements. The SC needs the right management approach with all the necessary measures, so as to support the employees in all requirements and enhance their capabilities through on-going training initiatives. Such a stance enables the management to formulate high value manufacturing strategies with the right engagement of the human element and with the deployment of the right technology-based processes, in both information systems (IS) and manufacturing processes to produce high quality products with higher value-added activities and customised services. There is also the need to undertake a revamp in the approach to promote innovative processes and products through dedicated innovative initiatives, both by the private sector and even through the Government support initiatives, by providing access to special grants and incentives to maintain and improve on the business sophistication pillar stone, through change management commitment in research and innovation, as referred in the WEF (2014a) report. Such a context need to be seen through the diligent effort that needs to be undertaken by all stakeholders to promote economical, social and environmental sustainability, by adhering to all safety practices, by complying to all standards and legal requirements and also to protect and harness the environment apart from economic prosperity initiatives. With such an approach, Malta's SMEs, within the context of the contemporary worldwide severe competition, needs to sustain and improve so

as to survive in the market, by leveraging its performance and innovative capabilities, through a high value-added manufacturing approach, so as to compete effectively, even relative to low cost countries, such as China, being one of the world's major economies.

1.8 Thesis structure

The thesis consists from a Grounded Theory based research methodology which requires that the researcher undertakes two literature reviews to meet the requirements of such a research strategy. The thesis structure is referred in Figure 1.1 and is also explained below.

Following this introductory chapter, the thesis proceeds with Chapter 2, which provides the literature review. Chapter 2 contains two literature reviews, dedicated to the a priori and the post literature reviews. The scope of the **a priori literature review** is to establish the: research objectives; research question; the design of the initial interview guide; literature gaps; and a broad in-sight of the phenomenon understudy with the establishment of the key significant concepts, which are provisional, in line with the Straussian Grounded Theory Method (Strauss & Corbin, 1998). The scope of the **post literature review** is to undertake a systematic literature review on all the emerged themes building the generated substantive theory and also to establish the literature gaps, so as to situate the emerged theory within the extant literature. This approach serves as a platform for Chapter 7 theoretical integration process, which is in line with both Straussian and Glaserian GTMs (Strauss, 1987; Glaser, 1998).

Chapter 3 justifies, selects, examines and outlines the Straussian Grounded Theory research methodology. It also justifies the role of the literature review, with special reference to the a priori and the post literature reviews. It includes an outline of the two sampling stages and the three coding techniques, with the support of the CAQDAS and storyline narrative approaches. It also explains the philosophical approach used to generate the theory from the data. Finally, reference is made to the adopted research quality criteria and the research ethics.

Chapter 4 is dedicated to the Open Coding (Phase 1) and Axial Coding (Phase 2) stages of the data collection and analysis process using the CAQDAS and storyline narrative approaches, with the support of memos and visual aids, in particular, logic diagrams as tables, code matrices as tree diagrams and conceptual frameworks as mind maps. Such two coding stages with all the tools referred, assist the theorising process, which finally establish a set of 22 tentative categories with their conceptual summaries.

Chapter 5 is dedicated to the refinement of the tentative categories, established from the previous chapter. This is achieved by designing a new interview guide based on the previous chapter emerged categories to undertake theoretical sampling as the research third phase (Phase 3), with a set of new interviews. Such a data collection and analysis process is focused on such emerged themes to achieve theoretical saturation of all the emerged theory. Such a chapter covers the Selective Coding stage, which as a result establishes the core theme that links all the emerged final action categories and outcome categories, within their contextual conditions, in line the coding paradigm (Strauss & Corbin, 1998).

Chapter 6 is dedicated to the generation of the SCI substantive theory, by outlining a summary memo of the final theory, which is a direct result of these iterative analytic processes of the two previous chapters in line with the research objective 1 (i.e. RO1). It also builds up the SCI holistic framework that represents the SCI phenomenon in line with the research objective 2 (i.e. RO2), based on SMEs within the Maltese manufacturing.

Chapter 7 outlines how the emerged theory from the data is integrated within the extant literature, in line with the GTM for logical extension and comparison of the emerged theory with the literature, by identifying the commonalities, the differences and the new (Strauss, 1987; Glaser, 1998) and also for theory building (Eisenhardt, 1989). The scope of this chapter also answers the research question and derives the study implications to the extant literature and also refines the emerged theory (Strauss, 1987; Glaser, 1998; Strauss & Corbin, 1998) and addresses the literature gaps.

Chapter 8, is the final chapter, it concludes the thesis by outlining the key findings and the research contributions generated from the data. In particular, the researcher draws out the research key ideas, some original insights and the substantive theory outline as the roadmap for practitioners. The researcher also establishes a set of propositions and the substantive theory implications to the extant literature. The research study further establishes the contributions to the methodology and to policy and practice and also includes the research limitations and ideas for future research.



Figure 1.1: Thesis structure (Author)

Chapter 2: The A Priori and Post Literature Reviews

2.1 Introduction

This chapter explains the role of the a priori and the post literature reviews, as the two main sections of this chapter. The first part (Section 2.2) is dedicated to the **a priori literature review**, which provides an overview of the literature that informs the research area. It establishes the foundations of supply chain integration (SCI) initiatives within manufacturing together with all the related literature, such as supply chain management (SCM), information technology (IT) and other concepts which define and explain the issues and challenges underpinning the phenomenon. It also establishes the literature gaps from both a SCI and an IT perspectives, the research objectives and question and leads to the decision of the research methodology, with its initial interview guide to kick-off the data collection and analysis process. The second part (Section 2.6) is dedicated to the **post literature review**, which is driven by Chapter 6 emerged theory from the data, as referred in Figure 1.1. The post literature review is used by the researcher to undertake the core part of the systematic literature review of the thesis, by focussing on the generated substantive theory with its emerged themes and establish the contemporary literature gaps in leadership, SCI, sustainability and manufacturing. The literature review, with its established gaps, is used in the discussion chapter (Chapter 7) to establish the commonalities and the differences from the emerged theory so as to derive the research contribution.

2.2 The a priori literature review

The systematic literature review is performed on the SCI phenomenon and its related topics for the researcher to be informed and understand the phenomenon understudy from a broad perspective through multidisciplinary theories within the extant literature. As a result several themes are derived from the extant literature to give a sound understanding of what constitutes SCI within manufacturing. Basically, the literature review within this section is used to establish the: research significance and its context; manufacturing supply chain (SC) outline; virtual networks within the SC; HVM within the SC; SCM challenging perspective; SCI within SCM; SC strategic review; SCI theoretical foundations; and technology deployment within the SC and manufacturing; and four significant concepts from the extant literature: information integration, strategic integration, strategic collaboration and information technology. Finally it is also established the: SCI challenges; literature gaps, both within SCI and IT; research objective; research question; and first interview guide.

2.3 The outline of the a priori systematic literature review

To achieve the systematic literature review, both standardised and customised search filters are used through established keywords under the researched phenomena, since the discipline of SCM has a broad spectrum of published research. The search inclusion criteria in the title, keywords and abstract are: supply chain; supply chain integration and manufacturing; and information technology. Sources used in the search are books from academic journal publishers, commercial academic search engines (such as Google Scholar), university catalogues and databases (such as EBSCO, ProRequest and Science Direct), trade journals, conferences proceedings and recognised websites with unpublished work which are more targeted to practitioners, such as such as Supply Chain Council (SCC); Global SC Forum; The Chartered Institute of Logistics (CILT); Council of SCM Professionals (CSCMP); Supply Chain Standard.com (SCS); Toolbox.com for IT; and Realtime NEXUS for IT.

The initial searches are based on relatively recent publications (i.e. 2000 to date). From the gathered academic papers, all cites which are considered very useful in assisting the systematic literature review, are also manually retrieved from the respective journal, to get the original sources, to promote better evaluation and synthesising of the literature to promote research rigour.

2.4 Overview of the supply chain integration theory within the a priori literature review

The term SCM was introduced by consultants in 1980s (Oliver & Webber, 1992) and started to gain structure as a management term and was being treated as logistics management outside the firm with suppliers and customers (Lambert & Cooper, 2000). Today SCM is being treated differently and is being defined to comprise from the integration of logistics together with the integration of all business processes (Lambert & Cooper, 2000; Cooper et al., 1997). Hence, logistics was then being treated as part of SCM. With the drastic increase in the global competition, the SCM paradigm has shifted to global sourcing with: worldwide distributed customers; low inventories or fast inventory turnovers; low cash-flow; low cost; and faster product (order) fulfilment cycles. This made the SC more sensitive to information and products flow with deeper engagement with all SC entities.

Similarly, the SCI evolution passed through several stages, with several descriptive terms and with the support of different information systems applications, such as JIT, cellular, lean, mass customisation, concurrent engineering (CE), flexible manufacturing systems (FMS), business process re-engineering (BPR), enterprise resource planning (ERP), customer relationship management (CRM), vertical integration and virtual integration (Stonebraker & Afifi, 2004). Another perspective, adding to the previous list of developments, included SCM, materials requirements planning (MRP), manufacturing

resource planning (MRP II), suppliers relationship management (SRM), advanced planning systems (APS), electronic data interchange (EDI), agile IS, decision support systems (DSS), extended enterprise application (EEA) and enterprise application integration (EAI) (Bayraktar et al., 2009). With such a timeline of activities made each SC set-up contingent of its employed technologies and processes, which also made a difference on the nature of the management approach requirements across the SC.

The concept of SCM is well established in the literature and is defined as 'a set of three or more entities (organisations or individuals) directly involved in the upstream and downstream flows of products, services, finances, and/or information from a source to a customer' (Mentzer et al., 2001, p.3). Though, the SCM concept lack consensus, since there exists 173 definitions on such a concept (Stock & Boyer, 2009), which is impinging that SCM is a complex concept. SCM is used to achieve competitive performance vis a vis cost, quality, flexibility and time performance (Daugherty et al., 1998; Lambert et al., 1998; Ketchen & Hult, 2007; Swafford et al., 2008; Flynn et al., 2010).

To achieve SCM, there is the need of SCI both within and outside the focal firm (Ketchen & Hult 2007; Swink et al., 2007; Ketchen, et al., 2008; Kim, 2009). The SCI concept, from a simplistic definition, is the linking of suppliers, manufacturers and customers (Calantone et al., 2002). From a more detailed perspective, SCI is defined as 'the degree to which a manufacturer strategically collaborates with its supply chain partners and collaboratively manages intra- and inter-organization processes' (Flynn et al., 2010, p.59). The SCI effectiveness is dependent on the context of the supply chain (Gimenez et al., 2012).

2.5 The foundational themes within the supply chain integration literature

2.5.1 The manufacturing supply chain theoretical outline

In today's industry, manufacturing is being based on lean, as an enhancement of mass production, and agile manufacturing (Yusuf & Adeleyei, 2002; Gunasekaran et al., 2008) that implements just in time (JIT) practices. The overall SC uses a pull-based system (Kros et al., 2006) to align the production and business processes throughout the SC to the changing demands of the chain's ultimate customers (Green & Inman, 2005). Pull strategies are customer focused and are employing a JIT-II-selling strategy tactics that integrates fully empowered sales representatives within the customer's purchasing process (Green & Inman, 2007). The industry, in the current economic situation, is consumption and not production limited (Pinto, 2009). The principle behind the pull strategy is sustained by Lambert (2008, p.32), where he declared that the 'supply chain starts with your customer network and flows

back through to your supplier network, and all the functions of the business need to be involved in managing the combined network'. Yusuf and Adeleyei (2002, p.4545) in an empirical paper in lean and agile manufacturing in UK, argue that 'integration is the means by which a company can deliver concurrently on a wide range of competitive objectives' and 'competing simultaneously on multiple competitive capabilities enhance performance better than a rather narrow focus on cost and quality'. In the new manufacturing paradigm, mass-produced components are distributed widely to dispersed factories around the globe that constitute a part of the supply chain according to the local markets key characteristics (e.g. cheap labour, labour skills set and proximity to markets). This is leading to the importance of the SC virtual enterprise (VE) environment.

From the above theory, VEs are an alliance of independent business processes or enterprises with each contributing 'core competencies' in areas such as design, manufacturing and distribution to the SC network (Gunasekaran, 1999; Rupp & Ristic, 2000; Kovacs et al., 2003). In manufacturing, lean and agile SCs constitute the competitive environment based on pull-SCs located in a global network forming up the VE to deliver the product faster and at a minimum total cost (Gunasekaran et al., 2008).

2.5.2 Global virtual networks within the manufacturing supply chain

Recent literature suggests that the process of fragmentation and global dispersion of firms' value chains has entered into a new paradigm shift based on a network of inter- and intra-firm relationships. Initially it has been advocated that all types of activities are potential subject to a shift to locations with the appropriate technological competencies, skills and knowledge to achieve the routine transactional tasks (e.g. Goldman et al., 1995; Adler et al., 1995). However, in retrospect, this shift is also applicable to more knowledge-intensive and proprietary tasks, including research and development (R&D) and innovation (Lewin & Couto, 2007; Kennedy & Sharma, 2009).

Such a change consists from a paradigm shift based on vertical integration, consisting of a wellintegrated single firm with an intra perspective, to an enterprise management approach, involving effective collaboration of several SC actors (Fine, 1988; Boardman and Clegg, 2001; Davis & Spekman, 2004). The enterprise management perspective outlines that the collaborative strategies may consist from a selection of either a vertical integration (VIE) or an extended (EE) or a virtual (VE) or a defunct (DE) enterprise respectively (Binder & Clegg, 2006, 2007; Clegg et al., 2012; Clegg & Wan, 2013). As a result the SC set-up may either deploy VIE or else a mix of both a EE or VE or DE collaborative relationships with various SC actors in its inter-organisational approach, according to the Collaborative Enterprise Governance (CEG) concept, which outlines the continuum of the operations strategy, in line with the Dynamic Enterprise Reference Grid (DERG), to achieve competitiveness (Boardman & Clegg, 2001; Binder & Clegg, 2006, 2007; Clegg et al., 2012).

The enterprise term has been defined to cover all entities that through partnerships or associations are engaged in economic activities (European Commission, 2003). The extended enterprise is based on SC set-ups where SC actors work closely together to consolidate their operations and add value in their value stream processes (Davis & Spekman, 2004; Binder & Clegg, 2006, 2007; Clegg et al., 2012; Clegg & Wan, 2013). The virtual enterprise refers to SC actors which exploit all collaborative relationships which are more focused on temporary and highly agile relationships by being proactive and innovative to meet such added value (Binder & Clegg, 2006, 2007; Clegg et al., 2012; Clegg & Wan, 2013). The defunct enterprise refers to no engagement between SC actors at all (Ibid).

Starr (1984, p.22) argues that 'global networking is crucial for gaining the competitive edge' within manufacturing. Today, the competitive edge is based on how companies are capable to organize their FMS factories and discrete value-added activities on the global scale, which results in the emergence of global manufacturing virtual networks (Shi & Gregory, 2005). Such networks are made up from the production plant and other firms which build up the SC under a contractual collaboration. The SC tiers, forming the virtual manufacturing organisations' environment, consist of manufacturers and service providers to provide a holistic approach (Shi & Gregory, 2005). This new paradigm is enabling the manufacturing environment to incorporate a more integrated solution-based service, where firms within the SC act as 'system integrators' to meet the needs of each other (Shi & Gregory, 2005). Such system integrators also meet the customers' diverse and fragmented demands.

From the above theory there is the need to adopt a holistic SCM approach in today's competitive scenarios, based on a global manufacturing virtual environment based on a number of SC actors as service providers to each other. Such an approach is needed both to meet the demanding industry requirements and the customers' demands, which needs to be based on all entities capabilities and core competences, ranging from repetitive manual operations to more knowledge intensive tasks, to achieve a holistic collaborative approach so as to promote a win-win mindset.

2.5.3 The High Value Manufacturing supply chain

An extended definition of manufacturing has been developed by the Institute for Manufacturing (Livesey, 2006) which equates manufacturing to the full cycle of activities from research and development, through design, production, logistics and services, sales and marketing and to end of life management, within an economic, environmental and social context. This view is based on the

approach that treats organizations as open systems with various management roles that in order to be successful have to interact with their environment.

A key distinction that this extended definition makes is that manufacturing and production are not the same. Production is but one activity of a manufacturing company. It may or may not be the defining activity of the company. For example, if a company has outsourced all of its production activities and is focusing on brand management or R&D, that company can still be a manufacturer. This definition is in-line with the new VE manufacturing paradigm.

High Value Manufacturing (HVM) is a relatively new concept that encourages the manufacturing sector to think about whole businesses and to consider a wider range of aspects, such as time-to-market, societal and environmental issues and a global strategy at a new level of focus (Livesey, 2006).

It has been defined that 'High value manufacturing is the application of leading-edge technical knowledge and expertise for the creation of products, production processes and associated services, which have strong potential to bring sustainable growth and high economic value' (Technology Strategy Board, 2012, p.10).

Such a HVM definition drives the need to establish business models to have superior value by managing the 'fragmented value chains including distributed manufacturing to support HVM' (Technology Strategy Board, 2012, p.5).

The use of SCM such as 'supply networks and value chains' are enablers of HVM and are a 'strategic and operational manufacturing management' (Technology Strategy Board, 2012, p.21). This leads to the importance of managing effectively the supply chain to act a sound platform for optimizing all manufacturing processes across all SC tiers up to the end customer.

This perspective leads to today's current theory that the manufacturing SC must be managed, through integration, collaboration and coordination as a single entity (Chan & Qi, 2003; Sanders, 2007). This enables that all players within the SC are to be linked to deliver the products at the right place and at the right time in the most efficient method (Copacino, 1997). Logistics is that part of the supply chain process that monitors and implements the flow of goods and information from the point-of-origin to the point-of-consumption in order to meet customers' requirements' (Bowersox, 1997, p.124). Logistics enables the integration and management of key business processes across the supply chain (Cooper et al., 1997) to remain competitive (Ragatz et al., 1997; Giunipero et al., 2005). Intermodal platforms requires efficient and flexible integration with the right consideration of technical, legal and

commercial aspects and an integrated management throughout the SC to decrease logistics costs and improve on the service levels (Cambra-Fierro & Ruiz-Benite, 2009).

Most discussions about value-added, in the context of HVM, assume that value is added at each stage of a SC from raw materials extraction, production, assembly and sales and services to the end customer, with a build-up on the relationships between companies, customers and suppliers. Such a value may take two forms along the SC: use (i.e. subjective value based on all parties in the SC) and exchange value (i.e. transaction value between parties) (Bowman & Ambrosini, 2000).

Value is also seen as a function of all stakeholders both within the micro and macro perspective and may take basically three types: financial, strategic and social value (Livesey, 2006). Such value types capture all stakeholders' interests, which includes financial indicators, strategic competitiveness and all sustainable and corporate social responsibilities respectively. The theory of value extends quite broadly from a concept of price of a good to abstract units of measurement, that bring together a variety of fields, such as manufacturing, distribution, operations management, marketing, strategy, location selection, finance, energy efficiency and so on (Livesey, 2006). From a marketing perspective value is considered a driver of customer satisfaction (Woodruff, 1997).

From the above theory, such a high value-added manufacturing approach throughout all the SC tiers, includes a holistic baggage of challenges based on the full cycle of activities, which is in-line with today's HVM based SC and is needed to remain competitive.

2.5.4 The manufacturing SCM challenging perspective

SCM important role, within the industry, has been described by Lambert and Cooper (2000, p.65) by the stating that 'one of the most significant paradigm shifts of modern business management is that individual businesses no longer compete as solely autonomous entities or brands, but rather as supply chains'. This statement is supported by various scholars such as Dyer (1994); Chopra and Meindl, (2001); Davis and Spekman, 2004; Fawcett and Magnan (2004); Patnayakuni et al. (2006); Binder and Clegg (2006, 2007); and Clegg and Wan (2013). Such a paradigm shift is also supported by Kovacs et al. (2003, p.166), where they stated that: 'Supply-chain management is one of the leading business process re-engineering, cost-saving and revenue enhancement strategies in use today'. This leads to the fact that the management of manufacturing is more complex since it has to deal with all the SCM roles (e.g. distribution, production and supply) and not only on the focal firm resources (Dyer & Singh, 1998).

The SC consists of both upstream and downstream tiers around the focal firm. The upstream tiers refer to all entities that supply the focal firm, such as suppliers, whilst the downstream tiers refer to all

customers, distributors and other firms that are supplied by the focal firm. The importance of the downstream SC tiers is clearly demonstrated since 'Executives would want to manage their supply chains to the point of consumption, because whoever has the relationship with the end user has the power in the supply chain' (Lambert & Cooper, 2000, p.68).

From the above theory, SCM from a manufacturing perspective outlines that 'a one size fits all solution' does not exist to any group of SC actors forming a SC (Childerhouse & Towill, 2000, p.338), due to the various modes of processes to be managed within different contexts. SCM needs a complex management approach of all SC actors' processes forming up the SC network, so as to achieve the coordination of all resources and the optimisation of all activities across the SC to obtain competitive advantages (Gunasekaran et al., 2008).

2.5.5 Supply chain integration within SCM

SCI originates from the Value Chain Model (Porter, 1985) with linkages between the firm's value chain (horizontal integration) and that of suppliers and customers (vertical integration) to achieve superior performance such as low cost and high speed (Frohlich & Westbrook, 2001). Supply chain management (SCM) is defined as the integration of materials and information based activities through improved supply chain relationships, to achieve a sustainable competitive advantage (Handfield & Nicols, 1999). Early SCI efforts ended up to a vertical integration approach, but nowadays, with the advent of a more competitive environment, the SCs are being set-up on a relationship between different SC entities to promote greater flexibility, but without ownership of the SC entity, but through dedicated collaborative arrangements, such as partnerships with contractual obligations (Stonebraker & Afifi, 2004). The term partnership was defined by Ellram (1995, p.41) as 'an ongoing relationship between two organisations which involves a commitment over an extended time period, and a mutual sharing of risks and rewards of the relationship'. Koh et al. (2007) reinforces the role of the SCI concept, by the declaration that successful organizations must now integrate and collaborate with supply chain partners to maintain competitive advantage.

Power (2005) classified SCM practices under three perspectives, namely SCI, strategy and planning, and implementation, where:

(a) SCI covers issues relating to integration of core processes across organizational boundaries through improved communication, partnerships, alliances and cooperation which may lead to removal of barriers between firms, better decision making, more efficiencies between firms, better SC visibility, identification of bottlenecks, more trust and commitment between firms, increased outsourcing by focal firm, cycle time reduction, reduced inventory and shorter lead times due to real time

communications based on IT, which all lead to reduce the operating costs and improvement in the customer service level.

(b) Strategy and planning examines supply chain management as a strategic matter for trading partners, along with factors relating to planning activities. The strategic approach for a manufacturing firm within the SC should be based on collaborative partnerships with suppliers, strong and direct relationships with customers and the coordination of disparate supply chain elements, without undergoing vertical integration, referred as virtual integration.

(c) Implementation issues concern factors critical for successful implementation, as well as issues specific to inter- and intra-organizational aspects of SC initiatives, such as staff involvement in the decision making, suppliers' involvement in the productivity measures, business processes optimisation and adoption of performance measurements (e.g. SCOR).

The virtual integration approach, referred by Power (2005) above, under the strategy and planning perspective, constitutes the SC paradigm shift from the traditional vertical integration approach to an innovative virtual integration approach. Within the SCM literature, vertical integration (forward and/or backwards) approach is a well established fact that any business takes to improve the performance of the firm through ownership of either upstream and/or downstream SC activities (Guan & Rehme, 2012). Nowadays the theory is focussing for the SC to deploy a virtual enterprise approach to optimize a global supply chain (Gunasekaran et al., 2008). This is in-line with the claim by Prahalad et al. (1994) where they stated that vertical integration is being replaced by virtual integration, where each firm within the SC concentrates on the processes that it performs best, leaving the rest to others. This approach is also substantiated by Magretta and Dell (1998, p.74) by stating that 'virtual integration [is a] means [where] you basically stitch together a business with partners that are treated as if they're inside the company'. Though, it is of pivotal importance that the entire process within the SC tiers is to be 'designed, managed and coordinated as a unit' (Bagchi et al., 2005, p.276). Furthermore, the enterprise management perspective, is also in line with the above, and outlines the various collaborative strategies based on vertical integration (VIE) or an extended (EE) or a virtual (VE) or a defunct (DE) enterprise respectively (Binder & Clegg, 2006, 2007; Clegg et al., 2012; Clegg & Wan, 2013), as referred in Section 2.5.2.

From the above theory it is deduced that SCI with its virtual integration approach and its value-added activities, forms up a VE made up from all the SC members which needs to perform effectively as a unit, to promote success and maintain competitive advantage, since the integrated management approach can reduce the negative effects associated with any of the SC members shortcomings (Simchi-Levi et al., 2000; Guillen, et al., 2007).

2.5.6 Supply chain integration strategic review

It is well established in the literature that the process by which management of a business evaluates the future opportunities and decides on any strategic action, in the long or short term, is referred as strategic planning. The objective of an integrated SC strategy is to synchronize the requirements downstream throughout all the SC tiers to reach a balance between high customer service and cost (Stevens, 1989).

The strategic perspective and importance of integrating suppliers, manufacturers and customers within the SC has been cited by various authors (e.g. Reck & Long, 1988; Power, 2005; Giunipero et al., 2005). Strategic action plans need to be generated based on the operations strategy specifications and then building on these plans, to design a set of performance measurements (PMs) across the SC tiers. The PMs are to be used to build a SC strategic performance management system to promote a learning and a continuous improvement approach. According to Neely et al. (2005), PMs are a set of metrics to quantify the efficiency and effectiveness of all actions. In view of the high levels of competition, the integration of all the business operations' activities and agents within the supply chain has become a strategic concern (Cambra-Fierro & Ruiz-Benite, 2009).

From the above theory a SC undertakes strategic planning and SCI as a strategic approach to synchronise all actions across the SC. Such a strategic approach with the right deployment of PMs, enable a SC group to quantify efficiency and effectiveness.

2.5.7 Supply chain integration theoretical underpinning foundations

The term 'integration' is defined as the 'act or process of combining two or more things so that they work together' (Oxford Dictionary, 2010, p.807) or as 'the unified control of a number of successive or similar economic or especially industrial processes formerly carried on independently' (Webster's 1966, p.1175). Integration was also defined as the 'quality of the state of collaboration that exists among departments that are required to achieve unity of effort by the demands of the environment' (Lawrence & Lorsch, 1967). Such collaborative actions depend on the personal and cultural differences from one firm to another, since these determine the organisational climates in all interactions (Lawrence & Lorsch, 1967; McKinnon et al., 2003).

The SCI concept is widely deployed in practice and has been defined to have two underpinning variables: the degree and the direction of the shared activities between two or more actors (Frohlich & Westbrook, 2001). The degree has been defined by the extent (i.e. depth) of the integration, which has been further denoted by the term 'arc' of integration, with a continuum of narrow to broad state, to

indicate the integration magnitude (Ibid). The direction has been defined on whether the integration is in the customer (i.e. downstream) or supplier side (i.e. upstream) or both respectively (Ibid).

Within the SC, the need for an integrated relationship between manufacturers and their supply chain partners is a well conceived requirement (Lambert et al., 1978; Hayes & Wheelwright, 1984; Watts et al., 1992). Empirical evidence shows that by integrating-specific internal supply chain functions, such as manufacturing and purchasing, will also lead to higher performance (Narasimhan et al., 2001; Stank et al., 2001; Frohlich & Westbrook, 2001; Flynn et al., 2010). In view of the above and from the extant literature, internal integration is a necessary step of the SCI process (Stevens, 1989; Rosenzweig et al., 2003; Vickery et al., 2003; Flynn et al., 2010). This is also substantiated by Pagell (2004), where it was advocated that a well-managed SC is then an integrated SC, since lack of integration indicates processes working at cross-purposes to reduce performance or else cases may exist when integration may be inhibited, if competition for resources among processes occurs within the organisation, that were ultimately serving the same end customer. Such an internal integration between different functional areas requires inclusive communication, joint accountability, strong working relationships and senior management involvement (Ellinger et al., 2006). The scope of the integration is to move from a traditional 'functional silos' approach to a more coordinated or concurrent approach, with both internal functional areas and external SC tiers working together (Pagell, 2004).

Integration has become a new core competence (Narasimhan et al., 2001), it increases value (Porter, 1985; Stank et al., 2001), and create the lowest total landed cost without sacrificing superior service (Stank et al., 2001), but requires separate parties to work together in a cooperative manner to arrive at mutually acceptable outcomes (O'Leary-Kelly & Flores, 2002), referred as a win-win situation.

Effective SCI involves mutual understanding, a common vision, shared resources, investments in the relationship and achievement of collective goals and integrated business processes which includes boundary-spanning activities by drawing suppliers and customers into the value creation process (Kahn et al., 1996; Tan et al., 1998) together with the focal firm, as the core of the integrated activities. Such an integration creates interweaved processes that cannot be easy replicated and as a result creates a competitive capabilities (Daugherty et al., 1998; Lambert et al., 1998; Ketchen & Hult, 2007; Swafford et al., 2008; Flynn et al., 2010).

Other researchers go on to argue that firms need to invest in their supply chain business processes and not only in information technology to perform effectively (Heinrich & Simchi-Levi, 2005). The SC must have integrated: the flow of information (Lee et al., 1997); physical materials (Stevens, 1990); and financial information (Mabert & Venkatraman, 1998) with its value chain trading partners. Such an action should promote less variability, better forecasts, better lead times and an improved customer
service with information flow (Simchi-Levi et al., 2004), promote cost savings and improvements with improved products flow (Malhotra et al., 2005) and better cash flow through workflow systems for monitoring all financial information (Malhotra et al., 2005).

SCI is considered to consist of a comprehensive collaboration among supply chain network members (Bagchi et al., 2005) and also as a source of leverage of resources and knowledge from suppliers (Fawcett & Magnan, 2004). The collaboration concept is defined as an 'interactive, constructive and knowledge-based process' leading to synergy (Hartono & Holsapple, 2004). High interactivity leads to innovation within the three areas of collaboration: 'across departments, within a department and inter-firm networks' (Maccoby, 2006). Literature also includes that collaboration facilitates streamlining of operations across the supply chain by 'replacing inventory and excess capacity with accurate information' (Taylor, 2004). According to Ajmera & Cook (2009), the drivers of collaboration are: responsiveness to variability, resource sharing, cost, and joint decision-making.

From the above theory, SCI is about collaborating firms who achieve economies of specialization through the SC, by exploiting the core competencies of all SC actors to optimize operations through a VE approach (Dyer, 1994; Cox, 1999; Lambert & Cooper, 2000; Fawcett & Magnan, 2004; Bagchi et al., 2005; Ketchen & Hult, 2007; Koufteros et al., 2007; Gunasekaran et al., 2008) instead of working alone. Schoenherr & Swink (2012, p.110) defined 'a theory of integration, i.e. a confirmation of the significant benefits that firms can realize by being strategically interconnected and aligned with their supply chain partners'. SCI results in streamlined processes and leverage of resources from all SC actors to promote higher performance and competitive capabilities (Pagell, 2004).

2.5.8 Technology deployment: SC information systems (ISs) and manufacturing processes

To date studies have suggested that high levels of coordination between organisations, which are necessary for improved supply chain performance, demand high levels of integration between partners' information systems (White et al., 2005). Such information sharing facilitates joint-decisions synchronisation across the SC to improve on the business performance (Simatupang et al., 2004).

Nowadays, IT is transforming the industry and quite recently, with the advent of innovative and diverse technological based applications, it has become feasible for firms to have the access to information that is accurate, timely and affordable (Bowersox & Calantone, 1998). The internetenabled supply chains have become powerful strategic weapons due to their unparalleled integration of information among partners at relatively low transaction cost (Sanders, 2007), which also enables organizations to integrate different legacy systems (Muffatto & Payaro, 2004). Nowadays the e-mail, containing text and cad files, trading documents, etc, has streamlined the SC operations, bridging the constraints associated with time and distance. Christopher (2000, p.38) substantiates the use of IT, by stating that it is 'creating a virtual supply chain. Virtual supply chains are information-based rather than inventory-based'. This argument is reinforced by Dell (1999, p.33) where he outlined that the 'new eBusiness technologies facilitate quick information sharing between downstream and upstream partners and enable companies like Dell Computer to trade inventory for information'.

Such an IT platform facilitates and meets effectively the SCI virtual enterprise approach (Gunasekaran et al., 2008). IT promotes value-chain transparency by allowing companies to reposition themselves in the chain and dynamically collaborate with companies to optimise their business position (Shi & Gregory, 2005; Cassivi, 2006). The contribution of IT within manufacturing extends beyond the role of providing an information systems, since it directly assists people in performing certain tasks or replaces them by doing the tasks that they once performed, through the use of advanced manufacturing technologies, through the use of productive tools (Gunasekaran & McGaughey, 2002), such as CAD, CAM, CIM and FMS. Though it is to be kept in mind that 'Collaboration is a result of human interactions which can only be supported by IT, one of which are e-business technologies, but not replaced' (Sanders 2007, p.1343).

White et al. (2005, p.397) state that 'the agility paradigm is the ability to form deeply integrated links with a wide range of trading partners and be able to quickly dissolve these and reform such deep linkages with new partners as required by changing market conditions'. Such an agility paradigm is enabled by agile information systems (White et al., 2005). The agility in SC can be defined as 'a business-wide capability that embraces organisational structures, information systems, logistics processes and, in particular, mindsets' (Christopher & Towill, 2002). Such SC agility, within a manufacturing context, is also substantiated by Gunasekaran et al. (2008) which is said to improve on lead times and cut on costs across all SC activities.

Inter-organisational systems (IOS) are used to transcends organisational boundaries to communicate with all SC actors, by deploying various IT applications, such as electronic data interchange (EDI), electronic funds transfer (EFT), ISs and SCM systems (White et al., 2005). IT, through dedicated ISs, is making such an enterprise management a technical reality to promote a competitive edge (Rodon et al., 2011; Clegg & Wan, 2013). The traditional ERP, referred as ERP I is undergoing a paradigm shift from a single organisation management approach to an inter-organisational collaborative environment based on e-commerce and web base technologies, referred as ERP II (Eckartz et al., 2009; Clegg & Wan, 2013). Such technology deployment (e.g. EAI, SOA and SaaS) is also supporting agile manufacturing (Banker et al., 2010).

One cannot exclude to mention the 'IT paradox' (Brynjolfsson & Yang, 1996) or 'productivity paradox' which still exists within the people's mind (Lim et al., 2004; Sriram & Stump, 2004).

Numerous explanations have been offered for this lack-of-productivity paradox (Sanders, 2007), since IT does not only always meet the expected benefits (Stratopoulos & Dehning, 2000).

From the above theory, ISs and IT applications are enabling information integration across the global markets and are also improving on manufacturing processes efficiencies and effectiveness. IT enables a virtual SCI environment, with the possibility of deeply integrated links, such as web based IOS. Such technology deployment promotes also SC agility in line with agile manufacturing, to meet the competitive market conditions and improved performance, but still does not replace the role of the human element.

2.5.9 Supply chain (SC) integration literature gaps

A coherent view of SCI and its measures has yet to be developed due to the different definitions between various scholars (Croom et al., 2000; Palomero & Chalmeta, 2012). SCI within the literature is rare (Pagell, 2004). Limited research has been conducted beyond the dyadic relationship or in parts of SCs containing smaller organisations (Cachon et al., 2007; Harland et al., 2007). SCI is still in its infancy (Harland et al., 2007). Research is quite limited to explain how to achieve SCI (Zhao et al., 2008) and how to determine the influencing factors of SCI (Flynn et al., 2010). Integration so far was associated with the operational processes without integrating the planning and design, so the returns have been limited (Ajmera & Cook, 2009). The development of models and approaches for an integrated SCM system will be a major step towards the optimization of SC priorities (Gunasekaran, 2004). Performance measurement is challenging because it is difficult to attribute performance results to one particular entity within the chain, both when the SC tiers are locally and globally (Hervani et al., 2005). There is a lack of investigation on understanding the SC measures which relates the objectives and motivations of various SC entities for the overall maximization of SC profit (Thakkar et al., 2009). Many conceptualizations of SCI are incomplete, leaving out the important central link of internal integration and also lacks a taxonomy of SCI that incorporates internal integration, as well as customer and supplier integration (Flynn et al., 2010). A holistic approach with a set of measures covering all SC actors (Holmberg, 2000) and how such processes can be coordinated within the focal firm and across the SC (Mentzer et al., 2001) are both lacking within the extant literature. This perspective is also substantiated by (Vickery et al., 2003, p.529), where they declared that 'the literature lacks a holistic theoretical framework' on SCI.

2.5.10 Information technology literature gaps within the supply chain context

Some firms have successfully integrated eBusiness technologies into their traditional bricks-andmortar business models, but others still struggle to implement and justify eBusiness initiatives (Devaraj et al., 2007). Our knowledge is especially weak concerning modern internet-based integration in all sectors (Bowersox et al., 2000; Frohlich & Westbrook, 2002). There is lack of empirical evidence on eBusiness technologies impact on performance within the context of SC (Devaraj & Kohli, 2003). To date there has been little empirical research within connected supply chains containing SMEs exploring larger and smaller firms' perceptions of the value of supply chain information integration (Harland et al., 2007). The payoff on SCM from IT investment has been referred as the 'IT paradox' or 'productivity paradox' (Freeman & Soete, 1990; Brynjolfsson et al., 1996; Lim et al., 2004).

2.5.11 The tentative research question

There is an extensive research on SCI and as a result this literature review has been focused on various parts of the SC since the researcher was not in a position to walk through all the diverse definitions and theories of SCI, due to the complexity and broadness of the discipline. Furthermore, this literature review is also supported by an overview of the **literature gaps** from a SCI and IT perspective (Sections 2.5.9 & 2.5.10) and also from the **systematic evaluation** of set of papers from the extant literature in a tabulated format, as referred in Appendix 2.

As a result the literature gaps outline that the extant literature lacks a comprehensive study on the key concepts that establish the intra- and inter-holistic SCI management approach. The evaluation of the set of papers (Appendix 2) provides a wide spectrum of the literature from which it is identified key four significant concepts building up SCI within the manufacturing sector that promote SC competitive capabilities. The identified four concepts are: information integration, strategic integration, strategic collaboration and information technology, as the independent variables, together with operational and financial performance measures, as the dependent variables that constitute the SC competitive capabilities. It is to be noted that these significant concepts are defined by several dimensions by undertaking a content analysis of all the studies, as referred in Appendix 2, so as to promote an in-depth level of understanding of SCI to assist the researcher to build on his theoretical sensitivity on the phenomenon in line with the GTM (Chapter 3).

As a result the outcome from such a literature review enabled the researcher to achieve two objectives:

(a) The formulation of the research question.

The research question that guides the study and supporting objectives is the following two-part question. First, which are the characteristics needed to achieve SCI within SMEs in the Maltese manufacturing? Second, which are the key factors that constitute SCI that promotes competitive performance within SMEs in the Maltese manufacturing?

(b) The design of the first open ended interview guide, as referred in Appendix 3.

2.5.12 Summary of the a priori literature review

The manufacturing SCI strategy should capture the internal integrated firm, together with the customers' and suppliers' external to the focal firm, beyond the dyadic relationships. Within the manufacturing sector, it is a well conceived requirement, that nowadays, to stay in business there is the need for an integrated strategic relationship between various firms and not to operate as autonomous entities (Dyer, 1994; Cox, 1999; Lambert & Cooper, 2000; Fawcett & Magnan, 2004; Bagchi et al., 2005; Ketchen & Hult, 2007). An integrated SCM operations strategy, design and implementation are defined by the term SCI. The extant literature from a cross disciplinary approach, based mainly on SCM, SCI and IT within the manufacturing context, reveals that SCI is depends on internal integrated efforts within each firm (i.e. internal integration) and all integrated activities across all firms (i.e. external integration) to leverage resources and knowledge (Fawcett & Magnan, 2004; Flynn et al., 2010). SCI ideally breaks down all functional 'silos' within and across the firms (Pagell, 2004; Stonebraker & Afifi, 2004).

A manufacturing SC is not about production but covers all the value-chain activities of the focal firm and all other firms building up the suppliers and customers' networks (e.g. OEMs, sub-contractors, third-parties, distributors and end-customers) within a context to meet the sustainability challenges (Livesey, 2006; Bayraktar et al., 2009). The new emerging manufacturing paradigm, is being referred to as HVM and is being based on distributed firms performing a full cycle of activities, such as R & D, innovation, production, distribution, advertising and customer support, around the globe, leading to the virtual enterprise (VE) concept (Livesey, 2006; Martinez et al., 2008). This management paradigm shift incorporates a change from vertical integration to virtual integration (Prahalad et al., 1994; Power, 2005; Gunasekaran et al., 2008), based on unity of efforts, to optimize the use of all core and distributed resources to form a single holistic SC. The manufacturing sector is also focussing on an ever changing improved automation and efficiencies, to produce with lower amount of people and lower costs, in-line with the globalised economy requirements (Droge et al., 2004; Gunasekaran, 2004). This new industry environment is being driven by a context having dynamic customers' needs with various other competitive SCs and a scenario based on a set of challenges, such as social, environmental, economic and legal responsibilities, regulations and standards compliance and stringent and turbulent economic conditions.

Such a new paradigm requires organisations to adopt a more decentralised management approach to focus on the individual competences and outsource others (Sink & Langley, 1997). At the same time

all firms need to employ all strategic efforts to collaborate with all SC members, to achieve the pull SC strategy approach, originated by customers in line with the pull-SC management approach (Kros et al., 2006; Green & Inman, 2007; Lambert, 2008). Such a strategic stance is attributed to lean and agile manufacturing, which promotes agility, flexibility, adaptability and responsiveness in all operations but also maintains cost effective solutions and quality (Yusuf et al., 1999; Sanchez & Nagi 2001; Yusuf & Adeleyei, 2002; Sherehiy et al., 2007; Gunasekaran et al., 2008). Though, it is to be noted that the shift to lean and agile manufacturing is primarily attributed to a cumulative build-up of the individual resources of all the firms, within the intra- and the inter-enterprise (Binder & Clegg, 2006, 2007; Clegg et al., 2012), to focus on improvement mechanisms, not only to eliminate waste, but also to promote flexibility and responsiveness across the SC (Gunasekaran et al., 2008; Clegg & Wan, 2013).

One of the key drivers of this change associated with lean and agile manufacturing is technology based on the internet platform to promote open technology based systems (Metcalfe et al., 2002; Muffatto & Payaro, 2004; White et al., 2005; Sanders, 2007). Such a change is promoting an IT integrated solution based service approach, to build and integrate the current and various legacy information systems, through systems integration, which is being deployed by each member within the SC. The technology driver is enabling the possibility for a high level of integrated information systems between all SC tiers to enhance all collaborative efforts (e.g. CPFR and VMI) (Maccoby, 2006; Baratt, 2004). This is also leading to information integration based on a high SC visibility and transparency (e.g. to minimize the negative effects from the bull-whip effect) and to meet the demanding trend of the agile and lean manufacturing concepts, by replacing inventory for information (Kearns & Lederer, 2003; Wang & Wei, 2007; Gunasekaran et al., 2008).

This level of integration within and across the SC is promoting effective and efficient operations with synchronised and real-time joint decision-making (Bagchi et al., 2005; Ajmera & Cook, 2009; Kim, 2009). The investment in IT based systems by all SC actors are sustaining the information-based virtual SC environment, which is in-line with the VE manufacturing environment (Christopher, 2000; Wang & Wei, 2007; Gunasekaran et al., 2008) together with the use of advanced manufacturing technologies (CAD, CAM, CIM and FMS) (Droge et al., 2004; Gunasekaran, 2004). It is a well conceived fact that IT, as a stand-alone resource does not create the edge in operations, but the mix of the components parts of the organisation human competences and all processes with IT, is what dictates IT value (Sanders, 2007). All these synergies between SCI and IT deployment, within the manufacturing context, are leading to the value-added concept, which is a cumulative build-up by each SC member, through strategic, financial and social value (Shi & Gregory, 2005; Livesey, 2006). To meet such a design and development of new products and services within the VE manufacturing

environment, needs a technological based infrastructure, such as enterprise resource planning systems (ERPs), for leveraging organizational competitiveness. The ERP centralised approach, which is focused on the individual firm, needs to be transformed through a distributed approach (Rupp et al., 2000; Kovacs et al., 2003; Clegg &Wan 2013), to enable the optimisation of the globalised SC, by adopting integrated based application software, based on a cloud-based SC environment (Metcalfe et al., 2002) or else on an enterprise application integration (EAI) systems to have interoperability between the multiple disparate systems (Loh et al., 2006). Interconnected ERPs through the VE environment using IT, would enable leaner production (Newell et al., 2003) and provide information visibility and supply chain flexibility, being the major advantages provided by IT support to achieve SCI (Wang & Wei, 2007; Gunasekaran et al., 2008; Clegg &Wan 2013).

2.6 The post literature review associated with the emerged theory

The post literature review is needed since the *a priori* literature review is based on a broad perspective on SCI theory within the extant literature and not focused on the emerged theory from the data. The role of the post literature review is used as a literature based platform for Chapter 7, since the latter chapter is used to integrate the emerged theory within the extant literature, as referred in Section 3.9. Though, it cannot be excluded that reference is also made to sources within the a priori literature review, so that Chapter 7 can fulfil its objective with a more comprehensive approach.

The core emerged theory from the substantive area, is based on a core theme and all of its five main themes and the set of sixteen sub-themes, as referred in Table 5.1. The post literature review objective is to systematically review the literature from the perspective of the emerged theory, by selecting the relevant scholarly work associated with such emerged themes, in a theme-by-theme approach, as referred in Section 2.6.2 below. To make this chapter more manageable, all the themes are included within this chapter but the other sixteen sub-themes, are referred in the Appendix 4. Finally, it is also established the literature gaps, focused on these four themes based on the extant literature: leadership; supply chain integration; sustainability; and manufacturing respectively (Section 2.6.10).

2.6.1 The outline of the post systematic literature review

The post literature review keywords used for the systematic literature review are based on the emerged themes. The main keywords are 'supply chain' AND 'integration', which was logically operated with a dedicated sequence of searches with an 'AND' logic with technology, manufacturing, leadership, strategy, management and holistic terms respectively. Further on, within such a systematic literature review, further keywords are used to cover all sixteen sub-themes as emerged under the holistic SCI

management approach, obtained from Section 6.3.6 and listed in Section 2.6.2 below. To ensure that the literature review covers a substantial number of papers which are relevant to the emerged theory, other keywords that emerged from this literature review itself, such as 'partnership'; 'power'; 'dominant players', are also included in the search, under the same logic referred above, to enrich the scholarly work on the SCI phenomenon. Throughout the systematic literature review, attention is also given to the papers ranking quality in area of OM and SCM, as referred in the Academic Journal Quality Guide by Harvey et al. (2010), to promote a high level of research quality.

Initially, this literature review includes a cursory analytic approach on a substantial number of papers that are found in the search, so as to classify their relevance, after going through their abstracts and contents to confirm their suitability. A systematic content analysis is then performed on a number of such selected papers to derive a rigorous review of the literature, which is initially broad but thorough and becomes more focused, to be relevant to the emerged theory. Further papers need to be selected during the papers' content analysis, from a substantial number of selected papers, due to their relative importance to the research focus, so as to enhance the literature synthesis quality, by including all the relevant and valid original sources, such as seminal works on each respective theme. Furthermore, the literature review also includes a special focus on the latest papers that are published in 2014/15 and those to-be published in 2015 to make reference to the latest contemporary research.

2.6.2 Overview of all the themes within the emerged theory

The study emerged substantive theory, is based on the core theme, together with a set of other five (5) main themes and sixteen (16) sub-themes. All the themes emerged from the substantive theory which are used in the systematic literature review are:

The core theme: Integrative Management and Leadership Approach (IMLA).

The six Main themes: Business Strategy; Manufacturing Strategy; and Supply Chain Strategy; Holistic SCI management approach; and Technology deployment; and Competitive capabilities.

The sixteen sub-themes under the <u>holistic SCI management approach</u> are: lean management; managing quality and sustainability; managing performance measurements; auditing operations; supply management; managing cash-flow; managing change and innovation; managing collaboration; managing culture; managing knowledge; managing customer service; managing information sharing; managing logistics; planning/forecasting; managing risks and managing trust.

2.6.3 The core category: Integrative Management and Leadership Approach (IMLA)

The concepts building up the emerged core category/theme, are referred as the Integrative Management and Leadership Approach (IMLA), namely it includes the 'leadership' and 'management' conceptual elements. From the data, the integrative management role is to integrate all people through the established organisational practices according to the planned objectives through a process approach to reach the necessary performance targets, both within and outside the focal firm. Similarly, the integrative leadership role shifts to a broader approach to meet organisational objectives by transforming and inspiring people, through an effective leadership approach, so that the business becomes innovative, empowered and adopts a dynamic approach based on a culture of teamwork, so as to create outstanding performance, both within the focal firm together with all of its stakeholders. Both the above concepts, together with all the other emerged concepts, emerged from the substantive theory, as referred in Section 2.6.2 above and in Table 5.1, are now reviewed in the literature.

2.6.3.1 The management concept synthesis within the IMLA core category

2.6.3.1.1 The management timeline

Management in the 20th century period has mainly been associated with a hierarchy of managers with a formal organisation set-up. The management theory passed through several phases between concern for production and concern for people. It started with the scientific or classical management approach, whereby Taylorism and Fordism movements advocated that there exist a set-of rules and procedures under a rigid hierarchy and with a division of labour with a top-down level of control with a mechanistic structure to achieve efficient operations (Kinicki & Williams, 2003). Henry Fayol identified the administrative model based on key management functions of **planning**, organizing, leading and controlling (Kinicki & Williams, 2003). Such an approach led the way to bureaucracy (e.g. Max Weber's concepts) which was associated with a formal organisation approach with clearly defined hierarchical structures with defined responsibilities at all levels that promoted rigidity and a lack of flexibility in the management approach (Mullins, 2005). The next management phase was more focused on the human relations approach, which generated a focus on the human social and psychological aspects, such as **motivation** (e.g. Maslow's hierarchy of needs) and **empathy** for the human element to promote empowered initiatives (such as the informal organisation), with an organic structure to promote performance improvements (e.g. Elton Mayo; Mary Parker Follett, Herzberg and McGregor). Then, the system theory followed, whereby managers need to view the organisation as a whole system with a number of sub-systems and where the system is located within an external environment. Such a system theory referred that for an organisation to be a success needs to adopt an

open system approach, to demonstrate the importance that an organisation needs the interaction of both the internal and the external environments for it to be effective (Von Bertalanffy, 1969).

2.6.3.1.2 The management role

The above management theoretical timeline is very simplistic but is presented to show the basic key principles from the seminal works behind the management theory. The management theory clearly outlines the need of managers to **manage effectively and efficiently** to meet both the **financial and human element** perspectives. It cannot be excluded that the way forward to date still makes reference to such classical management frameworks.

To-date, the role of management is to achieve objectives through people with some form of an organised structure, which tends to ensure that the **human element** is more **flexible and empowered** to **adapt to changes** due to the dynamic environment. The managing change approach was referred by Kurt Lewin (1951), where he derived the theory that explains the process of change management with a three stage model: unfreezing the current process; change to a new process; and freezing at the new level. From then on, the concept of change management has been embedded in various scholarly works associated with management of operations to promote radical (transformational) or incremental change (Johnson & Scholes, 1999; Kadar et al., 2014).

The management role is to make things happen, by developing people and reaching objectives to achieve results through people (Crainer, 1998). The role of the employees and the managers is also substantiated by one of the management gurus who stated that: 'Management is tasks. Management is a discipline. But management is also people. Every achievement of management is the achievement of a manager. Every failure is a failure of a manager' (Drucker, 1979, p.14). Management is not a label to a person but it is all about the activities to be performed, the authority and the responsibility (Mullins, 2005). The manager needs to manage complex and sometimes ambiguous situations making the management discipline to be not only a science but also an art (Watson, 1986). Furthermore, Drucker (1977) outlined that the manager, apart from **meeting the objectives, productivity and social responsibilities**, must also establish measurements on the individual and the organisation to **monitor performance improvements and optimisation** (e.g. performance measures, best practices and benchmarking).

2.6.3.1.3 SCM perspective

From the perspective of the SC, Lambert et al. (1998, p.5) identified that to 'integrate and manage all process links with all members across the supply chain, in most cases, be counter- productive, if not impossible', and that it is critical to identify the SC members who shall be 'allocated managerial attention and resources'. They further added that 'integrating and managing all business process links throughout the supply chain is likely not appropriate' (Lambert et al., 1998, p.7). They also advocated that the management of the business processes are to be classified into four approaches: managed; monitored; not managed and non-members. This perspective outlines that **some processes need to be integrated and managed but others just monitored** (e.g. audited) and the other two approaches, undertakes business as required with no integration and management at all. They further postulated that the 'lack of intercompany consistency is a cause of significant friction and inefficiencies in supply chains' (Lambert et al., 1998, p.9), which justifies the **need for SCI initiatives**. Though, it was clearly outlined that a decision needs to be taken to identify which 'key business processes need to be integrated and managed' (Lambert et al., 1998, p.11).

The SCM processes, as shown in Figure 2.1, are classified into two groups: 'physical and technical management components', which are tangible, and the 'managerial and behavioural management components', which are less tangible and visible, where the latter group determines the implementation of the former group to promote competitiveness (Lambert et al., 1998, p.11). The SC members need to focus on the management for the common good of the overall SC, so as to promote holistic optimised performance, even if there is the need to sub-optimise their individual performance (Cooper et al., 1997).





In fact, there is a **SCM paradigm shift** regarding the managerial and behaviour components, referred above, from a platform based on differentiation (i.e. specialisation of functional areas), untrusting attitudes, centralized decision-making, narrow and mechanistic tasks with untrained employees, to a new platform based on integration (i.e. formalization/collaboration), systemic trust, distributed decision-making, and broader and more participative, organic structures through continuously trained employees (Stonebraker & Afifi, 2004). The role of people with the right human resource management, within the SC context, is crucial to the success of the SCM approach (Ashby et al., 2012; Ellinger & Ellinger, 2014). Such a new paradigm implies that SCM is multifaceted, dynamic and highly complex, since there are instances where integration is useful but in other situations it is to no avail and where it varies from **short term adversarial to long term relationships** (Stonebraker & Afifi, 2004). Though, it is a generally accepted notion, that there is the need for **process improvements and an effective management approach** of all the intra- and inter-organizational processes across the supply chain (Flynn et al., 2010).

The management role within the SC context is to ensure that all SC members, especially buyers and suppliers, through operational linkages need to have in place 'the systems, procedures and routines that facilitate the more effective flow of goods, services and information'(Cox et al., 2003, p.137). Furthermore, they also added that such linkages 'can extend beyond a single dyadic relationship and operate across many exchange partners within the supply network' (Cox et al., 2003, p.137).

2.6.3.1.4 The summary of the management concept from both the focal firm and SC perspectives

In summary, managing an organisation refers to a **formal organization structure**, which has been defined as: 'the planned co-ordination of the activities of a number of people for the achievement of some common, explicit purpose or goal, through division of labor and function, and through a hierarchy of authority and responsibility' (Schein, 1988, p.15). Management occurs at all levels of the organisation within the command chain and there is the need for technical and social skills and also conceptual capabilities (Katz, 1974). Within the contemporary management literature it is pivotal to have cross-collaborative initiatives between people within the formal organisation set-up and also the acceptance of the **informal organisation structure**, that further breaks down the presence of any silo between people to promote teamwork and performance improvements (Stonebraker & Afifi, 2004). From a SC management perspective, the management approach to be adopted by the focal firm needs to include all the SC members, **beyond the dyadic members** (Cox et al., 2003), through all the necessary **dedicated collaborative initiatives** depending on each SC member role, since the SC needs to be managed from a holistic approach. The roles of **planning, organizing, leading and controlling operations** remain the core elements of the management functions, but nowadays such

elements are taking different forms to meet the dynamic and complex changes together with the introduction of the ever changing technology based applications.

Managing change and innovation are both crucial in today's challenging business environment due to such dynamic scenario in the manufactured products and processes being deployed to meet the competitive market and the demanding customers' expectations with a level of **flexibility and responsiveness**. Nowadays, managing effectively is not only considered from both a **social** and an **economical perspective**, in line with the classical management approach, but also needs to cater for an **environmental perspective**, by employing environmental friendly measures and 'green operations' to promote environmental sustainability in line with the 3BL sustainability (Elkington, 2002; Bojarski et al., 2009; Buyukozkan & Berkol, 2011; Ashby et al., 2012; Gimenez et al., 2012; Lee et al., 2012; Caniels et al., 2013; Glover et al., 2014).

2.6.3.2 The leadership concept within the IMLA from the focal firm perspective

2.6.3.2.1 The leadership timeline

In the late 1960s and 1970s new emerging theories about the central role of people within the organisation led to the distinction between management and leaders. This divide is also debatable todate, but it had driven managers to accept the facts that leaders need to align, involve and motivate people, whilst managers need to plan, organize and control people (Levine & Crom, 1994; Rue & Byars, 2005). In the **industrial era**, the leadership theories focused on the **individual leader** (Gronn, 2002), based on the tripod ontology resting on three main pillars of leadership theories: leader, follower and common goals (Bennis, 2007). Northouse (2007, p.3) defined leadership as a leader-to-follower relationship based on a 'process whereby an individual influences a group of individuals to achieve a common goal'. The literature clearly advocates that effective leadership contributes to organisational performance and competitive advantage (Avolio, 1999; Bass, 1990; Yukl, 2002; Ardichvili & Manderscheid, 2008). Effective organizational leadership vis a vis the organisational performance, employees and customers, is a determining factor for the success or failure of any business (Xu & Thomas, 2011). The human element, apart from the commercial element, is a crucial factor within the global business so as to have the right people with the appropriate skills and talent (Rothwell, 2002; Tarique & Schuler, 2010; Xu & Thomas, 2011).

2.6.3.2.2 Strategic leadership/shared leadership against individual leadership

As advocated by McCauley-Smith et al. (2013, p.85;99) within the educational sector, an effective leadership approach needs a shift from an '**individual leadership**' stance based on a dualistic or

dyadic nature, to an '**integrated leadership**' based on empowerment across the 'organisation structures' and by 'sharing of good practice through an interdependent leadership approach'.

The contemporary leadership literature, within this **knowledge era** (Uhl-Bien et al., 2007), due to the current competitive environment, is driving an increased awareness for corporate intelligence (McKelvey, 2001) so as to promote learning and exploit all organisational stakeholders' contribution for performance improvement. Such a current leadership approach is ideally focused on a collective leadership or **shared leadership** approach and not based on a **single individual** (House & Aditya, 1997), which is referred as connected leadership (Martin et al., 2007). This means that the current business scenario shall be based on leadership at all management levels (O'Reilly et al., 2010). Such a strategic leadership approach is inferred to be based on absorptive capacity (i.e. ability to learn), adaptive capacity (i.e. ability to change) and managerial wisdom (i.e. ability to understand all social actors) which all contributes to both strategic leadership and organisational effectiveness (Boal & Hooijberg, 2000).

Strategic leadership, within the strategic management literature, refers to a leadership approach within the organisation that shifts beyond the dyad set-up to a relationship across different organisations (Hunt, 1991; Finkelstein & Hambrick, 1996) which is referred as a macro perspective. Whereas supervisory leadership is a sub-set of strategic leadership since it primarily focuses on the dyad relationships and on the organisation lower levels tasks and individual performance (House & Aditya, 1997), which is referred as a micro perspective. Other leadership theories, such as transformational leadership among others, have built the set of leadership portfolio who emphasize the interpersonal processes between leaders and followers (Boal & Hooijberg, 2000).

2.6.3.2.3 Transformation leadership trait

Transformational leadership drives the followers to see the deeper purpose in their work, treats them as unique and valuable human beings, and instils in them a belief in what the leader is trying to achieve (Ardichvili & Manderscheid, 2008). The transformational leadership is based on engagement of both the leader and the follower/s to promote motivation and commitment to perform effectively. Such a transformational leadership approach produces powerful outcomes (Wang & Zhu, 2011). Bass & Avolio (2000) outlined the five basic components of transformational leadership: idealized attributes, idealized behaviours, inspirational motivation, intellectual stimulation and individualized consideration.

These five components advocated by Bass & Avolio (2000), which are also referred as the 4 I's, are defined as follows: **'idealized influence (attributed)'** develops a relationship between the leaders and the followers attributed to the charisma of the leader, based on the leader's moral attributes such as ideals and values; **'idealized influence (behaviour)'** considers their followers' needs before the leader's needs, with a collective concern focussed on values, beliefs with a moral and ethical concern for all decisions; **'Inspirational motivation'** provides a motivational element for the individual and the organisation with a challenging approach to all problems; **'Intellectual stimulation'** encourages followers to adopt an innovative approach by heuristic methods to discover new ways of doing things; and **'Individualized consideration'** refers to particular concern to the individual needs and individual development.

The above five elements comprises charismatic and ethical leadership values from the idealized influence attributed and behaviour and also from the inspirational motivation components (Kreitner et al. 1999), which together promote social responsible business practices, which are pivotal in today's institutional environment to promote social sustainable operations. The human element plays also a key part from the individualized consideration component, in line with the servant leadership style. Though, it cannot be underestimated that new emerging forms of leadership are focussed on leadership and power sharing respectively, to promote empowerment in a decentralised leadership approach, as McCauley-Smith et al. (2013, p.85) advocated that: 'New organizational forms, in particular, those of a collaborative nature, imply a sharing of leadership and an espoused sharing of power enabled and facilitated through empowerment and disbursing of prior centrally controlled leadership behaviors'.

'Power has been defined as 'the ability of an actor to influence another to act in the manner that they would not have otherwise' (Emerson, 1962, p.32). This is leading to the need of the integrated leadership concept, with a shift from **individual leadership** to **interdependent leadership** style (McCauley-Smith et al., 2013). Such a stance is creating a leadership mechanism which is shared and not focused at the top levels but is distributed at all management levels, which is referred as distributed leadership structure (Carson et al., 2007), where people across the organisation structure, through a cross-functional approach, are involved in decision-making, build on common values and beliefs and also share common practices to keep the 'big picture' always in mind. Such a distribution of leadership and power is also substantiated by Deming (1986, p.54), where it was stated that 'Management must adopt and institute leadership at all levels of organisation'. To view the individual leadership power and its influence over the interdependent leadership in the right perspective, the extant literature clearly indicates the pervasive influence of the leading role of top management (i.e. individual leadership style) on all the people across all management levels (Tsui et al., 2006; Ling et al., 2008).

2.6.3.2.4 Servant leadership trait

Greenleaf (1970) also introduced the concept of the servant leadership with his foundational essay, named as 'The Servant as Leader', where his theory advocates that a leader's primary motivation and role is that as a service to others, where he stated that: 'The great leader is seen as servant first, and that simple fact is the key to his greatness' (Greenleaf, 1970, p.2). The **Servant Leadership style** (Greenleaf, 2002) is another key theory of leadership focused on the leader's approach, where its premise is based on the **distribution of power to followers**, which goes a step further from the transformational leadership 'individualized consideration' component, by giving a **deeper concern to the individual**, by considering also his/her self-actualization. Furthermore, in transformational leadership, the individual is the aim of the leadership approach, assuming that such an employee relationship leads to the achievements of the goals (Smith et al., 2004).

2.6.3.2.5 Transactional leadership trait

The **transactional leadership approach** is the most pervasive used approach with a set of established objectives and tasks, as an exchange process between the superiors and the subordinates, based on **reward or punishment vis a vis performance** (Burns, 1978; Bass & Avolio, 1994, 2000; Bass, 1997). Such an approach may be useful in some cases but may rigidify and stifle improvement and innovation.

2.6.3.2.6 Situational leadership trait

The **situational leadership**, which falls under the umbrella of contingency theory of leadership (Fiedler, 1967; Lawrence & Lorsch, 1967; Thompson, 1967), was developed by Hershey and Blanchard (1969), who shifts from the leadership traits or behaviour to **contingencies of the situations**, and who advocated that there is no one best way to influence people, but different leadership styles are needed to meet each unique situations depending on the maturity of the followers. It is to be noted that the leader and the followers influence the micro conditions of the firm, as advocated by Liden and Antonakis (2009, p.1597), by stating that: 'the context of the organization is heavily influenced by individual characteristics of leaders and followers'.

2.6.3.2.7 The role of power and control in leadership: Empowerment and trust

Nowadays, in view of the demanding changes in the market, it is considered pivotal to have more effective measures of control through leadership. As a result such conditions are transforming

leadership through a paradigm shift to decentralisation and empowerment (Gretton, 1995; Lucas, 2000; McCauley-Smith et al., 2013). This is substantiated by Obeng (2001, p.85) where it was advocated that 'Control is the hallmark of the Old World while leadership is a measure of the New'. The control element was already referred as one of the management key functions in Section 2.6.3.1.1. Though, one cannot exclude the fact that control is also considered as a common element in both the leadership and management concepts, since it is a measure of how people behaviour is influenced in the course of all actions. The **power concept** in leadership is the **key source of control**, where within a business it refers to the leader's power to influence people's behaviour (Mullins, 2005). As advocated by French and Raven (1968, p.259): 'The phenomena of power and influence involve a dyadic relation between two agents which may be viewed from two points of view: (a) What determines the behavior of the agent who exerts power? (b) What determines the reactions of the recipient of this behavior?'

Furthermore, the control over the employees' performance differs as a function of trust the employees perceive about their managers, where the more trust perceived, the less is the need for an exerted control and vice versa (Handy, 1993). Such a **level of trust** determines the **level of empowerment** adopted by the employee, since the level of authority allotted to each employee determines his/her level of autonomy and self-control adopted to decide and act (Mills & Friesen, 2001). The empowerment approach is used to decentralise the organisation, so as to improve its performance with the increase in the assigned authority and responsibility to all employees across the command chain, in a top-bottom approach, to promote greater flexibility and responsiveness (Rue & Byars, 2005).

The influence of any leader is function of the leadership approach adopted in the leadership process (e.g. transactional and transformational style), which in turn influences the employees for unity in actions. Furthermore, French and Raven (1968) identified five main sources of power to achieve the change: **reward power; coercive power; legitimate power; referent power; and expert power**. Such sources of power create a perception by the recipient (e.g. employee) to undergo a **psychological change depending on each power type** (e.g. reward power will give a perception in terms of reward; coercive power will give a perception of punishment in case of failure; legitimate power will give a perception of the skills and knowledge possession; and referent power will give a perception of the identity associated with the person). The change within the recipient is associated with the employees' behaviour, opinions, attitudes, goals, needs and values and any other aspect from the psychological field (French & Raven, 1968).

Another power oriented classification which determines the level where decision-making occurs, is the based on **three leadership styles: autocratic, democratic and laissez-faire styles** (Mullins, 2005), where the democratic style focus on the group dynamics to share the power and promote teamwork mentality in all decisions and established management policies and procedures and not as it happens in the autocratic style, where the focus is on the leader, having an absolute control. On the other hand, the laissez-faire leadership style is considered as a passive approach due to the absence of leadership (Antonakis et al., 2003).

2.6.3.3 Leadership theories from the SC perspective

2.6.3.3.1 The strategic leadership approach across the SC

The main focus of the leadership approach in the previous review focused on the leaders and followers within an organisational context, but it excludes the wider contextual perspective to include interorganisational systems, with both a national and international scope (e.g. Osborn & Marion, 2009), such as SC set-ups. This macro perspective falls under **strategic leadership**, which covers beyond the dyad set-up to a relationship across different organisations (Hunt, 1991; Finkelstein & Hambrick, 1996). The leadership approach, within a wider context, also needs to apply both soft (e.g. trust, commitment and attitude) and hard (e.g. contractual obligations, product design, process and information technology) skills between all SC actors, so as to collaborate together (Whipple & Frankel, 2000; Shub & Stonebraker, 2009). Such a leadership approach, which runs across the SC, is a challenging approach, since all SC members are independent entities, and not as it happens within the organisation based set-up with a direct command and control approach (Sydow et al., 2011) or as it happens under a merger or acquisition with established common managerial practices (Palmer et al., 1993). Though, from any form of leadership stance adopted by any SC member on each other, the power and leadership structure surely affects the level of commitment of SC members (Cooper et al., 1997).

From a SC perspective, it is postulated that 'when the leadership role is taken, a firm's internal business process can become the supply chain business process' to promote synergies across the SC (Lambert et al., 1998, p.10). From the research carried out by Lambert et al. (1998, p.8) it transpires that there is no need to manage effectively all the SC business processes, but it is advocated that at least the focal firm surely needs to **integrate the key processes** with particular selected SC members, which are referred as the 'managed process links'. Such an integrated management approach across the SC is a very challenging task since 'executives are striving to interpret and determine how to manage the company's supply chain network and achieve the potential of SCM' and also several

measures need to be taken not 'to waste valuable resources when supply chains are not integrated, appropriately streamlined and managed' (Lambert et al., 1998, p.15).

2.6.3.3.2 The Leadership trait within SCM: A stakeholder theory perspective

For an effective SC set-up, from the perspective of the stakeholder theory, the leadership role within the SC context, need to classify all interactions between the focal firm and its stakeholders as either of a **primary** or of a **secondary importance** (Mitchell et al., 1997). Freeman (1984, p. 46) defined a **stakeholder** as 'any group or individual who can affect or is affected by the achievement of the organization's objectives'. For the focal firm to deploy an **effective management approach**, with all the relevant stakeholders, the stakeholder's salience concept needs to be understood, which is defined as 'the degree to which managers give priority to competing stakeholder claims' (Mitchell et al., 1997, p.854). The stakeholder's salience is based on the stakeholders' classes, consisting from one or more of these attributes: '(1) the stakeholder's power to influence the firm, (2) the legitimacy of the stakeholder's relationship with the firm, and (3) the urgency of the stakeholder's claim on the firm' (Mitchell et al., 1997, p.854).

Such a classification determines the approach to be taken by each stakeholder, which is based on the prioritizing of the social relationship according to the power type, the established requirements and the urgency of the request, to meet both static and dynamic commitments (Mitchell et al., 1997). Such a theory overlaps with French and Raven (1968) referent and legitimate power types, from the five sources of power, as referred above, but the stakeholder theory extends from a dyadic to multiple set of agents such as SC set-ups (i.e. a number of stakeholders).

Co and Barro (2008) builds on Mitchell et al. (1997)'s attributes and typologies of the stakeholder theory and provided a framework for analyzing stakeholder management strategies in supply chain collaboration to establish whether a **cooperative or aggressive collaborative strategy** is to be adopted in the relationship. The collaborative strategy selection depends on the joint function of self-perception (i.e. focal firm perceived relative power to the stakeholder), perception of others (i.e. level of importance) and group perception (i.e. group climate and efficacy) (Ibid). In view of these three perceptions a decision needs to be taken by the focal firm to establish which strategy from the two options (i.e. cooperative or aggressive) lead to the most fruitful outcome and at the same time build or sustain the established relationship by minimising the risk for the focal firm to be closed out from the collaborative arrangement (Ibid). It was established that when the level of trust among stakeholders is low, the focal firm may choose an aggressive strategy to collaborate effectively, whilst when there

exist a level of interdependence, a perception of urgency and that benefits will be incurred, may choose a cooperative strategy to collaborate effectively (Ibid).

2.6.3.3.3 SC actors' power based relationships: The power matrix

The strategic relationships continuum vary from an **arm's length** (i.e. transactional based) to **close collaborative relationship** (i.e. partnership based) type (Cox et al., 2003; Chicksand, 2015; Cuevas et al., 2015). The **partnership concept**, between a buyer and a supplier, is defined by: 'an on-going collaborative relationship between two legally separate organisations, based upon a commitment to the equal sharing of the costs, risks and rewards derived from working together', which is normally adopted when both parties have either low or high power attributes vis a vis the power structure (Chicksand, 2015, p.10). In other instances, there may be the need to adopt a lower level of collaboration, referred as an arm's length, when the relationship between the two actors may have either the supplier or the buyer in the dominant relationship, meaning that both have different power attributes, referred as power differentials (Chicksand, 2015), as referred within the context of the buyer and supplier collaborative approach or by a partnership include: 'lower transaction costs; improved management of complex purchases; dealing with contractual uncertainty; the acquisition of scarce resources; cost reduction and functional stability; improved business stability; and increased organisational legitimacy' (Chicksand, 2015, p.2).

Such a sense-giving approach for all the possibilities of different **power attributes**, is represented by the buyer and supplier power matrix (Chicksand, 2009), in Figure 2.2. Such a power matrix implies that for supplier dominance and/or independence relationship, the most appropriate sourcing strategy for the buyer is reactive sourcing (supplier selection) (Cox et al., 2004a). Whilst for a buyer dominance and/or interdependence relationship, the most appropriate sourcing strategy for the buyer is a proactive sourcing (supplier development) (Ibid). With an extended dyad relationship, a proactive SCM can be adopted to manage all the relevant SC tiers (Ibid).

As referred in Figure 2.2, the buyer and supplier power resources are defined by three variables: **utility, scarcity of suppliers and information scarcity**. This implies that such power resources determine the buyer power over the supplier power and vice versa, so that the matrix includes all the possible variations and also defines the dominance and type of dependencies (i.e. interdependence or independence). The meaning of the power resources variables, as depicted in Figure 2.2, are that: utility refers to how useful or important each supplier is for each other (e.g. volume of orders whether useful or not); scarcity refers to whether the items provided by each actor are available in the market

or restricted (i.e. scarce of suppliers); and information scarcity refers to the availability of the relevant information on each actor to assess each other' performance (i.e. SC visibility) (Chicksand, 2009).



Figure 2.2: The buyer and supplier power matrix (Source: Chicksand, 2009).

2.6.3.3.4 SC actors' collaborative relationships: Buyer-Suppliers relationship type framework

To understand the power dependencies relations, Cox et al.' s (2003) framework in Figure 2.3 outlines the four generic buyer-supplier power structures, namely the: buyer dominance, interdependence, independence and supplier dominance vis a vis the 'way of working' (i.e. arm's-length or collaborative relationship) and 'share of surplus value' (i.e. financial value or its operational centrality). To understand each SC actor power, from a management perspective, it can also be analyzed from the resource dependency theory, since it explains the relationship between entities and how SC actors depend on the **shared resources of each other**, which may be both tangible, such as supplier support, information sharing and shared goals and intangibles, such as investment in equipment to meet buyer's needs (e.g. Ramsay, 1996).

According to Cox et al. (2003), such a framework outlines that there are two variable concepts. One concept defines the '**way of working**' based on two types of relationships, referred as arm's-length and collaborative, which create a continuum of low contact (e.g. contractual information) to close and proactive value-added relationship (e.g. joint commitments, leverage of resources and information

sharing) between actors (Ibid). The balanced relationship type explains the ideal level of relationship to meet the two actors' requirements, either through an arm's length or collaborative approach, where both SC actors adopt a non-adversarial relationship, making them interdependent on each other, which further outlines that the non-adversarial collaborative approach is the closest type of relationship between SC actors, referred by the term partnership (Cox et al., 2003; Hingley, 2005; Chicksand, 2015). The other concept is the '**share of surplus value**', referred as buyer skewed or supplier skewed, based on the division of surplus value in buyer-supplier relationships, which also creates a continuum of **adversarial** static value to adversarial growing and dynamic appropriate value, and a balanced position between the two extremes based on a **non-adversarial** arm's length or collaborative effort (Cox et al., 2003).



Figure 2.3: Buyer–supplier relationship types (Source: Cox et al., 2003, p.138)

2.6.3.3.5 The aims of collaborative relationships

The SC members engage in collaborative relationships for **balancing of power** for the **mutual benefit** (Dyer & Singh, 1998; Leonidou et al., 2008; Hausman & Johnston, 2010) but others disagree with this ideal scenario, stating that the scope is to **gain individual competitive advantage** (Hingley, 2005; Caniels & Gelderman, 2007). Cox (2004c) has taken a more comprehensive perspective by indicating that each firm, within the SC, performs with a level of power asymmetry, so as to gain, from both as an individual entity (e.g. status) and as part of a group, from both commercial and operational benchmarks and also enhances its individual competitive advantage. Such a collaborative relationship

decision is a crucial step forward, since every SC member needs to understand its capabilities, depending on its power, to 'appropriate more of the value for themselves if they are able so to do' (Cox, 1999, p.172) to promote business success. The power varies from power symmetry to power asymmetry, as referred in Figure 2.2, which determines the depth of the cooperative relationship, such as an arm's length or in-depth collaborative relationship (Cox et al., 2003; Chicksand, 2015; Cuevas et al., 2015). Trust shifts from a positive to a negative attribute when the relationship between actors shifts from the power symmetry to the power asymmetry continuum (Huxham & Beech, 2008).

The power exerted from the dominant player (e.g. distribution of risks, benefits, etc), as already referred in Section 2.6.3.2.7, may take various forms such as reward power; coercive power; legitimate power; referent power; and expert power (French & Raven, 1968) so that the other parties, through such business relations, either gain or lose benefits. According to Yeung et al. (2009) the two leadership roles, adopted by the dominant firm, are classified as either **non-coercive** or **coercive power** approaches, which depend on the type of relationship (e.g. arm's length or close collaborative) and track record of performance (e.g. more stable orders, financial penalties, revenue sharing). Co and Barro (2008), as already referred above, also provided a framework in line with Yeung et al.'s (2009) theoretical perspective for either adopting a **cooperative** or an **aggressive** collaborative strategy based on different set of criteria.

The ideal collaborative scenario is referred by the situation when the dominant firm builds on trust, commitment and transparency with the weaker (i.e. dependent) firms (Cox, 2004; Ambrose et al., 2010) and applies a non-coercive power approach and/or use a joint combination from the five power forces, as referred by French and Raven (1968), not to produce a counter-productive effect on the performance of any SC member, which may even lead to termination or damage to the relationship (Benton & Maloni, 2005).

Furthermore, situations may exist when the dominant player may tend to exploit the relationship with the dependent SC members through **opportunistic behaviour**, by assigning to such SC actors below average benefits, instead of employing an **equitable stance** (Heide, 1994; Hingley, 2005) and furthermore owns and controls key resources that appropriate value (Cox, 1999). The relationship climate between SC actors is very dynamic, since both the dominant and the dependent players, while being engaged with a current set of SC actors, are also in search for new opportunities, which may seek more rewarding partners or players, with the termination of the contractual obligations (Weitz & Jap, 1995).

2.6.3.3.6 SC actors' value-adding roles

SC actors, such as suppliers, may also seek to **increase the dependence** of the buyers by undertaking investments initiatives to enhance interdependence between them (Ramsay, 1996) and at the same time enhance its competences and capabilities to promote added-value. Borys and Jemison (1989, p. 241) defined the concept of value as 'the process by which the capabilities of the partners are combined so that the competitive advantage of either the hybrid or one or more of the parties is improved'. The SC is an extended network of dyadic relationships, with several stages that add value all across the SC tiers until they reach the end-customer to deliver the final customer perceived value (Cox et al., 2001). In fact, an extended definition on value, outlines that it can take different forms: value to the customer, referred as the customer's value proposition; the value-adding processes across the SC, from one SC entity to the next until the final customer; and the monetary value for the company, referred as the value appropriation (e.g. return for the investment) (Cox et al., 2001). Such a value added perspective vis a vis the value proposition and value-adding processes is in line with Porter's (1985) value-added theoretical perspective. Furthermore, it is to be noted that value consists from both financial and non-financial value (i.e. monetary value and brand name/image value respectively). Such SC thinking, based on Cox et al.'s (2001) three forms of values, enables the focal firm to focus effectively how: to win the market by the right product to meet the customers' best perceived value; to design processes that optimise the activities for best value-added products/services with the contribution of all SC tiers; and to establish the best return on the capital employed (Cox et al., 2001).

The focal firm within the SC context, as the buyer, the underpinning rationale is to establish the most appropriate sourcing strategy to build a robust relationship management approach to promote **best practice** and the **best performance outcome** (Cox et al., 2004a). In such a way, the focal firm, may be subjected to three levels of satisfaction, referred as satisfied, more or less satisfied or dissatisfied (Cox et al., 2004a). Ideally a proactive SCM approach is implemented among all SC actors, but in reality, due to the different contextual conditions of all SC actors determined by their power regimes (Cox et al., 2001; Glyn, 2001) or power circumstances (Cox et al., 2004a), such an ideal approach, in practice, is difficult to achieve.

2.6.3.3.7 SC actors' alignment for performance improvement

Once the sourcing strategy is selected, there is the need for **aligning** it effectively (Cox et al., 2004b). In practice, an ideal alignment is the main objective for **performance improvement** and **leverage of resources** from the dyad relationship and also from all SC actors, but if not achieved, it can take two

other forms of alignment (Ibid). One form is when the relationship is misaligned and sub-optimal, with the possibility of remedial action to re-align the relationship in place and the other form is a misaligned and dysfunctional relationship, with no chance to remedy and achieve alignment (Ibid). Such a latter alignment relationship has a no way out to remedy and the buyer is recommended to seek an alternative supplier (Ibid). To remedy problems associated with the sourcing strategies implementation, every SC actor needs to be informed, through training, development and knowledge sharing, to understand the best practices that need to be followed (Meehan & Wright, 2012). It cannot be excluded that the sourcing strategies with their alignment between all SC actors, need also the internal alignment of each individual organisation, within its functional areas, so as to have both external and internal alignment in place (Cox et al., 2004a).

In a nutshell, such a collaborative scenario makes it crucial for all SC members to build on an ideal relationship climate, based on the **right power deployment** to promote a win-win approach with **trust-building** (Hingley, 2005; Terpend & Ashenbaum, 2012), to minimise any actor from seeking other partners or players in other SCs, causing a break-out from the relationship or else a dysfunctional or a misalignment or a sub-optimisation of the SC operations respectively, with the resultant negative consequences on the other SC actor (Cox, 2004a, 2004c). Such a **win-win approach** between SC actors is not the only scenario, since certain asymmetric power circumstances, between SC actors may exist, that requires a dyad relationship to adopt a **win-lose approach**, since it may also be beneficial for both parties (Cox, 2004d).

2.6.3.4 Summary of the leadership concept from both the focal firm and SC perspective

2.6.3.4.1 The leadership traits: transformational, transactional, servant and situational

All the above leadership styles within the focal firm, especially the transformational and transactional leadership styles may be mapped on the two dimensional Managerial/Leadership Grid axes (Blake & Mouton, 1985) where they juxtaposes two leadership approaches, referred as the Task and Relationship/People behaviours. The task attribute refers to the 'Concern for production', which emphasizes the leader's importance given to the accomplishment of the tasks (i.e. transactions) to promote a high level of performance. The relationship/people attribute refers to the 'Concern for people', which emphasises the leader's importance given to the individuals' needs and expectations. Leaders need to ensure a concern for productivity but at the same time need to build trust among all employees to unleash people's passion and promote knowledge sharing to foster creativity (Fawcett & Magnan, 2004).

The key leadership theories, from the extant literature, were focussed on the importance of the **transformation, transactional** and **servant leadership** styles, which are vital for leadership practices and its success. The Leadership Grid (Blake & Mouton, 1985) clearly outlines that the leadership styles consist of a continuum between a concern for both the leader's objectives and the employees' needs, where both are mutually inclusive and dynamic since ideally the outcome shall end up to a winwin situation for both actors. Such a continuum of leadership style is in line with the **situational leadership** approach since every situation is contingent and may need to take dedicated measures to achieve effective leadership especially due to the dedicated organisational climate (Hershey & Blanchard, 1969). This means that leadership approach consists from a leader, a follower and the contextual conditions (Tannenbaum & Schmidt, 1973; Tosi, 1991; Osborn et al., 2002; Liden & Antonakis, 2009).

The contextual conditions were further defined, within the leadership practices, and were classified to describe the organisational context to be under one of the referred situations of: stability, crisis, dynamic equilibrium, and edge of chaos (Osborn et al., 2002). The researcher from the literature considers that the attributes of the transformational leadership overlaps with authentic leadership (Avolio & Gardner, 2005) vis a vis the leadership values and ethical concern, where the ethical perspective nowadays is considered key in various literatures fields. In such a case all leaders need to embed in all decisions and practices, both ethical and moral principles which are away from the contemporary corruption cases and hidden agendas (Avolio & Gardner, 2005) which are also pivotal both within and across the SC to win all stakeholders' support, trust and commitment to promote an effective SCI approach.

2.6.3.4.2 The integrated leadership approach

Within a single organisation set-up, the leadership style is ideally based on a **decentralised approach**, where all employees are empowered across all management levels to promote a cross-functional and teamwork environment by adopting a leadership power sharing stance, with an **interdependent and integrated leadership approach** with all functional areas (McCauley-Smith et al., 2013). Within a SC set-up, one needs to consider, that for an effective leadership it must be seen not only from a top-bottom approach but from a 'leading out and leading in' approaches (Useem, 2001), since the employees need to constantly interact with their managers and leaders within and outside the firm with all of its stakeholders. Though, it is assumed in the literature, whichever is the context, such as buyer to supplier or employee to supervisor relationship, the leadership power need to be based on the same foundation principles (Heide & George, 1988). The effective leadership platform is reached through joint initiatives between all involved stakeholders as substantiated by Olivares et al. (2007, p. 79),

where they stated that 'Although individual-based leader development is necessary for leadership, it is not sufficient. Leadership requires that individual development is integrated and understood in the context of others, social systems, and organizational strategies, missions, and goals'.

In retrospect, an effective leadership approach needs the right **leadership style with a holistic stance** to ensure that all stakeholders' requirements are considered, to promote synergy of all actions with a teamwork vision (Bass et al., 2003; Schippers et al., 2008). Furthermore, the leader needs to have the right mindset and competence, as advocated by Cheung et al. (2012, p.3923) by stating that: 'having all the data in the world will not help to improve a company's profitability if the data are not correctly interpreted'. This approach reinforces the need of right leadership skills across all management levels, which are achieved through ongoing training and leadership development initiatives (McCauley-Smith et al., 2013), so that all relevant leaders are informed and well versed to promote effective decision making within all the SC dynamic contextual conditions. The quality of the management approach, within the SC context, is also function of the management leadership approach, which together with other seven dimensions, namely training, employee relations, customer focus, quality data and reporting, supplier quality management, product/service design, and process management, all contribute to the SC performance (Kaynak & Hartley, 2008).

2.6.3.4.3 Effective leadership with the right strategic power approach

An effective leadership approach also needs to consider that to reach effectively all SC stakeholders, need to interact according to their salience: which is function of the power of the focal firm vis a vis the SC member; the established practices jointly agreed upon; and the urgency or critical need of the matter (Mitchell et al., 1997). Such a SC context in general is determined by the individual power of all SC members which drives them to pursue different types of influences on other SC actors, but above all they try to extend the internal business processes also across the SC business processes, through either **coercive or non-coercive power strategies** (Yeung et al., 2009). Such a power influence approach is function of three power resources, referred as levels of utility, supplier's scarcity and information scarcity, to meet both commercial and operational benchmarks (Chicksand, 2009). Trust between SC actors builds with more power symmetry (Huxham & Beech, 2008). The deployment of power needs to be managed effectively, through a **cooperative or non-coercive approach** (Co & Barro 2008; Yeung et al., 2009), so as to build a positive climate within all relationships, so that it does not produce counter-productive effects on the overall SC performance, with the ultimate break-out of any of the SC actor from the SC network (Cox, 2004a, 2004c).

2.6.3.4.4 A holistic and integrated management and leadership approach

The effective management and leadership approach within the SC context needs to incorporate a holistic and inclusive understanding of both the financial and the physical properties of the SC (Cox et al., 2001). In such a way the value appropriation is to be mapped with the physical value added processes (Ibid). Furthermore, the focal firm needs to map also the power attributes of all SC actors to manage all the relationships with each other, since there can be more than one SC actor engaged with the focal firm, both as dyad or else as an extended network (Ibid).

Such a theoretical analytical approach to SC thinking, through power regimes or power circumstances of all SC actors, imply that every SC member needs to establish who gets what, how, where and when in the SC (Cox et al., 2001). This shows that SCI is not always possible due to various valid reasons (e.g. due to differing power). **SCI** is likely to occur for **interdependent SC actors** or when the focal firm is the buyer and it is close to the customer and can also apply power dominance (Ibid). It is well established in the literature that SCI increases the focal firm power, apart from other benefits, relative to the other SC actor (Horwitch &Thietart, 1987; D'aveni & Ravenscraft, 1994). Management needs to understand the power of each SC actor, since circumstances may exist where any SC actor may block value to another SC actor (Cox et al., 2001) or else a SC actor may have the power to undermine the overall SC (Glyn, 2001), since the SC is strong as its weakest link. Power regimes outline that there exists a hostile environment with a self-regarding effort by SC actors to appropriate value (e.g. opportunism from an RBV perspective) (Cox et al., 2001). This also implies that the complex network of SC actors needs **dedicated relationship building** with each other to enable business success (Ibid).

2.6.4 Business strategy theme

2.6.4.1 Corporate and business strategies

Johnson and Scholes (1999, p.10) define **strategy** as 'the direction and scope of an organization over the long-term: which achieves advantage for the organisation through its configuration of resources within a changing environment, to meet the needs of the markets and to fulfil stakeholder expectations'. The strategic decisions are of a complex nature due to an uncertainty and a volatile economy and with a competitive and dynamic market.

Due to the nature of the study, mainly focused on the SME sector type of firms, the terms **corporate and business strategies** are assumed by the researcher as common terms, due to the relative small size of the firms' understudy, where the focus is only based on a single business unit and wherever any SME formed part of a Multinational firm, no consideration has been included on the mother's company Corporate strategy, with its different business portfolio. The term corporate strategy as a result is merged into the business strategy, being a single business company (Thompson et al., 2005). Such a stance is also used to simplify the systematic literature synthesis of all the emerged themes from the data. The business strategy classifies the strategic management approach to consist from three levels which cover the strategic, tactical and operational levels respectively, so as to unify all functional areas together with all SC members with their capabilities and limitations (Skinner, 1969; Stonebraker & Afifi, 2004; Swink et al., 2007).

The overall business strategy is based on the mission of the firm, which outlines the values and expectations of all the stakeholders including the customers. The strategic management approach is based on the inter-relationship between the **strategic analysis**, **strategic choice and strategy implementation** (Johnson & Scholes, 1999). Such a three step process is used to ensure that the firm understands its environment and its strategic capabilities (i.e. strategic analysis) to take into account all of the relevant information so as to derive the strategic options to decide (i.e. strategic choice) and then proceed with its implementation. Such a strategic approach needs to be established effectively by selecting the best option (i.e. strategic fit) so that the focal firm achieves a competitive advantage (Ibid). The significance of such competitive capabilities is based on the fact that those SCs that lag behind and fail to deliver the value will lose in the competitive market (Cox, 1999).

2.6.4.2 The business operations strategy

The way forward for the firm's business strategy needs to have an operations strategy based on a **holistic integrated SCM approach** in-conjunction with a **continuous improvement mindset** in all operations so as to achieve a **sustainable competitive advantage** (Chen & Paulraj, 2004; Lummus et al., 2008). Slack et al. (2004, p.67) defines the operations strategy by stating that it 'concerns the pattern of strategic decisions and actions which set the role, objectives and activities of operations'. The operations strategy has several performance objectives to achieve and these are being referred as the **competitive capabilities** (Johnson & Scholes, 1999), which are further discussed in Section 2.6.9.

Not all firms can reach all competitive capabilities in the same manner, due to the adopted complexity and different **operations strategic priorities**. Such a complexity is attributed to the performance and decision-making variations that may originate by different stakeholders forming up the SC, which may need to jointly-undertake the necessary strategic trade-offs for a win-win outcome. Such a **strategic trade-off** concept was referred by Skinner (1969) who outlined the importance to establish a clear set of competitive priorities. Similarly Hayes and Wheelwright (1984) in their seminal paper, substantiated the importance of having strategic priorities and also trade-offs and synergies vis a vis competitive performance dimensions. They classified that there also exists four types of **operations strategies** sources that the focal firm may adopt, to achieve a competitive advantage within manufacturing (Ibid). The four operations strategies were divided into two types referred as neutral (i.e. reactive approach) and supportive (i.e. proactive approach), where both are considered from an internal and external perspective to either maintain or improve the competitive performance. Such a competitive advantage perspective is also discussed in Section 2.6.9 from Porter's seminal works, which is dedicated to review the literature related to competitive capabilities as an outcome of the strategic implementation.

2.6.5 Manufacturing strategy theme

2.6.5.1 Manufacturing strategy contribution to competitiveness

From the seminal paper of Skinner (1969), it was claimed as pivotal the direct bi-directional relationship between the corporate/business strategy with the manufacturing strategy and the importance of the manufacturing strategy contribution to the firm's competitive edge. According to Skinner (1969, p.136), the manufacturing function must not become a 'corporate millstone', due to a non-competitive production system, but must become a 'competitive weapon' with a top-down management approach, based on a continuous decision-making commitment to implement the right trade-offs to derive the relevant manufacturing policies, which are to be linked to the competitive and strategic facts. Such a stance needs to ensure that the top management gives due importance to the manufacturing strategy since it adds to the firm's 'arsenal of competitive weapons' (Skinner 1969, p.145). Furthermore, Hayes and Wheelwright (1984) developed a set of six (6) World Class Manufacturing practices, based on empirical research on Japanese, German and US firms, which enables firms to achieve outstanding performance and to build competitive strength. These six best practices are: Workforce skills and capabilities; Management technical competence; Competing through quality; Workforce participation; Rebuilding manufacturing engineering; and Incremental improvement approaches (Hayes & Wheelwright, 1984). Such a best practice approach within the manufacturing practices needs to be extended from an internal productivity perspective to both suppliers and customers' requirements. As defined by Garetti and Taisch (2012, p.84) 'manufacturing includes all industrial activities from the customer to the factory and back to the customer, thus including all the different kinds of services that are connected to the manufacturing chain'. Such an era led to concurrent engineering (Winner, et al., 1988), lean management (Womack et al., 1990), agile manufacturing (Youssef, 1994), JIT (Flynn et al., 1999), mass customisation (Duray et al., 2000) and efficient consumer response (Hoffman & Mehra, 2000).

The manufacturing strategy role in the contribution to **competitive performance** is also function of the product/process matrix position due to the nature of production method used (job, batch, line and continuous) and the nature of the product (i.e. type and volume) (Hayes & Wheelwright, 1979a, 1979b). Such a stance is used since the production method determines the competitive approach due to the lower costs, associated with say continuous production methods. Similarly a higher value-added product manufactured type, when it incorporates increased attractive features (e.g. dedicated product design, higher quality, lower price and high dependability), so as to create a competitive edge over others. Such Skinner's and Hayes and Wheelwright's theories both outline that the manufacturing strategy is a direct antecedent to a competitive strategy.

2.6.5.2 The manufacturing strategy and manufacturing leadership practices

The implementation of the manufacturing strategy is based on the operations strategy key categories classified under the structure and infrastructure decisions (Hayes & Wheelwright, 1984; Hayes et al., 2005), which built on Skinner (1969) work, by giving more importance to the **manufacturing leadership practices** to pursue the operations strategy, through the business's manufacturing policies and people, to promote a competitive edge. Such two decision areas promote an improved leadership approach by configuring and aligning these key operational strategic areas with the relevant priority assigned by the business to promote competitive edge.

Another organisation alignment approach related to the Hayes and Wheelwright (1984) operations strategic stance is the McKinsey 7-S model (Waterman et al., 1980). The 7-S model has seven internal factors that need to be considered for an effective strategic implementation process to achieve a competitive edge. These factors are classified as three hard elements and four soft elements, where strategy is one of the main elements since without a clear strategy, the organisation will lack direction and impetus. The seven elements are strategy, structure and systems which are referred as the hard elements and the **shared values**, skills, style and staff which are referred as the soft elements. The model focuses on the 'shared values' element, since it makes all the other elements aligned together, since the cultural values of all people reinforces unity and teamwork which are two recipes required for effective and efficient performance. Furthermore, such people shared values importance, shows the key role of the human element, within the strategic implementation process, meaning that people are a key and invaluable asset (Rothwell, 2002; Tarique & Schuler, 2010; Xu & Thomas, 2011). Such a shared value can be seen from both manufacturing, by the processes coordination and also from a SCI perspective, by the intra- and inter-organisational practices.

2.6.5.3 The manufacturing strategy and SCI

The **manufacturing strategy** is directly linked to the **SCI strategy**, as advocated by the (Frohlich & Westbrook, 2001) seminal paper, when they asserted that the 'new millennium upstream and downstream integration with suppliers and customers has emerged as an important element of manufacturing strategy'. This perspective was also referred by Flynn et al. (1999), where they extended the work of Hayes and Wheelwright in 1980s, by advocating that SCI within and outside the focal firm is a crucial approach to face all world class manufacturing problems associated with the key operational objectives of cost (i.e. quality and features), dependability (i.e. specifications, time and service), quality (i.e. process quality) and flexibility (i.e. product and volume).

In today's industry, manufacturing is being based on lean manufacturing, as an enhancement of mass production, when it focuses on cost and efficiencies, and deploys agile manufacturing, for a more responsive approach at the downstream side of the SC (Yusuf & Adeleyei, 2002; Gunasekaran et al., 2008; MacBryde et al., 2013). The manufacturing process needs to implement JIT practices and the overall SC is using a pull-based system (Kros et al., 2006), to align the production and business processes throughout the supply chain to meet the changing demands of the chain's ultimate customers (Cox et al., 2003; Green & Inman, 2005). Pull strategies are customer focused and by adopting a JIT-II-selling strategy, employ tactics that **integrates** fully empowered sales representatives within the customer's purchasing process (Green & Inman, 2007). As already referred in the a priori literature review, Section 2.5.1, Yusuf & Adeleyei (2002), based on an academic study focused on lean and agile manufacturing, contended that **integration** can deliver concurrently on multiple competitive capabilities.

Such agile techniques importance focused on the time of SC response to the customers' request has also been substantiated by Cox et al. (2003). Furthermore, **agility** within the manufacturing SC context, is also considered as a pivotal characteristic of **HVM**, since the manufacturing firm needs to build on their capabilities, innovative and environmental sustainability approaches (Martinez et al., 2008) for the success of the SC within the dynamic and competitive environment (Agarwal et al., 2007; Braunscheidel & Suresh, 2009; MacBryde et al., 2013). The new innovative way of manufacturing is to deploy a paradigm shift, for a **service led competitive manufacturing strategy** (Neely, 2009; Baines et al., 2009), which is based on a shift from product selling to an integrated product and service offering (Baines et al., 2007). Such an approach is referred as the **servitization of manufacturing** (Lightfoot et al., 2013) or the integrated solutions of manufacturing (Davies et al., 2006). MacBryde et al. (2013) supports this perspective by asserting that manufactures are engaging in activities which are complementary to the core production process, by adopting an approach that

67

involve the customers in both the design and service activities to enhance the overall service value proposition through enhanced servitization.

From a SC perspective, the **internal SCI** focuses on the focal firm manufacturing processes and strategies to fulfil the customers' requirements and efficiently interact with all upstream and downstream SC members to achieve **competitive advantage** (Ragatz et al. 1997; Flynn et al., 1999; Frohlich & Westbrook 2001; Paulraj et al., 2008; Flynn et al., 2010). The external linkages with all SC members, are not after a functional organisational structure, as it happens in the internal processes of the focal firm, but needs all the functional areas to work together in a **synchronous and unified process approach** (Flynn et al., 2010; Richey et al., 2010). It cannot be excluded that the manufacturing processes, to be in line with all lean and agile processes need the **enabling processes of IT/IS** (Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014). The IT infrastructure is facilitating and creating more effective IT based **information sharing capabilities** (e.g. sales and purchasing information) and **flexibility within all manufacturing processes**, both within and outside the focal firm, with all of its SC actors, such as in JIT inventory tracking, and in flexibility for product development, production, logistics, suppliers and supply base respectively (Gunasekaran et al., 2008; Jin et al., 2014) and also helps to build competitive performance of the overall manufacturing SC (Gosling et al., 2010).

2.6.5.4 The manufacturing 'make' or 'buy' strategic decision

The manufacturing strategy, as already referred, is interlinked with the SCI strategy, since the manufacturing process may be made inter-dependent on other firms forming up the SC. In fact various scholarly work have focused on the strategic decision whether a manufacturing process is kept in-house or is preferably outsourced (i.e. make or buy) to achieve or sustain its competitive advantage (Harrigan, 1985; Klein, 1988; Bajec & Jakomin, 2010). This strategic decision may either be based on the **core competency of the focal firm** (Cox, 1999), where the core activities are performed in-house and peripheral activities are outsourced, in line with the RBV (Barney, 1991) and extended RBV (Matthews, 2003) respectively. Such an approach exploits the resources of the focal firm and of its suppliers to improve on the overall competitiveness (Cox, 1999; Baloh et al., 2008). The other perspective may be based from a **cost efficient rationale** by the focal firm and of its suppliers to improve on the overall competitiveness (Williams, 1985; McIvor, 2010). In reality the decision is multifaceted since it is function of a complex set of other contextual conditions such as suppliers' availability, focal firm's products portfolio and/or capacity of production, sector type and proximity of suppliers.

2.6.6 Supply Chain (SC) Strategy theme

2.6.6.1 SC strategy and supply chain integration to achieve strategic alignment

The SC strategy is concerned with the **integration and coordination decisions** within the focal firm in conjunction with the set-up to be implemented with all stakeholders (Gunasekaran et al., 2008; Green et al., 2014). The SC strategy is complex concept in the SC literature (Rose et al., 2012) and consists from both a **strategic and operational capabilities** with their **alignment** (Cox, 1999; Kim, 2009) function of the contextual conditions (Cox, 1998). Such SC foundations with its variations imply that a contingent model needs to be adopted to fit each business situation (Stonebraker & Afifi, 2004). The SC strategy also needs to adapt and structure its resources to effectively align to the changing competitive conditions (Thompson, 1967; Hofer, 1975; Kim, 2009). Within such a competitive environment, an **extended integrated collaborative approach** is needed in all the activities, within the focal firm and across the SC, to deploy a time-based SCM approach, so as to achieve time compression, flexibility and agility (Stonebraker & Afifi, 2004; Lin et al., 2006). The organisations also need to fit their **environmental sustainability** requirements within their business strategy and across the whole SC, including suppliers' selection and logistics (Vachon & Klassen, 2006; Hsu et al., 2013; Caniels et al., 2013).

2.6.6.2 SC strategy from a manufacturing and marketing perspective

The SC strategy may be based on the type of product under manufacture, whether it is a **functional or innovative type**, so as to focus effectively, referred as matching, either through **efficiency or responsiveness** respectively (Fisher, 1997), by formulating the ideal SC strategy to promote business success. Fisher's model gives a **manufacturing against a marketing strategic perspective** of doing business to make a profit. The innovative approach is more customer focused, since an effective change management approach is needed to meet the innovative changes the customer wants (i.e. being responsive to all dynamic market, needs agility and flexibility) which in turn generates revenue with improved performance. According to Fisher (1997) the performance improvement with an innovative product is larger than that gained through efficient processes with a functional product (i.e. low cost basic product from a manufacturing perspective).

The SC Strategy is about effective decision making to promote success and to build on the **competitive capabilities** (Bojarski et al., 2009; Kim, 2009). A strategy that leads to performance improvement does not derive competitive capabilities and as a result the strategic approach does not make a firm a winner in the market but only a qualifier on the road to competitiveness. This is the key focus given by the seminal article by Fisher (1997), which postulates that the adopted SC strategic

approach is function of the type of product to ensure that the overall SC strategy fits effectively the business objectives, so as to meet what the customers want and to be competitive, even at the expense of less efficient processes, unless they are managed in such a way to minimise such inefficiencies.

2.6.6.3 Strategic supply chain integration with all SC actors

The SC strategy is concerned with all suppliers and upstream SC members' relationships across the SC to give a competitive edge through improved performance (Carter et al., 2000; Scannell et al., 2000; Kim, 2009). Through **strategic supplier integration**, the focal firm acquires and shares both information and knowledge on operational, technical and financial aspects, to improve on the overall capabilities and performance (Swink et al., 2007). Similarly, through downstream strategic SCI, the focal firm builds on all the customers' relationships across the SC, to give a competitive edge through performance improvement (Ellinger et al., 2000; Kim, 2009). Such a **strategic customer integration** approach acquires and assimilates the customers' information and knowledge to establish their preferences and requirements so as to establish the relationships with ongoing customer contacts and interactions (Swink et al., 2007).

The business strategy with the right strategic decision making, needs to establish the effective SC strategy, so as to integrate in some form or in a limited way, all business functions both within and throughout all the SC tiers (e.g. inventory, logistics, purchasing and supplies), backed by an IT infrastructure, to achieve both manufacturing and performance improvements (Carr & Smeltzer, 1999; Wisner, 2003; Gonzalez-Benito, 2007). Such a SCI approach, with **process integration** across the SC, provides the right structure and platform that leads to competitive success (Eisenhardt & Tabrizi, 1994; Birou et al., 1998; Kim, 2009). Such an extended understanding of the focal firm from a **SCI perspective** implies that the focal firm resources are being jointly interfaced and aligned with that of all SC stakeholders, so as to exploit: the synergy of resources, which are embedded in all SC actors; the SC strategic alignment advantages; and a cumulative competitive edge.

2.6.6.4 SC strategic decision based on the manufacturing 'make' or 'buy' strategic decision

The focal firm needs to employ a cost/benefit analysis to establish the feasibility of the way forward, since SCI is risky and may not be appropriate in all situations (Stonebraker & Afifi, 2004). Such a new operational business perspective, from a wider approach to operations, where the focal firm deploys resources outside its boundaries to achieve an overall competitive edge, is supported by the extended RBV of a network of firms (Matthews, 2003) and the RBV of the focal firm (Wernerfelt, 1984, Barney, 1991). Such an operational perspective explains the advantages gained from the

synergy effects of all the SC actors with each other and within the focal firm to promote a unified and a distinct competitive SC.

Such a SC strategic decision is referred to as the '**core competence paradigm**', to decide which operations are to be retained and others to outsource, so as to compete effectively (Cox, 1999, p.169). Such a decision is a crucial step forward, since every SC member needs to understand its capabilities, depending on its power, to 'appropriate more of the value for themselves if they are able so to do' (Cox, 1999, p.172) to promote business success. This is the foundation principle within a SC having dominant players, who are able to own and control key SC resources with the appropriation of value, over other SC members (Cox, 1999, 2004a). This is in line with the **manufacturing strategy**, make or buy decision, as referred in Section 2.6.5.4 and the virtual, extended and vertical integration enterprise approaches respectively, as referred in the collaborative theme and a priori literature review, Sections A4.1.8 and 2.5.2 respectively.

Such a SCM approach needs to be seen from the right **strategic and operational perspective**, since an **integrated SC** should be ideally based on power equivalence and 'equity, trust and openness' and not on the power of any SC member to direct others, with a hierarchy of structural dominance and dependency (Cox, 1999, p.171). Though, depending on the **SC type and context**, such as in the automotive manufacturing sector, where there exist a standard and smooth flow of demand, it makes structural dominance feasible for dominant players to impose direction on weaker (i.e. dependent) SC participants within an integrated SC (Cox, 1999). With such an approach each SC member creates value-added operations (e.g. innovation, improved customer service, superior product attributes) and also manages effectively its appropriation of value (e.g. extracts above average profits, build brand name, strengthen competitive position), since both **value creation** and **value appropriation** are strategic objectives required for achieving sustained competitive advantage (Mizik & Jacobson, 2003; Zhang & Chen, 2008).

2.6.6.5 The SC strategy as a function of the dominant players

The SC power structure outlines that the overall SC set-up has a level of competition, which is an **inherent adversarial approach** between SC members (Chicksand, 2015), based on who shall own and control the SC resources. Such a business trend is substantiated by Cox (1999, p.173), by stating that 'Most suppliers are basically **opportunistic** rather than **deferential**, and have little real incentive to tie themselves to one customer unless they are forced to do so'. At the same time such a **dominant player based approach** promotes success as a result of: the innovative approach to promote value-added operations; minimising any threat from other SC members by managing effectively all SC
actors; and exploiting the appropriation of value (Cox, 1999). For such an innovative SC climate, ideally there must be a '**benign power structure**' between all SC actors so that the resources ownership and their control are managed effectively across the SC (Cox, 1999, p.173). This means that there is the need to retain power and undertake effective control over all SC members depending on the SC configuration (e.g. how the SC is structured with the margins of allocated profits), which also clearly outlines that a '**one-size-fits-all' model** does not exist, due to the unique characteristics of each SC (Heide, 1994).

Supply chain theory suggests that the 'supply chain should be managed from end-to-end' but in practice this achievement has several barriers and needs a considerable effort to reach beyond the first tier in both directions (Storey et al., 2006, p.763). Such a **challenging end-to-end approach** was articulated by Storey et al. (2006, p.763) assertion, by stating that: the 'Management of the supply chain was analogous to a relay race, with responsibility being passed from one company of actors to another'. Though, it cannot be excluded that extended dyadic SCs exist, where SCs consists from a focal firm that has the **overall control of various SC actors** (Cox et al., 2001; Frohlich & Westbrook, 2001). The SC strategy is also determined by the knowledge embedded within the firm management to formulate the right strategy, as advocated by Cheung et al. (2012, p.3907) that: 'the knowledge-based system helps to formulate supply chain strategies which are capable of adapting and responding to the supply chain evolution, and helps **to optimize supply chain configuration** leading to the long term success'.

2.6.6.6 The SC strategy through process integration: The SCOR practitioner's model

Practitioners, through the Supply Chain Council, have established the Supply Chain Operations Reference model (SCOR, 2010), to outline the importance of the strategic importance of SCI associated with all SC entities. The SCOR model outlines that for effective SC strategy, the SC set-up needs to implement an integrated approach across the SC, by **linking strategically all processes**, which include planning, sourcing, making, delivering, returning and enabling between all SC actors, namely suppliers, customers and manufacturers, so as establish performance benchmarks and promote best practices (Bolstorff & Rosenbaum, 2007). The SCOR model, as referred by Bolstorff and Rosenbaum (2007), can be used to achieve SCM excellence by deploying a project management approach. The SCOR methodology, to be implemented effectively, needs to identify the scope, by establishing the business context, building SC definition matrix and derive a project charter (Ibid):

(a) The business context is used to determine the context for the SC improvement, by analysing the strategic background (e.g. SWOT and CSF), financial performance (e.g. revenue and income and total

current assets and liabilities), internal (e.g. organisation structure, performance measures and SC span) and external profiles (e.g. PESTEL assessment, 5 Forces Model and stakeholder list) respectively.

(b) The SC definition matrix outlines all the markets, customers and channels against products, locations of firm units and suppliers. A priority ranking may also be done to identify all products contribution to the SC.

(c) The project charter outlines all stakeholders to align all deliverables and define the roles and responsibilities to achieve business objectives by including the risks and dependencies on other SC members and control procedures.

2.6.7 Holistic supply chain integration (SCI) management theme

2.6.7.1 The holistic SCI approach within its contextual conditions

The holistic view of SCM has been considered as an important perspective for the formulation of the SC strategies (Wilding, 1998; Perona & Miragliotta 2004; Power, 2005; Fawcett et al., 2008; Bayraktar et al., 2009; Cheung et al., 2012). The holistic approach, from a theoretical perspective, is considered crucial to set-up a **broad integration of customers and suppliers** with the focal firm (Frohlich & Westbrook, 2002; Power, 2005). The term demand chain management (DCM), is also used to describe such a **holistic view**, since it explains how SC actors shall manage and coordinate all the links from the customer backwards to the suppliers, based on an extensive integrated approach to have an adequate understanding of the demand (Vollmann et al., 2000; Frohlich & Westbrook, 2002; Juttner et al., 2007). Lambert and Cooper (2000), in line with the DCM approach, outlined three marketing-oriented business processes, referred as CRM, customer service management and demand management. It has been clearly outlined that both the demand chain (i.e. customer side) and the supply chain (i.e. supplier side), forming up the two ends to the SC, although their focus differs, in line with Fisher (1997) market and product perspectives, both need to contribute effectively together in the pursuit of all goals (Juttner et al., 2007).

The SC is embedded in a **hostile environment** (i.e. short product lifecycle; competition; small sized orders; speed to deliver and to market) (Cheung et al., 2012). This context is clearly advocated by Cheung et al. (2012) by stating that: 'Globalization and proliferation of multi-national companies, joint ventures, strategic alliances and business partnerships have changed the business environment. For survival under the pressure of turbulence, it is critical for companies to reorganize and optimize their supply chain effectively, by taking a holistic view of the supply chain' (Cheung et al., 2012, p.3923). Such a holistic perspective is a need, since after all, although all SC actors deal with different roles (e.g. different material handling and processes), they are fulfilling the same customers' orders (Jiao et al., 2006). Furthermore, it is to be noted that on the **facilitators or inhibitors** for an effective

strategic SCM approach, needs to consider such issues and challenges not individually but from a **holistic perspective** to promote SC success (Fawcett et al., 2008). Typical facilitators are people empowerment and support, collaboration and information integration, whilst inhibitors are resistance to change, misaligned objectives, lack of collaboration, conflicting cultures and lack of trust between SC actors (Ibid). It cannot be excluded the importance to overcome the focal firm internal integration mechanisms challenges, due to the inbuilt barriers subjected to the silos thinking between functional areas and with a lack of cross-functional commitment to operations (Bowersox et al., 2000).

2.6.7.2 Holistic SCI approach: Agile and optimised SC and manufacturing operations

The holistic approach across the SC could be made possible with the advent of the **internet based technologies**, since through the web, one is able to provide an integrated platform for customers and suppliers for all activities across the SC (Frohlich & Westbrook, 2002). In fact DCM (Section 2.6.7.1) has been referred as the most powerful web-based manufacturing strategy (Frohlich & Westbrook, 2002). To achieve such a holistic process, there is also the need to have **productive based processes** across the SC, which need the **swift movement** of materials throughout the whole SC process, based on value-added processes, a high speed of flow without bottlenecks (i.e. competitive throughput time) and to narrow the process manufacturing variations for a more even flow (Schmenner & Swink, 1998). Such a stance is in line with the **theory of swift, even flow** (Ibid). Nowadays, a common saying used is: 'time is money'. As a result, organizations today, need to complete their tasks quickly to reap a greater profit based on a triple-A SC which needs to be **aligned, adaptable** and **agile** (Lee, 2004; Ishaq et al., 2012).

The **SC agility** needs to take a holistic approach to the SC based on a proactive strategic stance, to be more responsive to the customer demand (Power, 2005). Gunasekaran et al. (2008, p.550) support both the **holistic** and **agile approach**, since they advocate that **agile manufacturing** 'should not only be based on responsiveness and flexibility, but also on the cost and quality of goods and services'. The principle behind the pull strategy is sustained by Lambert (2008, p.32), where he declared that the 'supply chain starts with your customer network and flows back through to your supplier network, and all the functions of the business need to be involved in managing the combined network'. He further added that the way forward is 'the move to a more **holistic view** and finding a way to make decisions that are **optimal** for the whole of the business'. Such a holistic significance has also been advocated by Bayraktar et al. (2009, p.137) by stating that there is the need to get all the relevant information from all IT systems from SC actors in a 'DSS for **holistic decision making**', so as 'to consolidate the information from various systems together, in order to make the most suitable decision considering all

constraints. This will assist forecasting; optimize resource planning and operational efficiency; and ultimately provide a more accurate costing'.

Cohen and Lee (1988) have also referred the crucial role for the **optimisation of all processes**, such as material supply, production and distribution. This is in line with the principle that the new SCI paradigm is designated to **consolidate all the SC members' resources and capabilities**, for the benefit of the whole SC to achieve performance improvement, since the performance of all SC actors influence each other's performance (Huo, 2012). Furthermore, Fawcett et al. (2008, p.36) added that all potential SC 'barriers' need to be seen in 'combination or holistically', so as 'to separate the trees from the forest'.

In general, process management refers to a set of interlinked processes within the organisation in a concerted effort to improve organisational processes (Benner & Tushman, 2003). Within the context of manufacturing, a process approach improves manufacturing efficiencies, cost reductions, customer service and profitability (Hammer & Stanton, 1999). Such a process approach was also substantiated by Lambert and Cooper, (2000) where they declared that for the competitiveness of the whole SC to act as a **single enterprise**, there is the need to shift from managing individual functions across the SC, to **integrating activities into key SC processes**. Over the last years process management practices were applied in both manufacturing and in upstream/downstream SC activities. Quality-related initiatives, such as TQM, excellence models, ISO certification procedures and Six Sigma programmes, have become the norm in such processes (Benner & Tushman, 2003).

2.6.8 Technology deployment (in information systems and automation) theme

2.6.8.1 Information Technology (IT) as an enabler of SC and manufacturing processes

The IT strategy is ideally aligned to the business strategy to undertake the necessary investments. In practice, within the SMEs scenario, the IT strategy is reactive and implemented when there is the need to gain from IT investment or else it is an outcome from the bottom line to be able to perform works effectively (Ciborra, 1997).

Being in this information age, SC actors are highly dependent on an **effective information processing** approach, to develop and manage knowledge across the SC through IT applications (Dyer & Singh, 1998; Ofek & Sarvary, 2001; Schoenherr & Swink, 2012). The deployment of technology also has the capabilities to eliminate waste in several processes by making them **more efficient** (Cox, 1999; Gunasekaran, 2004; Sanders, 2008). IT is **enabling all collaborative processes**: through connectivity of all SC actors; to capture all real time information; to implement the necessary information

management techniques to promote SC visibility; to enable value added services; and to achieve a holistic management approach across the SC (McAfee, 2002; Chesbrough & Teece, 2002; Koudal & Wellener, 2003; Fawcett et al., 2007; Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014). Furthermore, 'seamless, real-time processes are already the status quo in both individual plants and entire companies. The next wave of business process redesign addresses interplant and interenterprise processes, known as collaborative processes' (Kagermann et al., 2011, p. xiii). This collaborative process scenario is compliant with the needs of SCI within the SC context based on the **process approach** (Lambert & Cooper, 2000; Benner & Tushman, 2003).

The importance of IT adoption is without doubt very beneficial for SMEs survival (Gagnon & Toulouse, 1996), since it is well cited that IT capability leads to superior performance (Determirhan et al., 2007). Though, one cannot exclude the fact, that practitioners within SMEs, are facing a challenging investment scenario, from a technology perspective, due to lack of capital (Bayraktar et al., 2009), even in view of all Governments initiatives and support for investment in technological solutions (e.g. provision of funds, incentives and capital guarantees schemes).

2.6.8.2 IT promotes SCI

With the advent of new technology applications (e.g. content management solutions, data warehouses, search engines and social networks), such an environment is solving a lot of demanding **information management** issues and challenges, which is leading to a paradigm shift from a platform of 'information overload' to an effective 'knowledge management' platform (Kagermann et al., 2011, p.27). Furthermore, in certain processes, the human element interplay is changing since the 'role of people in the fulfilment process is changing progressively from processor to process monitor and enhancer' (Kagermann et al., 2011, p.104). The human approach role is further elaborated by the fact that there is a growing consensus for IS implementation success (e.g. ERPs), by adopting a teamwork approach so as to **integrate, collaborate** and **share their expertise on technology based platforms** to capture information so as transform and exploit knowledge (Sarker & Lee, 2003; Faraj & Sambamurthy, 2006; Fawcett et al., 2007; Schoenherr & Swink, 2012). Though, it cannot be excluded the **key role of the human element** in all problem solving and critical thinking abilities, since humans are able to process information with an amazing efficiency and often perform better than the most highly sophisticated machines (Halpern, 2003; Kuhn, 1999).

In view of the above the IT scenario, with an environment based on the internet (referred by the cliché term as the '**Internet of Things**' (IoT)), with various interconnected systems (e.g. mobile phones, tablets, laptops, computers, etc), IT is **enabling and integrating all processes** across the SC, together with the customers' engagement, to promote enhanced products and processes, such as fast 'flow of

information', 'automation' based activities and 'postponement' activities (Bowersox et al., 2000; Frohlich & Westbrook, 2002; Parker & Anderson, 2002; Kagermann et al., 2011). Such a stance, with the advent of new technologies Kagermann et al. (2011, p. xiii) advocates that: 'From management's perspective, this emerging capability will necessitate **dismantling and rebuilding value chains**'. That is, the process which was used yesterday may be different from that of today, due to the advent innovative technology based solutions leading to new competitive strategies, which if deployed effectively, will add value to the overall process and also achieve a competitive edge, such as within the ERP II scenario (Clegg & Wan 2013), such as innovative trends through e-business; contracting out of operations; efficiencies through networking across the SC; and globalised services/manufacturing such as 3PL and 4PL). Such a transformation environment brought about by technological changes is also substantiated by Germain & Droge (1998), since they advocated that technology is creating the need to change the organisation structure.

2.6.8.3 IT creates a competitive edge: A RBV and Learning Theory perspective

From the RBV of the firm (Wernerfelt, 1984; Barney, 1991) and the extended RBV beyond the focal firm (Matthews, 2003), it is being outlined that the SC flexibility as a competitive edge is being created not from the investment in technology, but how the technology is being used, through its dedicated information sharing mechanisms (e.g. ERP customisation of the integrated processes to improve on the supplies, production, logistics and product flexibility), both within and across the SC, which are treating both the **information management** and the **IT infrastructure capabilities as resources**, which may be distinct from other SCs (Prajogo & Olhager, 2012; Schoenherr & Swink, 2012). From the learning theory perspective, IT investment provides the capabilities to **exploit the organisation current performance** (e.g. improves the efficiencies of information managed processes) and/or **explore new innovative processes** (e.g. undertaking new technological based investments) (March, 1991). Such two organisational actions to achieve competitiveness are referred as the exploitation and exploration initiatives respectively (Ibid).

IT has been recognised in the last two decades as a source of competitive advantage due to provision of systems that **assist organisational processes** for developing and deploying relational capabilities (Bharadwaj, 2000; Hult et al., 2002; Paulraj et al., 2008; Sanders, 2008; Gunasekaran et al., 2008). At the same time, with integrated information systems deployment between SC actors, they are enabling the focal firm to be informed from a seamless approach on suppliers' poor performance, unpredictable customers' demands and uncertain business environment (Bayraktar et al., 2009). Though, it cannot be excluded that IT is not only an **enabler role** but may be an **inhibitor**, due to outdated legacy systems, inadequate information systems and security risks due to hackers, and where IT is treated as solution rather than as an enabler (Fawcett et al., 2007, 2008). Other inhibitors are resistance to

change from employees, lack of IT vendor support, lack of integration commitment between SC actors, incompatible and disparate systems and shortage of skills (Bayraktar et al., 2009). Attention shall also be given on all IT investments since it does not mean that there will be a correct approach in IT usage (Sanders, 2008).

2.6.9 Competitive capabilities theme

2.6.9.1 Sustainable competitive capabilities within the focal firm and across the SC from a RBV perspective

Porter's seminal work in strategy literature, the **competitive forces model** (Porter, 1980), outlines how a firm, through its activities, can achieve a competitive strategy within the industry by analysing all the forces associated with the market sector and the environment to achieve a **strategic fit**. Porter's **value chain model** is another model that explains how a firm needs to think strategically about the activities involved in the business to assess their costs and their differentiation to promote value-added at a premium price and create a competitive advantage (Porter, 1985). The value is defined as 'the amount the buyers are willing to pay', where the firm to make a profit needs that the 'value-activities' are greater than the overall cost of production (Porter, 1985, p.38). Such sources of competitive advantage can be attributed to various activities, where with a **holistic view of the firm**, the set of activities will sum-up to an overall **competitive advantage**, by creating a differentiated value proposition such as pricing, branding, communication and promotion (Porter, 1996; Juttner et al., 2007). From a manufacturing perspective, such competitive approach leads to a competitive strength relative to its primary competitors in the market (Ward et al., 1994).

The sustainable competitive advantage or the above-average performance, within the industry can be created by adopting the strategic choice from the **three generic strategies**, as referred by Porter's Generic Competitive Strategies matrix (Porter, 1985), as either cost leadership or differentiation or focused strategy respectively. The competitive advantage assesses the competitive strategy (Porter, 1985). Porter further argues that 'the enduring competitive advantages in a global economy lie increasingly in local things - knowledge, relationships, motivations - that distant rivals cannot match' (Porter 1998, p.78). Nowadays, from the extant literature, sustainable performance or **sustainable competitive advantage** not only refers to an economical perspective but also from a social and environmental perspective (i.e. triple bottom line) (Elkington, 2002; Luken & Stares, 2005; Lee, 2008; Lee & Klassen, 2008; Glover et al., 2014; Hsueh, 2015). The environmental perspective alone has become a source of competitive advantage (Vachon & Klassen, 2006; Ketchen & Hult, 2007; Hsu et al., 2013; Caniels et al., 2013), as referred in Section A4.1.2, since the sustainable SC is able to differentiate itself from other SCs through more social responsible practices in both resources usage to

protect the environment and also in the treatment of employees (Buyukozkan & Berkol, 2011). The sustainable value-added approach that leads to a competitive edge within the focal firm is substantiated by the **RBV of the firm**, where such distinct resources cannot be matched or replicated by competitors due to the four criteria: value, rareness, imperfect imitability and non-substitutability (Wernerfelt, 1984; Barney, 1991). Similarly, within the SC context, the added value by all SC actors in all sustainable measures is also in line with the **extended RBV of the overall SC** (Matthews, 2003) which also leads to the overall SC competitive edge.

2.6.9.2 SCM competitive capabilities from the value-chain model perspective: Vertical integration or outsourcing (virtual integration)

The value chain concept is considered the root of the supply chain management theory within the extant literature (Mascarenhas et al., 2004; Ramsay, 2005). The activities within the value-chain are inter-related to assess their value and costs to promote value-added operations in all the processes (Fawcett & Magnan, 2004) (e.g. expensive raw material needs to adopt lean manufacturing throughout the process to safeguard efficiencies). The investment in technology within the organisation functional areas and outside the organisational boundaries to provide relevant, timely and accurate information is a source of competitive capabilities (Fawcett et al., 2007; Bayraktar et al., 2009). This scenario is also substantiated by Porter & Heppelmann's (2014) study on the use of smart technology within both manufacturing processes and products, where the competitive strategy results in a set of competitive advantages attributed to the IT-driven transformation through smart connected devices providing three perspectives: physical, smart and connectivity with the provision of IoT. The four areas of advantages are: monitoring, control, optimization and autonomy. Such value-activities are referred as the 'linkages within the value-chain' which promote competitive advantage through optimization and coordination (Porter, 1985, p.48). Within the SC context, the SC consists from various stakeholders who need to link all activities in line with the value-chain approach, to promote effectiveness and efficiencies and to promote a value-added approach in all chain of activities, referred by Porter (1985, p.55) as the 'vertical scope'. The competitive advantage attributed to the competitive capability, through a vertical scope (Porter, 1985), complements the well established term in SCM literature, referred as vertical integration or vertical linkages (Swink et al., 2007; Guan & Rehme, 2012). Furthermore, such a vertical integration approach may be one of the options that to manage effectively the firm, needs to **increase ownership** across the SC, so as to achieve growth and efficient SC operations (Guan & Rehme, 2012) whilst for others there may be the need to outsource to achieve the same objectives (Frohlich & Westbrook, 2001; Kroes & Ghosh, 2010) and deploy virtual integration.

The **horizontal linkages** of the value-chain activities are attributed to the internal integrated activities performed by the organisation to promote a competitive advantage (Porter, 1985). These integrated activities also constitutes the principle of SCI across the SC, with the difference that the activities' values and costs need to consider all the **SC members' activities** that contribute to the overall SC process to promote a win-win mindset leading to competitiveness and profitability (Dyer, 1994; Lambert et al., 1998; Ketchen & Hult, 2007; Swafford et al., 2008; Flynn et al., 2010; Zhao et al., 2013). The competitive advantage is achieved by the **integration** of the internal core competencies of the focal firm with those of the supplier, customers and other stakeholders' resources appropriately (Costantino & Pellegrino, 2010).

2.6.9.3 Competitive capabilities through quality management: A RBV perspective

Furthermore, the quality based established procedures, such as ISO 9001 standard, has become a prerequisite and in-built within the internal management processes to interact effectively with all stakeholders from a holistic approach (Singh et al., 2011). As a result going beyond 'basic' implementation quality standard to the 'advanced' and 'supportive' implementation stages (Prajogo et al., 2012; Huo et al., 2014) creates a competitive advantage over other SCs, through both implicit (e.g. effective information management capabilities) and explicit (e.g. state-of-the-art automated equipment) resources. Such a competitive advantage is achieved by the firm-specific and difficult-toimitate resources, such as physical assets (e.g. dedicated equipment), financial (e.g. improved cash flow), intangible (e.g. brand name) and human resources (e.g. distinct people skills), as outlined by the **RBV** of the firm (Wernerfelt, 1984; Barney, 1991; Teece et al., 1997; Green & Inman, 2005; Kim, 2009). Furthermore, Teece et al. (1997, p. 516) outlined the importance that the competitive advantage needs to result in dynamic capabilities to meet the flexibility needed in the marketplace, by stating that: 'The firm's ability to integrate, build and reconfigure internal and external competencies to address rapidly changing environments'. Such a dynamic capabilities are needed in today's changing, innovative and competitive environment. The external stance, from a SC perspective, was also reinforced by (Porter, 1987) as the cross-business collaboration, which is also in line with the extended RBV of the overall SC (Matthews, 2003).

Such an advanced and management support in **quality implementation** stages are a step forward to ensure that all SC stakeholders' interacting processes with the focal firm, at both upstream and downstream, go **beyond** the established ISO based quality practices, to promote proactive and innovative actions for **improved operational performance**. SCM is considered as a crucial operations strategy for organizational competitiveness in the twenty-first century, with the direct involvement of people (Gunasekaran et al., 2008). The new context associated with SCM, is that

today competition is not focused between firm to firm competitiveness, but on entire supply chains (Dyer, 2000; Christopher & Holweg, 2011).

2.6.9.4 Competitive capabilities from an individual assigned priority or from a cumulative approach

Initially **competitive capabilities** were considered as mutually exclusive and still a manufacturer needs to be selective on conflicting priorities such as flexibility and delivery (Skinner, 1985) but afterwards he re-stated his position that such competitive capabilities are based on a **dynamic trade-off** (Skinner, 1996). There are other sources in the literature that substantiate that competitive capabilities may be achieved by a firm in order of an **assigned priority** (e.g. Noble, 1995) and others who declare that firms can achieve very well in a multiple set of competitive capabilities (Flynn & Flynn, 2004) and furthermore that they are **mutually reinforcing or cumulative** (Swink et al., 2005). In fact, it is also well cited in the literature that operational or manufacturing capabilities lead to business performance either in **individually** or **in concert** (Vickery et al., 1997; Flynn et al., 1999; Rosenzweig et al., 2003; Swink et al., 2007).

The current frequently cited SC key **operational competitive capabilities** for the focal firm, which are also applied at both the upstream and downstream sides of the SC are: **cost, flexibility, speed/delivery and quality** (Hayes & Wheelwright, 1979b; Frohlich & Westbrook's 2001; Chen & Paulraj, 2004; Swink, et al., 2007; Ketchen & Hult, 2007; Ketchen, et al., 2008; Kim, 2009; Richey et al., 2010; White et al., 2010). Similarly, the financial or **business competitive capabilities** are **sales, market share and growth** (Johnson, 1999; Flynn et al., 2010) and **return on assets, return on sales and return on investments** (Vickery et al., 2003; Rosenzweig et al., 2003; Flynn et al., 2010).

2.6.9.5 Competitive capabilities direct/indirect relationship to SCI operational performance outcomes

From the literature, the evidence of the effects of the mediating variables of SCI within the internal focal firm are conflicting and other times differ. For example, innovation and operational performance directly promotes business performance improvements (Vickery et al., 2003; Koufteros et al., 2005). Other scholars' evidence supports also a direct relationship of internal SCI to business performance improvements (Frohlich & Westbrook, 2001; Swink et al., 2007; Bayraktar et al., 2009; Zhang & Huo, 2013). Similarly, internal integration evidence also differs on the effect on the operational performance since it can have both direct effect (Saeed et al., 2005) or an indirect effect (Koufteros et al., 2005; Gimenez et al., 2012). Furthermore, Schoenherr & Swink (2012) inferred, by distinguishing

between the four operational capabilities and stating that internal integration has no moderating effect on quality and cost dimensions, but it is affirmative on delivery and flexibility dimensions. Such a difference in the research results may be attributed to the fact that SCI has three dimensions (i.e. supplier, customer and intra focal firm), which all need to be aggregated together to have consistent outcomes (Swink et al., 2007) or else it could be attributed to the different dimensions and variables used to measure the SCI construct (Alfalla-Luque et al., 2012).

Though, it cannot be excluded from the literature, that internal integration is the foundation of **SCI initiatives** across the SC and that it also provides the bridge between the supplier and the customer integration initiatives to promote **performance improvement** and/or **operational capabilities** which in turn leads to improved business performance (Frohlich & Westbrook, 2001; Koufteros et al., 2005; Swink et al., 2007; Bayraktar et al., 2009; Flynn et al., 2010; Schoenherr & Swink, 2012) to achieve competitive capabilities (Rosenzweig et al., 2003; Bayraktar et al., 2009).

2.6.10 Literature gaps from the post literature review

2.6.10.1 Leadership literature gaps

The established leadership studies have significant relevance to each SC actor, from a leader-follower approach, applicable for each SC tier micro organisational set-up (i.e. from an internal organisational perspective). But there are few scholarly works which explicitly considers network based leadership in the macro context of inter-organisational networks (Sydow et al., 2011), such as supply chain set-ups (i.e. leadership approach adopted by SC members across the SC). The leadership domain within the 'management of integrated supply chain' is a 'largely ignored area' (Defee at al., 2010, p.764). They additionally declared that:

'This perspective suggests that supply chain integration may be partly due to the active leadership and influence of one organization within the supply chain. Interestingly, however, no theory currently exists that explains how a firm becomes the supply chain leader and maintains that role over time. Moreover, the role and impact of the other "follower" organizations within the supply chain has been virtually ignored in the context of their alignment and support of the supply chain leader' (Defee at al., 2010, p.764).

Also, Lambert et al. (1998, p.12) assert that 'The managerial and behavioural components were, in general, less well understood and more difficulties were encountered in their implementation', relative to the physical and technical management components (e.g. organisation structure, communication and information flow structure, etc). Stonebraker & Afifi (2004, p.1142) also contend that there is the need to investigate the 'leadership proactiveness' within the SC context. Furthermore, such a challenging SC management approach has also been queried by Storey et al. (2006, p.769), by

questioning who shall manage the SC from all the SC actors, by stating that: 'Who could and should have this responsibility?'

The role of people is pivotal in all SC management and leadership implementation initiatives, in fact the literature indicates that the role of people is often not explicitly addressed within SCM (McCarter et al., 2005) and furthermore there is lack of research on the integration between the HRM and SCM disciplines (Lengnick-Hall et al., 2013). Such lack of significance given to the human element, within the SCM literature, can also be relatively seen when compared with the literature dedicated to the 'green' dimension within the sustainable SCM perspective, since the latter takes a pivotal role in various scholarly works (Ashby et al., 2012). But with respect to the human element, there exist 'no equivalent use of the social element' despite the key role the people have in all needed skills and in the formation of all relationships across the SC (Ashby et al., 2012, p.506). Finally, Shub and Stonebraker (2009, p.32) declared that 'There is little in the literature that describes the relationship between human resource activities or organization variables and supply chain success'.

2.6.10.2 Supply chain integration literature gaps

SCI from an external perspective is lacking, since Swink et al. (2007, p.153) identified that 'past research provides little information regarding the relative and respective contributions of strategic customer integration and strategic supplier integration activities' (Swink et al., 2007, p.153). Furthermore, Guan and Rehme (2012, p.188) from a study focused on the customer integration gaps, emphasized that 'Downstream integration has received little attention in supply chain research'. SCI studies within manufacturing are also limited since Swink et al. (2007, p.153) added that 'Little attention has been given to the impacts of integration on manufacturing plant competitive capabilities'. Schoenherr & Swink (2012, p.110) also contend that 'theory surrounding supply chain integration is still underdeveloped', and as a result they recommended that 'Further studies should also aim to collect dyadic or even triadic integration measures', so that research studies cover a more holistic SCI approach. Such SCI holistic gaps was also supported by another study which outlined the need that future research has to consider multiple levels of firms relationships (Pero et al., 2010). Most of the SCI scholarly work are focused only on the dyadic level (Kang & Kim, 2010; Xiao et al., 2010; Dass & Fox, 2011), which as a result are causing lack of literature associated with the multiple direct and indirect relationship with all SC members (Holweg & Bicheno, 2002; McFarland et al., 2008; Dass & Fox, 2011). Finally, Cheung et al. (2012, p.3907) emphasize on the importance of deploying knowledge and not only information to improve on SC performance, by stating that 'Most of the previous research work for supply chain integration is information driven instead of knowledge

driven'. Bessant et al. (2003) also inferred such a lack of knowledge management within SCM literature.

2.6.10.3 Sustainability literature gaps

Most of this research within the SCM literature with respect to sustainable practices has been fragmented and considered single activities in isolation (Svensson, 2007; Gold et al., 2010; Glover et al., 2014), and as a result scholarly studies need to consider the triple bottom line measures from a holistic perspective. SC actors, such as suppliers, need to be engaged in such a green SC initiatives in-conjunction with the focal firm, which has also received little attention by scholars (Caniels et al., 2013). It is considered pivotal that SC related scholarly work need to engage Green Supply Chain Management (GrSCM), in fact it was stated that: 'further research in the integration of environmental management with ongoing SC operations' is being recommended in recent works associated with environmental performance (Bojarski et al., 2009, p.1747).

2.6.10.4 Manufacturing literature gaps

The manufacturing sector needs further research in the area of 'holistic research approach integrating technological, managerial and business-model solutions together with a change in the customer behaviour, with the perspective of a more sustainable manufacturing for the production of more sustainable products and services' (Garetti & Taisch, 2012, p.88).

The outcome attributed to SCI within manufacturing is lacking, since as referred in Section 2.6.10.2, it was stated that: 'Little attention has been given to the impacts of integration on manufacturing plant competitive capabilities' (Swink et al., 2007, p.153). In view of the dynamic changes within manufacturing from an operations management perspective, there is lack of literature to empirically establish the foundations of the paradigm shift on how the service provision is being put into practice within the HVM context (Wilkinson et al., 2009). Such a perspective, was substantiated by Macbryde et al. (2013, p.1594) by stating that 'there is little empirical evidence indicating whether firms in developed countries are indeed changing their mode of operation and moving into HVM' and they also added that 'These gaps in knowledge are particularly important in relation to smaller firms which are largely ignored in the existing literature, and yet form a significant part of so many developed economies'. They finally added that manufacturing success needs a strategic approach over the traditional efficiency focus, by declaring that a 'more holistic approach is required bridging the gap between higher level strategic work and the detail of operational activity'.

2.6.11 Summary of the a priori and post literature reviews

The a priori literature review (Section 2.2) is mainly focused on SCI and IT within the manufacturing SMEs. Such an a priori literature review is carried out before the initiation of the data collection and analysis process, so as to inform the researcher on the tentative research objectives, by taking a broad perspective on the theory underpinning SCI. Such a step situated the researcher with a good understanding of the key SCI issues and challenges from the extant literature, as referred in Section 2.5, supported by its summary in Section 2.5.12. Such a literature review also enabled the researcher to filter out the literature gaps (Section 2.5.9 & 2.5.10) so as to formulate the tentative research question (Section 2.5.11). The literature review with its established significant concepts (Appendix 2) also assisted the researcher to design the open-ended interview guide, to kick-off the research initial data collection and analysis (Appendix 3).

The post literature review (Section 2.6) is carried out with the completion of the data collection and analysis and after establishing the emerged theory, as referred in Chapters 5 and 6. Such a literature review objective is to establish the contemporary theory on all the emerged themes and the contemporary literature gaps respectively, so that the researcher is in a position to establish the commonalities and the differences between the emerged theory and literature, so as to derive the research contribution, as referred in Chapter 7.

The post literature review, as guided by the emerged theory, outlines that the business strategy needs to set-up a manufacturing strategy in-conjunction with the SC strategy, by involving the right SC actors and employ an integrated approach to develop all operations into a competitive weapon across various elements of performance (e.g. innovation through improved products and processes; low cost of operations through efficient processes; and low purchasing costs of all raw materials). The business strategy places demands on the manufacturing strategy to implement the necessary production plans, with all sustainable measures, from the triple bottom line perspective and quality standards, to meet the competitive capabilities (e.g. products low costs, operational cost cutting measures, high quality levels, sustainability and R & D initiatives, in both processes and products).

The focal firm cannot operate in a vacuum and needs to set-up an effective SC strategy, in-conjunction with relevant SC actors, to achieve a holistic SCI management approach to achieve competitive capabilities. Such a holistic process approach is enabled by technology deployment, in both manufacturing automation and information systems. The manufacturing technology is deployed to improve the efficiencies of all processes within the focal firm through automation. The information systems provide an information sharing platform to exploit all resources, such as information and knowledge, both within the focal firm and across the SC, through improved synergies, so as to

improve in both manufacturing (e.g. lean and agile manufacturing) and SC operations (e.g. real-time information sharing) to promote SC flexibility, responsiveness and visibility.

The SCI approach needs to ensure an internal integrated approach in all its strategies and tactics to be in a position to replicate such an integrated approach beyond the focal firm with all SC actors, to achieve an overall SC competitive performance. The holistic process approach, also needs a innovative SC strategic approach, to build on the manufacturing strategy innovative initiatives, to promote a continuous improvement approach in all processes, both within the focal firm and across the SC, by the involvement all SC actors. The holistic process approach needs to employ several measures and initiatives which involves all SC actors to unify together all SC stakeholders, which are to be based on a set of inter-related concepts since SCI is a multi-dimensional and cross-disciplinary concept, where one concept influences another and vice versa. Such a complex set of concepts are in line with the emerged theory themes under the holistic SCI management approach, as referred in Table 5.1, but there exists limited details on what constitutes such a holistic process approach, as already referred in Sections 2.6.7 and literature gaps in Sections 2.5.9 and 2.6.10.2.

The next chapter outlines the methodology, where it explains the role of both literature reviews and also the overall research process based on Straussian GTM, with its procedures and techniques in the data collection and analysis design and implementation together with the researcher's philosophical research paradigm to generate a valid and reliable substantive theory.

3.0 Chapter 3: Research Methodology

3.1 Introduction

This chapter provides the research methodological outline of the overall research process, by including the Straussian methodology Grounded Theory Method (GTM) (Strauss & Corbin, 1998) selection as the preferred option over the Glaserian GTM (Glaser, 1998), and also an overview of all the procedures and techniques within the GTM, with a representation of all the steps in a visual display. Justification to use the GTM within the supply chain management (SCM) discipline is also referred. The GTM seminal works are outlined together with the researcher's philosophical paradigm vis a vis the epistemological and ontological aspects for generating valid knowledge. The research quality criteria is referred to promote valid and reliable research and with all the ethical safeguards in line with the ESRC guidelines. The roles of the a priori and post literature reviews are referred within the research process. Finally, the research process itself is outlined, by explaining all the implementation measures taken in all sampling and data collection and analysis processes to promote scientific research rigour in line with the GTM research quality criteria.

3.2 Research methodology justification

Operations Management (OM) is an applied, problem-oriented and cross-disciplinary field (Amundson, 1998). OM based academic research needs to contribute to theory development from a multi-discipline perspective to produce relevant research within the relevant academic fields and which must also be accessible and understood from the practitioners' perspectives. The purpose of theory is to define, establish and explain relationships between concepts or constructs which forms up one element of scientific knowledge (Amundson, 1998). OM borrows from other fields for understanding the nature and purpose of theory such as sociology, economics, information and organisational theories among others, which makes the discipline as cross-disciplinary (Croom & Romano, 2000; Parente et al., 2008).

SCM falls within the OM discipline which lacks theory construction (Frohlich, & Westbrook, 2001; Pilkington & Fitzgerald, 2006) and is looking for less hypothesis testing (i.e. statistical methods) and more systematic observation through qualitative research approaches, due to the SCM innovative nature of all processes, both within and outside the focal firm (Fisher, 1997; Hayes, 2000). The researcher's cognitive lens adopted is based on Supply Chain Integration (SCI) with a multidisciplinary approach, mainly based on SCM and Information Technology (IT).

The research objective is to determine and explore how the focal firm, in-conjunction with all the stakeholders, within a SC set-up, is managing a holistic integrated approach between all players. The aim of the research is to derive a grounded theory based on a set of interviews, as the main source of the data collection approach, so as to establish inductively how the focal firm is achieving an integrated approach, within the focal firm and across the SC. The research question is to establish all the key factors and a theoretical framework that constitutes SCI within manufacturing SMEs to promote competitive performance.

The aim of the research is to select the most adaptable method to answer the research question, as referred by Bouchard (1976, p.402) who states that good research needs 'picking the most powerful method for answering that particular question'. Furthermore, the research methodology also needs to consider the epistemology and ontology, to establish the researcher's philosophical paradigm, since one cannot interpret research and generate knowledge from different paradigms within the same research method, due to different worldviews, not to risk research inconsistency as a result of the paradigmatic incommensurability (Lee & Lings, 2008).

Taking the two extreme ends of social science epistemologies, namely positivism and social constructionism, the research methodology has a set of determined characterisitcs depending on the philosophical approach of the researcher, with respect to the: research aim type (i.e. discovery or construction of reality); research techniques (i.e. measure or conversation); research analysis (verification/falsification or sense-making); and research outcome (i.e. casuality or understanding) (Easterby-Smith et al., 2008). The researcher's approach has adopted a research aim that constructs theory, as referred in Section 3.6.1. The research technique is conversation with the participants to generate data that outlines the theory under the practitioners' actions. The research design is sense-making by the on-going memos with constant comparison of data, during the data collection and analysis process, as referred in the three coding stages, within Section 3.10. The outcome is about the understanding what the data are saying, throughout the data collection and analysis process, to derive the substantive theory to promote understanding, to both the academics' and practitioners' communities, as referred in Section 3.9.

The decision to approach a study quantitatively or qualitatively depends on the research questions driving the study, prior work, the planned research design and the desired contributions the researchers wish to make (Edmondson & McManus, 2007). The suitability of the qualitative research strategy chosen by the researcher, over the quantitative counterpart approach, is valid since the research objective is exploratory with the objective to establish the characteristics that constitute the SCI phenomenon within the manufacturing sector. Such a qualitative research approach needs to

understand the meaning and knowledge from the data given by practitioners. The research boundary conditions are targeted to manufacturing SC, to ensure that the research is manageable and to engage in more focused research. SCI is based on the interactions between various SC members, which are all characterized by constant and dynamic technological changes and innovations, in both the manufacturing and SC processes and the manufactured products. The SCM discipline is still a developing management approach, especially in integration initiatives with the current developments, which is causing SCI to take an emerging paradigm shift (Pagell, 2004; Storey et al., 2006; Zhao et al., 2008; Schoenherr & Swink, 2012). With such literature gaps, a qualitative research approach is well suited due to the richness of quality data needed within the data collection process (Stern, 2007).

The research question and objective is to establish a broad perspective of the SCI and is therefore more practically reached by a qualitative methodology, to capture such a holistic phenomenon, where: emphasis is on exploration; it is naturally occurring and contextual; it is inductive; it is subjective; data is expressed in words; the researcher participates and collaborates with the participants; and it undergoes text analysis (Rudestam & Newton, 2007). It was further claimed that 'the frankness of the conversational responses reveal some of the depth that would be unlikely to be gained through a survey instrument' (Harland et al., 2007, p.1249). It is to be outlined that the GTM goes a step further than text analysis, since it is a method that digs deep within the text to undergo theoretical abstraction in an iterative process, with an 'interplay between induction and deduction' through constant comparison of the data to arrive at the emerged theory (Strauss & Corbin, 1998, p.137).

Qualitative theory building approaches, 'such as grounded theory method (GTM), are still not very widespread and rigorously applied in operations management (OM) research' (Binder & Edwards, 2010, p.232). Though, the GTM is well mentioned in information systems to discover process oriented descriptions and explanations within a context (Myers, 1997; Bryant, 2002; Urquhart et al., 2010). Such GTM applications substantiate the researcher's selected choice, since the research focus is mainly based on both the SCM and IT disciplines.

The researcher has used the GTM to focus on qualitative data but it is normally associated with both qualitative and quantitative data (Glaser, 1998). The researcher's statement that he is using a GTM based on a qualitative research method is a valid declaration, since such an option was taken because of the 'ease of highly meaningful data collection' (Glaser, 1998, p.29). The researcher also understands that no method can deliver the ultimate truth, although some methods are more suited than others for conducting research, no one can argue that a 'single method -- or a collection of methods -- is the royal road to ultimate knowledge' (Guba & Lincoln, 2005, p.205). The grounded theory approach, with such a research question, demonstrates consistency between the research questions and the methods used to answer these questions (Suddaby, 2006; Binder & Edwards, 2010).

3.3 Methodology selection

The research objectives' requirements led to the selection of a qualitative research method with a constructivist epistemology, based on the Grounded Theory Method (GTM) (Strauss & Corbin, 1998). The researcher's approach to the GTM is mainly governed by both the Glaserian (Classical) (Glaser & Strauss, 1967; Glaser, 1998) and the Straussian (Strauss & Corbin, 1998) grounded theory seminal works. Though, the researcher had to choose one from the two options and as a result preferred the Straussian GTM for its systematic process (Strauss & Corbin, 1998). This selection did not exclude the researcher from including any key reference to the Glaserian GTM within the research work, since both methodologies constitute some common roots but have some pivotal in-built differences.

The main differences are that the Glaserian methodology (Glaser,1992) outlines that: first, the research focus shall not be established prior to the research data collection and analysis, but needs to emerge from the analysis itself; second, that there is no need to review the extant literature before the research commencement, to ensure no preconceived ideas to influence the research process; and third, the approach of emergence shall be left open, with no systematic procedures, not to force the emerged theory out of the data. Whilst the Straussian methodology (Corbin & Strauss, 2008) outline that: first, an initial broad focus research question is required to guide the research process; second, undertaking an a priori literature review enhances the researcher's theoretical sensitivity and can be used to establish the research question/s; and third, the researcher can use analytic tools and techniques, such as the coding paradigm and CAQDAS, to assist the systematic theoretical conceptual process to inductively derive the emerged theory.

3.4 The Grounded Theory approach: Straussian methodology overview

The Straussian GTM research design is being outlined to demonstrate the systematic research approach adopted to promote research rigour. The researcher deploys the three Straussian GTM coding techniques, referred as Open, Axial and Selective Coding respectively. The data collection is based on two sampling techniques to generate the data. The GTM data management tools are based on both the CAQDAS (Computer-Assisted Qualitative Data Analysis Software) and the storyline techniques, supported by memos and visual aids all along the research process, in particular, logic diagrams as tables, code matrices as tree diagrams and conceptual frameworks as mind maps, to promote the research audit trail of all the data analysis. The researcher's constructivist philosophical stance is used, since it accepts the researcher's active engagement in the research data collection and analysis process, to generate valid knowledge. The research quality criteria is considered to promote validity and reliability in the generated substantive theory and in all the research process.

The research methodology process consists from the following steps, which although are depicted as a sequential process, the three coding stages within step 6 were of an iterative nature, which drives the researcher to move back and forth during the research process to build on theoretical sensitivity in the data, so as to promote research rigour. The key steps within the research methodology are referred in Table 3.1:

Step	Research methodology strategic steps	Research methodology description and scope
1	A priori literature review	Established gaps and research questions; enhanced the researcher's theoretical sensitivity; and designed the initial open-ended interview guide.
2	Research strategy	Qualitative approach to meet the inductive and exploratory nature of the research question.
3	Research method	Grounded Theory Method (GTM) to generate grounded theory based on the Straussian GTM.
4	Data collection	Mainly based on interviews; and supported by direct observation notes by the researcher and documents provided by participants and other accessed data sources (i.e. triangulation of data sources).
5	Sampling techniques	Open sampling (i.e. based on the first interview guide derived from the literature) and theoretical sampling (i.e. based on second interview guide as emerged from the data).
6	Data analysis	Open, Axial and Selective Coding stages with dual data analytic approaches based on CAQDAS and Storyline techniques supported by memos, logic diagrams (i.e. tabulations); conceptual frameworks (i.e. mind maps) and code matrices (i.e. tree diagrams). Such triangulation of analysis techniques assisted the theorising process through constant comparison so as to achieve theoretical saturation.
7	Research philosophy	Constructivist paradigm.
8	Research quality criteria	Fit (Credibility), Generality (transferability), Understandability (dependability) and Control (confirmability) (Glaser and Strauss, 1967).
9	Ethics	Ethical considerations in line with ESRC framework.
10	Theoretical account of the emerged theory and the conceptual framework with a set of theoretical propositions	Generated grounded theory with a set of key categories (i.e. one core; a set of 5 main categories with 16 sub-categories) and a conceptual framework.
11	Theoretical integration	Situated the generated substantive theory within the extant literature to derive the commonalities and the differences and the contribution.

3.5 Seminal works associated with the Grounded Theory Method (GTM)

The GTM was initially articulated in Glaser and Strauss's (1967) book 'The Discovery of Grounded Theory', after deriving a systematic analysis of terminal ill patients, with the objective to be used by social scientists in other topics to generate theory (Eisenhardt, 1989; Urquhart, 2001; Suddaby, 2006), without the need to have an a priori hypothesis (Glaser & Strauss, 1967). Over the years, the GTM developed into two distinct variants, one favoured by Glaser, referred as the classical GTM and the other by Strauss, referred as the Straussian GTM. A disagreement between the two sociologists, as referred in the Strauss and Corbin's book in 1990, caused the GTM to take two different approaches.

In the opinion of the researcher, although the Straussian GTM is being used in this research, both Glaser and Strauss seminal works, dated back in the Glaser and Strauss (1967) book, still represents the solid foundations of the GTM, although other books that followed, by both authors and even others, refined and clarified the principles, through their different approaches, associated with the research process to promote research rigour (e.g. Strauss, 1987; Glaser, 2002; Charmaz, 2006; Bryant & Charmaz, 2007). Such key common foundational concepts are: Theoretical Sensitivity; Constant Comparison; and Theoretical Sampling. In retrospect, the researcher wish to make it clear that primary attention during the research process is given to the Straussian GTM, but it is not being excluded that some concepts or techniques will be used from the Classical GTM and even other scholars, backed with the relevant reference to the sources.

A key issue that underpins the two GTM approaches is the philosophical paradigm that constitutes the theory generation. There was a movement away from the objectivist stance used in Classic grounded theory (Glaser & Strauss, 1967) to a position, where the role of the researcher was accepted to contribute to the sense-making process and that a multiple interpretation of human experience also contributes to derive the reality (Strauss & Corbin, 1994; Charmaz, 2000, 2006). In fact, the researcher's philosophical paradigm, as referred in Section 3.6.1, takes the constructivist GTM, which is still consistent with the Straussian GTM and in line with Charmaz (2006, p.130), where it is argued that researchers may come up with similar ideas but the theory construction may vary and hence she stated that the 'theory is dependent on the researcher's view; it does not and cannot stand outside of it', as it is treated by objectivists.

From the 'The SAGE Handbook of Grounded Theory', by Bryant and Charmaz (2007), the researcher's preoccupation on the different approaches adopted in GTM by Glaserian, Straussian and Charmaz, among others, has been completely eliminated, since it has been clearly outlined, that the GTM is a family of methods, where the researcher needs to make informed choices and needs to

articulate the rationale to support the choice (Bryant & Charmaz, 2007). This means that the researcher is allowed to adopt the key GTM foundations, referred by both Glaser and Strauss, as the main scholars who contributed to the seminal works on the GTM. As a result once any approach taken is defined and cited accordingly, then the methodology is considered to have a valid research design strategy. The main reason for the researcher to mainly deploy the Straussian GTM, is the fact that such a methodology employed a systematic approach and accepts an a priori literature review (Strauss & Corbin, 1990; Corbin & Strauss, 2008), whilst the Classical grounded theory methodology is open, flexible and parsimonious, where it focuses on latent structural patterns that constitute the emerged theory (Glaser, 1998).

3.6 Research philosophical approach and research quality

3.6.1 Epistemology

Epistemology defines the validity of the generated knowledge. The participants' statements (e.g. description of their actions) are analysed to construct out of such data the key themes. The term construction comes from a constructivist epistemology. The theory generation process is not only emerging and being discovered (Glaser & Strauss, 1967) but is being constructed from the emerged data, with the active engagement of the researcher's interpretation from the views of the participants through the transcribed data. This constructivist approach has been emphasized by the statement that the: 'mutual creation of knowledge by the viewer and viewed aims towards interpretive understanding of subject's meaning' (Charmaz, 2000, p.510). The data through the transcripts represents an account of the way reality was at the time of the interview, where participants creates their own meaning on the SC phenomenon based on their experience and knowledge (Corbin & Strauss, 2008). The researcher is engaged in a number of processes, under the GTM, such as constant comparison, Open Coding, storyline memo writing, Axial Coding and Selective Coding, which are all function of the researcher's reflexivity, which means that the analysis may also be coloured by the researcher (Strauss & Corbin, 1998; Charmaz, 2006). The researcher's reflexivity plays a key role in the way the analysis is undertaken, since it determines the researcher's abilities of creativity and discovery to construct theory from the data, referred as theoretical sensitivity (Strauss & Corbin, 1998; Corbin & Strauss, 2008).

The researcher cannot imagine to use the Classical GTM, since he cannot understand how it is possible to set aside all his previous knowledge during the research process to begin the research with a 'tableau rose' (Glaser, 1998, p.70) (i.e. tabula rasa or clean slate). Such a theoretical sensitivity is also highly influenced by the researcher's work experience, of around fifteen years in the industry and

with his professional knowledge on the subject matter and also as an academic for the last nine years). Though, the researcher considers that objectivity can be achieved through rigour, impartiality and consistency (Patton, 2002).

3.6.2 Ontology

The ontology of the research stance adopted is based on the condition that the physical and social reality is subjective with a multiple reality and context dependent. This means that reality is constructed by the science itself within a reality which is 'unstable, constantly changing and unavoidably subjectivity' exists (Lee & Lings, 2008, p.60). The researcher, being a human, during such an activity, cannot stand apart from the experience (i.e. be impartial). The human element cannot separate the self from the social context, leading to the researcher experience to be unavoidably influenced by such a context (Lee, 1999).

The scientist is no longer value free and objective in the data collection and analysis process, but reflects the personal values or beliefs, such as preferences, experiences and academic background. 'A God's eye frame of reference' is not fully possible (Van De Ven, 2007, p.45). Furthermore, particular attention should be given to objectivity (impartiality) of the information obtained during the qualitative research methodology, since as Lee and Lings (2008, p.120) refer, 'humans have a tendency to place more importance on information that supports their existing views than information which challenges those views'. Such an approach needs to be avoided, so as to maintain the rigour and validity of the findings by undergoing the constant comparison method on various participants' accounts, to derive the similarities and differences in the emerged theory with an audit trail of all the emerged theory. Furthermore, the CAQDAS and storyline techniques are used for data analysis, to promote a dual data analytic method triangulation, during the data collection and analysis processes to achieve trustworthy research by studying the same phenomenon through a different technique. Such research validity is further enhanced by including slices of data from polar or extreme (deviant) cases, which sustain cases that go against the norm, which further amplifies the variation in the emerged theory (Yin, 2009). With the implementation of such measures within the methodology introduced a level of robustness to make up for the researcher's influence on the interpretation of the data (Glaser & Strauss, 1967, Strauss & Corbin, 1998).

3.6.3 Research quality

The qualitative research approach has the researcher himself as the measuring instrument (Creswell, 2007). The qualitative research over the quantitative research approach can 'follow up ideas, probe responses and investigate motives and feelings, which the questionnaire can never do' (Bell, 2009,

p.157). The aim is to develop, capture or uncover substantive knowledge from the different processes and elements in the SC relationships, which can 'yield rich material and can often put flesh on the bones of questionnaire responses' (Bell, 2009, p.157), with theory building and refinement. The data is explanatory (i.e. reported in words rather than measurements or ratings) and special attention is given in the data analysis to promote rigour and to employ an engaged scholarly work and not to limit research outcomes validity and reliability. Though, the findings transferability (generalizability) properties, due to small the sample of the population under study, are characterised by 'contextual uniqueness and significance' Bryman (2008, p.378), which means that the findings are context specific and dependent and cannot be generalised to other situations due to contextual differences. The generalizability property is concerned to the theory generated grounded in data or to the 'quality of the theoretical inferences rather than to populations' (Bryman, 2008, p.392). The issue of transferability and contextual differences is further explored by Wasserman et al. (2009, p.371), where it was argued that 'Grounded theory attempts not only to build conceptual models specific to particular pieces of data, but for broader contexts'.

With such a perspective, the research outcome is being considered as more solid, since various countries across the globe, especially Small Island States, have similar contextual characteristics, such as competiveness and cultural values, but with different orientations, which may not affect the conceptual framework substantive key theory but may only influence the variations of the conceptual elements and their interactions within the overall process. Such a transferability property with its emerged hypothesis/propositions preferably needs to be substantiated through replication of the study in other countries to promote the validity of such generalisation across different geographical settings. It is to be noted that the research sample frame includes a mix of SMEs with all the concerned stakeholders across the SC, which are both locally and internationally engaged (e.g. Italy, USA, Germany and UK), which makes the conceptual substantive theory to cover a broader context with a richer generalizability.

Glaser and Strauss (1967) criteria for assessing the credibility, plausibility and trustworthiness of the grounded theory include: (a) fit; (b) understandable (i.e. understandability; (c) generality (i.e. generalizability); and (d) control. Thus the emerged theory should **fit** the substantive area (i.e. the generated propositions of the theory, fit the reality of the discipline), with the provision of being **understandable** to the people who apply the theory (i.e. the theory clearly promotes practical usage), include the **generality** principle to be applicable and flexible enough to guide problem-solving and processes implementation to a wide range of contextual situations and finally allow for **control**, by being able to undergo the necessary changes and refinements to make the theory relevant and useful over time. Corbin and Strauss (2008, p.302) declare that the two research quality criteria are

'trustworthy' and 'plausible'. Such a quality criteria approach is needed to ensure that the claims (propositions) made from the research are credible, valid, meets high standards of rigour and will advance the SCI theory in a scholarly way. The canons of good science were always a debatable subject by the research community to ensure that qualitative research maintain the standards of scientific research as quantitative research. Such a consensus has always been focused to derive the criteria needed to judge the merit of qualitative research and its generated theory.

The researcher, in view of such a multiple perspectives on the research quality criteria, also refers to Denzin and Lincoln (2005, p.24) who assert, in line with Glaser and Strauss (1967) scientific rigour, with particular reference to the constructivist paradigm, that 'credibility, transferability, dependability and confirmability, replace the usual positivist criteria of internal and external validity, reliability and objectivity'. Such an approach is also substantiated by Miles and Huberman (1994, p.277), where they established a practical list which need to be posed to promote trustworthy and authentic research, which are referred by: 'objectivity/confirmability of qualitative work; reliability/dependability/auditability; internal validity/credibility/authenticity; external validity/transferability/fittingness; and utilization/application/action oriented'. Henceforth. the researcher adopts the research quality criteria based on Glaser and Strauss (1967), as referred above, since they are in line with both Denzin and Lincoln (2005) and Miles and Huberman (1994), namely: fit matches credibility; generality matches transferability; understandability matches dependability; and control matches confirmability. All practical implications taken by the researcher for each research quality criteria are referred below:

- I. The constant comparison process applied in the data (i.e. slices of data) from both the same or different twenty-two transcribed interviews, across all the three coding stages, within all memos through the CAQDAS and the storyline approaches (Chapters 4 & 5), with narratives substantiated by *in vivo* excerpts (i.e. participants' meanings), kept the analysis grounded in the data and makes the substantive theory in line with the **confirmability** criteria by eliminating bias and preconceived concepts on the data and promotes transparency in the research analytic process, which makes the theory to take foothold in diverse situations to be **controllable** to meet the user differing requirements and to meet the dependability or fit criteria;
- II. A systematic approach is used through the CAQDAS and storyline approaches in all analytic coding stages (Chapters 4 & 5), as triangulation tools, which makes the substantive theory in line with the **dependability** criteria through an established consistent and rigorous audit trail, which promotes informed **understanding** of the substantive theory processes to be applicable by the user of the theory; and

- III. The final theory generated a substantive theory, which is specific (i.e. fit) to the discipline (Glaser & Strauss, 1967). The substantive theory is mainly focussed on the SCM and IT disciplines by building a SCI conceptual theoretical framework, with a set of theoretical hypotheses/propositions, which makes the substantive theory in line with the credibility criteria. Such a stance is achieved by ensuring that all analytic processes are grounded within the data, with the inclusion of variation within the key categories and sub-categories, as emerged from different views of all the participants through the constant comparison process, as outlined in the final conceptual storyline with its theoretical model (Chapter 6);
- IV. The emerged theory is open to development, since the iterative constant comparison process, was applied to various participants' data within different firms, which involved comparing the emerged theory with new data with every new participant. Such a stance makes the final theory tendency to adapt, or takes into account, to what the new data are saying, which makes the substantive theory in line with the **transferability** criteria, since it makes the theory to be generalizable to other contexts, once they fit or have a degree of similarity to each other.

It is to be noted that the inductive generated conceptual framework represents the theory as applied to the real world, but Lee and Lings (2008, p.123) refer to it, a 'model is a representation of the real world; but it is not the real world'. This accounts for the knowledge transfer problem (Van De Ven, 2007; Lee & Lings, 2008), which still needs to be challenged, due to the difference in the real world from the representation of the theoretical model.

As a result of the researcher's constructivist philosophical approach, in this Grounded Theory research methodology, is based on the researcher's engagement with the participants and various other sources of data, to promote data sources triangulation (e.g. interviews, documents and direct observation field notes) and also uses diverse analytic tools, to promote data analytic triangulation (e.g. CAQDAS and storyline with a set of memos), with the generation of knowledge. Such knowledge generation is a function of the researcher's theoretical sensitivity and hence his reflexivity, to have all the necessary insights to construct the theory from the interpretation of the data with a richer and a higher quality research approach. The generated theory is specifically valid to the sample under study and not generalizable to the whole population, due to unique contextual conditions, unless the contextual differences are well established to promote a wider contextual coverage. The reality is considered as subjective (multiple perspective) and socially constructed depending on all the actors (i.e. participants and researcher) and their context.

3.7 Research Temporal Design, Sampling Strategies, Sample Size and Unit of Analysis

3.7.1 Research temporal design

The research design is based on a cross-sectional type of design (Bryman, 2008), where the objective is to capture a snapshot of what is happening in the field. It is not the scope of the research to undergo a temporal assessment, as covered by longitudinal research. The constant comparison process under the GTM is used to capture in-depth data, so as to uncover all the processes and actions within the field understudy but not to include the changes in time from repeated visits to the participants.

3.7.2 Sampling strategies

The data is generated from two sampling stages, mainly based on the interviewing technique, being the primary mechanism used in various grounded theory studies (Birks & Mills, 2011), as referred in Section 3.9. The researcher had to play the role of the facilitator during the interviewing process, especially in the second sampling stage, by employing 'sensitizing', 'theoretical', 'practical' and 'guiding questions', to generate rich data, but at the same time had to watch out for any bias imposition both on the participant, during the interview or on the data, during the analysis (Corbin & Strauss, 2008, p.72).

All the participants, under both targeted samples, are focused on different types of manufacturing sectors to promote variation and richness in the data. Reference was made to the Malta Enterprise Directory 2012 (Malta Enterprise, 2012), for the list of the Maltese manufacturing firms, to be informed on the various sectors within manufacturing, to establish the right selection. Though, it is to be noted that to get access to the research respondents required more efforts than just by sending an e-mail or establishing a phone call. In fact all the respondents' selection, in the two sampling stages, was based on established contacts within the industry, through colleagues Engineers and Lecturers and other relatives and friends.

Every interview with the participants, in both sampling strategies, was digitally recorded and took around one to two hour time-span. The transcription time for each interview, together with the field notes taken from direct observation, took an average time of 20 hours to compile. The duration of the research data collection and analysis process took almost two years to complete, with several gaps dedicated to the analysis processes in between interviews to deploy the constant comparison process through various memos and summary memos both within the CAQDAS and storyline techniques.

3.7.2.1 First stage sampling strategy

The first stage of the sampling strategy combined the approaches of open sampling (Strauss & Corbin, 1998) and convenient sampling (Bryman, 2008). The reason for the selection of such a dual sampling stance is attributed to:

- Open sampling was used since the first participants selected in the research data collection were highly knowledgeable of their respective businesses and well versed with the supply chain responsibilities and in the role of IT at the workplace, to be in a position to detail out an in-depth account on the phenomenon in a broad and more flexible way from a practitioner's perspective, in line with the research criteria to generate rich data, considering that the researcher had no concepts in mind to probe upon; and
- Convenient sampling was used because the participants were readily accessible through established contacts, who also acted as gatekeepers. These were mostly Engineers or Managers, who were in the same class or in a different year, in my Graduate or Post Graduate education level training or else were referred by colleagues who worked in the manufacturing sector. It is to be noted that whenever no previous contacts to gatekeepers were available, the request to gain access to the firms to undertake the research interviews was always turned down.

Appendix 5 provides the list of these eight (8) participants with an anonymous coding label for ethical reasons. The sample matched the requirements of the study, to generate rich data, since it covered different types of firms' ownership and sector type, with different roles across the supply chain (e.g. suppliers, manufacturers, contractors and freight forwarders), which included both locally or internationally owned firms from various countries' subsidiaries and manufacturing sectors respectively. Such a sampling strategy was adopted for the first phase of the research method, to ensure that a good rapport exists with the participants and with such a level of trust, positioned the participants to generate rich and in-depth data for all the open-ended questions, as referred in the interview guide in Appendix 3. Almost all the interviews included a factory tour to enable the researcher to undergo direct observation of the whole manufacturing process.

Initially a pilot study was carried out to test the content validity of the interview guide and to undergo the relevant refinements with the support of my research supervisors. Such a sampling approach, included instances where participants adopted a very flexible approach and explained in detail all their views on the phenomenon, independent of the interview guide sequence of questions. There were some cases where participants also incorporated a critical perspective of some of their statements, on both current and retrospective data, such as by outlining incidents and examples from experience, to enrich the account of the information and how to sustain such SC practices. The researcher's priority was always focused to capture current experiences, over retrospective data, since living forward ideas that are being implemented leads to emerging explanations that are more interesting and innovative, than past actions. This **first stage** of the sampling strategy obtained data that was used in the first two phases of this research data collection and analysis, as referred in Tables 3.1 and 3.2.

3.7.2.2. Second stage sampling strategy

The **second stage** of the sampling strategy was based on theoretical sampling, with the scope of 'filling in' the gaps in the emerged theory (Corbin & Strauss, 2008, p.272). Theoretical sampling was left to last phase of the research process, to avoid varies pitfalls, such as 'premature closure' and 'redundant categories' (Charmaz, 2006, p.107). Such a sampling approach was focused on the emerged concepts, so as to be further explored, by a new interview guide, as referred in Appendix 6, containing a set of semi-structured questions, based on the emerged theory, represented by the set of emerged tentative categories from the Axial Coding stage in Chapter 5. The process of data collection and analysis was controlled by the emerging theory, whether substantive or formal (Glaser & Strauss, 1967). Such a new interview guide was backed with a probe list, to ensure that all participants remain in-line with the focused questions, by setting a set of prompts, so as to sample for concepts (Strauss & Corbin, 1990), as referred in Appendix 7.

The targeted sample was based on a sample of fourteen (14) relevant participants, in line with the research criteria of the study, to generate rich data, where some of them offered a follow-up visit, from the previous interview within the first sampling stage and others were new ones, who were ready to participate. The list of fourteen (14) participants that were included under this stage, are referred in Appendix 5, with an anonymous coding label for ethical reasons. The scope of this sampling stage was to: pursue intentionally and probe beneath the emerging theory; to enrich the research data collection and analysis process; to achieve theoretical saturation; and for 'generating theory' (Glaser & Strauss, 1967, p.45). The researcher during the theoretical sampling, needed to ask participants to outline their account on each emerging category and its properties in question, so that the researcher, through the data analytic approach, could establish the main concern within the data, identify relationships between concepts and to build-up further the variation in the categories' properties and dimensions, so as to reach theoretical saturation (Corbin & Strauss, 2008). Theoretical saturation refers to the data analysis process, where the emerged theory would have been developed, since it ceases to generate new conceptual properties with further additional data (Glaser & Strauss, 1967; Strauss & Corbin, 1990).

The data from this sampling stage was primarily used in the Selective Coding stage (Chapter 5), to establish the research core substantive theory, but was also used in the Axial Coding stage (Chapter 4)

to intensify the already emerged tentative categories, with respect to their properties and dimensions. The breakdown of all the research method phases, with their sampling strategies and coding stages, are referred in Table 3.2.

Research Method Phase	Coding Stage	Sampling Strategy Stage (Total of 22 interviews based on 17 firms)
1	Open Coding	Open Sampling based on 8 interviews on 6 firms (i.e. 2 of the firms offered to interview 2 participants)
2	Axial Coding	Open Sampling based on the previous data set
3	Selective Coding	Theoretical sampling based on 14 interviews based on 14 firms (i.e. 11 new firms and 3 firms with the 2 nd visit)

 Table 3.2: Research Methodology Phases, Coding Stages & Sampling Strategies (Author)

3.7.3 Sample size

The sample size dimension could not be determined before the data collection and analysis was started, since the GTM is an inductive process with evolving theory as new data is collected and analysed. The sample size was a function of the richness of the data collected, so that with the continuous data collection and analysis, this led to theoretical relevance and theoretical saturation. This means that the emerging concepts formed tentative categories, which were well developed and validated with the completion of the set of eight interviews within the open sampling stage. Such a number of participants' data were incorporated in the sample to saturate the key concepts, referred to as tentative categories and sub-categories. Further other two participants with other firms, were initially included in the sample after the first eight interviews data collection and analysis phase. Though, after listening to the digital recording and taking notes of the initial key in vivo quotes led to redundant results. These participants were excluded from the sample, since the respondents at this stage, had nothing new to contribute to the emerging theory. At the completion of the analysis of the first eight interviews, this phase was considered the key determinant of the first stage sample size. This initial stage sample size of eight (8) interviews enabled the completion of the first two key coding stages of the research, namely the Open Coding and Axial Coding, since the Selective Coding was assigned to the theoretical sampling stage, to avoid premature close-out of the emerged theory (Charmaz, 2006). The next data collection and analysis step, as already referred in Table 3.2, consisted from the theoretical sampling stage, which included another set of fourteen (14) interviews, to reach theoretical saturation, where the sample size followed the same principle implemented in the open sampling stage. This sampling stage was focused on the emerged theory, so as to elaborate and refine the conceptual and theoretical developments (Charmaz, 2006).

Stern (2007) advocated that 20 to 30 interviews are adequate to reach saturation. The overall number of interviews of this research amounted to twenty-two (22) which is in line with Stern's (2007) estimated number of interviews to reach theoretical saturation. The sampling approach, within the GTM, differs from statistical random sampling to select the right sample to represent the general population, since the overall population consists from quite a substantial number of SMEs within the manufacturing sector (Appendix 1). The sampling approach used was only focused on a relatively small number of SMEs, which were used to generate data that was required to reach theoretical saturation and emergence of the theory (Glaser & Strauss, 1967).

3.7.4 Unit of analysis

The **unit of analysis** of the research was the organisation, which consisted from any SC member or SC stakeholder. The **unit of observation** was the participant, who represented the role of the organisation or the stakeholder with his/her experiential expertise in all processes and functions. The GTM analysis needed to aggregate the participants' meanings and interpretation to understand the organisation strategies or actions, outcomes and contextual conditions. This theorising process had to respect rigour, by undergoing the constant comparison on the data with all participants, which is referred to as triangulation of data by using multiple participants. Such a triangulation may have introduced a level of ecological fallacy (Van de Ven, 2007), since inferences were being drawn from the individual/s to the group/s (i.e. from the participant to the organisation). Though, the nature of the iterative constant comparison process, was used as a tool, to ensure that the emerged theory is grounded in data, where in some cases even multiple participants were also interviewed within the same firm, as referred in Appendix 5, to have more in-depth comparative data, so as to focus on the key emerged general theory, by including its variations in all tentative categories.

3.8 Research ethics

The interview covering letter and consent form, together with the Interview guide, for the phase one of the research data collection process, are referred in Appendix 3. Such documents were submitted to all participants by e-mail to obtain a signed consent agreement before every interview. Such a consent form formally stated the researcher's ethical responsibility and the research data collection process, for the interview proposed session, to promote freedom of participation, confidentiality, anonymity in any future published work and trust respectively. The participant was always asked verbally in an ethical way whether s/he accepts that the interview will be recorded, to promote a sense of ethical respect by explaining the reason for the usefulness of recorded data, although it was one of the clauses included in the consent form. This is in-line with the ESRC requirements which state that 'Research staff and participants must normally be informed fully about the purpose, methods and intended possible uses of the research, what their participation in the research entails and what risks, if any, are involved'

(ESRC, FRE, 2010, p.3). The two interview guides were approved by the ethics committee and by academics, to be in-line with the ESRC requirements which state that the 'Research should be designed, reviewed and undertaken to ensure integrity, quality and transparency' (ESRC, FRE, 2010, p.3). Other ethical considerations, in line with the ESRC, FRE (2010), to ensure that all participants do not suffer any negative consequences from the research outcome, were that: all the digital recordings and transcribed data were held on a non-portable computer and protected by password and held in a secure place, together with the single back-up held, which was also held in a separate and secure place, where both provisions should minimise any leakage of data as a result of theft/loss; all the participants original data will be safely deleted/disposed from both sources, once the research is over; and that the transcribed data from the interviews was camouflaged from all sensitive information, by using coded labels, such as names of interviews, persons and companies, to ensure anonymity in all research published data.

3.9 Grounding the emerged theory in the data and the role of the a priori and post Literature reviews

The grounded theory methodology is classified as an inductive method, where theoretical sampling, is used as a deductive activity grounded in inductive derived categories (Glaser & Strauss, 1967; Glaser, 1978; Strauss & Corbin, 1998). The research process acts as a virtuous circle, where 'deductions for theoretical sampling fosters better sources of data, therefore better grounded inductions' (Glaser, 1998, p.43). The difference between an inductive and a deductive method relates to 'pacing'; if the researcher looks at data first and then forms the hypotheses, this is an inductive mode, or if the deduction, this is a deductive mode (Glaser, 1998). As a result the research has no need to establish an initial hypothesis from the literature, but this is derived in the course of the research once the core emerged theory is established (Glaser & Strauss, 1967).

The research methodology design is adapted from Lehmann's (2001) model, as depicted in Figure 3.1. The diagram shows that the GTM consists of multiple stages of data collection and analysis, referred as 'slices of data' (Glaser & Strauss, 1967, p.69), which are repeated until reaching theoretical saturation. The slices of data consist of multiple sources of data which are mainly generated from interviews and supported by observations field notes and documents from various sources, where the data is mainly of a qualitative nature. Data determines the theory generated since the more rich the data, the more theory emerges. The successive interviews throughout the data collection and analysis are not considered as case studies but slices of data (Glaser & Strauss, 1967), where all participants gives different data based on their knowledge which are collected and analysed by the researcher to

verify and validate the emerging theory through the constant comparison process, with the objective to reach theoretical saturation of the emerged theory. The theorising process is based on an inductive approach in the data, by comparing 'data to data, concept to data, concept to concept, and linking concepts back to the data' (Glaser, 1996, p.98), either from the same interview or other interviews, performed both in a successive and also in an iterative process in line with the constant comparison process.

The analysis consists of three coding techniques (Open, Axial and Selective Coding) in an iterative process to achieve an interconnection between emerged categories and to reach theoretical saturation with generation of the substantive theory. Memos are used as a sense-making process by enabling an audit trail to keep track of all the constant comparisons of all the theorising process, so as to flesh out the emergent concepts through theoretical abstraction.



Figure 3.1: The grounded theory research strategy (adapted from Lehmann, 2001)

At the open sampling stage of the research, the first set of interviews is based on an interview guide derived from the **a priori literature review**. Such a literature review is initially undertaken to guide the researcher, by establishing a tentative research question, based on the current discipline literature gaps and at the same time to promote research significance, by focussing on the current discipline emerging changes and dynamic challenges. Such an a priori literature review is also used to establish the current theories that builds up the research phenomenon, through the provision of significant concepts, with their dimensions, but which are kept 'provisional and conditional' (Strauss, 1987, p.12). Finally, it also assists the researcher to enhance his theoretical sensitivity, during the data collection and analysis process, to interact with the informants and also to grasp better the meaning of their statements within the interviews. In fact Creswell (2007) advocated that, the preparation for research in qualitative studies is essential, and some pre-research literature is necessary to frame the problem and provide a rationale for the study.

Theoretical sampling is used to capture refined data, once the theory starts to emerge, through further data collection and analysis based on a new interview guide. Such an approach is used to densify the emerged concepts variation in their properties and dimensions and also to achieve theoretical saturation. All the process leads to the generation of theory, within the empirical area, referred as substantive theory (Glaser & Strauss, 1967). At the theoretical sampling stage, the theory becomes dense with concepts and is also enriched and may be finally refined by relevant extant literature, by integrating the emerged theory with the literature, referred as theoretical integration (Strauss, 1987; Glaser, 1998; Strauss & Corbin, 1998). As a result, at this stage, the researcher is said to have 'discovered' a substantive theory (Glaser & Strauss, 1967). Substantive theories are applicable to the particular area of empirical enquiry from which they emerged (Ibid).

The generated substantive theory is related with the relevant literature so as to establish the contribution to the extant literature (Chapter 7). Such a contribution is determined by identifying the commonalities, the differences and the new, between the current literature and the generated substantive grounded theory (Glaser, 1998). Furthermore, the role of the extant literature should become very important at this stage, because the researcher needs to acquire sensitivity and knowledge on grounded concepts (Glaser, 1998). The **post literature review** is hence delayed to this final stage until the emerged theory is building up, to avoid any forcing on the emerged theory and is therefore read as a source of more data to be compared with existing grounded theory but not to control or derail it (Strauss, 1987; Glaser, 1998). Henceforth, the literature is used as a 'secondary source of data' (Strauss & Corbin, 1998, p.51). The literature review is directly related to the major subject areas based on the emerged concepts (i.e. emerged themes). Such an approach ensures and minimise the effect of preconceived concepts and properties from the literature on the generated theory. It is then,

and not before, that data from the extant literature contributes to the study (Eisenhardt, 1989; Urquhart, 2001).

3.10 The GTM research design three coding techniques

The aim of the research design is how one can generate empirical evidence to examine the research question (Lee & Lings, 2008). The analytic techniques are used as a tool to generate such evidence by establishing ways of probing in the data and thinking comparatively to generate ideas about what the data are saying and to establish the properties and dimensions of categories (Strauss & Corbin, 1998). To achieve such a systematic and transparent analytic process, the research methodology is based on three GTM key coding analytic stages:

3.10.1 Open Coding stage

The first stage of the analysis is Open Coding. Open Coding is the process to open the statements (as text or voice) outlined by the participants, to expose the meaning contained therein, in terms of actions, context/conditions and consequences categories respectively, referred to as the Coding Paradigm, within the micro and macro conditions in line with the Conditional/Consequential Matrix (Strauss & Corbin, 1998). In vivo Coding or Open Coding is an analytic process, performed on either single words, or phrases or sentences, which leads to an emerging concept or idea from the data. The Coding Paradigm is used as a framework to explicate clearly all the connections of the conditions, actions and consequences related to the phenomenon under investigation, to promote empirical grounding (Strauss & Corbin, 1990). Such a Coding Paradigm includes: the context/conditions which give rise to it, where it answers the 'why, where, how come and when' questions (e.g. the situations of the Maltese Industry and the international links for negotiation); the actions/interactions by which it is expressed, where it answers the 'by whom' and 'why' questions (e.g. strategic responses to issues/events/problems, such as collaboration and contractual agreements); and the consequences it produces, where it answers the 'what happens as the result of the actions/interactions' questions (e.g. effectiveness, efficiency and competitiveness) (Strauss & Corbin, 1990, 1998). The researcher coding approach in terms of the Coding Paradigm is used to promote a structured approach to facilitate the conceptual insight, but the main objective is to achieve explanations and to gain understanding of the phenomenon, within the micro and macro contextual conditions, as outlined in the conditional/consequential matrix (Strauss & Corbin, 1998). The process of Open Coding is to capture all potentially and provisional concepts in terms of the emerged codes grounded in the data through a set of **provisional categories** (Corbin & Strauss, 2008).

Such a theorising process is mainly initiated on the computer software application MAXQDA 2007,

with the relevant code memos for each open code and resumes through the storyline approach with the relevant memos and overall memos, in the next Axial Coding stage, as the first two key two data analytic approaches to data analysis, as referred to in Sections 3.11.1 and 3.11.2 respectively. The scope of using the storyline, as a follow-up of the CAQDAS approach, is to promote better freedom and creativity in-built in the GTM, since qualitative research software may 'numb', 'stultifies', and 'abort intuitive skill development', required in the 'grounded theory package' and to shift from the fragmented approach to a more holistic theorising process approach, since it may also 'hinders and cops out on the skill of doing grounded theory' (Glaser, 1998, p.185,186).

3.10.2 Axial coding stage

The second stage of the analysis is the Axial Coding analytic approach. There is no demarcation line between the Open Coding and Axial Coding, but it is a continuous and iterative process between the two coding stages, since emergence starts from the initial steps of the data analysis process (Strauss, 1987). In the Axial Coding, the discrete parts of data, within the Open Coding stage, as the open codes are re-assembled through theoretical abstract statements, to explain what is happening in the data and by deriving the relationships between the emerged tentative categories, by 'crosscutting or relating concepts to each other', in line with the Coding Paradigm (Corbin & Strauss, 2008, p.195). The coding paradigm is only used as a means and not the end in itself and as a result is used by the researcher to arrive at theory building.

The objective of this stage, as clearly advocated by Strauss and Corbin (1998, p.132), is not based on:

'identifying whether the conditions are causal, intervening, or contextual. Rather, what the analyst should focus on the complex interweaving of events (conditions) leading to a problem, an issue or a happening to which persons are responding through some form of action/interaction, with some sort of consequence'.

As a result the linking between concepts enables the theorising process to elevate significant concepts into a set of related tentative categories. Such an approach is needed to theorise from the data around the axis of one category at a time (Strauss, 1987). This theorising process, with reference to all concepts' relationships, is referred in Chapter 4, where the **tentative categories**, with all their interweaving connections with one another are generated. Such an analytic theorising process approach is mainly based on the storyline method with the support of: (a) narrative based memos, for each tentative category, with the selection of key *in vivo* quotes from the data; (b) logic diagrams for each tentative category, with the relevant properties and their dimensional variations and their conditions; and (c) conceptual frameworks, for each tentative category, with the related conditions and the consequences, in line with the Coding Paradigm, as referred in Section 3.10.1.
The role of the **tentative categories** is to capture more abstract concepts (i.e. concepts at broader levels of scale) to explain what the data are saying through their analytic power and 'potential to explain and predict' the phenomenon under study (Strauss & Corbin, 1998, p.113). The sequence of theorising starts from open codes (i.e. conceptual reference to represent the *in vivo* data) in Phase 1, where codes are then grouped to derive concepts, themes and categories, in such an orderly sequence in Phase 2, which all lead to the foundations of the emerged theory. The tentative categories are used as an abstract representation of either the actions, or conditions or consequences, to represent the data, which finally lead to the establishment of the core theme in next Selective Coding stage, in Phase 3, that links all the final tentative categories together.

3.10.3 Selective Coding stage

The third stage is the Selective Coding. In this stage the categories are integrated and refined, with the selection of a core category, together with other key categories and sub-categories, to represent the substantive theory. Such an integration process of main themes through the storyline technique gives the emerged theory greater explanatory power (Strauss & Corbin, 1998). Furthermore, the conceptual framework, as referred in Figure 6.1 represents the substantive theory, where such a framework approach is also used 'to organize their data, keep a record of their concepts and the relationships between them, and to integrate ideas' (Corbin & Strauss, 2008, p.125).

This integration process is referred to as the 'delimiting of the theory' to 'one or two core variable(s)', which should act as a guide for further data collection and analysis (Glaser, 1978, p.61, 72). The core category is said to cut across all other categories, in line with the Selective Coding process (Strauss & Corbin, 1998). By doing so, the research need to focus on one of the several basic social processes or conditions that would be present in the data, to serve as a basis for a generalised theory with a set of theoretical propositions (Strauss & Corbin, 1998; Binder & Edwards, 2010). Glaser (1998, p.190) outlines that the theory needs to have ideally 'the smaller the amount of concepts that account for the greatest variation in substantive behaviour', so as to resolve the main concern of the theory. Such a stance links all the categories around 'a core category to refining and trimming the resulting theoretical construction' (Corbin & Strauss, 2008, p.263). The refining and trimming is referred to the delimitation of the analysis to those significant variables affecting the core variable to contribute to parsimonious theory (Glaser & Strauss, 1967) which represents the final emerged pattern of social behaviour (Glaser, 2002).

3.11 Data collection and analysis data management tools and techniques

The Open Coding stage initiated the research analysis and is performed on all the eight interviews, in a systematic way by a qualitative data management tool, based on a computer software application. All the interviews are digitally recorded and transcribed to ensure that all the meaning and statements are all captured by the text, which are analysed in line with the Open Coding technique, referred in Section 3.10.1. The Open Coding stage, based on the CAQDAS, together with the Axial Coding stage based on the storyline approach, are the two data analytic processes. Both stages took almost three years to finalise, due to the in depth analysis and iterative nature of the research process having to organise interviews in a sequential order in time, with the necessary gaps to allow time for data analysis. As a result, such a data collection and analysis approach consists of a cumulative process of theorising sessions, from one interview to another in a sequence and also through cross-referencing and recursive mode, from one interview to another with no sequence whatsoever, due to the nature of the GTM analysis constant comparison process, to promote sense-making on the emerged theory.

3.11.1 Computer-assisted qualitative data analysis software (CAQDAS)

The CAQDAS used is the MAXQDA 2007 application (Kuchkartz et al., 2008), and is the same application used by Corbin and Strauss's (2008) book. Although the researcher had a choice between NVivo and MAXQDA, the latter application software is selected after reading the recommended positive comments on the ease of use by Flick (2006) and Corbin and Strauss (2008). The Open Coding stage is organised by the CAQDAS, where all emerged open codes are supported by descriptive and theorising memos to build and understand what the data are saying. The CAQDAS is used as a data management tool, to classify all the emerged codes in line with the coding paradigm (actions, context/conditions and consequences classification) and to promote an audit trail, along the research process, to promote research rigour. The MAXQDA application also provides a time saver in storing and coding, especially in the first phase of the research process, referred as the Open Coding stage, so as to minimise the effect of information overload. One needs to understand, based on the researcher's experience on this research work, that this software application is a tool for content analysis, by deriving frequency counts and also provides the facility to retrieve segments, in particular to see all in vivo quotes from the data associated with each code. Such information management provision facilitates the researcher's analytic process during the theorising process within all memos dedicated to each code. Though, it is to be noted, that the real grounded theory process, still remains in the researcher's mind and that these code memos within the CAQDAS served as a supporting tool in the grouping of the open codes to higher level codes, referred as tentative categories. This thinking process is the main reason to use the storyline approach in the Axial Coding stage, to serve as a follow-up analytic method of this CAQDAS stage based on the same data sample.

3.11.2 Qualitative storyline technique supported by displays

3.11.2.1 Qualitative storyline technique

The storyline analysis technique followed the CAQDAS analysis approach on the same data set. In practice, it cannot be excluded, that both the storyline and the CAQDAS approaches interweaved in an iterative process, to complete effectively such an analytic research process for all the eight interviews, but are outlined as sequential steps, so as to organise the research methodology presentation in a scholarly way. The storyline technique is used to understand the content and to uncover what the data are saying within the context of the sources (Patton, 2002) and in a systematic and prescriptive approach (Strauss & Corbin, 1998).

The storyline makes use of a narrative approach theorising process with *in vivo* references to the data, by maintaining an audit trail to ensure that all the emerged theory is grounded in data through the relevant evidence and by bringing the participants back to life. Such a data analytic approach needs to be applied throughout the research process, by moving between raw data (i.e. interview by interview *in vivo* quotes for each emerged theme), memos and the storyline itself, in line with both the constant comparison and the theoretical abstraction processes to discover the theoretical emerging patterns (Strauss & Corbin, 1998). It cannot be excluded that the storyline remains the conceptual interpretation by the researcher of all the data which is still a recommended analytic acceptable approach (Strauss & Corbin, 1998), but is dismissed by the classical grounded theory approach (Glaser, 1992). Though, one needs to put the right perspective on the storyline approach, since within the research community, the storyline, as a research strategy, is an underutilised approach within the GTM (Birks et al., 2009).

The storyline initially is used to compare data with data from the same or different interviews, so as to develop the tentative categories (Glaser, 1978). The approach included memos and overall memos (i.e. a summary memo is a memo based on one or more memos) from the transcripts (raw data) to summarise what the data are saying (Birks et al., 2007). Along this analytic approach, the researcher is immersed in the data and tries to understand and reflects upon, even at times by stepping back from the data, so as to establish what the data are saying, to grasp the key concerns of the emerging theory. The storyline approach, with the support of the storyline memos, is a process which includes the codes (i.e. concepts) identification, explanation and substantiation by quotes which are a construction and interpretation of the data (Strauss & Corbin, 1998). Hypothetical statements can then be used to identify relationships in line with the Axial Coding technique (Strauss & Corbin, 1990, 1998). As a result the final storyline needs to outline the theoretical constructs made up from the categories and

110

their relationships (Strauss, 1987; Charmaz, 2006). The variation within the data also builds the dimensional variations within the emerging concepts, both in the storyline and in the final theory, to increase the theory explanatory power (Strauss & Corbin, 1998). Furthermore, the storyline approach assists in the identification of gaps in the emerging theory, which is in line with the grounded theory discovery process (Charmaz, 2006). Such gaps need to be fulfilled by undergoing theoretical sampling (Strauss & Corbin, 1998), as referred in the next Selective Coding stage.

3.11.2.2 Displays or mind maps technique

Logic diagrams (i.e. tabulation) and conceptual frameworks (i.e. mind-maps) are also used throughout the data analysis to visually observe the relationships between themes/concepts and to outline all dimensional variations of all tentative categories, so as to assist the analytic and theoretical abstraction process, which are important visual elements of the integrative process in grounded theory (Miles & Huberman, 1994; Strauss & Corbin, 1998). Such diagrams helps to clarify and facilitates the analysis and evaluation of the theorising process of all the emerged categories, within the Axial Coding stage to derive the relationships between all emerged categories and their variations in a transparent process, as referred in the data analysis Chapter 4. The Code Matrix (i.e. Tree diagram) approach (Miles & Huberman, 1994) is used to demonstrate an audit trail and visual representation of all the emerged categories from the Open, Axial and Selective Coding stages respectively in the research analytic process (Table 5.1). The use of manual derived displays or mind maps, enable the researcher to focus, since they produce a condensed and comprehensive format of all the data as text, since text is 'weak and cumbersome form of display' but on the other hand displays are 'helpful for understanding the flow, location and connection of events' (Miles & Huberman, 1994, p.91, 93). Such a theorising process through the use of diagrams is supported in grounded theory work (Glaser & Strauss, 1967; Strauss & Corbin, 1998). In fact, Strauss and Corbin (1998, p.153) asserted that: 'diagrams are more useful than storytelling for sorting relationships between concepts'; 'valuable tools for integration'; enable the researcher to 'work with concepts rather than with details of data'; and diagrams are 'abstract representations of data'.

3.11.2.3 Triangulation of data analysis techniques and data sources

All the analysis from the data, with both the CAQDAS (e.g. through open codes and code memos) and the storyline techniques (e.g. through memos and overall memos), are all used in an iterative platform, based on constant comparison, to assist the theorising process so as to derive the key themes, capture the theory and the relationship/s between codes or categories. These two different techniques promote **triangulation of data analytic methods** to address the research validity (Miles & Huberman, 1994;

Strauss & Corbin, 1998; Denzin & Lincoln, 2005; Bryman, 2008; Corbin & Strauss, 2008). Such a stance does not refer to a mixed method research methodology but a dual approach to data analysis.

Multiple data sources are also used to promote trustworthy and authentic research through **data triangulation** (Miles & Huberman, 1994) (e.g. interviews, direct observation fields notes from factory tours and grey literature based documents, which includes both qualitative data, such as procedures manuals, information on each company on the web and Government portal and also quantitative data, such as annual financial accounts of each firm from MFSA and international mother companies). Such multiple sources of data are needed, since in grounded theory it is rarely to use interviews as the 'sole form of data collection' (Suddaby, 2006, p.635). Such triangulation of data, by using such different sources, is performed so that the secondary data could enrich and be compared to the emerging theory from the primary data in line with the notion as 'all is data' (Glaser, 1998, p.142). In general, qualitative studies should also triangulate their findings through multiple methods for their work to have greater impact (Binder & Edwards, 2010; Bluhm et al., 2010), but it is to be noted that this research is focussed on a single research methodology based on a GTM, with both multiple data sources and multiple data analytic techniques.

3.12 Implementation outline of the overall research data collection and analysis phases

3.12.1 Research method phase 1: Open sampling with Open Coding analysis technique assisted by CAQDAS

In this Open Coding stage, the analysis was performed on a set of eight transcribed interviews, denoted by interview one (1/INT1) to interview eight (6/INT8), as referred in Appendix 5. The interviews were guided by the first interview guide, as referred in Appendix 3, and in Section 3.7.2.1. The analysis was performed with the support of the CAQDAS, which was organised in line with the Coding Paradigm (Strauss & Corbin, 1998) as the analytic stance to establish systematically and to gain understanding of the phenomena, of the 'how' (i.e. the process) and of the 'why' (i.e. the structure or conditions) to capture the dynamic and evolving nature of events with their consequences (i.e. the outcomes) (Strauss & Corbin, 1998). Hence all the **open codes** were classified into: 'actions and reactions'; 'consequences and outcomes'; and context and conditions', as referred in Appendix 8. All interviews' transcription, together with field notes and secondary documentary data, were uploaded in sequence with every interview carried out. The analysis of each interview was undertaken at the earliest, possibly on the same day, with every interview transcription completion, to enrich the code memos with all the necessary theorising notes, included within each code, since the researcher would still have a clear picture of the participant explicit and implicit messages in the data.

Such a CAQDAS facilitated in a systematic way, the extraction of all *in vivo* segments of texts, by breaking the data, either in line-by-line or sentence or paragraph of in vivo text, from all the transcribed interviews, to be attached to each emerged code or open code (concept). Figure 3.2 depicts the MAXQDA four segments, namely the Document System, Text Browser, Code System and Retrieved Segment. In Figure 3.2, an example of an open code can be observed, which is labelled as 'communication' and which has a frequency count thirty-one (31) within the code system segment. Open Coding refers to the process of thinking outside the box, where it is well established that thinking is the heart and soul of the analytic process under qualitative research, so as to represent the data with an abstract term or word referred as a concept. Such an approach gave a set of fragmented codes from the data. The number of emerged codes was one hundred and thirty-two (132) codes from a 1472 selected in vivo quotes from the transcribed data, which amounted to around 30,000 words of transcriptions, as referred by a screenshot of all such open codes in Appendix 8. To achieve a more reader friendly approach, the data file from the MAXQDA with 132 codes was exported on a spreadsheet to represent all the emerged tentative codes on a single sheet, as referred in Appendix 9. Furthermore, the spreadsheet mathematical provision was used to enable sorting and filtering of all the codes function of the frequency count of less than 10 (i.e. < 10) for the data as derived from the CAQDAS, to capture the first set of forty-six (46) tentative open codes, as referred in Appendix 10.



Figure: 3.2 Screenshot of MAXQDA 2007 used for the Open Coding stage data collection and analysis (Author)

Such a data management tool enabled the researcher to compare interview by interview and also different parts within the same interview, which were considered similar and relevant, by extracting *in*

vivo texts from the transcribed interviews and grouping them under the same concept (i.e. open code) to promote a systematic and organised approach of theorising and sense-making on a concept by concept basis. The CAQDAS supported this theorising process by providing all the *in vivo* quotes in one document and in an orderly sequence for each open code (i.e. using the retrieval segment facility). At the same time such a compiled document for each code enabled the researcher to go through all the in vivo quotes one by one, which served as a validation process to confirm that each in vivo quote fits the emerged code. Such an analytic process also enabled the researcher to understand and theorise on what the data were saying and also whether such data could be included in other emerged codes. Such a constant comparison iterative process of various data from different participants enabled the researcher to derive each concept properties with its dimensional variation and at the same time initiated the approach towards theoretical saturation of each concept with all the eight interviews analysis. The concept's property is the characteristics (components) that define/explain an action (or group of actions) or a condition or an outcome of the concept, where the property can have dimensions. Such a theorising process was represented in a memo for each code within the CAQDAS, referred as the code memo, as illustrated in Figure 3.3. This systematic approach enabled the researcher to understand what the data were saying and at the same assisted in the theory generation iterative process. With such a systematic approach also promoted trustworthy research with the acceptable research validity and repeatability.

Open Coding refers to break the data to examine the meaning, to get in depth in the data and to think outside the box to get to the details to all emerging concepts in the initial phase of the research data collection and analysis. Such a stance built a solid foundation for the subsequent analysis stages having considered all possible interpretation from the data before deciding on the key issues covering the researched phenomenon.



Figure 3.3: Code memo sample from MAXQDA 2007 (Author)

Such an approach assisted the researcher to immerse himself in the data and try to conceptualise from the data what does the open code represents, by inserting all the theorising account made by the researcher, at any moment in time, in code memos attached to all open codes or sub-codes, as already referred in Appendix 9. An extract of these five open codes, from the overall forty-six (46) open codes, as extracted from Appendix 10, are being referred in Table 3.3 below, to demonstrate the audit trail of all open codes sources vis a vis the eight participants (i.e. Interview 1 to 8) and the frequency count values attributed to each open code by each participant.

		Interview numbers							
Filtered Emerged Categories (Based on									Total
frequency > 10)	1	2	3	4	5	6	7	8	Sum
Actions and reactions\Auditing & traceability of									
operations	5	4	0	3	2	4	3	4	25
Actions and reactions\Change management	2	7	1	1	1	2	0	1	15
Actions and reactions\Collaboration	4	2	10	8	2	1	4	2	33
Actions and reactions\Collaboration\Teamwork	4	2	0	2	0	0	0	4	12
Actions and reactions\Collaboration\Teamwork\employees									
engagement	0	8	1	1	0	0	1	2	13

Table 3.3: A demo showing the audit trail of five open codes (Author)

There were some open codes that were assigned *in vivo* labels and others were named with established names in the literature. Assurance was made to promote research rigour that every name assigned from the literature, really represents what the data were saying within the segment selected from the transcription linked to each open code, otherwise a more appropriate name that fits the data had to be assigned, to promote research rigour and avoid any forced concepts based on preconceived literature. For example, collaboration is a concept name found in the extant literature and the assigned properties were: teamwork; employees' engagement; duplication of effort; cross-functional operations; commitment; etc, as referred in Figure 3.4 and Appendix 9. Such a stance showed how the concepts earned their way throughput this phase and also their relationship with each other, in line with the GTM iterative process.



Figure 3.4: Snapshot of collaboration open code from MAXQDA 2007 (Author)

3.12.2 Research method phase 2: Open sampling with Axial Coding analysis technique based on the storyline analysis approach

In the storyline analysis stage, the same set of eight interviews was used and the analysis included a theorising process based on memos for each interview and overall memos to cover the either part of the set or the whole set of interviews in an iterative approach. Memos were based on a narrative approach to represent the theorising process based on a selection of key vivo quotes from the data, which represented exactly the common thread what the participants were saying. This stage began with the process of reassembling the same data set, which was fractured during the previous Open Coding stage through the key 46 open codes, so as to construct the relationship of the emerged tentative categories with their sub-categories, together with their properties and dimensions. The use of visual aids, in particular, logic diagrams as tables, code matrices as tree diagrams and conceptual frameworks as mind maps, were also used as indicated in Section 3.11.2.2. Such visual aids were used to build on the memos, so as to represent the relationship between the emerged concepts in line with the Axial Coding technique and at the same time assisted the sense-making and theorising process of the analysis.

Such a narrative approach, with the support of *in vivo* quotes, brought back to life, what the participants were saying. Furthermore, such an approach created an audit trail, for each emerged tentative category, by including only the key *in vivo* quotes from the data of all interviews. The researcher's decision to adopt such a limitation measure in the inclusion of only the key *in vivo* quotes from the data, was attributed to the voluminous amount of *in vivo* quotes derived from the constant comparison process from all interviews, so as to produce a concise and more manageable account of the data, which were of key relevance to the theorising process. The analytic approach was mainly

based on the researcher's skills to immerse himself in the data, where at times there was the need to dig deep in data and in other times to step back from the data, to establish what the data were saying and grasp the key concerns by compiling memos and overall memos. Such an analysis approach was very intensive and time consuming, since it was based on the researcher's thinking process and theoretical sensitivity to interpret what the data were saying, either line-by-line or sentence or paragraph of *in vivo* text.

The research method is an iterative and evolving process that leads to the emerged theory, which is not only based on the storyline approach of what the data are saying, but also on the construction and interpretation what the participant want to mean and disclose. There is the need of a sound theoretical sensitivity of the researcher to capture what the data are saying and how it fits in the research objective. It is to be noted that the GTM is an intuitive process of data collection and analysis, with a substantial level of creativity and theoretical abstraction from the data and hence the theorising and sense making process is not always logically demonstrated within the research document. The major part of the analysis work, especially as the research data collection and analysis progressed, was in the researcher's mind, which was based on the way the interpretation process of the data was used to capture and construct the necessary sense-making process. Initially such a sense-making process was demonstrated through descriptive memos, which was then followed by conceptualising memos with the research iterative process analysis approach. Theory emerged from the data but was being interpreted and constructed through the memoing process.

The emerged theory from this storyline approach, with its memos and overall memos, was also supported by the previous Open Coding stage, by an iterative constant comparison process approach, by using the list of forty six (46) key emerged open codes with their respective code memos. Such a process consisted from the comparison of the two stages, which enabled the researcher to build and/or refine on the emerged theory **to approach theoretical saturation** of all the emerged key codes. With such a storyline approach it enabled to densify the Open Coding stage and led to the successive Axial Coding stage to capture the relationship between codes in a more coherent approach to the emerged theory two (22) tentative categories. This is the conceptualising and sense-making process to represent the theory with a set of concepts, which are then grouped to form a few set of categories as potential indicators, referred as **tentative categories** (i.e. axial codes), which are related and interweaved with each other, to represent the emerging phenomenon (i.e. theory generation).

Furthermore, the storyline approach included logic diagrams and conceptual frameworks for each tentative category, so as to build on the memos and to represent visually the relationship between the

emerged concepts in line with the Axial Coding technique, which assisted the sense-making and theorising process of the analysis. All the twenty-two (22) emerged key concepts were represented in a storyline approach through a set of memos on a concept by concept basis, to define and explain the key concepts, with all its relationships to other key concepts or sub-concepts, which were all supported by the key in-vivo quotes from the data. The intention of the data analysis was that key concepts were going to be referred as categories and the sub-concepts as sub-categories respectively. The conceptual labels were to be used to represent the researcher interpretation on the data which were initially tentative open codes in the Open Coding stage, with all the relevant theorising process within the codes memos. Such an analytic process in this subsequent data collection and analysis phase was tracked through memos, all along the research process, to theorise on the emerged theory in an iterative process. Such an analytic approach cannot be attributed to a smooth analytic sequential process, since it needed a cyclical analytic stance from one interview to another, without any sequence whatsoever. In fact the memos became more accurate and complex as the analysis progresses after the overall constant comparison process of the whole set of eight interviews.

3.12.3 Research method phase **3**: Theoretical sampling with Selective Coding based on the storyline analysis approach under the GTM

This stage is referred to as the final data collection and analysis phase, which was focused on refining the theory of all emerged categories through the theoretical sampling technique to collect further data **to achieve theoretical saturation** of all categories. Such a stage included a newly designed interview guide, which was semi-structured, where all tentative emerged categories from the previous Axial Coding phase were included within the questions, to provide the possibility to generate data from the participants through direct questions, as referred in Section 3.7.2.2 and Appendix 6.

The sample included fourteen participants, which consisted from various stakeholders, which were selected to fit the sampling strategy to match the research criteria to generate rich data, by densifying the emerged theory with all the properties and dimensions, so as to promote theoretical saturation of all the emerged theory. The set of fourteen (14) transcribed interviews, amounted to a word count of around 35,000 words and were denoted by interview one (7TS/INT1) to interview fourteen (17TS/INT14), where the first digit refer to the firm identity code, as referred in Appendix 5 and the 'TS' refers the theoretical sampling phase. The objective of this theoretical sampling was to select the right audience and questioning the participants on the emerged theory to refine and build on the current analysis to reach theoretical saturation of the twenty-two (22) tentative emerged categories in the previous stage.

This stage interview guide, with questions focused on the emerged key categories, enabled the informants to intensify their account to more in depth explanations than the previous set of open ended interviews. Space was left for the participant to outline his/her views on each emerged concept, by guiding them by an open ended question on each category but at the same time probes were used, whenever necessary, to guide the participant to keep on track with the focused questions (Appendix 7). The aim of using probes was to lead all participants to disclose rich data (i.e. data with all the necessary details but with a high level of objectivity). Such a stance needed that the researcher had to keep an open mind but focus on the participant account for each question, to dig deep during the interview session, through an effective and active listening approach in an interactive way. Special attention had to be given so that the researcher did not introduce any bias, or not to force or lead the participant to answer what the researcher needed to hear and neither to force him/her to shift away from telling the real account from a practitioner's perspective. Such interviews were also accompanied by follow-up mails and phone calls to clarify any issue with the participants and to generate rich data and furthermore to fill any gaps in the emerged theory. Such a stance enabled to refine the theorising process and to densify the emerged tentative categories by digging deep in the data. Additionally, apart from the primary data, from all the interviews, the data collection included secondary data from various sources available to the researcher, such as documents handed by participants, on-line information from firms' web portals and field notes from the factory tours.

The theorising process was based on the Selective Coding analytic approach using the storyline technique, which consisted from the sense-making in line with the GTM, by undergoing an iterative constant comparison process of all the interviews' data, through theoretical abstraction, by stepping back from the data and asking: 'What is happening here?' and 'What is the core theme behind all this?' This is the conceptualising and sense-making process to generate the theory with a set of categories. Such an approach enabled the researcher to select the action based core category, which cuts across all the other five (5) main action categories with their sixteen (16) action sub-categories. Such a core category needed to explain holistically the research SCI phenomenon. Furthermore, it cannot be excluded the importance for the researcher to remain immersed in the data, by including the key *in vivo* quotes from the data, that were seen as most relevant to the theorising process transition to construct and discover the core theory through such an inductive approach from the data (Chapter 5). Furthermore, such an analytic process included a theoretical account of the emerged theory to promote understanding, by an in depth outline of the substantive theory behind the theoretical model derived from the data, as referred in Chapter 6.

The data was analysed in line with the Straussian Coding Paradigm model, to classify all emerged theory in terms of the actions, contextual data and consequences, which served as a robust platform for

the final theoretical framework. The audit trail of all the theorising process, that led to the emergence of the substantive theory from all the three stages, which included all the forty-six (46) open codes, twenty-two (22) tentative categories and finally the action core category, five (5) action main categories and sixteen (16) action sub-categories, together with the nine (9) contextual conditions and one (1) consequence sub-divided into eight (8) dimensions, is represented by the Code Matrix in Table 5.1. The final theoretical conceptual framework, which represents the final emerged theory, in line with the Coding Paradigm, is referred in Figure 6.1.

3.13 Chapter summary

This chapter has outlined that research methodology did not have an a priori hypothesis but an open ended research question to guide the research process and an open ended initial interview guide to take-off the research first data collection and analysis phase. It is clearly justified why the GTM is well suited as one of the most suitable research method for this research, since it has the capability to understand the conceptual relationships and to generate a substantive theory on the phenomenon understudy, which is exploratory and undergoing a dynamic paradigm shift (Glaser & Strauss, 1967).

The a priori literature review was performed to: set-up the initial open-ended interview guide and establish a tentative research question, which were used to take-off the research effectively; establish the literature gaps within the current literature to serve as a guide to the approach and perspective for the research purpose and to increase its significance; and enhance theoretical sensitivity of the researcher on *in vivo* data within the overall data collection and analysis process. Due attention was also given during the data analysis process, not to import any preconceived theories from the a priori literature on the data. Such an approach is in line with the Straussian GTM but goes against Glaserian GTM since the latter GTM declares a key 'concern that literature might contaminate, stifle or contaminate or otherwise impede the researchers' effort to generate categories' (Glaser, 1992, p.31) and furthermore that the emerged theory from the data through its 'interpretations become coloured by the preconceptions' (Glaser, 1998, p.71). The researcher's intended purpose is to achieve theory building through the constant comparison iterative process, which is to be a closely tied process, to generate theory with the right evidence from the data (Eisenhardt, 1989).

The research method with its data collection process, with the two sampling techniques and with its analysis three coding phases, looked at the data in line with the Coding Paradigm (Corbin & Strauss, 2008) (i.e. context, outcome and actions), so that there was an ongoing build-up of theory from the data under the constant comparison process within all the three coding phases until theoretical saturation of all categories had been reached. The principal data analytic approach used within the research, apart from the CAQDAS technique, was the storyline technique, with narratives

substantiated by key *in vivo* excerpts from the data, but both tools served as data analysis triangulation techniques to promote research rigour. Glaser (1992) had criticised the use of the storyline approach to generate the theory, since the researcher needs to use the correct analytic processes and procedures, where data needs to generate the theory. On the other hand, the researcher adopted the descriptive narrative, based on a grounded storyline approach, as a conceptualization tool, so as to aid theoretical development and integration (Strauss & Corbin, 1990). Hence, the researcher has used the storyline grounded in data, to arrive at the theory in line with the Straussian GTM, where the theory is generated from the data and not from the storyline. It cannot be excluded that the data that had to be analysed and coded was voluminous, which included an average of one-hour to two-hours duration of each interview, with a total of twenty two (22) interviews covering all the primary data, under both the open and theoretical sampling stages, which amounted to around 65,000 of words within all the data.

There was also the analysis of the secondary data sources which were excluded from the above word count reference and where they served as a triangulation of data sources (e.g. procedure manuals; firms' annual reports and other ancillary reports; content on the firms' website, field notes from interviews and factory tours and the general media). In every data analysis step, the interpretive and constructive approach was used to dig deep what the data were saying, by theorising and deriving the sense-making behind every statement or explanation or theme (e.g. 'KPIs are used to monitor the progress regarding material (raw components) and efficiencies of work carried out per station' (2/INT3) i.e. performance measurements serve as a reference to material usage and work efficiencies to promote lean management). To promote scientific research all the necessary research quality measures were taken to promote a trustworthy and a plausible research methodology and a valid generated substantive theory, in line with key grounded theory key scholars. Whenever possible, extreme cases within the data, where coded to derive the variation within the generated theory.

The emerged theory was kept as provisional in all the first two initial coding stages until the Selective Coding stage was reached and as a result all the emerged theoretical concepts were no longer then referred as tentative categories but as categories. The categories are assumed to be the 'cornerstones of developing a theory' (Strauss & Corbin, 1990, p.7), which leads to the generation of the substantive theory on the phenomenon. The final emerged theory objective, as outlined by Strauss and Corbin (1998, p.25), is to construct an 'explanatory scheme that systematically integrates various concepts through statement of relationships' so as to 'explain and predict events, thereby providing guides to action'. Such an objective has been met by the emerged substantive theory, as outlined in Chapter 6, which was represented by a conceptual account of all the categories with their properties and dimensions' variations (Section 6.3.6), together with the theoretical framework in Figure 6.1.

The post literature review, was carried out in the research final phase, in line with both Straussian and Glaserian GTMs (Strauss, 1987; Glaser, 1998), since this systematic literature review was used to integrate the emerged theory within the extant literature and also used the literature to refine the emerged theory (Section 7.4.7 & Figure 7.2), by treating the literature as the final set of data (Strauss & Corbin, 1998). The areas within the extant literature of this systematic post literature review were driven by the emerged themes from the data. This approach was based on the principle that the literature review acts on the generated substantive theory as a source of validation and to establish whether it fits, extends and refines the current knowledge in the field and also fills any literature gaps, so that the researcher can derive the similarities and differences and determines the research contribution (Strauss & Corbin, 1998).

Furthermore, both Grounded Theory approaches agree that the GTM establishes the emerged theory which is a 'de facto conclusive analysis', namely as the end product 'rather than a preliminary theory' and there is no need to be tested by quantitative methods (Glaser & Strauss, 1967, p.235). The researcher considers that the substantive theory has 'contextual uniqueness and significance' Bryman (2008, p.378) to the Maltese SMEs and cannot be generalised to other situations due to contextual differences. Hence, the transferability property needs to be substantiated through replication of the study in other countries to promote the validity of such generalisation across different geographical settings, unless such contextual differences are evaluated and applied to the current research to justify or not the validity of the substantive theory in other contextual conditions scenarios.

The following three chapters detail out the theorising and analytic processes carried out to generate the substantive theory on the SCI phenomenon, in line with the referred research methodology to meet the research quality criteria and scientific rigour.

Chapter 4: Data Collection and Analysis based on the Open and Axial Coding Techniques

4.1 Introduction to the overall data collection and analysis three phases

The following next two chapters are dedicated to examine the data of all twenty-two (22) interviews, based on seventeen (17) different firms from the two primary data collection sampling phases, referred as open sampling (i.e. 8 interviews) and theoretical sampling (i.e. 14 interviews) respectively. Secondary data was also used to capture other relevant and supporting data sources to promote data sources triangulation. The secondary data included both paper and electronic documents (Section 5.3.4). The data analysis approaches, consisted from three main phases, based on the three systematic coding stages, referred as Open, Axial and Selective Coding (Section 3.10), and as depicted in Table 4.1 below.

Phase Number	Coding type	Sampling Type	Quantity of	Data Analysis Approach
			Emerged codes	
			or	
			categories	
1	Open Coding	Open Sampling	46 tentative open	CAQDAS with code
(Chapter 4)		based on 6 firms	codes set	memos
		with 8 interviews	(Appendix 9 &	
		(i.e. two of the	10)	
		firms offered two		
		participants)		
2	Axial Coding	Open Sampling	22 tentative	Storyline, mind maps,
(Chapter 4)		based on the	categories	memos, and a code matrix.
		previous data set	(i.e. 7 key & 15	The theorising process
			non-key tentative	shifted from description to
			catergories, as as	conceptual memos (i.e.
			shown	categories memos and
			Table 4.10)	summary memos to
				encapsulate a set of
				categories)
3	Final Selective	Theoretical	1 core category, 5	Storyline, mind map and
(Chapter 5)	Coding	sampling based	main categories	with summary memos to
		on 11 new firms	with 16 sub-	derive the final emerged
		with 14	categories as	theory which is represented
		interviews (i.e. 3	shown	by the final code matrix
		of the firms	in Table 5.1	
		offered a 2 nd	(Total of 22	
		visit)	categories)	

Tuble fill biculture of all the dura concerten and unary sis phases (Traditor	Table	4.1:	Brea	kdown	list o	f all	the	data	collectior	ı and	anal	ysis	phases (Author	•)
---	-------	------	------	-------	--------	-------	-----	------	------------	-------	------	------	----------	--------	----

4.2 Phase 1: Data collection and analysis based on the Open Coding technique

The number of emerged open codes from such an analytic process amounted to 132 codes, with varies frequency counts for each open code, which amounted in total to 1472 counts (Appendix 9). Such **132 open codes** were filtered to generate **46 open codes**, based on a frequency count greater than ten, to

serve as the **initial tentative set of categories**, as referred in Section 3.12.1. The audit trail of such analytic results is referred in Appendices 9 and 10. The completion of this analytic approach with a set of 46 open codes (Table 4.1) put a start to the next Axial Coding stage, as described in Section 4.3.

4.3 Phase 2: Data collection and analysis based on the Axial Coding technique

4.3.1 Introduction to the Axial Coding stage

This second phase, referred as phase 2 (Table 4.1), outlines the Axial Coding analytic theorising process based on the storyline method. The purpose of the first phase was to generate open codes, whilst this second phase builds and refines the emerging theory by establishing the relationship between various concepts or tentative categories with their respective properties and dimensional variations. All emerged tentative categories, through the storyline approach, are mapped on a logic diagram and a conceptual framework, on a one by one category basis, to construct the overarching tentative categories to serve as the initial phase of theory building with potential and substantive ideas on the SCI phenomenon.

The forty-six (46) open codes from phase 1 are again compared with this phase data analysis emerged concepts. The same data set from phase 1 is used for this phase and the analytic storyline approach consists from a set of narratives based memos on the emerged key themes, supported by the selection of key *in vivo* extracts from the transcripts, to bring the participants back to life. The use of several theoretical memos serve as a source of validation of the theorising process, to ensure that the data analysis process follows a scientific and a rigorous methodical approach to produce a set of abstract codes, referred as axial codes (Strauss and Corbin, 1998). Such a phase focuses on the emerged theory, which transforms the previous phase set of forty-six (46) concepts to a new set of twenty-two (22) tentative categories, which are listed in Table 4.2 and also represented by the code matrix in Table 4.10.



 Table 4.2: List of tentative categories from the Axial Coding stage (Author)

4.3.2 Findings from the open sampling data collection under the Axial Coding analytic approach

In this Axial Coding stage, it is to be noted that fifteen (15) out of the twenty-two (22) tentative categories are being referred in the Appendix 11. Such an approach has been taken through a retrospective stance, since initially the actual research data analysis consisted of rich and in-depth theorising, through a storyline approach, of all the emerged theory, with no distinction of any tentative category whatsoever. The researcher had such an indication after the completion of the Selective Coding stage (Chapter 5), since at this Axial Coding stage he had no indication which were going to be the **seven (7) key tentative categories**, as marked in Table 4.2 above and Table 5.1 respectively.

This decision is attributed to the fact that this chapter only focuses on the most significant seven (7) axial codes, in line with the GTM key foundations, where such main themes form the core theorising process to be adopted in the next Selective Coding stage (Chapter 5). Such a stance is also adopted to make the research more manageable and focused on the salient part of the voluminous amount of data with its analysis which are most relevant to the emerged theory. Though, it is to be noted that due importance is also given to such **fifteen (15) non-key tentative categories**, since they still form part of the research conceptual framework foundations, as referred in both Chapters 5 and 6 respectively.

Within Appendix 11, the fifteen (15) non-key tentative categories theorising process includes the same analytic approach as adopted within the other 7 key tentative categories hereby.

4.3.2.1 Managing technology deployment tentative category

The managing technology deployment theme is about the use of technology to facilitate, drive and improve all SC operations and production processes, with higher productivity, through integrated information systems and automation. Furthermore, technology promotes **visibility and transparency across the SC** and **efficiencies** in all operations for the focal firm to be competitive.

To achieve SCI, technology deployment, through the use of **ICT** applications, may either consist from a **common platform** solution (e.g. EDI, ERP and database) and/or **disparate legacy systems** (e.g. workstations) which need to be interconnected with each other so that all these applications need to adapt to the SC and its manufacturing processes requirements. Two participants explained in detail the deployment of technology, by stating that:

'... [another system is used to] monitor the progress regarding material and efficiencies of work carried out per station not from ERP but from other application'; 'In-house systems are used to integrate with the ERP' (2/INT3); 'The ERP system does not cater for all needs being a standard system'; 'The scope is to adapt the product (software and hardware) to the manufacturing processes; 'For the SC to be fully automated, it is not that easy, since organisations ... have various systems in all areas ...' (4/INT5).

An **ERP** is a **proactive and powerful** tool for **integration** and **information management** that enables and drives the process of the focal firm with its SC members more competitively, especially with shared information facilities over the internet. Such a process approach was described by three participants, by stating that:

'The IT integrates and checks the MRP material and other application software, with all the prompts to manage effectively'; 'The role of IT assures the stock levels in a very efficient way'; 'The ERP is shared' (3/INT4); 'The IT is also driving the process of the SC, since suppliers and clients ... use their software ... to place the order or to see the orders' (5/INT6); and 'IT is an enabler of all processes ... to trigger the process' (14TS/INT11).

With an ERP system approach, it is not always possible to meet all the processes minor requirements and hence legacy systems, with a reactive approach, may still be used to meet such needs, as referred by one participant, by stating that:

'The software need to cater for such changes and some end up uses an excel as a time sheet, since to set and tweak the ERP, it is very expensive'; and 'This is a reactive and not real time approach' (4/INT5).

ICT enables SCI by providing a real-time information sharing platform from all systems used at the workplace or remotely through the web, with all other SC members (e.g. firms web portals and networks). Such tools enable an effective information management approach that promotes SC visibility across the SC for better responsiveness and efficiencies but the human element remains key in the relationship management with all SC members, as clearly explained by three participants, by stating that:

'IT systems are used real time to be updated with all information and changes' (2/INT3); and 'The order trend, with suppliers is integrated over internet (IT)'; 'Sales office through technology can manage the selling or buying, have a birds' eye view at one point'; 'Technology is a big enabler ... contributed to large increase in efficiencies'; and 'The on-line is effective and efficient, but the human element remains key to establish a long term relationship. The web portal is an asset but the human element is still needed' (17TS/INT 14).

The aim of the focal firm is to form a **holistic solution** to meet all SC members' **information management** needs and create a **flexible** approach in all SC processes, by **managing change and risks** effectively and to promote **business continuity** (e.g. identification of bottlenecks and delays in deliveries). Such a **flexible** provision by ICT applications needs a minimal effort from the user, since its information management capabilities are based on in-built formulas, which are flexible to calculate all the changes needed to adapt to any new requirements. Such technology enabling and holistic approach, was described by three participants, by stating that:

'...using IT to develop solutions from the data available and support the employees in all daily repetitive work, so that the effort is dedicated to analyse the data effectively and efficiently but generate reports automatically with a click of a button. The above is a business continuity measure against risk of failure not to lose the sale' (2/INT3); 'This helps collaboration between us all as a key issue' (5/INT 6); and 'IT helps out to flag out the issues and mitigate your risks' (11TS/INT8).

ICT facilitates the keep-up of an **audit** trail of all works, such as with the use of bar-coding, to improve on the overall efficiencies, historical records, inventory management, information updates and products follow-up across the SC, as referred by two participants, by stating that:

'IT systems are used to monitor the board per board basis, which stage it is, who was the person who did it' (2/INT3); 'Software is being used that manages information per component'; '... bar coding systems, to minimise and eliminate the stock inaccuracies, [are] used for issue of material' (4/INT5).

ICT enables the collection and monitoring of all activities to support **performance measurements**, through dashboard reporting and business intelligence applications to promote effective **decision making** and **competitive performance**. Such a measuring approach was emphasised by one

participant, by stating that: 'KPIs are used to monitor the progress regarding material and efficiencies of work carried out per station' (2/INT3).

ICT does not always eliminate all **duplication of tasks**, due to disparate applications, of an ERP and various legacy systems. With such a duplication of operations, SCI is still in place, but is against the lean management approach. Such a duplication was referred by one participant, by stating that:

'If SAP is used at both ends, then the data can be shared'; 'These are then keyed in the ERP. This is a double transaction due to the two systems, hence duplication'. 'The pdf need to be keyed, a level of duplication [exists]' (3/INT4).

Technology is also used in **production process technology** applications, such as in automation with advanced manufacturing techniques, to promote better quality, higher productivity and lean management, to achieve **high value manufacturing**, which employs less running costs and improved reliability. Though, it cannot be excluded, that the **human element** intervention is still needed within the process, as referred by two participants, by stating that:

'The technology investment in robotics etc, then the cheap labour is challenged'; 'The IT facilitated the processes; 'With full or semi automation machines (robotics arms etc) the cheap labour is won over, since one has 24/7 shifts and high quality, since human introduce errors'; 'Automation enables detection of efficiencies'; 'There exist processes and products that still need humans' intervention. Machines do not cater for all activities' (4/INT5). 'The production efficiency is higher, around 3 times or less, due to automation ...'; '... [automation] promotes better/higher reliability from a quality perspective' (6/INT8).

Logic diagram of managing technology deployment tentative category

Table 4.3 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Technology	Positive extreme dimensional	Negative extreme dimensional
deployment tentative category	variation	variation
properties		
Technology approach	Proactive	Reactive
Technology based system	ERP or common integrated	Disparate legacy systems
	database with minor use of legacy	
	systems applications	
Technology usage	Facilitate, drive proactively and	Management of basic information
	improve operations and enable	needs with a reactive approach
	informed decision-making	
Technology objective	Adapt technology to all	Adapt the processes to the
	requirements	technology

Automated process	Full automation through	Manual approach based on the use		
ratomated process	technology with high efficiency	of the human element with less		
	teenhology with high efficiency	of the number clement with less		
	reliability and lean	efficiency		
Information management	Manage effectively all information	Employ disparate information		
	to promote SC visibility and	sources with difficulties to meet		
	flexibility to be more responsive to	changes and responsiveness		
	changes and risks			
Information sharing	Shared information systems with	Isolated information systems		
	the involvement of the human			
	element to have a more effective			
	relationship			
Duplication of tasks	Fully integrated ICT based	Disparate ICT based solutions		
	solutions with no duplicated tasks	which needs duplication of efforts		
	(i.e. lean operations)	through multiple data entries		
		between systems		
Tracking of works	Audit trail of all products	No audit trail at all		
Performance measurements	Measuring all operations	No measuring at all		
Conditions	Technology investment in ERP or common databases or disparate lega			
	systems; and type of product determines the use of automation and/or			
	manual operations in manufacturing.			

Table 4.3: Managing technology deployment tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through the technology deployment tentative category, since it establishes an environment where all SC members together with the focal firm, can be interconnected with each other to share a common platform of information through the use of the right set of technological based software tools, either at the workplace or remotely through the web, to promote SC visibility. The technology used to manage all information needs of the overall SC, may consist from a **common information system**, such as an ERP or else a disparate based approach, based on fragmented ICT based legacy applications or else a mix of both. ICT facilitates the keep-up of an audit trail of all works to improve on the overall efficiencies, by measuring performance, to produce all information updates and product follow-up across the SC to assist decision-making. Technology is also used in **production process technology** applications, such as in automation with advanced manufacturing techniques, to promote better quality and higher productivity. The use of technology enables, drives and improves all SC and manufacturing operations. With the deployment of technology, the focal firm is able to adopt all the necessary flexible and proactive measures with minimal duplication of tasks and with the necessary change management and risk management approaches to promote improved effectiveness and efficiencies and business continuity.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing technology deployment conceptual element is considered as part of the SCI phenomenon since it is an action category, where it promotes several consequences, as a result of technology investment in both automation and ICT applications. Automated technology consequences are reduction in human workforce, advanced manufacturing, high quality and reliability, higher productivity and lean operations. ICT applications consequences are SC flexibility, responsiveness and visibility, competitive performance, lean operations, holistic approach, manage change and risks, performance measurement and information management. The relationship of the managing **Technology deployment** tentative category with all other categories is represented by the conceptual framework in Figure 4.1.



Figure 4.1: Managing technology deployment category tentative category (Author)

4.3.2.2 Leading effectively tentative category

The leading effectively theme is about people who need to promote self-direction and commitment, by developing managerial development skills that promote an effective leadership approach at all management levels, to meet all innovative challenges and the established levels of performance and to have more power to control all operations, both within and outside the focal firm, to promote effective and efficient performance.

To achieve SCI, the focal firm needs an effective **leadership approach**, based on the **leadership style** and on the individual leaders' **cultural values**, as outlined by four participants, by stating that:

'Leadership and the team approach is promoting the SCI' (3/INT 4); 'Other firms have the same product but use different management styles and culture' (4/INT 5); and 'Leadership style is like the glue to match all management levels together' (5TS/INT2).

An effective leadership approach promotes staff to be **empowered** and **motivated** at all management levels, to **perform** effectively, in line with the performance benchmarks, as emphasised by four participants, by stating that:

'Teamwork and leadership is very effective at all levels such as strategic, tactical and operational' (3/INT 4); '.... HR incentives exist through performance measurement since one has more direction, benefits and motivation (5/INT6); and '... we use empowerment to take decisions' (6/INT 7); and 'The leadership style is more of cooperation type, 'people type of style', with me roaming around people' (2TS/INT3).

A transformational leadership style is an important type of leadership approach and is used by staff to apply an **innovative** and **proactive** approach to work. Such a transformational leadership approach was referred by three participants, by stating that:

'... [we] share the project works and innovative ideas ...' (1/INT1); 'Projects change to meet other customers' requirements so as to innovate' (1/INT2); and 'Innovation in the process, say the transition from the manual to automation process. The management tools being used are also innovative' (6/INT8).

The **transactional leadership style** is used by staff within the focal firm, to perform through a **cross-functional approach** so as to manage all processes effectively in a collaborative environment in line with **lean management**, to reach all targeted **performance benchmarks**. Such a transactional leadership approach was explained by three participants, by stating that:

'[Focal firm] 'focus[es] on cross-functional operations, open style policy, teamwork, performance measurements' (1/INT 2); and 'Lean applies to all departments' (6/INT8); and 'The departments need to have managers that interact effectively together' (5TS/INT2).

Though, it cannot be excluded that such a teamwork approach, within the focal firm, is function of staff cultural values, due to their silo thinking, which sometimes makes it challenging to achieve the necessary improvements, since:

'There is always room for improvement and there is always rivalry at work between departments' (6/INT 7); and 'The management need to remove all silos, especially from staff operating on the same management level' (5TS/INT2).

Such a transactional leadership approach, in certain cases needs to adopt a situational leadership approach, to effectively manage all SC members, by implementing the right negotiating skills and implementing the right contingent situations, as clearly referred by one participant, by stating that:

'[The focal firm] needs negotiating skills, and you need not be weak with your customers Similarly, [with] the suppliers you need to be strong ...' (5/INT6); 'Such a stance need special measures which are different from other well established sources of suppliers' (5TS/INT2).

Furthermore, for the focal firm to lead effectively, it needs to deeply engage with all SC members with an **integrated management approach** to build on the **power of control** across the SC, as outlined by one participant, by stating that: '.... *the more attachment you have with all SC members (in depth), the more is the better since you will have more control*' (5/INT 6).

The **performance measurements** create a rewarding environment to facilitate an effective leadership approach, since staff are involved and become more committed to meet such targets, which in turn also promotes **continuous improvement**. Such a leadership style measurement culture has been outlined by a participant, by stating that:

'We measure and discuss everyday all operations. Workers are involved in all decisions'; '[Firm] performance is also measured and service issues are being measured and gaps identified to ensure improvement' (1/INT2).

The charismatic and humanistic quality of a leader promotes better engagement by all employees at all management levels, which shows the effectiveness of a human friendly leadership style (i.e. a mix of transformational and servant leadership styles). The leadership style need to be sensitive to the employees' needs, as referred by two participants, by stating that: '.... corporate surveys ... rate the management to employees' effectiveness' (1/INT1); and 'We work as a family' (TS/INT4).

Leading effectively needs the commitment of the focal firm **human element** with the right **leadership** approach and with the support of **technology**, to align all **processes**, both within manufacturing and across the SC, and to promote a **lean management** approach so as to achieve **competitive performance**. The role of technology to achieve an effective leadership approach has been clearly outlined by four participants, by stating that:

'... using IT to ... support the employees in all daily repetitive work' (2/INT 3); 'The IT integrates ... [and generates] all the prompts to manage effectively' (3/INT4); 'The IT facilitated the processes'; 'The IT is also driving the process of the SC....'; 'This helps collaboration between us all as a key issue' (5/INT6).

Logic diagram of leading effectively tentative category

Table 4.4 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Leading Effectively tentative	Positive extreme dimensional Negative extreme dimensi				
category properties	variation	variation			
People cultural values to work in	Teamwork employed across all	Silo thinking with lack of			
a team	management levels with a cross-	teamwork with an ineffective			
	functional approach and with a	leadership approach			
	charismatic leadership style				
Innovative approach	Continuous improvement with a	Status quo approach to			
	transformational leadership-	innovation			
	approach				
Commitment by all people	Proactive approach	Reactive approach			
Technology based processes in	Technology based information	Manual processes both in			
SC and manufacturing	management systems and	information and production			
	automated processes				
Performance measuring	Excellence in performance	Meeting average performance			
	benchmarks and with a				
	transactional leadership				
	approach				
Conditions	Investment in ICT and automation technology; staff cultural values				
	on teamwork deployment against silo thinking; transactional,				
	charismatic and transformational	l leadership styles to fit every			
	situation; and SC members' commitment to interact and work				
	together with the focal firm.				

 Table 4.4: Leading effectively tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through the leading effectively tentative category, as it promotes an empowered and motivated workforce to perform effectively and efficiently. Leading effectively depends on the leadership style and cultural values retained by the leaders at all management levels. Leading effectively mainly needs a mix of both transformational and transactional leadership approaches to promote innovation and a cross-functional approach, both within and across the SC. An integrated approach across the SC assigns more leadership power of control to the focal firm. Performance measurements facilitate an effective leadership approach since measuring drives motivation and commitment to meet and/or exceed all targets. It cannot be underestimated that the leader's humanistic approach (i.e. servant leadership style) has to be considered a key foundation principle behind leading effectively, since all leaders need to be sensitive to all employees' humane based requirements to promote an empowered and motivated workforce. Circumstances may exist when a leader needs to adopt a situational leadership approach to meet contingent situations. Technology deployment facilitates the decision-making process in all operations by providing an effective

information management approach with real-time information provision to promote an effective leadership approach in both SC and manufacturing operations.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The leading effectively conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as effective and efficient performance, lean operations, performance improvement, innovation and continuous improvement. The moderating variables are technology deployment, leader's leadership style and culture and staff culture. The relationship of the managing **leading effectively** tentative category with all other categories is represented by the conceptual framework in Figure 4.2.



Figure 4.2: Leading effectively tentative category (Author)

4.3.2.3 Managing process within the SC and manufacturing tentative category

The managing process within the SC and manufacturing theme is about establishing all the requirements needed by the focal firm to achieve an integrated SC with all its SC members, by deriving a **holistic process approach** to achieve effective and efficient performance.

To achieve SCI, the SC and manufacturing processes of the focal firm must employ **lean management** to **optimise all processes**, both within and across the SC, in **one holistic process** from both a **technical** and **logistics** perspective to improve on the overall **customer service**. A more indepth integrated process approach with SC members exists, when the outsourced activities form an integral part of the focal firm manufacturing operations. Such a process approach requirements, were clearly explained by seven participants, by stating that:

'Internally needs the right structure ... with a holistic responsibility ... have separate entities integrated under the Operations management' (1/INT 1); '[Focal firm] focus[es] on cross-functional operations' (1/INT 2); 'SCI is in depth since the outsourced activities involves part of the core processes' (3/INT 4); 'Lean removed the waste in the processes, to be more responsive to the customer requirements' (6/INT8); '... quality concept is a continuous process approach across all functional areas to make the SC holistic' (5TS/INT2); 'SCI is the integration of processes and assets both internal and external' (13TS/INT10); and '... the manufacturing of today is a process approach based on both the manufacturing and services' (15TS/INT 12).

All manufacturing processes differ from each other much more than the SC processes, but the overall process needs to be lean, streamlined and to adopt a level of **collaboration** with the right allocation of **priorities**, so as to be **flexible** enough to meet all commitments effectively and efficiently and with right **customer service**. The **management leadership style** may also influence the manufacturing process approach to differ even for firms within the same manufacturing sector. The SC and manufacturing process was described by four participants, by stating that:

'Restructure was made and the SC needs to be lean' (1/INT1); 'End of month budget is done to ensure improvement in processes' (1/INT2); 'Priority is carefully monitored ... not to escalate the problems with the customers'; 'Integration and collaboration is needed to promote flexibility to build the relationship' (3/INT4); and 'One full standard system does not exist to work throughout. The peripheral systems yes are standard ... [but] manufacturing still vary'; 'Other firms have the same product but use different processes due to different management styles and culture' (4/INT5).

All operations within and across the SC needs to be **measured** and **audited** to promote an **audit trail** of all activities and to undertake **performance measurement** to serve as reference to establish **performance improvements** of the overall process in line with established **KPIs**. Such an approach is easier to achieve for the focal firm since it is more challenging to implement for SC members outside the focal firm, due to the **different cultures** and objectives. Such a measurement and auditing perspective, was referred by four participants, by stating that:

'For inter-company the performance is more easy to be dealt with due to same values, culture and language with same MBOs. But with local suppliers and others external European suppliers to firm, requires the validation and audit parameters/criteria as established with the corporate KPIs and its suppliers' (1/INT1); '... gaps [are] identified to ensure improvement' (1/INT2); 'We can establish hence through the audit trail, such as the process, the date of production and the people who worked on it' (5/INT6); and 'The holistic approach is governed by the focal firm KPIs' (5TS/INT2).

The overall SC and manufacturing process needs to comply with all **quality standards and sustainable and legal requirements**, to provide a product, with all of its components, with the right quality and in line with the customers' specified requirements. As a result all SC and manufacturing

processes, with all human interventions, need to adapt to all specifications and procedures stipulated by such standards and requirements to eliminate any risk of failure. The quality of processes refer to all **logistics** and **production** activities that range from all human activities and interactions, down to every component that makes up the product. The sustainable measures refer to the impacts on the waste management and products movements, cost and cash flow management and employee wellbeing respectively. The sustainability perspective of the overall SC and manufacturing process approach was highlighted by four participants, by stating that:

'[We] implemented such EHS programme ... system sustainability OK'; 'Manufacturing variance refers to how we are cost effective to the budget'; 'The stock needs to be made available without ending up into cash flow problems' (1/INT2); 'Environment and CSR,... needs traceability on client and supplier waste management processes' (5/INT 6); '... to comply to the respective country standards and audits for certification purposes e.g. ISO and TUV within the global SC' (6/INT7); H & S we use OHSAS 18,001'; 'Operational excellence value is key here, because managers are expected to improve processes, people and systems'; 'Quality is an integral part of the model' (6/INT 8).

Risk management is to be employed in all activities to ensure that at all stages in the process, within and across the SC, meet the expected scheduled performance and are not subjected to any bottlenecks. Furthermore, ICT applications need to be used by all employees as a supporting tool in all their operations, to identify any shortcomings and to remedy at the earliest. Such a risk management approach, was explained by three participants, by stating that:

"...using IT to develop solutions... and support the employees' (2/INT3); "Though, an infallible SC model does not exist. One needs to reduce the chance to end up in these conditions and not be prepared. This is risk assessment, which needs to be shared across the SC with all members' (5/INT6); and 'The products and/or customers are being tracked and identified that have the possibility to fall behind' (6/INT8).

The focal firm and all SC members need to employ a **cross-cultural based knowledge** mindset, which rests on common values and beliefs, by **sharing information** to **collaborate effectively**. The focal firm, with all SC members, needs to negotiate and establish attractive **contractual agreements**, to meet all the process requirements competitively. Such an environment promotes all members **human element** to be **empowered** and to take **informed and effective decisions**, with a level of flexibility, for a **win-win outcome** and with a **continuous improvement** approach. Such a process approach based on knowledge transfer and collaborative environment was referred by four participants, by stating that:

'... material reduction and validation of new resin material used to reduce cost and better quality to improve efficiency, hence the knowledge sharing occurs on these matters'; '... share the project works and innovative ideas ..' (1/INT 1); 'We have 3 shifts and dig deep in case of problems. In fact, Focus groups are organised ... to find in depth ... the bottlenecks'; '...empower their responsibilities across all departments' (1/INT2); '... setting up the

contracts and undergo negotiations with the suppliers ..'; 'Suppliers and buyers are well related between us' (2/INT3); and 'The decisions are also flexible to discuss to promote improvements'; 'The relationship is healthy with suppliers and inform us on new products/changes ...' (3/INT4).

Technology deployment, through ICT tools, such as ERP, facilitates and drives information based processes through prompts and alerts, to implement more **efficient and effective** operations, over the slower manual operations. The internet is providing a means of interconnection to all SC members over the web, which through ICT applications, promote SC visibility, both within and across the SC, to be updated with real time information. With an ERP system approach, it is not always possible to meet all the processes minor requirements and hence legacy systems may still be used. Such an ICT based process environment was highlighted clearly by four participants, by stating that:

'IT solutions ... makes the work processes better, avoid double effort and ... data is generated from barcodes to be more creative'; 'IT systems are used real time to be updated' (2/INT 3); 'The IT integrates and checks,... with all the prompts to manage effectively' (3/INT4). '... the material is available but the planning is still pending to be done manually where all other processes are all automated'; 'The order trend, with suppliers is integrated over internet (IT)'; 'The SC uses cloud computing and groupware systems,which comprises interfaces through the web with sales office ... excel as a time sheet [is used], since to set and tweak the ERP, it is very expensive' (4/INT5); and 'The IT is also driving the process of the SC' (5/INT6).

Similarly, for automated production processes, such machines reduce the level of human intervention in manufacturing by employing less people and at the same gives the innovative edge needed in manufacturing for the local firms to survive and remain competitive, as explained by one participant, by stating that:

'The technology investment in robotics etc, then the cheap labour is challenged since it is still human'; 'The IT facilitated the processes'; 'With full or semi automation machines ... one has 24/7 shifts and high quality, since human introduce errors which are avoided' (4/INT5).

Though, the importance of the **human element** cannot be excluded, since human intervention with its technical and management capabilities is still needed to manage and improve the overall process and also to perform certain tasks which cannot be automated. It is further noted that the human element, in particular processes, has the capability to undertake better inspection of products defects, as referred by three participants, by stating that:

Machines do not cater for all activities'; 'Mistakes occur, when equipment and software is excellent, but they remove the human element [from the overall process]' (4/INT 5); '... the manual labour can identify better some defects' (6/INT 7); and '...although the processes are automated needs the human element to make the job complete' (10TS/INT7).

Logic diagram of managing process management within the SC and manufacturing tentative category

Table 4.5 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Process within the SC	Positive extreme dimensional	Negative extreme dimensional		
and manufacturing tentative	variation	variation		
category properties				
Optimised process	Efficient and effective process	No optimised process with lack of		
	approach with the support of the	human involvement		
	technical and management human			
	capabilities to optimise the process			
Process integration with other	Integrated process based on cross-	Not integrated process		
players	cultural knowledge			
Process monitoring through	Measuring and auditing of the	No measurement at all		
performance measurement	process activities to promote			
	improvements			
Process quality and sustainability	Process in line with all quality	Lacking in both quality standards		
	standards and sustainability	and sustainable measures within		
	measures (i.e. triple bottom line)	the process		
Process streamlining	Lean process with immediate	Process with several shortcomings		
	identification of bottlenecks to			
	remedy at the earliest			
Process automation	Automated activities in information	Manual process approach		
	management and manufacturing to			
	be innovative			
Conditions	Technology deployment in ICT and automation; human technical and			
	management capabilities; contractual agreements; effective management			
	and leadership style; and cultural values of focal firm staff and SC			
	members.			

 Table 4.5: Managing process within the SC and manufacturing tentative category (Author)

 Conceptual summary on the tentative category

SCI is enabled through the managing process tentative category, since the overall SC requires a holistic driving force with the right leadership style, to pull all SC members together in a continuous improvement and collaborative approach with the focal firm, to achieve effective and efficient performance. Lean management enables the overall process optimisation to reach the expected productivity levels, in line with the established focal firm's performance measurements and KPIs. Such lean operations need to be validated to respect all quality standards and sustainable and legal requirements, to be awarded the certified approvals. Risk assessments of all actions need to be in place to identify any unforeseen bottlenecks at the earliest and to remedy in time, to minimise the possibility of any process disruption to promote business continuity. Such a holistic process approach, with the use of technology in ICT and automation, with the right engagement and interfacing capabilities of the human element, are crucial requirements to optimise the overall process.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing process within the SC and manufacturing conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as optimisation of all processes, lean operations, a holistic approach to all operations, SC flexibility, win-win outcome, continuous improvement, efficiency and effectiveness, sustainability, customer service, business continuity and competitive operations. The moderating variables are lean management, human element capabilities, collaboration, cash-flow management, effective management style, performance measurement, audit of operations, risk management, quality standards and sustainable and legal compliance, risk management, information sharing, knowledge management and technology investment in ICT applications and automation. The relationship of the managing process within the SC and manufacturing tentative category with all other categories is represented by the conceptual

framework in Figure 4.3.

Human element capabilities



Figure 4.3: Managing Process within the SC and manufacturing tentative category (Author)

4.3.2.4 Managing supply chain integration tentative category

The supply chain integration (SCI) theme is about taking all the necessary measures to build a relationship between the focal firm and all SC stakeholders so as to create a **trustful** environment and a team work approach, depending on each SC member role. Such a teamwork approach needs to exploit each SC member capabilities with the necessary synergies, in line with all required **quality** standards, sustainable measures and legal requirements, to promote lean operations with the necessary flexibility and to perform effectively and efficiently so as to achieve a competitive edge.

SCI is key within the overall manufacturing and SC operations, since it needs that all collaborative initiatives, between the focal firm together with all SC members, to be linked in a common platform to meet jointly all quality standards and the triple bottom line, so that the overall SC performs effectively and efficiently. Such a SCI holistic approach was referred by four participants, by stating that:

'SCI is the core in the daily activities'; 'If you follow the success of the organisation, this is attributed to the SCI. This is not an overstatement but a reality' (1/INT1); 'Practically the firms are merging with each other'; '... one has no option to stay out of the SCI initiatives, due to regulations and other commitments you need to integrate otherwise you are out of business' (5/INT6); 'SCI for us is cross-functional and integrated approach' (1TS/INT5); and 'We have two forces. The customer satisfaction and the production (profit)' (10TS/INT7).

SCI may vary from a **close** (in-depth) to **arm's length relationship** with other SC members. The integration depth is function of each SC member's role vis a vis the focal firm. SC members who form part of the Corporate or else deliver a core part of the focal firm's products, backed by years of experience in the relationship, tend to become more closely integrated, with a higher level of commitment in all interactions, in such a way to become partners of each other. Other SC members who are involved in the basic exchange of orders have a minimal relationship engagement and are referred as suppliers. Such a SCI variation between SC members has been referred by two participants, by stating that:

'The SCI depth is a mixture. We have suppliers of the compressors, as the core activity since our product builds on their product and hence these suppliers are partners and have an indepth relationship, based on years of experience ... but others are referred as suppliers, that give us commodities products' (5/INT 6); and 'We are a bit fragmented and not integrated' (6/INT 7).

For an **in-depth integrated SC** the focal firm with all its SC members shall deploy a cross-functional **teamwork** approach between people, to promote **lean management** with a **holistic perspective**. Such an integrated approach promotes a more **collaborative effort**, between the focal firm and the different

business units of other SC members, both in the upstream and downstream sides of the SC, so as to adjust to the necessary **priorities** and to promote **SC flexibility**. It cannot be excluded the importance of the **knowledge level** and capabilities of all people within all business units to have the right expertise in place for the job assigned, so as to avoid any person to be forced to a situation in 'juggling many balls' (6/INT7). There must be **contractual agreements** to formally establish the nature of the **commitment** between SC members, but a **trustful environment between people** is critical for such a relationship to give an effective customer service. Such a collaborative approach between all SC members has been described by five participants, by stating that:

"...integrating through an operations department concept with a holistic responsibility with all the roles and have separate entities integrated under the Operations management' (1/INT1); "... all orders are managed according to the priority"; "Integration and collaboration is needed to promote flexibility to build the relationship"; "Training is given to eliminate waste (zero defects) more focused on the production"; "Even office work need to be lean" (3/INT4); "When a client engages with a manufacturer, both engage in a trustful environment ... apart from the **formal contracts**" (4/INT5); "One needs to build the SC and have expertise for each section, ... to focus on some of the key areas of the SC'; "If you were not integrated as the way we are, it will limit the customer service due to fragmented operations" (6/INT7); and "... effective and efficient SCI [depends] on the human element for the relationship between people" (9TS/INT6).

When the focal firm is subjected to a poor performer SC member, it shall adopt vertical integration initiatives, if possible, due to the contractual obligations, so as to bypass or solve such a problem, in line with the lean management approach. Once the contractual agreement is terminated, the focal firm may either take over or else outsource such a role. A poor performer SC member leads to an arm's length SC member relationship and a fragmented SC. Such a problematic situation, with other SC members and its solution, has been described by four participants as follows:

'Suppliers' control is being reduced through vertical integration'; 'Such a vertical integration promotes lean management'. 'Ownership is in the input and process but not in the output to the customer' (1/INT 1); 'The objective of the focal firm is to transfer all the manufacturing operations locally, referred as vertical integration'; 'Spin-off from [Corporate to an intermediary] as for the sales, delivery and transport ... Malta has been restricted' (1/INT 2); and 'We are obliged contractually to use [downstream intermediary] to reach all Europe' (1TS/INT5).

The **overall performance** of an integrated SC shall improve on the **flexibility** through **information sharing** and **communication** initiatives by all SC members, since '*Communication is a must for the* SCI depth, that all data arrives to all people and at the same time' (6/INT8) and 'We use SCI, by talking more with each other, departments are cross-functionally operating'; 'We are more flexible to meet each other' requirements' (16TS/INT 13).

The **leadership** approach shall empower employees and shall be used as a driver for the **integration** between staff. It can either takes the form of a **transactional leadership** approach, based on various performance measures, or else takes the form of a **transformational leadership** approach, based on an innovative and change management perspective of all operations or else a mix of both type of styles to adapt to the situation. Such a transformational leadership and transactional leadership approaches, were referred by three participants, by stating that:

'... [we] share the project works and innovative ideas ...' (1/INT1); 'Projects change to meet other customers' requirements so as to innovate'; '[Managers need] to empower their responsibilities across all departments and not focus on a departmental level only'; '[Focal firm] focus[es] on cross-functional operations, open style policy, teamwork, performance measurements' (1/INT2); and 'Leadership and the team approach is promoting the SCI' (3/INT 4).

The focal firm needs to integrate itself, whenever possible, with suppliers and customers through collaborative agreements in stock supplies, outsourced works and in all sales deliveries respectively since the SC is strong as its weakest link. Such an integrated approach, has been described by two participants, as follows:

'We bring the raw material and supply to local sub-contractors that do the moulding'; 'The sales companies coordinate with the logistics centre to promote integration through the same system' (3/INT 4); and '... if the manufacturers are not given the integration with the suppliers, they will not buy from them' (4/INT5).

ICT remains an **enabler** of the overall SC integrated process. **ICT** improves on the **SCI depth**, since it promotes **SC visibility**, by having all SC members updated and aligned with all **information**. Such **information management** shall assist the **risk management** across the SC, since immediate identification of all shortcomings is possible through such information, so that the affected SC member/s take all the necessary timely actions and remedies to promote **business continuity**. The deployment of ICT applications has been described by two participants, by stating that:

'... using IT to develop solutions from the data available and support the employees in all daily repetitive work The above is a business continuity measure against risk of failure' (2/INT3); '... monitoring is also done today by smart phones, where managers are mobile and can see the orders, make new orders and raise issues. Thanks to these tools these are pushing more integration across the SC' (4/INT 5).

The deployment of **technology** in manufacturing, by using advanced and integrated processes, with the validation of the overall production process, promotes **SCI** within the processes through a **high value manufacturing** approach, since: *'The [use of] IT facilitated the processes. Apart from the IT software there is the Automation machines ...'; 'With full or semi automation machines (robotics arms*)

etc) the cheap labour is won over' (4/INT5); and 'Validation of processes in the shop floor ensures this integration' (6/INT8).

Logic diagram of managing SCI tentative category

Table 4.6 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing SCI tentative category	Positive extreme dimensional	Negative extreme dimensional		
properties	variation	variation		
Teamwork approach	Synergies in all actions between people within and across the SC in line with all quality standards and the triple bottom line	Conflicting approach between people with fragmented operations across the SC		
SC level of control	Ownership of various SC tiers promotes improved control management (i.e. vertical integration)	Ownership of focal firm only limits the control on other SC tiers especially for low performing SC members under contractual obligations		
Integration level of depth	Closely integrated approach through an information sharing platform to promote an effective and efficient customer service	Arm's length integrated approach with lack of information sharing and gaps in the overall customer service		
Knowledge capabilities of SC members	Staff with specialised skills and with the right expertise	Staff end up to perform multi- tasking with lack of expertise in each task		
SC holistic performance	SC members with full commitment and support to focal firm	SC members who perform below the performance expectations of the focal firm		
Conditions	Technology deployment in ICT and automation; trust between SC members and the focal firm staff; effective leadership approach; and contractual agreements (e.g. technical and commercial)			

 Table 4.6: Managing SCI tentative category (Author)

Conceptual summary on the tentative category

SCI within the focal firm and outside with all SC members may vary from a close to arm's length relationship. There exist **contractual agreements** to establish formally the nature of the commitments between SC members, but a **trustful environment** is crucial for such a successful relationship. A level of consensus and compromise between all SC members is always needed to make all SCI initiatives fit. For an **in-depth integrated SC** the approach to be taken by the focal firm, both within and outside it, needs to be based on a cross-functional **teamwork** approach, to promote **lean management** and **optimised processes** for a **win-win** outcome for all SC members. The focal firm may adopt **vertical integration**, to employ improvements in its **lean management approach**, by
replacing poor performing SC members, once the contractual obligations are terminated. In an **arm's** length SC, the focal firm mainly focuses on the internal synergies and keeps a distance with the outside SC members, causing a fragmented SC. The **overall SC performance** improves, due to the synergy effect of all the members through such a holistic stance. The leadership style has a direct influence on the overall SCI approach, since all people within all SC tiers are motivated and empowered to change and innovation and at the same time perform their daily operations effectively and efficiently. The ICT remains an enabler of the overall process within the SC. ICT improves on the SCI depth, since it promotes SC visibility, by having all SC members updated and aligned with all information. Such information management assists the risk management approach across the SC by identifying any shortcomings at the earliest. The deployment of technology within manufacturing, by using advanced and integrated automated processes based on high value manufacturing, further promotes SCI within the overall manufacturing process.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing SCI conceptual element is an action category, where with the support of various moderating variables, promotes several consequences, such as SC visibility and flexibility, sustainability, effective and efficient performance, effective customer service, optimised performance, holistic process approach, business continuity, win-win outcome, lean operations, high value manufacturing and competitive performance. The moderating variables are lean management, performance measurement, change and innovative approach, quality standards with environmental, economical and social practices and legal responsibilities, collaboration, trust, risk management, leadership approach, information management, contractual agreements, information sharing and technology deployment in ICT applications and automation. The relationship of the managing SCI strategy tentative category with all other categories is represented by the conceptual framework in Figure 4.4.



Figure 4.4: Managing SCI tentative category (Author)

4.3.2.5 Managing the SC strategy category

Managing the SC strategy theme is about a strategic approach to SCM whereby a decision needs to be taken on the SC set-up with respect to the focal firm with all other stakeholders, to perform effectively

and efficiently. Such a decision needs to be taken in-conjunction with the manufacturing strategy. Such a SC strategy determines the established SCI depth of relationship between all SC members, being function of the role each SC member.

Managing the SC strategy is about a pull-SC strategic approach which places the customer at the heart of the SC strategic significance since it initiates the overall process based on the make-to-order strategic approach. The SC strategy needs to decide on the SCI strategy which may vary in the depth of the collaborative relationships with various stakeholders, both within the focal firm and outside it, to perform all activities in an optimised manner and with the right priority of all orders, for a win-win outcome. The internal integration remains of crucial importance to the overall integrated efforts across the SC. The integrated approach needs to effectively manage all the different individual objectives of all stakeholders to promote a unified and a common approach across the SC, with minimal conflicts and a balanced approach, since the **overall SC strategy** is function of all the SC members' **individual strategies** together with the **focal firm strategy**. The role of the SC strategy has been referred by five participants, by stating that:

'The intra SC has been a CSF' (1/INT1); '.. it is the pull system or sales driven, since we do not build to stock, except volume runners' (2/INT3); 'We have suppliers of the compressors, as the core activity ... hence these suppliers are partners ... but others are referred as suppliers' (5/INT 6); 'The various stakeholders are after the same criteria but these may be in conflict with their individual targets/criteria'; '... the cross-functional approach, levels out or is balanced vis a vis the individual interest' (6/INT8); and '... all the measures are needed and all are important since you are strong as the weakest link' (10TS/INT7).

The SC strategy adopts more generic and standard SC processes, relative to manufacturing practices, since the latter is more sector-specific with dedicated processes, with a level of variation from one manufacturing sector to another one. The difference between the SC and the manufacturing strategies has been referred by a participant, by stating that:

'This process is the core of the company, namely the production and manufacturing differs a lot' [from one sector to another]'; 'The standardised processes are the peripherals of the SC such as client integration and raw material ordering' (4/INT5).

All SC operations, both within and outside, are audited to promote sustainable operations and legal compliance. Such a sustainable approach has been described, by two participants, by stating that:

'EHS is the focus ...'; 'The results obtained with the audit were very promising ... system sustainability OK'; 'Environment and CSR [are being respected]' (1/INT 2); and 'Legislation such a **WEEE** [is also respected]' (2/INT3).

The SC strategy must undertake the **selection of suppliers** with respect to the trademark and products quality to have the right materials for manufacturing through an effective supply management approach in line with lean operations, without ending up into stock-out situations or cash flow problems due to excess stock. Furthermore, suppliers or third party manufacturers, who may also supply products or forms part of the focal firm manufacturing process, need to have their quality compliance assured. Such an assurance can be achieved by incorporating all the necessary control measures on all SC members, through auditing of their operations, both by the focal firm and by external certified bodies, to identify any risk associated with cutting corners or shortcomings in any of the quality standards. Such an effective management of supplies has been referred by four participants, by stating that:

'But with local suppliers and other external European suppliers to firm, requires the validation and audit parameters/criteria' (1/INT1); 'We negotiate on the trademark from the available suppliers due to the specs compliant with the design, since not all components are 100% compatible' (2/INT3); 'The products need to have all details according to that specified, to be UL certified'; 'Quality manager monitors samples, to operate a lean process with the right quality, defects identification, inspection etc'. (3/INT4); and 'The stock needs to be made available [but] without ending up into cash flow problems' (5/INT6).

The suppliers and third party manufacturers, who form part of the overall SC, are not only required to meet the quality specifications, but are also obliged to respect all delivery lead times not to disrupt the overall SC timing operations, as one participant outlined, by stating that:

'The supplier had promised to send a number of boards as agreed. We checked for such a delay and discovered such a shortcoming. He informed us that he had a problem in the manufacturing process' (5/INT6).

The SC strategy of firms within Malta, as a Small Island State, requires an effective logistics management to overcome all the challenging objectives associated with the selection of the **logistics routes and delivery modes of transportation**, considering the relative high volume of export and import transfers over the low volume in local sourcing. Three participants explained such a logistics challenging situation, by stating that:

'Logistics is the main problem in Malta' (2/INT3); 'Most of the products are for export, locally the market is small' (3/INT4); and 'SC in Malta I consider it as a phenomena, with various difficulties since one needs to consider various factors ... weather is a factor, ... we are surrounded by sea and considering that sourcing locally is minimal' (5/INT6).

The operations, under the SC strategy, are responsible of all materials and products movements needed within manufacturing, so that the final product is delivered to the customer in the right time, at the right place and with the right price. Such an approach needs that the focal firm, with the support of all SC members, builds a quality based process associated with both manufacturing and with all

collaborative initiatives with all stakeholders. Such a process needs to have full SC visibility of all operations to measure performance, through the provision of all the information, so as to be flexible enough to effectively adapt to any suppliers' delivery problems, the customers' changing requirements and the necessary after sales support with any reverse logistics requirements. Such a SC strategic approach, has been referred by four participants, by stating that:

'The customer has the right to return the product' (2/INT 3); 'Integration and collaboration is needed to promote flexibility to build the relationship' (3/INT4); 'The client [as the focal firm] has access to suppliers, the stock available and the customer [focal firm] is aware of the order and the shipping details' (4/INT5); and 'Quality is an integral part of the model and cannot be separated from the SC and manufacturing process. Quality is part of the product, and part of the communication with respect to timely feedback'; '... the right product, at the right time and at the right price ... the right [product] quality' (6/INT 8).

Contractual agreements are needed to formalise the collaborative relationship with all SC members to meet all quality standards and also to cover both legal and sustainable requirements. This explains the need to have the right manufacturing strategy in-conjunction with the right SC strategy, so that both strategic approaches are weaved together to derive a holistic SCI management approach. SCI strategy remains crucial both within and outside the focal firm, especially with SC members who are key players within the SC set-up. Such a holistic process approach is needed in both manufacturing and SC operations and with an in-depth collaborative effort between all involved to promote lean and sustainable SC operations. Such a holistic approach has been referred by six participants, by stating that:

'Internally there is SCI' (2/INT3); 'SCI is in depth since the outsourced activities involves part of the core processes'; 'Even office work need to be lean' (3/INT4); and 'SCI today is not an option to integrate or not. The internationalization in the business world, one needs to integrate, to move in line with the market'; '... one has no option to stay out of the SCI initiatives, due to regulations and other commitments' (5/INT 6).

The focal firm, in its SC and manufacturing strategies, needs to decide whether to go for **vertical integration** or **outsource** certain activities, with the objective, in either case, to achieve **lean management** and be more **competitive**. If the decision results in outsourcing, there is the need for the focal firm to promote an in-depth SCI set-up with the relevant SC member, especially if the outsourced part is part of the core activities. Wherever possible, local sourcing is preferred from overseas, due to proximity advantages. Such a decision is not a quick fix solution but a long term strategic approach. Such a SC strategic decision, in-conjunction with the manufacturing strategy, has been referred by three participants, by stating that:

'Getting a supply from Mexico is not lean'; 'Malta facility is an intra perspective and lean, since you are taking ownership of the SC tiers' (1/INT1); 'We are looking for the long term and not only focussing on the short term problems'; 'In Malta subcontracting is cheaper,

stock is kept low,... customer's market is closer, port is closer and work more efficiently having all the resources in hand' (1/INT2); 'SCI is in depth since the outsourced activities involves part of the core processes' (3/INT4).

SCI is a growth SC strategy and to be maintained may need the focal firm to increase its SC ownership span, especially if other SC members are not meeting the focal firm's expectations, as one participant explained, by stating that:

'The business model is working but can operate more effectively and efficiently if taken ownership of this missing part, due to opportunity of growth is large but with [such sales and marketing downstream intermediary], this is a bottleneck'; '... one of the 5 strategic priorities on international growth... needs the right manufacturing platform with an integrated SC ... ' (1/INT1).

ICT applications, through a common database (e.g. ERP) enable and drive all processes to promote visibility and traceability of all work and products all across the SC. It cannot be excluded that ICT applications may be limited or based on fragmented legacy systems that are not interfaced with each other and as a result introduce a substantial level of duplication in all tasks. Such an approach with an ideal common database solution or else with a combination of applications, is referred by two participants, by stating that:

'... [another system is used to] monitor the progress regarding material and efficiencies of work carried out per station not from ERP'; 'In-house systems are used to integrate with the ERP' (2/INT3); and 'The ERP does not cater for all the company needs and firms use such satellite systems'; 'For the SC to be fully automated, it is not that easy, since organisations originally used to have various systems in all areas'; 'This is an open process thanks to the software available, where a closed loop exists between manufacturer, client and supplier'; '... interfaces through the web with sales office, in the global world, have access to all information'; 'On the other side of the SC, the firm also knows the suppliers' stock levels, through access given by suppliers to the firm ... ' (4/INT5).

Logic diagram of managing the SC strategy tentative category

Table 4.7 outlines the main characteristics of this tentative category in a logic diagram (Strauss and

Managing the SC strategy	Positive extreme dimensional	Negative extreme dimensional	
category properties	variation	variation	
Pull SC driven demand	Make to order except in make to	Make to stock of all products	
	stock for volume runners		
SCI depth	Close collaborative relationships	Arm's length relationships both	
	internally and externally (e.g.	internally and externally with	
	partnership with SC members	fragmented and unbalanced SC	
	engaged in core operations)	operations	
Quality conformance across the SC	All SC members (including the	No auditing of SC members on	
to manage risks	focal firm) are audited for quality	quality at all	
	compliance		
Risk management of all SC	Identify all shortcomings through	No risk assessment at all	
operations	performance measurement to		
	remedy at the earliest (e.g. stock		

Corbin, 1990).

	and cash flow)			
Supply management	Effective lead time management Ineffective lead time management			
SC Operations	Effective and efficient standard and	Lack of effectiveness and		
	lean operations with a holistic	efficiencies within the SC standard		
	integrated approach across the SC	operations and with a fragmented		
		approach		
SC visibility	Information sharing mechanisms in	Lack of information sharing		
	place to provide all the needed			
	information (e.g. traceability)			
Logistics management	Delivery of merchandise at the	Inefficient and ineffective logistics		
	right location, at the right time and			
	at the lowest cost			
Strategic planning horizon and	Long term effective solutions	Quick fix temporary solutions		
significance				
Sustainability and legal compliance	Compliance to all sustainable	Lack of sustainable operations and		
	measures (i.e. triple bottom line)	legal shortcomings		
	and legal requirements			
SC ownership span	Ownership span is increased to	SC effectiveness and efficiencies is		
	promote effective and efficient	subject to other SC members, with		
	operations	low performing intermediaries		
SC Flexibility in operations	Adapt to all changing requirements	Lack of flexibility to meet changes		
	by SC members	by SC members		
SC growth initiatives	Investment in greater SC span	No investment to increase		
	ownership to ensure effectiveness	ownership at all		
	and efficiency			
Conditions	Manufacturing sector; contractual ag	reements; technology deployment in		
	ICT applications; and firm location within a Small Island State with low			
	local sourcing.			

Table 4.7: Managing the SC strategy tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled by the right SC strategy tentative category, since the focal with its pull SC strategic approach, needs to establish an integrated approach with all of its SC members, to optimise all operations and to perform effectively and efficiently for a win-win outcome. The internal integration of all the operations within the focal firm constitutes the key foundation of the effectiveness and efficiency of the overall SC strategy. Such a SC strategy needs to promote quality assurance in all products and processes and also sustainable operations, with legal compliance. Such sustainability is assured by undergoing the necessary performance measurements and auditing of all operations and processes within and across the SC. Such a measurement scenario is needed to establish at the earliest all shortcomings from the scheduled performance, by taking the most effective decisions to promote business continuity through an effective risk management approach. Supply management is effectively managed by ensuring that all products meet all quality standards and all lead times in all

deliveries respectively, so as to manage cash flow effectively and promote lean operations. Logistics management needs to establish the most cost effective and efficient routes for all deliveries and at the same time promote sustainability, considering the challenges subjected to firms located in Small Island State countries. SC visibility is needed for all collaborative initiatives with all SC members through effective information sharing on all requirements. Contractual agreements are put into place to have a formal commitment in all collaborative initiatives. The SC strategy needs to determine the most effective span of operations and SCI depth (i.e. close relationship or an arms' length approach) with all different SC members, and whenever necessary to undergo vertical integration or outsourcing to other parties. Such a SC strategy on the span of operations and the SCI depth also depends on the manufacturing strategy. The SC strategy in-conjunction with the manufacturing strategy need to adopt a closely integrated approach depending on the role of the SC member who is either: a single supplier which provides customised products or services; or a contractor who performs a core part of the manufacturing process; or an intermediary in the downstream side of the SC, who acts a bridge to the customer. Technology is the enabling tool in both production and ICT applications. Technology may be deployed to enhance the production processes to increase the depth of automation and similarly to enhance information systems through the deployment of common databases to provide effective information sharing with a common system to overcome the shortcomings provided by the use of fragmented legacy systems.

Tentative relationship between all emerged potential categories in line with the coding paradigm

Managing the SC strategy conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as customer service, optimised processes, win-win outcome, priority management, sustainability, SC visibility and flexibility, holistic strategic approach, competitive operations, business continuity and lean operations. The moderating variables are collaboration, logistics management, cash flow management, information management, SCI, quality processes, lean management, contractual agreements, supply management, risk management, effective management and leadership, performance measurement, technology deployment and auditing operations. The relationship of the managing the SC strategy tentative category with all other categories is represented by the conceptual framework in Figure 4.5.



Figure 4.5: Managing the SC strategy tentative category (Author)

4.3.2.6 Managing the manufacturing strategy tentative category

Managing the manufacturing strategy theme is about the need for the focal firm to deploy the right manufacturing approach, through the make or buy decision, to be in line with all engineering practices, quality standards and all sustainable and legislative measures to adopt a lean management with a continuous improvement approach. Such a manufacturing strategy needs to consider the SC strategy, so that all the SC members perform effectively and efficiently in conjunction with the focal firm, in either outsourced works or in supplies deliveries.

Managing the manufacturing strategy, whenever possible, being dependent on the type of manufacturing sector, needs to focus both on automated SC processes and automated manufacturing processes to gain from high efficiencies in SC operations and in manufacturing productivity to promote lean management. The automated manufacturing approach needs to deploy advanced manufacturing processes to achieve high value-added manufacturing, in line with the required technical engineering manufacturing standards (e.g. ISA) and product quality standards (e.g. ISO, GMP and BSI). Managing the manufacturing strategy requires the effective management of all manufacturing and SC processes, as indicated by three participants, by stating that:

'... [there are] processes [which are] still done manually ... This will be a bottleneck in the supply chain. That is the material is available but the planning is still pending to be done manually where all other processes are all automated' (4/INT5); '... to comply to the respective country standards and audits for certification purposes (e.g. ISO and TUV) within the global SC' (6/INT 7); and 'Quality is an integral part of the model and cannot be separated from the SC and manufacturing process'; 'The product is not high value added but we use high value manufacture' (6/INT8).

It is a well established fact that a one size fits all manufacturing model does not exist. As a result not all manufacturing processes can be fully automated and there is the need of the human element within manufacturing. As a result the manual intervention, with its lower productivity and accuracy levels, needs to be managed effectively and efficiently, not to create a bottleneck in the overall process. The human element based manufacturing activities are: process re-engineering; linking of all the processes; flexibility in certain operations to meet any variations in the processes; and detection of certain problems with a more effective approach than automated testing. The role of the human element has been described by two participants, by stating that:

'There exist processes and products that still need humans' intervention'; 'Machines do not cater for all activities'; '... the production and manufacturing differs a lot [from one sector to another]'; 'As a result one full standard manufacturing model does not apply'; 'Mistakes occur, when equipment and software are excellent, but they remove the human element from it ...'; 'They need to undergo business process engineering effectively and deliver the training needs to staff appropriately' (4/INT5); and '... the manual approach can adjust for variations [better]' (6/INT8).

Similarly, the SC processes are not always possible to be all automated vis a vis its information management processes, due to particular manual based tasks within the overall process. In the pharmaceutical sector, for example, the manufacturing process needs a high level of validation of all processes, both within manufacturing and across all SC operations, with strict legal compliance to

provide advanced medical care products without compromising human health. The level of automation in the SC operations is function of the manufacturing sector, as described by one participant, by stating that:

'... [there are] processes [which are] still done manually (such as excel sheet for planning) but others are OK. This will be a bottleneck in the supply chain'; 'There exists processes and products that still need humans' intervention' (4/INT5).

The manufacturing strategy needs to determine the make or buy decision, to establish whether a manufacturing process, is outsourced or performed in-house, to promote competitive operations. With such a strategic decision, the focal firm determines the span of the manufacturing processes to promote efficiencies in line with lean operations (e.g. outsourcing in cases where there exists opportunities for lower production costs and higher quality) and effectiveness (e.g. outsourcing to focus better on the core activities of the focal firm or to meet large demands exceeding the firm's capacity). Though, it cannot be excluded, that some SC members may be assigned to perform a core part of the manufacturing process, due to the nature of the manufacturing sector or in other cases SC members are provided the raw materials and the moulds to perform part of the manufacturing. In such a case SCI efforts need to be given a high level of priority and with special attention to the intellectual property protection, with the result that such SC members become partners with the focal firm and are no longer treated as suppliers or subcontractors. As a result such a strategic manufacturing decision on the 'make', depends on a complex set of variables such as: competitiveness vis a vis the cost and quality as the result of full ownership of the whole process; type of activity (i.e. whether core and noncore); centralised or decentralised level of control over the operations; direct customer contact; volume of units to be manufactured relative to its firm's full capacity; cost feasibility of engagement in the activities to get a competitive price; local sourcing to gain from proximity issues; supplier's expertise; availability of capabilities, capacity and resources to manufacture within the focal firm; funds availability for the investment to make in-house; intellectual property protection importance; and investment recovery time-span to make in-house. Such a make or buy decision to promote lean operations, has been described by nine participants, by stating that:

'Intellectual protocol (IP) is an issue so to as keep the IP of the product [protected]' (2/INT3); 'SCI is in depth since the outsourced activities involves part of the core processes' (3/INT4); '... local sub-contractors ... do the moulding'; 'Our volume is low to outsource a large part of the work' (3/INT 4); 'Lean removed the waste in the processes, to be more responsive' (6/INT8); 'Once the volume is high enough it is feasible to be in-house otherwise we outsource' (8TS/INT4); 'There are some functions which cannot be farmed out ... [to avoid] road to failure' (9TS/INT6); 'Some orders may be small and we hence outsource since this will be expensive for us' (10TS/INT7); and 'The outsourcing is more expensive then working with your staff and you are also missing out the client information and requirements' (13TS/INT10).

Such operational efficiencies within manufacturing, in line with lean management initiatives, may be improved with a continuous improvement approach, either by undergoing cost savings in materials purchasing, or by using new and cheaper raw materials through ongoing research in new materials or through process re-engineering to manufacture at lower costs. Supply management efficiencies may also be improved, whenever possible, by employing a consignment stock approach with suppliers, to save on running operational costs with better cash flow management and improve flexibility in manufacturing. Such cost efficiencies measures, has been described by two participants, by stating that:

'... to ensure improvement in processes, cost savings in purchasing, [and] operations efficiencies' (1/INT2); and 'we use consignment stock, ... we have more supplies and assist to be more flexible to change plans' (6/INT 8).

The manufacturing process needs to adopt a lean management approach and adhere to the established quality standards and all sustainable measures and legal requirements, by maximising the use of resources and eliminate all waste in all processes, both within manufacturing and SC operations. To meet such standards and requirements, the focal firm needs to audit all operations, both within and outside it, to promote assurance and compliance by all involved. Such a quality and sustainable approach in all SC and manufacturing operations, has been described by four participants, by stating that:

'Manufacturing variance refers to how we are cost effective to the budget' (1/INT2); '... to operate a lean process with the right quality' (3/INT4); 'Environment and CSR, we have to separate waste, and ISO 14000 is an environment standard, needs traceability on client and supplier waste management processes' (5/INT 6); and 'Lean removed the waste in the processes, to be more responsive to the customer requirements and is faster' (6/INT8).

For the manufacturing process to be **innovative** and competitive, does not require cheap labour, but depends on the manufacturing strategy to implement increased investment in automated production technology, to promote an increase in the productivity measures, improve on the rejection rates and maximise on the value-added operations through value stream-mapping deployment within the process. Another **innovative approach** is to undergo the necessary **change management** approach in products design and its components to promote availability of stock. **ICT applications**, through an effective information management approach in all workflows, also enables all processes to be **innovative** and optimised and at the same time provides traceability and auditing of all key activities, such as products work-in-process; suppliers' lead time; logistics timely deliveries and inventory levels. Such a technology based process, also enables the identification of any shortcomings at the earliest, so as to remedy accordingly and to eliminate any risk of failure. Such **innovative strategic decisions**, inconjunction with the mother company, wherever applicable, are needed to promote an effective **strategic behaviour and decision-making approach** of all implemented processes by the focal firm,

across all the three management levels, namely **strategic, tactical** and **operational** and externally with all SC members, to promote an **effective risk management approach**. **Performance is measured** to determine the overall strategic performance of the focal firm in line with the **focal firm's KPIs**. Such a competitive manufacturing operational approach has been described by nine participants, by stating that:

'... capital investment for a manufacture area ... which shall be state of the art' (1/INT1); 'R & D is the focus of the firm to ensure that the product is cutting edge ...' (2/INT3); 'Risk management is important since all operations are being monitored to deliver' (3/INT 4); 'A SC from A to Z, the cheap labour does not influence and one can become innovative'; 'The technology investment in robotics etc, then the cheap labour is challenged'; 'One needs to ensure an audit trail or traceability of every product'; 'Defective products need to be detected at the earliest'; 'Robotics, software and BPR with proper systems is leading to a workflow from the day of the order to the final product; 'The IT facilitated [all] the [manufacturing] processes'; 'Teamwork and leadership is very effective at all levels such as strategic, tactical and operational. Malta takes all the levels decisions' (3/INT 4); 'The turnover increase through IT, with a little increase in workforce' (4/INT5); 'Value-stream boards to outline the floor operations... for the key lines in the firm' (6/INT7); 'The management team has the KPIs and these build on the shop floor efforts and performance measures' (6/INT 8); 'Manual labour leads to failure due to the high costs required, one needs automation to be productive' (8TS/INT4); and 'A stitch in time saves nine, ... if a defect is not established from the start, then rework will be very expensive' (10TS/INT7).

Technology deployment plays a key role within manufacturing since apart from the investment in equipment for automation to achieve advanced manufacturing techniques and processes, ICT applications are used to facilitate all processes within manufacturing and SC operations by providing to all staff the real-time information needed to take informed decisions in a timely manner across the overall process. The role of ICT has been described by three participants, by stating that:

'IT systems are used real time to be updated with all information and changes' (2/INT3); 'The IT facilitated the processes'; 'This is an open process thanks to the software available, where a closed loop exists between manufacturer, client and supplier'; 'Technology is a big enabler in this case and has contributed to large increase in efficiencies' (4/INT5); and 'The production efficiency is higher, around 3 times or less, due to automation ...'; '.. [automation] promotes better/higher reliability from a quality perspective' (6/INT8).

Logic diagram of managing the manufacturing strategy tentative category

Table 4.8 outlines the main characteristics of this tentative category in a logic diagram (Strauss & Corbin, 1990).

Managing the Manufacturing	Positive extreme dimensional Negative extreme dimensio		
Strategy category properties	variation	variation	
Advanced manufacturing processes with technology deployment in automation	High value-added manufacturing based on automation with an effective and efficient manual based processes	Manual approach to manufacturing with lack of value-added and effectiveness and efficiency in the human element deployment	
Information systems	Technology based approach	Manual based approach	
Manufacturing Process Re- engineering	Value-stream maps to innovate and improve manufacturing operations	No improvement in manufacturing processes	
Manufacturing span of operations	Vertically integrated to take ownership of all operations	Production ownership only (e.g. assembly plant)	
Outsourcing operations	Employing a partnership approach to outsourced core activities to integrate effectively	Arms' length integration with all suppliers/contractors independent on the outsourced activity	
Lean management	Efficient operations with low waste, less cost in materials and operations and competitive production turnaround time	Lack of efficiency in all/some operations	
Sustainability measures	Effective social, environmental and economical sustainability	Lack of effective social, environmental and economical sustainability	
Quality standards	In line with all quality standards with the deployment of the necessary validation procedures	Lack of conformance to quality or cutting corners in certain quality measures with lack of validated procedures	
Flexibility in operations	Flexible approach	Lack of flexibility	
Conditions	Technology deployment in ICT and automation; manufacturing sector; outsourcing or in-house activities; intellectual property protection; and country of products' destination.		

Table 4.8: Managing the manufacturing strategy tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled by the right manufacturing strategy tentative category, since it establishes an environment where the focal firm deploys advanced manufacturing techniques, based on the relevant sector technical engineering manufacturing standards (e.g. ISA) and product quality standards, with a lean management approach in all manufacturing and SC activities, to promote high value-added manufacturing and with a continuous improvement approach across the SC. The role of the human element remains key within the manufacturing strategy, since it meets the necessary interfacing needs in all manufacturing process, to promote flexible, innovative, streamlined and value-added operations, through optimisation of all activities. The manufacturing strategy needs to decide on the manufacturing span of activities, through the 'make' or 'buy' decision, based on a complex set of

variables, to deploy either vertical integration or outsourcing of a particular set of activities, to promote effective and efficient operations. The overall manufacturing and SC activities need to employ sustainable operations, to strengthen the business image to all customers. The manufacturing and SC operations need to be monitored in real time, with the support of information systems, based on technological applications, so as to manage effectively all operations with the necessary auditing and traceability of processes and products. With such an approach, any shortcomings from the scheduled performance can be established, so that the right decision is taken to remedy such problems at the earliest, not to disrupt the established workflow with an effective risk management approach, to promote business continuity. Effective supply management need to be used to promote efficient manufacturing and SC operations by: saving on purchasing costs; buying new and cheaper raw materials; and employing a consignment stock set-up with suppliers, to promote stock availability at no additional costs with better flexibility to meet demands.

Tentative relationship between all emerged potential categories in line with the coding paradigm

Managing the manufacturing strategy conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as efficiency in manufacturing and SC operations, competitiveness, flexibility in operations, manufacturing innovative approach, high-value added manufacturing, business continuity, continuous improvement and sustainability. The moderating variables are quality and sustainable processes, advanced manufacturing technology, ICT applications, lean management, human element manual intervention, risk management, supply management, SCI, effective decision-making, change management/innovation and auditing. The relationship of managing the manufacturing strategy tentative category with all other categories is represented by the conceptual framework in Figure 4.6.



Figure 4.6: Managing the manufacturing strategy tentative category (Author)

4.3.2.7 Managing the business strategy tentative category

The business strategy theme determines the manufacturing strategy and the SC strategy. The business strategy for the local SME firm, under a multinational set-up, is determined by the Corporate's strategy. The Corporate strategy and the business strategy also determine the marketing strategy so as to establish whether the products to be manufactured shall be of a premium or generic design to focus either on a niche market or a broad market.

The local manufacturing firm business strategy for a SME, based on a multinational company set-up, is normally targeted to manufacture products which are in line with the mother company products portfolio, which may be either of a low or high value added nature or a mix of both. There is the need for an ongoing investment in technology, in both manufacturing and information management, so as to

promote more efficient and effective operations through improved automation and information based processes. Such a business strategy based on the use of technology has been described by five participants, by stating that:

'We are high value added firm' (2/INT3); 'All starts from the top management direction...'; 'We in fact invested in the ERP system,... using the EU funds' (5/INT6); 'The product is low cost such as sets of tubes and chamber up to high value products ...' (6/INT7); and 'Automation investment increased'; '... we use high value manufacturing' (6/INT8).

The marketing people, in advanced technological products SMEs, need to establish, through a customer feedback mechanism, whether the value-added features of customised products, through strategic diversification, are adding value to the customers' applications. Such a stance is needed to ensure that the premium costs attributed to such products, justifies the customised product investment, so as to compete effectively. The designed features of the manufactured product need to be fully exploited by the customers, otherwise the focal firm needs to shift away from the niche market, by deploying the right marketing strategic approach has been described by two participants, by stating that:

'... we may tailor our products to the customer need'; 'The sales people and the clients all over the world are well informed on the customers' behaviour'; 'Our product is sold to big customers and large enterprises, referred as niche marketing'; 'The technology is quite advanced and more expensive than a normal customer' (2/INT3); and 'the customers were using only default settings, without exploiting the whole set of features' (2TS/INT3).

The sales and marketing strategy for SMEs is normally under the responsibility of the corporate or an assigned contractor/partner/distributor, due to the local SME lack of resources, but the focal firm still needs to employ a high level of commitment to sustain such an integrated organisational set-up. The corporate may even purchase the products from competing manufacturing plants, in the same line of business, to instigate a high level of competitive behaviour on the focal firm performance. Such a sales and marketing strategy approach has been described by two participants, by stating that:

'Sales and marketing group abroad have the key link and in contact with the customerthey coordinate with us for the orders'; '[The focal firm has] a customer support officer ... to understand the impact of the quality, service and cost on the markets. These are [mother company] people to pool all the information from the market. This is another depth of SCI' (6/INT8); and 'Marketing company can purchase from us and even from our competitors. This shows that competitive actions taken by our [mother company]' (16TS/INT13).

A marketing strategy, to penetrate new markets, is to be based on an on-going relationship with new customers, by adopting flexible contractual agreements to adapt to the customers' requirements and to build a track record over time to retain existing customers and also win new customers. The focal firm winning strategy is not only based on signing contractual agreements with customers or other

downstream side SC members, but needs to use a continuous proactive approach with direct contact with such customers, to convince all type of customers that it is providing an effective customer service and an excellent product vis a vis its competitors. Such a business strategic winning behaviour has been described by five participants, by stating that:

'... we do contractual agreements. These are incentives to produce effectively' (2/INT3); and 'We try to convince customers to purchase from us. We design and produce to show the clients that we are capable to do it. You need to proof yourself'; 'If we wait for the contract to be signed, this leads to failure of the overall business. My saying was that: "we must go with a better valve". Continuation of investment and overcome obstacles without losing heart' (8TS/INT4); '... we are giving a service under a manufacture' (10TS/INT7); 'The firm key qualities are flexibility in the approach to negotiations terms and conditions' (11TS/INT8) and '... manufacturing of today is a process approach based on both the manufacturing and services' (15TS/INT 12)'

Most of the local manufacturing firms' sales are mainly dedicated for export since the local market is relatively small. Such products are either: delivered direct to the customers; or use Corporate's subsidiaries or distributors to deliver products to the customers; or use overseas warehouses as repositories of finished products, to deploy a consignment stock approach, so as to improve on the overall customer service. Though, the possibility to hold finished products in stock is not always possible, since some products, due to their intellectual property rights, cannot be sold to the public. The focal firm may also need to innovate its current products by re-designing new products, either in case the stock for the previous design is no longer available or else to launch innovative and competitive products. Such a competitive and innovative stance has been described by four participants, by stating that:

'Malta has the local staff work closely with US staff in collaboration for all projects, innovation and new products ...' (1/INT1); '[Mother company in USA] considers the canister price per unit that identified that we are better than China. Projects change to meet other customers' requirements so as to innovate' (1/INT2); 'We try to phase out such component, by redesigning new products or part of the product'; 'The design may be changed due to the bottlenecks of lack availability in the market of parts' (2/INT 3); and 'Most of the products are for export, locally the market is small' (3/INT4).

The business strategy determines the manufacturing strategy and SC strategy by establishing a set of policies and Key Performance Indicators (KPIs). The business strategy is established in-conjunction with the corporate, for multinational firms, to promote consistency in the Corporate's strategy among its subsidiaries. Such a business strategy within the focal firm is applied across all management levels down to the bottom line and also across the SC, with all the SC members, to build up the overall manufacturing and SC strategy. The local firm formal organisation structure is key to the SC performance with the deployment of the right performance measurements, in line with all KPIs, across

all management levels, so as to establish the tactical management approach needed to adhere to all policies and procedures in line with the business strategy. Such KPIs, at the same time, are used to establish the performance gaps to promote the necessary process improvements in all the operational management level, both within and outside the focal firm. Such a business strategic approach has been described by four participants, by stating that:

'For inter-company the performance is more easy to be dealt with ... But with local suppliers and others external European suppliers to firm, requires the validation and audit parameters/criteria as established with the corporate KPIs and its suppliers' (1/INT 1); '[Downstream side intermediary] performance is also measured and service issues are being measured and gaps identified to ensure improvement' (1/INT 2); 'The delivery performance is all over the staff' (3/INT 4); and 'Also the performance measurement needs to be part of the firm strategy of the future' (5/INT 6).

The corporate strategy of the multinational company must not create obstacles in the focal firm business strategy but must integrate itself with the local plant operations to promote a one unified strategy and sustain the necessary empowerment initiatives through effective communication, so that the focal firm can effectively manage all operations. The focal firm may vary from a fully fledged manufacturing plant, which manufactures the whole product, or else acts as a contractor, which manufactures only part of the product and forms a relatively small part of the SC due to the limited resources (e.g. the focal firm may be an OEM or Tier 1 up to Tier 'n' of the SC). The parent company corporate's strategy may have other priorities which may not be in line with those of the local firm strategic objectives. As a result, such a strategic actions, until the Corporate awards the necessary approvals. Though, it is a must for the focal firm to deploy all its internal and external operations in an integrated approach to promote SCI with an effective and efficient performance. Such a challenging unification of the Corporate and business strategic approaches have been described by two participants, by stating that:

'All starts from the top management direction ...'; 'They decide the direction and create new policies. I start my work from the policy onwards' (5/INT6); and '[The firm is a] bit fragmented and not integrated'; '[The] actual management is outside the plant'; 'Not everything is in our control'; 'If you were not integrated as the way we are [locally], it will limit the customer service'; '[Overseas SC members are involved by] weekly meeting through video conferencing call with the [overseas] teams abroad to ensure the performance is smooth' (6/INT 7).

Logic diagram of managing the business strategy tentative category

Table 4.9 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing the business strategy	Positive extreme dimensional	Negative extreme dimensional	
category properties	variation	variation	
Strategic diversification	Value-added and customised	Low value-added and basic product	
	product at a premium cost	at low cost	
Contractual agreements based on	Flexible contractual agreements to	Rigid contractual agreements and	
both a manufacturing and a service	meet changing customers' cannot meet customers' ch		
approach	requirements with a proactive and requirements		
	innovative approach		
Innovative products	Manufacturing new and innovative	Manufacturing the same products	
	products		
Corporate strategy	Corporate strategy is matched and	Corporate strategy is neither	
	integrated with the focal firm	matched nor integrated to focal	
	business strategy (i.e. unified	firm business strategy (i.e.	
	strategic approach)	different assigned priorities)	
Business strategy is based on	Employ performance	Performance measurements are	
performance measurements across	measurements at all management	lacking at all management levels	
all management levels	levels in line with the Corporate		
	KPIs		
Conditions	Technology investment in automation and information management;		
	economic recession; SME limited resources; and Small Island State		
	characteristics.		

Table 4.9: Managing the business strategy tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled by the business strategy tentative category, since it establishes an environment where the focal firm together with the Corporate establish the business strategic direction. The business strategy must be in place to promote a one unified strategy with the Corporate and sustain the necessary empowerment initiatives through effective communication, so that the focal firm can effectively and efficiently manage all of its operations. The business strategy is the source of the SC strategy, manufacturing strategy and marketing strategy. The business strategy for a SME, based on a multinational company set-up, is normally targeted to manufacture products which are in line with the mother company product portfolio, which may be either of a low or high value added nature. To achieve an effective marketing strategy deployment, considering the limited resources available by the local firm, the Corporate takes the responsibility directly or assign it to an intermediary, to take care of such a marketing strategy implementation, in-conjunction with the full collaborative commitment of the local firm. The business strategy through its marketing strategy establishes the product portfolio to be used to compete in the market so as to establish whether to manufacture customised or broad generic products. The local firm, with the support of other SC members in the marketing strategy, is not only after the right contractual agreements to promote further sales, but is mainly committed to win customers' requirements by providing proactive actions with innovative and competitive products.

The Maltese local market, being a Small Island State, needs to import and export most of the raw materials and finished products respectively. The focal firm, through its manufacturing strategy and SC strategy, needs to transform itself across time and adopt a dynamically changing approach, to manufacture competitive products across the globe. Such a transformation needs that the focal firm undergoes the necessary technological investment in automation and ICT applications, so as to improve in all processes and at the same time meet effectively any economic recession, which from time to time hit the market. The business strategy determines the manufacturing and SC strategies by establishing a set of policies and Key Performance Indicators (KPIs) to promote SCI both within the focal firm and outside with all of its stakeholders. Such a business strategy, with the established KPIs, needs to integrate the focal firm, across all of its management levels down to the bottom line and also across the SC, to build up the overall manufacturing and SC strategy with a set of best practices. Such KPIs, at the same time, are to be used to establish the performance gaps to promote the necessary process improvements, both within and outside the focal firm.

Tentative relationship between all emerged potential categories in line with the coding paradigm

Managing the business strategy conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as innovative products, competitive products, customised products, generic products, effective customer service and effective and efficient performance. The moderating variables are marketing strategy, SC strategy, manufacturing strategy, customer relationship management, contractual agreements management, technology deployment in production and information management and performance measurement. The relationship of managing the business strategy tentative category with all other categories is represented by the conceptual framework in Figure 4.7.



Figure 4.7: Managing the business strategy tentative category (Author) 4.4 Overview of the chapter

Within this chapter all the twenty two (22) emerged tentative categories are represented in a storyline approach through a set of memos on a category by category basis, to define and explain the key concepts, with all their relationships to other key concepts or sub-concepts. All the tentative categories are supported by the selection of the key in-vivo quotes from the data, due to the voluminous amount of data and visually mapped by logic diagrams and conceptual frameworks to assist the theorising process. The aim of the data analysis is that key concepts are going to be referred as categories and the sub-concepts as sub-categories respectively, as part of the theoretical abstraction process within the GTM. The seven (7) key categories, which were established after the completion of the Selective Coding stage in the next chapter, were all analysed in this chapter, whilst the remaining fifteen (15) non-key categories were also analysed with the same approach, but were shifted into Appendix 11 to make this chapter more manageable.

The conceptual labels are used to represent the researcher's interpretation on the data, in line with the Coding Paradigm (Strauss and Corbin, 1998), which were initially tentative open codes, in the Open Coding phase, within Section 4.2, which were fragmented but represented what the data were saying. It is then followed by the Axial Coding analytic process, within this Section 4.3, through the storyline

approach, which is tracked through memos, all along the research process, to theorise on the emerged theory in a narrative and iterative process. In fact the memos within the storyline approach, for each tentative category, become more accurate and complex as the analysis progresses throughout the eight interviews, by going back and forth to each category to refine and summarise each memo.

The outcome of this Axial Coding process can be seen in the Code Matrix shown in Table 4.10. The matrix shows the audit trail of how the forty-six (46) open codes, which are theoretically abstracted to twenty-two (22) tentative categories from the Open Coding phase to the Axial Coding phase. The CAQDAS and the storyline approach are used as triangulation tools to promote research rigour in the theorising process. The tentative categories are central themes that connect all open codes, which are still grounded in data. The next chapter will show the final step of the analysis, by using the Selective Coding as the final analytic approach, so as to finalise all tentative categories around a core category with a set of five (5) main categories with their sixteen (16) sub-categories (Table 5.1).

Phase 1: 46 Open codes (Note: 30 Actions & 7 Conditions Annendix 10 refers)	Phase 2: 22 Tentative categories		
(Note: 59 Actions & 7 Conditions Appendix 10 refers)	(7 Kov Tontative Categories)		
Leadership style: management effectiveness: and	Leading Effectively		
priority of works			
	Managing business strategy		
Manufacturing process; sub-contractors &	Managing manufacturing strategy		
outsourcing	Managing process within SC and manufacturing		
Ownership of operations	Managing SC Strategy		
ICT applications	Managing Technology deployment		
SC members support; SC strategic action and	Managing SCI		
growth; SCI activities; SCI depth; and time			
management			
Quality of processes & EHS Standards	Lean Management, Sustainability and Quality		
(Sustainability); Lean management; and standards	Standards Compliance		
implementations			
Information sharing	Managing Information sharing/communication		
Risk management	Managing Risks		
Collaboration; teamwork; employees' engagement;	Managing Collaboration		
cross-functional operations; communication; and			
coordination			
Training & share innovative ideas	Managing Knowledge		
Performance measurements	Managing Performance measurements		
Delivery actions; logistics; and transportation	Managing Logistics		
management			
Plans and forecasts	Planning & Forecasting		
Inventory management; supply management; and	Supply management		
lead time			
Change management; flexibility; and innovation	Managing Change & Innovation		
Customer relationship management	Managing Customer Service		
Trust	Managing Trust		
	Managing Culture		
Auditing & traceability of operations	Auditing Operations		
	Managing Cash-flow		

Table 4.10:	Open Cod	ing to Axial	Coding (Code Matrix	(Author)
--------------------	-----------------	--------------	----------	-------------	----------

Chapter 5: Data Collection and Analysis based on the Selective Coding Technique

5.1 Introduction to Selective Coding stage

The third phase of the data analysis, referred as phase 3 (Table 4.1), is based on the semi-structured interview guide under the theoretical sampling phase, as referred in Appendix 6. This phase consists from fourteen (14) semi-structured in depth interviews based on eleven (11) new firms, since three (3) of the first sample firms accepted a 2^{nd} visit (Appendix 5). The data collection and analysis approach focuses on these twenty-two (22) emerged tentative categories but remain open to any additional categories or properties from the emerged theory categories that may emerge at any moment in time.

5.2 Findings from the theoretical sampling data collection under the Selective Coding analytic approach

The theorising process in this phase refines and build on the previous Axial Coding phase with its twenty (22) emerged categories. Such a phase assists in the build up of all emerged tentative categories, with additional new properties with their variation, wherever applicable, to achieve theoretical saturation. Such updates are included in the previous chapter, as a matter of practicality, to minimise any duplication of analysis' content, but the key emerged categories' analysis, are all included in a narrative form and substantiated with key *in vivo* quotes within this Chapter. The Selective Coding stage selects six (6) key categories from the 22 tentative categories, through theoretical abstraction and the remaining 16 tentative categories are selected as sub-categories of one (1) of the six (6) key categories, as explained in the next Section 5.3 and in the code matrix Table 5.1. Such an analytic approach is an iterative process and is derived, through sense-making and theoretical abstraction, which establishes the core category with its five (5) main categories in a meaningful way, based on the constructivist paradigm approach, to generate the grounded theory.

5.3 The Coding Paradigm to represent the emerged theory

In this stage the categories are integrated and refined, with the selection of the core category, together with other five main categories and sixteen sub-categories, to represent the substantive theory. The theorising process is based on the Coding Paradigm model (Strauss and Corbin, 1998), as the tool which also guided all the previous two coding stages. Henceforth, such a model exposes the meaning contained within the emerged theory in terms of the **actions, context** and **consequences,** which also serve as a robust platform for the final theoretical framework to promote research rigour. The overall

analytic process with all the coding stages, including this Selective Coding stage, is referred in Figure 5.1.



Figure 5.1: Outline of the overall GTM coding data analytic approach (Author)

5.3.1 Actions Categories of the theory: The core category with the five main categories

The core category that emerges from this study is labelled as an **'integrative management and leadership approach'**, namely abbreviated IMLA. SCI to be implemented effectively needs the integrated management approach of all processes within the focal firm and across the SC, in a unified and holistic process approach, but also needs the right integrated leadership approach to drive all SC actors in concert, to empower all staff and to facilitate all management initiatives, so as to have all people, within and outside the focal firm, ready to collaborate and to perform effectively and efficiently, in line with the focal firm KPIs. Though, there exist a variation of the leadership approach continuum, where at one end there is the strategic and the supervisory leadership approaches, based on a shared leadership style, where the leadership stance incorporates all management levels and all key SC actors to perform effectively through a teamwork approach. While at the other end there is the single centric or individual leadership approach, where the leadership style is mainly focused on the top management team based on a centralised management approach. Such an integrative management and leadership approach, with its continuum, has been referred by eight participants, by stating that:

'... to empower their responsibilities across all departments' (1/INT2); 'Leadership and the team approach is promoting the SCI'; 'Synergy and coordination is the order of the day; 'Teamwork and leadership is very effective at all levels such as strategic, tactical and operational' (1/INT4); '... performance measurement needs to be part of the firm strategy of the future' (1/INT6); '... All starts from the top management direction, from the Directors ... **They decide the direction and create new policies**. I start my work from the policy onwards ...' (5/INT6); '... employing an open door policy, within the focal firm and a controlled open book, with all external entities'; 'To make all ends meets, one needs to remain in [his/her] tip toes, to ensure that all processes are integrated' (5TS/INT2); 'The holistic approach is governed by the focal firm KPIs' (2TS/INT3); 'SCI needs total control of the process ... ' (15TS/INT12); and '... [SCI] needs a commitment from both ends' (16TS/INT13).

It is crucial to have the right fit between the **business, manufacturing** and **SC strategies** respectively, in line with the quality standards and all sustainable measures, for each manufacturing sector, with the deployment of the right **technology in automation and information management**. But there is also the need for an **integrated management** and **leadership approach**, to act as a catalyst and a driving force to make all ends meet and to be flexible enough to decide and adapt to all situations, so that the overall **holistic SCI management process** meets the right set of SC competitive capabilities. The human element, within the IMLA, remains key in the collaborative initiatives with all SC members up to the end customer and at the same time it is also needed within manufacturing processes, in some way or another, even where a high level of automation exist. Such a unified strategic management stance with the deployment of a holistic SCI management approach, enabled by technology deployment, with a certain level of human element involvement, has been referred by eleven participants, by stating that:

Internally needs the right structure ... with a holistic responsibility with all the roles and have separate entities integrated' (i.e. holistic SCI management process); and 'This needs the right manufacturing platform with an integrated SC' (i.e. Manufacturing strategy) (1/INT1); 'Workers are involved in all decisions'; 'We are flexible to meet changes'; ' [adopt an] open style management where managers are part of the workforce' (i.e. IMLA) (1/INT2); '[the manager] supports, gives the stimulus and advice at the right time and quality of message to us in all operations, but empower us to do our job' (1/INT4); 'There exist processes and products that still need humans' intervention. Machines do not cater for all activities' (i.e. human element); (4/INT5); 'Quality in all processes is key together with sustainable measures' (TS/INT1); 'Leadership style is like the glue to match all management levels together' (5TS/INT2); 'All staff need as a team to have the right mentality, right people and be rowing in the same direction and strategy' (i.e. **Business strategy**); '... the workforce [need] to be adaptable to implement the changes effectively' (1TS/INT5); 'Holistic approach is based on information sharing and technology' (i.e. **Technology deployment**) (9TS/INT6); *([the success is attributed to] the change management culture that accepts to be flexible and* adapt to the customers' needs' (11TS/INT8); '... people is the most important asset in the SC' (i.e. human element); 'The key fundamentals in SC is to take the right decisions to avoid mistakes otherwise you miss targets' (i.e. SC strategy) (13TS/INT10); 'The on-line is effective and efficient, but the human element remains key to establish a long term relationship' (i.e. human element) (17TS/INT 14).

5.3.1.1 The sixteen sub-categories within the holistic SCI management approach

- I. Audit with a systematic evaluation of all SC operations, both within and outside the focal firm, to meet all quality standards together with all sustainable and legal obligations, in an integrated approach, as referred by two participants, by stating that: '... *suppliers, who have to stick to our standards and processes and we visit them and make audits...'* (6/INT7); and 'Clients we do credit checks on them, especially if they are new' (10TS/INT7).
- II. **Manage cash flow** through an effective management of the incoming and outgoing of cash associated with the managing of supplies and sales of products based on an integrated approach with the respective SC members, as referred by a participant, by stating that: '... we use consignment stock by getting the material and we do not pay until we use it' (6/INT7);
- III. Undertake collaborative initiatives with all SC entities to promote relationship building between people and to perform effectively, as referred by a participant, by stating that: '[Departments] are assessed on ... collaborative effort ...'; 'Teams are formed to support and report' (1/INT 1);
- IV. Manage change effectively by mastering the complex and dynamic changes from both upstream and downstream SC sides, as referred by two participants, by stating that: 'We are flexible to meet changes, to meet new customers' specs' (1/INT 2); and 'we use consignment stock, ... we have more supplies and assist to be more flexible to change plans, having more material in hand' (6/INT 8);
- V. **Managing information sharing** by capturing and sharing the right information at the right time, within and across the SC, so as to take all the necessary timely collaborative actions, so as to promote an integrated approach to all operations, as referred by two participants, by stating that: *'Data of progress is visible to all through monitors in the shop floor'; 'From suppliers' side, we share stock levels...'* (2/INT 3); and '*The sales offices will be updated on their orders and the stock and state of production of the firm'* (4/INT5);
- VI. Manage effectively the customer service, as referred by two participants, by stating that:
 Direct contact with the customer... is critical to achieve, to win the customer feedback and [*its*] management' (1/INT 1); and '[provide customers with] the right product, at the right time and at the right price' and 'to ensure that the product has the right quality' (6/INT 8);
- VII. **Managing knowledge** effectively through information sharing mechanisms, within and across the SC, to have an informed mindset to promote effective decision-making to deploy state of the art processes through business process re-engineering, both within manufacturing and SC operations, as referred by two participants, by stating that: '... material reduction and validation of new resin material used to reduce cost and better quality to improve efficiency, hence the knowledge sharing occurs on these matters'; '... share the project works and

innovative ideas ..' (1/INT 1); and '... managers are expected to improve processes, people and systems' (6/INT 8);

- VIII. **Manage logistics** through the effective transportation of all merchandise, from both the upstream and downstream SC sides, in the right time, with the right cost and in the right place, as referred by a participant, by stating that: *'The logistics is driving the whole set-up'*; *'SCI within logistics is needed since I always tried to build a good relationship with suppliers and sales (overseas)'* (3/INT4);
 - *IX.* **Managing performance measurements** effectively by measuring people's behaviour, attitude, skills and commitment to work, to determine their contribution to the overall SC performance, as referred by two participants, by stating that: *'We have a SC measurement for every line'* (6/INT7); and *'The manufacturing process needs to report back on a daily basis and on a shop floor level declares on an hourly basis the quality and quantity in-line to the standards'* (6/INT8);
 - X. Undertake planning and forecasting to establish the necessary plans of all orders to be manufactured, based on customers' forecasted requirements or orders, to procure all the necessary raw materials, with the collaborative approach of all suppliers, to meet all scheduled deliveries in a timely manner, as referred by a participant, by stating that: '... agreements with the clients for a consignment stock and also with the suppliers, to balance out cash flow' (5/INT6);
 - XI. Managing risks in quality, sustainability and legal requirements, within the manufacturing and SC operations, so as to establish any shortcomings to remedy at the earliest, as referred by two participants, by stating that: '*Risk management is important since all operations are being monitored to deliver'; 'All obstacles are being monitored to ensure such smooth delivery'* (3/INT4); and 'We revisit the procedures and systems and measure the level of risks, ... one need to be practical and do not overdo it, to avoid analysis by paralysis' (1TS/INT 5);
- XII. Undertake effective supply management through all the necessary collaborative measures with all suppliers, to procure the necessary stock in time, with the right quality and with a competitive price, by managing cash flow effectively, as referred by three participants, by stating that: *'The latter suppliers are available in more than one, to manage risk of failure in the supply'* (1/INT1); *'... focus more on strategic issues, say to find new alternative sources of parts due to single source i.e. all the eggs in one basket [situation needs to be avoided]'* (2/INT3); and *'... every product has a number of components to complete and hence any missing one, will be a bottleneck'* (3/INT4);
- XIII. **Managing trust** is needed through effective collaborative relationships between people and organisations across the SC, to promote a level of teamwork (i.e. integrated activities) with all stakeholders, as referred by three participants, by stating that: *'When a client engages with a*

manufacturer, both engage in a trustful environment which promotes open communication apart from the formal contracts' (4/INT 5); 'We have suppliers of the compressors, as the core activity ... these suppliers are partners and have an in-depth relationship, based on years of experience' (5/INT6); and 'We have a good relationship with the customers and hence they trust us ...' (6/INT 7);

- XIV. Managing culture is needed so that the focal firm adopts a working attitude that promotes understanding and a flexible collaborative effort between all stakeholders, to achieve a more integrated approach with effective and efficient performance, as referred by three participants, by stating that: 'We use a culture to cater for the future and undergo cost savings' (1/INT 2). The EU countries are more understandable and flexible' (3/INT 4); and 'One needs to detrain and re-train, to ensure that new processes are being used, due to the previous culture. Not to use the old systems again on the new processes' (4/INT5);
- XV. Lean management is needed so that the focal firm together with all of its SC members integrate themselves to deploy a systematic approach in all processes to be efficient and effective in the deployment of all resources, as referred by three participants, by stating that: '.... with lean six sigma introduction ... this led to a better integration' (1/INT2); 'Even office work need to be lean to eliminate duplicated office activities' (3/INT4); and 'Lean and six-sigma are used in quality assessment to apply the principles in ... a systematic process'; 'Lean removed the waste in the processes, to be more responsive to the customer requirements and is faster' (6/INT8); and
- XVI. **Managing quality and sustainability** is needed to comply to all standard operating procedures and to align all process with a continuous improvement approach to promote quality assurance in all processes and at the same time comply to all sustainable measures (i.e. triple bottom line), as referred by four participants, by stating that: '... one has no option to stay out of the SCI initiatives, due to regulations and other commitments you need to integrate otherwise you are out of business'; 'Environment and CSR, we have to separate waste, and [respect] ISO 14000' (5/INT6); '... suppliers, who have to stick to our standards and processes' (6/INT7); 'Quality is an integral part of the model' (6/INT8); and '... quality concept is a continuous process approach across all functional areas to make the SC holistic' (5TS/INT2).

5.3.2 Outcomes Categories of the theory: Competitive capabilities

The **competitive capabilities** from the Axial Coding stage, in Section 4.3.2, include a conceptual framework for all the 22 emerged categories with their respective outcomes. The key eight (8) outcomes from all the tentative categories are: **SC flexibility; SC visibility; business continuity;**

continuous improvement; lean (optimised) operations; effective and efficient performance; sustainability and effective customer service.

5.3.3 Context categories of the theory: Contextual conditions

Malta is mainly undergoing business with EU member states, as a single market, with various advantages, based on a Free Trade agreement, but it also operates with the rest of European countries and other worldwide countries, such as USA, North and East Africa, Middle East and Asia.

SMEs in Malta, who form part of various recognised Multi-nationals, invest a lot in the right management and leadership approach, so that all staff performs effectively and efficiently, within all the established standards and winning behaviours by the Corporate. Most of the SMEs try to be fully fledged manufacturers to perform a wide portfolio of activities. Such a wide portfolio of activities has been outlined by a participant, by stating that:

'Today manufacturing is focused on high value added product based on innovation. Today the companies are more focused to have fully fledged companies or else they are performing a wider range of roles than 20 years ago. If not R & D, today the stages are more broad in skills' (12TS/INT9).

The firm context is based on a Small Island State (SIS) country, where the local Government is proactive in various supporting measures to promote Foreign Direct Investment (FDI) to contribute to the country GDP. The Government is a watchdog and a stakeholder of all the industry, especially in the manufacturing sector and related services for SMEs. The manufacturing sector is one of the Maltese four economies pillar stones, and it is also under scrutiny by both local and international bodies, through various local and world-wide indicators and ratings, such as the Maltese financial services authority (MFSA) and World Economic Forum (WEF), Standard and Poor's and Fitch Ratings. The Government is also creating an attractive package to all current investors in the manufacturing sectors, by reducing several running costs, such as reduction in licences, low lease rates for premises, the provision of investment guarantees for start-ups and tax exclusions for all exports. Furthermore, the Government is supporting the industry through incentives and funds/grants for all investment, to serve as a catalyst for innovation and to motivate the industry to remain competitive by investing in the state-of-the-art technology, such as ICT applications (e.g. ERPs) and automated equipment (e.g. heavy manufacturing machinery). Such a business context and orientation for SMEs that serve as attractive incentives for investment, due to their limited resources, has been described by five participants, by stating that:

'When there were funds to IT investment, we took the opportunity to invest in an ERP' (5/INT 6); 'we [are] after job creation or to launch new products' (8TS/INT4); 'Malta pays no 35% corporate tax'; 'We did a vertical integration and the cost is better-off as from Jan 2012 with support of Malta Enterprise... It was a ½ million Euro [for investment in new developments]' (1TS/INT5); 'Fiscal incentives exist to encourage export and FDI' (9TS/INT6); and 'SMEs suffer to invest in ERPs' (17TS/INT14).

The manufacturing SC is function of the cultural element, which is influenced by both the local firms' human element and also from all SC members, both locally and abroad. The Maltese companies' cultural values are very attractive to investors, due the current industry cut-throat competitive scenario, since they are creating a positive culture based on innovation and learning to attract future investments. The Maltese workers are highly competent, with relatively high intellectual and practical capabilities, and demonstrate a culture based on an outstanding potential to work hard and to be flexible with a multi-skilling capabilities and are committed to adopt a life-long learning approach. Companies' KPIs include culture as one of the antecedents. The Maltese culture is catalyst for change since the local people have been subjected to a very strong educational system where learning and training has always been a priority in people's core values. It is not being excluded that certain disciplines are lacking due to the emerging dynamic trends in both traditional and evolving manufacturing sectors (e.g. printing and technology based expertise respectively). Such an employee' culture, to be hard working and multi-skilled, has been described by four participants, by stating that:

'There is a trend that the Maltese people are leading to a workaholic life with a passion-based approach to work, by trying to keep in touch with all customers' needs on a 24/7 basis due to cut-throat competition' (5TS/INT2); 'Skills for such employees on printing is short of labour availability' (10TS/INT7); 'Maltese are very flexible to solve the problems but the German are more focused and accept the culture of specialisation. We in Malta are more widely skilled and flexible. We can do the same job with less people, having people with a wider expertise coverage' (16TS/INT13); and 'people in Malta are flexible, and today multi-skilling is a key skill of Maltese people...'; 'The problems may be diversified and the wide skills of workers, end up to solve the problem in a more efficient way due to such [wide] expertise' (17TS/INT14).

The Maltese island is a Small Island State and SMEs cannot exploit the economies of scales advantages, causing local SC members to run with a relatively less margin of profits. In fact, most of the sales associated with Multinationals SMEs are all export oriented. In Malta there are large also a number of SMEs, within the manufacturing sector, which are engaged with local firms as sub-contracting or else are focussing on a niche marketing strategy to build on their sales, both locally and abroad. The SMEs within the manufacturing sector is very wide, covering various sectors and has a substantial contribution to the Maltese economy. The niche products are designed to satisfy customised customers' requirements to compete with the large companies, by focussing on the value-added approach of the products' specifications. Such a small economies of scale business climate and

the business strategic focus on niche marketing to strengthen the SCI efforts, has been described by four participants, by stating that:

"... the local market, ... is relatively small' (2/INT 3); "The main sales are export"; ".. here we are only manufacture/production" (7TS/INT1); "In fact the focal firm, within the contextual conditions of its small economies of scale, being located in a small island state, undertakes niche marketing, as part of the core strategy of the firm, to promote success ... with a value-added approach" (5TS/INT2); and "SMEs in Malta is our motor of the economy" (17TS/INT14).

The external economy recessions and downturn in the global sales do not normally influence immediately the Maltese workforce and productivity throughput of the local manufacturing, due to its <u>resilience</u> within its economy, but the effect is instantly felt in achieving an effective supply management, due to the need of the global sourcing of stock and to solve all the associated problems of inaccuracies in the forecasts figures submitted by customers and other SC members. Recessions create an imbalance in the stock availability in the market, especially for certain products, which make the SC management more challenging to manage all stock. Products which are not commodities, such as medical products and innovative products for smart homes, are normally less affected by such recessions, due to the stability of demand trends in such sectors. Such a challenging supply management approach and the Maltese economic resilience, has been described by two participants, by stating that:

'Recession of last year, caused 4 day week not as abroad who incurred redundancy'; 'Forecast was made as usual and due to recession the forecasts being requested were not met by the key suppliers since they gave us very long lead times'; 'This is an innovative line of activity and the sales are constant independent of the market recession' (3/INT 4); and 'The 2008 recession are still affecting all VMIs procedures with all firms. The suppliers used to push stock but now after the recession there is lack of stock availability' (5/INT 6).

The Maltese workforce is no longer competitive with respect to <u>cheap labour</u> since the minimum wage has increased, though it cannot be excluded that most of the salaries of professional people are relatively lower than their overseas equivalent. Such a salary assessment relative to other countries, has been referred by two participants, by stating that:

'The issue of minimum wage when compared to the Asian countries is no longer competitive, since in Malta this minimum wage has increased' (3/INT 4); and 'The salaries are low compared to abroad' (17/TS/INT 14).

The Maltese economy has no natural resources and depends on the importation of all needed resources associated with the manufacturing sector. The low economies of scale is influencing various suppliers engaged with the local manufacturing sectors' orders, since the supplies requirements are relatively low in volume and will not be feasible for such suppliers. The **local infrastructure** with its advanced

technological applications and services are key to the SC context, although it cannot be excluded that improvements can be made in certain institutions such as in the standards authority and customs. The sea transportation services, is a key requirement to an efficient SC for sea locked islands, but such privatised maritime logistics remains the main problematic issue in Malta, with its infrequent sea shipping services and with the relative high fixed cost in all logistics operations, with respect to the high port charges and high merchandise transport cost. Such a costly logistics situation is attributed to the low economies of scale, when compared to other logistics services in world-wide countries, including EU member states, due to the relative low order volumes of all order levels from the local market. As a result, Malta is being subjected to relatively higher fixed costs to meet all logistics requirements, when compared to the larger revenue incurred by other global players. The transportation choice from Malta to any destination is restricted to either by air or sea. Once the EU or other nearby countries main land is reached, the land transport mode is used to minimise costs. The air freight is relatively expensive compared to sea freight, but is very fast and with a better logistics services. The local infrastructure, that forms part of the logistics backbone, has a competitive freight forwarders network with a large substantial number of players with low customer loyalty, but remains limited, due to various infrastructural limitations such as: a single airport; two cargo handling operators; two sea-shipping terminals; and two containers depots. On the other hand, Malta's strategic location, being in the middle of the Mediterranean sea, with its transhipment-hub, is giving a competitive logistics position to the Eastern Countries suppliers, being nearer to such countries. The transhipment-hub terminal is being exploited by carrying the local merchandise requirements over other large freight forwarders routes, to minimise all associated local logistics costs. Such a Small Island State with its small economies of scale disadvantages and the strategic location of the Island, has been described by seven participants, by stating that:

'... locally the market is small' (3/INT 4); '... today we have a private focused company instead of Sea Malta' (5/INT 6); 'The Freeport bottleneck is not on speed or logistics but in terms of cost... The port charges are very high'; 'This may be attributed to the economies of scales' (1TS/INT5); 'Airlines for logistics are more available. Such as Lufthansa, low cost airlines etc. Cost of the item is high, it will be sent by air [to be feasible]' (10TS/INT7); 'Malta is at the peripheral of EU but is the 1st call for Europe for sea shipments from Asia'; 'Furthermore we are an Island and insular' (12TS/INT9); 'The orders are relatively small due to the small size of the Maltese sector and hence our fixed cost is high relative to overseas'; 'we cannot buy direct from China due to the large volume requirement by the seller'; 'This also applies to ships ... We need [to] exploit on the service of others, to benefit from the others' large orders'; 'Today the Freeport made the difference in principle, since it is providing support for transhipment'; 'Monopolies within the Maltese context ... impinge on SCI. This leads to less competition and apart from the price the systems are not efficient. The airport is one; two terminals both private. Ground handling is Airmalta and Global air. The facility is the same one. You have no choice to select from'; 'Containers are in Hal-Far and Freeport depots. The containers are our tools'; 'Hence the infrastructure is limited'; 'The Freight Forwarders (FF) are numerous and hence client loyalty is not found' (15TS/INT12); and 'We are the fringe to north of Europe. The shipments are twice a week' (16TS/INT13).

The SC running costs are also function of the changing local **electricity tariffs** since such a cost is one of the key running costs of operations, especially for SMEs. The investment in training by the industry is being greatly facilitated by **Government supporting funds** under the EU platform. The country **political, legal** and **social stability** is promoting stability of the SC, with no interruptions and possible performance degradation due to no political conflicts and neither corruption in the Maltese social and business life. The country educational system is a key strategic benchmark within the EU members, since the Maltese **educational level** is very competitive in various disciplines, which are needed to **support the industry**. The country is enforcing that all the industry complies with all **social responsibilities**, such as in **waste management** across the SC, including the end life of the products. The SC needs a solid **legal framework** for the global SC, so that all members can settle all disputes effectively, to promote fairness to all. Such a sound political, social and legal stability, environmental responsibilities and a high educational investment focus, has been described by three participants, by stating that:

'The social aspect, locally does not affect the SC... instability and corruption [exist in other countries]'; 'Technology in Malta is very updated'; 'Environment and CSR, we have to separate waste'; '... our legal framework is very robust, since they support the business against fraud, as compared to other larger countries, due to loop holes in their legal framework'; 'Sometimes, during a crisis there [abroad], we cannot get the finished products and anything out of the country' (6/INT 7); 'Small subcontractors may be drastically affected by rise in electricity charges or by any other introduced charges by the government' (7TS/INT1); and 'Nitrogen is used in large quantities. Such a gas ... it is not used a lot by other firms'; 'Hence the price is very expensive' (16TS/INT13).

From the primary data, associated with the context analysis, it transpired that the nine (9) key emerging contextual themes needs to be taken in consideration by all SMEs through all strategic actions to remain in business. These themes, as referred in Table 5.1, may be both macro conditions, which are the influencing factors within the external business environment and micro conditions, which refer to the focal firm internal factors. The theorising process of such **nine (9) themes** from the data refers:

- I. Most of the local SMEs need to undergo trading mainly with EU countries due to the Free Trade zone for EU member states, but must not exclude any commerce across the global world, although trading needs to incur settlement of the associated global tariffs;
- II. The SMEs need to exploit the local Government and EU deep commitment to encourage investment and innovation within the manufacturing sector, by applying for all relevant incentives and attractive opportunities of investment for innovation, to enhance the firms' position relative to the current cut-throat competitive scenario (e.g. funds for ERP and advanced manufacturing equipment investments). It cannot be excluded that some local

services' charges are relatively expensive relative to other competing SMEs across Europe and the globe, due to Malta's small economies of scale (e.g. shipping and electricity services);

- III. The local SMEs need to exploit the **Maltese employees' educational and collaborative culture**, based on learning and multi-skilled capabilities, with a high availability of professional certified people in varies disciplines, with multi-lingual competences, both within the focal firm and other locally engaged entities, by building an excellent workforce with a teamwork mentality that promotes an innovative approach to work and to meet effectively all change management initiatives;
- IV. The SMEs are located within an Island with a small economies of scale, which is the cause of the relatively less margin of profits to all SC members, although sales are mainly focused on exports. To balance such a negative effect, the SMEs may need to target to niche markets, to exploit their limited resources, by manufacturing value-added and customised products with a premium price, where large sized firms, find it difficult to focus upon to increase the profit margin. Such low economies of scale is also affecting the logistics management, by being subjected to relatively more expensive charges for all transportation of materials or finished products, due to a single airport; two cargo handling operators; two sea-shipping terminals; and two containers depots. On the other hand, all SMEs need to exploit the advantages for all Eastern originated shipments, since they are faster to be received, relative to other countries, due to Malta's ports strategic location, being an Island in the middle of the Mediterranean sea;
- V. Local SMEs supplies are being drastically affected by **economic recessions** since it is heavily dependent on the importation of various goods from different sources across the globe. As a result special measures need to be taken to manage the stock levels effectively, against stock-out situations, since recession influence the world economy and Malta being a sea locked island, requires more usage of sea or air modes of transport, which are relatively more expensive than other competing countries, to get all the required material in time; and
- VI. The local and the multinational SMEs benefit from a stable political, legal and social stability which are key requirements for the stability and integrity of all SC operations, negotiations and settlement of disputes.

5.3.4 Results from the analysis of the secondary data sources

5.3.4.1 Participant Observation

From participant observation it was derived that: a strong relationship existed between management and staff from the way they treat and respect each other (i.e. human element aspect) and also from the way they treated their suppliers and customers and the way they discussed their pending issues and orders; there was a strict compliance climate among all staff, to firmly adhere to all quality standards and sustainable measures, by having an organised workplace with in-built quality checkpoints along the process, through both manual and automated means and with well established procedures for the collection of waste to promote an effective recycling mechanism; the use of bar-coding was being implemented in the warehouse and all across the plant to promote an audit trail of the stock being used; the warehouses were very organised with different shelving systems (e.g. space savings provisions) and different areas for incoming and outgoing material to promote an organised approach to warehousing and inventory management (e.g. dedicated consignment stock areas were also present under lock and key); manual and electronic dashboard reporting were being used through both manual white-boards and electronic boards, where all information was being retrieved from shared databases to promote awareness of all orders work-in-progress; and the use of kaizen board approach with names of employees well visible in the company premises with updated notes on various challenges to improve upon. Such Kaizen board notes were included by various employees to be visible to all staff and to update the state of improvement with board markers, to promote a continuous improvement culture among all people in all the manufacturing and SC processes and other key processes.

From the participant observation analysis, it transpired that the key emerging themes were: an effective management and leadership approach sensitive to the human element within the firm; a collaborative environment within and across the SC with special commitment to quality standards compliance and sustainability measures; auditing the overall process to ensure works traceability; deployment of technology in both automation and information systems to increase productivity through advanced manufacturing techniques and to maintain all the information regarding all works in progress respectively; and employ a continuous improvement culture based on performance measurements of all manufacturing and SC key processes.

5.3.4.2 Analysis of Documentary Evidence

From the hard documents given and on-line content on the firms' websites, it was clearly evident that: the focus on the quality standards used in all processes were very rigorous, in some cases, it was more than others, which were supported by on-line manuals and with visual aids hanging all along the factory walls to serve as a symbolic recognition of all sound values that need to serve as guidelines to promote awareness and to encourage compliance to all standards and sustainable procedures; customers, whenever possible, were given access to company database to submit or upload their orders details depending on the nature of the product (e.g. books are uploaded and requests are sent by mail); the company websites were used as a showcase to show the high level of service quality, lean

179
operations and certified standards awarded to each firm (e.g. excellence award/s; EHS standard award/s; various quality standard certifications); and all staff were assigned access to common information systems to ensure that all works were being monitored to deliver, to avoid customer surprises, by registering all work in regular intervals to establish at the earliest any shortcomings all along the process so as to remedy at the earliest to promote business continuity.

From the above document analysis, it transpired that the key emerging themes were: quality standards and sustainable measures compliance, to create a climate based on a continuous improvement approach; information sharing through technology based systems to keep all employees informed and manage knowledge effectively to promote effective decision-making to eliminate any risk of failure; and an effective customer service.

5.4 The final emerged theory

The emerged theory is represented by the six final action key categories, eight competitive capabilities and nine contextual conditions, which can be seen in the code matrix Table 5.1, where the overall audit trail of the three analytic stages are outlined. The matrix shows how the forty-six (46) open codes are theoretically abstracted into twenty-two (22) tentative categories and then to six (6) key categories, namely with one (1) core category and five (5) main categories with sixteen (16) subcategories.

Phase 1:	Phase 2:	Phase 3:
46 Open codes	22 Tentative categories: 7 key	6 Key categories (1 Core & 5 main
(Note: 39 Actions & 7	& 15 non-key tentative	categories) plus 16 sub-categories
Consequences/Context	catergories	
conditions, Appendix 10 refers)		
Leadership style; management	Leading Effectively	Core Category: Integrative Management
effectiveness; and priority of	(Phase 3 Core Category)	and Leadership Approach (IMLA)
works (3)		
	Managing business strategy	Main Category (1): Business Strategy
	(Phase 3 Main Category 1)	
Sub-contractors & outsourcing	Managing manufacturing	Main Category (2): Manufacturing
(1)	strategy	strategy
	(Phase 3 Main Category 2)	
Ownership of operations; SC	Managing SC Strategy	Main Category (3): SC strategy
strategic action and growth (2)	(Phase 3 Main Category 3)	
ICT applications (1)	Managing Technology	Main Category (4): Information and
	deployment	automated technology deployment
	(Phase 3 Main Category 4)	
Manufacturing process (1)	Managing process within SC	Main Category (5): Holistic SCI
	and manufacturing	management process approach with
	(Phase 3 Main Category 5)	16 sub-categories:
SC members support; SCI	Managing SCI	1. lean management;
activities; SCI depth; and time	(Phase 3 Main Category 5)	2. managing quality and
management (4)		sustainability:
		······································

Quality of processes; EHS	Lean Management,	3. managing performance
(sustainability); Lean	Sustainability and Quality	measurements;
management; and standards	Standards Compliance	4. auditing operations;
implementation (4)		5 supply management
Information sharing; and	Managing Information	6 managing cash flow:
communication (2)	sharing/communication	0. managing cash-now,
Risk management (1)	Managing Risks	/. managing change and
Collaboration; teamwork;	Managing Collaboration	innovation;
employees' engagement; cross-		8. managing collaboration;
functional operations; and		9. managing culture;
coordination (5)		10. managing knowledge;
		11. managing customer service:
Training & share innovative ideas	Managing Knowledge	12 managing information sharing:
(1)		12. managing logistics:
Performance measurements (1)	Managing Performance	13. managing logistics,
	measurements	14. planning/forecasting;
Logistics; delivery actions; and	Managing Logistics	15. managing risks; and
transportation management (3)		16. managing trust.
Plans and forecasts (1)	Planning & Forecasting	
Inventory management; supply	Supply management	
management; and lead time (3)		
Change management; flexibility;	Managing Change & Innovation	
and innovation (3)		
Customer relationship	Managing Customer Service	
management (1)		
Trust (1)	Managing Trust	
	Managing Culture	
Auditing & traceability of	Auditing Operations	
operations (1)		
	Managing Cash-flow]
		•

Eight Competitive Capabilities

SC flexibility; SC visibility; business continuity; continuous improvement; lean (optimised) operations; effective and efficient performance; sustainability (3BL); and effective customer service.

Nine Contextual Conditions

Macro: EU Free trade zone; Global commerce outside EU needs to pay special tariffs; EU/Government incentives for investment and innovation; Malta small economies of scales makes most of the services relatively expensive (e.g. port charges, transport, electricity); Malta's strategic location for sea transport for EU, North Africa and Asia; Current cut-throat competitive scenario; political, legal and social stability; and Global or EU Economic recessions disrupts the procurement of all imports. **Micro:** Human resources competitive capabilities and professional competences promote good governance of SMEs.

Table 5.1: Code Matrix for all categories within the research phases (Author)

5.5 Overview of the chapter

In this chapter all the twenty two (22) emerged key concepts derived from Chapter 4, within the Axial Coding phase, were again reviewed in this Selective Coding phase within this chapter, to conclusively derive the set of final key categories. The theorising process is based on the coding paradigm model (Strauss and Corbin, 1998). Henceforth, such a model exposes the meaning contained within the emerged theory, with its core category, referred as the IMLA with all other categories in terms of their

actions and consequences within a context. The contextual conditions of the Maltese SMEs manufacturing sector are described in terms of the micro and macro contextual conditions, to ensure that the theory is contextualised within the data, in line with the conditional/consequential matrix (Strauss and Corbin, 1998) to promote research rigour.

The conceptual labels are used to represent the researcher's interpretation on the data, which are initially forty-six (46) tentative open codes in the Open Coding phase, with all the relevant theorising process within the codes memos. Then, the open codes are theoretically abstracted to twenty-two (22) tentative categories in the Axial Coding phase and finally, these tentative categories within this Selective Coding phase, are theoretically abstracted in terms of a core category and five (5) main categories. One of the main categories captures the sixteen (16) sub-categories. Such an analytic process, in all these three data collection and analysis phases, are tracked through memos, all along the research process, that constitute the theorising process on the emerged theory in an iterative approach in line with the constant comparison process. Such an analytic approach cannot be attributed to a smooth analytic sequential process, since it needs a cyclical analytic stance from one interview to another, without any sequence whatsoever and also from all the three coding stages between each other, as referred in Figure 5.1. In fact the theoretical memos for each phase become more accurate and complex as the analysis progresses, throughout the twenty-two (22) interviews, by going back to each category to refine and summarise each memo. The outcome of the Open Coding, Axial Coding and Selective Coding processes in terms of the categories can be seen in the Code Matrix Table 5.1.

The next chapter generates a concluding summary memo, which briefly describes the emerged SCI theory in a short narrative form with a special focus on the core category together with reference to all the five main categories and sixteen sub-categories with their dimensional variations. Furthermore, the substantive theory is also represented with the SCI theoretical framework with all actions and consequences which are all situated within the contextual conditions, as guided in Table 5.1.

Chapter 6: Theory Generation

6.1 Introduction to the substantive theory generation

The previous analytic approaches, referred in Chapters 4 and 5, within both the open sampling and the theoretical sampling phases respectively, have explored the primary data in an iterative process for the tentative categories with their properties and dimensions to include the variations in the emerged theory on the SCI phenomenon. The role of the secondary data also supported such an analysis.

The focus of this chapter is to generate a concluding summary memo, which briefly describes the emerged SCI theory in a short narrative form, which is finally represented by the emerged SCI theoretical framework, as referred in Figure 6.1. Such a theoretical framework represents the generated substantive grounded theory, which explains the SCI theory within the manufacturing SMEs in Malta.

6.2 Summary of the emerged grounded theory

The **definition** of the generated substantive theory with its component parts is: To achieve supply chain integration an **integrative management and leadership approach (IMLA)** needs to be deployed at the strategic, tactical and operational management levels, with due consideration of each SC actor contextual conditions, so as to establish and implement the right strategies for the business, manufacturing and SC, within the focal firm and in-conjunction with all SC actors, based on a holistic SCI management approach, which is enabled by technology, so as to achieve competitive capabilities.

To achieve the holistic SCI management approach, the firm needs to integrate the business strategy and the manufacturing strategy, with the SC strategy. The SC strategy defines the management approach that shall be taken by the focal firm employees at all management levels (from top to bottom) through all the processes, with all the external network of SC stakeholders. The SC and manufacturing processes are enabled through both information technology and automated technology based applications. A holistic SCI management approach can be achieved by employing an IMLA to ensure that all stakeholders' different strategic priorities are managed and led effectively and efficiently. This IMLA is the driving force that enables the achievement of a trade-off with all the SC members, from end customers to the suppliers, within their different business sectors' scenarios and with their different contextual conditions. Such a trade-off is needed to select the most effective decision that leads to the overall SC performance optimisation for a win-win situation to all, with the direct involvement of all relevant stakeholders. The decision making process is not after any particular SC member but aims to establish the common good to all involved so as to effectively meet the expected customer service (e.g. material quality issue from suppliers; delivery target dates vis a vis customers or suppliers issues; capacity issue brought about by production overload; and faulty machinery hindering production issues).

6.3 The research theory based on the core category around all of its categories

6.3.1. The Core Category: The Integrative Management and Leadership approach (IMLA)

The core category, referred as the **IMLA**, binds the SCI theory together. This category runs all along the **organisation** (i.e. the **focal firm employees**), since it starts from top management and cuts across all other management levels (i.e. strategic, tactical and operational) and includes all the SC stakeholders, such as suppliers, intermediaries and customers, who all perform in a holistic approach to achieve common goals. The lack of such a conceptual approach (i.e. IMLA) may lead to a situation of SCM failure, since in the absence of an IMLA, the lack of such an effective management approach may transmit a breaking domino effect on all the SC members' linkages, which are needed to hold the SC tiers to stick together, even if all the necessary strategic decisions with respect to the overall business, manufacturing, and SC are in place. This IMLA, when used effectively, creates the holistic commitment of the human resources, as the backbone of all the process activities, to achieve teamwork and employee engagement across all management levels and in all tasks, so as to synchronise all cross-functional efforts with a quality based approach in all activities both within the focal firm and across the SC and within all the manufacturing processes. An IMLA needs to involve all the SC tiers, from the upstream to downstream sides of the SC, which shall also include the end customers, so that the focal firm can achieve and excel in its mission, objectives and competitiveness. Though, it is to be noted that there exist a variation of the leadership approach, where at one end there is both the strategic and the supervisory leadership approaches, based on a shared leadership style, where the leadership stance incorporates all management levels and key SC actors and at the other end of the continuum, there is the single centric or individual leadership **approach**, where the leadership style is mainly focused on the top management team through a centralised approach.

The **IMLA**, within the focal firm, is further defined by the way that all management at all levels: cares and develops people through all interactions and training to promote a learning culture; builds on **trust** and **empowerment** of all people but does not focus on control; leads people to excel in their doings and not only manages them, but by exploiting every **individual capabilities** in all **decisions**, both individually and as a team; creates and transforms the teams and not only focuses on formal structures

and systems, to promote flexibility and to accept a controlled level of informality (i.e. informal organisation); and manage all changes and risks effectively. The IMLA key leadership styles as emerged from the data are four: **transformational leadership style**, to meet change and innovation through people; **servant leadership style**, to be sensitive for people's needs and also meet business objectives; **situational leadership style**, to meet contingent situations; and **transactional leadership style**, to meet the routine flow of transactions.

In a nutshell, the key leadership qualities deployed within the IMLA adapts to today's dynamic business climate, with an openness consideration to change and innovation and with full commitment to listen and engage people at all management levels to perform effectively and efficiently. As a result, the **IMLA** mainly deploys a **transformational leadership style**, but does not exclude the other three styles, so as to manage effectively the integrated SC both from an intra- and inter-organizational perspective. Such as stance consists from a change management approach, which will drive the focal firm's readiness for creativity and innovation and develops a **human relationship** approach based on sound ideals and values, to motivate all people at all management levels. Furthermore, such a transformational leadership style exploits the people's flexibility and preparedness to keep up with external business demands and pressure, with an **unconditional approach** to succeed, and through the organization's ability to instill in all employees, at all management levels, a sense of **process ownership** that promotes commitment and engagement with the right **decision-making** mindset to perform effectively and efficiently.

The **IMLA** is used to achieve SCI by dedicated initiatives and by creating a collaborative mindset that is based on a mix of both management and leadership concepts within and across the focal firm, with all the stakeholders, for a winning behavioural approach, which goes beyond the **contractual agreement terms and references**, so as to earn respect and build long term relationships. The IMLA approach needs to apply the necessary measures of leverage through all management and leadership techniques, so as to unify the practices of both **conceptual elements (i.e. management and leadership as the two elements)** and to draw the best from the two approaches to optimise all stakeholders' contribution to achieve best practices and a competitive SC for a winning strategy.

This **IMLA approach** is based on a **mutual understanding** and level of empowerment across all management levels, which not only promotes **effective management** in a cross-functional approach, but also develops a driving **leadership approach** that inspires people to employ innovative, flexible and transformational processes in a continuous improvement approach. This **continuous improvement approach** is achieved: by the commitment of all people across all management levels to enable better ways how to achieve best practices in the right approach through individual and team

value-added contributions, such as in the reduction of materials cost through the use of cheaper and better materials; by employing improved processes in production and across the SC, to become more lean, through more effective quality procedures and standards; by taking initiatives to maintain and improve on all sustainable measures (i.e. triple bottom line); by the deployment of all the workforce skills, through a continuous learning approach to achieve competitive levels of productivity; and to manage all risks effectively to promote business continuity.

IMLA core category properties are: open door policy; open management style; culture of trust and teamwork in all collaborative efforts at all management levels (i.e. strategic, tactical and operational); management need to lead by example; common objectives to work together within and outside focal firm; collaborative and joint-problem solving initiatives; adapt a common culture to make up for the different cultures of all SC members; close or an arm's length integration; holistic process approach; manage effectively all manufacturing and SC operations; adopt preventive measures; minimise any risk of failure (i.e. risk management); involve all stakeholders in all decisions and changes (i.e. apply an integrated management approach through change management and innovative initiatives in both products, manufacturing and SC processes); effective decision-making approach; leadership style needs to give direction and to unify all people in one team (i.e. integrated leadership approach); breakdown of all functional silos; management needs to employ a common mindset culture based on a two-way relationship with a **human element** that promotes understanding, appreciation, respect and with an honest approach, both within and outside the focal firm (i.e. humane orientation); make the employees understand and share the firm all strategic changes and innovative developments (i.e. change management approach); and performance measurements to review the overall SC performance and take remedial actions in all variances to keep in line with all KPIs.

6.3.2 The business strategy

The company **business strategy** is determined by the top management (e.g. Chairman, CEO, Directors and senior management) to establish the strategic direction. The company may vary between a fullyfledged manufacturing firm to a firm focused on a relatively limited part of manufacturing (i.e. contract based manufacturer or sourcing firm) with either a **local or a multinational ownership** (i.e. local subsidiary), where both type of firms need a dedicated organisation structure. The **business strategy**, in-conjunction with the **sales and marketing strategy**, are used to determine the strategic portfolio of the company, such as undergoing diversification or enter new markets and also on how to manage all operations through its **business strategic decisions** on all actions across all the management levels. Typical strategic decisions focus on: the cost cutting measures in all resources; the level of understanding and empathy to **human resources** requirements in training and support initiatives; the level of investment in technology (i.e. ICT and automation); the management policies and quality procedures to be adopted to be in line with all standards and legislative requirements, within the particular business manufacturing sector; the sustainability measures (i.e. the triple bottom line); the strong cultural values to be ingrained to perform in a cross-functional approach; the customer level of interaction and support to be adopted to each customer segment to achieve the right valueadded approach per segment; and other best practices, related directly or indirectly, with the sixteen sub-categories under the holistic SCI management approach, as referred in the research theoretical model in Figure 6.1.

The top management needs to align the business strategy, with both the **manufacturing strategy** and the **SC strategy**. The manufacturing strategy needs the support of the SC strategy to reach effectively all SC members through an established SC structure. The SC strategy needs **SCI** to define the nature of the focal firm processes integration type with all of its members, within the SC structure, from the suppliers to the end customers, under a pull-SC strategy approach (e.g. employing a partnership or close collaborative initiatives or using an arm's length relationship). It is to be considered that the emerged SCI theory is based on a set of **contextual conditions**, which have an influence on the interaction of all the theoretical concepts, as referred in Table 5.1, Figure 6.1 and Section 5.3.3.

6.3.3 The manufacturing strategy

The manufacturing strategy is responsible of all the manufacturing processes to be deployed depending on the manufacturing sector, to be in line with all engineering practices, quality standards and all sustainable and legislative measures and lean management. The focal firm manufacturing strategy determines the span of the manufacturing process to be adopted. Such a strategic decision depends on a complex set of variables which need to be considered based on the '**make' or 'buy' decision** to determine the best feasible and practical solution between the 'make' and the 'buy' options.

The **'buy' decision** refers to an outsourcing strategy with the use of contractors to perform part of the manufacturing. Such an outsourcing stance may reach a relatively high level of manufacturing efficiencies, since contractors can deploy the necessary manufacturing processes at lower costs and higher efficiencies and produce the products with the right delivery time, competitive price and a high level of quality, in line with the established contractual agreement. Such a sourcing stance may also be supported by the focal firm, by supplying them the required raw materials or providing the moulds, to promote a high level of SCI and at the same time the focal firm remains responsible of the works and protects the intellectual property rights.

On the other hand the **'make' decision** depends on various variables such as: competitiveness vis a vis the operational cost and quality as the result of full ownership of the whole process; type of activity (i.e. the nature of value-added operation, whether core and non-core); level of control over the operations; volume of units to be manufactured; cost feasibility of engagement in the activities to get a competitive price; availability of capabilities and resources to meet the demand with the right turnaround time; funds available for the investment to make in-house; and investment time-span to make in-house (i.e. how long it takes to reap the investment).

As a result the manufacturing strategy, wherever possible depending on the relevant sector, shall heavily focus on technology deployment through automation so as to promote high accuracy, productivity and competitiveness with the right change management approach to promote a continuous improvement approach. Technology deployment in ICT applications, enable information management and information sharing all along the manufacturing process to promote visibility and traceability with all the needed updated notifications and assists in the improved efficiencies of the overall process, by being informed of the overall manufacturing equipment effectiveness and on any disruption from the scheduled productivity level to undertake the necessary recovery actions to minimise the risk of any failure.

The **human element** plays a pivotal role in all manufacturing and SC processes for his/her intellectual and management abilities and flexibility, but if used to perform processes that can be automated, this will lead to a higher cost, due to the lower rate of human based productivity. All the necessary preventive and corrective actions need to be undertaken within the overall process, both within the focal firm and outside with all SC members, to identify all obstacles at the earliest so as to take the right decisions not to disrupt the manufacturing process and to promote an effective risk management approach to minimise waste, in line with lean management.

6.3.4 The supply chain strategy

The SC strategy is function of the size of the firm (e.g. micro, small and medium); the type of **outsourced activities** (e.g. core and non-core outsourcing); the type of **sector** (e.g. beverages, wet and dry foods, electronics, medical accessories and automotive parts); the type of **product** (e.g. high value added product to low cost low value-added product, such as industrial AC equipment or battery valves or medical disposal items such as pipes, drips canisters); the **number of intermediaries** involved in the SC, such as distributors at the upstream and downstream sides respectively (e.g. distributors of supplies and distributors of the finished products) and the **human element**, that links all the SC members and the internal focal firm staff together, into one whole entity. Such a **holistic human**

element approach is achieved through different types of collaborative initiatives, with a certain level of flexibility to adapt to the demanding and changing requirements of all parties, to build up a responsive SC with the necessary hard and soft skills of all involved individuals, which are needed to promote effective teamwork across the SC depending on the role of each SC actor.

Henceforth, in the upstream side of the SC, SCI is function of the type of supplies and the established contractual agreement terms and references. Suppliers may supply part of the core components of the final product, which will need both SC members to engage in a close collaborative relationships as partners, function of the SC members human element, based on technical (knowledge-based) and operational information (such as stock levels, prices and quality). In case of commodities type of supplies, SC members need to engage at an **arm's length** relationship, by using suppliers to supply only raw materials, which will only exchange operational information. Similarly, in the downstream side of the SC, SCI depends on the type of product and whether an **intermediary** exists between the focal firm and the end customer. For high value added products, there is the need to employ close collaborative relationships, from a human element perspective, with the customer, in line with the established contractual agreement terms and references, to promote a level of understanding and a competitive edge in the given value-added services (e.g. such as site visits and demos; products traceability; and technical support). Whilst for other volume runners, the collaborative relationship type with the end customer is relatively lower, if not negligible, which may be focused on advertising the product, such as on social media or on the web or on the general media. With regards to the outsourcing strategy, a large firm tends to outsource more than a small firm due to the availability of the organisational resources, which as a result can manage effectively all interactions with all SC members and at the same time, can focus better on the core competences (i.e. stick to the knitting).

6.3.5 The IMLA in-conjunction with the business, manufacturing and supply chain strategies

The **business Strategy, manufacturing strategy** and the **SC strategy** with its **SCI strategy**, they all rest on the **IMLA**. The **IMLA** variation is defined by following two extreme ends. At one end, the company need to employ an effective management approach to meet its KPIs and its quality standards, sustainable work practices and legislative requirements of the manufacturing sector, with the right leadership approach, to influence and motivate the employees, in a **humane approach**, to perform effectively and efficiently. Such a stance ideally needs that all management levels have to involve all concerned staff in all decision-making activities and at the same time all staff need to adopt empowered initiatives and a culture of change management approach, in line with the **strategic leadership approach**, based on the distributed leadership style. Such a proactive effort is in line with the lean management approach, so as to gain from cost savings and higher quality in all processes,

with a value-added and holistic process approach, to manage all risks effectively and efficiently. The other extreme end of the variation is a situation where the management does not involve the concerned staff in all decision-making and as a result the staff adopt a reactive approach to work, in line with the **individual leadership approach**. In the latter leadership approach, management lacks a **humane appreciation** of the employees' contribution, since there exists a management and technical divide, where the quantities count more than the quality of the manufactured products.

6.3.6 The holistic SCI management approach.

The IMLA is implemented through a holistic SCI management approach across all the SC, both within and outside the focal firm. This holistic process approach consists of a complex and interrelated set of concepts so that all the operations of all SC members create the necessary synergy, inconjunction with the focal firm, so that the overall SC will optimise and align all processes in one holistic approach vis a vis an IMLA. A one size fits all model does not exist, since every SC needs to adopt the right holistic process according to each contextual conditions. The holistic process approach is represented by a set of sixteen conceptual actions, as referred in Figure 6.1 below. The responsibility of the process ownership is ideally achieved through team leaders across all the organization levels with the support of the individual employees empowered initiatives. The holistic process approach, from the human element perspective, shall be based on human competences which shall be multi-skilled for staff to adopt a teamwork approach both within the focal firm and with all SC actors. The manufacturing process must be in line with all quality based procedures to meet all standards and sustainable measures, in line with the respective manufacturing sector certified bodies.

On the other hand, the holistic process shall not be based on using the **human element** with a multitasking approach, with lack of focus on all activities, by doing everything but with lower efficiency, which may also create friction and shortcomings in performance, due to the lack of uniformity and quality-based practices.

The holistic process approach shall give the expected competitive edge to improve the overall SC performance for a win-win situation to all SC members. Any SC member failure or bottleneck within the SC operations will cause a SC disruption, unless the necessary risk management measures are in place, from a holistic perspective, to withstand such a domino negative effect, from breaking down the SC.

The holistic SCI management approach is enabled and sometimes even driven by technology deployment but its effectiveness is also dependent on the IMLA, since it governs the human element

of the whole process who implements all information sharing initiatives and technological based processes. With such an IMLA, the focal firm, in-conjunction with all the SC members interconnected by an information platform, through information systems and automation based technological based applications, the human element still plays a key role to achieve the necessary integrative approach and team stimulus needed to perform effectively and efficiently to achieve a competitive performance.

The **holistic SCI management approach** consists from the following dimensions with the referred properties and their variations:

- I. Auditing Operations (i.e. A variation exists by either auditing all internal and external processes (e.g. audit suppliers; audit internal in-house processes and external audits of SC members; and audits by Corporate). The other extreme end variation is performing no auditing at all);
- II. Managing Cash-flow (i.e. A variation exists that either adopts a VMI or consignment stock to improve on cash-flow with favourable payment agreements with OEM suppliers or distributors and with the efficient use of all internal resources to be economically sustainable (i.e. by adopting the right cost cutting measures). The other extreme end variation is a situation of overstocking and lack of effective cost management in all resources, with the result of problems associated with both cash-flow and stock obsolescence);
- III. Managing Change/Innovation (i.e. A variation that either the focal firm focuses on a continuous or periodical improvement approach to be innovative, creative and lean, so as to use the latest advanced processes with the support of latest technology and engage in innovative products. The other extreme end variation is a situation where people from the focal firm, and from different or all management levels, offer resistance to change to maintain a culture of status quo);
- IV. Managing Collaboration (i.e. A variation exists where the focal firm externally adopts either an arm's length or a close relationships with sub-contractors, suppliers or subsidiaries. Such collaborative relationships depend on the role of each SC actor and are supported by contractual agreements (e.g. partnering with other SC members, such as suppliers and customers especially for high-value products). The other extreme end employs functional silos both within and outside the focal firm with no commitment to build any type of close relationship. Internally the collaborative initiatives may reach a conflicting and critical level, when the top management creates a management and technical divide, where all functional areas lack the teamwork environment. Externally with other SC members, the focal firm employs opportunism with no commitment to work together as a team);

- V. Managing Culture (i.e. A variation exists where the SC needs to adapt and work in conjunction with various SC members with different cultures across the globe. The focal firm needs to be sensitive and flexible to build sound management practices with the right mindset and values based on a collaborative relationship to promote understanding and synergy both within and across the SC. The sources of differences are attributed to the different: values assigned to time management; calendar working days as a result of their local festivities; level of sensitivity allocated to others' needs (i.e. a one-sided view against a more flexible and understandable approach may be taken in collaborative initiatives); levels of details and support needed in all interactions (i.e. high to low level); and approach to the skills requirements for the assigned job roles (i.e. broad skills against narrow skills capabilities). As a result the focal firm needs to build a culture that excels in all operations and adopts all the necessary contingency measures against possible bottlenecks, both internally and externally, due to the various members' differences in values and deviations in their commitments that form up the SC, so as to recover from any shortcomings and to maintain the expected level of performance. The other extreme end is based on a SC network with various conflicts and disagreements in all interactions both within and across the SC due to the lack of understanding and cooperation culture originated from both the focal firm and any SC member);
- VI. Managing Customer service (i.e. A variation exists that either the focal firm focuses on a commitment to promote proactive efforts to provide an edge in all customer service quality (such as on-site visits and real-time high value-added support with customised specifications, especially for high-value added products through niche markets) with the right sales and marketing strategy. The other extreme end is a situation where the focal firm takes a reactive or minimal effective customer service approach, where the support provided consists from a basic quality of service, targeted for the broad market);
- VII. **Managing Information sharing** (i.e. A variation exists where the focal firm exploits all available types of information and knowledge to use each other' resources and capabilities so as to build a well informed mindset through effective communication of all information to the right person and at the right time, from the suppliers down to the end customers. The information exchanged may include operational, tactical and strategic information so that the recipient, either within or outside the focal firm, will be in a position to take informed decisions to reach a win-win situation. The other extreme end is a situation where information is shared to cover operational quantities and with no information and knowledge sharing at all);
- VIII. **Managing Knowledge** (i.e. A variation exists that either adopts cross-cultural knowledge initiatives across the SC to promote learning and change and with ad hoc on-going training to

all staff both within and outside the focal firm, with special attention to protect sensitive information through non-disclosure agreements. The other extreme end is a situation where no training is given at all to the staff within the focal firm and with no knowledge sharing across the SC and with no provisions to protect the focal firm knowledge and intellectual property from any leakages to competitors);

- IX. Lean management (i.e. A variation exists where all operations are aligned and optimised to promote value-added activities, by minimising waste in both manufacturing and administrative processes in line with all quality standards and sustainable measures. It is a systematic approach to identify problems at the earliest and to promote a continuous improvement approach in all operations with a level of flexibility to achieve effective and efficient performance. The other extreme end is a situation where there exists wastage of all types of resources, such as reworking and duplicated activities without quality based processes and sustainable measures and an operational environment based on a lack of drive to optimise and align the overall process);
- X. **Managing Logistics** (i.e. A variation exists where transportation is monitored through GPS tracking and information updates of all movements on the freight forwarder web portals, to promote SC visibility of all merchandise movement to take immediate recovery action/s in case of any shortcoming/s. The necessary measures are adopted by the focal firm to establish the mode of transport with the most sustainable route, to ensure timely delivery (e.g. employ groupage approach with other shipments). The other extreme end is a situation where transportation delays are incurred without any updates what is happening and without having the possibility to have any access to information to take the necessary provisions to make up for any delivery problems);
- XI. Managing Performance Measurements (i.e. A variation exists where measuring is deployed on all the focal firm key work practices, so that all staff are informed on all performance achievements, and to promote a continuous improvement approach in line with the KPIs across all management levels, including the bottom-line. Upstream and downstream SC members are also included in such performance measurements, so that any bottlenecks are identified and immediately rectified so that the focal firm meets the customer side targets, with the right check and balances within the overall SC process. All the necessary initiatives are implemented to avoid measuring the wrong activities and/or failing to extract the right values. The other extreme situation is that of measuring nothing at all and keeping employees in the 'dark');
- XII. **Planning/Forecasting** (i.e. A variation exists where the focal firm takes all the necessary actions to have the information from all downstream SC members, which although may be subjected to various errors, is needed to derive the master production schedule. Such forecasts

drives the procurement process in a timely manner, with limited buffer stock dedicated to volume runners to effectively manage the risks of stock-out and cash flow respectively. The other extreme is a situation of no forecasts at all and the stock inventory approach adopted is based on past information or on experience which may result in stock-out or excess stock situations that may easily turn to obsolete stock with cash flow problems);

- XIII. Managing Quality & Sustainability (standards & legislations) (i.e. A variation exists where the focal firm and all SC members need to be in line with the established manufacturing sector quality standard procedures (e.g. quality manual) and international certified bodies (e.g. Underwriters Laboratories (UL) standard, ISO and BSI) and sustainable practices. The quality standards procedures and sustainable measures promote high quality practices, a safe workplace and an environmental conscious and safe product to achieve a sustainable and high quality product from an economical, social and environmental perspective respectively. Particular sectors use tougher standards and require adherence to legislative requirements, such as in automotive and pharmaceuticals sectors (e.g. TS 16949 and GMP). In such sectors all processes need to be inspected, validated and measured to capture any deviations from the scheduled quality standards. The manufacturing process associated with all the products shall employ traceability measures to promote recovery at any point in time due to any The other extreme end refers to situation with lack of unacceptable quality level. conformance to quality either by the focal firm or by any SC member, where standards are not enforced and followed in line with the established procedures and sustainable measures, which may end up to excessive waste and rejects and even uncertified products leading to the possibility of SC failure);
- XIV. **Managing Risks** (i.e. A variation exists where all operations are assessed, monitored or audited to promote quality assurance and business continuity, in line with the contractual agreements and to enable the necessary systematic and timely recovery actions from any problematic situation (e.g. lack of supplies due to delays). Risks are mitigated by taking the necessary preventive measures against such bottlenecks (e.g. use multiple supplier arrangements or reservation of stock). The necessary mark-ups within all costings are included to make up for any unforeseen increase in prices to maintain economic feasibility in all orders. The other extreme end refers to a situation where no risk assessment and/or mitigation is undertaken and failures are accepted as they come, leading to the possibility of SC failure);
- XV. **Supply management** (i.e. A variation exists where stock is purchased function of demand/forecasts, quality (e.g. specifications), price and lead-time to adhere to all standards, sustainability and legislations so as to promote both an effective stock turnover and cash flow respectively. The other extreme end refers to a situation where stock is purchased to promote

low cost with no safeguards against stock-outs due to low order levels or obsolete stock due to excessive volumes and its associated cash flow problems); and

XVI. Managing Trust (i.e. A variation exists where trust is used to build long term relationships and collaborative initiatives between all staff, both internally and externally of the focal firm. All the information and knowledge exchanged, promotes team building and represents the real requirements of all SC members to promote value-added activities in line with lean management. The other extreme end variation is a situation where silos are built both between staff within and outside the focal firm, where information exchanged is not reliable and may cause the SC bullwhip effect due to inflated requirements, leading to excess inventory associated problems, such as low inventory turnover, obsolete stock and negative cash flow).

6.3.7 The technology deployment

Technology deployment in IT applications and automation technology acts as an enabler of the holistic SCI management approach and may adopt two extreme process variations. It may either be focused on latest advances and applications to promote business process engineering in all workflows, within the focal firm and across the SC, through a high level of automation based technology. Such an approach is used to promote informed decision-making for a winning manufacturing and SC strategy, within the focal firm across all management levels and even outside with all SC members, based on an effective information sharing platform to promote an open, common and real time communications information systems (e.g. ERPs or database based applications). Such a technology deployment enables all processes to promote a high level of integration of all human interactions with streamlined processes across the SC and of all manufacturing processes (e.g. automation) to improve on the business performance. Technology adapts to all processes, and may also be used to drive all information systems and automated processes, for improved quality based processes, where within manufacturing, it enables a high level of production volumes with lower costs per unit to promote a competitive edge in all operations (e.g. achieving high levels of productivity with the right accuracy and efficiency through automation). On the other extreme end, the deployment of IT and manufacturing technology may be resisted by the workforce and instead mainly implement verbal communication methods with pen and paper methods and deploy manual manufacturing systems, which are preferred over technology based automated manufacturing, resulting in competitiveness failure.

Attention shall be given in cases of IT holistic technology based applications, such as with the implementation of an ERP or a custom database, where such a system may drive the focal firm to be less flexible and create a rigid approach to adapt to minor or basic applications requirements, causing

the focal firm to become victim to the system. Under such conditions, the users will have no other option, but to bypass the current rigid IT based system and deploy traditional-based legacy systems, by patching up the rigid systems with add-on applications, to promote flexibility and meet such requirements. Furthermore, in particular job production sectors, the manufacturing processes may be heavily manually based, with minimal technology based automation, since the nature of the products cannot be manufactured otherwise (e.g. switchgear and printing manufacturing sector). Though, it cannot be excluded that, within almost all manufacturing processes, a level of human element is always needed, since it plays a key role within the overall manufacturing process. Any manual approach that needs to be adopted must ensure that the human element must not create any bottlenecks in productivity within the manufacturing process, but acts only as an interface within the overall process.

6.3.8 Competitive capabilities

The competitive capabilities, as the outcome of the model, is driven by the IMLA, through the commitment and empowerment initiatives of all involved staff (i.e. human element), within the SC and manufacturing operations, made up from the right strategies (i.e. business, manufacturing and SC strategies) and actions (i.e. holistic SCI management approach enabled by technology) within a context. It is to be noted that the holistic SCI management approach has a set of sixteen actions (Figure 6.1), which serve as the guiding yardstick for all to follow, which need to be managed effectively with the right trade-off for the common good over the individual benefits of all SC actors, which are function of their contextual conditions, so as to serve as a means to an end for the overall SC competitive capabilities.

The key outcomes from all the tentative categories were: SC flexibility and visibility; business continuity; continuous improvement; optimised and lean operations; effective and efficient performance; sustainability and effective customer service. Such an integrated approach serves as a platform for all SC actors to create value-added and optimised operations and at the same time respect all the quality and sustainable measures to produce a competitive product with a competitive service to the end customer. Such an approach enables the focal firm with all its SC actors to build distinct competencies relative to other SCs to be competitive in the market so as to promote success.



Figure 6.1: The emerged SCI theoretical model (Author)

6.4 Conclusion

This chapter has presented the final stage of the grounded theory data analysis. By applying the Straussian GTM, the core category, labelled as the 'Integrative Management and Leadership Approach' earned its way to link all the emerged theory together. The generated substantive theory is based on the core category with the relation to other five (5) main categories, where one of the main categories encapsulates 16 sub-categories, as referred in Figure 6.1 and Table 5.1.

The research rigour was maintained by undertaking all the data collection and analysis process in Chapters 4 and 5 in line with the Straussian GTM three coding techniques, to promote reliable and valid findings, as already referred in the research methodology Chapter 3. Special attention was given to the constant comparison process and memos writing, to promote an audit trail of all analysis and to ensure that all interpretations are in line with the confirmability and control research quality criteria (Glaser & Strauss, 1967). Such a stance enabled the researcher to theorise on the data with the right sense-making approach to ensure that all participants are put back to life within all data analysis, by substantiating all the emerged theory grounded in data with key selected in vivo quotes from the data, in line with the fitness (i.e. credibility) research quality criteria (Ibid). All research participants selected, formed part of the population within the manufacturing sector in Malta, who were highly knowledgeable of their respective businesses and well versed with the supply chain responsibilities and in the role of IT at the workplace, in line with the dependability research criteria to generate rich data (Ibid). Such a stance also contributed to the generation of an understandable theory that also fits the SCI phenomenon understudy in line with the research quality criteria (Ibid). Furthermore, the generated theory within this Chapter through the final theoretical memo has taken a holistic perspective approach and with a level of variation in all categories, due to their properties dimensional variations, which both makes the research flexible enough to comply with the GTM generality and fitness principles of research quality criteria (Ibid). Henceforth, the grounded theory generated, being in line with Glaser and Strauss's (1967) criteria of fitness, generality, understandable and control, promotes research rigour since it meets the GTM quality assessment principles based on the credibility, plausibility and trustworthiness (Section 3.6.3).

This chapter is also used to drive the post literature review (Section 2.6), since at this stage all the analysis have been completed and the substantive theory has been generated with all its emerged themes in line with the GTM (Strauss, 1987; Glaser, 1998). The post literature review is directed by all the emerged final categories to situate the substantive theory within the extant literature. With the completion of the post literature review, the researcher was in a position to proceed with the research discussion (Chapter 7), to integrate the emerged theory within the extant literature, so as to identify the commonalities, the differences and the new (Glaser, 1998).

Chapter 7: Discussion - Theoretical Integration

7.1 Introduction

This chapter will integrate the emerged theory from the data within the extant literature in line with the GTM for logical extension and comparison of the emerged theory with the literature (Strauss, 1987) and also for theory building (Eisenhardt, 1989). To achieve such a stance, a comprehensive and systematic post literature review was carried out based on all emerged themes, as referred in Section 2.6, with the completion of the data collection and analysis process in Chapter 6.

The research question that guides the study and supporting objectives is the following two-part question. First, which are the characteristics needed to achieve SCI within SMEs in the Maltese manufacturing? Second, which are the key factors that constitute SCI that promotes competitive performance within SMEs in the Maltese manufacturing?

The scope of this chapter is to relate the substantive theory with the literature, to answer the research question and to derive the study implications to the extant literature. The substantive theory is also refined by making reference to the literature. This chapter also explores all the relationships between all emerged themes within the theory of this study which is building up the SCI theoretical model.

Henceforth, the study findings are related with the relevant literature so as to situate the generated theory within the extant literature, identify where the research fills the literature gaps and also to establish the research contribution. Such research discussion and its contribution are determined by identifying the commonalities, the differences and the new, between the current literature and the substantive grounded theory (Glaser, 1998).

7.2 The emergent theory within the extant literature

The study research aim is to generate a theory on supply chain integration (SCI) by focusing on the manufacturing sector SMEs within Malta. The research main objective is to establish the key SCI characteristics. The data was obtained from a set of twenty two (22) in-depth interviews, based on seventeen (17) firms, in various sectors within manufacturing, so as to capture all the differences, since the supply chain (SC) is contextually dependent on the type of product. The study derived a core theme from the data, based on a set of other main themes and sub-themes which constitute the foundations of the generated theory on SCI. All the themes have a set of properties and dimensions with a level of conceptual variation for each theme, which constitute the SCI theory. All the themes were defined, explained and depicted in the theoretical model, as referred in Figure 6.1, after the completion of the data collection and analysis in Chapters 4, 5 and 6.

7.3 A high level overview of the key findings: The IMLA explanation, originality, significance, definition and challenges

7.3.1 The IMLA explanation and its originality

The research findings show that the theory for SMEs to achieve supply chain integration (SCI) requires an integrative management and leadership approach (IMLA). The IMLA, as the research core theme, is considered to be a concept within the SCI theory and it is defined to be both a driver and a binding force of all the SC actors. The IMLA conceptual term foundations are based on organisational behaviour theory, which guides professionals and practitioners in their relationships, engaged within the manufacturing based SC, both within any firm and also with all of its SC stakeholders.

The IMLA is based on the foundations principles built from the management and leadership concepts respectively. From a management perspective the IMLA promotes mutual understanding between people so as to comply with the established procedures in line with all the quality and sustainable measures across all the management levels to achieve competitive capabilities. From a leadership perspective the IMLA inspires people to employ innovative, flexible and transformational leadership processes with a continuous improvement approach to achieve competitive capabilities. The IMLA sustains the established cliché that management (Watson, 1986) and leadership (Bennis, 2007) are not only scientific disciplines but are also performing arts and need a level of creativity and wisdom to establish the common good both within the focal firm and across the SC.

The core theme IMLA, as the name implies, is unique within the academic literature. It represents the concept needed to achieve SCI, being based on the dual concepts of management and leadership but in an integrated approach. The empirically generated theoretical model, as referred in Figure 6.1, is also unique, since the way all the concepts are inter-linked together in a comprehensive way is also original. The IMLA is also significant and contributes towards the gaps in literature for more theory regarding how SC actors deploy SCI initiatives within the manufacturing SMEs sector to achieve competitive capabilities. The definition of the SCI theory based on the IMLA is referred in Section 6.2 and is depicted in Figure 7.1 below, based on a simplified version of Figure 6.1, by outlining the emerged core category with its five main categories and the outcome category.



Figure 7.1: The SCI theoretical model based on the emerged main categories (Author)

7.3.2 The study significance and literature gaps from a multiple theoretical perspective The study significance is attributed to various literature gaps:

(a) From the supply chain management literature it was inferred that research is quite limited to explain how to achieve SCI (Zhao et al., 2008) and how to determine the influencing factors of SCI (Flynn et al., 2010). Furthermore, there is little empirical research within connected supply chains containing SMEs exploring larger and smaller firms' perceptions of the value of supply chain information integration (Harland et al., 2007). Finally, it was stated that the social element, 'in terms of labour, skills and the forming of relationships' is a neglected research area (Ashby et al., 2012, p.506) and that 'downstream integration has received little attention in supply chain research' (Guan & Rehme, 2012, p.188).

- (b) From the leadership literature, it was asserted that the 'management of integrated supply chain' is a 'largely ignored area' (Defee at al., 2010, p.764). There are few scholarly works which explicitly consider network based leadership in the macro context of inter-organisational networks (Sydow et al., 2011). The study validity is reinforced by assertion that the 'theory surrounding supply chain integration is still underdeveloped' (Schoen & Swink, 2012, p.110). The interest towards 'power and dependence in buyer–supplier relationships' ... 'has been wavy since 1950s. However, the studies discussing the interplay between these two important topics of relationship management are rare' (Kahkonen et al., 2015, p.151).
- (c) From the manufacturing literature, the manufacturing sector needs further research in the area of 'holistic research approach integrating technological, managerial and business-model solutions together with a change in the customer behaviour, with the perspective of a more sustainable manufacturing for the production of more sustainable products and services' (Garetti & Taisch, 2012, p.88).

7.3.3 Overview of the challenges to achieve SCI within the literature relative to the research core theme

The study outlines that supply chain integration (SCI) requires an integrative management and leadership approach (IMLA), as the core theme of the theoretical model, to achieve the necessary competitive capabilities. Although the leadership and management concepts are well represented in the literature within various academic disciplines, they are more focused from an organisational perspective and exclude the wider contextual perspective to include inter-organisational systems (e.g. Osborn & Marion, 2009). Lambert et al. (1998, p.12) declared that within the SCM context, 'The managerial and behavioural components were, in general, less well understood and more difficulties were encountered in their implementation', relative to the physical and technical management components (e.g. organisation structure, communication and information flow structure). It was also inferred that, 'Managing the entire supply chain is a very difficult and challenging task' (Lambert & Cooper, 2000, p.68) and it was contended that more SCI is beneficial and better than less integration (Swink et al., 2007). Furthermore, Sydow et al. (2011) inferred that studies that combine leadership and management concepts to achieve SCI, within the SC context within inter-organisational networks, is rare.

The research findings complement and address these gaps by focussing on the importance of the management and leadership concepts. The IMLA, with its empirically derived theoretical model, outlines how to manage the overall SC to meet all challenging scenarios both within and outside the focal firm.

7.3.4 Comparison of the key contemporary theoretical models from the literature with the emerged theory

A set of four scholarly works are being referred to situate the core research findings based on the IMLA within the extant literature. There are few other papers which mentioned the terms 'management' and 'leadership' to show the importance of such conceptual elements in the SC business processes but were not the main focus of the studies (Lambert et al., 1998; Cooper et al., 1997; Cowan et al., 2015).

7.3.4.1 Integrated chain management study

A close concept to the IMLA is the integrated chain management (ICM), which refers to a literature review based on German and Dutch manufacturing (Seuring and Muller, 2007). Seuring and Muller (2007) studied the ICM concept, within the context of sustainable SCM to achieve sustainable competitive advantage. In this paper abstract, ICM was described as the 'supply chain management that takes environmental and social issues into account' (Seuring and Muller, 2007, p.699). Such a definition left out the economical dimension, but within the paper content, it makes reference to both the economic activities and firm's profitability. The research findings complement this paper and outline that the deployment of the social, economical and environmental practices has been found pivotal and contributes to a sustainable competitive advantage. Such research findings also corroborate with sustainable SCM practices within the extant literature (Elkington, 2002; Seuring & Müller, 2008; Buyukozkan & Berkol, 2011; Sarkis, 2012; Wu et al., 2012; Caniels et al., 2013; Glover et al., 2014; Hsueh, 2015).

The ICM theoretical framework within this paper was based on a set of transactional and transformational processes to reach competitive sustainability. ICM was also analysed from three theoretical perspectives: the material and information flow (i.e. production and logistics management); the strategy and cooperation (i.e. to actively manage product and material flows); and the industrial network (i.e. exchange of materials to promote sustainable practices). The three perspectives scope was to establish how to manage effectively all the elements, with all the necessary synergies (e.g. symbiosis), efficiencies and recycling, to promote an overall supply chain sustainability of all materials usage within manufacturing from the cradle to the grave. The findings, based on the IMLA, complement and also serve as a theory extension of the ICM concept to achieve competitive capabilities, since the emerged theory covers all the above three foundations perspectives and also provides a more comprehensive theory. The emerged theory is more comprehensive since it is based on the IMLA and supported by the business, manufacturing and supply chain strategies respectively.

which all determine the holistic SCI management approach, based on set of sixteen concepts, enabled by technology deployment.

The ICM transactional and transformational processes, within this paper, were mainly based on the material flow, whereas the research findings considered also the transactional and transformational processes from a management and leadership perspective, together with the consideration of the servant and contingent leadership approaches. The IMLA also adopted a special focus on the human or social element at all management levels, which is only defined in the ICM model but is not tackled. Furthermore, the paper literature review did not locate any publications which included the social issues, so the IMLA with its human element special consideration within this thesis, is an extension of the ICM theoretical perspective both within this paper and to other referred papers within this scholarly case.

7.3.4.2 SC leadership and SC followership study

Defee et al. (2010) focused on a transformational leadership approach within a SC for both leaders and followers, which is also a close concept to the IMLA theme. The paper focused on two concepts referred as SC leadership and SC followership, using a research methodology based on a simulation game environment, provided by several US Universities. This academic work included only the transformational leadership approach and excluded all other types of leadership approaches, as referred in this thesis's findings, which are transactional, servant and contingent or situational, which are considered pivotal to meet the SC dynamic conditions. Furthermore, the paper only focused on three mediating constructs and two outcomes, namely information availability, communications and rewards, as the mediators, which led to efficient and effective performance, as the outcomes. Within the thesis's findings, a total of sixteen constructs are included to represent the holistic SCI management approach, which all cover this paper three constructs. Similarly, this thesis's findings

Such a paper demonstrated that SC followership contribution played a significant role in the SC performance outcomes, once there exist an information sharing platform and also when they were given the opportunity to contribute ideas and make decisions and were also included in a holistic reward scheme. The thesis's findings, corroborate this academic work, since it is identified that the holistic SCI management process approach requires intra- and inter-organisational synergies by deploying: the right performance measurements both within the focal firm and across the SC; and joint-decision making, between the focal firm and all stakeholders (i.e. as followers), so that the overall SC is optimised and aligned effectively. Similarly, the thesis's findings infer that holistic SCI management approach deploys an information sharing mechanism, enabled by technology in information systems and automation, to exploit knowledge from each other and to promote an

informed mindset to the right person and at the right time. Such technology deployment adds to the body of research within this paper's model.

7.3.4.3 Quality management practices study

Kaynak and Hartley (2008) inferred an empirical model, quite similar to the IMLA based theoretical model within this thesis work. The paper was based on quality management practices based on eight dimensions as antecedents that led to a set of performance outcomes (i.e. inventory management, quality, financial and market performance respectively). In the paper, the management leadership, was considered from a quality based approach, and was referred as the core driver of all the other seven dimensions: training, employee relations, customer focus, quality data and reporting, supplier quality management, product/service design, and process management.

The paper deployed a quantitative research methodology and the implications of the study suggested that management leadership, within the SC context in both manufacturing and services firms, was pivotal in creating a culture of continuous improvement, open communication and cooperation based on policies and structures that enable their organizations to be customer focused. Such a theory is also complemented by the thesis's findings through the holistic SCI management approach dimensions with the customer service as outcome with its optimised and lean operations. Furthermore, the paper inferred that the integration of both upstream and downstream processes led to improved quality performance, which is also corroborated by the research findings holistic SCI management approach. The thesis's findings are consistent with this paper theoretical model, since the IMLA complements it, by being the core driver of the findings theoretical model. The thesis's findings consider that both effective and efficient performance represent the operational and business indicators which complements the paper's financial and market outcomes. The paper inventory and quality performance outcomes were also covered within the findings by the supply management and managing quality and sustainability themes, but as antecedents within the holistic SCI management approach.

Within the research findings, the product/service design antecedent, referred in the paper, is implicitly indicated within the findings by the knowledge management (i.e. sharing of knowledge within the focal firm and across the SC), manage change/innovation (i.e. flexible and innovative design), manufacturing strategy (i.e. with its engineering and quality standards) and sustainability (i.e. ensuring low cost and environmental awareness in the product design) themes.

Finally, the findings infer that the holistic SCI management process approach includes a more comprehensive set of dimensions and is also enabled by technology deployment, which adds to the SCI conversation since this is lacking in this paper's model.

7.3.4.4 Supplier integration mechanisms study

Lockstrom et al. (2010) also inferred a theoretical model focused on focal firm leadership approach (i.e. buyer side) as the key antecedent to achieve SCI with suppliers so as to promote SC performance improvement. The research findings complements such a paper and extends on its theory, since the IMLA based theoretical model represents a more holistic picture of how to achieve SCI, by including not only the upstream side but also the downstream side of the SC.

This paper was based on empirical research to establish the factors that facilitated and inhibited supplier integration within the automotive manufacturing SC context in China. The research methodology used a multiple-case study approach on thirty firms with a grounded theory analytic stance. The results indicated that buyer-side leadership effectiveness was an important antecedent for building motivation, trust, and commitment among suppliers which facilitated strategic alignment and all collaborative capabilities and which acted as key enablers for successful supplier integration. The research findings support such an assertion, since the IMLA together with the holistic SCI management approach are creating a motivated environment, both within the focal firm and across the SC with all members, where trust is pivotal in all commitments and collaborative initiatives.

The paper outlined that there exist four main types of leadership styles: coaching/cooperative; situational; assertive; and delegating. In the research findings there also exist four leadership styles, where the situational style is common and is deployed with the scope to meet contingent situations, whilst the other three styles are all covered by the findings except that this thesis includes an added style. Meaning that the transformational leadership style, which is used to meet change and innovation through people, overlaps with this paper coaching/cooperative, assertive and delegating approaches, due to the relationship building, influencing values and empowerment characteristics within such a leadership style; the servant leadership style, to be sensitive for people's needs and also meet business objectives, overlaps with the cooperative approach due to the individual concern characteristics within this leadership style; and the transactional leadership style, was lacking in the paper, where it is used to meet the routine flow of transactions. Such a transactional leadership style expands on this paper theory, since it adds to the current conversation by giving a more comprehensive SCI approach.

The paper classified the achievement of supplier integration outcome, on two categories: supplier collaboration readiness and collaborative supplier capabilities. These categories were represented by a comprehensive set of dimensions which clearly demonstrated the multi-dimensionality nature of the SC constructs to achieve SCI. The research findings complements such a theoretical model from the

literature since SC actors need to be both ready for the relationship commitment and also need to have the necessary capabilities to interact together, as referred in the IMLA core theme with the support of the business, manufacturing and SC strategies respectively with the holistic SCI management approach.

From the paper, the supplier collaboration readiness dimensions were: quality mindset/customer focus; top management support; strategic alignment; willingness to learn/improve; long-term orientation; and trust. The collaborative supplier capabilities were process management; communication; performance management; autonomous problem solving; organizational learning; planning; and engineering/innovation.

The research findings constitute all collaboration readiness dimensions through: customer service for the quality mindset/customer focus dimension; IMLA for the top management support dimension; business, manufacturing and SC strategies for the strategic alignment dimension; holistic SCI management approach employs knowledge management for the willingness to learn/improve dimension; IMLA is a long term approach for the long-term orientation dimension; and holistic SCI management approach employs trust as one of the common dimension to this paper.

The research findings constitute all collaborative supplier capabilities dimensions through: holistic SCI management approach for the process management dimension; holistic SCI management approach employs information sharing for the communication dimension; holistic SCI management approach employs performance measurement for performance management dimension; IMLA incorporates a decision-making approach for the autonomous problem solving dimension; holistic SCI management approach employs knowledge management for the organizational learning dimension; holistic SCI management approach employs planning and forecasting for the planning dimension; and holistic SCI management approach employs change and innovation for the engineering/innovation dimension.

7.4 The emergent themes within the extant literature

7.4.1 The core theme: Comparison of the integrative management and leadership approach (IMLA) within the extant literature

7.4.1.1 The Management concept within the IMLA based on the focal firm and across the SC

From the literature, as referred in Section 2.6.3.1 in the literature review of the management concept, within an organisational context, managing effectively through people to achieve results is a key concern within the human resource management (Drucker 1979; Crainer, 1998) and also within the SC

context (Ashby et al., 2012; Ellinger & Ellinger, 2014). The management role is about meeting the objectives, productivity and social responsibilities but also needs to establish measurements on the individual and the organisation to monitor performance improvements and optimisation (Drucker, 1977). Nowadays, managing effectively within an organisation does not only consider the social and economic perspectives, but also needs to cater for an environmental perspective, by employing environmental friendly measures and 'green operations' to promote environmental sustainability (Bojarski et al., 2009). In fact contemporary research considers a sustainable SCM (SSCM), which means that SCM needs to be sustainable within the social, environmental and economical goals, also referred as the triple bottom-line (Elkington, 2002; Vachon & Klassen 2006; Cater & Rogers, 2008; Seuring & Müller, 2008; Buyukozkan & Berkol, 2011; Carter & Easton, 2011; Ashby et al., 2012; Sarkis, 2012; Wu et al., 2012; Caniels et al., 2013; Glover et al., 2014; Hsueh, 2015).

Such a management literature overview is complemented by the findings, since from a social perspective the IMLA is about managing people both within focal firm and across the SC, where the human element remains the backbone of all the processes. The human element within the focal firm needs to be unified in one team to breakdown all silos through a cross-functional approach, so as to employ a common mindset to promote understanding, appreciation and respect and furthermore to promote an innovative approach on the job. The findings also show that the social element needs to extend to all stakeholders, by employing a teamwork approach with all different SC actors, based on dedicated inter-organisational collaborative approaches, depending on the SC actor role with respect the focal firm (e.g. a core supplier or a one-off supplier). Such a collaborative approach needs to synchronise and optimise all activities and measure the overall performance to meet all scheduled objectives and productivity, by managing the whole SC set-up from a holistic approach. The social sustainable approach was clearly declared by the CEO of one firm, who stated that:

'My policy is to demolish all boundaries and work as a team. I destroy all cells or niches, since cells do not talk to each other if they are created' (8TS/INT4).

Similarly, from an economical perspective, the findings imply the need to manage cash flow effectively and to employ lean operations to promote effective and efficient performance with the least of waste in all types of resources and processes. The economical sustainable approach was explained by the operations manager of a firm, who stated that:

'The stock needs to be made available without ending up into cash flow problems' (1/INT2).

Finally, from an environmental perspective, the findings show that environmental sustainability needs to be established at source by the SC and manufacturing strategies and is part and parcel of the holistic SCI management approach, within the focal firm and across the SC, to implement all environmental sustainable measures within the product design and materials usage together within both

manufacturing, such as in product design and waste management approach and SC processes, such as in logistics. The environmental sustainable approach was explained by two participants, who stated that:

'... audited and implemented such EHS programme' (1/INT1); 'Environment and CSR [are both being respected]' (1/INT2).

The findings clearly sustain the extant literature since triple bottom line sustainability is a crucial element underpinning the manufacturing and SC strategies and also within the holistic SCI management approach, as referred in the sixteen dimensions within the theoretical framework in Figure 6.1, and as referred Table 5.1. The SC sustainability needs to cater for all processes within the focal firm in-conjunction with all the upstream and downstream SC actors, since any failure in compliance from any of the SC members, will breakdown the overall SC sustainability objective (e.g. certified green operations by suppliers and logistics providers need to be in harmony with the focal firm). The SC sustainability is pivotal since it needs to incorporate all SC actors, as explained by one participant, who stated that:

"... suppliers, who have to stick to our standards and processes and we visit them and make audits..." (6/INT7).

7.4.1.2 The Leadership concept within the IMLA based on the focal firm and across the SC: Strategic leadership style against individual leadership style

From the seminal works within the contemporary literature on leadership, based on the knowledge era (Uhl-Bien et al., 2007), within an organisational context, as referred in Section 2.6.3.2, the strategic leadership approach needs to shift beyond the dyad approach to a relationship approach, based across different organisations (Hunt, 1991; Finkelstein & Hambrick, 1996). Such a strategic leadership approach is complemented by supervisory leadership approach, where the latter is focused on the dyad relationships and on the organisation lower levels tasks and individual performance (House & Aditya, 1997). The **strategic leadership** approach is ideally focused on a collective or shared or distributed leadership approach (Carson et al., 2007), which is referred as connected leadership (Martin et al., 2007) and not based on a single individual (House & Aditya, 1997). As a result, an effective leadership style needs to be instituted at all levels of the organisation (Deming, 1986; O'Reilly et al., 2010) and shifts to an interdependent leadership style (McCauley-Smith et al., 2013).

On the other hand, the **individual leadership** approach, from the literature, refers to the industrial era, based on hierarchical structures, where the leadership theories were focused on the individual leader (Gronn, 2002), using the tripod ontology: leader, follower and common goals (Bennis, 2007). Ideally, the contemporary shared leadership approach constitutes the way forward to meet the current and

dynamic competitive environment, through corporate intelligence (McKelvey, 2001). With such a corporate wide intelligence approach, based on both a knowledge and a learning environment, needs to be deployed so as to exploit all organisational stakeholders' contribution for both strategic leadership and organisational effectiveness (Boal & Hooijberg, 2000). Though, it cannot be excluded that the current leadership theories, in reality are not adopting the shared leadership approach, but are still embedded in the individual leadership approach, due to the past leadership experiences and cultural influences from the industrial era (Gronn, 2002). Furthermore, the extant literature clearly takes into account the pervasive influence of the individual leadership approach due to the leading role that top management takes on all the people across all management levels (Tsui et al., 2006; Ling et al., 2008).

The research findings complements the two leadership approaches referred within the literature, since from the emerged theory, there exists a variation of the leadership approach within the IMLA. The IMLA variation from the findings imply that at one end, there is both the strategic and the supervisory leadership approaches, based on a shared leadership style, where the leadership stance incorporates all management levels and key SC actors. At the other end, there is the single centric or individual leadership approach, where the leadership style is mainly focused on the top management team. As a result, the findings mainly outline that the focal firm establishes the business, manufacturing and SC strategies, either through a shared leadership approach, in line with the IMLA through the involvement of all relevant management levels across the command chain in-conjunction with all key SC actors or else such strategies are only established by the top management team, through an individual leadership approach, by implementing the IMLA between top management only.

Four participants distinguished and explained the strategic leadership approach, through a shared leadership style, as follows:

'Workers are involved in all decisions'; '... to empower their responsibilities across all departments' (1/INT2); '...leadership is very effective at all levels such as strategic, tactical and operational' (3/INT 4); 'Leadership style is like the glue to match all management levels together'; '... employing an open door policy, within the focal firm and a controlled open book, with all external entities'; 'To make all ends meets, one needs to remain in [his/her] tip toes, to ensure that all processes are integrated' (5TS/INT2); and 'I [as CEO] fully involve myself to lead by example'; 'We delegate both upwards and downwards' (8TS/INT4).

One participant outlined the individual leadership approach as follows:

'... All starts from the top management direction, from the Directors ... They decide the direction and create new policies. I start my work from the policy onwards ... ' (5/INT6).

With such an organizational leadership perspective, the findings also substantiate that such an integrated leadership approach is not only applicable within the focal firm but also within the SC

context between different SC actors. Such a statement adds to a new conversation to the leadership theory within the SCM literature beyond the SC dyad.

The findings further denote that the leadership approach incorporates the necessary strategic decisions to determine on how to effectively lead all operations, both within the focal firm, at all management levels, together with all of its stakeholders across the SC. Such an approach is determined by both the manufacturing and the SC strategies in-conjunction with the holistic SCI management approach, to ensure that the focal firm with all of its stakeholders meet both the collective SC goal and all the SC members' individual goals, in line with the tripod ontology (Bennis, 2007). Though, it cannot be excluded, that the findings also outline that, as defined by the holistic SCI management approach foundations, the focal firm deploys a change management and an innovative approach on how to lead effectively through effective knowledge management and joint collaborative initiatives with all stakeholders, as defined by McKelvey's (2001) corporate intelligence, and at the same time promote an effective strategic leadership approach beyond the focal firm, as defined by Hunt (1991) and Finkelstein and Hambrick (1996), so as to manage effectively the overall SC. Such findings also expand on the strategic leadership literature within SCM. The importance to go beyond the focal firm and undertake knowledge sharing initiatives was described by one participant, who stated that:

'Knowledge sharing is function of SC member (who he/she is). One must share, otherwise we cannot work' (8TS/INT4).

7.4.1.3 The IMLA Four Leadership Traits: Transformational, Transactional, Servant and Situational leadership styles

Based on the seminal work on leadership theories, between leaders and followers, which also fall under strategic leadership approach, are the **transformational leadership** and the **transactional leadership styles** (Burns, 1978; Bass, 1985; Bass & Avolio, 1994, 2000). Bass and Avolio (2000) outlined the five basic components of **transformational leadership** (Section 2.6.3.2.3): idealized attributes and idealized behaviours, inspirational motivation, intellectual stimulation and individualized consideration, which together focus on the leaders and followers relationships and also promote social responsible business practices, which are pivotal in today's institutional environment to promote social sustainable operations.

The findings complement such two leadership theories, especially the **transformational leadership** style, since the IMLA is mainly based on a continuous improvement approach based on change and innovative initiatives to transforms teams (i.e. intellectual stimulation) and not only focuses on formal

structures and systems, so as to promote flexibility and accepts a controlled level of informality to adapt to today's business dynamic business climate. The IMLA key focus is to develop the human capabilities (i.e. individualized consideration) and the relationship approach between leaders and followers (i.e. idealized influence attributed) based on sound ideals and values (i.e. idealized influence behaviour) and to motivate all people at all management levels, both within the focal firm and across the SC (i.e. inspirational motivation), which adds to the conversation focused on the key foundations of the transformational leadership style and to the SCM theory on leadership. Such a leadership approach was explained by five participants in this way:

'... we share the project works and innovative ideas ..' (1/INT 1); 'We are flexible to meet changes, to meet new customers' specs' (1/INT 2); 'We are shifting to new technology (3/INT4); and 'Continuous change is the firm policy' (5/INT 6); and 'R & D is integrated with the plant, so that all manufacturing is backed by a strong team of expertise from R & D' (8TS/INT4).

From the literature, the **transactional leadership** approach is the most pervasive used approach with a set of established objectives and tasks as an exchange process between the superiors and the subordinates based on reward or punishment vis a vis performance (Burns, 1978; Bass & Avolio, 1994; Bass, 1997). From the findings, the holistic SCI management approach needs to the transactional leadership style to optimise and align all processes, within the focal firm and outside it with all of its stakeholders. Such a stance can be achieved by employing all the necessary measures under the holistic SCI management approach. Basically, such a holistic process meets the routine flow of transactions in line with the established performance measures, such as KPIs, among other measures, to mention a few, such as planning, auditing and risk management to promote an effective customer service among other competitive capabilities (Figure 6.1). Such findings add to the conversation on the transactional leadership style theory and to the SCM theory on leadership. Such a routine flow of transactions with their measurements was referred by one participant as follows:

'We measure and discuss everyday all operations'; and '[Firm] performance is also measured and service issues are being measured and gaps identified to ensure improvement' (1/INT2).

From the literature, Greenleaf (1970) also introduced the concept of the **servant leadership** where his theory advocates that a leader's primary motivation and role is that as a service to others, which as a result leads to successful outcomes. Such a theory is also complemented by the findings, since the IMLA, includes instances where leaders employ a more sensitive approach to human needs by showing exceptional caring behaviour to meet individual requirements such as for health problems or special requests for support to undergo dedicated training for self-development. Such findings add to the conversation on the servant leadership theory and to the SCM theory on leadership. Such a management concern for people was explained by two participants, who stated that:

'Human resources is not so dignifying, since they are not resources but people as individuals'; 'People respect you as you respect back' (9TS/INT6); and '.... the human element remains key to establish a long term relationship [with SC actors]...' (17TS/INT 14).

The **situational leadership**, which falls under the umbrella of contingency theory of leadership (Fiedler, 1967; Lawrence & Lorsch, 1967; Thompson, 1967; Hershey & Blanchard, 1969), shifts from the leadership traits or behaviour to contingencies of the situations, where different leadership styles are needed to meet each unique situations. From the findings there exist situations where such contingency measures are taken, as a result of unforeseen bottlenecks or risks originated from both internal and/or external sources, attributed to the various members' differences in values and/or deviations in their commitments, so as to recover from these shortcomings and maintain the expected level of performance (e.g. pre-booking of stock due to stock shortages or to meet unforeseen delays in deliveries). Such findings add to the conversation on the situational leadership style theory and SCM leadership theory. Such a situational/contingency leadership approach was sustained by two participants, who stated that:

'Risk is a key issue, since business continuity is done and is catered by pre-planned actions for all type of contingencies' (14TS/INT11); and '... hence contingency measures need to be taken to bypass such a failure' 15TS/INT12.

From the findings, the emerged theory denotes that the IMLA incorporates all the above listed four leadership styles to achieve the SCI approach, both within the organisational context of the individual SC actors and also in their inter-organisational relationships across the SC. Such leadership qualities are described as: transformational leadership style, to meet change and innovation through people; servant leadership style, to be sensitive for people's needs and also to meet the business objectives; situational leadership style, to meet contingent situations; and transactional leadership style, to meet the routine flow of transactions in line with the established performance measures. Such an emergent theory extends the leadership contemporary conversation within the both leadership and SCM literatures, since it sustains the strategic leadership theory by outlining that the established leadership theories within the organizational context are also applicable to the SC context.

7.4.1.4 The IMLA's collaborative types (close collaborative and arm's length) & processes

From the SCM literature, as referred in Sections 2.6.3.1.3 and 2.6.3.3.1, there is the need to identify which SC links need to be managed, since it is challenging and may not even render feasibility, to manage all SC links unless they are key business processes (Lambert et al., 1998). The management

approach varies from one SC member to another, with a continuum based on close collaboration (e.g. key business entities) to an arm's length (e.g. one-off suppliers).

From the findings, the SC and manufacturing strategies determine the relationship to be adopted with all of its stakeholders. The manufacturing strategy through its 'make' or 'buy' decision, establishes the most feasible approach to promote effectiveness and efficiencies with all of its stakeholders. The SC strategy determines how the focal firm engages with each SC actor, such as through close collaborative efforts or arm's length relationships, depending on the role assigned to each SC actor vis a vis the focal firm manufacturing operations. The referred findings add to the conversation within the SCM literature within manufacturing. Such collaborative types, was explained by one participant, who stated that:

'We have suppliers of the compressors, as the core activity ... these suppliers are partners and have an in-depth relationship, based on years of experience and history record but others are referred as suppliers, that give us commodities products and may be new suppliers. You need to pay before you purchase for such suppliers ...' (5/INT6).

From the literature, the SCM processes may be classified into two types: physical and technical management components and managerial and behavioural management components (Lambert et al., 1998), which implied that such a dual approach is needed for an effective management approach. The scope of the management approach across the SC is to manage from a holistic perspective all SC actors, even at the expense to sub-optimise the individual performance of one or more SC members for the common good of the overall SC (Cooper et al., 1997). The role of the management is to ensure that all SC members, especially buyers and suppliers, through operational linkages, need to have in place 'the systems, procedures and routines that facilitate the more effective flow of goods, services and information' (Cox et al., 2003, p.137). Furthermore, they also added that such linkages 'can extend beyond a single dyadic relationship and operate across many exchange partners within the supply network' (Cox et al., 2003, p.137).

From the findings, the IMLA with the manufacturing and SC strategies in-conjunction with the holistic SCI management approach, manages effectively, from both an engineering and management perspective. The manufacturing strategy, with the IMLA adopted by team leaders and knowledge workers, need to deploy both advanced manufacturing processes, to achieve high value-added manufacturing, in line with the required technical engineering manufacturing standards (e.g. ISA) and product quality standards (e.g. ISO, GMP and BSI) and also to deploy all the necessary managerial practices to manage the human element effectively. The SC strategy, with the IMLA adopted by team

leaders and staff, need to manage effectively with all of its SC actors, through the holistic SCI management approach, through the established collaborative initiatives between all people within the focal firm and across the SC, to meet each other' commitments, even at the expense of a win-lose situation of anyone from the SC actors. Such findings complement both the technical and management components outlined by Lambert et al. (1998), also sustain the arguments by Cooper et al. (1997) win-lose situations and finally highlights the importance of all operational requirements needed by all SC actors as referred by Cox et al. (2003), since in case of the latter, the findings extends this theoretical argument by outlining the need for holistic SCI management approach being based on a set of systems, procedures and routines as established by the sixteen measures under the holistic SCI management approach, as referred in Figure 6.1. The HVM based on a holistic SC process approach for the common good of all SC actors, was explained by one participant who stated that:

'Quality is an integral part of the model and cannot be separated from the SC and manufacturing process'; 'The product is not high value added but we use high value manufacture'; and 'The various stakeholders are after the same criteria but these may be in conflict with their individual targets/criteria' (6/INT8).

7.4.1.5 The IMLA vis a vis SCI and performance

With a focus on the leadership concept, within the inter-organisational SC context literature, as referred in Sections 2.6.3.3.1 and 2.6.3.4.2, it was established that the leadership strategies are pivotal for SCI and performance (Whipple & Frankel, 2000; Osborn & Marion, 2009; Shub & Stonebraker, 2009). From the findings, the IMLA, clearly substantiate that the leadership strategies contribute to both the achievement of SCI, being the core theme of this study and also to performance, as referred in Figure 6.1, with the model outcome, being made up from a comprehensive set of eight competitive capabilities, where one of which is effective and efficient performance. Such research findings extend all the cited scholarly work theories, referred above, meaning that leadership strategies contribute to SCI and performance.

7.4.1.6 IMLA's collaborative initiatives: cooperative/non-coercive and aggressive/coercive strategies

The management and leadership approach across the SC is very challenging, since all SC actors are independent entities and not as a single organization (Lambert et al., 1998; Sydow et al., 2011). The priority assigned to establish which SC actors are to be managed effectively, is guided by the stakeholder salience within the stakeholder theory, meaning that the SC actor is treated according to
the **power type**, the **established requirements** and the **urgency of the request**, to meet both static and dynamic commitments (Mitchell et al., 1997). The power concept within the leadership literature is considered as a key source of control by influencing the people's behaviour (French & Raven, 1968, Mullins, 2005). Co and Barro (2008) builds on Mitchell et al.' s (1997) attributes and typologies of the stakeholder theory in supply chain collaboration, to establish whether the focal firm should deploy a **cooperative** or **aggressive** collaborative strategy with the other stakeholder (Section 2.6.3.3.2). The collaborative strategy selection depends on above three criteria so as to decide for most fruitful outcome and at the same time build or sustain the established relationship without failure. Yeung et al. (2009) supports such a theory by labelling the two collaborative strategies as **non-coercive** and **coercive power** approaches.

From the findings, the emerged theory is being refined by the terms used within the literature, meaning that to achieve SCI the collaborative initiatives within the IMLA need to adopt a cooperative/non-coercive approach, which is away from any forced imposition of an aggressive/coercive approach (Co & Barro, 2008; Yeung et al., 2009). From the findings, the latter approach needs to be adopted only as a final resort, after the cooperative approach fails to achieve the expected results, not to disrupt the relationship. The research findings extends Co and Barro' s (2008) lack of trust dimension as a key condition associated with the adoption of an aggressive approach, by including the final resort condition which serves as an added decision criteria. Such a finding adds to Co and Barro's (2008) aggressive strategy in supply chain collaboration, based on the stakeholder theory. This option was explained by a participant who stated:

'Even when problems exist, I try to coordinate in a positive approach not to demolish relationship, like a give and take situation' (3/INT4).

Furthermore, from the leadership literature, within the SC context, as referred in Section 2.6.3.3.5, the nature of the cooperative strategy is function of the SC actors' power symmetry to asymmetry continuum (Cox et al., 2003; Hingley, 2005; Chicksand, 2015), since such a power difference between the dyad, determines the level of trust. The trust decreases inversely proportionally with such power difference between SC actors (Huxham & Beech, 2008). From the findings, the focal firm needs to establish dedicated collaborative initiatives with all of its SC stakeholders. Furthermore, the findings outline that such relationships vary from an arm's length to a close collaborative approach, depending on the nature of the product and the role assigned to each SC actor, in line with the contemporary literature except that the literature uses the terms non-adversarial and adversarial relationships to demonstrate the continuum of such a relationship variation (Cox et al., 2003; Hingley 2005; Chicksand, 2015). In this case, the literature refines the emerged theory on IMLA based on an arm's

length to a close collaborative approach, in the collaborative relationships with SC actors, since the relationship may take the form of either a non-adversarial or an adversarial relationship.

The findings also complements Cox et al.'s (2003) framework, referred in Figure 2.3, since there exist cases from the data, when suppliers are power dominant (i.e. supplier dominance) and the focal firm needs to accept de facto the assigned **contractual terms and conditions**, and there are also other cases when suppliers engage in a deep commitment to work together with a power sharing approach and even invest their resources within the buyer premises. Similarly, from the data, the focal firm may have relatively higher power than its stakeholders (i.e. buyer dominance), and determines the contractual terms and conditions, which the supplier needs to accept. Furthermore, from the data, there also exist other cases where the focal firm engages with a power sharing and with an interdependent approach and invest within either the supplier or customer premises and vice versa. Such an emerged theory also corroborates Chicksand's (2009) buyer-supplier power matrix, as referred in Figure 2.2.

Within the literature, the cooperative strategy is more appropriate: for change management; to demonstrate the interdependence; and as a sign of contribution of benefits to each other (Co & Barro, 2008). The emerged theory extends the outcome of a cooperative strategy within the literature, by underscoring the scope of such a cooperative strategy by two other objectives: to undertake a continuous improvement (Section 6.3.1) and to meet the everyday challenges, being both IMLA properties through, SC flexibility (Section 6.3.4), since the change management objective is in line with the literature. Such a finding from the study extends the knowledge on the cooperative strategy approach referred by Co and Barro's (2008) paper and hence contributes to the SCM literature.

7.4.1.7 The IMLA's human element from a social sustainable perspective within the SCM literature

From the SCM literature, it is currently considered pivotal that organisations need to undertake the necessary investments in all human resources for social sustainable practices, through training, performance evaluation and rewards (Jabbour et al., 2010; Wagner, 2013), since within the global business it is a must to have the right people with the appropriate skills and talent (Rothwell, 2002; Sarkis et al., 2010; Tarique & Schuler, 2010; Xu & Thomas, 2011). Such an approach is taken since the role of people with the right human resource management is crucial to the success of the SCM approach (Ashby et al., 2012; Ellinger & Ellinger, 2014). Management also needs to safeguard the employees' health and safety and their equal treatment (Leire & Mont, 2010). Though, it cannot be excluded the key role of the human element in all problem solving and critical thinking abilities, since humans are able to process information with an amazing efficiency and often perform better than the

most highly sophisticated machine (Kuhn, 1999; Halpern, 2003). Within the SC context, the human resource strategies such as the role of the human resources activities (e.g. training) and organisation variables (e.g. structure), through the transaction-based (e.g. efficient processes) and emerging relationship-based (e.g. employee loyalty and commitment) respectively, were all proposed theoretically to contribute to SCI and performance (Shub & Stonebraker, 2009). The human element knowledge also enables SCI through the people's capabilities to understand all the changes needed to adapt, in a flexible way, to all requirements and to meet all competitive performance measures (Gowen & Tallon, 2003; Gunasekaran et al., 2008; Cheung et al., 2012; Ellinger & Ellinger, 2014).

From the findings, the basic attributes and values of the IMLA, as referred in Section 7.4.1.3 above, consists from the four leadership traits. These leadership styles are mainly focused on the human element, where managers and leaders exercise their roles to meet the dynamic and challenging business objectives. The leadership styles need to employ the necessary management and leadership skills based on a substantial level of charisma, empathy and ethical behaviour, to win people's commitment, motivation and trust. Such a stance is needed to serve as a spring-board in all synergies between all staff at all management levels, both within the intra- and inter-organisational interactions. The IMLA takes roots in the human relationships gained over the years of experience and engagement, both within and outside the focal firm. The human element plays a pivotal role in the management of all operations, since it forms up the backbone of all the collaborative initiatives within the focal firm and SC actors' activities. Staff within the focal firm needs to be safeguarded against all hazards, trained and updated with all strategic changes and innovative developments, to enhance their capabilities and build on a trustful relationship so to promote engagement, involvement and commitment in all challenges. The human element collaborative initiatives, within the focal firm and with all SC stakeholders, consist from sharing of ideas, information and knowledge on the SC and manufacturing processes and in meeting all needed changes, due to in-built human flexibility and competence capabilities, even where the manufacturing processes incur a high level of automation (e.g. the human element role may still be crucial even if s/he only performs the interface part within one automated process to another). Such findings add to the conversation on the role of the human element and the leadership theories within the SCM and manufacturing literatures. The important role of the human element was explained by three participants who stated that:

'There exist processes and products that still need humans' intervention'; 'Machines do not cater for all activities' (4/INT5); '... the manual approach can adjust for variations [better than machines]' (6/INT8); 'If you miss the knowledge from people, you may miss the product success'; 'We work as a family' (8TS/INT4).

Furthermore, the findings indicate that the human element performance depends on the way people are treated, appreciated and respected by each other in an honest approach, independent from the

management level and/or position within the firm and also in the way management invests in the human resources self-development through training. Management and leaders may lack the human appreciation of the employees' contribution, since there may exist a management and technical divide. Such an outline of the emerged theory, associated with the human element, also corroborates the referred literature outline associated with the human element (Section 6.3.1), which adds to the conversation on the pivotal importance of people within the SC context. The human element performance is function of the management respect as highlighted by two participants who stated that:

'We believe in respect to all and between all' (8TS/INT4); and 'Human resources is not so dignifying, since they are not resources but people as individuals'; 'People respect you as you respect back' (9TS/INT6).

The role of the people was also underscored within the manufacturing strategy literature as one of the four infrastructure decision areas that contributes to the operations strategy (Hayes & Wheelwright, 1984; Hayes et al., 2005). Furthermore, from the organisational literature, transformational leadership drives people to see the deeper purpose in their work, treats them as unique and valuable human beings, and instils in them a belief in what the leader is trying to achieve (Ardichvili & Manderscheid, 2008). Such a view is also substantiated by Bass and Avolio (2000) where people's needs are considered before the leader's needs with a deep belief in moral and ethical concern and with a motivational climate to adopt an innovative approach in the way of doing things.

From the referred human element findings outlined above, through the rich insight and detailed account, as referred in Sections 6.3.1 up to 6.3.6, the findings contribute to HRM discipline within the SCM, organisational and manufacturing literatures respectively. The emerged theory focuses on the role of the human resources activities of both the intra- and inter-organisational relationships, which are pivotal for the success of the manufacturing and SC operations, since the literature on the social element or the human element, regarding the labour, skills and the forming of relationships, are a neglected area of research (McCarter et al., 2005; Shub & Stonebraker 2009; Ashby et al., 2012; Lengnick-Hall et al. 2013). One participant explained the pivotal importance of the role of the human element within manufacturing, who stated that:

'Mistakes occur, when equipment and software are excellent, but they remove the human element from it ...' (4/INT5).

Furthermore, the emerged theory, as refined by the literature, outlines that such a relationship approach between people, must not be based on a coercive or aggressive approach, but must be based on a unity and an understanding culture based on a cooperative approach, as referred in Section 7.4.1.6 above. As a result, a cooperative approach between the SC actors, from the emerged theory, promotes a sense of community with a dedicated concern for all mutual relationships and creates a motivated

workforce based on cross-cultural trust, that enable a holistic SCI management approach of the overall SC and manufacturing operations, with the contribution of all the SC stakeholders for a win-win outcome. It is to be noted that circumstances may also exist, when anyone of the SC actors may end up to a win-lose situation in any particular transaction, but still retains the position and the obligations within the SC, not to lose the trustful relationship built over time between people and looks forward for the future win-win outcomes. Such an emerged theory supports the literature since the SC interfirm relationships may either lead to a win-win or win-lose outcome or a level of imbalance of distributed benefits (Cox, 2004d; Hingley, 2005; Claycomb & Frankwick, 2010; Terpend & Ashenbaum, 2012).

7.4.2 Business strategy and the IMLA within the supply chain context

From the literature, due to the nature of the study, being focused on the SME manufacturing sector, the terms corporate and business strategies were assumed to be common terms (Section 2.6.4.1), due to the relative small size of the firms within the research (Thompson et al., 2005). The business strategic management approach consists from three management levels that covers the strategic, tactical and operational levels, and needs to unify all functional areas together with all SC members with their capabilities and limitations (Skinner, 1969; Stonebraker & Afifi, 2004; Swink et al., 2007).

From the findings, as already defined in Sections 6.2 and 7.3.1, the IMLA serves as a driver for the focal firm strategic, tactical and operational management levels, so as to formulate the business strategy, even by considering the SC actors requirements, once a shared leadership approach is adopted based on a strategic leadership style. Such findings complement the literature focused on the strategic management literature and also contribute to the leadership theory within the business strategy and SCM literatures.

From the literature, the operations strategy of the focal firm refers to the firm's business strategy implementation, with a continuous improvement approach of all operations and with an integrated SCM approach, to achieve a sustainable competitive advantage, by focussing on the entire SC (Narasimhan & Carter, 1998; Chen & Paulraj, 2004; Lummus et al., 2008).

The findings complement the operations strategy literature, since as already referred in Section 7.3.1 and 7.4.1.3, the IMLA, especially through the transformational leadership style, adopts a continuous improvement approach based on change and innovative initiatives, to transform teams (i.e. intellectual stimulation) and not only focuses on formal structures and systems, so as to promote flexibility and accepts a controlled level of informality to adapt to today's business dynamic business climate to

promote competitive capabilities. Such findings also add to the operations strategy conversation within the SCM context.

From the literature, the strategic management approach is based on the inter-relationship between the strategic analysis (i.e. understands its environment and its strategic capabilities), strategic choice (i.e. gather all relevant information), strategic fit (i.e. to decide effectively on the best option) and strategy implementation which determines the focal firm competitive advantage (Johnson & Scholes, 1999) (Section 2.6.4.1). The scope of the SC to meet its competitive capabilities is needed since those SCs that lag behind and fail to deliver the value will lose in the competitive market (Cox, 1999).

From the findings, the business strategy in-conjunction with the manufacturing and SC strategies respectively determine the strategic fit of the overall SC, as referred in Sections 7.4.3 and 7.4.4, after undertaking the strategic analysis and choice. With the overall three strategies in place, with the holistic SCI management approach, promotes the overall competitive capabilities. Such findings complement the literature focused on the strategic management (Johnson & Scholes, 1999).

7.4.3 Comparison of the manufacturing strategy emerged theory within the extant literature

7.4.3.1 High value manufacturing and sustainability within SMEs

From the extant literature it was established that for developed countries, with high cost economies, such as Malta, SMEs within manufacturing are adopting HVM practices, based on the triple bottom line sustainable practices in all processes, both within manufacturing and also across all SC operations, so as to achieve competitive advantage (Martinez et al., 2008; Technology Strategy Board, 2012). The SC entities forming up the SC need to be integrated as a single entity (Chan et al., 2003; Sanders, 2007) and wherever possible focus on virtual integration rather than vertical integration to optimize the use of all core and distributed resources to form a single holistic SC (Prahalad et al., 1994; Power 2005; Wang & Wei, 2007; Gunasekaran et al., 2008). The term sustainability today integrates social, environmental and economic responsibilities (Elkington, 2002; Bojarski et al., 2009; Buyukozkan & Berkol, 2011; Gimenez etal., 2012; Lee et al., 2012; Caniels et al., 2013; Glover et al., 2014).

From the findings, the current approach being adopted by SMEs, within the Maltese manufacturing, is to ensure that all SC actors forming up the SC build on each other' value added operations with high quality (e.g. ISO) and professional engineering (e.g. ISA) standards respectively and with the right provisions for sustainable practices, so as to promote a competitive approach based on holistic value-proposition to the customer. The manufacturing operations trends, in line with the literature, are to focus on virtual integration rather than vertical integration, with value-added manufacturing to gain from outsourcing manufacturing efficiencies based on integrated initiatives and make use advanced

manufacturing techniques in line with HVM, so as to operate as a one holistic SC. Contextual circumstances for one firm, within the data, favoured the in-house option of a particular manufacturing process and had to vertically integrate locally that upstream side process, to improve on the overall performance. This emerged theory corroborates with the manufacturing literature and adds to the contemporary conversation on manufacturing strategy and SCM literatures and also emphasizes on the importance of HVM.

7.4.3.2 Manufacturing Strategy and Supply Chain Strategy for the SC competitiveness

From the extant literature, the manufacturing strategy determines the type of SC strategy to be adopted with all SC actors. The manufacturing strategy needs to decide whether a manufacturing process is kept in-house or is preferably outsourced (i.e. make or buy) to sustain its competitive advantage (Harrigan, 1985; Klein, 1988; Bajec & Jakomin, 2010). This strategic decision may either be based on the core competency of the focal firm (Cox, 1999), where the core activities are performed in-house and peripheral activities are outsourced, in line with the resource-based view (Barney, 1991), so as to exploit the resources of the focal firm and of its suppliers to improve on the overall competitiveness (Cox, 1999; Baloh et al., 2008). The other option may be based from a cost efficient rationale by the focal firm, in line with the transaction cost analysis (Williamson, 1985), to exploit the cost of operations of the focal firm and of its suppliers to improve on the overall competitiveness (Williams, 1985; McIvor, 2010). In reality this decision is multifaceted since it is function of a complex set of other contextual conditions, such as suppliers' availability, focal firm's products portfolio and/or capacity of production, sector type and proximity of suppliers.

The findings complements the literature by outlining that the manufacturing strategy is function of the make or buy decision, with their relative advantages, as referred in Section 6.3.3. Such a decision has a direct effect on the SC strategy approach to be adopted, since it establishes whether to go for vertical integration (i.e. make) or virtual integration (i.e. buy), as referred in Section 7.4.4.1. The data outlines that such a decision is very complex and has a quite comprehensive laundry list which is also based on several contextual conditions of each firm to determine the final decision, in particular, whether the focal firm shall make or buy a specific part of the manufacturing operations strategy. Such findings not only complement the literature but adds to the conversation, since it was established various conditions that determine the make or buy manufacturing decision, together with the SC strategic options, based on either a vertical integrated or a virtual integrated approach respectively with the engaged SC actors, to determine all the needed actions to achieve the SC competitive capabilities.

7.4.3.3 The IMLA and the manufacturing strategy for a competitive strategy

From the extant literature, the value proposition for a competitive strategy is mainly based on a set of priorities and trade-offs based on a top-down management approach with right manufacturing policies which are linked to the competitive and strategic facts in line with Skinner (1969)'s focus on the importance of the manufacturing strategy to competitiveness and Hayes and Wheelwright's (1984) best practices (Section 2.6.5). Skinner (1969) also focused on the importance of the manufacturing leadership practices to pursue the operations strategy, which were later followed by the two manufacturing strategy categories, dedicated to the structural decisions and infrastructure decisions areas (Hayes & Wheelwright, 1984; Hayes et al., 2005). Furthermore, management commitment is dedicated to streamline and align the manufacturing strategy in line with McKinsey 7-S model seven elements, namely strategy, structure and systems (hard elements) and the shared values, skills, style and staff (soft elements) (Waterman et al., 1980). The key management focus, with reference to the 7-S model, is the people shared values to promote unity and teamwork, where the human element is given a pivotal role within the strategic implementation process, as clearly established in the literature, that people are a key and invaluable asset (Rothwell, 2002; Tarique & Schuler, 2010; Xu & Thomas, 2011).

From the data, as referred in Sections 6.3.1 up to 6.3.6 and Section 7.4.1.7 above, the IMLA, gives crucial importance to the effective management of people, which is synonymous to the shared value element referred in the 7-S model. This is further emphasized in the findings with the deployment of the shared leadership approach both within the focal firm and across the SC. Such an emerged theory can be clearly seen, both from a manufacturing perspective, in the processes coordination initiatives between various people from different functional areas and also from a SCI perspective, by both intraand inter-organisational synergies and work practices. Such an emerged theory adds to the contemporary knowledge on the pivotal role of manufacturing leadership, as referred by Skinner's (1969) and Hayes & Wheelwright's 1984 scholarly works, and at the same time addresses the gaps of leadership practices within the context of manufacturing literature (Liden & Antonakis, 2009).

7.4.3.4 The manufacturing strategy and the operational capabilities

From the extant literature, the manufacturing strategy is directly linked to the SC integrated activities of both upstream and downstream to link all stakeholders with the focal firm (Ragatz et al., 1997; Flynn et al., 1999; Frohlich & Westbrook 2001; Paulraj et al., 2008; Flynn et al., 2010) to meet the key operational objectives of cost (i.e. quality and features), dependability (i.e. on specifications, time and service), quality (i.e. process quality) and flexibility (i.e. product and volume) (Flynn et al., 1999). Furthermore, it was referred in both the a priori and post literature reviews, that in today's industry, the manufacturing is based on lean and agile manufacturing with a broad focus on competitive

operational capabilities with special attention on responsiveness and flexibility (Yusuf et al., 2002; Gunasekaran et al., 2008; MacBryde et al., 2013) and innovation (Binder et al., 2007).

From the emerged theory, the empirical based theoretical model, referred in Figure 6.1, represents the operational objectives with two key competitive capabilities outcomes, namely effective and efficient performance and SC flexibility. Furthermore, the literature supports this study theoretical model by giving a comprehensive outline of all the referred operational objectives, as indicated in the literature above, since it is pivotal for the manufacturing process to meet the low cost, right delivery time, high quality and flexibility objectives to achieve the overall SC competitive capabilities. Such a literature is used to refine the emerged theory outline on theoretical model competitive capabilities outcomes, vis a vis the operational performance outcome, as referred in Section 7.4.7, with refined SCI theoretical model in Figure 7.2.

7.4.3.5 The servitization of manufacturing

From the literature, the competitive manufacturing scenario need to be seen within the context of European Technology Platform (ETP), through the 'Manufuture' report and the Community Research and Development Information Service (CORDIS), under the EU funding programmes for research projects, through the 'Horizon 2020 framework programme for research and innovation' such as 'Manutelligence', where all contextual manufacturing effort is being focused on a product-service emerging trend based on servitization of manufacturing (Lightfoot et al., 2013) or integrated solutions of manufacturing (Davies et al., 2006). From the emerged theory, the product-service emerging trend is supported by the data and it is to be noted that the product type determines the nature of the relationship with the customer, since the sectors which included higher value-added products, the more are the dedicated customer service measures, which result in a more in-depth servitization approach. Such a statement extends the work focused on the servitization of manufacturing within the extant literature. The emerged theory (Figure 6.1) has the effective customer service as one of its competitive capabilities.

7.4.4. Comparison of the SC strategy emerged theory within the extant literature

7.4.4.1 The IMLA within the SC Strategy with its virtual enterprise approach: Vertical or Virtual Integration

From the literature, the SC strategy is concerned with the integration and coordination decisions within the focal firm in-conjunction with the set-up to be implemented with all SC stakeholders (Gunasekaran et al., 2008; Green et al., 2014) to achieve competitive capabilities (Kim, 2009). The SC strategy also needs to adapt and structure its resources to effectively align to the changing

competitive conditions (Thompson, 1967; Hofer, 1975; Kim, 2009). Such SC foundations due to the variation in conditions, implies that a contingent model needs to be adopted to fit each business situation (Stonebraker & Afifi, 2004). The new emerging manufacturing paradigm, is being referred to as high value manufacturing (HVM) based on the virtual enterprise (VE) concept (Shi & Gregory, 2005; Livesay, 2006; Martinez et al., 2008). This management paradigm shift incorporates a change from vertical integration to virtual integration (Prahalad et al., 1994; Power, 2005; Gunasekaran et al., 2008), based on unity of efforts, to optimize the use of all core and distributed resources to form a single holistic SC.

The findings extend the contemporary conversation on SC strategy since it serves as a platform for the manufacturing strategy so that through a holistic SCI management approach, achieves the overall SC competitive capabilities. The findings also outline that the SC strategy needs to be established inconjunction with the manufacturing strategy and in view of today's competitive scenario, the SC is normally based on virtual integration rather than vertical integration, in line with the literature from both the SC strategy perspective, as already referred above, and from the manufacturing strategy perspective, as referred in Section 7.4.3.1. Though, cases still exist from the evidence within the data, where vertical integration needs to be incurred so as to overcome the problems associated with particular SC actors, as referred in Section 7.4.3.1. As a result such an emerged theory gives a more complete picture of the SC strategy to that referred in the literature. Such different situations, as indicated by the data, imply that the SC strategy is not a one-size-fits-all approach, whereby such a statement corroborates with the SC strategy literature on the need to take SC contingency measures (Stonebraker & Afifi, 2004) such as vertical integration, to optimise its operations. Furthermore, the findings sustain the literature both on the contingent approach, through the situational leadership style, and also on the HVM approach, with its virtual enterprise concept, through the strategic leadership style by involving all SC actors, based on the IMLA. Such findings sustain the applicability of the IMLA to meet different SC strategic scenarios within the SCM literature and also add to the leadership theory within the SCM literature.

From the literature, the focal firm needs to employ a cost/benefit analysis to establish the feasibility of the way forward, since SCI is risky and may not be appropriate in all situations (Stonebraker & Afifi, 2004). The SC strategy needs to decide on how to gain from the optimum performance from each SC members so as to eliminate the risks associated with sub-optimisation effect that may result from friction between SC actors optimal (Cox et al., 2004a; Narayandas & Rangan, 2004). The way forward is not to blindly employ an integrated approach with all stakeholders, but to understand the power struggle over the appropriation of value between all SC actors at every point of the SC with their relevant contextual conditions and with due consideration of all competitors in the market and

then decide the type of relationship to be deployed with each SC member (Cox, 1999). The pivotal role of employing effective relationships has also been supported by key management gurus, such as Peter Drucker (1999), who advocated that business growth and expansion have to be based on alliances, partnerships, and all kind of relationships.

The findings imply that the SC strategy apart from establishing the type of relationships (Section 7.4.1.4), also needs to set-up the necessary terms and references of all contractual commitments to determine the cost/benefit of each engagement through the contractual terms and conditions with all SC actors, as already referred in Section 7.4.1.6. Under both the vertical integration and especially in the virtual integration approaches, due to the difficulties to manage the external SC actors, the focal firm SC strategy, with the holistic SCI management approach (Section 7.4.5.2), needs to employ all the necessary measures, to align and optimise the overall SC performance. Such findings complements the literature focused on SCM and SCI, since the focal firm needs to unify all SC stakeholders' individual objectives to be balanced effectively to achieve optimised overall operations, through either vertical or virtual integration, since the effectiveness of the overall SC strategy depends on the success of the individual strategies of all SC actors.

7.4.4.2 SC strategy design to fit different manufacturing sectors

From the literature, the SC strategy is a complex concept in the SCM literature (Rose et al., 2012) and consists of both a strategic and operational capabilities and with their alignment (Cox, 1999; Kim, 2009) function of the contextual conditions (Cox, 1998). Due consideration needs to be given to the type of product (i.e. functional or innovative type) to achieve a strategic fit by matching the appropriate SC design based on either a manufacturing approach with efficient processes or a marketing approach with responsive processes (Fisher, 1997), as referred in Section 2.6.6.2. From the findings, the firms covered both functional and innovative types of products, though from the data, a certain level of innovation was also being employed in the functional products type, such as: deploy special materials with better characteristics (e.g. disposal medical accessories improved features); employ different ingredients (e.g. meeting different customers' taste in food manufacturing); and implement improvements in appearance (e.g. printing of books with different types of edge colouring in the paper). Such an approach dedicated to the functional product types contributes for an innovative final product with an improved competitive edge in the market. Such findings challenges the literature, by adding to the conversation on Fisher's (1997) matching criteria for performance improvement, meaning that success is also achieved by employing an innovative approach even in functional products.

7.4.4.3 The sustainable SC strategy

The literature outlines that the SC strategy of the focal firm needs to fit its environmental sustainability requirements within its business strategy and also across the whole SC, including the selected suppliers and logistics providers (Vachon & Klassen, 2006; Hsu et al., 2013; Caniels et al., 2013). It is to be noted that the term SC sustainability today integrates social, environmental and economic responsibilities (Elkington, 2002; Bojarski et al., 2009; Buyukozkan & Berkol, 2011; Gimenez et al., 2012; Lee et al., 2012; Caniels et al., 2013; Glover et al., 2014; Hsueh, 2015). The findings, from the SC strategy perspective, show that sustainability is pivotal in all strategic decisions, both within the focal firm and across the SC with all stakeholders. The data outline that SC strategy serves as a driver for the holistic SCI management approach which needs to have in place all sustainable measures from the triple bottom line perspective. Furthermore, the data show that the sustainability measures are more demanding in certain sectors, since tougher standards exist with respect to both quality and/or legal obligations, such as in the automotive (e.g. TS16949 as standards) and pharmaceutical (e.g. GMP as standards and legal requirements). Such findings add to the SC strategy sustainability and manufacturing conversation, by outlining the need for a sustainable holistic SCI management approach to achieve competitiveness and by describing the demanding requirements of specific manufacturing sectors.

7.4.4.4 The SC Strategy enabled by technology deployment for performance improvements

From the literature, the SC strategy is concerned about upstream (Carter et al. 2000; Scannell, Vickery, and Dröge, 2000; Kim, 2009; Swink et al., 2007) and downstream (Ellinger et al., 2000; Kim, 2009; Swink et al., 2007) relationships to give a competitive edge with performance improvement. The SC strategy to achieve integration, both within the focal firm and throughout all the SC tiers (e.g. logistics and supplies), needs to be backed by an IT infrastructure, to achieve both manufacturing and performance improvements (Carr & Smeltzer, 1999; Wisner, 2003; Gonzalez-Benito, 2007) and competitive success (Eisenhardt & Tabrizi, 1994; Birou et al. 1998; Kim, 2009). The SC capabilities may result in distinctive competitive capabilities within both the focal firm, in line with the resource-based view (Wernerfelt, 1984, Barney, 1991) and also from a wider approach to operations, where the focal firm deploys all resources outside its boundaries with its SC actors, in line with the extended resource-based view of a network of firms (Matthews, 2003). Furthermore, it is well established by the literature, adding to the above assertion, that the role of IT promotes performance improvement, that IT also enables all collaborative processes, through: connectivity of all SC actors; it captures all real time information; it adopts the necessary information management techniques to promote SC visibility; it provides value added services; and it achieves a holistic management approach of the SC

(McAfee, 2002; Chesbrough & Teece, 2002; Koudal & Wellener, 2003; Fawcett et al., 2007; Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014).

The findings complement the SC strategy literature, since the data established that SCI is achieved by employing an IMLA, together with the SC strategy in-conjunction with the business and manufacturing strategies, through the holistic SCI management approach, enabled by technology, to achieve effective and efficient performance. The findings also add to the conversation within the literature focused on both upstream vis a vis a core and non-core product types and downstream vis a vis a high-value added and a low-value added product types, since different collaborative initiatives are dedicated to meet such different conditions at both ends of the SC, as referred in Section 7.4.1.4.

Furthermore, the findings also complement the literature on the deployment of IT, as referred above, since the data reveals that the holistic SCI management approach is enabled by technology deployment in both information systems and automation, to promote effective and efficient performance. The emerged theory infers that the SC strategy with its challenging targets needs to establish the necessary visibility of all operations, through information sharing mechanisms adopted by all SC actors, to have all the needed information, to remedy all problems and adapt to all changes so as to promote SC flexibility and responsiveness. Such a technology deployment enables all processes to promote a high level of integration of all human interactions with streamlined workflows across the SC (e.g. between different functional areas of the focal firm and outside) and of all manufacturing processes (e.g. automation) to improve on the business performance. This emerged theory outline adds to the body of knowledge on information systems and in automation within manufacturing and SC context.

7.4.4.5 The SC strategy long term effects

From the extant literature, the overall SC is using a pull-based system (Kros et al., 2006), to align the production and business processes throughout the supply chain to the changing demands of the chain's ultimate customers (Cox et al., 2003; Green & Inman, 2005; Green et al., 2007; Lambert, 2008). From the findings it is established, in line with the literature, that the SC strategy is a pull-SC strategic approach, which places the customer at the heart of strategic significance. But, not located within the literature, is that the SC strategy through **SCI promotes growth** and from the data it was also established that the focal firm may need to increase the SC ownership span (i.e. vertical integration) to meet improved performance. Furthermore, it is also inferred, from the emerged theory, that the SC strategy is a **long term strategic** approach, since most of decisions and collaborative measures are not quick fix solutions, but involves long-term commitments and investments. In fact from the literature, it was inferred that the focal firm prefers to secure a long-term relationships with a few dedicated

suppliers to create an improved value and yield lower transaction and production costs (Bayraktar et al., 2009) but no direct reference was found that such an approach is part of the SC strategy formulation except for the collaborative enterprise management approach (Clegg & Binder, 2007, 2008; Clegg et al., 2012; Clegg and Wan, 2013), as referred in Section 2.5.2. As a result, the findings add to the SC strategy literature by inferring that the SC strategy formulation is a long term strategic approach and that through SCI it promotes growth.

7.4.5 Holistic SCI management approach

7.4.5.1 The SC strategy from a holistic SCI management approach enabled by technology

The literature broadly defines supply chain integration as the extent to which a firm is strategically interconnected and aligned with its supply chain partners (Das et al., 2006; Jayaram et al., 2010). This means that SCI is a strategic approach which is needed to achieve the running of the focal firm through the integrated operations within all of its functional areas and also with all its stakeholders' network. As a result supply chain integration approach, both within the focal firm and across the SC, promotes performance improvements (Braunscheidel & Suresh, 2009; Flynn et al., 2010; Koufteros et al., 2007; Swink et al., 2007; Wong et al., 2011; Zhao et al., 2011). The broad integration set-up of customers and suppliers to form a holistic supply chain management approach has been considered of pivotal importance for the formulation SC strategies that are optimal for the whole SC (Wilding, 1998; Lambert & Cooper, 2000; Perona & Miragliotta 2004; Power, 2005; Lambert, 2008). This holistic approach across the SC could be made possible with the advent of the internet based technologies, since through the web, one is able to provide an integrated platform for customers and suppliers for all activities across the SC, to take the most appropriate decisions by considering all constraints (Frohlich & Westbrook, 2002; Bayraktar et al., 2009). Furthermore, IT applications (e.g. ERPs and databases) provide the ability to capture all real time information and adopts the necessary information management techniques to promote SC visibility, enable value added services and to achieve a holistic management approach of the SC (McAfee, 2002; Chesbrough & Teece, 2002; Koudal & Wellener, 2003; Fawcett et al., 2007; Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014).

This holistic approach is in line with the principle that the new SCI paradigm is designated to consolidate all the SC members' resources and capabilities, for the benefit of the whole SC to achieve performance improvement, since the performance of all SC actors influence each other's performance (Huo, 2012). Furthermore, Fawcett et al. (2008, p.36) added that all potential SC 'barriers' need to be seen in 'combination or holistically', so as 'to separate the trees for the forest'. Though, it cannot be excluded that SCI needs a level of commitment and investment from all SC actors, but as Schoen and Swink (2012, p.100) advocated that: 'In general, however, researchers have supposed that the benefits

derived from integration activities outweigh their associated costs, leading to overall greater levels of operational performance'.

The findings clearly sustain that SCI starts from the focal firm and extends throughout with all stakeholders, to set-up a unified SC, with dedicated collaborative initiatives depending on the role of each SC member. The data also indicates that the scope of the SCI is to achieve a win-win outcome for all SC actors and this cannot be achieved unless the focal firm is in a position to consider the big picture, in particular, from a holistic approach. The holistic SCI management approach, through integrated operations, both within the focal firm and across the SC, promote improved synergies with all SC actors, which altogether contribute to the SC competitive capabilities. From the data SCI does not only provide benefits but may bring about negative issues, such as greater inflexibility (i.e. needs switching costs to change systems due to established processes with other SC actors), decreased innovation (i.e. locking to particular approaches with suppliers), costly investments to integrate with partners (e.g. invest in common technologies to promote further synergies with suppliers). Furthermore, the findings also outline that the holistic SCI management approach, which is enabled by technology, facilitate all processes but does not replace the human element, both within the manufacturing and the SC processes respectively. Such findings complement the literature on the significance to apply a holistic approach within the SC context and also adds to the SCI literature on the several challenges that the focal firm may need to address to remain competitive and also adds to the conversation on the deployment of IT and the human element role within the SCM literature to achieve a holistic approach.

7.4.5.2 Holistic SCI management within manufacturing

A common term to holistic SCM is the Demand Chain Management (DCM) which is used to describe how the SC actors shall manage and coordinate all the links from the customer backwards to the suppliers, based on an extensive integrated approach to have an adequate understanding of the demand (Vollmann et al., 2000; Frohlich & Westbrook, 2002; Juttner et al., 2007). Such a holistic significance has been found useful to complete all tasks quickly to reap a greater profit based on a triple-A SC, meaning that the SC processes need to be aligned, adaptable and agile (Lee, 2004; Ishaq et al., 2012) in line with the theory of swift, even flow, where the manufacturing processes variations are narrowed to promote a more streamlined flow (Schmenner & Swink, 1998). Gunasekaran et al. (2008, p.550) support this view and advocate that agile manufacturing 'should not only be based on responsiveness and flexibility, but also on the cost and quality of goods and services'.

A common saying used nowadays is 'time is money'. As a result, the findings, clearly outline, that all SC operations are employing lean operations to streamline all manufacturing and SC processes and to

minimise all waste in all types of resources, and also to optimise, align and improve all operations from a holistic perspective, in line with the literature. The holistic SCI management approach through the IMLA is adapted to fit and manage effectively each SC actor with its particular contextual conditions. The IMLA achieves this through the holistic SCI management approach with its sixteen conceptual actions, which are used to meet the expected competitive capabilities, such as SC flexibility with an effective customer service, among others, as referred in Figure 6.1. Such sixteen measures include a number of pivotal requirements to be implemented to achieve this holistic process approach, in particular, by the measurement of all operations across the SC together with the compliance to all quality based processes and with the deployment of all sustainable measures, in both manufacturing and SC operations, as already referred in Sections 6.3.6 and 7.3.1. Such findings sustain the arguments referred by all cited scholarly works on SCI with its holistic significance by outlining all measures that constitute such a holistic approach by addressing the lack of literature on holistic measures. Furthermore, it also adds to the leadership theory within SCI and the importance of streamlined activities within the manufacturing literature.

7.4.6 Technology deployment in information systems and automation

From the extant literature, the role of technology is pivotal in enabling all information based processes (Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014) and processes within production automation and across the SC, such as supply management and logistics to mention a few (Gunasekaran et al., 2008; Jin et al., 2014). Being in this information age, the businesses within the SC are highly dependent on an effective information processing approach to develop and manage knowledge across the SC through information technology applications (Dyer & Singh, 1998; Ofek & Sarvary, 2001; Schoenherr & Swink, 2012) and with the deployment of technology to eliminate waste in several processes, by making them more efficient (Cox, 1999; Gunasekaran, 2004; Sanders, 2008). The importance of IT adoption is without doubt very beneficial for SMEs survival (Gagnon & Toulouse, 1996), since it is well cited that IT capability leads to superior performance (Determirhan et al., 2007). In fact, information technology has been recognised in the last two decades that it is a source competitive advantage due to provision of systems that assist organisational processes for developing and deploying relational capabilities (Bharadwaj, 2000; Hult et al., 2002; Paulraj et al., 2008; Sanders, 2008; Gunasekaran et al., 2008; Clegg & Wan, 2013).

Furthermore, from the literature, in certain processes, the human element interplay is changing since the 'role of people in the fulfilment process is changing progressively from processor to process monitor and enhancer' (Kagermann et al., 2011, p.104). The human approach role is further elaborated by the fact that there is a growing consensus for IS implementation success (e.g. ERPs), by adopting a teamwork approach, so as to integrate, collaborate and share their expertise on technology based platforms to capture information so as transform and exploit knowledge from each SC actor (Sarker & Lee, 2003; Faraj & Sambamurthy, 2006; Fawcett et al., 2007; Schoenherr & Swink, 2012; Clegg &Wan, 2013). Though, it cannot be excluded the key role of the human element in all problem solving and critical thinking abilities, as referred in Section 7.4.1.7. From the RBV of the firm (Wernerfelt, 1984; Barney, 1991) and the extended RBV (Matthews, 2003), it is being outlined that the SC flexibility with its competitive edge is being created not from the investment in technology, but how the technology is being used, through its dedicated information sharing mechanisms (e.g. ERP customisation of integrated processes), both within and across the SC, which are treating both the information management and the IT infrastructure capabilities as resources, which may be distinct from other SCs (Prajogo & Olhager, 2012; Schoenherr & Swink, 2012).

In fact from the data, the technology remains crucial, especially for particular manufacturing sectors that implement heavily based automated activities, since technology investment needs to be considered within the business strategy, by involving all key stakeholders, both within the focal firm and all the relevant key stakeholders. Nowadays, the technology investment is needed, to meet the current dynamic and competitive business environment, so as to ensure that all automated based processes efficiencies and throughput are not limited by the human element intervention, but are function of the equipment and machinery. Though, it cannot be excluded that the human element remains pivotal within the manufacturing (Section 7.4.1.7), since not all operations can be automated and furthermore there is always the need to exploit the human capabilities and competences in: reengineering; linking of processes; flexibility in certain activities; and detection of problems. Social sustainable measures are also being addressed, through technology based applications, for safeguarding the health and safety of the human element, such as in lifting and handling heavy loads, through robotics and for fumes protection, through ventilated environments and through technologybased training, for individual self-development. As a result the findings also outline that ICT facilitates the repetitive administrative work through the use of information management applications and also promotes knowledge sharing, such as with the use of databases or ERPs and also enables and improves, and in some cases even drives, the overall manufacturing and SC processes to promote effectiveness and efficiencies. The findings add to the conversation of technology deployment within the literature, with special focus on the role of the human element in view of technology based applications, which adds to the social sustainability literature within SCM and manufacturing (Section 7.4.1.7). The findings also contributes to knowledge within the manufacturing and SCM technology literature, since technology within academic work was either considered alone or with a relatively reduced set of constructs in several models and in some other cases it was also excluded from SCbased holistic models.

7.4.7 The competitive capabilities

Porter's seminal work within the strategy literature (Section 2.6.9) refer to Porter's (1980) competitive forces model and how activities contribute to a competitive strategy; Porter's (1985) value chain model and how firms assess the costs and the differentiation to create a competitive advantage; and Porter's (1985) Generic Competitive Strategies matrix guides between cost leadership or differentiation or focused strategy respectively, to achieve a sustainable competitive advantage. Such sources of competitive advantage can be attributed to various SC actors' activities, where with a SC holistic view, the set of activities will sum-up to an overall competitive advantage (Porter, 1996; Juttner et al., 2007).

From the findings, the data shows that most of the focal firms adopt a value-stream mapping approach to monitor all value-chain processes, both within the focal and across the SC to promote effectiveness and efficiencies. The firms understudy had different type of products and deployed different competitive strategies and as a result had adopted different value-propositions in line with Porter's strategy literature.

SCM is considered as a crucial operations strategy for organizational competitiveness in the twentyfirst century, with the direct involvement of people (Gunasekaran, et al., 2008). The contemporary context associated with SCM, is that today competition is not focused between firm to firm competitiveness but between entire supply chains (Dyer, 2000; Christopher & Holweg, 2011). This competitive edge is substantiated by the resource based view of the firm, where such distinct resources competitors cannot match or replicate due to the four criteria: value, rareness, imperfect imitability and non-substitutability (Wernerfelt 1984; Barney 1991). Within the SC context, the investment in technology within the organisation functional areas and outside the organisational boundaries to provide relevant, timely and accurate information is a source of competitive capabilities (Fawcett et al., 2007; Bayraktar et al., 2009) in line with the extended-RBV (Matthews, 2003). Within a SC, various stakeholders may need to link all activities in line with the value-chain approach (Porter, 1985), to promote both effectiveness and efficiencies and a value-added approach in all chain of activities, referred as vertical integration or vertical linkages (Swink et al., 2007; Guan & Rehme, 2012). On the other hand, the horizontal linkages of the value-chain activities are attributed to the integrated activities performed by the organisation to promote a competitive advantage (Porter, 1985). The integrated activities across the SC, need to consider the activities' values and costs of all the SC members' activities that contribute to the overall SC process, so as to promote a win-win mindset leading to competitiveness and profitability (Dyer 1994; Lambert et al., 1998; Rosenzweig et al., 2003;

Ketchen & Hult, 2007; Swafford et al., 2008; Bayraktar et al., 2009; Flynn et al., 2010; Zhao et al., 2013). It is to be noted that 2

The findings complement the SCM literature, since social sustainability, through the role of the human element, has been considered pivotal within the IMLA and in all processes. The deployment of technology, is also in line with the IT literature, since it has also been considered crucial in all information based and automated based processes, within the holistic SCI management approach, which is being enabled and sometimes even driven, by the IT deployment. Such a holistic process approach contributes to the overall SC effective and efficient performance with a continuous improvement approach to promote competitive capabilities. Such findings add to the conversation on SCI initiatives through technology deployment, both within the focal firm and with all SC actors.

The contemporary frequently cited SC key competitive operational capabilities for both the focal firm and with all upstream and downstream SC sides are: cost, flexibility, speed/delivery and quality (Hayes and Wheelwright, 1979b; Frohlich and Westbrook's 2001; Kim, 2009; Richey et al., 2010; White et al., 2010). Similarly, the financial or business competitive capabilities are sales, market share, and growth (Johnson, 1999; Flynn et al., 2010) and return on assets, return on sales and return on investments (Rosenzweig et al., 2003; Flynn et al., 2010). It is well cited in the literature that operational or manufacturing capabilities lead to business performance either in individually or in concert (Flynn et al., 1999; Rosenzweig et al., 2003; Swink et al., 2007).

From the findings, the competitive capabilities were considered as the outcome of the theoretical model. The emerged theory is hereby refined by the literature, since the theoretical model, as referred in Figure 7.1 above, must be extended beyond the competitive capabilities, as referred in Figure 7.2 below. Meaning that the competitive capabilities are all valid but are considered as intermediate outcomes, and that the actual outcomes are the operational (i.e. cost, flexibility, speed/delivery and quality) and the business (i.e. sales, market share, growth, return on assets, return on sales and return on investments) performance indicators respectively, as in line with the literature above and as shown in Figure 7.2.



Figure 7.2: The final SCI theoretical model (Author)

7.5 Summary of the discussion chapter

The multidisciplinary aspect of this research takes into account aspects of strategy, management, leadership, manufacturing, SCM, OM and technology in automation and in ICT applications, among other conceptual elements building up such disciplines. It is quite common that research within SCM is based on a multidisciplinary approach (e.g. Croom & Romano, 2000; Parente et al., 2008).

The research generated a substantive theory and derived a conceptual framework where the core theme to achieve a SCI management approach is the IMLA, which includes an organisational management set-up based on the focal firm and on all of its external SC stakeholders. The research stance had to appropriate such an interdisciplinary perspective, both directly and indirectly, with several intersection areas within different disciplines, to build a solid foundation of the key concepts that are involved in the design and management of an integrated SC set-up. It is to be noted that the research main focus of the post literature review was focused on supply chain management within manufacturing with special attention on all emerged themes from the data as the literature main keywords.

As a result the breadth and depth of the topic was very extensive and was a very challenging task to derive a synthesis of such a comprehensive and voluminous scholarly work and to finalise the research sense-making process to relate such a literature to the generated substantive theory with its conceptual framework. Henceforth, although the emerged core theory was based on both a management and leadership concepts, the research scope was not focused on the critique of the management and leadership theories, but was mainly focused to answer the research question on how to achieve SCI, with a special focus on SCM literature based on a holistic significance, which incorporated a complex set of concepts.

Henceforth, this chapter, with the support of the comprehensive post literature review (Section 2.6), as summarised in Table 7.1, provided explicit links of the literature to the conceptual model and also enabled the researcher to derive the study implications to the extant literature, by integrating the emerged theory within the relevant literature for logical extension and comparison (Strauss, 1987) and for theory building (Eisenhardt, 1989). Such an approach also enabled the researcher to fill the literature gaps, refined the theoretical framework (Strauss & Corbin, 1998; Charmaz, 2006), as referred in Figure 7.2, and also identified the research contribution (Glaser, 1998; Strauss & Corbin, 1998). It cannot be excluded the role of the a priori literature review (Section 2.3), which also provided a broad but a solid foundation of the literature to assist the final sense-making process within this chapter.

Integrative management and leadership approach (IMLA):

- Seuring and Muller's (2007) study used the **integrated chain management (ICM)** concept, within the context of sustainable SCM, based on a set of transactional and transformational processes on three perspectives, in particular material and information flow (i.e. production and logistics management); the strategy and cooperation (i.e. to actively manage product and material flows); and the industrial network (i.e. exchange of materials to promote sustainable practices) to reach competitive sustainability.
- Defee et al.'s (2010) study focused on a **transformational leadership approach** within a SC for both leaders and followers, with three mediating constructs and two outcomes, namely information availability, communications and rewards, as the mediators, which led to efficient and effective performance, as the outcomes.
- Kaynak & Hartley's (2008) study was based on quality management practices based on eight dimensions as antecedents that led to a set of performance outcomes. The **management leadership**, was considered from a quality based approach, and was referred as the core driver of all the other seven dimensions: training, employee relations, customer focus, quality data and reporting, supplier quality management, product/service design, and process management. The outcomes were inventory management, quality, financial and market performance respectively.
- Lockstrom et al.'s (2010) study inductively derived a theoretical model focused on focal firm **leadership approach** (i.e. buyer side) as the key antecedent to achieve SCI with suppliers so as to promote SC performance improvement, where the buyer-side leadership effectiveness was found to be a pivotal antecedent for building motivation, trust, and commitment among suppliers which facilitated strategic alignment and all collaborative capabilities and which acted as key enablers for successful supplier integration.

Management concept from the IMLA:

Managing effectively through people to achieve results within the firm and across the SC is a key concern (Drucker 1979; Crainer, 1998; Ashby et al., 2012; Ellinger & Ellinger, 2014); monitor performance improvements and optimisation (Drucker, 1977); to have the right people with the appropriate skills and talent within and across the SC (Rothwell, 2002; Sarkis et al., 2010; Tarique & Schuler, 2010; Xu & Thomas, 2011; Ashby et al., 2012; Ellinger & Ellinger, 2014); there is the need to ensure employees' health and safety and their equal treatment (Leire & Mont, 2010); human element is pivotal in all problem solving and critical thinking abilities (Kuhn, 1999; Halpern, 2003); the human resource strategies contributes to SCI and performance (Shub & Stonebraker, 2009); the role of the people contributes to the operations strategy (Hayes & Wheelwright, 1984; Hayes et al., 2005); management is about achieving results through people (Crainer, 1998); management discipline is not only a science but also an art (Watson, 1986); 'key business processes need to be integrated and managed' (Lambert et al., 1998, p.11); collaboration is a continuum based on close collaboration (i.e. partnership) to an arm's length type of relationship (Lambert et al., 1998; Cox et al., 2003; Chicksand, 2015; Cuevas et al., 2015); there is the need to establish the common good of the overall SC (Cooper et al., 1997); and the operational linkages (i.e. systems, procedures and routines) for the effective flow of goods, services and information need to extend beyond a single dyadic relationship (Cox et al., 2003).

Strategic leadership from the IMLA concept:

It includes a leadership approach which consists from a leader, a follower and the contextual conditions (Tannenbaum & Schmidt, 1973; Tosi, 1991; Osborn et al., 2002; Liden & Antonakis, 2009); consists of soft (e.g. trust, commitment and attitude) and hard (e.g. contractual obligations, product design, process and information technology) skills between all SC actors, so as to collaborate together (Whipple & Frankel, 2000; Shub & Stonebraker, 2009; McCauley-Smith et al., 2013); shifts beyond the dyad approach to a relationship approach, across different organisations (Hunt, 1991; Finkelstein & Hambrick, 1996); is focused on a collective or shared or distributed leadership approach (Carson et al., 2007; Martin et al., 2007); is instituted at all levels of the organisation (Deming, 1986; O'Reilly et al., 2010); adopts a shared leadership approach and is achieved through corporate intelligence (McKelvey, 2001); is an interdependent leadership style focussed on leadership and power sharing respectively, to promote empowerment in a decentralised leadership approach (McCauley-Smith et al., 2013); achieves organisational effectiveness (Boal & Hooijberg, 2000) and cannot be based on a single individual (House & Aditya, 1997), though it cannot be excluded that the individual leadership style still exists and it is based on hierarchical structures focused on the individual leader (Gronn, 2002; Tsui et al., 2006; Ling et al., 2008); is pivotal for SCI and performance (Whipple & Frankel, 2000; Osborn & Marion, 2009; Shub & Stonebraker, 2009); with its leadership structure surely affects the level of commitment of SC members (Cooper et al., 1997); with its power concept within the leadership literature is considered as a key source of control by influencing the people's behaviour through either a cooperative or aggressive collaborative strategy

(i.e. non-adversarial and adversarial relationships) (French & Raven, 1968; Mitchell et al., 1997; Cox et al., 2003; Hingley, 2005; Mullins, 2005; Co & Barro, 2008; Yeung et al., 2009; Hingley, 2005; Chicksand, 2015); and needs the right power deployment in all collaborative relationships to promote a win-win approach with trust-building (Hingley, 2005; Terpend & Ashenbaum, 2012) although a win-lose approach cannot be excluded which may also be beneficial for both parties (Cox, 2004d).

Leadership traits within the IMLA:

Transformational leadership: is based on five basic components, which are idealized attributes, idealized behaviours, inspirational motivation, intellectual stimulation and individualized consideration (Bass & Avolio, 2000); it drives people to see the deeper purpose in their work, treats them as unique and valuable human beings, and instils in them a belief in what the leader is trying to achieve (Bass and Avolio, 2000; Ardichvili & Manderscheid, 2008); and it produces powerful outcomes (Wang & Zhu, 2011).

Transactional leadership styles: is based on a set of established objectives and tasks, as an exchange process between the superiors and the subordinates, leading to reward or punishment vis a vis performance (Burns, 1978; Bass & Avolio, 1994, 2000; Bass, 1997).

Servant leadership: is based on the leader's primary motivation and role, to be a service to others and with distribution of power to followers (Greenleaf, 1970, 2002); and such an employee relationship leads to the achievements of the goals (Smith et al., 2004).

Situational leadership: is based on the contingency theory of leadership (Fiedler, 1967; Lawrence & Lorsch, 1967; Thompson, 1967; Hershey & Blanchard, 1969).

Business strategy:

It establishes the direction and scope of an organization over the long-term to meet the needs of the markets and to fulfil stakeholder expectations by the configuration of resources within the contextual environment (Johnson & Scholes, 1999); to unify all functional areas together with all SC members with their capabilities and limitations (Skinner, 1969; Stonebraker & Afifi, 2004; Thompson et al., 2005; Swink et al., 2007); and operations strategy implements the business strategy by focussing on the entire SC (Narasimhan & Carter, 1998; Chen & Paulraj, 2004; Lummus et al., 2008).

Manufacturing strategy:

It needs manufacturing leadership practices to pursue the operations strategy, through the business's manufacturing policies and people, to promote a competitive edge (Skinner, 1969; Hayes & Wheelwright, 1984; Hayes et al., 2005); with respect to the people shared values importance, as referred in McKinsey 7-S model (Waterman et al., 1980), outlines the key role of the human element, within both the manufacturing and SC leadership processes (Rothwell, 2002; Tarique & Schuler, 2010; Xu & Thomas, 2011); deploys HVM practices, based on the triple bottom line sustainable practices in all processes, both within manufacturing and also across all SC operations, so as to achieve competitive advantage (Martinez et al., 2008; Technology Strategy Board, 2012); decides on the make or buy option so as to sustain its competitive advantage (Skinner 1969; Harrigan, 1985; Klein, 1988; Bajec & Jakomin, 2010); decides of the make or buy based on the core competency of the focal firm (Cox, 1999; Baloh et al., 2008); exploits the cost of operations of the focal firm and of its suppliers to improve on the overall competitiveness (Williams, 1985; McIvor, 2010); and is based on lean manufacturing for cost and efficiencies, and deploys agile manufacturing, for a more responsive approach at the downstream side of the SC (Yusuf & Adeleyei, 2002; Gunasekaran et al., 2008; MacBryde et al., 2013) to meet the changing demands of the chain's ultimate customers (Cox et al., 2003; Green & Inman, 2005), referred as servitization of manufacturing (Lightfoot et al., 2013).

SC strategy:

Is needed to perform well in the 'competitive market' (Cox, 1999); is concerned with the integration and coordination decisions within the focal firm in-conjunction with the set-up with all SC stakeholders (Gunasekaran et al., 2008; Green et al., 2014); needs to consider the variation in conditions by undertaking a cost benefit analysis to determine the focal firm way forward for the type of SCI and also implies that a contingent model needs to be adopted to fit each business situation (Cox, 1999; Stonebraker & Afifi, 2004); is not a 'one-size-fits-all' model, since it does not exist, due to the unique characteristics of each SC (Heide, 1994); is function of the new emerging manufacturing paradigm, referred to as high value manufacturing (HVM) based on the virtual enterprise (VE) concept (Shi & Gregory, 2005; Livesay, 2006; Martinez et al., 2008); this management paradigm shift incorporates a change from vertical integration to virtual integration, based on unity of efforts, to optimize the use of all core and distributed resources to form a single holistic SC (Prahalad et al., 1994; Power, 2005; Gunasekaran et al., 2008; Cheung et al., 2012); from an enterprise management perspective may deploy either a selection or a mix from VIE, EE, VE and DE collaborative relationships among all SC actors in its inter-organisational approach, according to the collaborative enterprise governance, which outlines the continuum of the operations strategy, in line with the Dynamic Enterprise Reference Grid (DERG), to achieve competitiveness (Board & Clegg, 2001; Binder & Clegg, 2006, 2007; Clegg et al., 2012; Clegg and

Wan, 2013); within a SC, various stakeholders may need to link all activities in line with the value-chain approach (Porter, 1985), to promote both effectiveness and efficiencies and a value-added approach in all chain of activities, referred as vertical integration or vertical linkages (Swink et al., 2007; Guan & Rehme, 2012); needs to decide on how to gain from the optimum performance from each SC members so as to eliminate the risks associated with sub-optimisation effect that may result from friction between SC actors optimal (Cox et al., 2004a; Narayandas & Rangan, 2004); consists from both strategic and operational capabilities and with their alignment (Cox, 1999; Kim, 2009) function of the contextual conditions (Cox, 1998); is function of the type of product (i.e. functional or innovative type) to achieve a strategic fit, by matching the appropriate SC design based on either a manufacturing approach with efficient processes or a marketing approach with responsive processes to be competitive (Fisher, 1997); is based on a pull-based SC system (Kros et al., 2006), to align the production and business processes throughout the supply chain to the changing demands of the chain's ultimate customers (Cox et al., 2003; Green & Inman, 2005; Green et al., 2007; Lambert, 2008); must consider long-term relationships with a few dedicated suppliers to create an improved value and yield lower transaction and production costs (Bayraktar et al., 2009); and SC needs to cater for social, environmental and economical goals, also referred as the triple bottom-line (Elkington, 2002; Vachon & Klassen 2006; Cater & Rogers, 2008; Seuring & Müller, 2008; Bojarski et al., 2009; Buyukozkan & Berkol, 2011; Carter & Easton, 2011; Ashby et al., 2012; Gimenez et al., 2012; Lee et al., 2012; Sarkis, 2012; Wu et al., 2012; Caniels et al., 2013; Hsu et al., 2013; Glover et al., 2014; Hsueh, 2015).

Holistic SCI management approach (with its 16 sub-categories as referred in Figure 7.2):

Supply chain integration is the extent to which a firm is strategically interconnected and aligned with its supply chain partners (Das et al., 2006; Jayaram et al., 2010); the SC entities forming up the SC need to be integrated as a single entity (Chan et al., 2003; Sanders, 2007) and wherever possible focus on virtual integration rather than vertical integration to optimize the use of all core and distributed resources to form a single holistic SC (Prahalad et al., 1994; Power 2005; Wang & Wei, 2007; Gunasekaran et al., 2008); there is the need to shift from an intra management perspective to an optimized management approach across the SC with the effective involvement of all SC actors, since rarely a single firm has the entire technical expertise (Boardman and Clegg, 2001); the manufacturing strategy is directly linked to the SC integrated activities of both upstream and downstream to link all stakeholders with the focal firm (Ragatz et al., 1997; Flynn et al., 1999; Frohlich & Westbrook 2001; Paulraj et al., 2008; Flynn et al., 2010); holistic supply chain management approach has been considered of pivotal importance for the formulation SC strategies that are optimal for the whole SC (Wilding, 1998; Lambert & Cooper, 2000; Perona & Miragliotta 2004; Power, 2005; Fawcett et al., 2008; Lambert, 2008; Bayraktar et al., 2009; Cheung et al., 2012); take the most appropriate decisions by considering all constraints across the SC (Frohlich & Westbrook, 2002; Bayraktar et al., 2009); all potential SC 'barriers' need to be seen in 'combination or holistically', so as 'to separate the trees for the forest' (Fawcett et al., 2008, p.36); 'the benefits derived from integration activities outweigh their associated costs, leading to overall greater levels of operational performance' (Schoen & Swink, 2012, p.100); the Demand Chain Management (DCM) is based on an extensive integrated approach across the SC to have an adequate understanding of the demand (Vollmann et al., 2000; Frohlich & Westbrook, 2002; Juttner et al., 2007); SC processes need to be aligned, adaptable and agile (Lee, 2004; Ishaq et al., 2012); and the leadership style needs a holistic stance to ensure that all stakeholders' requirements are considered, to promote synergy of all actions with a teamwork vision (Bass et al., 2003; Schippers et al., 2008).

Technology deployment:

The SC strategy to achieve integration, both within the focal firm and throughout all the SC tiers (e.g. logistics and supplies), needs to be backed by an IT infrastructure, to achieve both manufacturing and performance improvements (Carr & Smeltzer, 1999; Wisner, 2003; Gonzalez-Benito, 2007) and competitive success (Eisenhardt & Tabrizi, 1994; Birou et al. 1998; Kim, 2009); the role of technology is pivotal in enabling all information based processes (Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014) and processes within production automation and across the SC, such as supply management and logistics (Gunasekaran et al., 2008; Jin et al., 2014); the role of IT promotes performance improvement and also enables all collaborative processes, through: connectivity of all SC actors; real time information provision; information management techniques to promote SC visibility; value added services; and a holistic management approach of the SC (McAfee, 2002; Chesbrough & Teece, 2002; Koudal & Wellener, 2003; Fawcett et al., 2007; Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014); IT through dedicated ISs are making such enterprise management a technical reality to promote a competitive edge (Rodon et al., 2011; Clegg & Wan, 2013); the holistic approach across the SC could be made possible with the advent of the internet based technologies, since through the web, one is able to provide an integrated platform for customers and suppliers for all activities across the SC (Frohlich & Westbrook, 2002); the traditional ERP, referred as ERP I is undergoing a paradigm shift from single organisation management approach to an inter-organisational collaborative environment based on e-commerce and web base technologies, referred as ERP II (Echartz et al., 2009; Clegg & Wan, 2013). Such technology deployment (e.g. EAI, SOA and SaaS) is also supporting agile manufacturing (Banker et al., 2010); IT applications (e.g. ERPs and databases) provide the ability to capture all real time information and adopts the necessary information management techniques to promote SC visibility, enable value added services and to achieve a holistic management approach of the SC (McAfee, 2002; Chesbrough & Teece, 2002; Koudal & Wellener, 2003; Fawcett et al., 2007; Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014); there is the need to use of smart technology within both manufacturing processes and products to promote competitiveness which is attributed to the IT-driven transformation through smart connected devices providing three perspectives: physical, smart and connectivity with the provision of IoT. The four areas of advantages are: monitoring, control, optimization and autonomy (Porter & Heppelmann, 2014); the businesses within the SC are highly dependent on an effective information processing approach to develop and manage knowledge across the SC through information technology applications (Dyer & Singh, 1998; Ofek & Sarvary, 2001; Faraj & Sambamurthy, 2006; Fawcett et al., 2007; Schoenherr & Swink, 2012) and with the deployment of technology to eliminate waste in several processes, by making them more efficient (Cox, 1999; Gunasekaran, 2004; Sanders, 2008); information technology with its systems assists organisational processes for developing and deploying relational capabilities (Bharadwaj, 2000; Hult et al., 2002; Paulraj et al., 2008; Sanders, 2008; Gunasekaran et al., 2008; Clegg &Wan, 2013); and all relevant information from all IT systems from SC actors can be consolidated by using 'DSS for holistic decision making' to make the most suitable decision (Bayraktar et al., 2009, p.137).

Competitive capabilities (operational indicators & business indicators):

Today's competition is not focused between firm to firm competitiveness but between entire supply chains (Dyer, 2000; Christopher & Holweg, 2011); an effective SC strategy needs to adapt and structure its resources to effectively align to the changing competitive capabilities (Thompson, 1967; Hofer, 1975; Bojarski et al., 2009; Kim, 2009); the SC capabilities may result in distinctive competitive capabilities within both the focal firm, in line with the resource-based view (Wernerfelt, 1984, Barney, 1991) and also from a wider approach to operations, where the focal firm deploys all resources outside its boundaries with its SC actors, in line with the extended resource-based view of a network of firms (Matthews, 2003); the various SC actors' activities sum-up to an overall competitive advantage (Porter, 1996; Juttner et al., 2007); SCM is considered as a crucial operations strategy for organizational competitiveness with the direct involvement of people (Gunasekaran, et al., 2008); supply chain integration approach, both within the focal firm and across the SC, promotes performance improvements (Braunscheidel & Suresh, 2009; Flynn et al., 2010; Koufteros et al., 2007; Swink et al., 2007; Wong et al., 2011; Zhao et al., 2011); agile manufacturing 'should not only be based on responsiveness and flexibility, but also on the cost and quality of goods and services' (Gunasekaran et al., 2008, p.550); IT capability leads to superior performance (Determirhan et al., 2007); information technology has been recognised in the last two decades that it is a source competitive advantage due to provision of systems that assist organisational processes for developing and deploying relational capabilities (Bharadwaj, 2000; Hult et al., 2002; Fawcett et al., 2007; Paulraj et al., 2008; Sanders, 2008; Gunasekaran et al., 2008; Bayraktar et al., 2009; Clegg &Wan, 2013); from the RBV of the firm (Wernerfelt, 1984; Barney, 1991) and the extended RBV (Matthews, 2003) SC flexibility with its competitive edge is being created not from the investment in technology, but how the technology is being used, through its dedicated information sharing mechanisms (e.g. ERP customisation of integrated processes), both within and across the SC, which are treating both the information management and the IT infrastructure capabilities as resources, which may be distinct from other SCs (Prajogo & Olhager, 2012; Schoenherr & Swink, 2012); the integrated activities across the SC, with the consideration of the activities' values and costs of all the SC members' activities, contribute to the overall SC process with a win-win mindset leading to competitiveness and profitability (Dyer 1994; Lambert et al., 1998; Rosenzweig et al., 2003; Ketchen & Hult, 2007; Swafford et al., 2008; Bayraktar et al., 2009; Flynn et al., 2010; Zhao et al., 2013); SC member creates value-added operations (e.g. innovation, improved customer service, superior product attributes) and also manages effectively its appropriation of value (e.g. extracts above average profits, build brand name), since both value creation and value appropriation are strategic objectives required for achieving sustained competitive advantage (Mizik & Jacobson, 2003; Zhang & Chen, 2008); internal integration is the foundation of SCI initiatives across the SC since it also provides the bridge between the supplier and the customer integration initiatives to promote performance improvement and/or operational capabilities which in turn leads to improved business performance (Frohlich & Westbrook, 2001; Koufteros et al., 2005; Swink et al., 2007; Bayraktar et al., 2009; Flynn et al., 2010; Schoenherr & Swink, 2012); operational or manufacturing capabilities lead to business performance either in individually or in concert (Flynn et al., 1999; Rosenzweig et al., 2003; Swink et al., 2007); business growth and expansion have to be based on alliances, partnerships, and all kind of relationships (Peter Drucker, 1999); SC strategy is concerned about upstream (Carter et al. 2000; Scannell, Vickery, and Dröge, 2000; Kim, 2009; Swink et al., 2007) and downstream (Ellinger et al., 2000; Kim, 2009; Swink et al., 2007) relationships to give a competitive edge; sustainable performance or sustainable competitive advantage not only refers to an economical perspective but also from a social and environmental perspective (Elkington, 2002; Luken & Stares, 2005; Lee, 2008; Lee & Klassen, 2008; Glover et al., 2014; Hsueh, 2015); the environmental perspective alone has become a source of competitive advantage (Vachon & Klassen, 2006; Ketchen & Hult, 2007; Hsu et al., 2013; Caniels et al., 2013).

- **Operational indicators:** operational objectives of cost (i.e. quality and features), dependability (i.e. on specifications, time and service), quality (i.e. process quality) and flexibility (i.e. product and volume) (Hayes & Wheelwright, 1984; Flynn et al., 1999); competitive operational capabilities with special attention on responsiveness and flexibility (Yusuf et al., 2002; Gunasekaran et al., 2008; MacBryde et al., 2013) and innovation (Binder et al., 2007); competitive operational capabilities for both the focal firm and with all upstream and downstream SC sides are: cost, flexibility, speed/delivery and quality (Hayes and Wheelwright, 1979b; Frohlich and Westbrook's 2001; Kim, 2009; Richey et al., 2010; White et al., 2010).
- **Business indicators:** financial or business competitive capabilities are sales, market share, and growth (Johnson, 1999; Flynn et al., 2010) and return on assets, return on sales and return on investments (Rosenzweig et al., 2003; Flynn et al., 2010).

Table 7.1: Summary of the conceptual model categories within the literature (Author)

The next chapter concludes the thesis by outlining the key findings with a set of propositions, the emerged theory outline and the research contributions, limitations and suggestions for future research.

Chapter 8: Conclusion

8.1 Introduction

This chapter concludes the thesis by outlining the key findings and the contributions generated from the data through the Grounded Theory methodology, based on Corbin and Strauss (2008). The GTM research design comprised of two literature reviews, namely the initial and the final literature reviews respectively. The initial literature review mainly assisted the researcher's theoretical sensitivity, in such a way that the researcher was in a better position to understand what the data were saying, throughout the research data collection and analysis process. The final literature review was mainly focused on the emerged themes from the research findings, which were used to contrast the emerged theory implications within the extant literature. Furthermore, the final literature review also played a crucial role at the end of the data analysis process, since the extant literature served as another source of data (Strauss, 1987; Glaser, 1998; Charmaz, 2006), which was used to refine the outcome part of the final emerged theory within the theoretical model, as referred in Figure 7.2.

The researcher's objective is to a build theory on how to achieve SCI by establishing the key characteristics through a cross-disciplinary approach, so as to identify and describe its underpinning theoretical foundations. The generated substantive theory to answer the research question (Section 1.6) is very comprehensive and consists from a multiplicity of factors which have a varying degree of impact on what constitutes SCI with its respective outcomes and within a set of both micro and macro contextual conditions (Table 5.1). As a result the researcher from this study has drawn out the research key ideas, some original insights and propositions and also established the substantive theory implications to the extant literature and to policy and practice. Furthermore, the research also included the research limitations and ideas for future research.

The generated substantive theory is focused on how to achieve SCI, which is a concept within the discipline of SCM, which is in turn a sub-field of operations management. The research touches various other disciplines to promote a holistic understanding and to show that SCI within manufacturing is a multidisciplinary phenomenon, with a wide-thinking perspective. As a result the SCI foundations are based on several disciplines, other than SCM, such as technology deployment (i.e. information systems and manufacturing technology), organization behaviour (i.e. management and leadership roles), psychology (i.e. human behaviour role) and sociology (i.e. SC actors' role in a society).

The inductively derived emerged theory with its theoretical model is original (Figure 7.2) and is also compared with other models within the extant literature, to derive the commonalities and the differences (Section 7.3.4). Furthermore, the researcher has examined and theorised on the research phenomenon by using well established seminal theories to promote a robust theorising approach, based on strategic management (such as RBV and extended RBV), business ethics (such as stakeholder theory), sociology (such as institutional theory), economic theory (such as transaction cost analysis (TCA)) and social exchange theory (such as resource dependency theory).

Such a research study promoted theory extension of the main fields under focus (e.g. SCM, SCI and leadership) and solidified the contributions, by not only contributing to the latest current disciplines, such as operations management and SCM among other main fields in the literature, but also contributed, through their application, to the established and conventional theories which are used in various scholarly works to promote understanding of the emerging concepts and the theorising processes within the SCM and OM extant literature, such as RBV, TCA and stakeholder theory.

This chapter summarises the research scope with the substantive theory main issues and challenges in the first section. The second section is focused on the research significance. The third section is dedicated to the emerged core theory outline. The fourth section is focused on the relevance of the core theme, which is followed by the main themes, with additional snapshot of all the sub-themes, as a roadmap to practitioners. The fifth section lists the theoretical and methodological contributions, which is followed by the contributions to practice and to policy. The final section links the research study with the research limitations and future research.

8.2 Research scope and the substantive theory main issues and challenges

The data clearly showed that the supply chain operations within the Maltese manufacturing sector SMEs have a certain set of homogenous indigenous characteristics, although with a certain level of variation. The key manufacturing based SC process is assumed to consist from four main stages: procurement, fabrication, assembly and distribution among all other supporting and inter-linking related stages, such as research and development, quality inspections, testing, waste management and reverse logistics. The homogenous characteristics refer to the: flat organisation structures; promising cultural values of people at the workplace, based on high intellectual capabilities and skills (e.g. bilingual and highly certified) and with a deep commitment to learn and to work together (e.g. team spirit); proximity to local contractors (i.e. mainland logistics) and to customs/shipping terminals (i.e. sea and air logistics); high quality standards implementation; ICT applications (e.g. barcode readers and databases) and process automation equipment deployment (e.g. robotics); an effective

management approach focused on both SC and manufacturing based processes, with a particular focus on lean operations, quality certification and legal compliance; and streamlining of operations based on a deep concern on environmental sustainable measures, together with a special focus on the social and the economical sustainable measures.

It is to be noted that a main variation existed vis a vis the innovative commitment embedded within the human element, which was more visible in the processes for high technology-based manufacturing sectors, since the technology-based products, such as electronic devices and equipment, are very dynamic and undergo a relatively faster pace of developments in the market relative to other sectors, such as healthcare, printing, food and toys. Though, it cannot be excluded that with today's advent of technology based processes and knowledge sharing environment across the SC and with the customised services' provision to a wide type of customers' segments, innovative developments in products and processes are also touching all types of sectors, but with a different level of intensity, which makes the business scenario very dynamic and challenging.

From the above contextual outline, the research has produced a parsimonious model of a set of antecedents with their relationships and a set of outcomes, embedded within different micro contextual conditions of each SC actor. Although all SC actors have the macro local contextual conditions common, they have different macro international contextual conditions, due to the different products' destinations. The model, as referred in Figure 7.2, is based on a set of six inter-organizational main conceptual elements that all stakeholders building up the SC, including the focal firm, need to manage effectively to achieve competitive capabilities. It is to be noted that SCI practices need to walk through several functional areas within the focal firm and furthermore, also need to have access to different SC actors across the SC. Such a stance requires an integrative management and leadership approach (IMLA) with a level of cross-cultural knowledge and charisma and a high level of expertise on how to go about such relational approaches, so as to be sensitive enough to manage such critical strategic moves to promote successful overall SC performance. This argument is being put forward, since the SC crosses several boundaries, both within the focal firm and across all SC tiers, where access is function of all players, with a mix of dependent and independent entities. To take command of external SC actors is very challenging, since depending on each SC actor role, the interactions may vary from non-adversarial to adversarial type, within both the arm's length and the close collaborative types of relationships respectively. Furthermore, it is the norm, that the collaborative initiatives, to achieve each other' requirements, are mainly based on a cooperative approach, function of the role of each SC actor, which is built on a trustful relationship over time. Though, it cannot be excluded that other circumstances may also exist, where the lack of trust, may drive any SC actor to take an aggressive stance. Such a cooperative collaborative environment is pivotal within the SC operations,

since SCI consists from various interactions and needs to be performed effectively to exploit from each other all the resources, so as to promote business continuity from all dynamic situations and to achieve competitive performance for a win-win outcome to all SC actors.

From such a Maltese manufacturing sector context and the emerged theory outline situates the significance of the core research theme, where the IMLA is needed to meet the demanding holistic SCI objectives in view of all the challenging complexities and contingent situations.

8.3 Research significance

Existing literature is replete with both conceptual and empirical scholarly work on SCI within manufacturing, but somehow lacks holistically integrating models (Sections 2.5.9 & 2.6.10.2) that encapsulate the SCI phenomenon in terms of various conceptual elements in one model, since most of the theoretical models are fragmented and focus only on a few number of concepts at a time (Pibernik & Sucky, 2006; Svensson, 2007; Flynn, et al., 2010; Lorentz et al., 2012; Schoenherr & Swink, 2012).

Furthermore, the extant literature is also well supported by scholarly work focused on the dimensions of SCM practices from a variety of perspectives, but there is lack of literature on how to manage and lead the overall SC (Lambert & Cooper, 2000; Fawcett et al., 2008; Flynn et al., 2010). In particular, within the SCM extant literature various scholars derived: **SCM conceptual frameworks** (e.g. Chicksand's (2009) Power Matrix; Cox et al.'s (2003) Buyer-Supplier Relationship Types Matrix; Fisher's (1997) Model); **SCM principles** (e.g. Cooper et al. (1997); Lambert et al. (1998); Stonebraker & Afifi (2004)); and **SCM best practices** (e.g. SCOR Model (SCOR, 2010). But, holistic models are lacking, and furthermore, it is clearly being referred within the literature, that SCI is complex, dynamic and contingent to every situation between each dyadic SC relationship or within an extended relationship approach between various SC actors (Vickery et al., 2003; Pero et al., 2010; Garetti & Taisch, 2012; Macbryde et al., 2013). In fact, the scope of the research addresses the gap for a holistic SCM approach, as referred by the statement within the garment manufacturing industry, that:

'Globalization and proliferation of multi-national companies, joint ventures, strategic alliances and business partnerships have changed the business environment. For survival under the pressure of turbulence, it is critical for companies to reorganize and optimize their supply chain effectively, by taking a holistic view of the supply chain' (Cheung et al., 2012, p.3923).

The emerged themes within the holistic SCI model were also contrasted within the extant literature, to derive the commonalities and the differences and it transpired that there were many areas of

congruence but there were also a number of new insights. The main difference is attributed to the core emerged theme, which is original, in particular the holistic stance undertaken, since no literature seems to capture such a comprehensive approach.

Within the global scenario, the core theme, being based on a management and leadership approach, has also captured the importance within the World Economic Forum report (WEF, 2014b), where it was outlined the importance of high performance leadership best practices in international organisations, within EU, whereby it outlined the key areas of focus that facilitates good leadership:

⁽¹⁾ selecting and re-electing leadership on merit, (2) managing performance, (3) setting and evaluating ethical standards, (4) developing and retaining talent, (5) setting strategic priorities, (6) engaging with a wide range of stakeholders, and (7) evaluating independently and effectively' WEF (2014b, p.5).

Such leadership awareness initiatives, referred by WEF, are key guidelines for practitioners so that all firms set-up the right instruments, establish sound and ethical principles and put in place accountability mechanisms for effective governance, both within the focal firm and across with all SC actors. Such an approach is in line with the research IMLA core theme objectives within this study. The leadership approach within this thesis is used to fit the situation and there exists both structured and unstructured situations which need to be catered by the four leadership traits in-built within the IMLA.

8.4 The research emerged theory outline based on the core theme: The IMLA

The emerged core theory, as referred in Section 6.3, implies that for the focal firm to achieve SCI, it needs to implement an IMLA with the support of its business, manufacturing and SC strategies, which together drive the holistic SCI management approach, enabled by technology deployment in manufacturing and in information systems to achieve competitive capabilities.

The IMLA creates direction and alignment across the SC without any boundaries (i.e. beyond the dyad), so that the overall SC will be in a position to pursue a multiple set of competitive priorities. With such an approach, the IMLA needs to be focused on the use of all the SC actors' resources and capabilities effectively and efficiently to be sustainable in all actions. Such a stance enables the overall SC to promote value-added operations and achieves performance improvements in both the manufacturing and the SC processes to achieve its competitive capabilities.

As a result the focal firm, being part of the SC, needs to have the right set-up with a number of different stakeholders, who in-conjunction needs to build a collaborative commitment to fit in each others' strategic goals for a win-win outcome. It is not being excluded that a win-lose situations may even occur from time to time, for both the focal firm and any other SC actor, but the scope of SCI is based on long-term relationships and is not on quick fix solution, and as a result all SC stakeholders consider long term benefits over short term failures to build on the established SC relationships. The SC is as strong as its weakest link, where the SC needs to apply a set of holistic performance metrics, through its holistic SCI management approach, to detect such bottlenecks, so as to prevent from any domino effect originated from any SC actor's failure over the overall SC. Such a stance enables the focal firm to achieve performance beyond a level that it could have achieved on its own through the effective commitment, synergy and empowerment of all SC members.

The IMLA, depending on the management leadership values, ideally adopts a strategic leadership approach, based on a shared leadership mindset, but may else adopt an individual leadership approach, based on a centralised leadership mindset. The IMLA is rooted within four types of leadership styles: transformational; servant; transactional and situational styles respectively, to meet different situations, as referred in Section 6.3.1.

From the findings, it was also derived that to achieve SCI, the management and leadership practices are instrumental in getting the work done at all management levels, with a higher influence on the bottom line. Considering the small size of the firms understudy, being SMEs within the manufacturing sector, the interaction between all management levels with the bottom line employees is evident, since even the CEO tries, on a daily basis, to keep in touch with all staff, to build on a positive relationship between all workers and the top management. Though, one cannot exclude the pronounced effect, of the one-to-one contact and strong relationships that exist between the supervisors and the line managers with all the bottom-line staff. All such interactions constitute SCI initiatives based on an IMLA with a level of shared leadership variation between all staff within the focal firm. Similarly, such a relationship environment is also established with all SC actors across the SC depending on the role of each stakeholder.

The SC also needs to include the customer, since apart from the pull-SC mentality, which has been a widely accepted business principle (Kros et al., 2006; Green et al., 2007; Lambert, 2008), where the customer also needs to be given its due importance. Such a customer interaction commitment, requires that the SC success depends on an open mind and an inclusive mindset of all the SC players, based on both the customers' and SC stakeholders' interests. Though, it is to be noted that the effective customer service approach depends on the manufacturing sector/product type and the SC

actor role. As a result, there are firms which act as sub-contractors and make only a part of the whole product, or deliver a service, so the customer is another SC member (e.g. a distributor to a firm). Other firms are fully fledged manufacturers and have a direct link to the customer and as a result the service offered is direct to the final customer, with the possibility to engage in a more closely integrated customer-relationship approach to promote an improved customer service.

In summary, the research main contribution is that the study inductively derived a conceptual framework which extends the current literature, which is mainly based on a dyadic relationship (Croom et al., 2000; Cachon et al., 2007; Xiao et al., 2010; Huang et al., 2012) to a substantive theory that captures all the SC members. The emerged theory outlined that for an integrated SC to achieve its competitive capabilities in different criteria (e.g. sustainability and efficient and effective performance), as referred in the SCI conceptual framework Figure 7.2, requires a complex set of concepts which revolves around the IMLA. Any failure of any SC member to meet the established performance targets may have a negative domino effect on the overall SC. As a result the recommended **roadmap for all SC actors** to follow so as to achieve SCI, based on all the other five (5) main strategic actions, which revolves around the IMLA, is referred in Appendix 12. Though, one needs to keep in hindsight that finally the ultimate pivotal objective for the overall SC is to win in the market and to stay in business.

8.5 Theoretical, methodological and practitioners' contributions

The scope of this research is to produce novel ideas through the extension of current management and leadership theories within the SC context, together with SCI theory and a SCI theoretical framework to assist academics in contemporary and future research and assist practitioners in implementation of best practices to improve overall SC performance.

8.5.1 Theoretical contributions

From this research six theoretical contributions, with a set of seven propositions as summarised in Table 8.1, can thus be identified from the emerged theoretical themes, associated with SCI within the manufacturing SMEs supply chain:

First, this study presents a unique empirical research on SCI initiatives in SMEs within the Maltese manufacturing sector.

To my knowledge, research focused on SCI initiatives within the manufacturing sector in Malta, as a Small Island State, does not exist. Though, one cannot exclude the fact that SCM has been under the focus of several Maltese practitioners, in both manufacturing and services sectors, over a number of years, more than the commitment taken by the local academic institutions. Only lately tertiary education curricula has introduced SCM dedicated courses, except for the disparate individual commitments taken by some people a number of years ago, to push the subject from the business arena, through one-off training programmes, so as to drive policy makers to recognise the importance of such a subject. Hence, such a contribution adds to both the extant literature on SCI and to the Maltese academic work.

Second, the research contributes to the generation of a holistic SCI substantive theory, together with its conceptual framework (Figure 6.1), which is firmly grounded in an in-depth analytic approach based on a broad and multidisciplinary perspective of the data, gathered from SMEs within the Maltese manufacturing sector.

Such a contribution refers to the generated theory based on a holistic theoretical rationale, which captures all the SC members within a SC set-up, beyond the dyad relationship. Such a theory establishes an understanding of what constitutes supply chain integration to meet the overall SC competitive capabilities, within today's dynamic and challenging competitive environment. Such SCI holistic models are scarce or sparse in the extant literature (Vickery et al., 2003; Pero et al., 2010; Garetti & Taisch, 2012; Macbryde et al., 2013). Hence, such research, contributes to the SCI holistic models within the extant literature (Prahalad et al., 1994; Frohlich & Westbrook, 2002; Power 2005; Shi et al., 2005; Fawcett et al., 2008; Gunasekaran et al., 2008; Bayraktar et al., 2009). Furthermore, the extant literature is also used to refine the theoretical model's outcome to strengthen its significance, by extending the competitive capabilities outcome, as an intervening antecedent concept, to both operational and business indicators outcomes, as referred in Figure 7.2.

Third, the research addresses the importance of manufacturing and SC sustainability, technology, quality and lean management bodies of knowledge, in line with the extant literature, where it was established that it is a common belief that both manufacturing and SC strategies need to be targeted according to both the 3BL (triple bottom line) or Triple Ps (people, planet and profit) and the lean management objectives, with the right use of technology, to achieve competitive capabilities.

This research further extends these bodies of knowledge, by stating that this sustainable approach needs to be implemented in all processes including also the administrative and managerial based operations within all manufacturing and SC processes. Such an approach needs the adoption of due

diligence and lean procedures through empowered initiatives, in both technical standards (e.g. ISA) and quality standards (e.g. ISO and GMP) together with all managerial approaches, at all management levels, with the right use of technology in ICT and automation to promote economical, social and environmental sustainable processes. The findings established the need of technology deployment in both information management and knowledge sharing initiatives (Sarker & Lee, 2003; Faraj & Sambamurthy, 2006; Fawcett et al., 2007; Schoenherr & Swink, 2012) together with the support it gives to all manufacturing and SC operations to promote competitive efficiencies (Gunasekaran et al., 2008; Bayraktar et al., 2009; Cheung et al., 2012; Abdullah & Musa, 2014; Jin et al., 2014). Technology is also used to eliminate waste in several processes, in line with lean management, by making them more efficient (Cox, 1999; Gunasekaran, 2004; Sanders, 2008). The technology deployment also promotes social sustainability, such as in health and safety issues (e.g. use of robotics in handling heavy items). The findings add to the conversation of technology deployment in manufacturing and SC operations and also to social sustainability within SCM and manufacturing literatures'. The research findings corroborate with the sustainable SCM literature (Elkington, 2002; Seuring & Müller, 2008; Buyukozkan & Berkol, 2011; Sarkis, 2012; Wu et al., 2012; Caniels et al., 2013; Glover et al., 2014; Hsueh (2015). Furthermore, the research findings also complement the lean management literature (Womack & Jones, 1996; Juttner et al., 2007; Bayraktar et al., 2009). It is not being excluded though, that some manufacturing sectors may even demand more stringent sustainability measures, having tougher standards in quality and/or legal measures. Such a contribution also refers to the fact that the economy is inevitably shifting the business climate to a scenario based on low cost operations, with low carbon footprint and with high societal responsibilities. Such a key concern outlines that all SC stakeholders need to implement both sustainable and lean best practices: through the use of environmental friendly resources; by employing environmental focused logistics and by adopting a lean management approach to ensure minimisation of labour, material usage and waste, in line with the 4 R's (reduction, remanufacturing, recycling and reuse); through less energy consumptions based on improved and efficient technology based systems; through less labour costs with improved operations and automation investment; and through more organised human-based activities with the right management and leadership approach.

Fourth, the research established that all SMEs through an IMLA, need to assess and revise on an ongoing basis and with a continuous improvement approach their manufacturing and SC operations, both internally and externally with all the SC stakeholders, to achieve a holistic SCI management approach, enabled by technology, through high quality and sustainable processes, with a particular focus on both managing change/innovation and SC flexibility/responsiveness respectively, so as to improve on the overall SC competitive capabilities.

Such a contribution refers to the fact that a holistic approach enables that all processes are to be innovative and lean, in line with both the quality standards and sustainability measures, which are all summing up to a set of competitive capabilities, as referred in the previous contribution. The literature advocates that to achieve a SCI across the overall SC from a holistic perspective is very challenging (Lambert et al, 1998; Cox, 1999; Lambert & Cooper, 2000; Storey et al., 2006). But, such a holistic approach is a need in today's competitive market, so that the SC becomes responsive and flexible as a unit (Cooper et al., 1997; Cox et al., 2003; Gunasekaran et al., 2008). Such responsiveness and flexibility measures are needed so that the overall SC is in a position to continuously improve all operations through an effective change management approach to promote innovation and maintain or improve on its competitive capabilities (Gowen & Tallon, 2003; Gunasekaran et al., 2008; Cheung et al., 2012; Ellinger & Ellinger, 2014). As a result, the Maltese SMEs, with such an innovative-driven approach with avante-garde technology based solutions, due to their limited use of resources and low labour count, will be able to drive themselves much easier than large firms, so as to achieve high levels of improvements, in line with both sustainable practices and high value manufacturing. Such a contribution adds to the conversation on competitive capabilities literature, since the holistic view of the firm with its value-added activities sum up to such competitive capabilities (Porter, 1985, 1996, 1998; Juttner et al., 2007) by giving a competitive edge over its competitors (Ward et al., 1994). Furthermore, such a contribution also adds to the SMEs literature and to Fisher's (1997) model vis a vis functional products, since SMEs are undertaking innovative initiatives even for such products, which also adds to the competitive manufacturing capabilities. Finally, the technology deployment within the organisation functional areas and outside the organisational boundaries, which provides relevant, timely and accurate information is also considered as a source of competitive capability (Fawcett et al., 2007; Bayraktar et al., 2009; Clegg & Wan, 2013). As a result, the innovative approach taken on functional products and the deployment of technology, are both considered sources of competitive advantage, which are also substantiated by the RBV of the firm (Wernerfelt, 1984; Barney, 1991) and the extended RBV of the overall SC (Matthews, 2003).

Fifth, the research established that the overall SC, through its IMLA, holistically need to involve the human element in building up all the SC up to the end customer, through all the necessary collaborative initiatives, to achieve competitive capabilities. Such a holistic SCI management approach, through the involvement of all the key stakeholders, which is enabled by technology, needs to build on trust and to jointly-derive the necessary business, SC and manufacturing strategies across all management levels, which are all needed to achieve the SC competitive capabilities.
Such a contribution refers to the fact that the IMLA is the managerial and driving force to achieve such a holistic SCI management approach and is also the theoretical concept that acts as the glue that holds the SC members all together in unity, with no boundaries whatsoever, so as to promote synergies and competitive capabilities. This research addresses Sydow et al. (2011) call for further research on the use of management and leadership concepts to achieve SCI. Furthermore, this study substantiates the need of leadership based research across different organisations (Hunt, 1991; Finkelstein & Hambrick, 1996; Osborn & Marion, 2009), even up to the end-customer through downstream integration (Guan & Rehme, 2012). Such a stance cannot be achieved without deploying the right mix of the human element capabilities with the technology based applications, in both ICT and automation, to promote SC visibility, flexibility and responsiveness. It is of pivotal understanding that technology does not replace the human element, but enables and sometimes even drives processes, through improved efficiencies, but the human element remains a crucial asset in both interfacing and decisionmaking processes. Such a theory based on the pivotal role of the human element complements the need to have the right people with the appropriate skills and talent (Rothwell, 2002; Sarkis et al., 2010; Tarique & Schuler, 2010; Xu & Thomas, 2011) and also substantiates the argument that the role of people with the right human resource management is crucial to the success of the SCM approach (Ashby et al., 2012; Ellinger & Ellinger, 2014). With such a contribution, the focal firm set-up within its functional areas, and even with all SC tiers, with the appropriate collaborative initiatives, will ensure to deploy the human element appropriately, through a holistic SCI management approach, enabled by technology, to achieve the SC competitive capabilities. Hence this study also substantiates the argument that the leadership strategies are pivotal for SCI and performance (Whipple & Frankel, 2000; Osborn & Marion, 2009; Shub & Stonebraker, 2009). Furthermore, in line with Co and Barro's (2008) cooperative and aggressive strategies, all SC actors need to play a pivotal part in all collaborative initiatives. Such a social element requires a new platform based on integrated initiatives (i.e. formalization/collaboration), systemic trust, distributed decision-making and broader and more participative organic structures through continuously trained employees, in line with strategic leadership focused on a distributed leadership approach (Hunt, 1991; Finkelstein & Hambrick, 1996). Such a contribution adds to human resource management, in line with social sustainability criteria, SCM, leadership, strategic management and technology applications, in information systems and manufacturing, within the extant literature.

Finally, this research extends the current set of definitions on SCI by incorporating a more comprehensive theoretical understanding of how to achieve SCI from a holistic approach through the deployment of an IMLA together with its five action main categories and sixteen action sub-categories to achieve competitiveness within a context.

Such a contribution refers to the fact that the emerged conceptual framework is based on a multidisciplinary approach focused on OM, SCM, IT, human resource management, management, leadership, sociological and psychological disciplines. Such a substantive theory is based on the IMLA with its empirical derived theoretical framework is an original contribution to the SCI phenomenon within the extant literature (Zhao et al., 2008; Flynn et al., 2010; Pero et al., 2010; Garetti & Taish, 2012; Schoenherr & Swink, 2012). The IMLA with its four types of leadership traits, it was inductively derived that: with the transformational leadership style, it meets change and innovation through people; with the servant leadership style, it is sensitive for people's needs and at the same time meets the business objectives; with the situational leadership style, it meets contingent situations; and with the transactional leadership style, it meets the routine flow of transactions in line with the established performance measures. This study complements the theories on the four leadership traits: Transformational (Burns, 1978; Bass, 1985; Bass & Avolio, 1994, 2000); Servant (Greenleaf, 1970); Situational (Fiedler, 1967; Lawrence and Lorsch, 1967; Thompson, 1967; Hershey & Blanchard, 1969); and Transactional (Burns, 1978; Bass & Avolio, 1994; Bass, 1997). Such a research also adds to the theory on strategic leadership, based on a distributed leadership approach across the SC within the leadership, SCM and manufacturing extant literatures, in line with McCauley-Smith et al.' s (2013, p.99) inference, where they stated that there is the need of 'sharing of good practice through an interdependent leadership approach'. Though, such a contribution does not exclude the fact that some SMEs may still have a top management style that adopts an individual leadership approach.

N.O.	Theoretical Propositions
1	To achieve supply chain integration the IMLA needs to be deployed at the strategic, tactical
	and operational management levels, with due consideration of each SC actor contextual
	conditions, so as to establish and implement the right strategies for the business,
	manufacturing and SC, within the focal firm and in-conjunction with all SC actors, based on
	a holistic SCI management approach, which is enabled by technology, so as to achieve
	competitive capabilities.
2	The IMLA needs to adopt a strategic leadership approach to build on the collective effort of
	all people based on four types of leadership traits to achieve both intra and inter-SCI, where:
	the transformational leadership style, needs to meet the change and innovation through people;
	the servant leadership style, needs to be sensitive for people's requirements and at the same
	time meets the business objectives; the situational leadership style, needs to meet contingent
	situations of all actors; and the transactional leadership style, needs to meet the routine flow of
	transactions in line with the established performance measures.
3	The business strategy through both the manufacturing strategy with its make or buy
	decision and the SC strategy collaborative initiatives with either an arm's length or close
	relationships determined by the role of each SC actor, need to apply the IMLA with the
	involvement of all SC actors, so as to jointly establish, from a holistic SCI management
	approach, the most effective and efficient operations set-up so as to promote competitive
	capabilities.
4	All SMEs through an IMLA, need to assess and revise on an ongoing basis and with a

	continuous improvement approach, their manufacturing and SC operations, both internally
	and externally with all the SC stakeholders, to achieve a holistic SCI management
	approach, enabled by technology, through lean, high quality, sustainable (3BL or Triple Ps)
	and HVM processes, with a dedicated focus on both managing change/innovation and SC
	flexibility/responsiveness respectively, so as to improve on the overall SC competitive
	capabilities.
5	The overall SC, through its IMLA, holistically needs to involve the human element in
	building up all the SC up to the end customer, through all the necessary collaborative
	initiatives, to achieve competitive capabilities. Such a holistic SCI management approach
	with its 16 sub-categories (Figure 7.2), through the involvement of all the key stakeholders,
	which is enabled by technology, needs to build on trust and to jointly-derive, through a
	common mindset, the necessary business, SC and manufacturing strategies across all
	management levels, which are all needed to achieve the SC competitive capabilities.
6	Technology deployment in IT applications and automation needs to adapt to all information
	and manufacturing processes requirements and needs to be an enabler of the holistic SCI
	management approach, to promote operational and knowledge sharing initiatives together
	with streamlined and optimised workflows, both within the focal firm and across the SC, to
	achieve competitive capabilities.
7	The IMLA together with its business, manufacturing and SC strategies with the holistic
	SCI management approach enabled by technology promote competitive capabilities,
	which consist of: SC flexibility and visibility; business continuity; continuous improvement;
	optimised and lean operations; effective and efficient performance; sustainability and effective
	customer service. All such capabilities are antecedents to both operational and business
	indicators (Figure 7.2).

 Table 8.1: Theoretical Propositions from the emerged substantive theory (Author)

8.5.2 Methodological contributions

This research has used a grounded theory approach to generate a substantive theory that explains SCI initiatives within the Maltese manufacturing SMEs. As a result such a study extends the applicability of grounded theory based research within SCM and related OM disciplines since it was found as the 'best-fit' methodology since the research objective is after theory verification and generation (Suddaby, 2006; Binder & Edwards, 2010), where the emerged theory was verified and validated by the iterative constant comparison process (Section 3.9).

Furthermore, such a study answers the call for such a GTM methodology usage, since 'it is not very widespread and rigorously applied in operations management (OM) research' (Binder & Edwards, 2010, p.232).

Finally, although the researcher has clearly indicated the systematic approach taken by the GTM approach (Strauss & Corbin, 1998), the overall theorising process was not a continuous sequential process, but there were instances, that the researcher needed to conceptualise at different stages to assist the theorising process, through theoretical abstractions, so as to reach a higher level of understanding of what the data were saying to arrive at the emerged theory.

8.5.3 Contributions to practice and to policy

The substantial part of the research is focused on the role of practitioners within manufacturing and SC operations, since the primary data constituted a key part of the research commitment with its collection and analysis. Various secondary data sources were also retrieved to understand the implications of policies and practices. In general, the Maltese manufacturing sector has always been considered one of the pillar stones of the Maltese economy, due to its substantial contribution to the economy GDP. Furthermore, such outstanding performance has currently captured the attention of the EU community since 2004, with Malta's EU membership and also within world-wide reports and indicators, such as those published by the WEF (2014a) (Appendix 1), among other rating companies, such as Eurostat, Standards and Poor's and Fitch Ratings respectively.

The Maltese manufacturing has been classified by the WEF within an innovative-driven phase (WEF, 2014a), which is the final stage in the three development stages of global competitiveness. As a result, the Maltese economy is considered as a country with substantial potential and economic prosperity, especially in business sophistication and innovation, such as manufacturing. In fact, Malta's SMEs, currently are undertaking substantial investment in specialised manufacturing, which in turn are increasing the commitment to training and development of all employees (Ernst & Young, 2013). In the global scenario, SMEs have always been considered to have a pivotal role within the economy of most emerging countries in the generation of employment and economic growth (Demirbag et al., 2006). Furthermore, SMEs play a key role within SC set-ups by taking various roles, such as suppliers, distributors, producers and customers (Hong & Jeong, 2006). Though, one cannot exclude the challenging reality subjected to the Maltese SMEs within its specific contextual conditions, vis a vis the small local market, the geographic isolation and the lack of economies of scales. Such contextual conditions are contributing to the negative repercussions to all local SMEs, especially in their negotiations of transactions with the global suppliers and SC actors, due to the relatively low order levels, which are sometimes well below their acceptable minimum orders. Such a small economy is also contributing to the relative high fixed cost attributed to logistics in all supply chain transportation requirements due to the low order volumes.

As a result, considering such issues and challenges, it is of pivotal importance for policy makers to serve as a supporting platform for all SCI initiatives to be adopted by SMEs by listening to all stakeholders' requirements, which are both directly and indirectly related to the manufacturing and supply chain operations. Such an approach will establish the overall and specific needs of the industry and the public community, which also needs the direct and active involvement of representatives from the key academic communities to bridge the gap between the practitioners and the academics. The

interaction process of all stakeholders shall not be based on one-off solutions, through say launch of white papers, but the process need to entail and establish an on-going dialogue, through various sources of communication with all involved actors. Such an approach could be achieved by: organising face to face meetings and events; establishing committees made up from the key industry's stakeholders with the relevant Government officials to set-up public and private partnerships; and deploying on line forums to encourage participation of key stakeholders, based on various communities, such as local, EU and worldwide representatives, since nowadays the industry is borderless and runs all across the globe. Such a stance is needed so that all policies and innovative development programmes (e.g. FP 7) should be developed with the right and professional approach and with the full collaboration between the industry practitioners, the policy makers and the academic communities, so as to meet all societal challenges with a sustainable approach, in line with the social, economic and environmental measures to promote SMEs competitiveness.

In a nutshell, the SCI initiatives, from the practitioners' perspective may lead to various improvements, where the key ones are: synergy in all collaborative initiatives between SC actors to achieve outstanding competitive performance across the SC; lead time effective management with more dedicated effort with all SC members to meet or exceed competitive performance benchmarks; employ an effective waste management approach in all resources usage, such as actions at source and a dedicated commitment for the reduction of tasks' duplication to promote efficiencies; improved and proactive customer support in all interactions to promote an effective customer service; ownership of ideas and responsibilities to enhance a motivated workforce with improvements in productivity to achieve best practices and innovation; supply management cost-savings measures through better negotiation of purchasing by gaining both better prices and credit terms to achieve economic sustainability; and implementing sustainable processes across all the SC tiers to create a positive social climate at the workplace and to achieve a sustainable product to the end customer.

8.6 Research limitations

The research offered a credible, valid and trustworthy substantive theory based on a GTM, as already referred in the methodology Chapter 3, which is attributed to the several precautionary measures to promote such research rigour in line with the scientific research agenda. Though, this research work is subjected to different sources of limitations, which in some way or another, could not be avoided, as referred by the salient issues below:

First, the research focuses on the Maltese SMEs manufacturing sector. Hence all generalisations to the theory may not be applicable outside this manufacturing sector environment, such as services and not even within other contexts, such as in other countries, even within other Small Island States.

Second, the sample is limited, since not all SC members could be included for every focal firm understudy so as to cover the whole SC. If they were included, more rich data could have been available, so as to refine or validate further the theorising process. Such a shortcoming could not be completely solved, since one of the reasons was attributed to the fact that some SC actors forming up part of the focal firm's SC (e.g. suppliers, upstream and downstream distributors and customers) were not accessible, being located in a foreign country. Another reason was attributable to the limited available research time-span, since the data collection was not based on a one-off exercise period from a number of participants, but consisted from an iterative process of almost two years of data collection and analysis, consisting from face-to-face meetings, where the good rapport with the participant was considered pivotal to capture rich data.

Third, the research did not identify the theoretical effect of each particular leadership style from the four established traits and neither developed further theory on the effectiveness of each trait individual leadership elements (e.g. inspirational, charisma) that contributed to each competitive capability, but considered the overall competitive capabilities from a holistic perspective.

Fifth, in some cases data collection was limited due to the controlled and limited access given, since for some firms, the data obtained had to rely on a single informant, who may have added a level of subjectivity and bias to the research response, since a single person could have a limited oversight of all the SC operations, which are complex and span several functional areas. Other limitations for capturing data from the firms, was attributed to the fact that some firms declined my request for participation, stating that they were not interested to participate at all. The latter approach is part of the ethical research process.

Sixth, this study theoretical richness may have been limited by the use of a cross-sectional research design, which only captured a snapshot of what the data were saying at one moment in time. Because SCI is a long-term strategic commitment and not a quick fix approach between all customers, suppliers and manufacturers, such a SCI is developed over time and is also function of different product lifecycle stages. As a result a longitudinal study would have captured different levels of SCI across a timeline and would have rendered more rich data based on more variations in the integrative capabilities and the type of integration adopted approach between SC actors. Such a shortcoming to overcome would have been very challenging since it would have entailed a much longer research data collection period, which is very limited in such studies.

Eight, the research has produced a parsimonious model of a set of antecedents and outcomes of interorganizational conceptual elements that constitute a SC made up from the focal firm and a number of stakeholders, embedded within the contextual conditions of the focal firm. Such a holistic model comprises various concepts which include a theory based on substantial breadth but the in-depth explanations were mainly focused on the six main categories (i.e. the main constructs) and not on the sixteen sub-categories, within the holistic SCI management approach, due to the research limitation of word count.

Ninth, the research dedicated limited depth regarding the servitization of manufacturing (Lightfoot et al., 2013) or integrated solutions of manufacturing (Davies et al., 2006) since within the emerged theory, the customer service emerged as one of the sub-themes, although currently such a concept is considered pivotal in the literature, due to the current customer focused initiatives being undertaken by SMEs for improved and value-added customer services.

Tenth, the research with the deployment of the systematic approach, in line with the GTM techniques and procedures (Strauss & Corbin, 1998; Corbin & Strauss, 2008), may have stifled the theory emerging process. In fact the process, in several instances, needed an iterative analytic stance, to shift bi-directionally, from one stage to another. Though, such a methodological and organised approach was found very useful and practical due to researcher's embedded approach on how he treats all his challenges and issues, both when he was a practitioner for fifteen years, as a Senior Engineer and currently, as an academic, as a Senior Lecturer for last nine years.

8.7 Suggestions for future research

The substantive theory with its theoretical model is very comprehensive and consists from a multiplicity of factors, which have a varying degree of impact on SCI, which were not quantified. Hence future research may study such concepts, to derive their receptive weighted significance to the SCI initiatives to achieve competitive capabilities.

The proposed framework, together with the seven propositions (Table 8.1), shall be used to direct future researchers to investigate the role of all conceptual elements relationship, through a large scale study from a quantitative approach, to enhance its validity and generalizability and also to sustain or invalidate such a substantive theory with its theoretical framework. The sample under such research shall incorporate other contexts or adopt a cross-cultural study within EU or across different worldwide countries. Furthermore, such research may be also focused on a set of Small Island States to derive the commonalities and differences in such homogenous type of countries, such as Malta.

Future research may determine the particular leadership style (e.g. transformational) and/or leadership elements (e.g. inspirational and participative) needed to achieve SCI with each respective competitive capability.

It will be fruitful for future research to examine the evolution of SCI patterns across time by undertaking a longitudinal type of research design.

Future research may focus more on the individual emerged sub-categories in relation to the inductively derived holistic SCI management approach within this study, since due to the research limitations of word count, there was lack of space to include such an analytic depth in such sub-themes.

Future research needs to explore the customer service approach within the context of research focused on the servitization of manufacturing (Lightfoot et al., 2013) or integrated solutions of manufacturing (Davies et al., 2006).

8.8 Summary of the conclusion

This chapter highlighted the generated substantive theory, the research significance and the roadmap for implementing such an emerged theory in practice (Appendix 12).

The theoretical contribution is based on the identification of the SCI theory with its key factors, which are required to achieve competitive capabilities for the Maltese SMEs within manufacturing. Such contributions are supported by seven propositions for furthering the theory within SCI through future quantitative positivistic research approach, thus making the research even more robust. It is to be noted, that in reality a perfect inter-firm integrated environment with equitable sharing of mutual benefits, between all SC actors, to achieve all competitive capabilities, does not exist. Such theoretical contributions mainly provided: an original contribution, through the IMLA and the emerged theory with its theoretical framework; a unique research focused on the SCI within the Maltese manufacturing sector; an answer to the literature gaps associated with both the holistic SCI models and also to the management and leadership concepts within the SCM literature; and an added conversation to the extant literature vis a vis sustainable SCM, technology, lean management and competitive capabilities.

The methodological contribution is based on the fact that it extends the applicability of grounded theory based research within SCM and related OM disciplines by answering to the call for such a GTM methodology usage within operations management research and also highlights the need that the GTM systematic process also needs a level of flexibility to conceptualise and promote theoretical abstraction.

The practical contribution is based on the fact that the Maltese SMEs have a substantial number of opportunities, in both the local and global scenario, since Malta is competing with both EU and other World-wide countries, having an innovative-driven economy within manufacturing. This scenario shall encourage all local SMEs to implement SCI initiatives to exploit from all the competitive advantages offered by such integrated efforts with all SC actors. Such SCI initiatives in-conjunction with the high-value manufacturing investments, shall promote both higher revenues and also an improvement on the manufacturing sector contribution to Malta's GDP. This cannot be achieved without the support and involvement of the local Government and EU through incentives and investment schemes.

This chapter also examined the limitations of the research, which highlighted the key shortcomings which were the research relative small sample size, the nature of the cross-sectional research design, the restricted access to multiple participants and to a wider set of SC actors. Finally, it was also outlined the opportunities to conduct future research to extend this thesis study and also to refine the contemporary extant literature on the SCI phenomenon, in particular to: undertake in-depth research in certain concepts, that could not be unpacked by this research so as to understand the underpinning theory behind them, such as servitization of manufacturing; conduct cross-cultural studies within EU or across different worldwide countries, with the inclusion of Small Island States to promote further research generalizability; to examine the evolution of SCI patterns across time by undertaking a longitudinal research approach; and to establish from the comprehensive and complex set of dimensions, within the holistic SCI management approach, the weighted significance to the SCI substantive theory by distinguishing the level of importance of each factor to achieve competitive capabilities.

List of references

Abdullah, Z., & Musa, R. (2014). The Effect of Trust and Information Sharing on Relationship Commitment in Supply Chain Management. *Procedia - Social and Behavioral Sciences*, 130, 266 – 272.

Adler, P. S. (1995). Interdepartmental interdependence and coordination: the case of the design/manufacturing interface. *Organization Science*, 6 (2), 147–167.

Agarwal, A., Shankar, R., & Tiwari, M. K. (2007). Modeling agility of supply chain. *Industrial Marketing Management*, 36(4), 443-57.

Ageron, B., Gunasekaran, A., Spalanzani, A. (2012). Sustainable supply management: an empirical study. *International Journal of Production Economics*, 140, 168–182.

Ajmera, A., & Cook, J. (2009). A multi-phase framework for supply chain integration. *Advanced Management Journal*, 74(1), 37-47.

Alfalla-Luque, R., Medina-Lopez, C., & Dey, P. K. (2012). Supply chain integration framework using literature review. *Production Planning & Control*, 1–18.

Almuiet, M. Z., & Salim, J. (2013). Knowledge Flow in Supply Chain Manufacturing: Case Study in Food Manufacturing Firm. *Procedia Technology*, 11, 463 – 470

Alvarez, G., Pilbeam, C., & Wilding, R. (2010). Nestlé Nespresso AAA sustainable quality program: An investigation into the governance dynamics in a multi-stakeholder supply chain network. *Supply Chain Management: An International Journal*, 15(2), 165–182.

Ambrose, E., Marshall, D., & Lynch, D. (2010). Buyer supplier perspectives on supply chain relationships. *International Journal of Operations and Production Management*, 30(12), 1269–1290.

Amundson, S, D. (1998). Relationships between theory-driven empirical research in operations management and other disciplines. *Journal of Operations Management*, 16, 341–359.

Anderson, E. W., Fornell, C. & Lehmann, D.R. (1994). Customer satisfaction, market share, and profitability: findings from Sweden. *Journal of Marketing*, 58(3), 53-66.

Armstrong, M. (2006). A handbook of human resources Management Practice (10th ed.). London: Kogan Page.

Aronsson, H. & Brodin, M. H. (2006). The environmental impact of changing logistics structures. *International Journal of Logistics Management*, 17(3), 394-415

Ashby, A., Leat, M., & Hudson-Smith, M. (2012). Making connections: a review of supply chain management and sustainability literature. *Supply Chain Management: An International Journal*, 17(5), 497-516.

Avolio, B. J., & Gardner, W. L. (2005). Authentic leadership development: Getting to the root of positive forms of leadership. *The leadership Quarterly*, 16(3), 315-338.

Ajmera, A., & Cook, J. (2009). A multi-phase framework for supply chain integration. *Advanced Management Journal*, 74(1), 37-47.

Angelou Economics (2010). Retrieved December, 21, 2010 from: <u>http://www.timesofmalta.com/articles/view/20100323/local/malta-urged-to-focus-on-seven-industry-target-for-growth.299583.</u> Antonakis, J., Avolio, B. J., & Sivasubramaniam, N. (2003). Context and leadership: An examination of the nine-factor full-range leadership theory using the Multifactor Leadership Questionnaire. *Leadership Quarterly*, 14, 261-295.

Ardichvili, A., & Manderscheid, S. V. (2008). Emerging Practices in Leadership Development. (Special issue). Advances in Developing Human Resources, 10(5), 619-631.

Avolio, B. (1999). Full leadership development. Thousand Oaks, CA: Sage.

Bagchi, P. K., Ha, C. B., Skjoett-Larsen, T., & Soerensen, B. L. (2005). Supply chain integration: a European survey. *The International Journal of Logistics Management*, 16(2), 275-294.

Baines, T., Lightfoot, H., & Evans, S. (2007). State-of-the-art in product-service systems. *Proceedings IMechE.*, Vol. 221 Part B: Journal of Engineering Manufacturing, 1543-1552.

Baines, T., Lightfoot, H., & Peppard, J. (2009). Towards an operations strategy for product-centric servitization. *Journal of Operations & Production Management*, 29(5), 494-519.

Bajec, P., & Jakomin, I. (2010): A Make-or-Buy Decision Process for Outsourcing. *Distribution Logistics Review*. Promet – Traffic & Transportation, 22(4), 285-291.

Baloh, P., Jha, S., & Awazu, Y. (2008). Building strategic partnerships for managing innovation outsourcing. *Strategic Outsourcing: An International Journal*, 1(2), 100–121.

Banker, R. D., Chang, H., & Kao, Y. (2010). Evaluating cross-organizational impacts of information technology – an empirical analysis. *European Journal of Information Systems*, 19, 153-167.

Barney, J. B. (1991). Firm resources and sustained competitive advantage. *Journal of management*, 17, 99-120.

Barua, A., Konana, P., Whinston, A.B., & Yin, F., (2004). An empirical investigation of Net-enabled business value. MIS Quarterly, 28(4), 585–620.

Baratt, M. (2004). Understanding the meaning of collaboration in supply chain. Supply Chain Management, 9(1), 30-42.

Barratt, M., & Oliviera, A., (2001). Exploring the experiences of collaborative planning initiatives, International, Journal of Physical Distribution and Logistics Management. 31(4), 266–289.

Barringer, B., & Harrison, J. (2000). Walking a tightrope: creating value through inter-organisational relationships. *Journal of Management*, 26(3), 367-403.

Bass, B. M. (1990). Bass and Stogdill's handbook of leadership. New York: Free.

Bass, B. M. (1997). Does the transactional – transformational leadership paradigm transcend organizational and national borders? *American Psychologist*, 52, 130–9.

Bass, B. M., & Avolio, B. J. (1994). *Improving Organizational Performance Through Transformational Leadership*. Sage Publications.

Bass, B. M., & Avolio, B. (2000). *Multifactor leadership questionnaire*. Redwood City, CA: Mind Garden.

Bass, B.M., Avolio, B.J., Jung, D.I., & Berson, Y. (2003). Predicting unit performance by assessing transformational and transactional leadership. *Journal of Applied Psychology*, 88, 207–18.

Bayraktar, E., Demirbag, M, S. C., Lenny, K. S. C., Tatoglu, E., & Zaim, H. (2009). A causal analysis of the impact of information systems and supply chain management practices on operational performance: Evidence from manufacturing SMEs in Turkey. *International Journal of Production Economics*, 122, 133–149.

Bell, J. (2009). Doing your research project. (4th ed.). England: Open University Press.

Benton, W. C., & Maloni, M. (2005). The influence of power driven buyer/seller relationships on supply chain satisfaction. *Journal of Operations Management*, 23(1), 1–22.

Benner, M. J., & Tushman, M. L. (2003). Exploitation, Exploration, and Process Management: The Productivity Dilemma Revisited. *Academy of Management Review*, 28(2) 238–256.

Bennis, W. (2007). The challenges of leadership in the modern world: Introduction to the special issue. *American Psychologist*, 62(1), 2-5.

Berasategi, L., Arana, J., & Castellano, E. (2011). A comprehensive framework for collaborative networked innovation. Production Planning and Control: The Management of Operations, 22(5–6), 581–593.

Berrone, P., Cruz, C., Gomez-Mejia, L. R., & Larraza, M. (2010). Socio emotional wealth and organizational response to institutional pressures: do family controlled firms pollute less? *Administrative Science Quarterly*, 55, 82–113.

Bertram, G. (2004). On the convergence of small island economies with their metropolitan patrons. *World Development*, 32(2), 343–364.

Bessant, J., Kaplinsky, R., & Lamming, R. (2003). Putting supply chain learning into practice. *International Journal of Operations and Production Management*, 23(2), 167–184.

Bharadwaj, A. S. (2000). A resource-based perspective on information technology capability and firm performance: an empirical investigation. *Management Information Systems Quarterly*, 24(1), 169–19.

Bijlsma-Frankema, K., & Costa, A. C. (2005). Understanding the trust–control nexus. *International Sociology*, 20(3), 259–282.

Binder, M., & Clegg, B. T. (2006). A conceptual framework for enterprise management. *International. Journal of Production Research*, 44(18/19), 3813-3829.

Binder, M., & Clegg, B. T. (2007). Enterprise management: a new frontier for organisations. *International Journal of Production Economics*, 106(2), 409-430.

Binder, M., & Edwards, J. S. (2010). Using grounded theory method for theory building in operations management research. A study on inter-firm relationship governance. *International Journal of Operations & Production Management*, 30(3), 232-259.

Binder, M., Gust, P., & Clegg, B. (2007). The importance of collaborative frontloading in automotive supply networks. Journal of Manufacturing Technology Management, 19(3), 315–333.

Birks, M J., Chapman, Y B., & Francis, K. L. (2007) Grounded theory methodology in action: transforming method in exploring outcomes for baccalaureate nursing graduates in Malaysian Borneo. In: Morales, A. (ed), *Distance Education Issues and Challenges*. New York: Nova pp. 45–63.

Birks, M., Mills, J., Francis, K., & Chapman, Y. (2009). A thousand words paint a picture: The use of storyline in grounded theory research. *Journal of Research in Nursing*, 14, 405.

Birks, M., & Mills, J., (2011). Grounded theory. A practical guide. London: SAGE Publications

Birou, L. M., Fawcett, S. E., & Magnan, G. M. (1998). The product life cycle: a tool for functional strategic alignment. *International Journal of Purchasing and Materials Management*, 34(2), 37–51.

Blake, R. R., & Mouton, J. S. (1985). The Managerial Grid III. Gulf Publishing Company.

Blome, C., Schoenherr, T., & Eckstein, D. (2014). The impact of knowledge transfer and complexity on supply chain flexibility: A knowledge-based view. *International Journal of Production Economics*, 147, 307–316.

Bluhm, D. J., Harman, W., Lee. T. W., & Mitchell, T. R. (2010). Qualitative Research in Management: A Decade of Progress. Journal of Management Studies, doi: 10.1111/j.1467-6486.2010.00972.x.

Boal, K., & Hooijberg, R. (2000). Strategic leadership research: Moving on. *The Leadership Quarterly*, 11(4), 515-549.

Boardman, J. T., & Clegg, B. T. (2001). Structured engagement in the extended enterprise. *International Journal of Operations and Production Management*, 21, 795–811.

Bojarski, A. D., Laínez, J. M., Espuna, A., & Puigjaner, L. (2009). Incorporating environmental impacts and regulations in a holistic supply chains modeling: An LCA approach. *Computers and Chemical Engineering*, 33, 1747–1759.

Bolstorff, P., & Rosenbaum, R. (2007). *Supply chain excellence. A handbook for dramatic improvement using SCOR Model* (2nd ed.). US: American Management Association.

Bontis, N. (2002). *Managing organizational knowledge by diagnosing intellectual capital: Framing and advancing the state of the field*. In: *World Congress on Intellectual Capital Readings*, Bontis, N. (Ed.), Butterworth-Heinemann, Boston, MA, pp.13-56.

Borys, B., & Jemison, D. B. (1989). Hybrid arrangements as strategic alliances: theoretical issues in organizational combinations, *The Academy of Management Review*, 14(2), 234-249.

Bouchard, T. J. (1976). Field research methods: Interviewing, questionnaires, participant observation, systematic observation, unobtrusive measures. *In:* DUNETTE, M. D. (ed.). *Handbook of industrial and organizational psychology*. Chicago: Rand McNally, p.363-413.

Bowersox, D. J. (1997). Integrated Supply Chain Management: A Strategic Imperative, in Annual Conference Proceedings. Council of Logistics Management. Chicago, Illinois, 181–189.

Bowersox, D. J., & Calantone, R. J. (1998). Executive insights: global logistics. *Journal of International Marketing*, 6(4), 83-93.

Bowersox, D. J., Closs, D. J., & Stank, T. P. (2000). Ten mega-trends that will revolutionize supply chain logistics. *Journal of Business Logistics*, 21(2), 1–16.

Bowman, C., & Ambrosini, V. (2000). Value Creation Versus Value Capture: Towards a Coherent Definition of Value in Strategy, *British Journal of Management*, 11, 1 - 15.

Braunscheidel, M. J., & Suresh, N. C. (2009). The organizational antecedents of a firm's supply chain agility for risk mitigation and response. *Journal of Operations Management*, 27(2), 119–140.

Braunscheidel, M. J., Suresh, N. C., & Boisnier, A. D. (2010). Investigating the impact of organisational culture on supply chain integration. *Human Resource Management*, 49(5), 883 – 911

Brundtland Commission (1987). Our common future, report of the World Commission on Environment and Development. Retrieved April 9, 2015, from: http://www.un-documents.net/our-common-future.pdf.

Bryan, J. R. J., Sinkovics, R. R., & Kim, D. (2010). Drivers and Performance Outcomes of Relationship Learning for Suppliers in Cross-Border Customer–Supplier Relationships: The Role of Communication Culture *Journal of International Marketing*, 18(1), 63–85.

Bryant, A. (2002). Re-Grounding Grounded Theory, *JITTA: Journal of Information Technology Theory and Application*; 4, 1.

Bryman, A. (2008). Social Research Methods. (3rd ed.). Oxford: University Press.

Brynjolfsson, E., & Yang, S. (1996). Information technology and productivity: a review of literature. *Advances in Computers*, 43, 179–214.

Burns, J. M. (1978). Leadership. New York: Harper & Row, 1978.

Buttery, E. A., & Leung, T. K. P. (1998). The difference between Chinese and Western negotiations. *European Journal of Marketing*, 32(3/4), 374-89.

Buyukozkan, G., & Berkol., C. (2011). Designing a sustainable supply chain using an integrated analytic network process and goal programming approach in quality function deployment. *Expert Systems with Applications*, 38, 13731–13748.

Cachon, G. P., & Fisher, M. (2000). Supply chain management and the value of shared information. *Management Science*, 46(8), 1032–1048.

Cachon, G. P., Randall, T., & Schmidt, G. M. (2007). In search of the bullwhip effect. *Manufacturing & Service Operations Management*, 9(4), 457–479.

Cadden T., Marshall, D., & Cao, G. (2013). Opposites attract: organisational culture and supply chain Performance. *Supply Chain Management: An International Journal*, 18(1), 86 - 103

Cagliano, R., Caniato, F., & Spina, G. (2006). The linkage between supply chain integration and manufacturing improvement programmes. *International Journal of Operations and Production Management*, 26(3), 282–299.

Calantone, R. J., Cavusgila, T., & Zhao, Y. (2002). Learning orientation, firm innovation capability, and firm performance. *Industrial Marketing Management*, 31, 515–524.

Cambra-Fierro, J., & Ruiz-Benite, R. (2009). Advantages of intermodal logistics platforms: insights from a Spanish platform. *Supply Chain Management: An International Journal*, 14(6), 418–421.

Camerinelli, E. (2009). *Measuring the Value of the Supply Chain: Linking Financial Decisions with Supply Chain Performance*. Gower Publishing, Burlington, VT.

Caniels, M. C. J., & Gelderman, C. J. (2007). Power and interdependence in buyer-seller relationships: A purchasing portfolio approach. *Industrial Marketing Management*, 36, 219–229.

Carr, A. S., & Smeltzer, L. R. (1999). The relationship among purchasing benchmarking, strategic purchasing, firm performance, and firm size. *The Journal of Supply Chain Management*, 35(4), 51–60.

Carson, S. J., Madhok, A., & Wu, T. (2006). Uncertainty, opportunism, and governance: The effects of volatility and ambiguity on formal and relational contracting. *Academy of Management Journal*, 49(5), 1058–1077.

Caniels, M. C. J., Gehrsitz, M. H., & Semeijn, J. (2013). Participation of suppliers in greening supply chains: An empirical analysis of German automotive suppliers. *Journal of Purchasing & Supply Management*, 19, 134–143.

Carmichael, D. G., & Balatbat, M. C. A. (2010). A contractor's analysis of the likelihood of payment of claims. *Journal of Financial Management of Property and Construction*, 15(2), 102-17.

Carson, J. B., Tesluk, P. E., & Marrone, J. A. (2007). Shared leadership in teams: An investigation of antecedent conditions and performance. *Academy of Management Journal*, 50, 1217–1234.

Carter, P., Carter, J., Monczka, R., Slaight, T., & Swan, A. (2000). The Future of Purchasing and Supply: A Ten-year Forecast. *Journal of Supply Chain Management*, 36(1), 14-26.

Carter, C. R., & Easton, P. L. (2011). Sustainable supply chain management. *International Journal of Physical Distribution & Logistics Management*, 41(1), 46–62.

Carter, C. R., & Rogers, D. S. (2008). A framework for sustainable supply chain manage- ment: moving toward new theory. *International Journal of Physical Distribution and Logistics Management*, 38(5), 360–387.

Cartwright, S., & Cooper, C. L. (1993). The role of cultural compatibility in successful organisations marriage. *Academy of Management Executive*, 7(2), 57-70.

Cassivi, L. (2006). Collaboration planning in a supply chain. Supply Chain Management: An International Journal, 11(3), 249-58.

Cavinato, J. L. (1999). Fitting purchasing to the five stages of strategic management. European *Journal of Purchasing and Supply Management*, 5(2), 75–83.

Cavusgil, S., Tamer, R., Calantone, J., & Zhao, Y. (2003). Tacit knowledge transfer and firm innovation capability. *Journal of Business and Industrial Marketing*, 18(1), 6–21.

Chan, F. T. S., & Qi, H. J. (2003). Feasibility of performance management system for supply chain: a process-based approach and measures. *Integrated manufacturing systems*, 14(3), 179-190.

Chandra, C., & Kumar, S. (2000). Supply chain management in theory and practice: a passing fad or a fundamental change? *Industrial Management & Data Systems*, 100(3), 100–114.

Chapman, R. L., Soosay, C., & Kandampully, J. (2003). Innovation in logistics services and the new business model: a conceptual framework", *International Journal of Physical Distribution & Logistics Management*, 33(7), 630-650.

Charmaz, K. (2000). Constructivist and objectivist grounded theory. In: N.K. Denzin & Y.Lincoln (Eds). *Handbook of Qualitative Research*. (2nd ed.). Thousand Oaks, CA: Sage, pp. 509-535.

Charmaz, K. (2006). Constructing Grounded Theory: A Practical Guide Through Qualitative Analysis, London: Sage.

Chatman, J. (1989). Improving interactional organizational research: A model of person-organization fit. *Academy of Management Review*, 14, 333-349.

Chen, I. J., & Paulraj, A. (2004). Towards a theory of supply-chain management: the constructs and measurement. *Journal of Operations Management*, 22(2), 119–150.

Chen, I. J., Paulraj, A., & Lado, A. A. (2004). Strategic purchasing, supply management, and firm performance. *Journal of Operations Management*, 22(5), 505–523.

Chen, M. C., Yang, T., & Li, H. C. (2007). Evaluating the SC performance on IT based interenterprise collaboration. *Information and Management*, 44, 524–534.

Chen, H., Daugherty, P. J., & Landry, T. D. (2009). Supply chain process integration: a theoretical framework. *Journal of Business Logistics*, 30(2), 27-46.

Chen, H. L. (2012). Empirical behavioral analysis of project contractors' supply-chain payment terms. *Supply Chain Management: An International Journal*, 17 (3), 277 - 289

Cheng, J. H., Yeh, C. H., & Tu, C.W. (2008). Trust and knowledge sharing in green supply chains. *Supply Chain Management: An International Journal*, 13(4), 283–295.

Chesbrough, H. W., & Teece, D. J. (2002). Organizing for innovation: when is virtual virtuous? *Harvard Business Review*, 8(8), 127.

Cheung, C.F., Cheung, C.M., & Kwok, S. K. (2012). A Knowledge-based Customization System for Supply Chain Integration. *Expert Systems with Applications*, 39, 3906–3924.

Chicksand, D. (2009). Creating a sustainable UK farming and food industry: An analysis of partnership thinking as a solution to the problems in the UK farming and food industry. PhD Thesis Birmingham: The University of Birmingham.

Chicksand, D. (2015). Partnerships: The role that power plays in shaping collaborative buyer–supplier exchanges. *Industrial Marketing Management*. Doi: http://dx.doi.org/10.1016/j.indmarman.2015.03.019.

Childerhouse, P., & Towill, D. (2000). Engineering supply chains to match customer requirements. *Logistics Information Management*, 13(6), 337-45.

Cho, D. W., Lee, Y. H., Ahn, S. H., & Hwang, M. K. (2012). A framework for measuring the performance of service supply chain management. *Computers & Industrial Engineering*, 62, 801–818

Choi, T. (1995). Conceptualising continuous improvement: Implications for organisational change, Omega, *International Journal of Management Science*, 23(6), 607-624.

Choi, T. Y., & Hartley, J. L. (1996). An exploration of supplier selection practices across the supply chain. Journal of Operations Management, 14(4), 333-43.

Chopra, S., & Meindl, P. (2001). *Supply Chain Management: Strategy, Planning, and Operation*. New Jersey: Prentice Hall.

Chong, M. Y., Chin, J, F., & Hamzah, H. S. (2012). Transfer of total productive maintenance practice to supply chain. *Total Quality Management*, 23(4), 467–488.

Christopher, M. (2000). The agile supply chain – competing in volatile markets. *Industrial Marketing Management*, 29(1), 37-44.

Christopher, M., & Holweg, M. (2011). Supply Chain 2.0: managing supply chains in the era of turbulence. *International Journal of Physical Distribution & Logistics Management*, 41(1), 63-82.

Ciborra, C. U. (1997). De Profundis? Deconstructing the concept of strategic alignment. *Scandinavian Journal of Information Systems*, 9(1) 57–82.

Clark, K. B. (1996). Competing through manufacturing and the new manufacturing paradigm: is manufacturing strategy passé? *Production and Operations Management*, 5(1), 42–58.

Clegg, B., Chandler, S., Binder, M., & Edwards, J. (2012). Governing inter-organizational R&D supplier collaborations: a study at Jaguar Land Rover. *Production Planning & Control*, pp. 1-19, iFirst.

Clemons, E. K., Reddi, S. P., & Row, M. C. (1993). The impact of information technology on the organization of economic activity: The 'move to the middle' hypothesis. *Journal of Management Information Systems*, 10(2), 9–35.

Clegg, B., T., & Wan, Y. (2013). ERP Systems and Enterprise Management Trends: a Contingency Model for the Enterprization of Operations. *International Journal of Operations and Production Management*, 33, (11/12), 1458-1489.

Co, H. C., & Barro, F. (2009). Stakeholder theory and dynamics in supply chain collaboration *International Journal of Operations & Production Management*, 29(6), 591-611.

Coase, R. H. (1937). The nature of the firm. *Economica*, 4(3), 386–405.

Cohen, M. A., & Lee, H. L. (1988). Strategic analysis of integrated production distributed systems: models and methods. *Operational Research*, 36(2), 216–28.

Cokins, G. (2003). Measuring profits and costs across the supply chain for collaboration. *Cost Management*, 17(5), 22-29.

Cooper, M. C., Lambert, D. M., & Pagh, J. D. (1997). Supply Chain Management: More Than a New Name for Logistics. *The International Journal of Logistics Management*, 8(1), 1–13.

Copacino, W. C., & Raton, B. (1997). *Supply Chain Management: The Basics and Beyond*. FL: St. Lucie Press/APICS Series on Resource Management.

Corbin, J., & Strauss, A. (2008). Basics of Qualitative Research, Techniques and Procedures for Developing Grounded Theory. (3rd ed.). CA: Sage.

Cordeiro, J., & Sarkis, J. (1997). Environmental proactism and firm performance: evidence from security analyst forecasts. *Business Strategy and Environment*, 6(2), 104–114.

Costantino, N., & Pellegrino, R. (2010). Choosing between single and multiple sourcing based on supplier default risk: A real options approach. *Journal of Purchasing & Supply Management*, 16, 27-40.

Cousins, P. D., & Menguc, B. (2006). The implications of socialization and integration in supply chain management. *Journal of Operations Management*, 24(5), 604–620.

Cousins, P., Lawson, B., & Squire, B. (2008). Performance measurement in strategic buyer supplier relationships: the mediating role of socialisation mechanisms. *International Journal of Operations and Production Management*, 28(3), 238-58.

Cowan, K., Paswan, A, K., & Steenburg, E. V. (2015). When inter-firm relationship benefits mitigate power asymmetry. *Industrial Marketing Management*. http://dx.doi.org/10.1016/j.indmarman.

Cox, A. (1998). Clarifying complexity, Supply Management, 28 January.

Cox, A. (1999). Power, value and supply chain management. Supply Chain Management: An International Journal, 4(4), 167-75.

Cox, A., Sanderson, J., & Watson, G. (2001) Supply chains and power regimes: Toward an analytic framework for managing extended networks of buyer and supplier relationships. *Journal of Supply Chain Management*, Spring, 37, 2.

Cox, A., Londsdale, C., Watson, G., & Qiao, H. (2003). Supplier relationship management: A framework for understanding managerial capacity and constraints. *European Business Journal*, 15(3), 135–145.

Cox, A. (2004). *Strategic outsourcing: avoiding the loss of critical assets and the problems of adverse selection and moral hazard*. Business Briefing: Global Purchasing & Supply Strategies, IFPMM, Business Briefings Ltd., London.

Cox, A., Watson, G., Lonsdale, C., & Sanderson, J. (2004a). Managing appropriately in power regimes: relationship and performance management in 12 supply chain cases. *Supply Chain Management: An International Journal*, 9(5) 357-37.

Cox, A., Lonsdale, C., Sanderson, J., & Glyn Watson, G. (2004b). Business Relationships for Competitive Advantage: Managing Alignment and Misalignment in Buyer and Supplier Transactions. Palgrave Macmillan, Basingstoke.

Cox, A. (2004c). *Win–Win? The paradox of value and interests in business relationships*. London: Earlsgate Press.

Cox, A. (2004d). Business relationship alignment: on the commensurability of value capture and mutuality in buyer and supplier exchange. *Supply Chain Management: An International Journal*, 9(5), 410-420.

Craighead, C.W., Blackhurst, J., Rungtusanatham, M. J., & Handfield, R. B. (2007). The severity of supply chain disruptions: design characteristics and mitigation capabilities. *Decision Sciences* 38(1), 131–156.

Crainer, S. (1998). *Key Management Ideas: Thinkers that Changed the Management World* (3rd ed.). Financial Times: Prentice Hall.

Creswell, J. W. (2007). *Qualitative Inquiry and Research Design: Choosing Among Five Approaches*. (2nd ed.). CA: Sage.

Croom, S., & Romano, P. (2000). Supply chain management: an analytical framework for critical literature review. *European Journal of Purchasing & Supply Management*, 6, 67-83.

Cuevas, J. M., Julkunen, S., & Gabrielsson, M. (2015). Power symmetry and the development of trust in interdependent relationships: The mediating role of goal congruence, *Industrial Marketing Management*. Doi: <u>http://dx.doi.org/10.1016/j.indmarman</u>.

Curran, J., & Blackburn, R. A. (2001). Researching the small enterprise. London: SAGE Publications.

Dahlgaard, J. J., & Dahlgaard-Park, S. M. (2006). Lean production, six sigma quality, TQM and company culture. *The TQM Magazine*, 18(3), 263–281.

Darnall, N., & Sides, S. (2008). Assessing the performance of voluntary environment programs: does certification matter? *Policies Studies Journal*, 36(1), 95-117.

Das, A., & Narasimhan, R. (2000). Purchasing competence and its relationship with manufacturing performance. *The Journal of Supply Chain Management*, 36(2), 17–28.

Das, A., Narasimhan, R., & Talluri, S. (2006). Supplier integration – finding an optimal configuration. *Journal of Operations Management*, 24(5), 563–582.

Dass, M., & Fox, G. L. (2011). A holistic network model for supply chain analysis. *International Journal of Production Economics*, 131, 587–594.

Daugherty, P. J., Stank, T. P. & Ellinger, A. E. (1998). Leveraging logistics/distribution capabilities: the effect of logistics service on market share. *Journal of Business Logistics*, 19(2), 35-52.

D'aveni, R.A., & Ravenscraft, D.J. (1994). Economies of integration versus bureaucracy costs: does vertical integration improve performance? *Academy of Management Journal*, 37(5), 1167–1206.

Davies, A., Brady, T., & Hobday, M. (2006). Charting a path towards integrated solutions. *MIT Sloan Management Review*, 47(3).

Davis, E., W., & Spekman, R., E. (2004). *Extended Enterprise: Gaining Competitive Advantage Through Collaborative Supply Chains*. Financial Times Prentice-Hall, New York, NY.

De Brito, M. P., & Van Der Laan, E. A. (2008). Supply chain management and sustainability: Procrastinating a holistic integration. IEMC – Europe 2008: *International engineering management conference proceedings*, pp. 389–392.

Dechurch, L. A., Hamilton, K. L., & Haas, C. (2007). Effects of conflict management strategies on perceptions of intragroup conflict. *Group Dynamics: Theory, Research, and Practice*, 11(1), 66–78.

Defee, C. C., & Fugate, B. S. (2010). Changing perspective of capabilities in the dynamic supply chain era. *The International Journal of Logistics Management*, 21(2), 180-206.

Defee, C. C., Stank, T. P. T., & Esper, T. (2010). Performance implications of transformational supply chain leadership and followership. *International Journal of Physical Distribution & Logistics Management*, 40(10), 763-791.

Dell, M. S. (1999). Direct from Dell: Strategies that Revolutionized an Industry. New York: Harper Collins

Deming, W. E. (1986). *Out of the Crisis*. Cambridge, Mass.: MIT Press)

Demirbag, M., Koh, S. C. L., Tatoglu, E., & Zaim, S. (2006). TQM and market orientation's impact on SMEs' performance. *Industrial Management & Data Systems*, 106(8), 1206–1228.

Denzin, N. K., & Lincoln Y. S. (2005). *The Sage Handbook of Qualitative Research* (3rd ed). London: Sage Publications.

Determirhan, D., Jacob, V. S., & Raghunathan, S. (2007). Strategic IT investments: the impact of switching cost and declining IT cost. *Management Science*, 53(2), 208–226.

Devaraj, S., & Kohli, R. (2003). Performance impacts of information technology: is actual usage the missing link? *Management Science*, 49(3), 273–289.

Devaraj, S., Krajewski, L., & Wei, C. W. (2007). Impact of eBusiness technologies on operational performance: The role of production information integration in the supply chain. *Journal of Operations Management*, 25, 1199–1216.

DiMaggio, P. J., & Powell, W. W. (1983). The iron cage revisited: institutional isomorphism and collective rationality in organizational fields. *American Sociology Review*, 48(2), 147–160.

Doney, P. M., Barry, J. M., & Abratt, R. (2007). Trust determinants and outcomes in global B2B services. *European Journal of Marketing*, 41(9/10), 1096–1116.

Dos Santos, B., & Sussman, L. (2000). Improving the return on IT investment: the productivity paradox. *International Journal of Information Management*, 20(6), 429–440.

Dreyer, H. C., Alfnes, E., Strandhagen, J. O., & Thomassen, M. K. (2009). Global supply chain control systems: a conceptual framework for the global control centre. *Production Planning & Control*, 20(2), 147-157.

Droge, C., Jayaram, J., & Vickery, S. K. (2004). The effects of internal versus external integration practices on time-based performance and overall firm performance. *Journal of Operations Management*, 22(6), 557–573.

Drucker, P. F. (1977). People and Performance. Heinemann.

Drucker, P. F. (1979). Management. Pan Books.

Duray, R., Ward, P.T., Milligan, G.W., & Berry, W.L. (2000). Approaches to mass customisation: configurations and empirical validation. *Journal of Operations Management*, 18, 605–625.

Dyer, J. H. (1994). Dedicated assets: Japan's manufacturing edge. *Harvard Business Review*. 72, 6 (November-December 1994), 174-178.

Dyer, H. J. (2000). *Collaborative Advantage: Winning through Extended Enterprise Supplier Networks*. New York: Oxford University Press.

Dyer, J. H., & Singh, H. (1998). The relational view: cooperative strategy and sources of inter organizational competitive advantage. *Academy of Management Review*, 23(4), 660–679.

Eckartz, S., Daneva, M., Wieringa, R., & Hillegersberg, J. V. (2009). Cross-organizational ERP management: how to create a successful business case? *SAC'09 Proceedings of the 2009 ACM Symposium on Applied Computing*, Honolulu, HI, USA.

Edmondson, A. C. & McManus, S. E. (2007). Methodological fit in management field research. Academy of Management Review, 32, 1155–79.

EEA (2012). European Environment Agency. Retrieved April, 20, 2015, from: <u>http://www.eea.europa.eu/soer</u>.

EFQM (2003). Assessing for Excellence, Brussels, Belgium. Retrieved March, 3, 2010, from: http:// http://www.efqm.org/EFQM.pdf.

Easterby-Smith, M., Thorpe, R., & Jackson, P.R. (2008). *Management Research* (3rded.). London: Sage

Eisenhardt, K. M. (1989). Building Theories from Case Study Research. Academy of Management Review, 14(4), 532-550.

Eisenhardt, K. M., & Tabrizi, B. N. (1994). Accelerating adaptiveprocesses: product innovation in the global computer industry. *Administrative Science Quarterly*, 40 (1), 84–110.

Elkington, J. (2002). *Cannibals with forks: the triple bottom line of 21st century business*. Oxford: Capstone, Reprint.

Ellinger, A. E., Daugherty, P. J., & Keller, S. B. (2000). The Relationship Between Marketing/Logistics Interdepartmental Integration and Performance in U.S. Manufacturing Firms: An Empirical Study. *Journal of Business Logistics*, 21(1), 1-22.

Ellinger, A. E., & Ellinger, A. D. (2014). Leveraging human resource development expertise to improve supply chain managers' skills and competencies. *European Journal of Training and Development*, 38(1/2), 118-135.

Ellinger, A. E., Scott, B. K., & Hansen, J. D. (2006). Bridging the Divide Between Logistics and Marketing: Facilitating Behaviour. *Journal of Business Logistics*, 27(2), 1-27.

Ellinger, A., Shin, H., Northington, W. M., Adams, F. G., Hofman, D., & O' Marah, K. (2012). The influence of supply chain management competency on customer satisfaction and shareholder value. *Supply Chain Management: An International Journal*, 17(3), 249 – 262.

Ellram, L. M. (1995). Partnering pitfalls and success factors. *International Journal of Purchasing and Materials Management*, 31(2), 36-44.

Ellram, L., Tate, W., & Billington, C. (2004). Understanding and managing the services supply chain. *Journal of Supply Chain Management*, 40(4), 17–32.

Emerson, R. M. (1962). Power-dependence relations. American Sociological Review, 27, 31-41.

Ernst & Young (2013). Market gaps in access to finance and the feasibility of new financing instruments in the EU addressing the credit needs of Maltese businesses. Malta Business Bureau, Malta.

Esper, T. L., Ellinger, A. E., Stank, T. P., Flint, D. J., & Moon, M. (2010). Demand and supply integration: a conceptual framework of value creation through knowledge management. *Journal of the Academy of Marketing Science*, 38(1), 5–18.

ESRC (Economic and social research council), FRE (framework for research ethics). (2010). Retrieved March, 3, 2010, from: http://www.esrcsocietytoday.ac.uk/ESRCInfoCentre/opportunities/research_ethics_framework/

Estampe, D., Lamouri, S., Paris, J.L., & Brahim-Djelloul, S. (2010). A framework for analysing supply chain performance evaluation models. *International Journal of Production Economics*, doi:10.1016/j.ijpe.2010.11.024.

European Commission (2003). Commission recommendation of 6 May 2003 concerning the definition of micro, small and medium sized enterprises. *Official Journal of the European Union*, L124, 1422, 36-41.

European Commission. (2008). Horizon 2020. Retrieved October, 21, 2014, from: <u>http://ec.europa.eu/programmes/horizon2020/</u>

Eurostat. (2014). *Key Figures on Europe*. Luxembourg: Publications Office of the European Union. Retrieved March, 30, 2015, from: <u>http://ec.europa.eu/eurostat/documents/3930297/6309576/KS-EI-14-001-EN-N.pdf</u>.

European Union. Retrieved May, 21, 2013 from: http://ec.europa.eu/eurostat/documents/3930297/5969538/KS-EI-13-001-EN.PDF.

Faraj, S., & Sambamurthy, V. (2006). Leadership of information systems development projects. *IEEE Transactions on Engineering Management*, 53, 238–249.

Fawcett, S. E., & Magnan, G. M. (2004). Ten guiding principles for high-impact SCM. *Business Horizons* 47/5, September-October (67-74).

Fawcett, S. E., Magnan, G. M., & McCarter, M. W. (2008). Benefits, barriers, and bridges to effective supply chain management. *Supply Chain Management: An International Journal*, 13(1), 35-48.

Fawcett, S.E., Osterhaus, P., Magnan, G.M., Brau, J.C., & McCarter, M.W. (2007). Information sharing and supply chain performance: the role of connectivity and willingness. *Supply Chain Management: An International Journal*, 12/5 (2007) 358–368.

Feretti, I., Zanoni, S., Zavanella, L., & Diana, A. (2007). Greening the aluminium supply chain. *International Journal of Production Economics*, 108, 236–245.

Fiedler, F. E. (1967). A theory of leadership effectiveness. New York: McGraw-Hill.

Fine, C. H. (1998). *Clockspeed: Winning Industry Control in the Age of Temporary Advantage*. Perseus Books, MA.

Finkelstein, S., & Hambrick, D. (1996). *Strategic leadership: Top executives and their effects on organizations*. New York: West Publishing Company.

Fisher, L. M. (1996). How strategic alliances work in biotech. *Strategy and Business*, First Quarter, 1-7.

Fisher, L. M. (1997). What is the right supply chain for your product? *Harvard Business Review*, Mar-Apr, 105-116.

Flick, U. (2006). An introduction to qualitative research. (3rd ed.). London: Sage Publications.

Fitzgerald, L., Johnston, R., Brignall, T. J., Silvestro, R., & Voss, C. (1991). Performance measurement in service businesses. London: CIMA.

Flynn, B. B., & Flynn, E. J. (1999). Information-processing alternatives for coping with manufacturing environment complexity. *Decision Sciences*, 17(3), 249–269

Flynn, B.B., & Flynn, E. J. (2004). An exploratory study of the nature of cumulative capabilities. *Journal of Operations Management*, 22(5), 439–458.

Flynn, B., Schroeder, R.G., & Flynn, J. (1999). World class manufacturing: an investigation of Hayes and Wheelwright's foundation. *Journal of Operations Management*, 17(3), 249–269.

Flynn, B.B., Huo, B., & Zhao, X. (2010). The impact of supply chain integration on performance: A contingency and configuration approach. *Journal of Operations Management*, 28, 58–71.

Francis, V. (2008). Supply chain visibility: lost in translation? Supply Chain Management: An International Journal, 13(3), 180-4.

Freeman, C., & Soete, L. (1990). Fast structural change and slow productivity change: some paradoxes in the economics of information technology. *Structural Change and Economics Dynamics*, 1(2).

Freeman, R.E. (1984). Strategic Management: A Stakeholder Approach. Boston, MA: Pitman.

French, J. R. P., & Raven, B. (1968). The Bases of Social Power. In: Cartwright, D. and Zander, A. F. (eds) *Group Dynamics: Research and Theory* (3rd ed.). Harper and Row.

Frohlich, M.T., & Westbrook, R. (2001). Arcs of integration: an international study of supply chain strategies. *Journal of Operations Management*, 19(2), 185–200.

Frohlich, M.T., & Westbrook, R. (2002). Demand chain management in manufacturing and services: web-based integration, drivers and performance. *Journal of Operations Management*, 20, 729–745.

Frosch, R. A., & Gallopoulos, N. E. (1989). Strategies for manufacturing. *Scientific American*, 261(3), 144–152.

Gagnon, Y., & Toulouse, J. (1996). The behavior of business managers when adopting new technologies. *Technological Forecasting and Social Change*, 52(1), 59-74.

Garetti, M., & Taisch, M. (2012). Sustainable manufacturing: trends and research challenges. *Production Planning & Control*, 23(2–3), 83–104

Gebauer, H., Friedli, T., & Fleisch, E. (2006). Success factors for achieving high service revenues in manufacturing companies. *Benchmarking: An International Journal*, 13(3), 374–386.

Gebauer, H., & Fleisch, E. (2007). Managing sustainable service improvements in manufacturing companies. *Kybernetes*, 36 (5/6), 583–595.

Gebauer, H. (2008). Antecedents for the transition from products to services. *Journal of Applied Management and Entrepreneurship*, 13(3), 4–19.

Germain, R., & Droge, C. (1998). The conext organisational design and performance of JIT buying versus non-JIT buying firms. *International Journal of Purchasing & Materials Management*, 34(2), 12-18.

Germain, R., & Iyer, K. N. S. (2006). The interaction of internal and downstream integration and its association with performance. *Journal of Business Logistics*, 27(2), 29–53.

Gimenez, C., Sierra, V., & Rodon, J. (2012). Sustainable operations: Their impact on triple bottom Line, *International Journal of Production Economics*, 140(1) 149–159.

Giunipero, L.C., Denslow, D., & Eltanatwy, R. (2005). Purchasing/supply chain management flexibility: moving to an entrepreneurial skill set. *Industrial Marketing Management*, 34, 602-613.

Glaser, B. G. (1978). Theoretical Sensitivity. Mill Valley, CA: Sociology Press.

Glaser, B. G. (1992). Basics of Grounded Theory Analysis. Mill Valley, CA: Sociology Press.

Glaser, B. G. (1996). Sociology. Press Mill Valley, CA: Sociology Press.

Glaser, B. G. (1998). Doing Grounded Theory: Issues and Discussions. USA: Sociology Press.

Glaser, B. G. (2002). *Constructivist Grounded Theory*? Forum Qualitative: Qualitative Social Research, 3(3), 12.

Glaser, B. G., & Strauss, A. L. (1967). *The Discovery of Grounded Theory: Strategies for Qualitative Research*. Chicago, IL: Aldine Publishing Company.

Glover, J. L., Champion, D., Daniels. K. J., & Dainty A. J. D. (2014). An Institutional Theory perspective on sustainable practices across the dairy supply chain. *International Journal of Production Economics*.

Glyn, W. (2001). Subregimes of power and integrated supply chain management. *Journal of Supply Chain Management*, Spring, 37, 2.

Gold, S., Seuring, S., & Beske, P. (2010). Sustainable supply chain management and interorganizational resources: a literature review. *Corporate Social Responsibility and Environmental Management*, 17(4), 230–245.

Goh, A. (2006). A strategic management framework for leveraging knowledge innovation, *International Journal of the Computer, the Internet and Management,* 14(3), 32–49.

Goldman, S. L., Nagel, R. N., & Preiss, K. (1995). *Agile Competitors and Virtual Organizations: Strategies for Enriching the Customer*. New York: Van Nostrand Reinhold.

Gomm, M. L. (2010). Supply chain finance: Applying finance theory to supply chain management to enhance finance in supply chains. *International Journal of Logistics: Research and Applications*, 13(2), 133–142.

Gonzalez-Benito, J. (2007). A theory of purchasing's contribution to business performance. *Journal of Operations Management*, 25(4), 901–917.

Gonzales-Benito, J., & Gonzales-Benito, O. (2006). The role of stakeholder pressure and managerial values in the implementation of environmental logistics practices. *International Journal of Production Research*, 44(7), 1353-1373.

Gosling, J., Purvis, L., & Naim, M. M. (2010). Supply chain flexibility as a determinant of supplier selection. *International Journal of Production Economics*, 128, 11–21.

Goswami, S., & Mathew, M. (2005) Definition of innovation revisited: An empirical study of Indian information technology industry. *International Journal of Innovation Management*, 9(3), 371–383

Grant, R. (1996). Prospering in dynamically-competitive environments: organizational capability as knowledge integration. *Organization Science*, 7, 375–387.

Green, K. W. Jr., & Inman, R. A. (2005). Using a just-in-time selling strategy to strengthen supply chain linkages. *International Journal of Production Research*, 43(16), 3437-53.

Green, K. W., & Inman, R. A. (2007). The impact of JIT-II-selling on organizational performance. Industrial Management & Data Systems, 107(7), 1018-1035

Greenleaf, R. K. (1970). The servant as leader. Indianapolis, IN: The Robert K. Greenleaf Center.

Gronn, P. (2002). Distributed leadership as a unit of analysis. *The Leadership Quarterly*, 13(4) 423-451.

Green, Jr. K. W., Inman, R. A., Birou, L. A., & Whitten, D. (2014). Total JIT (T-JIT) and its impact on supply chain competency and organizational performance. *International Journal of Production Economics*, 147, 125–135

Gretton, I. (1995). Taking the Lead in Leadership. Professional Manager, January 1995, 20-2.

Grover, V., Teng, J. T. C., & Fiedler, K. D. (2002). Investigating the role of information technology in building buyer-supplier relationships. *Journal of the Association for Information Systems*, 3, 217–245.

Guan, W., & Rehme, J. (2012). Vertical integration in supply chains: driving forces and consequences for a manufacturer's downstream integration. *Supply Chain Management: An International Journal*, 17(2), 187 – 201.

Guba, E. G., & Lincoln, Y. S. (2005). Paradigmatic controversies, contradictions, and emerging confluences. In: *Handbook of qualitative research* (3rd ed.). N. K. Denzin & Y.S. Lincoln, Sage, London, pp. 191-215.

Guillen, G., Badell, M., & Puigjaner, L. (2007) A holistic framework for short-term supply chain management integrating production and corporate financial planning. International. *Journal of Production Economics*, 106, 288–306.

Gulati, R. (1995). Does familiarity breed trust? The implications of repeated ties for contractual choice in alliances. *Academy of Management Journal*, 38, 85–112.

Gunasekaran, A. (1999). Just-in-time purchasing: an investigation for research and applications. *International Journal of Production Economics*, 59(1–3), 77–84.

Gunasekaran, A. (2004). Editorial Supply chain management: Theory and applications. *European Journal of Operational Research*, 159, 265–268.

Gunasekaran, A., Lai, K., & Cheng, T. C. E. (2008). Responsive supply chain: A competitive strategy in a networked economy. *Omega*, 36, 549 – 564.

Gunasekaran, A., & McGaughey, R. (2002). Editorial Information technology/information systems in 21st century manufacturing. *International Journal of Production Economics*, 75, 1–6.

Gunasekaran, A., Patel, C., & Tirtiroglu, E. (2001). Performance measures and metrics in a supply chain environment. *International Journal of Operations & Production Management*, 21 (1/2), 71-87.

Hall, E. T. (1976). Beyond Culture. Garden City, NY: Anchor Press.

Hall, R. W. (1987). Attaining Manufacturing Excellence. Homewood, IL: Dow Jones-Irwin.

Halpern, D. F. (2003). *Thought and knowledge: An introduction to critical thinking* (4th ed.). Mahwah, NJ: Erlbaum.

Hammer, M., & Stanton, S. (1999). How process enterprises really work. *Harvard Business Review*, 77(6), 108-118.

Hammer, M. (2004). Deep change: how operational innovation can transform your company. *Harvard Business Review*, 82(4), 84-96.

Handfield, R. B., & Nichols, E. L. (1999). *Introduction to Supply Chain Management*. New Jersey: Prentice Hall.

Handfield, R., & Bechtel, C. (2002). The role of trust and relationship structure in improving supply chain responsiveness. *Industrial Marketing Management*, 31(4), 367–382.

Handy, C. B. (1993). Understanding Organizations (4th ed.). Penguin.

Harland, C, M., Caldwell, N, D., Powell, P., & Zheng, J. (2007). Barriers to supply chain information integration: SMEs adrift of eLands. *Journal of Operations Management*, 25, 1234–1254.

Harrigan, K. R. (1985). Vertical integration and corporate strategy. *Academy of Management Journal*, 28(2), 397-425.

Hartmann, E., Kerkfeld, D., & Henke, M. (2012). Top and bottom line relevance of purchasing and supply management. *Journal of Purchasing & Supply Management*, 18, 22–34.

Hartono, E., & Holsapple, C. (2004). Theoretical foundations for collaborative commerce research and practice. *Information Systems and E-Business Management*, 2(1), 1-30.

Harvey, C., Kelly, A., Morris, H., & Rowlinson, M. (2010). Academic Journal Quality Guide. The Association of Business Schools, v4.

Hausman, A., & Johnston, W. J. (2010). The impact of coercive and non-coercive forms of influence on trust, commitment, and compliance in supply chains. *Industrial Marketing Management*, 39(3), 519–526.

Handfield, R. B., & Melnyk, S. A. (1998). The scientific theory-building process: a primer using the case of TQM. *Journal of Operations Management*, 16(4), 321–339.

Hayes, R.H. (2000). Toward a 'new architecture' for POM. *Production and Operations Management*, 9(2), 105-10.

Hayes, R. H., Pisano, G., Upton, D., & Wheelwright, S. (2005). *Operations, Strategy and Technology: Pursuing the Competitive Edge*. New York: John Wiley.

Hayes, R. H., & Wheelwright, S. C. (1979a). Linking manufacturing process and product life cycles. *Harvard Business review*, (January-February), 133.

Hayes, R. H., & Wheelwright, S. C. (1979b). The dynamics of process-product life cycles. *Harvard Business review*, (March-April), 127.

Hayes, R. H., & Wheelwright, S. C. (1984). *Restoring Our Competitive Advantage*. New York: Wiley.

He, Y., & Lai, K. K. (2012). Supply chain integration and service oriented transformation: Evidence from Chinese equipment manufacturers. *International Journal of Production Economics*, 135, 791–799.

Heide, J. B., & George, J. (1988). The role of dependence balancing in safeguarding transactionspecific assets in conventional channels. *Journal of Marketing*, 52(1), 20–35.

Heide, J. B., & John, G. (1992). Do norms matter in marketing relationships. *Journal of Marketing*, 56(4),32–44.

Heide, J. B. (1994). Interorganizational governance in marketing channels. *Journal of Marketing*, 58(1), 71–85.

Heinrich, C. E., & Simchi-Levi, D. (2005). Do IT investments really change financial performance? *Supply Chain Management Review*, 9 (May 4), 22–28.

Hershey, P., & Blanchard, K. H. (1969). *Management of organizational behavior*. Englewood Cliffs, New York: Prentice Hall.

Hervani, A., Helms, M., & Sarkis, J. (2005). Performance measurement for green supply chain management. *Benchmarking: An International Journal*, 12(4), 330-353.

Hingley, M. K. (2005). Power to all our friends? Living with imbalance in supplier-retailer relationships. *Industrial Marketing Management*, 34(8), 848–858.

Ho, D. C. K., Au, K.F., & Newton, E. (2002). Empirical research on SCM: a critical review and recommendations. *International Journal of Production Research*, 40(17), 4415-4430.

Hoch, J. E., & Dulebohn, J. H. (2013). Shared leadership in enterprise resource planning and human resource management system implementation. *Human Resource Management Review*, 23, 114–125

Hoffman, J. M., & Mehra, A. (2000). Efficient consumer response as a supply chain strategy for grocery businesses. *International Journal of Service Industry Management*, 11(4), 365–73.

Hofstede, G. (1991). Cultures and Organisations: Software of the Mind. McGraw-Hill, Maidenhead.

Hofstede, G., & Bond, M. H. (1988). The Confucius connection. From cultural roots to economic growth. *Organizational Dynamics*, 16, 4-21.

Holmberg, S. (2000). A systems perspective on supply chain measurements. *International Journal of Physical Distribution & Logistics Management*, 30(10), 847-68.

Holweg, M., & Bicheno, J. (2002). Supply chain simulation—a tool for education enhancement and endeavor. *International Journal of Production Economics*, 78, 163–175.

Hong, P., & Jeong, J. (2006). Supply chain management practices of SMEs: from a business growth perspective. *Journal of Enterprise Information Management*, 19(3), 292–302.

Hong, P., Kwon, H. B., & Roh, J. J. (2009). Implementation of strategic green orientation in supply chain: an empirical study of manufacturing firms. *European Journal of Innovation Management*, 12(4), 512–532.

Horwitch, M., & Thietart, R.A. (1987). The effect of business interdependencies on product R&Dintensive business performance. *Management Science*, 33(2), 178–197.

Hou, Y., Xiong, Y., Wang, X., & Liang, X. (2014). The effects of a trust mechanism on a dynamic supply chain network. *Expert Systems with Applications*, 41, 3060–3068

House, R. & Aditya, R. (1997). The Social Scientific Study of Leadership: Quo Vadis? *Journal of Management*, 23(3), 409-473.

Hsueh, C. F., (2015). A bilevel programming model for corporate social responsibility collaboration in sustainable supply chain management. *Transportation Research*, Part E 73, 84–95. http://dx.doi.org/10.1016/j.tre.2014.11.006.

Hsu, C. W., Kuo, T. C., Chen, S. H., & Hu, A. H. (2013). Using DEMATEL to develop a carbon management model of supplier selection in green supply chain management. *Journal of Cleaner Production*, 56, 164-172.

Huang, Z., & Gangopadhyay, A. (2004). A simulation study of supply chain management to measure the impact of information sharing. *Information Resource Management*, 17(3), 20-31.

Huang, Y. T., Liu, C.Z., & Wang, Y.F. (2014). Audit Quality of the Integrated Supply Chain – Evidence from Taiwan Electronic Industry. *Journal of Accounting and Finance*, 14(4).

Hult, G. T. M., Ketchen, D. J. Jr., & Nichols, E. L. Jr. (2002). Examination of cultural competitiveness and order fulfillment cycle time within supply chains. *The Academy of Management Journal*, 45(3), 577-86.

Hult, G. T. M., Ketchen, D. J. Jr., & Slater, S. F. (2004). Information processing, knowledge development, and strategic supply chain performance. *Academy of Management Journal*, 47(2), 241-54.

Hunt, J. G. (1991). Leadership: A new synthesis. Newbury park, CA: Sage

Huo, B. (2012). The impact of supply chain integration on company performance: an organizational capability perspective. *Supply Chain Management: An international Journal*, 17(6), 596-610.

Huo, B., Han, Z., & Prajogo, D. (2014). The effect of ISO 9000 implementation on flow management. *International Journal of Production Research*, 52(21), 6467–6481.

Huxham, C., & Beech, N. (2008). Inter-organizational power. The Oxford handbook of interorganizational relations. New York: Oxford University Press.

Ishaq, M. I., Hussain, N., Khaliq, W., & Waqas, M. (2012). A Review on Triple-A Supply Chain Performance. *East Asian Journal of Business Management*, 2-2, 35-39

Ismail, H. S., & Sharifi, H. (2006). A balanced approach to building agile supply chains. *International Journal of Physical Distribution & Logistics Management*, 36(6), 431-44.

ISO14040 (1997). *Environmental management - life cycle assessment - principles and framework*. Tech. Rep. ISO. Retrieved December, 20, 2014 from: http://www.iso.org/iso/catalogue_detail?csnumber=37456.

Jabbour, C. J. C., Santos, F. C. A., & Nagano, M. S. (2010). Contributions of HRM throughout the stages of environmental management: methodological triangulation applied to companies in Brazil. *International Journal of Human Resource Management*, 21, 1049-1089.

Jain, V., Benyoucef, L., & Deshmukh, S. G. (2008). What's the buzz about moving from 'lean' to 'agile' integrated, supply chains? A fuzzy intelligent agent-based approach. *International Journal of Production Research*, 46(23), 6649-77.

Jiao, J. R., You, X., & Kumar, A. (2006). An agent-based framework for collaborative negotiation in the global manufacturing supply chain network. *Robotics and Computer-Integrated Manufacturing*, 22, 239–255.

Jin, Y., Vonderembse, M., Ragu-Nathan, T. S., & Smith, J. T. (2014). Exploring relationships among IT-enabled sharing capability, supply chain flexibility, and competitive performance. *International Journal of Production Economics*, 153, 24–34

Johnsen, T. E. (2009). Supplier involvement in new product development and innovation: taking stock and looking to the future. *Journal of Purchasing and Supply Management*, 15(4), 187–197.

Johnson, J. L. (1999). Strategic integration in distribution channels: managing the interfirm relationship as a strategic asset. *Academy of Marketing Science Journal*, 27(1), 4–18.

Johnson, G., & Scholes, K. (1999). Exploring Corporate Strategy (5th ed.). Europe: Prentice Hall.

Johnston, D. A., McCutcheon, D.M., Stuart, F. I., & Kerwood, H. (2004). Effects of supplier trust on performance of cooperative supplier relationships. *Journal of Operations Management*, 22(1), 23-38.

Jonsson, P., Kjellsdotter, L., & Rudberg, M. (2007). Applying advanced planning systems for supply chain planning: three case studies. *International Journal of Physical Distribution & Logistics Management*, 37(10), 816-834.

Jonsson, P., & Mattsson, S. A. (2013). The value of sharing planning information in supply chains. *International Journal of Physical Distribution & Logistics Management*, 43(4), 282-299

Josi, A. W., & Campbell, A. J. (2003). Effect of environmental dynamism on relational governance in manufacture-supplier relationships: A contingency framework and an empirical test. *Journal of the Academy of Marketing Science*, 31, 176–188.

Juran, J. M. (1988). Juran on Planning for Quality. New York: Free Press.

Juttner, U., Christopher, M., & Baker, S. (2007). Demand chain management-integrating marketing and supply chain management. *Industrial Marketing Management*, 36, 377 – 392.

Juttner, U., Peck, H., & Christopher, M. (2003). Supply chain risk management: outlining an agenda for future research. *International Journal of Logistics Research and Applications*, 6(4), 197-210.

Kadar, M., Moise, L.A., & Colomba, C. (2014). Innovation Management in the Globalized Digital Society. *Procedia - Social and Behavioral Sciences*, 143, 1083 – 1089

Kahkonen, A. K., Lintukangas, K., & Hallikas, J. (2015). Buyer's dependence in value creating supplier Relationships. *Supply Chain Management: An International Journal*, 20(2), 151 - 162

Kahn, K, B., & Mentzer, J, T. (1996). Logistics and interdepartmental integration. *International Journal of Physical Distribution and Logistics Management*, 26(8), 6-14.

Kahn, K., & Mentzer, J. (1998). Marketing's integration with other departments. *Journal of Business Research*, 42, 53–62.

Kaplan, R. S., & Norton, D. P. (1992). The balanced scorecard – measures that drive Performance. *Harvard Business Review*. January-February, pp. 71-90.

Kagermann, H., Osterle, H., & Jordan, J. M. (2011). *IT-Driven Business Models: Global Case Studies in Transformation*. John Wiley & Sons Inc., Hoboken, New Jersey.

Kale, S. H. (1984). Dealer Perceptions of Manufacturer Power and Influence Strategies in a Developing Country. *Journal of Marketing Research*, 23, (4/11), 387-393.

Kang, J. H., & Kim, Y. D. (2010). Coordination of inventory and transportation managements in a two-level supply chain. *International Journal of Production Economics*, 123, 137–145.

Karande, K., Ha, J., & Singhapakdi, A. (2008). The role of contextual factors in relationship commitment of buyers to foreign suppliers: A survey of Korean importers. *Industrial Marketing Management*, 37, 856–862.

Katz, R. L. (1974). Skills of an Effective Administrator, *Harvard Business Review*, September–October, 90–102.

Kaufman, A. K., Wood, C. H., & Theyel, G. (2000). Collaboration and technology linkages: a strategic supplier typology. *Strategic Management Journal*, 21(6), 649-63.

Kaynak, H., & Hartley, J. L. (2008). A replication and extension of quality management into the supply chain. *Journal of Operations Management*, 26(4), 468-89.

Kearns, G. S., & Lederer, A. L. (2003). A resource-based view of strategic IT alignment: how knowledge sharing creates competitive advantage. *Decision Sciences*, 34(1), 1–29.

Kemppainen, K., & Vepsalainen, A. P. J. (2003). Trends in industrial supply chains and Networks. *International Journal of Physical Distribution & Logistics Management*, 33(8), 701-19.

Kennedy, R. E., & Sharma, A. (2009). *The Services Shift: Seizing the Ultimate Offshore Opportunity*. Pearson Education, New Jersey.

Ketchen, D., & Guiniupero, L. (2004). The intersection of strategic management and supply chain management. *Industrial Marketing Management*, 33, 51–56.

Ketchen, D., & Hult, G. T. (2007). Bridging organization theory and supply chain management: The case of best value supply chains. *Journal of Operations Management*, 25(2), 573–580.

Ketchen, D. J. Jr., Rebarick W., Hult, G. T. M., & Meyer, D. (2008). Business Horizons, 51, 235-243

Khazanchi, S., Lewis, M. W., & Boyer, K. K. (2007). Innovation-Supportive Culture: The Impact of Organizational Values on Process Innovation. *Journal of Operations Management*, 25(4), 871–884.

Kim, A. W. (2009). An investigation on the direct and indirect effect of supply chain integration on firm performance. *International Journal of Production Economics*, 119, 328–346.

Kim, D. Y., Kumar, V., & Kumar, U. (2012). Relationship between Quality Management Practices and Innovation. *Journal of Operations Management*, 30(4), 295–315.

Kim, K. K., Park, S. H., Ryoo, S. U., & Park, S. K. (2010). Inter-organizational cooperation in buyersupplier relationships: both perspectives. *Journal of Business Research*, 63 (8), 863–869.

Kinicki, A., & Williams, B. K. (2003). *Management: A practical introduction*. Boston: McGraw-Hill Irwin.

King, A, A., & Lenox, M, J. (2001). Lean and green? An empirical examination of the relationship between lean production and environmental performance. *Production and Operations Management*, 10(3), 244.

Klein, B. (1988). Vertical integration as organizational ownership: the Fisher Body-General Motors relationship revisited. *Journal of Law, Economics & Organization*, 4(1), 199-213.

Klein, K. J., & Sorra, J. S. (1996). The Challenge of innovation implementation. Academy of Management Review, 21(4), 1055-80.

Klimov, R., & Merkuryev, Y. (2008). Simulation model for supply chain reliability evaluation. *Technological and Economic Development of Economy Baltic Journal on Sustainability*, 14(3), 300–311.

Koh, S. C. L., Demirbag, M., Bayraktar, E., Tatoglu, E., & Zaim, S. (2007). The impact of supply chain management practices on performance of SMEs. *Industrial Management & Data Systems*, 107(1), 103-24.

Koudal, P., & Wellener, P. (2003). Digital loyalty networks: Continuously connecting automakers with their customers and suppliers. *Strategy and Leadership*, 31(6), 4 - 11.

Koufteros, X. A., Cheng, T. C. E., & Lai, K. H. (2007). Black-box and gray box supplier integration in product development: antecedents, consequences and the moderating role of firm size. *Journal of Operations Management*, 25 (4), 847–870.

Koufteros, X. A., Vonderembse, M., & Jayaram, J. (2005). Internal and external integration for product development: the contingency effects of uncertainty, equivocality, and platform strategy. *Decision Sciences*, 36(1), 97–133.

Kovacs, G. L., & Paganelli, P. (2003). A planning and management infrastructure for large, complex, distributed projects—beyond ERP and SCM. *Computers in Industry*, 51, 165–183.

Krapfel, R.E., Salmond, D., & Spekman, R. (1991). A strategic approach to managing buyer-seller relationships. *European Journal of Marketing*, 25(9), 22-37.

Kreitner, R., Kinicki, A., & Buelens, M. (1999). *Organizational Behaviour*, First European edition, McGraw-Hill.

Kros, J. F., Falasca, M., & Nadler, S. S. (2006). Impact of just-in-time inventory systems on OEM suppliers. *Industrial Management & Data Systems*, 106(2), 224-41.

Kroes, J. R., & Ghosh, S. (2010). Outsourcing congruence with competitive priorities: impact on supply chain and firm Performance. *Journal of Operations Management*, 28(2), 124-43.

Kuhn, D. (1999). A developmental model of critical thinking. Educational Researcher, 28, 16–25.

Kuckartz, U., Dresing, T., Radiker, S., & Stefer, K. (2008). Qualitative Evaluation. Der Einstieg in die Praxis (2nd ed.). Wiesbaden: VS Verlag für Sozialwissenschaften.

Kuk, G. (2004). Effectiveness of vendor managed inventory in the electronics industry: Determinants and outcomes. *Information and Management*, 41(5), 645–654.

Kwon, I. W. G., & Suh, T. (2005). Trust, commitment and relationships in supply chain management: a path analysis. *Supply Chain Management: An International Journal*, 10(1), 26-33.

La Londe, B. J. (1983). A reconfiguration of logistics systems in the 80s: strategies and challenges. *Journal of Business Logistics*, 4(1), 1–11.

Lambert, D. M., Emmelhainz, M.A., & Gardner, J. T. (1999). Building successful partnerships. *Journal of Business Logistics*, 20(1), 165–181.

Lambert, D. (2008). University of Auckland Business Review Spring 2008. Retrieved July, 20, 2012 from: <u>http://www.shortcourses.ac.nz/_docs/AGlobalViewOfSupplyChainManagement2008.pdf</u>

Lambert, D. M., & Cooper, M. C. (2000). Issues in Supply Chain Management. *Industrial Marketing Management*, 29, 65–83.

Lambert, D. M., Cooper, M., C., & Pagh, J., D. (1998). Supply chain management: Implementation issues and research opportunities. *International Journal of Logistics Management*, 9(2), 1-19.

Lambert, D. M., Robeson, J. F., & Stock, J. R. (1978). An appraisal of the integrated physical distribution management concept. *International Journal of Physical Distribution and Materials Management*, 9(1), 74–88.

Lamming, R., Caldwell, N., & Harrison, D. (2013). Developing the concept of transparency for use in supply relationships. *British Journal of Management*, 15, 291-302.

Lawrence, P. R., & Lorsch, J. W. (1967). Differentiation and integration in complex organizations", *Administrative Science Quarterly*, 12(1), 1-47.

Laudon, K, C., & Laudon, J. P. (2000). *Management Information Systems: Organization and Technology in the Networked Enterprise*. New Jersey: Prentice Hall.

Lee, A. H. I., Kang, H., Hsu, C., & Hung, H. (2009). A green supplier selection model for high-tech industry. *Expert Systems with Applications*, 36, 7917–7927.

Lee, H. L. (2004). The triple-A supply chain. *Harvard Business Review*, 82(10), 102-12.

Lee, H. L., Padmanabhan, V., & Whang, S. (1997). Information distortion in supply chain: The bullwhip effect. *Management Science*, 43(4), 546–558.

Lee, H. L., So, K. C., & Tsang, C. S. (2000). The value of information sharing in a two-level supply chain. *Management Science*, 46(5), 626-643.

Lee, N., & Lings, I. (2008). *Doing business research: a guide to theory and practice*. SAGE Publications Limited.

Lee, S. M., Kim, S. T., & Choi, D. (2012). Green supply chain management and organizational performance. *Industrial Management & Data Systems*, 112(8), 1148-1180.

Lee, S., & Klassen, R. D. (2008). Drivers and enablers that foster environmental management capabilities in small and medium-sized suppliers in supply chains. *Production and Operations Management*, 17(6), 573–586.

Lee, S. Y. (2008). Drivers for the participation of small and medium-sized suppliers in green supply chain initiatives. *Supply Chain Management: An International Journal*, 13(3), 185–198.

Lee, T. W. (1999). Using Qualitative Methods in Organizational Research. Thousand Oaks, CA: Sage.

Lederer, A. L., Mirchandani, D.A., Sims, K. (2001). The search for strategic advantage from the World Wide Web. *International Journal of Electronic Commerce*, 5(4), 117–133.

Lehmann, H. P. (2001). A Grounded Theory of International Information Systems. PhD Thesis, University of Auckland, New Zealand.

Leidner, D. E. (2010). Globalization, culture, and information: towards global knowledge transparency. *Journal of Strategic Information Systems*, 19(2), 69–77.

Lengnick-Hall, M. L., Lengnick-Hall, C. A., & Rigsbee, C. M. (2013). Strategic human resource management and supply chain orientation. *Human Resource Management Review*, 23(4), 366-377.

Leonidou, L.C., Talias, M.A., & Leonidou, C.N. (2008). Exercised power as a driver of trust and commitment in cross-border industrial buyer–seller relationships. *Industrial Marketing Management*, 37(1), 92–103.

Levine, S., & Crom, M. (1994). The Leader in You, Simon and Schuster. In: Peter Speck (2006). *Teamwork in Palliative Care: Fulfilling or Frustrating?* Oxford University Press.

Lewin, A. Y., & Couto, V. (2007). Offshoring Research Network 2006 Survey Report: Next Generation Offshoring - The Globalization of Innovation. Booz Allen Hamilton, Chicago.

Lewin, K. (1951). Field Theory in Social Science. Harper & Row, New York, NY.

Lewins, K. (1947). Frontiers in group dynamics: concept, method, and reality in social science: Social equilibria and social change. *Human Relations*, 1, 5-41.

Li, S., Ragu-Nathan, B., Ragu-Nathan, T. S., & Rao, S. S. (2005). Development and validation of a measurement for studying supply chain management practices. *Journal of Operations Management*, 23, 618–641.

Li, X., Chung, C., Goldsby, T. J., & Holsapple, C. W. (2008). A unified model of supply chain agility: the work-design perspective. *International Journal of Logistics Management*, 19(3), 408-35.

Liden, R. C., & Antonakis, J. (2009). Considering context in psychological leadership research. *Human Relations*, 62(11), 1587-605.

Lightfoot, H., Baines, T., & Smart, P. (2013). The servitization of manufacturing: A systematic literature review of interdependent trends. *International Journal of Operations & Production Management*, 33(11/12), 1408-1434

Lim, J. H., Richardson, V. J., & Roberts, T. L. (2004). Information technology investment and firm performance: a meta-analysis. In: *Proceedings of the 37th Hawaii International Conference on Systems Sciences*, pp. 1–11.

Lin, C., Kuei, C. H., & Chai, K. W. (2013). Identifying critical enablers and pathways to high performance supply chain quality management. *International Journal of Operations & Production Management*, 33(3), 347 – 370.

Lin, C. T., Chiu, H., & Chu, P. Y. (2006). Agility index in the supply chain. *International Journal of Production Economics*, 100 (2), 285-99.

Lindgreen, A., & Wynstra, F. (2005). Value in business markets: what do we know? Where are we going? *Industrial Marketing Management*, 34, 732–748.

Ling, Y., Simsek, Z., Lubatkin, M. H., & Veiga, J. F. (2008). The impact of transformational CEOs on the performance of small- to medium-sized firms: Does organizational context matter? *Journal of Applied Psychology*, 93, 923–34.

Livesey, F. (2006). *Defining High Value Manufacturing, Institute for Manufacturing*. University of Cambridge.

Loh, T. C., Koh, S.C.L., & Simpson, M. (2006). An investigation of the values of becoming an extended enterprise. *International Journal of Computer Integrated Manufacturing*, 19(1), 49–58.

Lysons, K., & Farrington, B. (2006). *Purchasing and Supply Chain Management*. Pearson Education, Upper Saddle River, NJ.

Lucas, E. (2000). Tooling Up for Leadership. Professional Manager, September, 10.

Lummus, R. R., Vokurka, R. J., & Krumwiede, D. (2008). Supply chain integration and organizational success. S.A.M. Advanced Management Journal, 73(1), 56–63.

Luken, R., & Stares, R. (2005). Small business responsibility in developing countries: a threat or an opportunity? *Business Strategy and the Environment*, 14, 38-53.

Mabert, V. A., & Venkatraman, M. A. (1998). Special research focus on supply chain linkages: Challenges for design and management in the 21st century. *Decision Sciences*, 29(3), 537–550.

MacBryde, J., Paton, S., & Clegg, B. (2013). Understanding high-value manufacturing in Scottish SMEs. *International Journal of Operations & Production Management*, 33(11/12), 1579 – 1598.

Maccoby, M. (2006). Creating collaboration. Research Technology Management, 49(6), 60-62.

Magretta, J., & Dell, M. (1998). The power of virtual integration: an interview with Dell Computers' Michael Dell. Harvard Business Review, 76, March/April, 72-84.

Malhotra, A., Gosain, S., & Sawy, A. O. (2005). Absorptive capacity configurations in supply chains: gearing for partner enabled market knowledge creation, *Management Information systems Quarterly*, 29(1), 145-187.

Malta Council of Science and Technology. (2011). *Malta's National Strategic Plan for Research & Innovation 2011-2020: A Vision for Knowledge-Driven Growth*, draft for consultation. Retrieved March, 30, 2015 from: http://mcst.gov.mt/files/uploaded/National%20Strategy%20DRAFT.pdf.

Malta Enterprise. (2012). Retrieved February 21, 2012 from: http://www.maltaenterprise.com/en/support. Manderson, A. K. (2006). A systems based framework to examine the multi-contextual application of the sustainability concept. *Environment Development Sustainability*, 8, 85–97.

ManuFuture Platform. (2006). STRATEGIC RESEARCH AGENDA assuring the future of manufacturing in Europe. Manufuture High Level Group and Support Group. Retrieved April, 10, 2015, from: <u>http://www.manufuture.org</u>.

March, J. G. (1991). Exploration and exploitation in organizational learning. *Organization Science*, 2(1), 71–87.

Marcotte, F., Grabot, B., & Affonso, R. (2009). Cooperation models for supply chain management. *International Journal of Logistics Systems and Management*, 5(1/2), 123-153.

Markley, M. J., & Davis, L. (2007). Exploring future competitive advantage through sustainable supply chains. *International Journal of Physical Distribution & Logistics Management*, 39(9), 763-74.

Martin, A., Willburn, P., Morrow, P., Downing, K., & Criswell, C. (2007). The changing nature of leadership (Whitepaper). Greensboro, NC: Center for Creative Leadership.

Martinez, V., Neely, A., Guanjie, R., & Smart, A. (2008). *HVM – Delivering on Promise*, Advanced Institute of Management Research, London.

Mascarenhas, O. A., Kesavan, R., & Bernacchi, M. (2004). Customer value-chain involvement for cocreating customer delight. *The Journal of Consumer Marketing*, Nov, 21(7), 486-496.

Matopoulos, A., Vlachopoulou, M., Manthou, V., & Manos, B. (2007). A conceptual framework for supply chain collaboration: empirical evidence from the agri-food industry. *Supply Chain Management: An International Journal*, 12(3), 177-86.

Matthews, J. (2003). Strategizing by firms in the presence of markets for resources. *Industrial and Corporate Change*, 12, 1157–1193.

McAfee, A. (2002). The impact of enterprise information technology adoption on operational performance: an empirical investigation. *Production and Operations Management*, 11(1), 33–53.

McCarter, M., Fawcett, S., & Magnan, G. (2005). The effect of people on the supply chain world: some overlooked issues. *Human Systems Management*, 24, 197–208.

McCauley-Smith, C., Williams, S., Gillon, A.C., Ashley Braganza, A., & Ward, C. (2013). Individual Leader to Interdependent Leadership: A Case Study in Leadership Development and Tripartite Evaluation. *Advances in Developing Human Resources*, 15(1) 83–105.

McEvily, B., Perrone, V., & Zaheer, A. (2003). Trust as an organizing principle. *Organization Science*, 14(1), 91–103.

McFarland, R. G., Bloodgood, J. M., & Payan, J. M. (2008). Supply chain contagion. *Journal of Marketing*, 72(2), 63–79.

McIvor, R. (2010). The influence of capability considerations on the outsourcing decision: the case of a manufacturing company. *International Journal of Production Research*, 48(17), 5031–5052.

McIvor, R., Humphreys, P., & McAleer, E. (1997). The evolution of the purchasing function. *Strategic Change*, 6(3), 165–179.

McKelvey, B. (2001). Energising order-creating networks of distributed intelligence: improving the corporate brain. *International Journal of Innovation Management*, 5(2), 181-212.

McKinnon, A. C., S. Browne, M., & Whiteing A. (2010), *Green Logistics. Improving the Environmental Sustainability of Logistics*. London, UK: Kogan Page.

McKinnon, J. L., Harrison, G. L., Chow, C. W., & Wu, A. (2003). Organizational culture: association with commitment, job satisfaction, propensity to remain, and information sharing in Taiwan, *International Journal of Business Studies*, 11(1), 25-44.

Meehan, J., & Wright, G. H. (2012). The origins of power in buyer-seller relationships. *Industrial Marketing Management*, 4, 669–679.

Mendelson, H. (2000). Organizational architecture and success in the information technology industry. *Management Science*, 46(4), 514-29.

Menor, L. J., Kristal, M. M., & Rosenzweig, E. D. (2007). Examining the Influence of Operational Intellectual Capital on Capabilities and Performance. *Manufacturing and Service Operations Management*, 9(4), 559–578.

Mentzer, J. T., DeWitt, W., & Keebler, J. S. (2001). Defining supply chain management. *Journal of Business Logistics*, 22(2), 1–25.

Mentzer, J. T., Min, S., & Bobbitt, L. M. (2004). Toward a unified theory of logistics. *International Journal of Physical Distribution & Logistics Management*, 34(8), 606-27.

Metcalfe, D., Meringer, J., & Roitman, J. V. (2002). *eMarketplaces: Rebound and deliver*. US: Forrester Research, Inc.

Meyer, J., & Rowan, B. (1977). Institutionalized organizations: formal structure as myth and ceremony. *American Journal of Sociology*, 83(2), 340–363.

Miles, M. B., & Huberman, A. M. (1994). *Qualitative Data Analysis: An Expanded Sourcebook*. (2nd ed.). CA: Sage, Thousand Oaks.

Miller, D. (1988). Relating porter's business strategies to environment and structure: analysis and performance implications. *Academy of Management Journal*, 31(2), 280–308.

Mills, D. Q. & Friesen, G. B. (2001). 'Empowerment'. In: Crainer, S. and Dearlove, D. (eds), *Financial Times Handbook of Management* (2nd ed.). Financial Times Prentice Hall.

Miocevic, D. (2011). Organizational buying effectiveness in supply chain context: Conceptualization and empirical assessment. *Journal of Purchasing & Supply Management*, 17, 246–255

Mitchell, R. K., Agle, B. R., & Wood, D. J. (1997). Toward a theory of stakeholder identification and salience: defining the principle of who and what really counts. *Academy of Management. The Academy of Management Review*, 22(4), 853.

Mizik, N., & Jacobson, R. (2003). Trading Off Between Value Creation and Value Appropriation: The Financial Implications of Shifts in Strategic Emphasis. *Journal of Marketing*, 67 (January), 63–76

Mol, M. J. (2003). Purchasing's strategic relevance. *Journal of Purchasing & Supply Management*, 9(1), 43–50.

Mondragon, A. E., & Lalwani, C. (2011). Measures for auditing performance and integration in closed-loop supply chains. *Supply Chain Management: An International Journal*, 16(1), 43–56

Morash, E. A., & Clinton, S. R. (1998). Supply chain integration: Customer value through collaborative closeness versus operational excellence. *Journal of Marketing Theory and Practice*, 6(4), 104–120.

Morgan, R. M., & Hunt, S. D. (1994). The commitment-trust theory of relationship marketing. *Journal of Marketing*, 58(3), 20–38.

Msanjila, S. S. & Afsarmanesh, H., (2011). On modelling evolution of trust in organisations towards mediating collaboration. Production Planning and Control: *The Management of Operations*, 22 (5–6), 518–537.

Muffatto, M., & Payaro, A. (2004). Integration of web-based procurement and fulfillment: a comparison of case studies. *International Journal of Information Management*, 24, 295–311.

Mullins, L. J. (2005). *Management and Organisational Behaviour* (7th ed.). Pearson Education Limited, FT Prentice Hall.

Myers, M. D. (1997). Qualitative Research in Information Systems. MIS Quarterly, 21(2), 241-242.

Narasimhan, R., & Das, A. (2001). The impact of purchasing integration and practices on manufacturing performance. *Journal of Operations Management*, 19(5), 593–609.

Narasimhan, R., & Kim, S. W. (2001). Information system utilization for supply chain integration. *Journal of Business Logistics*, 22(2), 51–75.

Narasimhan, R., Swink, M., & Kim, S. W. (2006). Disentangling leanness and agility: an empirical Investigation. *Journal of Operations Management*, 24(5), 440-57.

Narayandas, D., & Rangan, V. K. (2004). Building and sustaining buyer-seller relationships in mature industrial markets. *The Journal of Marketing*, 68(3), 63–77.

Neely, A. (2009). Exploring the financial consequences of the servitization of manufacturing. *Operations Management Research*, 1(2).

Neely, A., Adams, C., & Kennerley, M. (2002). The performance prism: The scorecard for measuring and managing business success. London: FT Prentice-Hall.

Neely, A.D., Gregory, M.J., & Platts, K.W., (2005). Performance measurement system design: a literature review and research agenda. *International Journal of Operations and Production Management*, 25(12), 1228–1263.

Neely, A., Richard, H., Mills, J., Platt, K., & Bournce, M. (1997). Designing performance measures: a structured approach. *International Journal of Operations and Production Management*, 17(11), 1131–1152.

Neuman, J., & Samuels, C. (1996). Supply chain integration: vision or reality? *Supply Chain Management*. 71(2), 7.

Newell, S., Huang, J. C., Galliers, R. D., & Pan, S. L. (2003). Implementing enterprise resource planning and knowledge management systems in tandem: fostering efficiency and innovation complementarity. Information and Organization, 13, 25–52

Norman, P. M. (2004). Knowledge acquisition, knowledge loss, and satisfaction in high technology alliances. *Journal of Business Research*, 57(6), 610–619.
Nyaga, G. N., Lynch, D. F., Marshall, D., & Ambrose, E. (2013). Power asymmetry, adaptation and collaboration in dyadic relationships involving a powerful partner. *Journal of Supply Chain Management*, 49(3), 42-65.

Noble, M. A. (1995). Manufacturing strategy: testing the cumulative model in a multiple country Context. *Decision Sciences*, 26(5), 693-721.

Nooteboom, B. (2000). Trust as a governance device. In: M. Casson and A. Godley, eds. *Cultural factors in economic growth*. New York: Springer, 44–68.

Northouse, P. G. (2007). Leadership: Theory and practice (4th ed.). Thousand Oaks, CA: Sage.

NSO. (2012). Retrieved June, 30, 2013 from: http://www.nso.gov.mt/themes/theme/Malta_in_Figures_2012.pdf.

NSO. (2013). Retrieved November, 30, 2014 from: http://www.nso.gov.mt/themes/theme/Malta_in_Figures_2013.pdf.

NSO. (2014). Retrieved March, 30, 2015 from: http://www.nso.gov.mt/themes/theme/Malta_in_Figures_2014.pdf.

Obeng, E. (2001). 'New World Organization'. In: Crainer, S. and Dearlove, D. (eds), *Financial Times Handbook of Management* (2nd ed.). Financial Times Prentice Hall.

Ofek, E., & Sarvary, M. (2001). Leveraging the customer base: creating competitive advantage through knowledge management. *Management Science*, 47(11) 1441-56.

Ogden, J. A., Rossetti, C. L., & Hendrick, T. E. (2007). An exploratory cross-country comparison of strategic purchasing. *Journal of Purchasing and Supply Management*, 13(1), 2–16

Olivares, O. J., Peterson, G., & Hess, K. P. (2007). An existential-phenomenological framework for understanding leadership development experiences. *Leadership and Organization Development Journal*, 28, 76-91.

Oliver, R. K., & Webber, M. D. (1992). Supply Chain Management: Logistics Catches up with Strategy. In: Christopher, M.G. *Logistics, The strategic issue*. Chapman and Hall, London.

O'Leary-Kelly, S. W., & Flores, B. E. (2002). The integration of manufacturing and marketing/sales decisions: impact on organizational performance. *Journal of Operations Management*, 20(3), 221–240.

Olsen, L. L., & Johnson, M. D. (2003). Service equity, satisfaction, and loyalty: from transaction specific to cumulative evaluations. *Journal of Service Research*, 5(3), 184-95.

O'Reilly, C. A., Chatman, J., & Caldwell, D. F. (1991). People and organizational culture: A profile comparison approach to assessing person-organization fit. *Academy of Management Journal*, 34, 487–516.

O'Reilly, C. A., Cladwell, D. F., Chatman, J.A., Lapiz, M., & Self, W. (2010). How leadership matters: The effects of leaders' alignment on strategy implementation. *Leadership Quarterly*, *21*, 104-113.

Osborn, R. N., Hunt, J. G., & Jauch, L. R. (2002). Toward a contextual theory of leadership. *Leadership Quarterly*, 13, 797-837.

Osborn, R. N., & Marion, R. (2009). Contextual leadership, transformational leadership and the performance of international innovation seeking alliances. *Leadership Quarterly*, 20(2), 191–206.

Oxford dictionary. (2010). 7th Edition, Oxford University Press.

Pagell, M. (2004). Understanding the factors that enable and inhibit the integration of operations, purchasing and logistics. *Journal of Operations Management*, 22, 459–487.

Pagell, M., & Wu, Z. (2009). Building a more complete theory of sustainable supply chain management using case studies of 10 exemplars. *Journal of Supply Chain Management*, 45(2), 37–56.

Palmer, D. A., Devereaux, P. J., & Zhou, X. (1993). Late adoption of the multidivisional form by large U.S. corporations: Institutiona, political, and economical accounts. *Administrative Science Quarterly*, 38, 100-131.

Palomero, S., & Chalmeta, R. (2012): A guide for supply chain integration in SMEs, Production Planning & Control, DOI:10.1080/09537287.2012.698422.

Panayides, P., & Lun, Y. (2009). The impact of trust on innovativeness and supply chain performance. *International Journal of Production Economics*, 122(1), 35–46.

Parasuraman, A., Zeithaml, V., & Berry, L. L. (1988). SERVQUAL: A multiple-item scale for measuring consumer perceptions of service quality. *Journal of Retailing*, 64(1), 13–40

Parente, D., Lee, P., Ishman, M., & Roth, A. (2008). Marketing and supply chain management: a collaborative research agenda. *Journal of Business & Industrial Marketing*, 23(8), 520-8.

Parmigiani, A., Klassen, R., & Russo, M. (2011). Efficiency meets accountability: performance implications of supply chain configuration, control and capabilities. *Journal of Operations Management*, 29, 212–223.

Parker, G. G., & Anderson, E. G. (2002). From buyer to integrator: the transformation of the supplychain manager in the vertically disintegrating firm. *Production and Operations Management*, 11(1), 75–91.

Patnayakuni, R., Rai, A., & Seth, N. (2006). Relational Antecedents of Information Flow Integration for Supply Chain Coordination. *Journal of Management Information Systems*, 23(1), 13-49.

Patton, M. Q. (2002). *Qualitative Research & Evaluation Methods*. 3rd Ed., Sage.

Paulraj, A., Chen, I. J., & Flynn, J. E. (2006). Levels of strategic purchasing: impact on supply integration and performance. *Journal of Purchasing and Supply Management*, 12(3), 107–122.

Paulraj, A., Lado, A., & Chen, J. (2008). Inter-organizational communication as a relational competency: antecedents and performance outcomes in collaborative buyer–supplier relationships. *Journal of Operations Management*, 26 (1), 45–64.

Pero, M., Rossi, T., Noe, C., & Sianesi, A. (2010). An exploratory study of the relation between supply chain topological features and supply chain performance. *International Journal of Production Economics*, 123, 266–278.

Perona, M., & Miragliotta, G. (2004). Complexity management and supply chain performance assessment. A field study and a conceptual framework. *International Journal of Production Economics*, 90(1), 103–115.

Perotti, S., Zorzini, M., Cagno, E., & Mucheli, G. J. L. (2012). Green supply chain practices and company performance: the case of 3PLs in Italy. *International Journal of Physical Distribution & Logistics Management*, 42(7), 640-672.

Peteraf, M. (1993). The cornerstones of competitive advantage: a resource-based view. *Strategic Management Journal*, 14, 179–192.

Petersen, K. J., Handfield, R., & Ragatz, G. (2005). Supplier integration into new product development: coordinating product, process and supply chain design. *Journal of Operations Management*, 23, 371–388.

Pilkington, A., & Fitzgerald, R. (2006). Operations management themes, concepts and relationships: a forward retrospective of IJOPM. *International Journal of Operations & Production Management*, 26(11), 1255-75.

Pinto, J. (2009). *High-Value Manufacturing*, Automation World San Diego, CA, USA. Retrieved August 21, 2011 from: <u>http://www.jimpinto.com/writings/highvaluemfg.html</u>.

Porter, M. E. (1980). Competitive *Strategy: Techniques for Analyzing Industries and Competitors*. New York: Free Press.

Porter, M. E. (1985). Competitive Advantage. New York: Free Press.

Porter, M. E. (1987). *From Competitive Advantage to Corporate Strategy*. Harvard Business Review, May-June.

Porter, M. E. (1996). What Is Strategy? Harvard Business Review, November–December.

Porter, M. E., & Heppelmann, J. (2014). How Smart Connected Devices are Transforming Competition. *Harvard Business Review*, Nov 2014, 64-88.

Power, D. (2005). Supply chain management integration and implementation: a literature review. *Supply Chain Management: An International Journal*, 10(4), 252–263.

Prahalad, C.K., & Hamel, G. (1994). Strategy as a field of study: why search for a new paradigm. *Strategic Management Journal*, 15, special issue, 5-16.

Prajogo, D. I., & C. M. McDermott. (2011). The Relationship between Multidimensional Organizational Culture and Performance. *International Journal of Operations and Production Management*, 31(7), 712–735.

Prajogo, D., Huo, B., & Han, Z. (2012). The Effects of Different Aspects of ISO 9000 Implementation on Key Supply Chain Management Practices and Operational Performance. *Supply Chain Management: an International Journal*, 17(3) 306–322.

Prajogo, D., & Olhager, J. (2012). Supply chain integration and performance: the effects of long-term relationships, information technology and sharing, and logistics integration. International Journal of Production Economics, 135(1), 514–522.

Prajogo, D. I., & Sohal, A. S. (2003). The relationship between TQM practices, quality performance, and innovation performance: An empirical examination. *International Journal of Quality and Reliability Management*, 20(8), 901–918.

Rai, A., Patnayakuni, R., & Seth, N. (2006). Firm performance impacts of digitally enabled supply chain integration capabilities. *MIS Quarterly*, 30(2), 225-46.

Ragatz, G, L., Handfield, R, B., & Scannell, T, V. (1997). Success factors for integrating suppliers in the product development. *Journal of Product Innovation Management*, 14(3), 190-202.

Ramsay, J. (1996). Power measurement. *European Journal of Purchasing & Supply Management*, 2(2/3), 129-143.

Ramsay, J. (2005). The real meaning of value in trading relationships. International Journal of Operations & Production Management, 25, 549.

Ramsay, J., & Croom, S. (2008). The impact of evolutionary and developmental metaphors on purchasing and supply management: a critique. *Journal of Purchasing and Supply Management*, 14 (3), 192–204.

Ramsay, J., & Wagner, B. A. (2009). Organisational supplying behaviour: understanding supplier needs, wants and preferences. *Journal of Purchasing and Supply Management*, 15(2),127–138.

Reck, R. F., & Long, B. G. (1988). Purchasing: a competitive weapon. *Journal of Purchasing and Materials Management*, Fall, 2–8.

Rhee, J., Park, T., & Lee, D. H. (2010). Drivers of innovativeness and performance for innovative SMEs in South Korea: mediation of learning orientation, *Technovation*, 30(1), 65-75.

Richey, R. J., Roath, A. S., Whipple, J. M., & Fawcett, S. E. (2010). Exploring governance theory of supply chain management: barriers and facilitators to integration. *Journal of Business Logistics*, 31(1), 237-56.

Ring, P. S., & Van de Ven, A. (1994). Developmental processes of cooperative interorganizational relationships. *Academy of Management Review*, 19(1), 90–118.

Robinson, C. J., & Malhotra, M. K. (2005). Defining the concept of supply chain quality management and its relevance to academic and industrial practice. *International Journal of Production Economics*, 96(3), 315-37.

Rodon, J., Sese, F., & Christiaanse, E. (2011). Exploring users' appropriation and post-implementation managerial intervention in the context of industry IOIS. *Information Systems Journal*, 21, 223-248.

Rose, W., Mann, I. J. S., & Rose, S. (2012). A strategic perspective and taxonomy of supply chain strategies, IUP. *Journal of Operations Management*, 11(3), 6–42.

Rosenbloom, B., & Larsen, T. (2003). Communication in International Business-to-Business Marketing Channels: Does Culture Matter? *Industrial Marketing Management*, 32(4), 309–315.

Rosenzweig, E. D., Roth, A. V., & Dean, J. J. W. (2003). The influence of an integration strategy on competitive capabilities and business performance: an exploratory study of consumer products manufacturers. *Journal of Operations Management*, 21(4), 437-456.

Rothwell, W. J. (2002). Putting success into your succession planning. *Journal of Business Strategy*, 23, 32-37.

Rozemeijer, F. A., van Weele, A. J., & Weggeman, M. (2003). Creating corporate advantage through purchasing: toward a contingency model. *The Journal of Supply Chain Management*, 39(1), 4–13.

Rue, W. R., & Byars, L. L. (2005). *Management skills and applications* (11th ed.). McGraw-Hill Irwin NY.

Rudestam, K. E. & Newton, R. R. (2007): Surviving your dissertation. A comprehensive guide to content and process. (3rd ed.). Sage Publications.

Rupp, T. M., & Ristic, M. (2000). Fine planning for supply chains in semiconductor manufacture. *Journal of Materials Processing Technology*, 107, 390-397.

Saccomano, A. (1998). 'Keeping SCOR'. Traffic World, 255(13), 27-8.

Saeed, K. A., Malhotra, M. K., & Grover, V. (2005). Examining the impact of inter-organizational systems on process efficiency and sourcing leverage in buyer supplier dyads. *Decision Sciences*, 36(3), 365–396.

Sako, M. (2006). Does trust influence business performances? In: R.M. Kramer ed. *Organizational trust: a reader*. Oxford: Oxford Publishing.

Sanchez, L. M., & Nagi, R. (2001). A review of agile manufacturing systems. *International Journal of Production Research*, 39(16), 3561-3600.

Sanders, N. R. (2007). An empirical study of the impact of e-business technologies on organizational collaboration and performance. *Journal of Operations Management*, 25, 1332 1347.

Sanders, N. R. (2008). Pattern of information technology use: the impact on buyer-suppler coordination and performance. *Journal of Operations Management*, 26 (3), 349–367.

Saranga, H., & Moser, R. (2010). Performance evaluation of purchasing and supply management using value chain DEA approach. *European Journal of Operational Research*, 207(1), 197–205.

Sarker, S., & Lee, A. S. (2003). Using a case study to test the role of three key social enablers in ERP implementation. *Information & Management*, 40, 813–829.

Sarkis, J. (2003). A strategic decision making framework for green supply chain management. *Journal of Cleaner Production*, 11(4), 397–409.

Sarkis, J. (2012). A boundaries and flows perspective of green supply chain management. *Supply Chain Management: An International Journal*, 17(2), 202-216.

Scannell, T., Vickery, S., & Droge, C. (2000). Upstream Supply Chain Management and Competitive Performance in the Automotive Industry. *Journal of Business Logistics*, 21(1), 23-48.

Schein, E. H. (1988). Organizational Psychology (3rd ed.). Prentice-Hall.

Schippers, M. C., Den Hartog, D.N., Koopman, P.L., & van Knippenberg, D. (2008). The role of transformational leadership in enhancing team reflexivity. *Human Relations*, 61, 1593–616.

Schmenner, R. W., & Swink, M. L. (1998). On theory in operations management. *Journal of Operations Management*, 17, 97–113.

Schoenherr, T., & Swink, M. (2012). Revisiting the arcs of integration: cross-validation and extensions. *Journal of Operations Management*, 30, 99–115.

Schuler, R., & Jackson, S. E. (2014). Human resource management and organizational effectiveness: yesterday and today. *Journal of Organizational Effectiveness: People and Performance*, 1(1), 35-55.

SCOR. (2010). Retrieved October 5, 2011, from: http://www.supply-chain.org.

Seo, Y. J., Dinwoodie, J., & Kwak, D.W. (2014). The impact of innovativeness on supply chain performance: is supply chain integration a missing link? *Supply Chain Management: An International Journal*, 19(5/6), 733 – 746.

Seong-Tae, K., & Sang-Yoon, L. (2012). Stakeholder pressure and the adoption of environmental logistics practices. *International Journal of Logistics Management*, 23(2), 238-258.

Seuring, S. (2004). Integrated chain management and supply chain management: comparative analysis and illustrative cases. *Journal of Cleaner Production*, 12(8).

Seuring, S., & Muller, M. (2007). Integrated chain management in Germany - identifying schools of thought based on a literature review. *Journal of Cleaner Production*, 15, 699-710

Seuring, S., & Muller, M. (2008). From a literature review to a conceptual framework for sustainable supply chain management. *Journal of cleaner production*, 16(15), 1699-1710.

Sharma, M. K., & Bhagwat, R. (2007). An integrated BSC-AHP approach for supply chain management evaluation. *Measuring Business Excellence*, 11(3), 57-68.

Sharp, J. M., Irani, Z., & Desai, S. (1999). Working towards agile manufacturing in the UK industry. *International Journal of Production Economics*, 62(1/2), 155-69.

Sherehiy, B., Karwowski, W., & Layer, J. K. (2007). A review of enterprise agility: Concepts, frameworks, and attributes. *International Journal of Industrial Ergonomics*, 37, 445–460.

Shi, Y., & Gregory, M. (2005). Emergence of global manufacturing virtual networks and establishment of new manufacturing infrastructure for faster innovation and firm growth. *Production Planning & Control*, 16, (6), 621-631.

Shub, P. A. N., & Stonebraker, W. (2009). The human impact on supply chains: evaluating the importance of "soft" areas on integration and performance. *Supply Chain Management: An International Journal*, 14(1), 31 - 40.

Simatupang, T. M., Wright, A. C., & Sridharan, R. (2004). Applying the theory of constraints to supply chain collaboration. *Supply Chain Management: An International Journal*, 9 (1), 57-70.

Simchi-Levi, D., Kaminsky, P., & Simchi-Levi, E. (2000). *Designing and Managing the Supply Chain*. (1st ed.). New York: Irwin McGraw-Hill.

Simchi-Levi, D., Kaminsky, P., & Simchi-Levi, E. (2004). *Managing the supply chain: The definitive guide for the business professional*. New York, NY: McGraw-Hill.

Singh, P. J., & Smith, A. J. R. (2004). Relationship between TQM and Innovation: An Empirical Study. *Journal of Manufacturing Technology Management*, 15(5), 394–401.

Singh, P. J., Power, D., & Chuong, S. C. (2011). A resource dependence theory perspective of ISO 9000 in managing organizational environment. *Journal of Operations Management*, 29(1-2), 49-64.

Sink, H. L., & Langley, C. J. (1997). A managerial framework for the acquisition of third-party logistics services. *Journal of Business Logistics*, 18 (2), 163–190.

Skinner, W. (1969). Manufacturing: missing link in corporate strategy. *Harvard Business Review*, 47(3), 136-145.

Skinner, W. (1985). Manufacturing: The Formidable Competitive Weapon. New York: John Wiley.

Skinner, W. (1996). Manufacturing strategy on the S curve. *Production and Operations Management*, 5(1), 3-14.

Slack, N., Chambers, S., & Johnston, R. (2004). *Operations Management* (4th ed.). Harlow: Pearson Education.

Smith, J. S., Karwan, K. R., & Markland, R. E. (2007). A note on the growth of research in service operations management. *Production and Operations Management*, 16(6), 780–790.

Smith, B. N., Montagno, R. V., & Kuzmenko, T. N. (2004). *Journal of Leadership and Organizational Studies*, *10*(4), 80-91.

Solvang, W. D., & Hakam, M. H. (2010). Sustainable Logistics Networks in Sparsely Populated Areas. *Journal of Service Science & Management*, 3(1), 72-77.

Sorenson, O. (2003). Interdependence and adaptability: organizational learning and the long-term effect of integration. *Management Science*, 49(4), 446–463.

Sriram, V., & Stump, R. (2004). Information technology investment in purchasing: an empirical investigation of communications, relationships and performance outcomes. *Omega*, 32 (1), 41–55.

Srivastava, S. K. (2007). Green supply chain management: A state of the art literature review. *International Journal of Management Review*, 9, 53–80.

Sroufe, R., & Curkovic, S. (2008). An examination of ISO 9000:2000 and supply chain quality assurance. *Journal of Operations Management*, 26(4), 503-520.

Stank, T., Keller, S., & Closs, D. (2001). Supply chain collaboration and logistical service performance. *Journal of Business Logistics*, 22(1), 29-48.

Stank, T. P., Davis, B. R., & Fugate, B. S. (2005). A strategic framework for supply chain oriented logistics. *Journal of Business Logistics*, 26(2), 27-45.

Starr, M. K. (1984). Global Production and Operations Strategy. *Columbia Journal of World Business*, 19 (4), 17-22.

Stern, P. N. (2007). *On solid ground: Essential properties for growing grounded theory*. In: A. Bryant & K. Charmaz, eds. The Sage handbook of grounded theory. Los Angeles: SAGE Publications.

Stevens, J. (1989). Integrating the supply chain. *International Journal of Physical Distribution and Materials Management*, 19(8), 3–8.

Stevens, G. C. (1990). Successful supply chain management. Management Decision, 28(8), 25-30.

Stock, J. R., Boyer, S. L., & Harmon, T. (2010). Research opportunities in supply chain management, *Journal of the Academy of Marketing Science*, 38(1), 32-41.

Stonebraker, P. W., & Afifi, R. (2004). Toward a contingency theory of supply chains. *Management Decision*, 42(9), 1131-44.

Storey, J., Emberson, C., Godsell, J., & Harrison, A. (2006). Supply chain management: theory, practice and future challenges. *International Journal of Operations & Production Management*, 26(7) 754 – 774.

Stratopoulos, T., & Dehning, B. (2000). Does successful investment in information technology solve the productivity paradox? *Information & Management*, 38(2), 103–117.

Strauss, A. L. (1987). *Qualitative Analysis for Social Scientists*. Cambridge, CA: Cambridge University Press.

Strauss, A. L. & Corbin, J. (1990). Basics of Qualitative Research: Grounded Theory Procedures and Techniques. Newbury Park, CA: Sage.

Strauss, A. L., & Corbin, J. (1994). *Grounded theory methodology*. In: Handbook of qualitative research, eds. N.K. Denzin & Y.S. Lincoln, Sage, London, pp. 273-285.

Strauss, A. L., & Corbin, J. (1998). *Basics of Qualitative Research: Techniques and Procedures for Developing Grounded Theory* (2nd ed.). Sage Publications.

Suchman, M. C. (1995). Managing legitimacy: strategic and institutional approaches. Academy Of Management Review, 20(3), 571–610.

Suddaby, R. (2006). What grounded theory is Not. Academy of Management Journal, 49(4), 633-642.

Suzuki, M. (2014). Addressing key issues in technology innovation and transfer of clean energy technologies: a focus on enhancing the enabling environment in the developing countries. *Environmental Economics & Policy Studies*, 16(2), 157-169.

Svensson, G. (2007). Aspects of sustainable supply chain management (SSCM): conceptual framework and empirical example. *Supply Chain Management: International Journal*, 12(4), 262–266.

Swafford, P. M., Ghosh, S., & Murthy, N. (2008). Achieving supply chain agility through IT integration and flexibility. *International Journal of Production Economics*, 116(2), 288-97.

Swink, M., Narasimhan, R., & Kim, S. W. (2005). Manufacturing practices and strategy integration: effects on cost efficiency, flexibility, and market-based performance. *Decision Sciences*, 36(3), 427-57.

Swink, M., Narasimhan, R., & Wang, C. (2007). Managing beyond the factory walls: Effects of four types of strategic integration on manufacturing plant performance. *Journal of Operations Management*, 25, 148–164.

Sydow, J., Lerch, F., Huxham, C., & Hibbert., P. (2011). A silent cry for leadership: Organizing for leading (in) clusters. *The Leadership Quarterly*, 22, 328–343.

Tan, K. C. (2001). A structural equation model of new product design and development. *Decision Science Journal*, 32(2), 195–226.

Tan, K. C., Kannan, V. R., & Handfield, R. B. (1998). Supply chain management: supplier performance and firm performance. *International Journal of Purchasing and Materials Management*, 34(3), 2–9.

Tannenbaum, R., & Schmidt, W. H. (1973). How to Choose a Leadership Pattern. *Harvard Business Review*, May–June 1973, 162–75.

Tarique, I., & Schuler, R. S. (2010). Global talent management: Literature review, integrative framework, and suggestions for further research. *Journal of World Business*, 45, 122-133.

Taylor, D. A. (2004). Supply chains: A manager's guide. Boston: Addison-Wesley.

Technology strategy Board. (2012). *Progressive for future manufacturing*. Retrieved February 23, 2012, from <u>http://www.ifm.eng.cam.ac.uk/uploads/News/2012/23Feb</u>.

Teece, D. J., Pisano, G., & Shuen, A. (1997). Dynamic Capabilities and Strategic Management. *Strategic Management Journal*, 18(7), 509-533.

Terpend, R., & Ashenbaum, B. (2012). The intersection of power, trust and supplier network size: Implications for supplier performance. *Journal of Supply Chain Management*, 48(3), 52–77.

Thakkar, J., Kanda, A., & Deshmukh, S. G. (2009). Supply chain performance measurement framework for small and medium scale enterprises. *Benchmarking: An International Journal*, 16(5), 702-723.

Thome, A. M. T., Hollmann, R. L., Carmo, L. F. R. R., & Scavarda, D. (2014). Research synthesis in collaborative planning forecast and replenishment. Industrial Management & Data Systems, *Industrial Management & Data Systems*, 114(6), 949-965.

Thompson, J. D. (1967). Organizations in Action. New York: McGraw-Hill.

Thompson, A. A., Strickland, A. J., & Gamble, J. E. (2005). *Crafting and Executing Strategy* (15th ed.). New York: McGraw-Hill/Irwin.

Tohamy, N. (2003). Supply Chain Visibility Defined, Research Report. Forrester Research, Cambridge, MA.

Tosi, H. L. (1991). The organization as a context for leadership theory: A multi-level approach. *Leadership Quarterly*, 2, 205-228.

Tracey, M., & Neuhaus, R. (2013). Purchasing's role in global new product-process development projects. *Journal of Purchasing & Supply Management*, 19, 98–105

Trent, R. J., & Monczka, R. M. (2003). Understanding integrated global sourcing. International *Journal of Physical Distribution & Logistics Management*, 33 (7), 607–629.

Trkman, P., & McCormack, K. (2009). Supply chain risk in turbulent environments: a conceptual model for managing supply chain network risk. *International Journal of Production Economics*, 119(2), 247-258.

Tsoulfas, G. T., & Pappis, C. P. (2006). Environmental principles applicable to supply chain design and operations, *Journal of Cleaner Production*, 14(18), 1593-1602.

Tsui, A. S., Zhang, Z. X., Wang, H., Xin, K. R., & Wu, J. B. (2006). Unpacking the relationship between CEO leadership behavior and organizational culture. *Leadership Quarterly*, 17, 113–37.

Uhl-Bien, M., Marion, R., & McKelvey, B. (2007). Complexity leadership theory: Shifting leadership from the industrial age to the knowledge era. *The Leadership Quarterly*, 18(4), 298-318.

Urquhart, C., Lehmann, H., & Myers, M. D. (2010). Putting the Theory back into Grounded Theory: Guidelines for Grounded Theory Studies in Information Systems. *Information Systems Journal*, 20 (4), pp. 357-381.

Useem, M. (2001) 'How to Groom Leaders of the Future'. In: Pickford, J. (ed.) Financial Times *Mastering Management 2.0*. Financial Times Prentice Hall, p.297.

Vachon, S., & Klassen, R. D. (2006). Extending green practices across the supply chain: the impact of upstream and downstream integration. *International Journal of Operations & Production Management*, 26(7), 795–821.

van Echtelt, F. E. A., Wynstra, F., & van Weele, A. J. (2007). Strategic and operational management of supplier involvement in new product development: a contingency perspective. *IEEE Transactions on Engineering Management*, 54(4), 644–661.

Van Hoek, R. I. (1998). Measuring the unmeasurable –measuring and improving performance in the supply chain. *Supply Chain Management: An International Journal*, 3(4), 187-92.

Van De Ven, A. H. (2007). Engaged scholarship: A guide for organisational and social research. Oxford: University press.

Van Weele, A. J. (2010). Purchasing and Supply Chain Management: Analysis, Strategy, Planning and Practice. Florence, KY (5th ed.). Cengage Learning Services.

Vickery, S., Droge, C., & Markland, R.E. (1997). Dimensions of manufacturing strength in the furniture industry. *Journal of Operations Management*, 15(3), 317–330.

Vickery, S.K., Jayaram, J., Droge, C., & Calantone, R. (2003). The effects of an integrative supply chain strategy on customer service and financial performance: an analysis of direct vs. indirect relationships. *Journal of Operations Management*, 21, 523–539.

Vision. (2015). Retrieved April 5, 2015, from:

http://www.ictmalta.org.mt/category/content/strategies/malta-vision-2015.

Vlachos, P. I., & Bourlakis, M. (2006). Supply chain collaboration between retailers and manufacturers: Do they trust each other? *Supply Chain Forum: An International Journal*, 7(1), 70–80.

Vollmann, T. E., Cordon, C., & Heikkila, J. (2000). Teaching supply chain management to business executives. *Production and Operations Management*, 9(1), 81–90.

Von Bertalanffy, L. (1969). *General System Theory: Foundations, Development, Applications*. New York: George Braziller.

Wagner, M. (2013). Green human resource benefits: do they matter as determinants of environmental management system implementation? *Journal of Business Ethics*, 114, 443-456.

Walley, N., & Whitehead, B. (1994). It's not easy being green. *Harvard Business Review*, 72 (3), 46–52.

Walter, A., Muller, T. A., Helfert, G., & Ritter, T. (2003). Functions of industrial supplier relationships and their impact on relationship quality. *Industrial Marketing Management*, 32, 159–169.

Wang, E. T. G., & Wei, H. (2007), Inter organizational governance value creation: Coordinating for information visibility and flexibility in supply chains, *Decision Sciences*, 38(4), 647-674

Wang, P., & Zhu, W. (2011). Mediating role of creative identity in the influence of transformational leadership on creativity: Is there a multilevel effect? *Journal of Leadership & Organizational Studies*, 18, 25-39.

Ward, P. T., Leong, G. K., & Boyer, K. K. (1994). Manufacturing proactiveness and performance. *Decision Sciences*, 25(3), 337–358.

Wasserman, J, A., Clair, J, M., & Kenneth, K. L. (2009). Problematics of grounded theory: innovations for developing an increasingly rigorous qualitative method. *Qualitative Research*, 9(3) 355-381.

Waterman, R., Peters, T., & Phillips, J. (1980). Structure is not organization, *Business Horizon*, and J. Clued, 'Organizational alignment: the 7-S model', HBS Case 9-497-045.

Watson, T. J. (1986). Management, Organisation and Employment Strategy. Routledge & Kegan Paul.

Watts, C.A., Kim, Y.K., & Hahn, C. (1992). Linking purchasing to corporate competitive strategy. *International Journal of Purchasing and Materials Management*, 28, 2–8.

Webster's. (1966). Third New International Dictionary. William Benton, Chicago.

WEF. (2014a). The Global Competitiveness Report 2014–2015. Retrieved September, 11, 2014, from: http://www3.weforum.org/docs/WEF_GlobalCompetitivenessReport_2014-15.pdf

WEF. (2014b).Effective Leadership in International Organizations.Global Agenda Council onInstitutionalGovernanceSystems.September,11,2014,from:http://www3.weforum.org/docs/WEF_GlobalRisks_Report_2014.pdf.11,2014,from:

Weitz, B. A., & Jap, S. D. (1995). Relationship marketing and distribution channels. *Journal of the Academy of Marketing Science*, 23(4), 305–320.

Wernerfelt, B. (1984). A resource-based view of the firm. Strategic Management Journal, 5, 171-180.

Whipple, J. & Frankel, R. (2000). Strategic alliance success factors. *Journal of Supply Chain Management*, Summer, 21-8.

White, A., Daniel, E. M., & Mohdzain, M. (2005). The role of emergent information technologies and systems in enabling supply chain agility. *International Journal of Information Management*, 25, 396–410.

White, R. E., Ojha, D., & Kuo, C. C. (2010). A competitive progression perspective of JIT systems: evidence from early US implementations. *International Journal of Production Research*, 48(20), 6103–6124.

Whitfield, G., & Landeros, R. (2006). Supplier diversity effectiveness: does organizational culture really matter? *Journal of Supply Chain Management*, 42(4), 16.

Wilding, R. (1998). The supply chain complexity triangle: Uncertainty generation in the supply chain. *International Journal of Physical Distribution & Logistics Management*, 28(8), 559–616.

Wilkinson, A., Dainty, A., & Neely, A. (2009). Changing times and changing timescales: the servitization of manufacturing. *International Journal of Operations and Production Management*, 29 (5), 425-429.

Williamson, O. E. (1975). *Markets and Hierarchies: Analysis and Antitrust Implications*. New York: Free Press.

Williamson, O. E. (1985). *The Economic Institutions of Capitalism: Firms, Markets, Relational Contracting*. New York: Free Press.

Winner, R. J., Pennell, J. P., Bertrand, H. E., & Slusarczuk, M. M. (1988). *The role of concurrent engineering in weapons systems acquisition*. IDA R-338. Institute for Defense Analyses, USA.

Wisner, J. D. (2003). A structural equation model of supply chain: Management strategies and firm performance. *Journal of Business Logistics*, 24(1).

Woodruff, R. B. (1997). Customer value: the next source of competitive advantage. *Journal of the Academy of Marketing Science*, 25 (Spring), 139–153.

Wolf, C., & Seuring, S. (2010). Environmental impacts as buying criteria for third party logistical services. *International Journal of Physical Distribution & Logistics Management*, 40(1/2), 84-102.

Wolf, J. (2011). Sustainable supply chain management integration: A qualitative analysis of the German manufacturing industry. *Journal of Business Ethics*, 102, 221–235.

Wolters, T., James, P., & Bouman, M. (1997). Stepping-stones for integrated chain management in the firm. *Business Strategy and the Environment*, 6(3).

Womack, J., Jones, D., & Roos, D. (1990). *The Machine That Changed the World*. Rawlinson Associates, New York.

Womack, J. P., & Jones, D. (1996). Lean thinking: Banish waste and create wealth in your corporation. New York Simon and Schuster.

Wu, G., Ding, J., & Chen, P. (2012). The effects of GSCM drivers and institutional pressures on GSCM practices in Taiwan's textile and apparel industry. *International Journal of Production Economics*, 135(2), 618–636.

Xiao, T., Shi, K., & Yang, D. (2010). Coordination of a supply chain with consumer return under demand uncertainty. *International Journal of Production Economics*, 124, 171–180.

Xu, J., & Thomas, H. (2011). How can leaders achieve high employee engagement? *Leadership & Organization Development Journal*, 32, 399-416.

Yeung, J. H. Y., Selen, W., Zhang, M., & Huo, B. (2009). The effects of trust and coercive power on supplier integration. *International Journal of Production Economics*, 120(1), 66–78.

Yoon, Y., Guthan-Canli, Z., & Schwarz, N. (2006). The effect of corporate social responsibility (CSR) activities on companies with bad reputations. *Journal of Consumer Psychology*, 16(4), 377–390.

Youssef, M. A. (1994). Agile manufacturing; the battle ground for competition in the 1990s and beyond. *International Journal of Operations Production Management*, 14(11), 4–6.

Yukl, G. (2002). Leadership in organizations (5th ed.). Upper Saddle River, NJ: Prentice Hall.

Yusuf, Y. Y., & Adeleyei, E. O. (2002). A comparative study of lean and agile manufacturing with a related survey of current practices in the UK, International. *Journal of Production Research*, 40(17), 4545-4562.

Yusuf, Y. Y., Sarhadi, M., & Gunasekaran, A. (1999), Agile manufacturing: the drivers, concepts and attributes, International Journal of Production Economics, 62 (1/2), 33-43.

Zablah, A., Bellenger, D., & Johnston, W. (2004). An evaluation of divergent perspectives on customer relationship management: Towards a common understanding of an emerging phenomenon. *Industrial Marketing Management*, 33, 475–489.

Zhang, X., & Chen, R. (2008). Examining the mechanism of the value co-creation with customers. *International Journal of Production Economics*, 116(2), 242-250.

Zhang, M., & Huo, B. (2013). The impact of dependence and trust on supply chain integration. *International Journal of Physical Distribution & Logistics Management*, 43(7), 544–563

Zhao, M., Droge, C., & Stank, T. P. (2001). The effects of logistics capabilities on firm performance: customer focused versus information-focused capabilities. *Journal of Business Logistics*, 22(2), 91-107.

Zhao, X., Huo, B., Flynn, B. B., & Yeung, J. H. Y. (2008). The impact of power and relationship commitment on the integration between manufacturers and customers in a supply chain. *Journal of Operations Management*, 26, 368–388.

Zhao, X., Huo, B., Selen, W., & Yeung, J., (2011). The impact of internal integration and relationship commitment on external integration. *Journal of Operations Management*, 29 (1-2), 17–32.

Zhao, L., Huo, B., Sun, L., & Zhao, X. (2013). The impact of supply chain risk on supply chain integration and company performance: a global investigation. *Supply Chain Management: An international Journal*, 18(2), 115-131.

Zhu, Q., Sarkis, J., & Lai, K (2007). Initiatives and outcomes of green supply chain management implementation by Chinese manufacturers. *Journal of Environmental Management*, 85, 179–189

Zhu, Q., Sarkis, J., & Lai, K. H. (2008). Confirmation of a measurement model for green supply chain management practices implementation. *International Journal of Production Economics*, 111(2), 261-273.

Appendix 1: Overview of the Maltese manufacturing scenario

A1.1 The Maltese Business Scenario vis a vis the EU and Global Economy

The impact of the global financial and economic crisis on the industrial activities and the subsequent recovery, within manufacturing, can be clearly seen for the last seven years timeline by Eurostat (2014) for EU member states and a small selection of other world-wide countries, as referred in Figure A1.1 together with a detailed outline of their GDP and their GDP per capita for four different years in Table A1.1. The GDP of various countries contracted due to the 2009 economic global recession, which followed a recovery in 2010. In the year 2012, Brazil and the EU-28 were the only G20 members to record falling industrial output (Eurostat, 2014).

The Maltese wealth performance statistics with its track record in positive performance can be seen in the GDP per Capita, due its relatively high level of standard of living, when compared to other large EU countries, such as UK and Germany, being 86 relative to 104 and 126 as their respective position, with reference to the year 2012, but on the other hand Malta's GDP is then considered very low due its small economies of scale and its small market size, as referred in Table A1.1.



Figure A1.1: Industrial Production Index for EU and G-20 members (Source: Eurostat, 2014, p.101)



Table A1.1: GDP at current market prices, 2002–03 and 2011–13 (Source: Eurostat, 2014)

From a global perspective, both within and outside the EU-member states, the World Economic Forum (WEF, 2014a), in its Global Competitiveness Report (2014–2015), analyzed the competitiveness performance of 144 economies with their microeconomic and macroeconomic foundations of national competitiveness based on 12 pillars of competitiveness. Such a report listed all economies ranking according to their performance to derive each Global Competitiveness Index (GCI), as referred in Figure A1.2. The WEF report outlined that policymakers need to integrate the public sector together with the private sector, by even employing public-private partnerships, who need to work together to promote a more robust economic growth by adopting more-inclusive economies to gain from synergies of the two sectors for improvement not only in productivity but also in social development and environmental stewardship.

The 12 competitive pillars, referred in WEF report, are not independent, since they tend to reinforce each other and as a result a weakness in one area may have a negative impact in others. The World Economic Forum (WEF, 2014a, p.55) defines sustainable competitiveness as: 'the set of institutions, policies, and factors that make a nation productive over the longer term while ensuring social and environmental sustainability'. Furthermore it defines social sustainability as: 'the institutions, policies, and factors that enable all members of society to experience the best possible health, participation, and security; and that maximize their potential to contribute to and benefit from the economic prosperity of the country in which they live' (Ibid). Finally, it defines, 'environmental sustainability as the institutions, policies, and factors that ensure an efficient management of resources to enable prosperity for present and future generations' (Ibid).

From the above, the WEF report clearly outlined the importance that for the economy of any country to be competitive, in line with the competitive benchmarks, must also respect both the social and environmental sustainable factors, since such factors may have a domino negative effect if such actions are not implemented with due diligence (e.g. scarcity of resources, such as water and

agricultural products, extreme weather conditions, due to drastic climate changes, unhealthy environment, due to extreme pollution levels with more labour sickness, higher prices due to lack of resources, lack of sales due to customer green-expectations products, lack of workplace safety, etc).

Most of the businesses nowadays are hence focussing on measures and actions based on more efficient use of resources by using renewable sources of energy and investments in more green friendly products and processes in production and services, to remain competitive through innovative and avante-garde business solutions.



Figure A1.2: Global Competitive Index pillars of competitiveness (Source: WEF, 2014a, p.9)

Malta has been classified in stage 3 in its GCI, being located in the 'innovative-driven' phase within the world economy and with a ranking position of 41 in the year 2013-14, as referred in Figures A1.3 and A1.4, where all the pillars, from 1 to 10, as referred in Figures A1.2 and A1.4, result that the Maltese economy has a substantial competitive level of economic prosperity in view of its relative small domestic market size and its SIS economy. As a result, the Maltese wages have reached above the EU established minimum wage level, with a relatively low level of unemployment rate and with salaries which are adaptable to the position or role of employees. The economy has also reached a relative high level of improvement in the standard of living and in healthcare from a social perspective. The country has also enforced environmental regulations and is engaged in several environmental initiatives as part of the Government and private sector agendas, with a wide programme in renewable energy investments in photovoltaic systems across the Island, waste water treatment, fisheries investments, such as fish-farms and with several other investments, from an environmental perspective, to control and reduce carbon emissions in various sectors. Such an economy, with such measures, is investing not only to achieve an industry with sustainable economic performance but also to reap the benefits from both environmental and social sustainability, with businesses who are able to compete with new and unique products and using sophisticated production

processes (pillar 11) and by innovating new ones (pillar 12) and a the same time with dedicated training and educational programmes to all relevant people.



Figure A1.3: Maltese GDP (Source: WEF, 2014a, p.264)



Figure A1.4: Radar plot of Malta's GCI and Stages of development (Source: WEF, 2014a, p.264)

A1.2 The Economic Statistics of the Maltese Manufacturing SMEs

The Maltese economy over the last decade underwent a gradual shift from the manufacturing towards the services sector, but still manufacturing has still a pivotal role but with a declining contribution to the economy. Such a shift within manufacturing is explained by the fact that the industrial development began in the second half of the 1960s, and by the early 21st century the manufacturing sector was contributing to about 20% of the gross domestic product (GDP). Since the 1980s the manufacture of computer parts, instruments, and electronics, as well as a large variety of consumer products (toys, cosmetics, detergents, and foodstuffs), has been important. In the early 2000s, light manufacturing (e.g. pharmaceuticals, semiconductors, automotive, airplane parts and software) replaced much of the low-cost labour-intensive production that had earlier played a more important role in Maltese manufacturing. Pharmaceutical production in particular has grown rapidly, as a result of the patent law advantages that Malta gained upon EU membership.

For the last ten years, since Malta's EU membership, in every political legislation to-date, there has always been the allocation of substantial EU funds for the manufacturing sector, as part of good governance. Such initiatives are being used to continue to strengthen the local industry through incentives that assist in the development of new technologies and e-business, the sustainability of the energy sector, start-ups, the identification of new markets, research and innovation, international competitiveness and the environment, in line with the climate change EU policies. Further focused incentives are being dedicated to the private sector to attract further new FDI with fast processing of permits to launch the businesses with minimal bureaucracy and by establishing tax credits in all expenses incurred on internationalisation efforts.

This type of support is further enhanced through more grants in tax credits to incentivise and to trigger investment in line with the European Regional Development Fund (ERDF) Grant Schemes. The manufacturing industry investments and commercialisation are also enhanced by creating a common infrastructure, through a private-public partnership, for products design and prototyping of new innovative products and to adapt to the competitive pressures in the global market.

The average full-time employment in the manufacturing sector was 20,803 in 2010, with an industry sectors' segmentation of retail (40%), manufacturing (10%) and general services (50%), such as real estate, hospitality and financial services, as the key GDP sectors which contribute most to the Maltese economy. The local manufacturing industry continues to be characterised by a large number of small firms (those with less than 10 employees), which are referred as micro enterprises, which together with SMEs, are contributing to the nation's economy, job creation and innovation. The larger firms, however, account for over 90 percent of total manufacturing output. The portion of micro, small and medium enterprises in Malta represents 99,9% of the total number of enterprises, showing the relative few amount of large classified firms. SMEs are classified by 'less or equal' values in size, with respect to employee count and turnover, in million of Euros, where small sized firms are classified as (50, 10) and medium sized firms as (250, 50) respectively (European Commission, 2003).

Over 200 export-oriented foreign companies operate profitable manufacturing subsidiaries in Malta, within the micro, SMEs and large classifications respectively, which are also benefiting from attractive incentives. In May 2004, with Malta's accession to the EU, goods produced in the EU, or goods that were already in free circulation in the EU, were exempted from the payment of customs duties, and only goods exported to non-EU countries are subject to the payment of export duties. From an SME perspective in all sectors, they employ about two-thirds of the private sector workforce, and contribute to more than half of the country's value added (Ernst & Young, 2013).

The distribution of the manufacturing industry in Malta is governed by a large number of indigenous SMEs which includes also a substantial larger number of micro business units of less than 10 employees with less or equal 2 million of Euros of revenue (European Commission, 2003), which exceed the count for both all SMEs and large firms. In fact, the industry overall sectors (not only manufacturing) is mainly dependent on the micro sized firms since they amount to 68,743 units whereas the small, medium and large amount to 1572, 384 and 83 units respectively (NSO, 2012), as illustrated in Figure A1.5, which only excludes the micro-sized statistics. It is to be noted that these statistics need to be seen from the perspective of the contribution to the GDP, which is function of a complex set of variables, where the large-sized firms, although they are relatively few, contribute to the largest amount of revenue to the GDP. Malta has a strong economy with an overall manufacturing industry export revenue greater than 38% on the overall import in the year 2013, as referred in Table A1.2. The previous Table also shows the comprehensive list and different types of manufacturing sectors located in Malta, which are referred by the standard NACE coding system.



Figure: A1.5 Types of firms by size distribution in Malta with number of units (Left) and employment size (Right) (Source: NSO, 2012)





Table A1.2: Manufacturing by NACE code with imports and exports levels in Euros (Source: NSO, 2014)

The Maltese manufacturing sector, as it happens in most countries in the World, has a substantial contribution to the GDP. The GDP contribution and its GDP growth for a period of three years are referred in Table A1.3, where the contribution is increasing but the rate of growth is decreasing.



Table A1.3: Gross domestic product at market prices (Source: NSO, 2013)

A1.3 Malta's infrastructure within the manufacturing SC

Malta boasts some of the finest natural harbours in the world, although it is subjected to the insularity and peripherality challenges relative to various other EU member states. Extensive conventional and roll-on/roll-off services by national and international shipping lines carry freight and cargo from Malta directly to Mediterranean, North European, Middle Eastern and Asian ports. All factories are located within 30 minutes of the two harbours and the airport. For all local based SC actors, Malta is an ideal candidate that gains from local sourcing advantages and close-relationships, due to the proximity issues, as a small island, which promotes better teamwork and can exploit the enhanced collaborative environment with all stakeholders, through trust building, as a result of all the face-to-face interactions. Such close proximity characteristic, apart from an improved social climate also adds to more sustainable operations, with improvements in both economic and environmental targets.

The key infrastructural advantages attributed to Malta as a Small Island State are:

The Freeport: The Malta Freeport Corporation embraces three prime activities namely, container handling, industrial storage and oil products handling. The Freeport is recognised as a high profile transhipment hub and presently enjoys third place amongst all Mediterranean transhipment ports. It handles over one million TEUs per annum (twenty-foot equivalent units) and has network connections to over 95 ports world-wide.

Air transport: Air connections with major European destinations are efficient and frequent. Twelve legacy carriers operate scheduled air services to 37 destinations. The national airline, Air Malta, operates regular scheduled flights to the major European cities—a total of over 45 direct destinations with 200 flights a week. Low-cost carriers are gaining in popularity, accounting for over 25 per cent of all departures. The Malta International Airport, which handles some 2.5 million passengers annually, is a modern, spacious and efficient terminal.

Postal services: Postal services are efficient and reliable. Letters to Europe normally take two days to reach their destination. For faster service the major international courier service companies operate to and from Malta (e.g. UPS and DHL). The Malta Post Office operates an Expedited Mail Service (EMS Datapost) with guaranteed delivery times.

Telecommunications systems: Malta has one of the leading telecoms network backbone, with four mobile and broadband operators, which were upgraded according to plans drawn up by the International Telecommunications Union (ITU). International connections have been significantly expanded through satellite technology and a high-capacity fibre-optic cable links between Malta with Europe. Internet usage by enterprises stood at 94 per cent in 2010, while 70 per cent of households had access to internet at home.

Human resources: Malta is reaping the fruit of the massive investment in the educational sector in the past years so much so that 83 per cent of students were continuing with post secondary education (NSO, 2010). The government invested massively to improve educational standards and introduced major reforms including the college system, whereby students were getting a more holistic education. The labour force in Malta is a very productive one, highly educated, multi-lingual and extremely flexible with an excellent work ethic. The people are the greatest natural resource and the country has good availability of professional, managerial and technical staff as well as a ready supply of top graduates most of whom are technology-experienced.

The Research and Innovation Strategy (Vision, 2015) for Malta seeks to focus the national investments in Research and Innovation (R&I) for economic growth in the following priority areas: Energy and Environment; ICT; Health and Biotech; and Value Added Manufacturing.

Item N.o.	Dimension	Dimension variables	Specific Authors
1	Information integration (includes information associated with physical flow)	Information sharing with customers/suppliers (such as orders levels, involvement in design etc)	Bagchi et al. (2005); Cagliano et al. (2006); Das et al. (2006); Frohlich and Westbrook (2001); Devaraj et al. (2007); Koufteros et al. (2007); Lambert & Cooper (2000); Power (2005); Rosenzweig et al., (2003); Stank et al., (2001) ; Swink et al., (2007); Cassivi (2006).
		Information technology/e- business/information systems capabilities SC visibility	Bagchi et al., (2005);Chen et al. (2004); Cagliano et al. (2006); Devaraj et al. (2007); Power (2005); Vickery et al. (2003); Stank et al. (2001); Droge et al. (2004); Kim (2009). Bagchi et al. (2005); Devaraj et al. (2007);Kim
			(2009); Koufteros et al. (2007); Lambert & Cooper (2000).
		Strategic and Planning information access by various functional areas within (e.g. purchasing) and outside the focal firm	Bagchi et al., (2005); Cagliano et al. (2006).
		Information sharing risks (e.g. propriety information)	Bagchi et al. (2005).
		Forward and reverse information flow	Cagliano et al. (2006).
		Leverage information from others (suppliers, customers etc)	Cagliano et al. (2006); Koufteros et al. (2007).
		Align and streamline processes within and across the SC tiers	Cagliano et al. (2006); Kim (2009); Lambert & Cooper (2000).
2	Strategic integration (with	Manufacturing through lean/agile (technical initiatives)	Cagliano et al. (2006); Gunasekaran et al., (2004c); Kim (2009);Yusuf et al. (2002); Kros et al. (2006); Green & Inman (2005).
	suppliers, customers,	Manufacturing through ERP system	Cagliano et al. (2006); Power (2005).
	internal and product-	Computer aided manufacture (CAD, CAM, CIM etc)	Droge et al. (2004); Gunasekaran et al. (2004c); Vickery et al. (2003); Gunasekaran et al. (2002).
	process integration)	Standardized practices Knowledge and learning (culture)	Droge et al. (2004). Das et al.,(2006); Gunasekaran et al., (2004c); Koufteros et al. (2007); Pagell (2004).
		Strategic planning	Gunasekaran et al. (2004c); Power (2005).
		Virtual enterprise organisation structure (virtual integration or virtual organisation) as a form of partnerships, alliances and cooperation	Gunasekaran et al. (2004c);Kim (2009); Pagell (2004); Power (2005); Gunasekaran et al. (2008).
		Intra- and inter-managed organisational processes (forward and reverse information and physical flows)	Flynn et al. (2010); Gunasekaran et al. (2004c); Frohlich & Westbrook (2001); Kim (2009); Koufteros et al. (2007); Lambert & Cooper (2000); Rosenzweig et al. (2003); Vickery et al. (2003); Swink et al. (2007); Malhotra et al.(2005); Simchi-Levi et al. (2004).
		Competitive capabilities (cost leadership, differentiation and innovation)	Bagchi et al., (2005); Chen et al. (2004); Kim (2009); Lee (2000); Power (2005); Rosenzweig et al. (2003); Swink et al., (2007); Gunasekaran et al. (2008); Gunasekaran et al. (2004b); Koh et al. (2004); Flynn & Flynn (2004).
3	Strategic	Customer focus/close customer	Chen et al. (2004): Frohlich and Westbrook

Appendix 2: A Priori Literature Review Significant Concepts

	Collaboration (including tactical and operational)	Relationships (e.g. sales administration)	(2001); Bagchi et al., (2005); Narasimhan & Kim (2001); Droge et al. (2004); Rosenzweig et al. (2003): Stank et al. (2001)
		Strategic purchasing	Chen et al. (2004) ; Gunasekaran et al. $(2004c)$;
		Strategie purchasing	Vickery et al., (2003); Giunipero et al., (2005).
		Top management support	Chen et al., (2004); Stank et al., (2001).
		Information technology (e.g SC software)	Chen et al., (2004); Cagliano et al., (2006); Frohlich and Westbrook (2001); Power (2005); Vickery et al., (2003); Bagchi et al., (2005); Sanders (2007); Frohlich and Westbrook (2002); Harland et al., (2007); Devaraj et al., (2003).
		Supply network structure (close suppliers relationships)	Chen et al., (2004); Rosenzweig et al., (2003); Vickery et al., (2003): Giunipero et al., (2005)
		Long-term relationship with	Chen et al., (2004); Koufteros et al., (2007);
		suppliers and buyers (risk and	Stank et al. (2001); Zhao et al. (2008).
		reward management)	
		Supply base reduction	Chen et al. (2004); Koufteros et al. (2007)
		Communication	Chen et al. (2004); Cagliano et al. (2006);Das et al. (2006); Gunasekaran et al. (2004c); Koufteros et al. (2007); Pagell (2004); Power (2005).
		Inter and intra cross-functional	Chen et al. (2004); Cagliano et al. (2006); Das
		teams/resource sharing	et al., (2006); Koufteros et al., (2007);Lambert & Cooper (2000); Lee (2000); Rosenzweig et al., (2003); Stank et al., (2001); Vickery et al. (2003); Gunasekaran et al. (2004c); Ajmera et al. (2009): Pagell (2004); Zhao et al. (2008).
		Logistics integration (e.g.	Chen et al. (2004); Kim (2009); Rosenzweig et
		distributors/retailers)	al. (2003); Stank et al. (2001); Cambra-Fierro & Ruiz-Benite (2009).
		Information sharing (e.g. production information with both customers/suppliers)	Cagliano et al. (2006); Das et al., (2006).
		Collaborative planning and/or forecasting	Devaraj et al. (2007); Lee (2000); Stank et al. (2001).
		Inventory management (suppliers- side)	Bagchi et al. (2005).
		Joint decision-making and problem solving	Bagchi et al. (2005); Lee (2000); Stank et al. (2001); Narasimhan & Kim (2001); Das et al. (2006); Ajmera et al. (2009).
		Risk management from shared Information systems	Bagchi et al. (2005).
		Performance measurements	Pagell (2004); Rosenzweig et al. (2003); Stank
		(operational and business) and	et al. (2001) ;Vickery et al. (2003);Chen et al.
		rewards (includes both forward	(2004); Das et al. (2006); Deveraj et al. (2007);
		focal firm functional boundaries	(2009): Swink et al. (2007): Neely et al. (2005):
		and across the overall SC tiers)	Gunasekaran et al. (2001); Boone & Ganeshan (2007).
4	SC/IT	ERP-based system/e-business	Bagchi et al. (2005); Cagliano et al.,
	technology	(using XML on-line access on the	(2006);Deveraj et al. (2007); Power (2005);
		internet or virtual private network)	Vickery et al. (2003); Nurmilaakso (2008); Power (2005); Cassivi (2006); Laudon &

			Laudon (2000); Huang & Gangopadhyay (2004)
		Direct access IT infrastructure both intra and inter-organisational	Bagchi et al., (2005)
5	Operational performance measures	Customer satisfaction/responsiveness (support and speed of service)	Chen et al., (2004); Droge et al. (2004); Flynn et al. (2010); Gunasekaran et al. (2004c); Frohlich and Westbrook (2001); Kim (2009); Rosenzweig et al. (2003); Swink et al. (2007); Vickery et al. (2003)
		Cost (manufacture)	Chen et al., (2004); Das et al. (2006); Frohlich and Westbrook (2001); Rosenzweig et al. (2003) ; Stank et al. (2001); Swink et al. (2007)
		Quality	Chen et al., (2004); Das et al. (2006); Frohlich and Westbrook (2001); Rosenzweig et al. (2003); Swink et al. (2007)
		Delivery (speed)	Chen et al., (2004); Das et al. (2006); Droge et al. (2004); Flynn et al. (2010); Frohlich and Westbrook (2001); Rosenzweig et al. (2003); Swink et al. (2007); Vickery et al. (2003)
		Flexibility (process and product)	Chen et al., (2004); Das et al. (2006); Flynn et al. (2010); Gunasekaran et al. (2004c); Frohlich and Westbrook (2001); Rosenzweig et al. (2003); Swink et al. (2007)
6	Financial performance measures	Market share (performance)	Chen et al., (2004); Das et al. (2006); Droge et al. (2004); Flynn et al. (2010); Frohlich and Westbrook (2001); Swink et al. (2007)
		Return on investment	Chen et al., (2004); Das et al. (2006); Droge et al. (2004); Flynn et al. (2010); Frohlich and Westbrook (2001); Rosenzweig et al. (2003); Vickery et al. (2003)
		After sales profit	Chen et al., (2004); Das et al. (2006); Droge et al. (2004); Flynn et al. (2010); Frohlich and Westbrook (2001).
		Sales growth	Das et al. (2006); Droge et al. (2004); Flynn et al. (2010); Rosenzweig et al. (2003)

Appendix 3: Interview guide for convenient/open sampling

- 1. What is your position (CEO, Senior manager etc); experience in the industry and also on the particular job; education level; firm industry type; headquarters location; company number of employees and company revenue?
- 2. What do you understand by Supply Chain Integration (SCI)?
- 3. What is your experience on SCI initiatives within the organisation (intra perspective) and with other parties (inter perspective) for example with suppliers, firms, customers etc?
- 4. At which management level/s within the business does SCI takes place (for example within strategic, tactical and operational)?
- 5. How do you define the level (depth), if any, of SCI being employed with other firms/suppliers/customers?
- 6. What are the results/outcomes of SCI?
- 7. Are there any special circumstances when SCI is being implemented?
- 8. What type of investment, if any, has been done to promote and build SCI within and across the firms?
- 9. Which are the factors that are promoting/inhibiting SCI?
- 10. SCI is being considered locally (with Maltese firms) and/or internationally?
- 11. How do you consider SCI improvements with time (years of experience) of engagement with each firm/supplier/customer forming up each relationship with your firm?
- 12. How is SCI being achieved with others, considering the variation in: (a) standards (due to different organisational structures and processes); (b) policies (various perspectives and regulations); and (c) cultures (social elements such as behaviour, beliefs and values) held by the other firms engaged in the SC relationship?
- 13. Does the Maltese context (political, economical etc) contribute to the set-up or running of SCI with other firms/suppliers/customers engaged in the SC relationship?

Initial Letter to Executive Directors

(DATE) Name Address of Executive Director

Dear Executive Director,

My name is Ronald Cuschieri and I am conducting research for my Ph. D. dissertation which I am doing with Aston University, UK, at the Operations and Information Management Group, Business School. The objective of my dissertation is "To establish a framework that promotes Supply Chain Integration (SCI) for competitive performance within the Maltese Manufacturing."

I require some volunteers who would be prepared to participate in a research project designed to address the relevance of SCI, in adopting and implementing the criteria to derive the SCI theoretical framework.

Therefore, I am soliciting persons who have experience and expertise and are currently engaged in SCI within the business either/both internally (within their firm itself) or/and externally (with other firms, suppliers and customers) and who are located within any of the three management levels (strategic, tactical and operational). A minimum of a single participant (and a maximum of three participants), wherever applicable, from every management level will be preferably required to participate in the research.

This study requires the involvement and collaboration of a number of participants for a 45 minute interview (as per Interview Guide attached), which is preferably audio recorded. The participants should be knowledgeable of any part of the SCI mechanism and be willing to confidentially share the knowledge they have acquired via their experiences.

It would also be appreciated if access is given to all related documents to the SCI operations such as policies, procedures, etc. All material will of course be treated with confidentiality and will only be used as part of a wider research which does not reveal the identity of individual collaboration. All material given by the participant together with the information and audio recording will respect all data protection requirements and will be safely kept in a secure and private place against malpractice by third parties.

It would be further appreciated if after the completion of the set of interviews, a focus group session/s on SCI, based on the same interview guide, of around two hours duration is set-up, with the same participants or other nominees who had not already participated in the interviews, to create a brainstorming and interactive session. Preferably two to six persons per focus group would be required. This request is optional and no commitment is enforced on the participant who participates in the interview referred above.

Please forward the names of prospective interviewees and the focus group coordinator to me on the email **1** and I will contact your nominees directly for this research exercise, or feel free to have your nominees contact me directly.

It is hoped that together we will all gain from the intellectual exchange provided by this research effort. All the participants may be debriefed after the research is completed to gain from the participation to this qualitative based survey. Again, thank you so much for your time and consideration in this research project. My supervisor is Dr. Prasanta Dey, Aston University, Business School, UK, phone (44) 121 204 4011 and p.k.dey@aston.ac.uk. If you have any questions about this research, you may also contact Dr. Prasanta directly.

Sincerely,

Ing. Ronald Cuschieri (MBA, Henley, UK) Senior Lecturer, University of Malta

Consent Form Letter

(DATE) Name Address of Participant

Dear Participant,

My name is Ronald Cuschieri, Ph.D. candidate working on my dissertation which I am doing with Aston University, UK. I am soliciting some volunteer participants who would desire to be a part of a research project designed to establish a framework based on Supply Chain Integration (SCI). I am therefore seeking individuals who have experience and expertise and are currently engaged in SCI within the business either/both internally (within their firm itself) or/and externally (with other firms, suppliers and customers) within all the three management levels (strategic, tactical and operational). The title of the research project is "To establish a framework that promotes Supply Chain Integration for competitive performance within the Maltese manufacturing."

Participants should preferably respond to all the questions being asked but the interview will also be of value if a participant responds to only a part of the interview guide (as attached). The respondents' replies should preferably be audio recorded due to the nature of the research method for effective analysis and to promote research rigour and validity.

The participant may be a supplier, or a firm or a customer of a firm or a third party who is engaged in the whole Supply Chain activity related to the firm under research. Such a wide range of participants are being included to get a variety of perspectives across the entire Supply Chain network.

A follow-up interview may be required for the researcher to converse with you telephonically or faceto-face (depending on the preference of the respondent) to clarify and/or edit the data gathered.

This researcher is aware of the necessity of keeping information discussed strictly confidential. It may need to be noted that the survey questions are not of a personal nature but do ask the participants to address their experiences in a general nature. Anonymity and confidentiality are guaranteed and no real names will be used in any part of the research and no information will be disclosed to any other participant that may lead to the identification of the possible source of information. There would be no attempts at deceiving any of the participants at any time, although positive and negative responses would be reported accurately. For this research, all participants would be asked to volunteer, be free from deception, be confident of privacy and confidentiality, and be allowed to check accuracy throughout the research.

Please acknowledge your consent to participate in this research, by confirming the information below:

I, _____, do agree to participate in this confidential research and may be contacted on/at ______ concerning this research.

Your participation is done on a voluntary basis and you may withdraw from this study at any time of the process, without any repercussions or concerns that your lack of participation will alter the outcome of this study.

My personal email is

It is hoped that together we will all gain from the intellectual exchange generated by this research effort. All the participants may be debriefed after the research is completed to gain from the participation to this qualitative based survey. Again, thank you so much for your time and consideration in this research project. My supervisor is Dr. Prasanta Dey, Aston University, Business School, UK, phone (44) 121 204 4011 and <u>p.k.dey@aston.ac.uk</u>. If you have any questions about this research, you may also contact Dr. Prasanta directly.

Sincerely,

Ing. Ronald Cuschieri (MBA, Henley, UK) Senior Lecturer, University of Malta

Appendix 4: Holistic SCI management theme with its 16 sub-themes

A4.1 The sixteen sub-themes within the holistic SCI management approach

A4.1.1 Lean Management theme

Lean management theme is concerned about all techniques that need to be used to eliminate and/or reduce all types of waste associated in all manufacturing SC processes including all areas of the business (Womack & Jones, 1996; Cox, 1999; Juttner et al., 2007). Bayraktar et al. (2009, p.133), clearly state the importance of a lean management approach for the elimination of waste and to adopt low inventory strategies, since 'Lean practices have long dominated companies to accomplish excellence in their business operations' (Bayraktar et al., 2009, p.133). The lean management approach is function of the contextual conditions of the firm, since both the type of product and production volume, are the ideal candidates that determine the lean approach in operations.

The scope of the SC holistic process is to minimise waste in all operations within all the process by doing more with less (Juttner et al., 2007). The waste concept has been referred by the seminal works of Dr Shigeo Shingo with his works associated with the Toyota Production System (TPS) with the objective to eliminate the non-value-adding activities for more efficient operations from internal and external production processes, referred as the seven classic wastes of Shigeo Shingo: overproduction, waiting, transportation, unnecessary processing steps, stocks, motion and defects (Hall, 1987).

The quality management standard ISO 9000 is being adopted by lean manufacturing firms and in turn are also adopting ISO 14000, since lean is complementary to pollution and waste reduction (King et al., 2001), where they refer that 'green is lean'. Such an action will improve the environmental performance of the firm (Hart, 1997), especially in today's current triple bottom line perspective of doing business based on GSCM (Sarkis, 2012; Wu et al., 2012; Caniels et al., 2013; Glover et al., 2014; Hsueh, 2015). Lean manufacturing enables less reactive measures to treat waste (i.e. end-of-pipe treatment is avoided whenever possible) due to its proactive measures, but in practice this may lead to a paradoxical situation since it does not always lead to green activities, due to increased efficiencies (i.e. faster operations may generate more waste), less inventory (i.e. less economical use of stock due to higher costs for small orders) and small batch sizes (i.e. more changeovers in production) (King & Lenox, 2001) which all contribute to more waste of resources.

A4.1.2 Managing Sustainability theme

Sustainable development concept was first referred by the Brundtland Report (Brundtland Commission, 1987, p.16), which was defined by the '. . . development that meets the needs of the present, without compromising the ability of future generations to meet their own needs'. Furthermore, such environmental sustainable awareness and preoccupation was also highlighted at that time by Frosch & Gallopoulos (1989, p.144) where they stated that: 'An ideal industrial ecosystem may never be attained in practice, but both manufacturers and consumers must change their habits to approach it more closely if the industrialized world is to maintain its standard of living and the developing nations are to raise theirs to a similar level without adversely affecting the environment'. Such sustainability critical perspective was also referred in the United Nations 2005 World Summit. Furthermore, ISO standards, such as ISO 14001 (ISO14040, 1997), have been issued, based on a life cycle assessment (LCA), to serve as guidelines for environmental management processes for both the SC design and its operations (Bojarski et al., 2009). Additionally, various countries have already started to enforce and issue various legislations for controlling all the processes related to the industry, so as to protect the environment (Lee et al., 2009).

A more contemporary definition, which includes the triple bottom line perspective, is advocated by Carter and Rogers (2008, p. 368), where they define SSCM as: 'the strategic, transparent integration

and achievement of an organization's social, environmental, and economic goals in the systematic coordination of key inter-organizational business processes for improving the long-term economic performance of the individual company and its supply chains'.

Such a sustainable approach has been nowadays in-built in all scientific paradigms in various disciplines, such as SCM, so as to promote sustainable practices in all processes within the context where the resources and capacities are limited (Srivastava, 2007; Bojarski et al., 2009; Buyukozkan & Berkol, 2011). As a result, currently all societies worldwide, including all G-20 major economies, EU members, Governments all across the world, Educational Institutions and industries, have been engaged in such environmental conscious practices to promote a sense of ethical commitment for environmental responsibility around the globe down to all society members (Elkington, 2002; Manderson, 2006; Bojarski et al., 2009). In fact, the literature is replete with various academic work and practitioners' reports, which recommend strategies on how to promote environmental sustainability. The words such as SSCM; GSCM; green operations, triple bottom line, etc has now been integrated in various management literature (Elkington, 2002; Seuring & Müller, 2008; Buyukozkan & Berkol, 2011; Sarkis, 2012; Wu et al., 2012; Caniels et al., 2013; Glover et al., 2014; Hsueh, 2015). Integrated chain management (ICM), is also being used in various countries, such as Germany and Holland, where it defines supply chain management that also takes the environmental and social issues into account, which is leading to the conceptual development of the sustainable SCM concept (Wolters et al., 1997; Seuring, 2004; Seurring & Muller 2007), in line with the previous concepts (i.e. GSCM, SSCM etc). ICM highlights two dimensions, that is those of material and information flow and those of strategy and partnerships in the supply chain (Seurring & Muller 2007).

When customers procure products and goods, they are also expecting, due to their embedded culture of sustainability, that all their purchases originate from sustainable sources based on sustainable design and sustainable manufacturing practices with sustainable deliveries, such as green packaging (Vachon & Klassen, 2006; Zhu et al., 2008; Caniels et al., 2013). Otherwise, such customers are uncomfortable to proceed with the purchase of goods and services. Such a stance is a cyclical process, since the industry needs the customers and as a result all firms are putting all their efforts to meet all the customers' expectations to enhance their green reputation and not to avoid taxation measurements and intense scrutiny by Governments (Caniels et al., 2013).

The Organisations through the effective management of the SC need to comply with all environmental sustainable practices, such as buying practices and environmental impacts of various processes (Mentzer et al., 2001, Perotti et al., 2012; Sarkis et al., 2010; Wolf, 2011; Carter & Easton, 2011; Ageron et al., 2012; Caniels et al., 2013). Such an environmental friendly stance must not only be adopted to build on an appropriate image and/or meet minimum requirements and/or by avoiding any reputational risk, but to remain in business (Caniels et al., 2013). This is referred by Suchman (1995) as legitimate practice, even if it does not give an economic return (Meyer & Rowan, 1977; DiMaggio and Powell, 1983; Berrone et al., 2010). Environmental sustainability is related to waste, emissions and pollution reduction respectively, energy efficiency and a decrease in the consumption of hazardous materials (Darnall et al., 2008; Gimenez et al., 2012). Currently Governments are enforcing regulations and legislations to promote environmental protection within industrial specific sector requirements (Caniels et al., 2013). The organisations need to fit their environmental sustainability requirements within their business strategy and across the whole SC, including suppliers' selection and logistics (Vachon & Klassen, 2006; Hsu et al., 2013; Caniels et al., 2013) and ensure less waste in materials, energy consumption and labour respectively and adopt better use of technology (Berrone et al., 2010; Buyukozkan & Berkol, 2011; Glover et al., 2014). Such environmental practices may be more challenging to achieve in small firms relative to large firms, since the latter higher purchasing power, gives them an edge in undertaking all the necessary investments in implementing such sustainable measures (Caniels et al., 2013; Glover et al., 2014).

Sustainability within the SC 'provides competitive advantages to the companies as it offers the opportunity to differentiate from other companies by being equitable in the fair utilization of natural resources, prudent enough not to harm the environment, being socially responsible in terms of equal human development, ensuring health and safety of employees and contributing to humanity and the environment' (Buyukozkan & Berkol, 2011, p.13731). It is currently pivotal that organisations need to undertake the necessary investments in all human resources for sustainable practices, through training, performance evaluation and rewards (Jabbour et al., 2010; Wagner, 2013). The role of the human element all across the SC, including the end-users, is playing a key role, since such actors are being informed and made aware of all the initiatives of sustainability in all operations of the overall process and it has become a business necessity to employ such environmental friendly initiatives (Gonzales-Benito, 2005; Tsoulfas & Pappis, 2006; Vachon & Klassen, 2006; Zhu et al., 2008; Schuler and Jackson, 2014), which are directly or indirectly supporting such sustainable perspective (e.g. customers expect green-friendly products and to dispose such products appropriately with the intervention of the manufacturer).

The roadmap for the future 2020 in manufacturing research (ManuFuture Platform, 2006) is to transform the European manufacturing industry through a Strategic Research Agenda to derive a knowledge-base sector through five pillars and their associated new enabling technologies for the industrial transformation, with in-built sustainable measures, refer:

- new, added-value products and services;

- new business models;

- new advanced industrial engineering;
- new emerging manufacturing science and technologies; and
- transformation of existing R&D and education infrastructure to support world-class
- Manufacturing.

The International Energy Agency (EEA) is also promoting to deploy sustainable, reliable, affordable and clean energy both within and outside EU to deploy efficient technologies and systems optimisation in the energy consumption especially within the manufacturing sector (EEA, 2012), such as with the use of 'efficient compressors, sharing of resources across the SC and the deployment of integrated systems in the plants' (EEA, 2012, p.357).

The role of ICT in all communication and collaborative initiatives across the SC and the investment in manufacturing technology, based in automation of processes, are both enabling and integrating all the SC and manufacturing processes to promote a sustainable holistic integrated SC. Such a stance covers all the processes from a lifecycle management approach, with all the necessary information management capabilities, from the product design to the delivery, including the end-of-life of the product to ensure sustainable operations even within the product disposal phase, to ensure, that whenever possible, to reuse, remanufacture and recycle, so as to undergo annual savings (Hervani et al., 2005; Srivastava, 2007; Bojarski et al., 2009; Sarkis et al., 2011; Buyukozkan & Berkol, 2011). Examples for such provisions refer to technology assisted energy savings manufacturing mechanisms and tracing of all products with bar-coding systems to promote easily recovery of defects with less waste due to rework.

It cannot be excluded that the viability of taking green initiatives have been questioned regarding economic feasibility, since performance may subjected in a negative way due to the investments costs and time to undertake (Walley & Whitehead, 1994; Cordeiro & Sarkis, 1997). This negative view is debatable since other authors' views have supported its implementation, since it leads to sustainable competitive edge and profitability (Defee & Fugate, 2010). Such a balanced perspective was substantiated by others by advocating that environmental commitment is contingent since it may once be profitable and at other times it is not profitable at all (Zhu et al., 2007).

In a nutshell, the term sustainability today integrates social, environmental and economic responsibilities (Elkington, 2002; Bojarski et al., 2009; Buyukozkan & Berkol, 2011; Gimenez etal., 2012; Lee et al., 2012; Caniels et al., 2013; Glover et al., 2014) and hence environmental sustainability need to be backed by the other two objectives for any organisation to achieve overall sustainability. Such a theoretical perspective has brought at par the environmental sustainability with the already well established operational competitive objectives. The SCM well established operational objectives consist from cost, quality, flexibility and delivery (Hayes & Wheelwright, 1979b; Frohlich & Westbrook, 2001; Chen & Paulraj, 2004; Swink, et al., 2007; Ketchen & Hult 2007; Ketchen, et al., 2008; Kim, 2009; Richey et al., 2010; White et al., 2010), but with such environmental awareness, other objectives have come into play, such as reputation and branding (Yoon et al., 2006; Meehan & Wright, 2012; Caniels et al., 2013). For the focal in-conjunction with all of its stakeholders, to meet the social, environmental and economic challenges so as to promote sustainable operations, must have a broad spectrum of competence to perform effectively and efficiently in line with sustainable manufacturing 'Horizon 2020', under the EU FP 8 programme (EU Commission, 2008). Such an EU framework is promoting the industry and academics institutions to undertake research initiatives for scientific excellence in innovation and industrial developments; technologies innovation; and societal challenges solutions within the sustainability platform.

A4.1.3 Managing Performance Measurement theme

The measuring of performance is pivotal to be in place, for both the focal firm and the performance of all SC actors across the SC, especially with all its partners to disseminate best practices (Fawcett & Magnan, 2004; Kim et al., 2010). To implement an effective performance measurement system with a set of selected measures is a challenging process, since 'selecting the appropriate performance measures is challenging, due to the inherent complexity and interdependence of supply chains' (Flynn et al., 2010, p.60). Furthermore, the lack of availability of the right metrics (i.e. inconsistent and inadequate measures) have a direct influence on SCM competency which also restricts how SC can optimize their operations (van Hoek, 1998; Stock et al., 2010) and also act as a barrier to collaborative and effective SCM initiatives (Fawcett et al., 2008).

There are difficulties on what to measure (i.e. number of metrics) and how to measure performance to achieve a fully integrated SC, with a clear direction for improvement and achievements of goals and at the same time to adopt a balanced framework, which includes both financial and non-financial performance measures (Gunasekaran et al., 2001). Such a challenging perspective was substantiated by Sharma & Bhagwat (2007), where they declared that many companies failed to develop effective performance measures.

The focal firm needs to adopt a set of performance measures to address all the management levels within the organisation command chain (i.e. strategic, tactical and operational), and to assist the effective decision making process at each level (Gunasekaran et al., 2001). Such a measurement approach needs to extend beyond the focal firm and include all SC tiers up to the customers (Storey et al., 2006; Kim et al., 2010), so that the focal firm is in a position to follow its strategies and action plans effectively and to monitor improvements through an ongoing assessment (Cousins & Menguc, 2006; Kim et al., 2010). As Sroufe & Curkovic (2008, p.516), declared 'Without a proper performance measurement system it is difficult to measure, manage, and hold people accountable for their actions. The importance of performance measurement cannot be stressed enough, as good performance measurement systems facilitate a better understanding of processes and products both internally and externally'. Furthermore, the performance measures are also needed to implement the auditing practices of the overall SC (Mondragon et al., 2011).

The pivotal role of performance measures, to serve as a monitoring tool, through a set of established benchmarks, is clearly declared by Fawcett and Magnan (2004, p.72), that 'When performance is measured, performance improves'. Benchmarking objective is to employ a comparison of

performance either internally, within one's own organisation or externally to employ competitive performance with other firms in the same sector (i.e. functional benchmarking) (Estampe et al., 2010). The deployment of the external performance measurement approach also enables a source of comparison through the established sector benchmarks between other SCs peers and/or competitors, so as to lead to an assessment of the firm performance, through its KPIs, and also to promote continuous improvement approach (Gunasekaran et al., 2008; Bayraktar et al., 2009). Mondragon et al. (2011) also sustains such a perspective on the need for an integrated set of measurement tools to take into account both the forward and reverse directions of the SC, so as to establish the SC performance and take all the necessary measures to overcome all limitations in all activities across the SC (such as forecasting and sourcing).

A tool which also assists to measure performance across the manufacturing organisation, which is being used within the SCM concept, is referred as the value stream mapping (VSM) (Mondragon et al., 2011), which is also in line with Porter's (1985) value-chain primary activities. Gunasekaran et al. (2008, p.550) outlined that the five necessary basic functional activities in a value stream for a responsive SC based on a continuous improvement approach, include: '(1) procurement (e.g. maximum purchasing discounts), (2) inbound logistics (e.g. low transportation costs), (3) operations (e.g. low production costs), (4) marketing and sales (e.g. wide product range/high availability), and (5) outbound logistics (e.g. low transportation costs)'. Another tool is the balanced scorecard (Kaplan and Norton, 1992), which has also been applied in SCM to create such a balanced performance measurement approach (Bhagwat & Sharma, 2007).

'When performance is measured and reported, the rate of improvement accelerates' (Fawcett & Magnan, 2004, p.72). Additionally, Kaplan and Norton, (1992, p.71) declared that 'What you measure is what you get'. The performance measures need to include measures associated with both financial, such as profitability, return of assets (ROA) and cash to cash cycle) (Chen & Paulraj, 2004; Wisner 2003) and operational indicators, such as cost, time, quality, delivery and flexibility (Hayes & Wheelwright, 1979b; Neely et al., 1995; Ketchen, et al., 2008; Kim, 2009; Richey et al., 2010; White et al., 2010). Other useful operational indicators are adaptability to change and customer service (Flynn et al., 2010, p.60) and logistics performance (Germain & Iyer, 2006).

It cannot be excluded that performance measurement is even a pivotal tool to achieve performance excellence across the SC or between SC partners, as referred by Kim et al., (2010). They deployed the European Foundation for Quality Management (EFQM, 2003) excellence model, with its five enablers, four results criteria and eight concepts of excellence. Such an EFQM tool was used to evaluate the dyad or overall SC activities, efforts and performance, through a balanced performance assessment approach, so as to serve as a source of performance improvement (Kim et al., 2010). It cannot be excluded the importance of the SCOR model, as already referred in Section 2.6.6. and by the Balanced Scorecard role in performance measurements, as referred above. The ISO 9000:2000 based metrics also provide a very comprehensive measurement approach which is useful in both finding and solving problems (Sroufe & Curkovic, 2008).

A4.1.4 Managing Auditing Operations

Auditing within the SC context is needed to capture the dynamic performance of: materials forward and reverse flow with their logistics; the bi-directional information flow across the SC networks; and the level of integration between SC actors, to provide an overall picture of the closed-loop SC performance (Mondragon et al., 2011). With today's current economic turmoil and the adoption of sustainable legislations there is the need of robust auditing tools to identify the performance across the SC (Ibid). Undertaking auditing of upstream and downstream SC actors provides the ability to evaluate the risks associated with the revenue and purchasing cycles and the sustainability of the profit margin (Huang et al., 2014). Operating within a SC context, the audit mechanisms are beneficial for the focal firm to identify any best practices associated transfer problems to other SC actors and/or establish kaizen improvements (Chong et al., 2012).

A4.1.5 Managing Supply Management theme

With the current SC dynamic and competitive environment across different manufacturing sectors, it has become a normal praxis for the focal firm to focus on a few and critical suppliers (Chandra & Kumar, 2000; Chen & Paulraj, 2004; Fawcett et al., 2008; Cousins et al., 2008; Lamming et al., 2013). Such relationships with few suppliers need to be based on a partnership approach based on a long term relationship (Matopoulos et al., 2007; Baraktar et al., 2009) relative to the rest of the other suppliers, which are kept at an arm's-length, due to the non-critical nature of the procured items (Cadden et al., 2013). Supply management is concerned about an extended SC approach beyond the first-tier SC member/s, to have access to all SC actors, so as to build on the overall focal firm strategy by selecting suppliers on their strategic contribution capabilities (Carter et al., 2000).

The partner type of suppliers need to meet the basic characteristics such as cost, service and quality but also need to meet added criteria such as innovative approach, knowledge and capabilities and cultural fit (e.g. Cadden et al., 2010). Furthermore, such a partnership approach with such suppliers promotes value-added operations, with both lower transaction and production costs respectively (Baraktar et al., 2009). The optimized relationship management between the suppliers and the focal firm, as the buyer, from a TCE perspective (Coase, 1937; Williamson, 1975), promotes optimized transaction costs (i.e. less interaction costs to locate supplies from the market) (Krapfel et al., 1991). The scope of such a supply management approach is to exploit and leverage the power of suppliers (e.g. knowledge and expertise) through building of relationships to promote competitive advantage (Ketchen & Giunipero, 2004; Kahkonen et al., 2015).

The supplier and the buyer relationship is a dynamic concept and undertake improvement, once the buyer increases its attractiveness to the supplier for the mutual benefit (Ramsay & Wagner, 2009). The relationship quality consists from three dimensions: trust, commitment and satisfaction (Walter et al., 2003). Such a relational approach is based on the relationship quality dimensions and not on the contractual obligations (Heide & John, 1992).

The effectiveness of the purchasing and supply management approach is based on the implementation of SCI across the SC to provide value for the SC downstream side customers (Van Weele, 2010). On the other hand, the efficiency of the purchasing and supply management approach is based on the employment of accounting principles, such as overall purchase expenses, low inventory costs and high inventory turnover (Miocevic, 2011). Such an efficiency is pivotal in the purchasing and supply management, since it is well established the relative high costs associated with such a role (Ramsay & Croom, 2008), where every effort need to be done from both the business side (e.g. cost cutting through resources' sharing and lean approach) (Narasimhan & Das, 2001) and supply side (e.g. negotiate attractive prices and improved payment terms) (Chen et al., 2004) of the dyadic relationship.

Purchasing and supply management promotes operational capabilities, such as quality and innovation and technological capabilities developments in products and processes, in-conjunction with suppliers (Das & Narasimhan, 2000) and subsequently it contributes to the financial success of the firm (Saranga & Moser, 2010). The suppliers' quality of service in all supplies is a pivotal strategic concern for the focal firm and as a result the suppliers' performance is closely monitored as a source of quality assurance (Lysons & Farrington, 2006). Such a strategic significance through effective supply management adds value for the customers (Hayes & Wheelwright, 1984; McIvor et al., 1997; Chen et al., 2004; Lindgreen & Wynstra, 2005) and promotes a sustainable competitive advantage (Mol, 2003; Paulraj et al., 2006).

A4.1.6 Managing cash-flow

Cash-flow takes into account the inflows and outflows of cash based on the purchasing of raw materials and utilities and the sales of products to customers (Guillen et al., 2007). Guillen et al. (2007, p.288) further stated that:

'Traditional SC models focus solely on determining the profit or revenue-maximizing, or costminimizing production schedule in a SC that runs from suppliers to manufacturing plants to distribution facilities and finally to retail outlets. These models neglect the cash-flow consequences ...'

As a result, nowadays SCM competency needs to manage effectively the working capital efficiency as one of the major drivers to financial performance together with revenue growth and operating costs (Camerinelli, 2009). SCM competency depends on the level of integration across the SC (Chen et al., 2009; Fawcett et al., 2008). The supply chain finance is associated with the optimization of the financial structure and the cash-flow within the supply chain (Gomm, 2010) by the speed up of cash-flow between the focal firm and all other SC actors, through the settlement of all the necessary payments, to improve on the cash-flow performance (Carmichael & Balatbat, 2010; Cho et al., 2012).

Cash-flow is a dynamic payment process, where payments terms are based on a time-schedule, function of the performance within the dyad contractual agreements (Ellram et al., 2004). Firms need to predict and plan cash-flows in such a way to slow cash outflows and speed-up cash inflows to achieve improved return on investment (Chen, 2012). The ability to leverage information technology and process innovation to sustain cash-flow is regarded as a significant source of competitive advantage (Ellinger et al., 2012).

A4.1.7 Managing change and innovation

The change management approach has been defined as a shift from traditional values to new ideas and beliefs (Lewins, 1947) whilst the continuous improvement is defined as an iterative process of planning, changing and evaluating with the involvement of all SC actors to meet improvement (Choi, 1995). Goswami and Mathew (2005) infer that innovation is a very complex and a multi-dimensional concept related to a product or a process or a market and it can bring about a new change or introduce a new process or product or else adopting something new that has been successfully tried earlier. They further add that innovation is considered similar to organisational change and is a major source of competitive advantage. Kadar et al. (2014) sustain Goswami & Mathew's (2005) view but includes organisational change as an added dimension with the product or process or market and does not treat innovation as synonymous to organisational change.

Organisations nowadays need to adopt a change management approach to meet the current dynamic environment made up from changing markets with unpredictable demands, increasing global competition and ever changing customers' demands (Sharp et al., 1999). For survival and prosperity a firm needs to decide on the investments and competitive strategies (March, 1991). Within the organisation learning literature, for a firm to avoid suboptimal performance, there is the need to undergo improvements or a change management approach through a trade-off between two initiatives of exploitation (e.g. gains from efficiencies in production and implemented processes) and exploration (e.g. gains from innovation and risk taking measures, such as to adopt new approaches in products or processes) based actions (Ibid). It is to be noted that results from exploitation measures are more quick and precise relative to exploration measures, due to the different nature of the actions needed to be implemented (Ibid).

All processes need to have an acceptable level of quality but at the same time must adopt a continuous improvement approach so as to reduce costs (Parmigiani et al., 2011) and promote change (Prajogo &
Sohal, 2003). Quality standards, such as ISO 9000, when used in a proactive way and through an advanced implementation approach, (i.e. not just basic compliance), they become a catalyst of change (Prajogo et al., 2012; Huo et al., 2014). In fact, innovative companies create cutting edge products and services through various options, such as technological innovation, quality, efficiency or style (Miller, Furthermore, there exist a significant relationship between quality, from a TQM perspective, 1988). and innovative performance, especially with respect to process innovation, which is more pronounced than product innovation (Prajogo & Sohal, 2003). The product innovation refers to the level of new products; the speed of new product development; the number of products introduced in the market and the products that are first-to-market (Ibid). The process innovation refers to the technological competitiveness; updateness of technology used in products; speed of adoption of latest technological innovations; and the rate of change in processes, techniques and technology (Ibid). Such an innovative perspective must also be seen from a wider consideration of environmental sustainable practices, where innovative practices for low carbon technologies and clean energies is an international challenge to be part and parcel of all work practices and processes for manufacturers (Suzuki, 2014). Furthermore, GSCM initiatives have a significant impact on technological innovation (Lee et al., 2014).

Seo et al. (2014, p.733) argue that for manufacturers, engaged within the value-chain activities in various supply chains, state that: 'Increasingly, organisations are realising that their level of innovativeness in supply chains is integral to strategic success and long-term survival [and]... achievement of sustainable competitive advantage'. They further added that innovativeness is enhanced with more integration between SC actors and with the result of improved efficiencies and effectiveness in both physical and information flows. Innovativeness is about trying to capture new opportunities and ideas and not to build on the current strengths (Panayides & Lun, 2009; Rhee et al., 2010). Calantone (2002, p. 113) defined innovativeness as 'the capacity of a new innovation to influence the firm's existing marketing resources, technological resources, skills, knowledge, capabilities, or strategy'. A SC environment based on a concrete foundation of technology, knowledge and relationships facilitate innovation (Chapman et al., 2003). To achieve innovation-oriented processes there is the need for all SC actors involved to undertake major organisational changes (Tracey & Neuhaus, 2013).

Innovativeness has an indirect impact on SC performance through the mediation of internal and external SCI initiatives, as a catalyst for such performance improvements (Rhee et al., 2010; Seo et al., 2014). In fact, the importance of SCI was already asserted by Yusuf et al. (1999) by advocating that an integrated agile supply chain is the twenty-first century's enterprise paradigm with a winning strategy to meet the dynamic customers' requirements efficiently and effectively. Meeting the customers' needs and expectations with the right products, through organisational innovation, is the principle behind the customer focus approach (Juran, 1988). The change and risk management together with the continuous improvement are two from the ten key enablers of agility within manufacturing (Sharp et al., 1999). Change and innovation key enablers are culture (Prajogo & McDermott, 2011), organisational values (Khazanchi et al., 2007; Richey et al., 2010) and operational intellectual capital (Menor et al., 2007). The intellectual capital is defined as all the resources and capabilities that drive organisational performance and value (Bontis, 2000).

Furthermore, Klein and Sorra (1996) contend that organisational values need to fit the innovative practices to achieve the expected benefits. Hong et al. (2009) argue that motivation of people plays a key role in the commitment for any organisational change which is triggered by any distinct and challenging opportunity. Furthermore, Tracey et al. (2013, p.102) reinforce on the importance of the management commitment for change to meet innovation-oriented processes, by advocating that such processes need a 'call for managers willing to challenge entrenched bureaucratic and cultural barriers to make their firms more competitive given the existing global marketplace'. Kadar et al. (2014, p.1083) reinforce the need for the right management approach, by inferring that 'Management of

innovation is a complex task of leadership that aims at a systemic process of change throughout strategic and operational approaches'.

Agility is needed for all SC actors to have the capabilities to align all SC operations (Ismail & Sharifi, 2006) so as to adapt to the changes needed by the customers' demand (Narasimhan et al., 2006; Jain et al., 2008). Such a change management approach is met either proactively or reactively and in a flexible and timely manner (Li et al., 2008) to synchronise the supply and the demand (Swafford et al., 2008). Cases exist where agility within the SC is needed to meet effectively even unforeseen changes (Lee, 2004). The role of the upstream SC actors is accentuated by Johnsen (2009) by contending that the focal firm interacts and exploits the suppliers' innovative competencies to achieve product developments and innovative practices. Although an innovative approach, through upstream integrated synergies, has no direct lever for financial performance, it is a source of competitive edge for sustainable success (Hartmann et al., 2012). Such a firm's innovativeness orientation, for both suppliers and buyers, within the electronics SC context, is also advocated Bryan et al. (2010), by inferring that the innovative culture promotes relationship learning as a leverage mechanism by developing the learning capabilities with the result of improved products and services that creates competitiveness with such cross-cultural relationships and also leads to a higher level of performance.

A4.1.8 Managing collaboration

Hartono and Holsapple (2004, p.20) defined collaboration as 'an interactive, constructive, and knowledge-based process, involving multiple autonomous and voluntary participants employing complementary skills and assets, with a collective objective of achieving an outcome beyond what the participants' capacity and willingness would allow them individually to accomplish'. They further added that the resources are complementary since they are needed to complete the whole task inconjunction with other participants, and that resources may take the form of a participant's skill, who no other participant possess (e.g. diagnosing manufacturing problems) or an asset, that promotes jointwork (e.g. knowledge of financial planning). Collaborative initiatives also include joint planning and knowledge sharing on greener product design, environmental process updates and reduction on logistics waste (Vachon & Klassen 2006). A cooperative environment builds on trust, where both cooperation and trust conceptual elements, influence the SC members to implement GSCM initiatives and long term relationships (Caniels et al., 2013).

The collaborative paradigm within the SC context refers to organizations that operate within a network of inter-dependent relationships developed and fostered through strategic integration (Vachon & Klassen, 2006). Early SCI efforts ended up to a vertical integration approach, but with the advent of a more competitive environment, the SCs were being set-up on a relationship between different SC entities to promote greater flexibility, but without ownership of the SC entity, through dedicated collaborative arrangements, such as **partnerships** with contractual obligations (Stonebraker & Afifi, 2004). The term partnership was defined by Ellram (1995, p.41), as 'an ongoing relationship between two organisations which involves a commitment over an extended time period and a mutual sharing of risks and rewards of the relationship'. Partnering may create several benefits (Fisher, 1996; Barringer & Harrison, 2000; Cadden et al., 2013) and reference is made to Chicksand's (2015) comprehensive list: lower transaction costs; improved management of complex purchases; dealing with contractual uncertainty; the acquisition of scarce resources; cost reduction and functional stability; improved business stability; and increased organisational legitimacy. Such a paradigm shift from vertical integration to an enterprise management approach consists of several collaborative strategies, referred as vertical integration (VIE), or an extended (EE) or a virtual (VE) or a defunct (DE) enterprise respectively (Binder & Clegg, 2006, 2007; Clegg et al., 2012; Clegg & Wan, 2013). The inter-firm collaboration typical methods are Collaborative, Forecasting, Planning and Replenishment (CPFR), Continuous Replenishment (CR), Efficient Customer Response (ECR) and Vendor Management Inventory (VMI) (Co & Barro, 2008; Ajmera & Cook, 2009; Richey et al., 2010).

Supply Chain collaboration has emerged as one of the most promising paradigm for creating long term competitiveness, since together with IT, they provide a robust foundation to the integration of a set of independent firms (Ajmera & Cook, 2009; Clegg & Wan, 2013). Collaboration can increase its competitive edge by extending from an information sharing stance to a joint decision-making and sharing of resources approach, such as in product development and strategic planning (Choi & Hartley, 1996; Ajmera & Cook, 2009; Cadden et al., 2013). Collaboration streamlines all operations by replacing inventory and excess capacity with information (Taylor, 2004).

Nowadays, there is the need for an effective SC performance to focus on relationship mechanisms based on a mix of EE or VE or DE collaborative relationships with various SC actors rather than relying on vertical integration or VIE (Board & Clegg, 2001; Binder & Clegg, 2006, 2007; Wang & Wei, 2007; Clegg et al., 2012). Relationship governance within the SC context, is based on relational norms and joint actions for common goals (Josi & Campbell, 2003), which has caused a shift from a transactional to a collaborative approach, based on mutual trust, information visibility, less duplication, lean management and agility of operations to achieve SC flexibility (Wang & Wei, 2007).

The inter-organizational systems (IOS) within the SC context represent the dual role of IT as 'a control mechanism as well as a relationship building mechanism' (Grover et al., 2002, p. 223). IOS promotes **SCI** (Clegg & Wan, 2013), where an IOS is an IT based process approach to facilitate common operations across the various SC operations to enable collaborative decision making and performance control (Morash & Clinton, 1998) with a reduction in transaction costs, decease in opportunism and increase in monitoring mechanisms (Wang & Wei, 2007). Though, IT still remains as a tool that enables SC visibility, since the relationship building is a more important driver for SC adaptability/flexibility (Wang & Wei, 2007). The human relational element importance is substantiated by Sanders (2007, p.1343), where it was declared that: 'Collaboration is a result of human interactions which can only be supported by IT, one of which are e-business technologies, but not replaced'.

The technology driver is enabling the possibility for a high level of integrated information systems between all SC tiers to enhance all dynamic collaborative efforts (e.g. CPFR and VMI) (Baratt, 2004; Maccoby, 2006), which together with the internet based applications are recognised as the foundations of the future of e-manufacturing with the optimisation of all SC actors operations (Shi & Gregory, 2005).

This collaborative theme leads to today's current theory that the manufacturing SC must be managed, through integration, collaboration and coordination, as a single entity (Chan et al., 2003; Sanders, 2007). The role of the SCI, within all collaborative initiatives, remains pivotal, since successful organizations must now integrate and collaborate with supply chain partners to maintain competitive advantage (Koh et al., 2007) and also achieve operational efficiency (Co & Barro, 2008).

A4.1.9 Managing culture

Hofstede and Bond (1988, p.6) define culture as 'the collective programming of the mind that distinguishes the members of one category of people from those of another' and they added that the cultural traits are 'rather sticky and difficult to change in any basic fashion, although they can often be modified'.

Based on a scholarly paper of fifty-three cultures across the world, within the IBM subsidiaries, Hofstede and Bond (1988) empirically derived the four dimensions attributed to the culture's consequences on the management practices within such organisations: power distance (i.e. power differences between two persons or organisations); individualism/collectivism (i.e. variation from an individual to a collective attitude); masculinity/femininity (i.e. roles assigned function of sex); and uncertainty avoidance (i.e. the search for truth based on established rules and measures). Such cultural

values have a crucial influence on: the organisational structures, such as centralisation of resources when power distance is high and shifts to democratic leadership style when the power distance is low; the integration of groups, where the collectivist cultures are in a position to establish better synergies between people since they promote better group loyalties, than the individualistic approach; the distribution of roles, where men are more assertive and women are more nurturing; and the situations conditions, since with low uncertainty avoidance, people feel more tolerant to accept the cultural differences between people/organisations to work together (Ibid).

Hofstede and Bond (1988) four dimensions represent the key cultural values and beliefs foundations that determine the nature of collaborative initiatives across the SC, since although there may exist cultural differences, they need to be considered as complementary and not based on any member's superior cultural power (Ibid), so as to promote effective management practices across different organisations within the global SC. Later on, Hofstede (1991) added the fifth dimension, which refers to the long-term versus short-term orientation, which constitutes the principle of long-term based relationships vis a vis short term relationships which are function of the type of collaborative relationships engaged between SC members.

A collaborative culture within the SC context is an approach based on the shared values, beliefs and behaviours (Cadden et al., 2013). The term cultural fit, which refers to the compatibility between two parties, was originally associated with mergers and acquisitions Cartwright and Cooper (1993) but currently is being used within the SC context to denote the critical importance of such a concept, by outlining that it is a key requirement for the inter-organisational relationships between SC members to promote successful performance (Whitfield & Landeros, 2006; Fawcett et al., 2008; Braunscheidel et al., 2010). Furthermore, with the current competitive global scenario with the diversities of social cultures (Kale, 1984), based on continuous innovative developments, with the contribution of all SC actors, culture plays a crucial part within the SC strategy, since it is also an enabler of innovation in products or processes (Prajogo & McDermott, 2011). The diversity of social cultures was further explored by Rosenbloom and Larsen (2003), based on Hall's (1976) scholarly work on communication culture, to classify culture into high and low context cultural value. Such a dichotomy was used to categorize the two types of cultures that influence the approach adopted to the relationships between international distributors and exporters. The different approach to the norm was attributed to the high context cultures type of people, since to interact effectively, they rely more on frequent verbal communication mode (e.g. phone calls) than through IT-based text-based communication mode (e.g. e-mails), which is the current and well established approach used between low context based cultures such as in Western people (Ibid).

It has become a widely accepted view that organisational culture has a significant impact on both business and operational firm performance (Cadden et al., 2013). They further added that SC members, as a supplier to buyer dyadic relationship, may either adopt an adversarial or a collaborative culture respectively, which in turn will have a direct negative or positive effect on the overall SC performance respectively, ranging from ideally maximum, to moderate and to poor performance, within a cultural context which is classified as 'best cultural fit'; 'moderate cultural fit'; 'cultural mismatch'; and 'cultural misfit' (Cadden et al., 2013, p.96). Typical SC associated cultures are outcome focus, open exchange of ideas and information and goodwill, which all contribute to positive performance outcomes (Cadden et al., 2013). Such cultural values, within best-performing SCs, consist of suppliers who exploit their collaborative cultural values with the buyers, to strengthen their relationships, where at the same time the buyers shift their adversarial relationship to a more collaborative culture (Ibid). Such a stance causes an iterative strengthening of values between the dyad to improve on the SC performance (Ibid).

The cultural values of people within the organisation determine the teamwork initiatives commitment, within and outside the focal firm to meet all stakeholders' requirements. This is referred as the personculture fit drawn from the interactional psychology literature (Chatman, 1989; O'Reilly et al., 1991). The person and culture varies from one person to another and also from one firm to another respectively, which henceforth determine the attitudes and behaviour in all interactions within different organisational climates (Lawrence & Lorsch, 1967; McKinnon et al., 2003). This is why there must be a collective effort to understand each other values and established procedures through an integrated approach to cultivate the necessary synergies for a win-win situation. Such a SCI approach is enhanced through training initiatives, such as supporting programmes, to promote the necessary skills and knowledge on how to manage such interactions across the SC (Hammer, 2004; Kwon & Suh, 2005). In fact it is considered that the human behaviour is the root of all cultural associated barriers (e.g. resistance to change) and are very challenging to solve, since cultural issues are intangible and more difficult to manage, being based on judgment and not as other organisational tangible problems, having well established solutions (Fawcett et al., 2008).

Kaynak and Hartley (2008, p.472) infer the crucial role for managing effectively, within and outside the focal firm, in both manufacturing and servicing firms, by stating that 'Management leadership is essential in ensuring that a culture of continuous improvement, open communication, and cooperation is developed throughout the supply chain'. They also added that there is the need to adopt the right policies and structures to 'create a work environment in which employees' attention is focused on serving the customer'. This challenging management approach is pivotal to be effectively implemented, since the differences in the SC members' organizational culture create difficulties to achieve effective communication, collaboration and integration across the SC (McCarter et al., 2005). Such cultural differences also influences the negotiation process between buyers and sellers, since different negotiation styles needs to be adopted to suit the different cultural values embedded within each SC actor (Buttery & Leung, 1998). Furthermore, it was outlined that the cultural aspects takes a lower profile in Western negotiations relative to the Chinese negotiations, where the latter take a high context cultural value (Ibid).

Cowan et al. (2015) infer that the cultural values may play a crucial part in the broken relationships between partners within the globalised SC networks due to the cultural differences, as referred by Hofstede and Bond (1988).

A4.1.10 Managing knowledge

Knowledge management deals with how people acquire, exchange and disseminate knowledge to promote learning (Armstrong, 2006). Knowledge is a key resource (Blome et al., 2014) and may consist from either explicit or tacit knowledge (Armstrong, 2006). The explicit knowledge can be codified (e.g. stored digitally) whereas tacit knowledge is found in the people's mind and is acquired from people's experience (e.g. on the job expertise) (Ibid). Technology role is central to the explicit knowledge but it is used as a supporting tool for tacit knowledge (Ibid). Organisations build and accumulate on knowledge over time and such knowledge is embedded in all procedures and rules (March, 1991). The learning occurs by the organisation together with its individuals, so that the individuals build a social belief of the organisation (Ibid).

Within the SC context, tacit knowledge sharing occurs between buyers and suppliers at strategic level, such as in process reengineering and product development (Vachon & Klassen, 2006), where the focal firm, as the buyer, gets access to the supplier's invaluable knowledge based resources to promote product improvements (Tan, 2001), to enhance the customers' value proposition (Kaufman et al., 2000). Tacit knowledge is generated through the cross-functional teamwork approach adopted within the focal firm and joint problem-solving and relationship building outside the focal firm, which at the same time enables SCI build-up with the capabilities of flexibility and adaptability to change (Das et al., 2006). Blome et al. (2014) further sustain that the transfer of knowledge, both internally and externally to the focal firm, is also a key enabler of SC flexibility and is being regarded as pivotal in today's highly dynamic and vulnerable business environment (Blome et al., 2014). Though, it is to be

noted, that as part of the integrated approach between SC actors, all interactions consist from both explicit and tacit knowledge, which both facilitate learning to all participants (Das et al., 2006).

Such tacit knowledge within and across the SC, since it is more difficult to imitate and is a valuable intangible resource, contributes to the firm's competitive advantage (Peteraf, 1993), which is in line with the resource based view of the firm (Wernerfelt, 1984; Barney, 1991) and also in line with the extended resource-based view of a network of firms (Matthews, 2003). On the other hand attention needs to be given to ensure that external SCI, due to its dedicated integrated approach with the established set of few SC actors, may limit its exposure to the external environment, by hindering the assimilation of external knowledge from other sources (Sorenson, 2003) or else may be subjected to knowledge loss through leakages to competitors (Norman, 2004).

Within the context of SC relationships, the social and collaborative interactions between firms across the SC, transfer knowledge between each other. Such knowledge becomes a source of value creation and creates a learning environment, to foster new processes and products innovation (Cavusgil et al., 2003; Goh, 2006; Koufteros et al., 2007; Kahkonen et al., 2015) and may even generate new knowledge (Almuiet & Salim, 2013) to promote a competitive advantage (Dyer & Singh, 1998; Cheng et al., 2008; Blome et al., 2014). There can be suppliers that go beyond the provision of knowledge to the focal firm, but develop the products needed by the focal firm themselves and on their own initiatives (Petersen et al., 2005). Such an innovative climate, through knowledge management, across the SC also improves on other operational capabilities, such as quality and costs, as sustained by Tracey et al. (2013, p.103), by stating that: '... the team approach across organizations enabled a broad-based sharing of contextual knowledge that facilitated high levels of innovation and quality while simultaneously reducing costs and development time'.

The SC needs knowledge and not only information, although it is assumed that there will be no knowledge without information. Hence one needs to go an extra mile to get knowledge out of information by employing the right information management systems (e.g. using DSS, EIS, ERP, etc). Such a stance is advocated by Cheung et al. (2012), where it was highlighted the pivotal role of knowledge-based systems, based on automatic identification technologies (e.g. RFID and EPC information systems), network infrastructures (e.g. internet) and electronic data exchange technologies (e.g. ERP/EDI) within the supply chain landscape. They added that such a stance guides management with efficient and effective information sharing through SC visibility, so as to promote a holistic approach to all the SC activities for an overall performance optimisation and improvements. They further emphasised the importance for such information to be interpreted correctly to contribute to the strategy formulation. Gunasekaran et al. (2008), further contend such knowledge management thinking based on the deployment of technology, by arguing that within the context of an agile manufacturing based SC, one of the three major determinants of a responsive SC, is knowledge management and information technology management initiative (e.g. ERPs, groupware and internet based applications to automate and/or support all the human element based tasks, such as decisionmaking, to derive knowledge from information).

Purchasing and Supply management professionals need to have the right skills and knowledge so as to contribute to both top and bottom line management levels and promote value to the firm (Chen et al., 2004; Ogden et al., 2007) and to enhance cooperation and collaboration across the focal firm functional areas (van Echtelt et al., 2007) with the resultant improvements in both operational and financial performance (Rozemeijer et al., 2003).

The SMEs which are adopting a HVM stance are considered to incorporate the servitization approach to meet the customer service effectively. Such servitization approach also needs knowledge sharing initiatives, to support all collaborative activities associated with the design and service activities performed by the focal firm, even with the customers, apart from the suppliers (MacBryde et al., 2013) and also to capture detailed knowledge on each customer (Leidner, 2010). With today's embedded

culture on sustainability awareness, knowledge is also needed to build internal capabilities to participate even in GSCM initiatives both within and across the SC (Lee, 2008; Caniels et al., 2013).

A4.1.11 Managing customer service

In this current competitive environment, to manage the customer service effectively, the manufacturing firms within the SC need to meet the customers' demand with high quality and low cost products and with an agile response to adapt to all customers' dedicated and dynamic changing needs (Gunasekaran et al., 2008).

Ellinger et al. (2012, p.250) within the SC context advocate that: 'The ability to generate higher levels of customer satisfaction is regarded as an important differentiator and has therefore become a key element of many firms' business strategies'. This means that an effective customer service is becoming a source of competitive advantage (Gebauer & Fleisch, 2007; Smith et al., 2007). Such an effective customer service meets and may even exceed the customers' expectations (Olsen & Johnson, 2003) and contributes to higher returns (Anderson et al., 1994). Integrating the customer, such as by using JIT within the SC (e.g. JIT-selling and JIT-information), promotes improved organisational performance (Green & Inman, 2005). SCI from the literature has positive impacts on the customer service quality (Vickery et al., 2003; Green et al., 2014) and creates value in the customer experience (Esper et al., 2010). Rosenzweig et al. (2003) updated the extant literature on the relationship of the effect of SCI on both sales growth and customer satisfaction, from a direct to an indirect relationship through operational capabilities. SCI also promotes SC visibility, which is pivotal for all firms across the SC, to be informed with the right and timely information to meet the customers' demand in a synchronised approach and at the same time effectively manage all inventories across the SC (Stevens, 1989).

SCM has a pivotal role in the customer service strategy with the objective to serve the customers and build an effective relationship so as to match or customise the customers' expectations and with an overall improved customer service (Wisner, 2003). Such an approach may also deploy effective logistics to speed up the delivery process to the customer, such as through cross-docking and direct deliveries to the door (Wisner, 2003). The collaborative relationships between all SC actors, including suppliers and customers, need to solve problems and manage all complexities (Flynn & Flynn, 1999) and build on the knowledge and expertise of each other to reduce the concept-to-customer cycle time and costs and also improve the overall design effort (Ragatz et al., 1997) from product design through manufacturing to distribution (Frohlich & Westbrook, 2001). The customer relationship includes work practices related to complaint handling, customer satisfaction and long-term relationship establishment (Bayraktar et al., 2009). The customer relationship management (CRM) approach is deployed to develop and lever out the market intelligence for profit maximisation portfolio from customer relationships (Zablah et al., 2004).

Manufacturing firms from the extant literature needs to establish a service culture to increase their revenue by exploiting the value derived from the services given from the products manufactured to their customers (Gebauer et al., 2006). The manufacturing firms to transform themselves into service providers need to revamp both the internal set-up and the external relationships with all SC actors, so as to strengthen the customer's confidence and the supplier's credibility (He & Lai, 2012). Such a paradigm shift to improve on the customer service value proposition consists from a shift from product selling to an integrated product and service offering (Baines et al., 2007). Such an approach is referred as servitization of manufacturing (Lightfoot et al., 2013) or integrated solutions of manufacturing (Davies et al., 2006).

Furthermore, He and Lai (2012) outlined that competing on customer services, although the total offering are more intangible and difficult to evaluate, will render strategic benefits as result of higher barriers to entry and improved revenues, in line with Porter's (1985) competitive forces model.

Within the extant literature, it is repleted with manufacturing performance measures (Neely et al., 2002), but this is not the same for the service performance measures, due to their intangible properties (Doney et al., 2007). Though it cannot be excluded that there exist service performance measures and most commonly applied measures are based on either Parasuraman et al. (1988) or Fitzgerald et al. (1991).

The general used classification for services within manufacturing, which was based on an equipmentbased manufacturers research, is the product-based and customer action-based services (Gebauer, 2008; He & Lai, 2012), where the former refers to the product operations itself, such as after sales service and training and the latter refers to the support in the sales experience to enrich the customer value of the service, such as provision of integrated solutions from an in-depth relationship (He & Lai, 2012). Though, there is a key difference between the two services orientations, since the customer action-based service builds on the product-based service and needs more trust and engagement to share all the information and knowledge so as establish all the customised requirements and to optimise the value-proposition, for the dyad relationship to be a success (He & Lai, 2012). SCI through both operational and strategic integration initiatives, within and outside the focal firm, contribute to both product- and service- action based services, which in turn lead to competitive performance (He & Lai, 2012). Such an approach is also substantiated by Macbryde et al. (2013) where it was argued the importance that, within the manufacturing SMEs SC, in their practice to achieve HVM, need to implement the integration of the design, production and service activities at all management levels and across all functional areas across the SC to achieve a competitive advantage.

A4.1.12 Managing information

Effective communication fosters the flow of information and the focal firm must recognise the importance to share information, by being communicative, with both suppliers and customers to counteract the negative impacts of information safeguarding (Richey et al., 2010). The quality of information depends on the accuracy, adequacy and timeliness (Bayraktar et al., 2009).

All collaborative efforts need to be supported with the right information sharing initiatives to coordinate and improve on the procurement management and material flow through the necessary joint production planning, inventory and production scheduling (Vachon & Klassen, 2006). Information sharing is crucial in SCI (Kahn & Mentzer, 1998; Rai et al., 2006) which promotes SC visibility (Chopra & Meindl, 2001; Francis, 2008) to capture all data to promote improvements in all SC processes, effective decision-making and reduction of risks (Tohamy, 2003). Such a SC visibility promotes transparency of information across the SC which is pivotal for an agile SC (Yusuf et al., 2004). Cheung et al. (2012, p. 3923) defined supply chain visibility by outlining that it 'entails having a holistic aggregation of detailed information for the supply chain interactions, allowing macroscopic detailed view from one end of the supply chain to the other'.

Information gives value-added information by improving coordination across the SC (Hult et al., 2004), promotes learning (Grant, 1996) and breeds trust (Gulati, 1995). Flynn et al. (2010, p.61), also claimed that 'supplier and internal integration help manufacturers reduce mistakes and waste, through information sharing and joint planning and the internal integration provides the link between the suppliers and the customers'. This argument is also reinforced by (Dell, 1999, p.33) where, from the perspective of the company, it is considered that 'new eBusiness technologies facilitate quick information sharing between downstream and upstream partners and enable companies like Dell Computer to trade inventory for information'. On the other hand, from a supplier perspective, the supplier strategic benefits are function of the type and quality of the information passed by the buyer to the supplier (Sanders, 2008).

ICT applications strengthen SCI through the established technologies and systems, that enable access to real-time information and which promotes better customer and supplier integration and also lead to

an interactive effect between the two or more SC tiers (Frohlich & Westbrook, 2001; Devaraj et al., 2007; Fawcett et al., 2007). With timely information sharing, the interaction with suppliers is enhanced by faster decision making, which results in shorter lead times and smaller batch sizes (Cachon & Fisher, 2000). The customer interface, with web technologies, is provided with easier and faster access to information and improved flexibility to all requests and furthermore, speeds up the overall transaction experience (Lederer et al., 2001; Devaraj 2007). Such an improved customer's interface experience, through customers' on-line orders or suppliers' on-line information exchange, is giving the lead to the 'negotiating power of customers' (Cokins, 2003, p.23). As a result, the deployment of technologies and the value added benefits, depend on the extent of such information sharing which SC actors are willing to share (Lee et al., 2000; Patnay et al., 2006; Fawcett et al., 2007) otherwise all investment in technologies is useless (Fawcett et al., 2007).

Such information sharing, within and outside the focal firm with all SC actors, with their distinct way of information sharing mechanisms, transpires into a competitive competence (Handfield & Nichols, 1999), in line with both the resource based view of the focal firm (Barney, 1991) and extended resource based view from all the SC actors perspective (Matthews, 2003). Furthermore, it is also considered as a relational competency, from the relational based theory (Dyer & Singh, 1998), since through information sharing across the SC, the collaborative environment between all SC actors is enhanced (Paulraj et al., 2008) and at the same time reduces transactional costs with an improved customer service (Fawcett et al., 2007), where all these factors also promote a competitive advantage.

In spite of the importance and value added stance through information sharing initiatives, Kemppainen and Vepsalainen (2003, p.716) declare that 'visibility is still limited' across the SC and research indicates that firms consider 'a realistic view on the advantages and risks of information sharing', and as a result 'planning information is shared only selectively'....'relations [are] with selected partners' and 'companies hesitate to expand their coordination efforts beyond order processing and operational scheduling within the dyadic supplier-buyer relationships'. Furthermore, the lack of trust and fear from information leakages to other SC actors inhibits the sharing of certain information (Kuk, 2004). On the other hand due attention needs to be given to manage all information effectively to foresee situations against information overload (Mendelson, 2000).

A4.1.13 Managing logistics

Bowersox (1997, p.124) defined that: 'Logistics is that part of the supply chain process that monitors and implements the flow of goods and information from the point-of-origin to the point-of-consumption in order to meet customers' requirements'. Intermodal platforms requires efficient and flexible integration with the right consideration of technical, legal and commercial aspects and an integrated management throughout the SC to decrease logistics costs and improve on the service levels (Cambra-Fierro & Ruiz-Benite, 2009). Effective logistics management enables the integration and management of key business processes across the supply chain (Cooper et al, 1997; Markley & Davis, 2007; Stank et al., 2005) to remain competitive (Ragatz et al., 1997; Zhao et al., 2001; Giunipero et al., 2005). The competitive capabilities through effective logistics management are focused on both time and quality objectives and are based on four conceptual capabilities: demand-management, supply-management, information-management and coordination (Mentzer et al., 2004).

The concept of logistics integration refers to managing information and material flows (Frohlich & Westbrook, 2001). Unforeseen events require such a logistical integration to meet the flexibility needed in transportation of supplies (Cachon & Fisher, 2000). On the other hand, low logistical integration requires detailed contracts and provides limited flexibility in the transaction details and where information sharing is dedicated only to order monitoring (Vachon & Klassen 2006).

Nowadays, with the opening of markets and increased internationalisation, there is the need for logistics to guarantee deliveries in the right place and at the right time (La Londe, 1983; McKinnon et

al., 2010) and also to provide a customised service approach (Mentzer, 2004). Logistics has consequently become a decisive factor in the success of a supply chain (Solvang & Hakam, 2010). With the current focus on SC sustainability, Wolf & Seuring (2010) assert that logistics service providers nowadays are expected to meet environmentally friendly transportation. Such an environmental scenario need to be seen within the context where an estimate of 8 percent of CO2 emissions are attributed to freight transportation and a 3 percent to warehousing and goods handling (McKinnon, 2010). As a result, logistics, being an integral part of the supply chain, needs to be given its due concern for its environmental impact to promote sustainability (Aronsson & Brodin, 2006; Wolf & Seuring, 2010).

Seong-Tae and Sang-Yoon (2012) used the term environmental logistics practice (ELP) to define all the environmental activities related to logistics and supply chain management by manufacturers as well by logistics service providers, retailers, and wholesalers, which focus on minimizing negative environmental impacts while maximizing logistics efficiency in handling and distributing cargoes. They further defined ELPs as three interconnecting components, namely internal environmental management (IEM), environmental sourcing and packaging (ESP), and environmental process design (EPD). As a result Seong-Tae and Sang-Yoon (2012) inferred that the stakeholder pressure directly influence the adoption of ELP to build an eco-oriented culture across the SC so as to respond to all environmental challenges effectively and promote firm survival. Such a contextual scenario, demands that all manufacturing and logistics processes need to implement environmental-friendly based practices to enable an eco-oriented culture system wide across the SC to achieve an environmental based competitive capability (Ibid).

A4.1.14 Planning and forecasting

Strategic planning is about the focus, alignment and synchronization of all SC processes across all functional areas, such as manufacturing, marketing, technology and finance (Bayraktar et al., 2009). Such a strategic planning approach is based on a long term perspective and requires the set-up of all the necessary business and operational policies to promote performance improvement and a responsive SC (Gunasekaran et al., 2008). There is also the need to consider both economical and environmental issues within the planning process to promote a sustainable SC (Bojarski et al., 2009). The purchasing and supply management also forms a pivotal part of the strategic planning process (Cavinato, 1999) where an ongoing planning approach is deployed to identify inefficiencies (Neely et al., 1997).

The planning process needs the right information about both the demand and the supply sides of the SC, so that the focal firm will be in a position to establish or forecast all the information associated with the customers (e.g. a customer orders, point-of sales data, forecasts, planned orders and available customer stock) and supplies (e.g. availability of supplier stock, lead times, deliveries expected to be delayed, and advanced shipment notices respectively) (Jonsson & Mattsson, 2013). Such SC planning information, with its sales and operations planning, is enabled by the deployment of an integrated IT application, such as an advanced planning and scheduling (APS) tool (Jonsson et al., 2007; Marcotte et al., 2009) to promote SC visibility and synchronisation of all processes (Dreyer et al., 2009).

The Collaborative Planning Forecast and Replenishment (CPFR) is an established business process model used to enhance the customer visibility and to match the supply with the demand, with the necessary flow of all required raw materials, production goods and customers' deliveries (Thome et al., 2014). The CPFR has been defined by Thome et al. (2014, p.949) as 'a cohesive bundle of business processes whereby supply chain (SC) trading partners share information, synchronized forecasts, risks, costs and benefits with the intent of improving overall SC performance through joint planning and decision making'. The CPFR is one of the SC collaborative models, such as VMI, EDI and CR (Ajmera & Cook, 2009), to promote information sharing across the SC, since the collaborative effort needs to be built on a common culture of collaboration based on trust (Chen et al., 2007;

Fawcett et al., 2008b) and not only based on front-end agreements on all plans and forecasts (Thome et al., 2014).

A4.1.15 Managing quality

Managing quality within the SC context is crucial so as to exploit both all internal focal firm management practices together with all the external SC actors' practices and capabilities to derive a system wide quality plan (Robinson & Malhotra, 2005; Kaynak & Hartley, 2008; Sroufe & Curkovic, 2008; Lin et al., 2013). Furthermore, Kaynak and Hartley's (2008) model also outlined that quality management, within the SC context, is pivotal in the integrated system approach of all upstream and downstream process with the focal firm.

Managing quality has been defined by Lin et al. (2013, p.348) as: 'Supply chain quality management (SCQM) is concerned with designing, linking, and managing strategic and tactical activities of a supply chain system to prevent unwanted deviations, ensure accountability of channel partners, create system-level optimization, and build competent supply networks'. Quality is associated with SCI as declared by Robinson and Malhotra (2005, p. 319) who define SCQM as 'the formal coordination and integration of business processes involving all partner organizations in the supply chain channel'. Quality is pivotal in the product and service delivery (Bowersox et al., 2000; Bayraktar et al., 2009; Trkman & McCormack, 2009). Quality is one of the operational SC performance measures for competitive capabilities (Hayes & Wheelwright, 1979b; Schmenner & Swink, 1998; Ho et al., 2002; Swink et al., 2007).

QM (quality management) is considered synonymous to total quality management (TQM) (e.g. Singh & Smith, 2004; Kim et al., 2012). Furthermore, similarities of roadmaps and core values in quality management (i.e. TQM), is also compared to lean production and six sigma quality (Dahlgaard & Dahlgaard-Park, 2006).

Quality is one of the six world class manufacturing practices, as outlined by Hayes and Wheelwright (1984) and also as one of for manufacturing strategy infrastructure decision areas referred by Flynn et al. (1999). Quality has also been considered as part and parcel of all manufacturing and SC processes to perform effectively and efficiently as declared by Kim et al. (2010), where they defined that there exists three models to achieve corporate quality and business excellence: the European Foundation for Quality Management (EFQM) Excellence model, ISO 9000, and the Malcolm Baldrige National Quality Award (MBNQA).

With reference to Kim et al. (2010) the ISO 9000:2000 standard, it is also considered a prevalent quality management system approach which several organisations have obtained its registration and implemented it, both within the organisation and across its SC to promote significant benefits also in quality assurance (Sroufe & Curkovic, 2008). The ISO 9000:2000 standard, with its quality assurance characteristics, promotes better understanding of processes and achieves both lower costs and improved performance (Ibid). ISO implementation is needed not only to have the registration or the established procedures in place within the firm, but to serve as a measure for operations improvements, proactive commitment to implement quality standards integration across the SC and as a catalyst for change (Sroufe & Curkovic, 2008; Prajogo et al., 2012). To achieve such an innovative culture, the focal firm needs to go beyond the 'basic' implementation quality standard and shift to the 'advanced' and 'supportive' implementation stages, by taking new initiatives to operations and business with the full support of top management levels (Prajogo et al., 2012; Huo et al., 2014) so as to lead to a competitive advantage. From the extant literature ISO based established quality procedures are the core of all operations within manufacturing but they also need to integrate with them green initiatives across the SC with all SC members, to promote quality and meet the goals of environmental sustainability (e.g. ISO14001) in all processes (Sarkis, 2003; Feretti et al., 2007; De Brito & Van Der Laan, 2008; Caniels et al., 2013; Blome et al., 2014).

From the above, quality is also considered as a multidimensional concept, which refers to other perspectives such as supplier, management and information quality respectively. The dimension of quality data and reporting are considered an important asset in information quality, which is based on the accuracy, adequacy and timeliness of information, to promote a quality management approach (Li et al., 2005; Kaynak & Hartley 2008). The dimension of supplier quality management has also been substantiated by (Nyaga et al., 2013) by declaring that the quality of the relationships determines the willingness and type of collaborative initiatives. The dimension of management/leadership quality has also been substantiated by Robinson and Malhotra (2005), by declaring that both internal and external quality leadership approaches are necessary to promote sustainable competitive advantages.

Quality has also been considered as a paradox by the expression 'better quality-low cost', which has decreased its importance with the advent of sustainable SCs, where the competitive capabilities focus shifted on the social and environmental issues and challenges, than on quality and cost (Buyukozkan & Berkol, 2011, p.13731). Though, quality problems associated with manufacturing remains attributed to rework, scrap, machine and interrupted workflows (Clark, 1996; Flynn et al., 1999; Schemnner & Swink, 1998). Though, it is important to view quality from a holistic SC responsibility perspective, since every SC member, such as suppliers and all the processes within the focal firm with all other stakeholders, need to be accountable for any failure, even if the source of the problem originates from a quality issue in the suppliers' components quality (Trent & Monczka, 1999).

A4.1.16 Managing risks

Risks within the context of the SC has been defined by Juttner et al. (2003, p. 7) as: 'Supply chain risks comprise any risks for the information, material and product flows from original supplier to the delivery of the final product for the end user'. Risk management is considered as pivotal in SCI initiatives (Kahn & Mentzer, 1998). In fact Lambert and Cooper (2000) considered the risk and reward structure as one of the management component needed to promote SCI, since it affects the long-term commitment of all the SC actors within the SC.

There exists a SC risk associated with all the interactions of the focal firm with all other SC actors in the procurement of all needed resources to promote business continuity (Kahkonen et al., 2015). The SC operations are also subjected to a level of risk due to the nature of their complexities, if they are not managed and addressed effectively, such as in production and in view of the demand uncertainties (Craighead et al., 2007). In fact there is the need that businesses must exploit cross firms activities to improve in all operations and to 'share risks and rewards' to secure higher performance than would be achieved by operating alone (Lambert et al., 1999, p.166). Within the manufacturing SC the key attributed risks are associated with machines which perform below the scheduled performance and as a result may cause low productivity, planning disruptions and escalating production costs (Chong et al., 2012).

An integrated IT network enhances transparency within the firm and across the SC, so that the manufacturer can promptly detect variations in its operations as well as in suppliers' changes in production and delivery (Jin et al., 2014) and at the same time decrease all transaction risks (Clemons et al., 1993). Though, it cannot be excluded that firms takes a prudent and pragmatic approach on the benefits and risks of sharing information, and as a result a selective approach is taken on the information content shared through such integrated initiatives.

Due to the current global access to various stakeholders, the SC includes many actors with complex and different cultures and various political problems and characteristics, which may sometimes serve both as an advantage but at the same time may create great uncertainties and risks (Costantino & Pellegrino, 2010). For the focal firm to identify the appropriate strategy, needs to mitigate such risks and cope with such uncertainties, to promote success (Ibid). Such a mitigation of risks needs joint agreements with all SC actors to improve on the SC visibility and create a level of understanding to share all information on all associated risks sources, so as to prepare the necessary joint business continuity plans (Juttner et al., 2003).

SC risks have been classified in different ways, and the most basic classification adopted hereby is based on two scenarios, whether the risk is generated internally or externally (Klimov & Merkuryev, 2008). The SC risk is also considered as a function of external environment of all SC actors which is said to be situated in either a stable or dynamic environment (Hou et al., 2014). They further added that within such a complex dynamic environment, the overall SC network financial status not only depends on the focal firm's management skills but also from all SC actors' operational decisions.

A4.1.17 Managing trust

Nooteboom (2000, p. 53) defines that 'organizational trust is a constellation of behavioural trust, with organizational structure and culture acting as institutions that limit and guide behaviour of staff'. The effectiveness of all the SC actors engaged in a partnership relationship, within a SC, depends on their mutual trust, so as to meet the end customers' requirements (Vlachos & Bourlakis, 2006). Furthermore, trust with both upstream and downstream SC actors promote SCI initiatives for performance improvement (Zhang & Huo, 2013).

Trust in-conjunction with commitment among the SC actors, promote efficiency, effectiveness and productivity and also reduce conflicts and uncertainties across the SC (Dechurch et al., 2007; Cousins et al., 2008), otherwise there is the need to apply a coercive power approach to win cooperation from each other (French & Raven, 1968; Morgan & Hunt, 1994; Yeung et al., 2009).

Although trust-based SC collaboration is regarded as a risky approach (Carson et al., 2006) it is the way forward for leveraging resources from each SC actor for achieving a competitive edge (McEvily et al., 2003; Bijlsma-Frankema & Costa, 2005). The outstanding effect of trust can also be seen on the SC responsiveness improvement, since it takes place even when suppliers are more powerful, due to the power asymmetry between the dyad-relationship (Handfield & Bechtel, 2002).

Trust is based on the overall values of the human relationships and on the partner's reliability and integrity, which both promote a collaborative approach and a long term relationship (Morgan & Hunt, 1994; Berasategi et al., 2011). Trust emerges from the commitment, time, effort and dedication dimensions of all engaged SC actors (Msanjila & Afsarmanesh, 2011). As a result trust is also based on past experience that establishes a common understanding and builds up on common knowledge for mutual commitment (Ring & Van deVen, 1994). Though, an established axiom, which is function of trust, states 'trust but verify' (Tracey et al., 2013, p.103), meaning that verification may be needed to promote assurance, since blind faith may leads to associated risks.

Trust is a key determinant for suppliers to engage in joint planning activities and flexible arrangements taken with the focal firm as the buyer (Johnston et al., 2004). Furthermore, trust lowers the transaction costs (Sako, 2006), improves the average cycle time and in-time order fulfilment rate (Lin et al., 2005), promotes innovation (Panayides & Lun, 2009) and serves as a key foundation for stimulating effective relationship learning (Bryan et al., 2010). Trust is also needed for SC members to engage in a commitment to meet each others' requirements, such as in GSCM initiatives (Caniels et al., 2013) and also to promote enhanced environmental innovation (Pagell & Wu, 2009; Alvarez et al., 2010).

Firm	Sector	Type of Firm	Head-	Product	Revenue	Participant role
			count			
1/INT1 &	Medical devices	Part of a	104	Consumable	17 million	Participant 1:
1/INT2)		Global		medical devices	US	Director with 15
		Company		at low cost	Dollars	years experience
		(Headquarters				(Degree and
		USA) Rank				MBA)
		17 in the				Participant 2:
		Fortune 500				Operations
		classification				Manager with 15
						years experience
						(Technical
						Certificate and
						MBA)
2/INT3	Electronics	Part of a	150	High end	20 million	Director with 18
	(CCTV and	Global		customers	Sterling	year experience
	security	Company		relatively high		(Degree in
	systems)	(Headquarters		cost		Engineering)
		UK)				
3/INT4	Automation	Part of a	260	High to medium	18 million	IT and Logistics
	products	Global		end customers	Euros	Manager with 11
		Company		relatively high		years experience
		(Headquarters		to medium cost		(BSC Business
		Italy)				and Computing
						degree, BA
						(Honours) and
						MBA.
4/INT5	Software	Local Firm &	5	Corporate and	n/a	Director and CEO
	development	Contractor		individual		with 19 years of
	and systems			training and		experience.
	integration			support		
5/INT6	Heavy	Local Firm	180	Air-	14 million	Senior
	Industrial	(Originated		Conditioning	Euros	Management
	Equipment (AC	from		Industrial high		Team (20 years
	Units)	Germany and		value equipment		experience in SC
		shifted to		manufacturing		and Logistics)
		Malta)				

Appendix 5: A breakdown of all the participants under the both open sampling and theoretical sampling phases

6/INT7 &	Medical	Part of a	400	Low cost	n/a	Participant 1:
6/INT8	Devices	Global		medical		Senior Manager
		Company		products		with 25 years
		(Headquarters		including high		experience
		USA).		value units/sets		specialised in
		Local firm				logistics and
		uses high				warehousing
		value				(Diploma)
		manufacturing				Participant 2:
						Operations & HR
						Manager with 18
						years experience
						(Mechanical
						Engineer)
7TS/INT1	Toys and toys-	Part of a	800	Low cost toys	n/a	Purchasing
	sets	Global		as products		Manager with post
		Company		including		secondary level of
		(Headquarters		medium value		education with 20
		Germany).		units/sets		years of work
		Local firm				experience in
		uses high				operations
		value				management and
		manufacturing				purchasing
5TS/INT2	Heavy	Local Firm	180	Air-	14 million	Operations
(Same Firm	Industrial	(Originated		Conditioning	Euros	Manager and
5)	Equipment (AC	from		Industrial high		Engineer by
	Units)	Germany and		value equipment		Profession with 17
		shifted to		manufacturing		years of
		Malta)				experience
2TS/INT3	(CCTV and	Part of a	150	High to medium	20 million	Director with 18
(Same Firm	security	Global		end customers	Sterling	year experience
2)	systems)	Company		relatively high		(Degree in
		(Headquarters		to medium cost		Engineering)
		UK)				
8TS/INT4	Automotive	Private	120	Innovative	n/a	25 years Industry
	products	Foundation		products		Experience with
		based locally		medium to high		Degree in
		with founders		end		Electrical Power
		in Germany				and Professor of
		(no				the Electrical
		shareholding)				Engineering
						Department at
						University of

						Malta				
1TS/INT5	Medical	Part of a	104	Consumable	17 million	Director with 15				
(Same Firm	Devices	Global		medical devices	US	years experience				
1)		Company		at low cost	Dollars	(Degree and				
		(Headquarters				MBA)				
		USA) Rank								
		17 in the								
		Fortune 500								
		classification								
9TS/INT6	High Voltage	Part of a	80	Industrial high	n/a	30 years				
	Switchgear for	Global		value equipment		experience Senior				
	heavy electrical	Company		manufacturing		Engineer in				
	applications	(HQ in Libya)				switchgear for				
						electrical sub-				
						stations				
10TS/INT7	Printing on	Private locally	100	Low cost to	12 million	35 years				
	paper products	owned		medium value	Euro	experience in the				
				products (e.g.		printing industry				
				books and toys		Senior System				
				& food		Manager				
				packaging						
				boxes)						
11TS/INT8	Food	Private	400	Low to medium	70 million	Senior Manager				
	Manufacturing	(Originally		cost products at	Euros	having an MBA				
	(Dry and wet	partnership		high volume		and with 12 years				
	foods)	with Unilever		(e.g. dry power		experience in				
		UK (1967-		and liquid		production and				
		1990))		drinks and		quality assurance.				
				sauces)		Supported by				
						another Senior				
						Manager having				
						an MBA and with				
						15 years				
						experience in				
						supply chain and				
						logistics.				
12TS/INT9	Logistics	Private locally	160	Freight services	10 million	Senior Manager				
		owned			Euros	with 30 years				
						experience in				
						freight forwarding				
13TS/INT10	Pharmaceutical	Part of a	16	Medicinal	10 million	General Manager				
	sector	Global		products	Euros	with 20 years				
		Company		(Generic drugs)		experience				

		(HQ in India)				
14TS/INT11	Pharmaceutical	Part of a	300	Medicinal	5 million	Head of IT
	sector	Global		products	Euros	Infrastructure with
		Company		(Generic drugs)		13 years
		(HQ in USA)				experience
15TS/INT12	Logistics	Local and	18	Logistics	2 million	General Manager
		Private		Service provider	Euros	with 40 years
		ownership				experience with a
						Graduate level of
						education
16TS/INT13	Automotive	Part of a	600	'O' Rings or	20 million	Purchasing,
	parts	Global		sealing	Euros	logistics and IT
		Company		solutions		support with 14
		(HQ in				years experience
		Sweden)				
17TS/INT14	Systems	Consultancy	8	ICT services in	8 million	Senior IT
	Integration	in IT based		hardware and	Euros	Consultant with 13
	consultancy for	projects		software		year experience in
	manufacturing					manufacturing and
						IT.

Note: **1/INT1** refers to firm 1 and interview 1 (i.e. INT1 means interview 1) under the open sampling phase and similarly **17TS/INT** 14 refers to firm 17 and interview 14 (i.e. INT14) under the theoretical sampling (**TS**) phase.

Appendix 6: Interview guide for theoretical sampling phase

(1) Kindly indicate your position and experience; the company manufacturing industry sector; number of employees; product/s and/or services provided; and approximate revenue (turnover).

(2) Can you tell me about the management practices and key drivers adopted that promote or inhibit, if any, to determine the company supply chain integration (SCI) strategy so as to promote effective SC integration within and outside the firm? It may help to answer this question by referring to the checklist below, unless you feel to add any other issue/s.

- i. The link between the corporate strategy and the SC integration strategy;
- ii. Any decisions taken on the SC set-up (e.g. number of SC tiers or entities);
- iii. Which are the internal and external **SCI measures** or **initiatives** to get a holistic SC approach;
- iv. About the use of other **SC members resources**;
- v. The use of the **performance management** approach on the procedures and standards both internally and externally;
- vi. The leadership style used in change management and daily workflow;
- vii. The **assigned priorities**, if any, to meet various SC targets;
- viii. What is the **level of IT adoption** to enable and/or transform all business processes; &
- ix. Which are the initiatives taken on **information sharing.**

(3) How do you describe, through some examples or events, the overall SCI process foundations (within and outside the firm) in terms of the checklist below. It may help also to consider extreme events and even add any other ideas that you feel they have been omitted.

- i. Teamwork/collaboration efforts for relationship building;
- ii. Trust/relationship management;
- iii. Human/social elements (e.g. soft and hard skills);
- iv. Technology deployment (e.g. in machinery and ICT);
- v. Information sharing (level (e.g. visibility level of SC tiers) and type of information (e.g. operational; technical; financial and strategic));
- vi. Knowledge management and knowledge sharing across different cultures;
- vii. Lean management (vis a vis quality and corporate social responsibility (CSR));
- viii. Planning (forecasting);
- ix. Supply management and inventory management;
- x. Logistics management;
- xi. Performance measurements;
- xii. Risk management;
- xiii. Customer service; and
- xiv. Innovation/change management

(4) Which are the measures that you use to identify the success or failure of the overall SC performance.

(5) Given that the Maltese context, as a small island state, is distinct from other global SCs, how do you consider such similarities and differences.

Appendix 7: Interview guide for theoretical sampling for researcher with Probe List

(1) Kindly indicate your position and experience; the company manufacturing industry sector; number of employees; product/s and/or services provided; and approximate revenue (turnover).

(2) Can you tell me about the management practices and key drivers adopted, if any, to determine the company supply chain integration (SCI) strategy so as to promote effective SC integration within and outside the firm? You may answer this question by referring to the checklist below, unless you feel to add any other issue.

Probes List: (Emerged theory)

- i. The link between the corporate strategy and the SCI integration strategy (<u>Emerged theory:</u> there is alignment between **the corporate strategy** to **SCI strategy**);
- ii. **Any decisions** taken on the SC set-up (<u>Emerged theory</u>: decisions are taken on the length of the SC to be adopted (i.e. what is the number of SC members (suppliers, 3rd parties etc) in the pipeline and whether to **integrate** with them or undergo vertical integration, to bypass them or work alone (e.g. suppliers, distributors, outsourced entities or sub-contractors));
- iii. Which are the internal and external **SCI measures** or **initiatives** to get a holistic SC approach (<u>Emerged theory</u>: cross-functional approach, training, right leadership, no silos, audit, lean management, CSR, close customer contacts and others);
- iv. About the use of other **SC members resources**. (Emerged theory: Action is taken to **exploit SC members** resources to promote an **innovative approach** to achieve a competitive SC (e.g. shared leaning initiatives, knowledge transfer, outsourcing, trust building, communication for information sharing and others);
- v. The use of the **performance management** approach on the procedures and standards both internally and externally. (<u>Emerged theory</u>: A **performance management** approach (internal and external to the SC) is used which includes all procedures and standards (e.g. ISO for quality of processes; manufacturing (e.g. ISA 95) and social /environmental) to meet the company KPIs;
- vi. The **leadership style** used in change management and daily workflow. (<u>Emerged theory</u>: the type of **Leadership style** used is to promote SCI (e.g. transformational leadership for innovation/change and transactional leadership for daily workflow or processes efficiency improvements or other);
- vii. The **assigned priorities**, if any, to meet various SC targets. (<u>Emerged theory</u>: **priorities** are assigned to the allocation of resources, to meet all SC targets effectively such as speed, quality, flexibility and cost; and
- viii. What is the **level of IT adoption** to enable and/or transform all business processes. (<u>Emerged theory:</u> use of ICT tools (e.g. ERP, databases etc) and automation technologies in manufacturing (e.g. CAD, CAM, smart and advanced manufacturing systems etc)); and
- ix. Which are the initiatives taken **information sharing** (<u>Emerged theory</u>: information sharing enabled by ICT applications range from SC visibility and real-time updates to joint cross-cultural knowledge sharing).

(3) How do you describe, through some examples, the overall SCI process foundations (within and outside the firm) in terms of the checklist below. You may also consider extreme events and even add any other concept that you feel it has been omitted.

Probes: (Emerged theory)

- i. **Teamwork/collaboration**: (decentralised or centralised structure, cross-functional approach level of implementation (within all management levels, across functional areas and across the SC members));
- ii. **Trust/relationship management: (cultural values** such as shared values, commitment, effort, expectations and behavioural aspects);
- Human/social elements: employee skills (motivations, proactive/passive actions and aspirations for development); training (in-house and shared with SC members); (soft skills: values, empowerment, involvement and behaviour; and hard skills: compliance to established procedures and reporting in-line with ISO standards);
- iv. **Technology deployment:** (technology is used both within the manufacturing process (such as robotics) and as an ICT tool/s within/across all SC members up to the customer for real time communication and to align resources (process re-engineering) so as to enhance and/or transform all SC operations through information management and automation (such as risk, supply, inventory and logistics management respectively and performance measurement etc) (e.g. of ICT tools: EDI, ERP (SC software, business intelligence applications and dash-board reporting, databases etc), internet information exchange (e-mails, XML data exchange, social media etc) legacy systems and e.g. of automation tools: CAD, CAM, advanced manufacturing systems and embedded systems etc);
- v. **Information sharing:** (information is shared through information management; communication, SC visibility level used within and across (1st tier or all Sc tiers) to share with respect to level and type i.e. level (e.g. visibility level of SC tiers) and type of information (e.g. operational (delivery details, stock levels etc); technical (knowledge); financial and strategic)) or (operational; tactical and strategic);
- vi. **Knowledge management:** (knowledge is needed to understand the cross-cultural SC members capabilities and sharing of knowledge occurs both explicit and/or implicit);
- vii. **Lean management:** (standards and procedures are used to promote efficiencies such as in waste management, zero defects, maximise the use of resources for continuous improvement (six sigma) etc) and meet all CSR measures in-line with the firm KPIs);
- viii. **Planning (forecasting):** (manage the right stock level to avoid stock-out or excess stock in-line with cashflow management);
- ix. **Supply management and inventory management:** (supply management considers the lead time and stock guarantees to be respected by suppliers, VMI and/or consignment stock systems, multiple suppliers etc) and inventory management considers cashflow management, stock levels policies (minimum, buffer, EOQ or other);
- x. **Logistics management:** (e.g. transportations and CSR measures to promote sustainability);
- xi. **Performance measurements:** (audit tool and as a driver for action which includes performance measurement and audits with respect to social (e.g. voice of employee,

customer services etc); environmental (e.g. waste management, emissions, carbon footprint, etc); manufacturing (e.g. standards such as ISA 95) and SC (lead-time, supply deliveries, logistics of finished products etc);

- xii. **Risk management:** (vertical integration to reduce risks and improves supply management; delivery time/lead-time; process failure due to faults (business continuity); performance measures used to identify problems, supplier management against failure; investment in high-added manufacturing to improve on the machinery reliability etc); and
- xiii. **Customer service** (pull or push SC, close to the customer through web site, phone calls and other connections, relationship/s, customer classification (allocated priorities due to customers' profiles or segmentation) and follow-ups)

xiv. Innovation and Change management

(4) Which are the measures that you use to identify the success or failure of the overall SC performance.

Probes: (Emerged theory)

- i. Operational and business indicators (SC measures interlinked to corporate measures),
- ii. Compromise/trade-off between targets and optimise capabilities (resources) to achieve a win-win SC,
- iii. Low cost or differentiation strategy,
- iv. Lean management and quality processes (e.g. standard processes (ISO), TQM, continuous improvement, low cost, flexibility (agility), speed, etc),
- v. Customer service,
- vi. Employee motivation (through involvement, empathy and respect),
- vii. Flexibility to change/responsiveness,
- viii. Innovative and sustainable approach and also the competitive edge over other SCs.

(5) How do you consider the similarities and differences of the Maltese context, as a small island state, which makes the SC distinct from other global SCs.

Probes: (Emerged theory)

Local Environment: transportation of physical products (logistics) costs (maritime and shipping services, with equipped ports); Government/EU support to SMEs (e.g. investment packages support; incentives to investment; grants; funds; capital guarantee support for all investments); Political, Legal and regulatory frameworks promote stability in the Banking sector and Financial institutions to aid further investment opportunities; Highly skilled workers availability and with a substantial variation of academic skills; Cultural values, motivations and expectations of staff; Corporate social responsibility awareness.

Global economy: Selection of SC member is function of particular competences (e.g. cost per unit of production; vicinity to the customer/market (market proximity); cost and quality of raw material; cheap labour; recession affected countries.

t8	Code System	💥 📇 🔁 🖢 😢	V 🗗	x >		
t:	A		10 1	t		
	Code System			1472		
	- Actions and reactions			25	[A sting Category
				- 15		Action Category
	change inchagement			5		
	Collaboration			33		
	E		i i i	12		
	employees engagement		- Ē	13		
				7		
	Cross-functional operations			13		
	······································			5		
	Communication_being informed			31		
				1		Code Memo
	contact person			2		
	contractual agreements			10		
	coordination			11		
	Polotionski socionalk			2		
	Relationship management			9		
	Cost management			4		
				40		
				6		
	direct contact with the customer			2		
	customisation and client needs			1		
	Customer support			6		
	⊡		i i i i i i i i i i i i i i i i i i i	10		
	problem solving		Ē	10		
	delivery actions			14		
+O	Code System	** E Fa % On	w 2	- 1		
•••	code system			• •		
		440		16		
	material reduction	- (4)		4		
	recycle Waste management			4		
	flexibility			18		
	ICT applications			81		
	improvement in processes			9		
	information sharing			19		
	visibility		i i i	7		
	Innovation New products is the norr	n	ī	18		
	R &D			6		
				44		
	🖃 🕸 🖀 leadership style			12		
				3		
	management effectiveness			15		
				26		
	E Sector Construction			17		
	transportation management			20		
	manufacturing process			60		
	humans' intervention.		_	6		
	work in progress					
	added value manufacture			1		
	where the second			21		
				30		
	OP plane and forecasts			21		
	Comparison of orders work			12		
				23		
	······································			28		
	SC members support			32		
	SC strategic action & growth			23		
1.1						

Appendix 8: Laundry list of all Open codes from MAXQDA 2007 code segment (Author)



to	Code System 💥 📇	•	و ک	21 🚺	4	a :
t:						#
						1
						1
	Customer type					2
	🖃 🗠 🚰 experienced staff					7
	workforce attributes					7
						0
	Change oriented culture instilled					0
	Competitive cost					1
	Skills					0
						15
	🕬 🍘 International scope					6
	Interviewee role_director of manufacturer ar	nd a				7
	Interviewee_experience					14
	🕬 🍘 Language barrier					2
	👓 🖓 Local firm type					30
						7
	👓 🖓 local market					2
	👓 🖓 Natural resources (Small Island State)					2
	🖃 🗠 🚰 organisation structure					4
	🛶 🕞 functional areas available					3
						2
						27
						8
	👓 🖓 revenue of firm					5
	👓 🖓 SC Logistics as a problem					8
	SC member support				?	1
	👓 🖓 Size of firm					9
	Strategic position to Europe					2
	Proximity of transportation.					2
i	🚔 Sets					0

Appendix 9: CAQDAS Outcome of all Emerged Categories for all the Eight Interviews

Interviews

										Codes
All Emerged Categories	1	2	3	4	5	6	7	8	Total Sum	Count
Actions and reactions	0	0	0	0	0	0	0	0	0	0
Actions and reactions\Auditing & traceability of operations	5	4	0	3	2	4	3	4	25	1
Actions and reactions\Change management	2	7	1	1	1	2	0	1	15	1
Actions and reactions\Change management\resistance to change	0	3	0	0	2	0	0	0	5	1
Actions and reactions\Collaboration	4	2	10	8	2	1	4	2	33	1
Actions and reactions\Collaboration\Teamwork	4	2	0	2	0	0	0	4	12	1
Actions and reactions\Collaboration\Teamwork\employees engagement	0	8	1	1	0	0	1	2	13	1
Actions and reactions\Collaboration\Teamwork\Duplication of effort	3	0	0	2	2	0	0	0	7	1
Actions and reactions\Collaboration\Teamwork\cross-functional operations	2	2	0	1	0	0	5	3	13	1
Actions and reactions\Collaboration\commitment	1	3	0	0	0	0	1	0	5	1
Actions and reactions\Collaboration\communication_being informed	2	4	7	3	2	1	7	5	31	1
Actions and reactions\Collaboration\concerns	0	0	0	0	0	0	1	0	1	1
Actions and reactions\Collaboration\contact person	0	0	0	0	0	0	2	0	2	1
Actions and reactions\Collaboration\contractual agreements	1	2	2	2	1	0	1	1	10	1
Actions and reactions\Collaboration\coordination	2	2	1	3	0	0	1	2	11	1
Actions and reactions\Collaboration\prompt feedback	0	0	1	1	0	0	0	0	2	1
Actions and reactions\Collaboration\Relationship management	0	2	2	4	0	0	0	1	9	1
Actions and reactions\cost management	1	3	0	0	0	0	0	0	4	1
Actions and reactions\cost management\cash flow management	0	0	0	0	0	2	1	1	4	1

Actions and reactions\customer relationship management	0	3	7	7	5	6	7	5	40	1
Actions and reactions\customer relationship management\supported	0	3	3	0	0	0	0	0	6	1
Actions and reactions\customer relationship management\direct contact with the										
customer	2	0	0	0	0	0	0	0	2	1
Actions and reactions\customer relationship management\customisation and client										
needs	0	0	0	0	1	0	0	0	1	1
Actions and reactions\customer relationship management\customer support	0	4	1	0	0	1	0	0	6	1
Actions and reactions\decision-making	0	2	0	3	0	1	1	3	10	1
Actions and reactions\decision-making\problem solving	0	3	2	0	0	0	0	5	10	1
Actions and reactions\delivery actions	0	0	4	4	0	2	3	1	14	1
Actions and reactions\EHS (environmental, health and Safety)	2	12	0	0	0	1	1	0	16	1
Actions and reactions\EHS (environmental, health and Safety)\material reduction	1	2	0	0	1	0	0	0	4	1
Actions and reactions\EHS (environmental, health and Safety)\recycle/Waste										
management	0	0	0	0	3	1	0	0	4	1
Actions and reactions\flexibility	0	2	5	5	1	1	2	2	18	1
Actions and reactions\ICT applications	3	12	18	14	23	4	3	4	81	1
Actions and reactions\improvement in processes	1	2	0	0	0	1	3	2	9	1
Actions and reactions\information sharing	2	3	2	1	2	0	6	3	19	1
Actions and reactions\information sharing\visibility	0	1	1	0	2	0	1	2	7	1
Actions and reactions\Innovation_New products is the norm	5	3	4	0	0	1	4	1	18	1
Actions and reactions\Innovation_New products is the norm\R & D	0	0	1	3	0	1	1	0	6	1
Actions and reactions\Inventory management	3	4	6	6	6	5	9	5	44	1
Actions and reactions\leadership style	3	2	0	3	0	1	2	1	12	1

Actions and reactions\leadership style\open style policy	0	3	0	0	0	0	0	0	3	1
Actions and reactions\leadership style\management effectiveness	1	3	0	0	4	0	2	5	15	1
Actions and reactions\lean management	9	6	0	5	4	0	0	2	26	1
Actions and reactions\Logistics	0	1	2	5	1	2	6	0	17	1
Actions and reactions\Logistics\transportation management	0	6	1	2	2	1	4	4	20	1
Actions and reactions\manufacturing process	1	2	13	13	13	2	9	7	60	1
Actions and reactions\manufacturing process\humans' intervention.	0	0	0	1	4	1	0	0	6	1
Actions and reactions\manufacturing process\work in progress	0	0	0	0	0	1	1	5	7	1
Actions and reactions\manufacturing process\added value manufacture	1	0	0	0	0	0	0	0	1	1
Actions and reactions\ownership of operations	9	10	1	0	0	0	1	0	21	1
Actions and reactions\performance measurement	8	6	1	6	1	1	4	3	30	1
Actions and reactions\performance measurement\KPIs	2	0	1	0	0	0	2	1	6	1
Actions and reactions\plans and forecasts	0	2	5	3	1	3	1	6	21	1
Actions and reactions\Priority of orders/work	1	2	0	5	0	0	0	4	12	1
Actions and reactions\Quality of the processes	2	1	1	4	1	0	5	9	23	1
Actions and reactions\risk management	2	2	9	6	0	5	2	2	28	1
Actions and reactions\SC members support	6	19	2	1	0	0	2	2	32	1
Actions and reactions\SC strategic action & growth	8	2	4	2	0	2	2	3	23	1
Actions and reactions\SC_measure the velocity of the	2	0	0	0	0	1	0	1	4	1
Actions and reactions\SCI activities	3	6	4	1	4	8	4	6	36	1
Actions and reactions\SCI activities\Reverse logistics	0	0	0	3	0	0	0	1	4	1
Actions and reactions\SCI depth	10	4	5	8	7	4	5	4	47	1
Actions and reactions\Service centres	0	0	1	0	0	0	0	0	1	1

Actions and reactions\standards implementation	4	1	2	8	1	4	3	5	28	1
Actions and reactions\standards implementation\Legislation	0	0	3	0	0	2	1	1	7	1
Actions and reactions\sub-contractors & outsourcing	1	0	0	8	1	0	2	2	14	1
Actions and reactions/suppliers (multiple suppliers use)	1	0	0	0	0	0	0	0	1	1
Actions and reactions\Supply management	9	4	19	27	6	7	5	2	79	1
Actions and reactions\Supply management\reduce cost & prices	4	0	0	0	0	0	1	0	5	1
Actions and reactions\Supply management\stock codes	0	0	0	0	1	0	0	3	4	1
Actions and reactions\Supply management\Suppliers use there own systems	0	0	1	0	0	0	0	0	1	1
Actions and reactions\Supply management\validation of materials	3	0	0	0	0	0	0	0	3	1
Actions and reactions\Supply management\Lead time	4	0	4	3	0	3	1	0	15	1
Actions and reactions\time-management	0	1	2	3	4	0	1	0	11	1
Actions and reactions\time-management\Dynamic action	0	0	0	0	0	0	2	2	4	1
Actions and reactions\training share innovative ideas	4	4	2	1	1	0	2	1	15	1
Actions and reactions\training share innovative ideas\knowledge sharing	2	0	1	0	0	0	1	0	4	1
Actions and reactions\trust (intra and inter)	1	5	0	2	1	1	3	2	15	1
Actions and reactions\employees engagement	0	0	0	0	0	0	0	0	0	1
Actions and reactions\Duplication of effort	0	0	0	0	0	0	0	0	0	1
Actions and reactions\cross-functional operations	0	0	0	0	0	0	0	0	0	1
Consequences and outcomes	0	0	0	0	0	0	0	0	0	0
Consequences and outcomes\Country responsibilities	1	0	0	0	0	0	0	0	1	1
Consequences and outcomes\Country responsibilities\Labour employment and										
training	1	0	0	0	0	0	0	0	1	1
Consequences and outcomes\Country responsibilities\Add GDP	1	0	0	0	0	0	0	0	1	1

Consequences and outcomes\competitiveness	0	2	0	0	0	2	0	1	5	1
Consequences and outcomes\customer service level	1	0	0	0	0	0	0	2	3	1
Consequences and outcomes\customer service level\Customer visibility	0	4	1	0	0	0	0	0	5	1
Consequences and outcomes\customer service level\direct customer contact	1	0	0	0	0	0	0	0	1	1
Consequences and outcomes\customer service level\customer complaints	1	2	0	0	0	0	0	0	3	1
Consequences and outcomes\delivery performance	0	5	0	3	0	1	1	1	11	1
Consequences and outcomes\delivery performance\proximity to the market	0	3	0	0	0	0	0	0	3	1
Consequences and outcomes\delivery performance\SC bottlenecks	5	6	1	0	0	0	0	0	12	1
Consequences and outcomes\delivery performance\synergy and collaboration	2	0	1	0	0	0	0	0	3	1
Consequences and outcomes\delivery performance\cost reduction measures	2	0	0	1	0	0	0	0	3	1
Consequences and outcomes\delivery performance\service level agreements	1	2	0	0	0	0	0	1	4	1
Consequences and outcomes\delivery performance\sales volume	1	0	0	1	0	0	0	0	2	1
Consequences and outcomes\delivery performance\profit margin & Cashflow	0	1	0	0	0	1	0	0	2	1
Consequences and outcomes\delivery performance\Awards and Excellence										
achievement	2	6	0	0	0	0	0	2	10	1
Consequences and outcomes\delivery performance\Effectiveness	2	1	0	0	0	0	0	1	4	1
Consequences and outcomes\delivery performance\Efficiency	2	4	0	0	0	0	0	3	9	1
Consequences and outcomes\Market share_opportunity for growth	2	5	0	0	1	0	1	0	9	1
Consequences and outcomes\Market share_opportunity for growth\growth through										
SCI	1	3	0	0	0	0	1	0	5	1
Context/Conditions	0	0	0	0	0	0	0	0	0	0
Context/Conditions\competition	1	3	0	0	0	0	1	0	5	1
Context/Conditions\Culture	5	4	1	1	3	1	1	1	17	1

Context/Conditions\customer profile	0	0	0	0	0	0	0	0	0	1
Context/Conditions\customer profile\customer specifications	0	0	1	0	0	0	0	0	1	1
Context/Conditions\customer profile\customer knowledge	0	1	0	0	0	0	0	0	1	1
Context/Conditions\customer profile\Customer type	0	0	2	0	0	0	0	0	2	1
Context/Conditions\experienced staff	0	2	0	3	1	0	0	1	7	1
Context/Conditions\experienced staff\workforce attributes	0	1	0	0	4	0	2	0	7	1
Context/Conditions\experienced staff\workforce attributes\Labour mobility	0	0	0	0	0	0	0	0	0	1
Context/Conditions\experienced staff\workforce attributes\Change oriented culture										
instilled	0	0	0	0	0	0	0	0	0	1
Context/Conditions\experienced staff\workforce attributes\Competitive cost	1	0	0	0	0	0	0	0	1	1
Context/Conditions\experienced staff\workforce attributes\Skills	0	0	0	0	0	0	0	0	0	1
Context/Conditions\Government & EU support	3	2	1	1	1	5	1	1	15	1
Context/Conditions\International scope	1	3	0	1	0	0	1	0	6	1
Context/Conditions\Interviewee role_director of manufacturer and a	1	0	1	1	1	2	0	1	7	1
Context/Conditions\Interviewee_experience	2	2	3	3	1	1	1	1	14	1
Context/Conditions\Language barrier	0	0	0	0	1	0	0	1	2	1
Context/Conditions\Local firm type	11	2	4	6	1	2	1	3	30	1
Context/Conditions\Local industry support (subcontracting)	1	1	3	0	0	1	1	0	7	1
Context/Conditions\local market	0	0	1	1	0	0	0	0	2	1
Context/Conditions\Natural resources (Small Island State)	0	0	0	0	0	1	0	1	2	1
Context/Conditions\organisation structure	3	0	0	1	0	0	0	0	4	1
Context/Conditions\organisation structure\functional areas available	3	0	0	0	0	0	0	0	3	1
Context/Conditions\Politics	0	1	0	0	0	0	1	0	2	1

Context/Conditions\Product_Malta facility	7	2	11	3	0	1	2	1	27	1
Context/Conditions\recession	0	1	2	3	0	1	1	0	8	1
Context/Conditions\revenue of firm	1	1	1	1	0	1	0	0	5	1
Context/Conditions\SC Logistics as a problem	0	0	1	1	0	2	3	1	8	1
Context/Conditions\SC member support	0	1	0	0	0	0	0	0	1	1
Context/Conditions\Size of firm	1	1	1	1	2	1	1	1	9	1
Context/Conditions\strategic position to Europe	1	1	0	0	0	0	0	0	2	1
Context/Conditions\strategic position to Europe\proximity of transportation.	0	2	0	0	0	0	0	0	2	1
									1472	132

Frequency Count 132

	Interview numbers									
Filtered Emerged Categories (Based on frequency >									Total	
10)	1	2	3	4	5	6	7	8	Sum	Count
Actions and reactions\Auditing & traceability of										
operations	5	4	0	3	2	4	3	4	25	1
Actions and reactions\Change management	2	7	1	1	1	2	0	1	15	1
Actions and reactions\Collaboration	4	2	10	8	2	1	4	2	33	1
Actions and reactions\Collaboration\Teamwork	4	2	0	2	0	0	0	4	12	1
Actions and										
reactions\Collaboration\Teamwork\employees										
engagement	0	8	1	1	0	0	1	2	13	1
Actions and reactions\Collaboration\Teamwork\cross-										
functional operations	2	2	0	1	0	0	5	3	13	1
Actions and										
reactions\Collaboration\communication_being informed	2	4	7	3	2	1	7	5	31	1
Actions and reactions\Collaboration\coordination	2	2	1	3	0	0	1	2	11	1
Actions and reactions\customer relationship management	0	3	7	7	5	6	7	5	40	1
Actions and reactions\delivery actions	0	0	4	4	0	2	3	1	14	1
Actions and reactions\EHS (environmental, health and										
Safety)	2	12	0	0	0	1	1	0	16	1
Actions and reactions\flexibility	0	2	5	5	1	1	2	2	18	1
Actions and reactions\ICT applications	3	12	18	14	23	4	3	4	81	1

Appendix 10: CAQDAS Outcome of a Filtered Set (based on a frequency count > 10) of Emerged Categories for all the Eight Interviews

Actions and reactions\information sharing	2	3	2	1	2	0	6	3	19	1
Actions and reactions\Innovation_New products is the										
norm	5	3	4	0	0	1	4	1	18	1
Actions and reactions\Inventory management	3	4	6	6	6	5	9	5	44	1
Actions and reactions\leadership style	3	2	0	3	0	1	2	1	12	1
Actions and reactions\leadership style\management										
effectiveness	1	3	0	0	4	0	2	5	15	1
Actions and reactions\lean management	9	6	0	5	4	0	0	2	26	1
Actions and reactions\Logistics	0	1	2	5	1	2	6	0	17	1
Actions and reactions\Logistics\transportation										
management	0	6	1	2	2	1	4	4	20	1
Actions and reactions\manufacturing process	1	2	13	13	13	2	9	7	60	1
Actions and reactions\ownership of operations	9	10	1	0	0	0	1	0	21	1
Actions and reactions\performance measurement	8	6	1	6	1	1	4	3	30	1
Actions and reactions\plans and forecasts	0	2	5	3	1	3	1	6	21	1
Actions and reactions\Priority of orders/work	1	2	0	5	0	0	0	4	12	1
Actions and reactions\Quality of the processes	2	1	1	4	1	0	5	9	23	1
Actions and reactions\risk management	2	2	9	6	0	5	2	2	28	1
Actions and reactions\SC members support	6	19	2	1	0	0	2	2	32	1
Actions and reactions\SC strategic action & growth	8	2	4	2	0	2	2	3	23	1
Actions and reactions\SCI activities	3	6	4	1	4	8	4	6	36	1
Actions and reactions\SCI depth	10	4	5	8	7	4	5	4	47	1
Actions and reactions\standards implementation	4	1	2	8	1	4	3	5	28	1

Actions and reactions\sub-contractors & outsourcing	1	0	0	8	1	0	2	2	14	1
Actions and reactions\Supply management	9	4	19	27	6	7	5	2	79	1
Actions and reactions\Supply management\Lead time	4	0	4	3	0	3	1	0	15	1
Actions and reactions\time-management	0	1	2	3	4	0	1	0	11	1
Actions and reactions\training share innovative ideas	4	4	2	1	1	0	2	1	15	1
Actions and reactions\trust (intra and inter)	1	5	0	2	1	1	3	2	15	1
Consequences and outcomes\delivery performance	0	5	0	3	0	1	1	1	11	1
Consequences and outcomes\delivery performance\SC										
bottlenecks	5	6	1	0	0	0	0	0	12	1
Context/Conditions\Culture	5	4	1	1	3	1	1	1	17	1
Context/Conditions\Government & EU support	3	2	1	1	1	5	1	1	15	1
Context/Conditions\Interviewee_experience	2	2	3	3	1	1	1	1	14	1
Context/Conditions\Local firm type	11	2	4	6	1	2	1	3	30	1
Context/Conditions\Product_Malta facility	7	2	11	3	0	1	2	1	27	1
								ncy	1472	46
Note: Shaded parts show seven (7) consequences &										
Context codes. The remaining action codes amount to							key			
thirty-nine (39). A total of 46 codes.							codes		46	

Appendix 11: Axial Coding of the 15 non-key tentative categories

A11.1 Auditing Operations tentative category

Auditing operations theme is about a systematic evaluation of all SC operations, both within and outside the focal firm to meet all quality standards together with all sustainable and legal obligations, in an integrated approach, to promote effective and efficient performance.

To achieve an **integrated approach** within and across the SC, the focal firm needs to undergo internal and external **auditing of all operations**, to meet all **standards** in quality, **sustainable measures** together with **legal** compliance. The external auditing is carried either by the Corporate or by an established certified entity (e.g. BSI, UL), as indicated by four participants, by stating that:

'Corporate audits are more in depth and scrutinuous on quality'; '... audited and implemented such EHS programme'; 'Audited every year, by BSI ...' (1/INT1); 'The products need to have all details according to that specified, to be UL certified' (3/INT4); '... suppliers, who have to stick to our standards and processes and we visit them and make audits...' (6/INT7); and 'Clients we do credit checks on them, especially if they are new' (10TS/INT7).

The corporate audits on the focal firm are easier to perform due to the **ingrained common KPIs and culture** between each other, since with other SC members, the focal firm needs to enforce the auditing measures. One participant confirmed this, by stating that:

'For inter-company the performance is more easy to be dealt with due to same values, culture ...'; 'But with local suppliers and others external European suppliers... requires the validation and audit parameters/criteria as established with the corporate KPIs and its suppliers' (1/INT1).

The aim of the audits is to assess the focal firm and also its SC members on a regular basis and with a proactive approach, to ensure **matching of all operations** in a timely manner and to promote compliance to all of its **standard operating procedures** to achieve an **overall effective and efficient performance.** Two participants declared that the focal firm needs to be: 'Audited every year, by BSI'; and '... [audited] with respect to operations, management of inventory of suppliers, the lead times to match our lead time' (1/INT1); and 'Corporate auditors has introduced such leading indicators to implement proactive actions ... '(1/INT2).

Such an auditing approach is used to remedy at source and in a timely manner any variance from the established performance levels, to ensure no returns, with their resultant negative effect on the firm capacity and with its associated risks of failure in the effective customer service. Such problematic issues may originate due to various sources, which may either be dependent or independent of the focal firm, as clearly narrated by a participant, by stating that:

'The products and/or customers are being tracked ... due to front end delays due to material problems, causing risk not to be in time or due to quality issues originated from the incoming verification, such as raw material or from end customer complaints and hence instead of supplying the same product, we try to solve the problems such as quality issue ...' (6/INT 8).

Such an audit stance inspects the **quality of the products** and monitors the **inventory levels**, to ensure no stock-out, so that all **internal and external processes** are **managed effectively**, in line with the **lean management approach**, as referred by three participants, by stating that the:

'Quality manager monitors samples, to operate a lean process with the right quality, defects identification ...' (3/INT4); 'Stock audits are done to detect such kitting errors' (5/INT 6); and 'Lean and six-sigma are used in quality assessment ... to help the investigation process in a systematic process' (6/INT8).

ICT is a tool, which through bar-coding, provides **real time tracking of information and an audit trail** on all products to promote **SC visibility** and at the same time enable the effective **management of manufacturing** and **sales** to **manage** all **risks** effectively and efficiently, by detecting in real time all disconformities, as outlined by three participants, by stating that:

'Every stock has a barcode label to indicate the order number, date of purchase and description'; 'The logistics centre has to cater for the job order records and if not done the scope of the audit trail is lost' (3/INT4).

Logic Diagram of the Auditing Operations tentative category

Table A11.1 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Auditing operations tentative	Positive extreme dimensional	Negative extreme dimensional						
category properties	variation	variation						
Auditing to establish conformity	Full conformity to the established	Lack of conformity to the						
	performance criteria (e.g. KPIs),	established performance criteria						
	both within and outside the focal	(e.g. KPIs) and without a						
	firm, to be in line with the	systematic approach						
	systematic approach of lean							
	management							
Approach to audits	Proactive to remedy all variances	Reactive with delayed remedial						
	in a timely manner	actions						
Regularity of audits	Frequent/regular	Not frequent						
Source of auditing	Internal/External (e.g. focal firm	No audit at all						
	and certified bodies)							
Type of remedial actions for all	Taking the necessary actions in	Delayed action with the resultant						
shortcomings	real-time or at the earliest, to	risk of failure both within and						
	manage risks effectively within and	outside the SC						
	across the SC							
Conditions	Standards (quality and triple bottom line) and legal compliance adopted by							
	certified bodies either locally and/or abroad; Corporate role in Multinational							
	firms; legal and other requirements stipulated by the local Government and							
	foreign country of the products' destination; SC members culture; technology							
	deployment level; risk management approach to promote business continuity; and							
	SC visibility adopted for traceability of all defects.							

 Table A11.1: Auditing operations tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through the auditing operations tentative category, since it establishes an environment where all SC members together with the focal firm, from a holistic perspective, manage their operations effectively by matching all their performance to promote standard operating practices and also adopts a lean management approach and manages all risks to achieve competitive performance for a win-win situation. Such practices need to respect all quality standards and the triple bottom line and also comply to all legal requirements. To achieve effective auditing operations it needs technology deployment to promote real time SC visibility for a systematic evaluation of all SC operations and for the immediate detection of all problematic sources so as to take all the necessary remedial actions in a timely manner to promote business continuity.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The auditing operations conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several
consequences, such as quality standards and legal compliance, effective and efficient performance, business continuity, competitive edge and SC visibility. The moderating variables are the deployment of information technology through ICT applications and by a lean management and an effective risk management approach. The relationship of the auditing operations tentative category with all other categories is represented by the conceptual framework in Figure A11.1.



Figure A11.1: Auditing operations tentative category (Author)

A11.2 Managing Cash Flow tentative category

The cash flow management theme is focused on the effective management of the incoming and outgoing of cash associated with the managing of supplies and sales of products not to run into liquidity problems and at the same time ensure no stock-out or excess stock situations.

The cash flow management describes all the necessary cost management actions needed to ensure sustainable operations in all activities, especially in situations of economic recessions, as one participant clearly expressed that importance of: *'cost savings to make up for future situations that are negative such as recessions'* (1/INT2) and furthermore one participant added that all operations need to be managed with a pull-SC approach to promote less running costs by using the:

'End of month budget [which] is done to ensure improvement in processes, cost savings in purchasing, [and] operations efficiencies'; and '... we are approaching a case of built to order type of company, considering the cash-flow management, economic recession and the wide business portfolio' (2TS/INT3).

The cash flow management is based on the collaborative initiatives of both the focal firm and all the engaged SC members to promote a win-win situation, where focal firm needs to retain the right and minimum level of stock to manage **cash flow** effectively, since 'there is a guarantee that we will buy it but pay when we use it' (2/INT3); and 'The stock needs to be made available [but] without ending up into cash flow problems' (5/INT6).

The focal firm may need to incur additional unforeseen expenses due to the bottlenecks generated by upstream SC members which have a domino effect on the focal firm logistics up to the customers, to make up for such shortcomings, as indicated by a participant, by stating that:

"... we have bottlenecks with regards to suppliers' delivery times, due to breakdown in their machinery, ... So instead ... we need to have more shipments per week ..., which is a more costly shipment. This ... has a knock-on-effect right down on the SC, since the production cannot meet this month output' (6/INT7).

The focal firm adopts a lean management approach, through vertical integration initiatives, to promote better cash flow management, by having all inputs locally at the firm, using a JIT approach and with a lower operating cost with less stock tied in idle capital, as indicated by a participant, by stating that it was crucial to:

'Shift equipment to Malta, to do the components in Malta and not order to store which was leading to idle stock. Malta inventory may be kept low ... and just-in-time may be used' (1/INT2). He further added that lean management is maintained through local sourcing with its proximity advantages, since 'In Malta subcontracting is cheaper, stock is kept low,... customer's market is closer, port is closer and work more efficiently'; and he closed the matter by stating that 'We are operating lean and this helps, since business base on cost per unit'.

Adopting a consignment stock approach at both the upstream and downstream sides of the SC, through an **information sharing** mechanism between all parties, through ICT applications, is fundamental, as clearly referred by a participant by stating that: *'The IT is also driving the process of the SC, since suppliers and clients are requesting that all the members of the SC are to use their software applications to place the order or to see the orders'* (5/INT6). Such an approach promotes improved cash flow management and better **flexibility** in manufacturing operations due to the stock availability with no tied-up cost until usage and with improved credit terms. Such an approach was clearly indicated by two participants, by stating that a:

'A balance need to be made for the stock levels, ... with clients for a consignment stock and also with the suppliers, to balance out cash flow'; '...[suppliers] payment terms with an agreement with a longer period' (5/INT6); and '... we use consignment stock by getting the material and we do not pay until we use it' (6/INT7).

At the downstream side, some sales are provided on demand, by using warehouses near the customers, where the focal firm gains from immediate or faster cash flow recovery and also with an improved and effective customer service, due to faster product delivery. The participant explained such a SC downstream set-up by stating that: '... we hired a warehouse in Germany, adjacent to the customer premises'; 'Although we deliver before being paid, the payment is faster ...' (5/INT6).

Only when suppliers impose stock limits, may cause cash-flow problems, due to forced **purchasing** of excess stock with the result of idle capital tied in stock. The participant explained, by stating that what happens is that: *'The supplier force a minimum order quantity per item, this leads to a problem of over stock'* (INT4).

Logic Diagram of Managing Cash Flow tentative category

Table A11.2 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Cash Flow tentative	Positive extreme dimensional Negative extreme dimensional			
category properties	variation variation			
Supply management inventory	Minimal level of stock as required	Imposed minimum levels of stock		
levels	by the focal firm	by suppliers resulting in excess		
		stock levels		
Liquidity level	Favourable	Negative		
Efficiency in operations	High efficiency (e.g. cost savings	Low efficiency (e.g. sourcing from		
(Competitive and lean operations)	measures such as less running	overseas SC members suffers from		
	costs, faster deliveries through	lack of proximity costs associated		
	local sourcing and negotiate better	with logistics and due to higher		
	supply management prices)	levels of inventory requirements)		
Supply management transactions	Consignment stock or Vendor	Purchasing of stock without any		
type	Management Inventory for stock	concessions		
	supplies			
Sales deliveries	Supply products on demand with	Supply products on order with the		
	fast deliveries through warehousing	stipulated lead time for		
	close to the customers	manufacturing		
Collaborative initiatives	Close collaboration	Lack of collaboration		
Managing logistics expenses	Orders' levels supplied as	Frequent shipping arrangements to		
	requested with one shipment or	make up for suppliers'		
	from local sourcing to gain from	shortcomings to meet the complete		
	proximity advantages	orders		
Conditions	Economic recession; technology deployment in ICT applications; vertical			
	integration by the focal firm; and proximity of the customers and			
	suppliers due to the demanding logistics management needed due to			
	Malta's Small Island State characteristics.			

 Table A11.2: Managing Cash Flow tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through the managing cash flow tentative category, since it establishes an environment where all SC members together with the focal firm, from a holistic approach, **collaborate** effectively with each other with the minimal level of stock tied in capital not to run into liquidity problems. Managing of cash flow needs to be **sustainable** by operating efficiently and has to implement the necessary measures to make up for periods of recessions. The focal firm adopts a **lean management** approach in all operations by undergoing **vertical integration** and manufacture in-house, whenever possible, to promote more efficient processes. Consignment stock is used to improve on managing cash flow effectively by settling payments only on stock usage. The focal firm, whenever possible, needs to provide a faster supply of finished products and retrieve payments at the earliest, through say dedicated warehousing closer to the customer. With such an improvement in both upstream and downstream SC sides, the focal firm being both a buyer and a supplier, promotes a proactive process to achieve improved SC flexibility at both sides of the SC leading to a competitive edge and a win-win outcome. The deployment of technology based information systems is crucial to enable real-time information sharing for SC visibility.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing cash flow conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promote several consequences, such as the level of firm competitiveness in the market sector (e.g. efficiency), win-win situation for all SC members, effective customer service and flexibility across the SC. The moderating variables are collaborative actions adopted with all SC members, shared information initiatives, through technology deployment to promote real time information sharing mechanism and a lean management approach. The relationship of the managing cash flow tentative category with all other categories is represented by the conceptual framework in Figure A11.2.



Figure A11.2: Managing cash flow tentative category (Author)

A11.3 Managing Collaboration tentative category

The managing collaboration theme is about sharing activities and assigning responsibilities to teams, both within and outside the SC, to build strong relationships between people and to incorporate a **level flexibility** to meet effectively and efficiently all the established **performance targets**.

To achieve SCI all the entities making part of the SC, need to undertake **collaborative** initiatives to promote **relationship building** between people and to perform effectively, as clearly outlined by three participants, by stating that:

'[Departments] are assessed on ... collaborative effort ...'; 'Teams are formed to support and report' (1/INT 1); 'Suppliers and buyers are well related between us'; 'Good relationship [exist] between us' (2/INT 3); 'Integration and collaboration is needed to promote flexibility to build the relationship' (3/INT4); and '... the daily interactions are more based on hands on and face to face' (10TS/INT7).

The SCI depth may vary from a **close** to an **arm's length integration**, which both need to be supported by **contractual agreements**. Such an approach was substantiated by various participants, by stating that:

'local staff work closely with US staff in collaboration' (1/INT 1); '... we do contractual agreements' (2/INT3). 'Contracts exist with suppliers' (3/INT4); and '.... our product builds on their product and hence these suppliers are partners and have an in-depth relationship ..., but others are referred as suppliers ...' (5/INT6). Though, such contractual agreements with SC members may restrict the focal firm to engage with any other contractor, as one participant added, by stating that: 'contracts with certain suppliers and contractors are not possible and we cannot use' (2/INT1).

An arm's length collaborative effort, may consists from opportunism and lack of engagement, feedback and cooperation both within and outside the focal firm respectively, which may drive the focal firm to undergo duplicated activities and take special measures, to make up for such shortcomings, as clearly referred by three participants, by stating that:

'Sales and marketing are duplicated ...'; 'The business model is working but can operate more effectively and efficiently if taken ownership of this missing part, ... but with [this downstream sales intermediary], this is a bottleneck' (1/INT1); '... there is always rivalry at work between departments'; 'We still have internally silos thinking within departments' (6/INT7); and 'The various stakeholders are after the same criteria but these may be in conflict with their individual targets/criteria' (6/INT8).

A close collaborative effort refers to a situation where SC members work closely as a **team** or as partners, who are **flexible** and well **trained** to **negotiate** together to meet the expected performance and **change**, as confirmed by three participants: '... [to use] a decentralised approach' (1/INT1); 'Integration and collaboration is needed to promote flexibility to build the relationship' (3/INT4); and '... [to deploy effective] negotiating skills [across the SC] ... '(5/INT6).

With a close collaborative environment the focal firm and all the SC members, promote staff commitment, empowerment, involvement and synergy so as to engage collectively in a **trustful** environment to tackle all challenges in a **holistic perspective**, with a **cross-functional approach** for a **win-win** situation. This approach is clearly expressed by three participants, by stating that:

"... voice of employee in heard ...' (1/INT1); "Workers are involved in all decisions" (1/INT2); and "Synergy and coordination is the order of the day ...'; "Production planner works in-conjunction with the purchasers in the same office"; "The product sales support (PSS) in Malta coordinates/integrates with the overseas sales office and the end-user. PSS goes abroad to the end customer ... to listen and monitor our products ...'; "... collaboration, which if not done, it will work against the relationship building" (3/INT4).

With such a trustful environment, the focal firm will not be subjected to inflated requirements by Corporate's staff or other SC members, which may result in the disruption of the SC performance from the bullwhip effect, as indicated by one participant, by stating that: '... some of the requests may have a hidden agenda such as exploded needs and urgency' (3/INT4).

The focal firm, within and with all SC members, needs to integrate all operations effectively. The external SCI is more challenging to achieve, due to the different cultural values and performance objectives, as clearly indicated by one participant, by stating that: *'The intra was achieved and the inter needs more challenges to overcome'*; and *'For inter-company the performance is more easy to be dealt with due to same values, culture and language with same MBOs'* (1/INT 1).

SCI needs to adopt a level of trust to perform effectively between each other as partners, by building on their experiences, joint activities and track record. Such a trustful relationship has to be the daily norm and goes beyond the established contractual agreements, as highlighted by three participants, by stating that:

'Even when problems exist, I try to coordinate in a positive approach not to demolish relationship, like a give and take situation' (3/INT4); 'When a client engages with a manufacturer, both engage in a trustful environment which promotes open communication apart from the formal contracts' (4/INT5); and '.... our product builds on their product and hence these suppliers are partners and have an in-depth relationship, based on years of experience and history record based on problems solved and opportunities we passed together' (5/INT6).

Suppliers with a substantial level of trust with the focal firm, need to engage in various **value-added measures** (e.g. consignment stock), to promote both **synergy** and a **win-win** situation, since suppliers have guaranteed sales and the focal firm gains from improved cash flow management, as indicated by two participants, by stating that:

'The principle is a win-win situation through the SC ...'; and '... we also use consignment stock, where the supplier having a good relationship ...' (2/INT3).

The firm also needs to build trust with the customers, for managing collaboration effectively with the right customer priority, based on the type of service/product given, to promote further sales. Such an effective customer relationship has been declared by three participants, by stating that:

'In the selling side, we have close contact with distributors and end customers themselves, we sell to both of these entities' (2/INT3); '... all orders are managed according to the priority agreed within the logistics'; (3/INT4) and '... we hired a warehouse in Germany, adjacent to the customer premises' (5/INT6).

ICT tools need to be used for **information sharing**, so as to provide **SC visibility** and to facilitate such a collaborative process, so as to strengthen the relationships. Typical ICT tools, may be either mobile devices (e.g. tablets and smart phones apps) or any computer (e.g. desktop or workstation), which may need to be connected to the internet (e.g. web portals and social media) or networks connected to common databases (e.g. intranet with ERPs and EDIs). Such ICT tools were clearly emphasised by one participant stating that: 'We have the tools today (ERP), and we can see where the work state is. This is IT based'; and 'This helps collaboration between us all as a key issue' (5/INT 6).

Logic Diagram of Managing Collaboration tentative category

Table A11.3 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Collaboration	Positive extreme dimensional Negative extreme dimensional		
tentative category properties	variation	variation	
Relationship type	Close relationship	Arms' length relationship	
	(i.e. cooperation such as	(i.e. opportunism with forced	
	consignment stock for supply	minimum stock order levels)	
	management and customised fast		
	delivery approach using dedicated		
	warehousing near the customer)		
Contractual agreements	Promotes more commitment	Cannot engage with some suppliers	
		being inhibited by the established	
		contractual agreements with certain	
		SC members	
Culture	Common culture	Culture differences	
Trust	Trustful relationship promotes real	Lack of trust may cause inflated	
	requirements	requirements	
Teamwork	Proactive; decentralised;	Reactive (i.e. silo thinking)	
	involvement; and cross-functional		
Training	Knowledgeable Lack of knowledge		
Internal and external SC members'	Adaptable changes Rigid to changes		
flexibility			
Communication	Open communications	Closed communications	
	(i.e. information sharing)	(i.e. limited access to information)	
Problem solving	Constructive approach to problems	Destructive relationship approach	
	with a holistic perspective for a	to problems with silo thinking	
	win-win situation		
Priority	Customised priority	Priority based on scheduled dates	
Conditions	Level of trust between people; cultural differences among SC members;		
	inflated requirements submission by SC members to focal firm;		
	manufactured product type; SC member role (e.g. partner or supplier);		
	and technology deployment level.		

Conceptual summary on the tentative category

SCI is enabled through dedicated collaborative efforts since such a collaborative environment contributes to an improved effective and efficient **performance**, such as lower cost of **inventory** and better **customer service quality**. The collaborative environment is made up from sharing of activities and assigning responsibilities between people, within and across the SC, with either close or arm's length integrated efforts, which are supported by **contractual agreements** to promote a formal commitment to all requirements. A **level of trust** is needed to tackle all challenges from a **holistic perspective** for a win-win situation which goes beyond the commitments stipulated in the contractual agreements. The collaborative efforts need to incorporate dedicated engagements based on strong relationships, such as consignment stock and deliveries of products on demand, in both upstream and downstream sides of the SC respectively. **Technology deployment** enables SC visibility, through established information sharing mechanisms, across the SC. Such a **collaborative** approach facilitates the **manufacturing management of all operations**, the **change management** approach and the **leadership power of control and empowerment** of all people in all scheduled work and in their proactive actions.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing collaboration conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as collaborative action determine the level of the management and leadership effectiveness, holistic approach across the SC for a win-win situation, SC visibility and flexibility, change management and effective and efficient performance approach to achieve a competitive edge over other SCs. The moderating variables are the level of trust and the information sharing initiatives between all SC members. The information sharing is enabled by the technology deployment to promote real-time information exchange. The relationship of the managing collaboration tentative category with all other categories is represented by the conceptual framework in Figure A11.3.



Figure A11.3: Managing Collaboration tentative category (Author)

A11.4 Managing Change tentative category

The managing change theme is about mastering the complex and dynamic changes from both upstream and downstream SC sides and at the same time to be flexible and innovative in all manufacturing and SC operations within the focal firm to promote competitive edge.

To achieve SCI the focal firm needs the **employees' commitment**, together with the **management commitment and support**, to manage change effectively, as two participants clearly emphasized, by stating that:

'Full support to all staff in Malta and my CEO is appreciated [by the Corporate]' (1/INT2); 'Management sometimes has no time to analyse the current situations'; 'The management sometimes has reluctance to change'; and '[in other circumstances tries] to improve by implementing change' (4/INT 5).

Change is driven by **technology** and needs a favourable embedded **culture** of all people within the organisation to adopt a **flexible** and **proactive approach** (e.g. restructuring, improvements and cost savings). The change management approach needs an ingrained culture of a **continuous improvement** with the least resistance to change from all people, since one participant clearly indicated that: *One of the main problems is the resistance to change from certain people such as fear'*; *Certain cultural barriers are difficult to overcome, since you give them the technology but still they do not use it'* (4/INT5). Three other participants adopted a more positive approach to managing change and outlined that:

'Continuous change is the firm policy'; '... there is always room for improvement'; '... we need to go back to the drawing board ... to [use] new components' (5/INT 6); 'The [strategic] decisions are also flexible to discuss [and] to promote improvements on ... some processes'

(3/INT 4); and 'Continuous change is the need to survive'; '...changing the systems as needed' (17TS/INT 14).

Such flexibility can be also achieved through dedicated collaborative initiatives with certain suppliers, by deploying a consignment stock arrangement, by having all the necessary materials in hand, so as to effectively meet any new orders or changes in the manufacturing process, as explained by one participant, by stating that:

'From the suppliers at the input as raw material, we use consignment stock, ... we have more supplies and assist to be more flexible to change plans, having more material in hand' (6/INT 8).

Change promotes **innovation** within and across the overall SC to meet new customers' requirements or replace obsolete stock with a new product design or introduce new processes design in the overall manufacturing stages, so as to improve on the **risk management** approach to promote **business continuity**, as explained by three participants, by stating that: 'We are flexible to meet changes, to meet new customers' specs' (1/INT 2); 'The design may be changed due to the bottlenecks of lack availability in the market of parts' (2/INT 3); and 'Innovation refers to process improvement (Kaizen approach)' (6/INT7).

Logic Diagram of Managing Change tentative category

Table A11.4 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Change tentative	Positive extreme dimensional	Negative extreme dimensional	
category properties	variation variation		
Change management cultural	Continuous improvement culture	Resistance to change	
values			
Commitment to change	Proactive	Reactive	
Technology deployment	Enables and supports through	Manual approach	
	automated systems		
Flexibility	Adaptable processes	Rigid approach	
SC members collaborative	Dedicated consignment stock by	Orders are given by suppliers only	
initiatives	suppliers improves on change	on demand by the focal firm	
	management in manufacturing of		
	orders		
Innovative approach	New products and/or new	Same products and/or same	
	operating processes	processes	
Risk management	Measures taken to promote	Risks of stock-out due to lack of	
	business continuity (e.g. new	business continuity measures (e.g.	
	processes and design of products)	indifference to changes in the	
		supply of components)	
Conditions	Government and private institutions support to the investment for change		
	(e.g. local commercial banks and Government/EU funds for investment);		
	economic recessions; management and staff culture on support and		
	training initiatives.		

Table A11.4: Managing Change tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through the managing change tentative category, as it establishes an environment where all SC members, together with the focal firm, from a holistic perspective, manage change through a proactive approach with the commitment of all people involved. Change management requires an embedded culture to be flexible and adopt a continuous improvement approach to all challenges, with an effective risk management approach, to perform effectively and efficiently and to promote business continuity. Change management improves on the risks by adopting close relationships with suppliers, for more effective stock deliveries, such as with the implementation of a consignment stock approach. Technology investment in automation and in ICT based applications enable the change management approach of the focal firm to become more agile in meeting changes due to the automated processes and due to the availability of real time information. Such improvements refer to changes in manufacturing processes and in product re-design to promote innovation.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The change management conceptual element is considered as part of the SCI phenomenon, where with the support of various moderating variables, promotes several consequences, such as customer services improvements, processes improvements and products re-design to promote innovation, business continuity and a competitive edge. The moderating variables

are the collaborative initiatives by all stakeholders, risk management approach, the human element flexible and proactive approach and the investment in technology. The relationship of the managing change tentative category with all other categories is represented by the conceptual framework in Figure A11.4.



Figure A11.4: Managing Change tentative category (Author)

A11.5 Managing Information Sharing/Communication tentative category

The managing information sharing/communication theme is about capturing and sharing the right information at the right time, within and across the SC, so as to take all the necessary timely collaborative actions so as to promote an integrated approach to all operations and to achieve effective and efficient performance.

To achieve SCI, **information sharing** and **communication** are to be used to enable **integrated operations** with a **continuous improvement** approach with less duplication of efforts, both within the focal firm and across the SC, so as to promote effective and efficient performance. Such minimisation

of duplication and continuous improvement approaches respectively, was explained by two participants, by stating that: 'This access to information will avoid the making of any query to the supplier' (4/INT5); and 'The idea is written down on the board and is reviewed, monitored and implemented from the written information on the Kaizen board' (6/INT7). Such an information sharing environment is promoting an integrated approach, as emphasised by three participants, by stating that: 'Data of progress is visible to all through monitors in the shop floor'. 'From suppliers' side, we share stock levels...' (2/INT 3); 'The sales offices will be updated on their orders and the stock and state of production of the firm' (4/INT5); and 'Communication is a must for the SCI depth, that all data arrives to all people and at the same time' (6/INT8). The promoting factors of SCI, as clearly declared by a participant, is communication, by stating that: 'Definitely, communication between all external entities (firms etc) and internally between all departments' (5/INT 6).

The information shared can either be of a **knowledge** or of an **operational** nature respectively, to meet all collaborative requirements, as referred by one participant, by stating that: '... sales office ... to listen and monitor our products and their use and any problems'; and 'The logistics layer is important to pass information' (3/INT 4).

Face to face communication within daily meetings and among staff located physically together or virtually, wins over all other types of electronic communications (e.g. ERP or database systems and e-mails), as clearly expressed by three participants, by stating that 'We measure and discuss everyday all operations' (1/INT 2); the '*Planning and purchasing department are within the same office to promote better communication*' (2/INT 3); and 'We have weekly meetings through video conferencing call' (6/INT7).

Such information flow provides **SC visibility** which builds on the level of **trust** across the SC so as to promote improved **collaborative** initiatives for better **teamwork**. Such a trustful environment creates a business scenario, where the focal firm may adopt an open book with a continuous dialogue approach with the respective SC member, which goes beyond the contractual commitments. In fact three participants outlined the need for such a trustful environment, by stating that:

'The trust is needed since we share confidential information' (3/INT4). 'When a client engages with a manufacture both engage in a trustful environment, which promotes open communication, apart from the formal contracts' (4/INT 5); and 'Effective communication is key. Namely not pass the message, but entering in a continuous dialogue with others' (9TS/INT 6). Though, it cannot be excluded that some SC members do not give access to the focal firm to remain updated with all the necessary information, as referred by one participant, by stating that: 'The end customer of [the Focal firm] is not seen' (1/INT 2).

ICT applications, through in-house databases or through the internet, are the enablers for such realtime information sharing and communication mechanisms to promote SCI, as clearly outlined by two participants, by stating that: '*E-mails are being used with various customers'; 'Suppliers are contacted* ... by a simple e-mail' (3/INT 4); and '*The SC uses cloud computing and groupware systems* ... with accounting system, ERP, BOM ... which comprises interfaces through web with sales office'; '... these tools are pushing more integration across the SC' (4/INT5). Such information sharing, through technology, enables the **monitoring** and **management** of all operations of all products through the use of the bar-coding system, which provides an **audit** trail of all the processes in real-time, as referred by the same participant, by stating that: '*The barcode is used to track the items across the SC*' (4/INT5).

Such **information management** enables the focal firm to holistically manage all operations effectively and to detect any deviation in the scheduled performance in a timely manner, as one participant declared the importance to have: '... *a birds' eye view at one point';* and '*This needs information of all the processes'* (4/INT5). With an effective information management stance, it enables management to take **informed decisions** so as to remedy all problematic situations in real-time and achieve all targeted performance in a timely manner, as referred by one participant, by stating that: '... *based on the information, knowing future projects in the pipeline to have a clear visibility of*

all the projects and assist us in our plans and the material orders in place, we can achieve the targets' (2/INT3). All information needs to be made available, so that **performance is measured** and **all risks** are managed effectively, as declared by three participants, by stating that: '*Risk management is important since all operations are being monitored to deliver*' (3/INT4); '*The SC measurements are in place to measure all works in progress*' (6/INT7).

Logic Diagram of Managing Information Sharing/Communication tentative category

Table A11.5 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Information	Positive extreme dimensional	Negative extreme dimensional
Sharing/Communication	variation	variation
tentative category properties		
Depth of information sharing	Proactive and in-depth information	Limited information sharing with
	sharing (e.g. operational and	an inhibited approach
	knowledge based information) for	
	informed decisions	
Timely information sharing	Real-time	Delayed
Communication approach	Face-to-face verbal communication	Written communication based on
	with the support of electronic	IT applications only (e.g. e-mails)
	communication tools	
Monitoring and tracking	Performance measurements and	No measuring or auditing at all
performance to manage all risks	audit all processes to promote a	
	holistic SC visibility and to detect	
	any deviations in real-time	
Information sharing and	Close relationships leading to	Arms' length relationships
communication for collaborative	integrated operations with less	
efforts	duplication of tasks and with a	
	continuous improvement approach	
Conditions	Technology deployment in ICT appli	cations; and level of trust between
	SC members.	

Table A	11.5:	Managing	Information	Sharing/Co	ommunication	tentative of	category	(Author)
								(

Conceptual summary on the tentative category

SCI is enabled by sharing and communicating on both technical and operational information to promote SC visibility across the SC tiers. With the right information in hand, it enables the focal firm to drive production, identify problems, assign priorities by stock code and track the product/s and works with a continuous improvement approach across all SC tiers. Information is retrieved from every SC tier, such as from suppliers, contractors, logistics providers, customers and even from the internal focal firm itself, through either face to face or through electronic communication. Face to face communication, either physically or virtually, remains the best approach for effective communication but still needs the support of technology based information management systems (e.g. ERPs, e-mails and databases) to capture the needed information, since ICT enables monitoring and tracking of all works, such as through bar-coding. Once information is shared between the SC members, this initiates the process of communication to interact together, support each other and takes the necessary informed decisions to solve problems, to manage all risks and to promote a collaborative approach between each other. The level of information sharing depends on the level of trust between SC members which goes beyond the terms within the contractual agreements.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The information sharing/communication conceptual element is considered as part of the SCI phenomenon, where with the support of various moderating variables, promotes several consequences,

such as effective and efficient performance, collaboration, information management, performance measurement, risk management, SC visibility, informed decision-making and continuous improvement. The moderating variables are the level of trust between the respective SC members and the technology deployment through ICT based tools. The relationship of the managing information sharing/communication tentative category with all other categories is represented by the conceptual framework in Figure A11.5.



Figure A11.5: Managing Information Sharing/Communication tentative category (Author)

A11.6 Managing Customer Service tentative category

The managing customer service theme is to provide customers an effective service quality with 'the right product, at the right time and at the right price' and 'to ensure that the product has the right quality' (6/INT 8).

To achieve SCI, the focal firm needs to have a **direct contact** with all the SC members including the end customer and with **interconnected processes**, by **monitoring effectively** the upstream SC not to disrupt the downstream SC, so as to achieve the expected end **customer service quality**, as two participants explained, by stating that: '*Direct contact with the customer service quality*, as two *win the customer feedback and [its] management*' (1/INT 1). '*The client [as the focal firm] has access to suppliers, [updated on] the stock available*' (4/INT5). Though, it cannot be excluded that in some cases the focal firm has no direct connection with the customer and as a result misses crucial information, as clearly outlined by a participant, by stating that: '*The Maltese plant, has no contact with the end-user*' (3/INT 4).

The focal firm needs to assign different **priorities** to each customer to achieve the scheduled quality of service and to build a healthy **customer relationship**, where honesty is the best policy, as referred by one participant, by stating that: *'Production planner assigns the priorities or proceed according to FIFO, but we are flexible to the job order sequence, ... due to the type of customer'* (3/INT4). 'Honesty is the best policy, and we need to inform customer [on any shortcomings]' (10TS/INT7).

The focal firm needs to meet all the SC members including the customers' changing requirements to promote **SC flexibility**, as explained by one participant, by stating that the focal firm needs to: '... *undergo the necessary transformation to adapt also to the customer needs*' (5/INT 6).

The focal firm may undergo **vertical integration** in the downstream side of the SC, to become more **lean** and closer to the customer, so as to improve on the customer service to promote **SC responsiveness**, as described by two participants, by stating that:

'SC includes the [Company X as the distributor] up to the customer. This is a limiting factor of the SC'; '...customer side need to be lean' (1/INT1); and 'Lean removed the waste in the processes, to be more responsive to the customer requirements and is faster' (6/INT8).

The customer service needs to be integrated within the focal firm due to the nature of the SC, being a pull-SC, where all orders initiate from the customers' side and needs to be managed effectively to give the right support, as indicated by three participants, by stating that:

'The future is seen to be built from the information up to the end customer' (1/INT2); 'The customers are also imposing integration, so that they can see ... design features of the product but also on the process and product flow [delivery]' (4/INT 5); and 'If you were not integrated as the way we are, it will limit the customer service due to fragmented operations'; '.... we will attend for a hands-on visit ... [at the customers' premises, if there is the need]' (6/INT 7).

ICT applications remain a key enabler in all operations related to all stages of the product lifecycle and promote **information sharing** across all SC members down to the customer. This is clearly expressed by one participant, by stating that: *'The IT is also driving the process of the SC, since suppliers and clients are requesting that all the members of the SC are to use their software applications to place the order or to see the orders they have for you. Practically the firms are merging with each other'* (5/INT6).

Logic Diagram of Managing Customer Service tentative category

Table A11.6 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Customer Service	Positive extreme dimensional variation	Negative extreme dimensional variation	
Contact with the customer	Direct (i.e. vertically integrated in the downstream side of the SC)	No direct contact but through third parties	
Customer Integrated approach	Close interactions with honest approach e.g. on-site customers' visits	Arms' length	
Customer priority	Certain customers may need to be allotted a priority	No customer priority at all	
Meet customer's changing requirements	A responsive, flexible and adaptable customer support	A passive customer support	
Employ lean management to improve on the customer service	Lean operations through downstream vertical integration to be closer to the customer	No dedicated lean approach	
Information updates from all the SC members including the customers	Access to all information, enabled by ICT, from the customer to be aware of all requirements	No access to customer information	
Conditions	Technology deployment, through ICT applications, provides customers access for the order lifecycle management information.		

Table A11.6: Managing Customer Service tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through the managing customer service tentative category, since to perform effectively the focal firm needs to manage effectively the customer service to retain satisfied current customers and win others, through an ongoing relationship building, by adopting the right customer priority. Such a competitive edge is met through a lean management approach to provide the right products at

the right time and at an attractive price and with the right quality. Lean management is used so that the focal firm adopts less waste in all processes and to be more responsive to changing customers' needs. Such a lean approach is facilitated with a direct customer contact with no intermediaries in the SC, such as by undergoing vertical integration. ICT applications are used to facilitate all interactions with all customers, through all necessary information updates and follow-ups of all orders.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing customer service conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as SC flexibility and responsiveness, lean operations, customer service quality and customer relationship. The moderating variables are technology deployment through ICT applications, lean management and vertical integration. The relationship of the managing information sharing/communication tentative category with all other categories is represented by the conceptual framework in Figure A11.6.



Figure A11.6: Managing Customer Service tentative category (Author)

A11.7 Managing Knowledge tentative category

The managing knowledge theme is about knowledge acquisition through information sharing within and across the SC, to have an informed mindset to promote effective decision-making to deploy state of the art processes through business process re-engineering, both within manufacturing and SC operations.

To achieve SCI, the focal firm shall undergo **training** to promote **learning** and a **change management culture** among all staff in the deployment of newly designed processes. The learning culture needs to build on **knowledge**, so as to adopt an effective **decision making** approach to **reengineer** all manufacturing and SC processes with the right investment in **technology-based production processes** and solutions (e.g. advanced manufacturing systems) and with the required **integrated information systems** (e.g. ERP or common databases). Such an approach was referred by three participants, by stating that:

'They need to undergo business process engineering effectively and deliver the training needs to staff appropriately and on a continuous basis'; 'The order trend, with suppliers is integrated over internet (IT)'; 'The SC uses cloud computing and groupware systems,

.....which comprises interfaces through the web with sales office, in the global world' (4/INT 5).

Such a SC re-engineering approach will lead to a **higher quality** and a **lower operating cost** with **innovative processes**, both **within and across the SC**, to promote **lean operations**, as referred by three participants, by stating that: '... material reduction and validation of new resin material used to reduce cost and better quality to improve efficiency, hence the knowledge sharing occurs on these matters'; '... share the project works and innovative ideas ...' (1/INT 1); and 'Training is given to eliminate waste' (3/INT4).

The focal firm, in-conjunction with all SC members, from customers to suppliers, including the Corporate, need to build on a **cross-cultural knowledge** platform to **share information** in a teamwork approach. Such **knowledge sharing** initiatives lead to an informed mindset and promote **excellence** with a **continuous improvement**. Such an approach was described by three participants, by stating that:

'Particular staff (engineers) has been trained in lean six sigma and given Kaizen training' (1/INT1); 'The relationship is healthy with suppliers and inform us on new products/changes' (3/INT4); 'Products have been built through these partners (key-suppliers)'; and 'Operational excellence value is key here, because managers are expected to improve processes, people and systems' (6/INT 8).

Particular SC members may need a more dedicated knowledge sharing approach than others due to the difference in the cross-cultural knowledge and values, as explained by one participant, by stating that: *'The Asian market expects details and full support on a continuous basis. The EU countries are more understandable and flexible to meet all problems and discussions'* (3/INT4).

Logic Diagram of Managing Knowledge tentative category

Table A11.7 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Knowledge tentative	Positive extreme dimensional	Negative extreme dimensional	
category properties	variation	variation	
Training	Ongoing approach	Reactive approach	
Information sharing	Open platform to share all information within and across the SC	Minimal information sharing except for exchanging orders information	
Learning culture	Commitment to learn through training and knowledge sharing with all SC members	No commitment to learn both within and outside the SC	
Innovative approach to re-engineer	Proactive commitments to establish	Reactive approach to all possible	
operations (i.e. change management approach)	the necessary improvements through re-engineering	improvements	
Sensitivity to different SC	Dedicated effort to exchange	No efforts to meet cross-cultural	
members' culture to knowledge	knowledge with SC members who retain different cultural values	differences	
Conditions	Investment in technology in automation and ICT applications; Corporate support in all training initiatives; culture adopted by people in learning and change management; and commitment of all SC members to share knowledge.		

 Table A11.7: Managing Knowledge tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through managing knowledge tentative category, as it establishes an environment where all SC members, together with the focal firm, manage knowledge through a learning culture with the involvement of all stakeholders to reengineer all SC and manufacturing processes to promote higher quality, lower operating costs and innovative initiatives in line with lean operations. With such a building knowledge approach, the overall SC stakeholders will have an informed mindset with the aim to take effective decisions to promote excellence and to manage change with a continuous improvement approach.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing knowledge conceptual element is considered as part of the SCI phenomenon since it is an action category to promote several consequences such as a learning culture, innovative processes, effective decision-making, continuous improvement, lean operations and change management. The moderating variables are technology investment in automation and ICT based applications that moderate all information sharing initiatives, people culture to learning and change management culture and training programmes. The relationship of the managing knowledge category tentative category with all other categories is represented by the conceptual framework in Figure A11.7.



Figure A11.7: Managing Knowledge tentative category (Author)

A11.8 Lean Management and Quality Standards Compliance category

The lean management theme is about how a focal firm together with all its SC members integrate themselves to deploy a systematic approach in all processes to be efficient and effective in the deployment of all resources.

To achieve SCI, the focal firm needs to adopt a **lean management strategic approach** so as to adhere to the established quality standards, sustainable and legal requirements, so as to maximise the use of **resources** and eliminate the **waste** in all **processes**, both within the focal firm and across the SC. **Lean management** provides a systematic approach on how to manage all operations and at the same time improve on the overall firm performance and on the customer service. Three participants substantiated such a lean management approach, by stating that:

'.... with lean six sigma introduction ... this led to a better integration and such changes were possible to shift equipment and gain from transport costs, delays and smoother operations in Malta' (1/INT2); 'Even office work need to be lean to eliminate duplicated office activities' (3/INT4); and 'Lean and six-sigma are used in quality assessment to apply the principles in ... a systematic process'; 'Lean removed the waste in the processes, to be more responsive to the customer requirements and is faster' (6/INT8).

Lean management needs that the focal firm **restructures** itself and deploys a **cross-functional** approach in all operations and must comply to all defined quality **standards**, **legislations** and **sustainable responsibilities** in the sector. Such a lean approach needs to extend with all SC members to promote aligned processes and with a continuous improvement approach to promote business continuity. Four participants described the measurements needed to achieve lean management, by stating that:

'Restructure was made and the SC needs to be lean' (1/INT1); 'We are ISO certified' (2/INT 3); '.... the UL affects the SC flexibility in the use of material and stock'; 'All components need to be RoHS directive for Environmental, segregation of waste, minimize waste' (3/INT4); and 'Lean looks for efficiency in all processes with a continuous improvement approach with strict guidelines to EHS for all workers' (5TS/INT 2).

The focal firm may **vertically integrate** and/or use local contractors which are preferred from overseas **outsourcing** to be lean. Such a lean management approach, was explained by three participants, by stating that: 'Such a vertical integration promotes lean management'; 'SC full ownership [gives] a lot of benefits'; 'Getting a supply from Mexico is not lean' (1/INT1); '... we use locally to benefit from lead time' (2/INT3); '[local suppliers are] more flexible since I can push the supplier to perform in time' (7TS/INT1).

JIT shall be used, wherever possible, by submitting orders when they are needed, to improve on **inventory** and **logistics management** respectively. Such a JIT approach is function of the industry sector and external economic conditions such as recessions. One participant explained, by stating that: *'....just-in-time approach is being used which means no pre-orders are made, but when needed they are ordered'; 'Forecast was made as usual and due to recession the forecasts being requested were not met by the key suppliers (distributors) since they gave us very long lead times' (3/INT4).*

The SC need to use a low **inventory management** approach to minimise **costs** by deciding on the right trade-off between **ordering the right level of stock** and minimise the **logistics costs** associated with **stock movement**, by employing **JIT**, and whenever possible, avoiding **just-in-case** orders. Stock movement, whenever possible, needs to be moved directly to manufacturing and not to warehousing to minimise costs in logistics, but this is dependent on the **suppliers' commitment** to meet all **scheduled targets**. Wherever strong relationships exist with SC members, a consignment stock approach shall be deployed to gain from the advantages of stock in hand with improved cash flow due to less capital tied in inventory to promote **lean operations**. This situation was explained by two participants, by stating that:

'VMI or consignment stock is used. This is used to minimize the stock levels'; '... just-in-time approach is being used which means no pre-orders are made, but when needed they are ordered' (3/INT4); and '... put more products in the shipment to cut the costs' (6/INT7).

Information sharing through **ICT applications** shall provide a platform of real-time information to process all information at source and wherever possible to implement automated activities, to avoid duplication, which are needed to **enable** a lean management approach through efficient **processes** within and across the SC. Such lean operations were referred by three participants, by stating that:

'IT systems are used real time to be updated with all information and changes' (2/INT3); 'The role of IT assures the stock levels in a very efficient way'; 'The pdf need to be keyed, a level of duplication but cannot be avoided' (3/INT4); and 'Technology is a big enabler in this case and has contributed to large increase in efficiencies' (4/INT5).

Technology based production processes cause the transformation of processes through automation to become more efficient and lean, as described by one participant, by stating that: *'With full or semi automation machines (robotics arms etc) the cheap labour is won over, since one has 24/7 shifts and high quality'* (4/INT5).

Logic Diagram of Lean Management and Quality Standards Compliance tentative category Table A11.8 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Lean Management and Quality	Positive extreme dimensional	Negative extreme dimensional	
Standards Compliance tentative	variation	variation	
category properties			
Waste management	Minimise waste in all processes	No waste management at all	
	and materials usage		
Systematic approach (i.e. lean	Processes are all organised to	Processes do not follow any	
management)	follow all established standards	established standard with lack of	
	with a continuous improvement	improvements	
	approach		
Process approach	Process based on efficient and	Process lacks efficient and	
	effective approach	effective measures	
Ownership span	Vertical integration in both/either	Limited integration and ownership	
	downstream and/or upstream sides	initiatives outside the focal firm	
	of the SC to be more lean		
Sourcing	Outsourcing from local SC entities	Outsourcing is independent from	
	to gain from JIT operations and	proximity consideration of SC	
	proximity issues	entities	
Conditions	Technology deployment in both/either ICT and/or automation; industry		
	sector; economic recessions; and collaborative initiatives to gain from		
	special arrangement with suppliers (e.g. consignment stock)		

 Table A11.8: Lean Management and Quality Standards Compliance tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through lean management and quality standards compliance tentative category, since it establishes an environment where all SC members, together with the focal firm, integrate all operations with a systematic process to be efficient in the use of all resources and effective in meeting the established quality standards and sustainable and legal requirements. Such an approach maximises the use of all resources by minimising waste and by employing the necessary restructuring to promote a cross-functional approach, both within and across the SC, to promote lean operations and an effective customer service. Wherever possible the focal firm vertically integrates to gain from inhouse operations or use local sourcing to gain from proximity issues, to deploy JIT so as to gain from a low inventory management approach and effective logistics. Furthermore, the focal firm preferably deploys a consignment stock approach with its suppliers to benefit from improved cash flow management having less capital tied in stock. To deploy lean management, wherever possible, all information sharing applications shall deploy automated tasks through the use of ICT applications to manage all information effectively and similarly all production processes shall use automated technology to promote efficiency in manufacturing.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The lean management and quality standards compliance conceptual element is considered as part of the SCI phenomenon since it is an action category where with the support of various moderating variables, promotes several consequences, such as lean operations and sustainability, efficiency and effectiveness, performance improvements, effective customer service, cash flow, logistics and inventory management, continuous improvement, business continuity and information management. The moderating variables are technology deployment through ICT applications, lean management and vertical integration collaborative initiatives with suppliers, sustainability and legal requirements, information sharing with the support of technology deployment in ICT and automation. The relationship of the lean management and quality standards compliance tentative category is represented by Figure A11.8.



Figure A11.8: Lean Management and Quality Standards Compliance tentative category (Author)

A11.9 Managing Logistics tentative category

The managing logistics theme is concerned with managing the transportation of all merchandise from both the upstream and downstream SC sides in the right time, with the right cost and in the right place.

Managing logistics is a core activity within SCM, since it provides the movement of all raw materials and finished products across the SC, otherwise the manufacturing operations cannot commence or if delayed, may have a ripple effect on the overall SC performance, with the risk of failing to deliver in the scheduled time to meet the expected customer service, as one participant declared, by stating that: *'The logistics is driving the whole set-up'* (3/INT4). It cannot be excluded the importance to consider all logistics requirements of all in-house movements within the factory premises, due to the limited factory space availability, ending up to use fragmented factory units, as one participant highlighted, by stating that: *'Malta has limited space of land and hence most factory units are located apart, that is have their land which is distributed'* (4/INT5).

The transportation mechanism mode of transport (i.e. air, sea or land) is function of the type of merchandise (e.g. weight and volume), distance and geographical nature of the country, being either a land locked or sea locked type. Malta, as a Small Island State, with its insularity characteristics has a relatively expensive and less efficient transport and logistics infrastructure, to meet its tight lead times, due to its infrequent cargo shipping schedules, since it needs to use either sea or air, as the first shipment mode of delivery. Once the mainland is reached the next mode of transport needs to be selected, to resume its journey to reach all suppliers' or customers' localities across the globe, such as Europe, USA, North Africa and Asia, among other countries. Four participants explained such a challenging logistics situation, by stating that:

'Logistics is the main problem in Malta'; 'The Sea shipping routes are quite limited and are expensive'; 'Air travel is used for fast moving products and is expensive' (2/INT3); 'This firm has heavy stock/material and we do not consider airfreight and courier'; 'From Germany we have 5 days transit time'; We are using more the Asian market, with 28 days transit time' (5/INT6); and 'Malta is an island. This affects logistically, we take 3 weeks to get the deliveries. Abroad the delivery is 3 days, due to transportation on the road'; 'Freight by air is used for urgent orders' (14TS/INT11).

To achieve SCI the focal firm needs to collaborate effectively with both upstream and downstream SC members, including the freight forwarders. Managing logistics is very challenging and needs to respect both the **lean management** approach and all **environmental responsibilities**, by deploying **effective and efficient logistics management** (e.g. use the most efficient shipping method and effective shipping frequency; use the shortest route with the least carbon footprint; and to manage the capacity of the containers' space or use groupage). Three participants outlined such an effective and efficient logistics management, by stating that:

'Consideration is being given by freight forwarders for route planning in line with the roadmap 2050 for low carbon Europe' (2/INT3); 'SCI within logistics is needed since I always tried to build a good relationship with suppliers and sales (overseas)' (3/INT4); and '... stock being kept decreased due to such a faster process of delivery'; '... we made vast reduction with the [freight forwarder contractor] on the inventory terms'; '.... [we] put more products in the shipment to cut the costs'; 'Italy is near but the stock is not, but we only ship twice a week... so it affects the SC'; and 'The transportation ... have [abroad] feeder trucks everyday' (6/INT7).

Such an ideal lean management scenario may not always be possible to achieve since some suppliers may fail to meet all contractual obligations, which drive the focal firm to undertake more frequent shipments with an overall increase in the scheduled expenses, as one participant clearly declared, by stating that: '... we are not having the material on time. So instead of having one shipment per week with a container, we need to have more shipments per week which incurs more costs[this] is a major bottleneck' (6/INT7).

The SC, which may employ an upstream side **vertical integration**, needs to manage the **inventory** inline with the **lean management** principles to improve on the overall **performance**, such as to meet all stock requirements with the least stock, respect lead times and use lowest transportation cost, as it was referred by two participants, by stating that: *'It is part of the lean six sigma to produce these components here...[to] avoid the transportation costs'* (1/INT2); and *'Process improvement ... refers to the use of less material in our product ... less load in weight for shipping'* (6/INT8).

ICT applications, enable **visibility** of all merchandise' movements across the SC tiers (e.g. web portals transportation tracking services). ICT serves to **synchronise** all **processes** and **actions** and serve as a **risk management** tool to detect any divergence from scheduled target dates of all logistics to promote the necessary timely recovery actions and at the same time, for certain merchandise, evidence through data loggers are used as a guarantee on the temperature stability for the whole journey. Such an ICT support has been described by three participants, by stating that:

'[Freight forwarder] has a tracking means for logistics' (1/INT2); 'The logistics layer is important to pass information and promote visibility (3/INT4); and 'In Malta we use barcodes [few use RFID]'; '[Particular products are] transported by special containers with temperature controlled equipment where data loggers [are used]' (4/INT5).

Logic Diagram of Managing Logistics tentative category

Table A11.9 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Logistics tentative	Positive extreme dimensional Negative extreme dimension		
category properties	variation variation		
Sea Transportation availability	Frequent	Not frequent	
Space availability within factory	Organised factory space with no	Lack of space within factory resulting	
premises for effective warehousing	duplicated handling and movements	in disparate allocation of items with	
		handling duplication	
Environmental responsibility for the	Most environmental friendly route	No consideration to environmental	
selection of transport route with the		friendly route	
least carbon emission			
Dedicated services during	Temperature controlled environment	No dedicated service for merchandise	
transportation for effective customer	(i.e. reefer container)		
service			
Lead time management	Meeting targets	Delays in deliveries	
Transport available modes (sea, land &	Use of the most effective and efficient	No consideration to route effectiveness	
air)	route to deliver the merchandise	and efficiency	
Lean management based transportation	Lowest transportation cost, right	No lean considerations at all	
	shipping frequency, shortest route,		
	environmental friendly route and		
	exploit container space.		
Conditions	Manufacturing sector determines the type	e of transport mode (i.e. air, sea and land)	
	due to the product weight and volume; ICT applications to promote visibility of		
	all orders' movement by freight forwarders and couriers (e.g. tracking on web		
	portals); level of vertical integration of the focal firm across the SC; and		
	merchandise special requirements (e.g. temperature controlled transportation)		

 Table A11.9: Managing Logistics tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled by the managing logistics tentative category, as logistics needs to be managed effectively and efficiently to ensure that the movement of merchandise arrives at the right time, with the right cost and in the right place, so that the manufacturing process may start at the scheduled date and that the customers' scheduled deliveries are met by the agreed delivery date. Such an environment requires the collaboration of the relevant SC members, which together with the focal firm, interact to meet the scheduled customer service to promote a competitive SC. The location of the focal firm being in a Small Island State, such as Malta, makes managing logistics a more challenging role, being a sea locked island, which demands the use of either maritime or air transport service to get out of the country to the mainland of destination and vice versa. Logistics is part of the overall SC and manufacturing operations to promote lean management and to meet all environmental responsibilities. Any delays on the logistics commitments may need the focal firm to engage in timely recovery actions, to eliminate any risk of failure in the overall SC performance. ICT based tools are used to provide SC visibility to allow all SC members to track all orders by using hotline phone numbers or through on-line access to web portals, provided by freight forwarders or couriers on every order.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing logistics conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences such as customer service, lean management, SC visibility, risk management, effectiveness and efficiency and environmental responsibilities compliance. The moderating variables are ICT and the transport and logistics infrastructure. The relationship of the managing logistics tentative category with all other categories is represented by the conceptual framework in Figure A11.9.



Figure A11.9: Managing Logistics tentative category tentative category (Author)

A11.10 Managing Performance Measurements tentative category

The managing performance measurement theme is about measuring people's behaviour, attitude, skills and commitment to work, to determine their contribution to the overall firm performance. Such an assessment needs to cover not only the focal firm, but also all SC members, to promote an aligned SC with a continuous improvement approach.

To achieve SCI, **performance measurements** are used within the focal firm to cover all management levels down to the bottom line so as to monitor, on a daily basis, the overall performance effectiveness and efficiencies and at the same time create awareness to all staff of all their individual performance targets. The focal firm must then align its operations with all its SC members, by implementing **joint performance benchmarks**, to create awareness of such performance indicators to all involved, to promote the necessary **performance improvements** and settle all the necessary payments. With such a performance measurement approach it enables to uncover all the established gaps, both within and outside the focal firm, so as to meet all the scheduled performance targets effectively. The managing performance measurement scenario was explained by three participants, by stating that:

'[Firm] performance is also measured and service issues are being measured and gaps identified to ensure improvements'; 'We measure and discuss everyday all operations' (1/INT 2); 'KPIs are used to monitor the progress regarding material, efficiencies of work carried out per station' (2/INT 3); and 'The invoicing performance is monitored to ensure performance' (3/INT4).

The focal firm aligns all its internal and external operations with all SC members, through both **transactional** and **transformational leadership styles**, with an ongoing performance measurement approach. Both leadership styles ensure a **motivated workforce** all across the organisational levels, where all **collaborative efforts** are **coordinated**, both internally and externally to the focal firm, so as

to contribute to the overall **SC performance improvement** and to promote a **holistic competitive edge** across the SC. The importance of the leadership styles together with the deployment of performance measures, has been described by four participants, by stating that:

'Leadership competencies for example is within the appraisal' (1/INT1); '[Focal firm] focus[es] on cross-functional operations, open style policy, teamwork, performance measurements' (1/INT 2); '... through performance measurement since one has more direction, benefits and motivation' (5/INT 6); and 'The manufacturing process needs to report back on a daily basis and on a shop floor level declares on an hourly basis the quality and quantity in-line to the standards' (6/INT8).

The focal firm must use performance measurements to capture the attitude and performance of the individual employees, the teams and the overall organisation, as a means to assess and reward all employees' **engagement, commitment, skills and empowerment** at all the management levels. Such people management, through the deployment of performance measures, has been described by four participants, by stating that:

'... measurement is done on these competencies under the performance appraisal'; '....corporate surveys ... to show the engagement of the employees to the firm' (1/INT 1); 'The delivery performance is all over the staff' (3/INT4); 'We have a SC measurement for every line' (6/INT7); and 'The production bonuses are on a daily basis and based on three criteria, namely the attitude, quantity and quality at the individual level ... and team level' (6/INT8).

Such performance measurements are based on both lagging and leading daily indicators, which need to be included in the monthly reports in line with the KPIs. The lagging indicators consider performance measurements as a reactive approach whilst the leading indicators consider them as a proactive approach. The leading indicators promote a climate of performance improvement. The proactive and reactive approaches, through the use of performance measurements, have been outlined by one participant, by stating that:

'EHS has lagging factors. Lagging indicators measure the lost time accidents and lost time days measurements'; 'Corporate auditors has introduced such leading indicators to implement proactive actions... to prevent the incidents for better assessment' (1/INT2).

ICT remains the tool that **enables** the management of performance measurements, since ICT is able to facilitate the collection and monitoring of all measurements to provide all the information needed by all SC members, so as to be aware of the real time performance and of all the gaps and at the same time establish what needs to be done to meet the scheduled **competitive benchmarks**. ICT is also used to capture all information from processes associated with **automated production**, so as to track productivity on automated tasks and to promote all the necessary performance improvements. The role of ICT in managing the performance measurements, has been outlined by two participants, by stating that:

'... [another IT based system is used to] monitor the progress regarding material and efficiencies of work carried out per station ...'; 'IT systems are used real time to be updated with all information and changes' (2/INT3); and 'Automation enables detection of efficiencies' (4/INT5).

Cases exist where the performance does not represent the actual performance due to the way it is measured (e.g. measures exclude the firm maximum capacity, so rejecting orders due to full capacity is treated as lost orders). Such a management approach, based on an inappropriate application of performance measurements, has been emphasized by one participant, by stating that: *'The delivery performance does not consider the capacity of the firm'* (INT4).

Logic Diagram of Managing Performance Measurements tentative category

Table A11.10 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Performance	Positive extreme dimensional	Negative extreme dimensional	
Measurement tentative category	variation	variation	
properties			
People performance measurement	Motivated workforce with full	Reactive workforce with a passive	
with the right leadership approach	commitment and empowerment in	approach to improvements	
	a continuous improvement		
	approach		
Alignment of focal firm with SC	Aligned focal firm with all SC	No alignment of operations	
members in a holistic approach	members with joint established	between focal firm and all SC	
	performance benchmarks	members deploy disparate	
		performance measures	
Tracking of performance	Work is monitored and measured	No monitoring at all	
	in real-time		
Performance indicators	Based on leading indicators to	Based on lagging indicators which	
effectiveness	promote proactive actions	promote reactive actions	
Application of performance	Performance measurements are	Performance measurements are	
measures	applied effectively to represent the	applied inappropriately and does	
	reality of the performance	not represent the real performance	
Conditions	Technology deployment to create an information sharing platform of		
	performance measures, from both ICT and automation sources; and an		
	effective leadership style to meet all innovative changes (i.e.		
	transformational leadership) and/or daily operational commitments (i.e.		
	transactional leadership).		

Table A11.10: Managing Performance Measurements tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through the managing performance measurements tentative category since it establishes an environment where the focal firm together with all SC members are made aware of their current performance and the gaps that need to be addressed. Such a performance measurement scenario enables that the focal firm, together with all the SC members, with the deployment of the right leadership approach, will be able to collaborate and align all operations to meet all customers' requirements and promote a continuous improvement approach. It cannot be excluded that ICT creates an information sharing platform of all the measurements on both the employees' performance and on the automated production processes, to enable the necessary tracking of works and at the same time keeps the employees informed so as to promote improvements.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing performance measurements conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences such as performance improvements, holistic

competitive edge, improved customer service, continuous improvement and effective and efficient performance. The moderating variables are information sharing, through the deployment of technology, both in ICT and automation and an effective leadership style, to motivate the workforce at all management levels. The relationship of the managing performance measurements tentative category with all other categories is represented by the conceptual framework in Figure A11.10.



Figure A11.10: Managing Performance Measurements tentative category (Author)

A11.11 Planning and forecasting tentative category

The **planning and forecasting** theme is about establishing the necessary plans of all orders to be manufactured, based on customers' forecasted requirements or orders, to procure all the necessary raw materials, with the collaborative approach of all suppliers, to meet all scheduled deliveries in a timely manner.

To achieve SCI, the focal firm needs to undertake a diligent planning and forecasting approach to meet all orders effectively by establishing all the **inventory requirements**. The inventory management approach needs to maintain the **lowest levels of stock** to promote **lean** management and an effective **cash-flow** management. Furthermore, cash-flow management is enhanced by using a consignment stock approach with a number of suppliers. It cannot be excluded that SC planning needs to adopt a **level of flexibility**, since operational plans may need to be shifted due to lack of stock in inventory but may still meet the customers' agreed delivery dates. Such a planning and forecasting approach has been described by two participants, by stating that:

"... schedules and plans [are issued] and materials are purchased and shift plans if we cannot meet the targets due to lack of material"; "... we may shift the plan and accept the order late delivery and still meet the final customer targets" (2/INT3); and "The SC planning needs careful and detailed planning"; "... agreements with the clients for a consignment stock and also with the suppliers, to balance out cash flow" (5/INT6). The forecasting process is used to determine the customers' orders, as part of the planning process. The inventory levels are based on such **forecasts**, which are information figures, with a margin of error, since these are gathered from **customers' trends** or **orders**, in-line with the **pull SC** principle. Two participants outlined such a planning and forecasting process, by stating that: '... the plan is based on the market and statistics plans trends' (2/INT3); 'At least you have an estimate of the forecast' (4/INT 5); and 'The customer gives a forecast of the units needed' (5/INT 6).

The SC planning process is concerned with the implementation of all orders in a systematic approach in the order of the submitted requests, but the planned priorities may shift due to the customer type, as one participant outlined, by stating that: *'Production planner assigns the priorities or proceed according to FIFO, but we are flexible to the job order sequence, ... due type of customer'* (3/INT4).

Trust between SC members enables the focal firm to capture real **information**, based on the actual requirements of all members, with no inflated figures, not to create the negative results associated with the bull-whip effect, as described by a participant, by stating that: *Once the forecast is reliable and show real values'; and '... some of the requests may have a hidden agenda such as exploded needs and urgency'* (3/INT4).

Recession periods in the economy, depending on the product type, has a domino effect on both **availability of supplies** and **demand forecasts**, which may cause **SC risks** to increase, unless planning includes contingency measures. The product type determines the demand variation since not all products are affected different by recessions. To **manage such risks effectively**, the focal firm needs to deploy a planned buffer level of stock and that of finished products, whenever possible, to make-up for such uncertainties of supply and demand situations and erratic customers' orders. Such a situation of uncertainty was explained by three participants, by stating that:

'Forecast was made as usual and due to recession, the forecasts being requested were not met by the key suppliers since they gave us very long lead times'; 'This is an innovative line of activity and the sales are constant independent of the market recession' (3/INT 4); '.... we need to deal with production, to meet this excess units of 75 this week' (5/INT6); and 'Our market did not suffer due to recession may be due to type of product' (6/INT7).

SC planning is function of **logistics management**, since different **lead-times** exist due to the nature of the global SC, and as a result such transportation durations differences need to be managed accordingly to get all deliveries in time, as one participant outlined that: 'USA we use air freight and this is 1/3 of our sales otherwise 6 weeks by sea' (2/INT3).

The deployment of technology, through ICT applications, promotes an effective tool in the planning and forecasting process to promote SC visibility to procure all the inventory requirements in time, as two participants explained, by stating that: *'Sales customer system also exist, which gives a delivery date based on forecasts instead of guessing ... to have a clear visibility ... [to] achieve the targets' (2/INT3); and '... so that we supply to them such products... It is all done based on ERPs software' (6/INT7).*

Logic Diagram of planning and forecasting tentative category

Table A11.11 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Planning and forecasting	Positive extreme dimensional	Negative extreme dimensional	
tentative category properties	variation	variation	
SC planning and forecasting	Adjust the operational planned	Adhere to a rigid operational	
flexibility on raw materials and	schedules to have buffer levels to	schedules based on forecasts and	
finished products requirements	cater for to unforeseen conditions	orders	
	(e.g. recessions)		
Collaborative planning with	Sharing of all scheduled plans and	Every SC member issues schedules	
suppliers	collaborating with all relevant SC	with the delivery of all stock to all	
	members to promote high	orders function of individual	
	inventory turnover and better cash	supplier	
	flow (e.g. consignment stock)		
Flexible customer priority	Customer type priority	No priority at all	
adjustment			
Exchange of information forecasts	Real information exchanged	Inflated information is exchanged	
between SC members	between all SC members based on	between all SC members due to	
	a trustful relationship	lack of a trustful relationship	
Lead time values	Short lead times with air freight	Long lead times with sea freight	
Conditions	Level of trust among SC members; collaborative commitments between		
	SC members; economic recessions; logistics management across the		
	globe; the availability of an information sharing platform across the SC		
	by ICT applications.		

Table A11.11: Planning and forecasting tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through an effective planning and forecasting approach with the collaborative effort of all SC members. Such a planning and forecasting approach is mainly focused on effective inventory management by having less capital tied in stock, with lowest level of stock, to promote both lean operations and cash-flow management. Such inventory levels are based on forecasts, with a margin of error, since these are gathered from the customers' trends, in-line with the pull SC principle. Trust between SC members ensures the exchange of real information, with no inflated requirements, to avoid problems of excess stock, associated with the SC bullwhip effect. Furthermore, trust may result in dedicated arrangements, such as consignment stock by some suppliers. Recession periods in the economy has a domino effect on both supplies availability and demand forecasts, which may cause SC risks to increase, unless contingency measures are taken, by having a buffer level of stock against stock-out situations. Planning needs to employ effective logistics, since different lead-times exist, due to the global nature of supplies, having different transportation modes and durations, function of each origin or destination, to ensure no delays in the manufacturing process or in the finished product delivery process. With the current global SC developments, for a firm to remain competitive, it has become a need, whenever possible, to shift purchasing from Asia, which needs a much longer lead times than EU sources. ICT applications are used to have an information sharing platform between all SC members to promote SC visibility across the SC.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The planning and forecasting conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promote several consequences, such as effective inventory management, lean operations, cash-flow management, SC visibility and flexibility, customer service and risk management. The moderating variables are trust, logistics management, lean management and the deployment of technology through the use of ICT applications. The relationship of the planning and forecasting tentative category with all other categories is represented by the conceptual framework in Figure A11.11.



Figure A11.11: Planning and forecasting tentative category (Author)

A11.12 Managing Risks tentative category

The managing risks theme is about taking all the necessary measures in quality, sustainability and legal requirements, within the manufacturing and SC operations and also by monitoring and auditing operational activities, to establish any shortcomings so as to remedy at the earliest, without affecting the customer service quality.

To achieve SCI, **risk management** shall be focused on the **monitoring** of the overall SC with its manufacturing process activities, to remedy all shortcomings in a timely manner, by deploying the necessary contingency measures, so as deliver a **competitive service**. Such a risk management approach needs to effectively manage all the necessary **priorities**, ensures the right **inventory** levels and respects the **quality standards** and **sustainable** requirements. With such process requirements in place, the focal firm also needs to effectively manage the **lead times** of all orders against the effective management of **cash-flow**, not to order excess stock at the expense of liquidity problems, so as to deliver all **orders in time**, with the right **customer service**, so as to promote **business continuity** for a **win-win outcome** for all SC members. Such a risk management approach, has been clearly identified by three participants, by stating that:

'... to manage risk of failure in the supply of the raw materials' (1/INT 1); 'Priority is carefully monitored and to cater for shortcomings'; 'Risk management is important since all operations are being monitored to deliver'; 'The stock needs to be made available without ending up into cash flow problem'; 'Environment and CSR, we have to separate waste, and ISO 14000 is an environment standard, needs traceability on client and supplier waste management processes' (5/INT6); 'If we are in delay, a mechanism is triggered, to decide [on the actions to be taken] ...' (6/INT8); and 'Today quality systems are urging to undergo risk assessment'; 'The SC will [need to] identify the potential risks' (13TS/INT 10).

Inventory levels shall be according to the **forecasts** and **orders**, with due attention to forecasting errors, especially in **recession periods**. Such recession periods may drive the firm to take contingency measures by pre-booking the planned stock required, to guarantee its availability, and to effectively

manage the risk of stock-out and the inability of suppliers to meet all orders. Such a planning approach, to guarantee stock availability, has been explained by two participants, by stating that:

'We pre-book, even if there is no need of the material, not to risk losing the orders. This is against the just-in-time and pull system and we have to use just in case due to such worst conditions' (2/INT 3); and '... lack of delivery, mainly attributed to lack of supplies due the recession matter' (3/INT4).

Stock may become obsolete, which will drive the focal firm to use the grey market or else undertakes an **innovate** approach in its design, so as to use the new available stock in the market. Such a lack of stock situation measures is outlined by a participant, by stating that *'Grey market is used, firms, ...it is a market of second hand suppliers...'; '... we try to phase out such component, by redesigning new products or part of the product' (2/INT3).*

The inventory is **audited** to adjust the stock levels records against any misuse by the focal firm, to promote assurance that every individual component is available to minimise the risk of stock-out with the resultant negative effect on production failure. Such an audit approach, has been described by a participant, by stating that: '... every product has a number of components to complete and hence any missing one, will be a bottleneck' (3/INT4).

Effective supply management is needed, through due diligence actions, to manage all risks associated with all engagements with suppliers. A **multiple supplier** arrangement is preferred over a single supplier approach, to guarantee supplies and to enhance the **business continuity** against stock-outs. Though, it cannot be excluded that some components may need to be locked with a single supplier, due to the nature of the products. Furthermore, some suppliers may not meet the focal firm requirements due to the low order volumes, which drive the focal firm to order all supplies from a single supplier, to meet the imposed minimum order levels, with the resultant effect of having a substantial level excess of stock volumes and with the associated risks of locking to a single supplier. Such a challenging supply management approach, has been described by four participants, by stating that:

'... to find new alternative sources of parts due to single source, i.e. [avoid to put] all the eggs in one basket' (2/INT 3); '...the production will be at risk due to no stock ... [being] locked to a single supplier'; 'The supplier force a minimum order quantity per item' (3/INT4); and '.... may be a bottleneck in case this supplier fails and better to have two or more suppliers' (5/INT 6); and 'We do due diligence ... we validate a supplier, before we enter in any agreement'; 'We revisit the procedures and systems and measure the level of risks, ... one need to be practical and do not overdo it, to avoid analysis by paralysis' (1TS/INT 5).

An arm's length collaborative effort, may introduce risks associated with opportunism and lack of engagement, feedback and cooperation respectively, both within and outside the focal firm. Such a relationship may drive the focal firm to undergo duplicated activities and to take special measures of precautions, to make up for such shortcomings, when they occur, not to influence the overall SC performance, as referred clearly by two participants, by stating that:

'The [intermediary in the downstream side of the SC] is not meeting the expectation of the [corporate expected sales] and the initiatives are being pushed by [the corporate].... This is inhibiting the business model' (1/INT1); 'There is always room for improvement and there is always rivalry at work between departments'; 'We still have internally silos thinking within departments' (6/INT7); and 'The various stakeholders are after the same criteria but these may be in conflict with their individual targets/criteria' (6/INT8).

Malta, as a Small Island State, with its insularity characteristics has a relatively expensive logistics service with its infrequent cargo shipping schedules, since it needs to use either sea or air, as the first or last shipment mode to deliver or bring the products to or from the nearest mainland respectively.

Furthermore, due to Malta's small country size, various plants are ending up in using disparate sites to make up for the required plant size. Managing logistics is very challenging and needs to respect both the **lean management** approach and the **environmental responsibilities** by deploying **effective and efficient logistics management**. With such an approach a solid foundation based on a healthy **social and environmental responsibility** is built not to put to risk the image of the firm with any of its stakeholders. Additionally, it is a must to minimise the expenses associated with all logistics and inventory levels in stores, to manage all costs effectively but without incurring the risks of stock-out Such a logistics management challenging scenario was explained by four participants, by stating that:

'Logistics is the main problem in Malta'; 'The Sea shipping routes are quite limited and are expensive'; 'Air travel is used for fast moving products and is expensive'; 'Consideration is being given by freight forwarders for route planning in line with the roadmap 2050 for low carbon Europe' (2/INT3); 'The firms have problems due to physical space limitations of the building layout, logistics problems due to movement of products, say half an hour away of the buildings. Malta has limited space of land and hence most factory units are located apart' (4/INT5) 'This firm has heavy stock/material and we do not consider airfreight and courier'; 'From Germany we have 5 days transit time. In case for the electronics SC, one may use courier to win the lead times'; We are using more the Asian market, with 28 days transit time' (5/INT6); and 'Results also show that, we made vast reduction with the [freight forwarder contractor] on the inventory terms'; and '.... [we] put more products in the shipment to cut the costs' (6/INT 7).

ICT tools integrate all sources of information and remain an enabler of all processes, by undergoing **information management**, to promote information sharing within and across the SC and by having access to all information updates on all activities that promotes **SC visibility**, so that the focal firm is always **informed** in real time on any shortcomings by managing risks effectively through **immediate recovery actions**. The deployment of ICT to assist the managing risk approach, has been explained by two participants, by stating that:

'The IT integrates ... [gives] all the prompts to manage effectively'. '.... the correct data is important otherwise the production will be at risk due to no stock' (3/INT4); and 'SC transparency/visibility is important along the whole SC, so that the suppliers, contractors and clients can detect all the flow such as shipping problems'; 'The workflow thanks to technology has a complete picture with all the details ... [of the overall process performance]' (4/INT 5).

The **technology deployment** in **production processes** improves the manufacturing capabilities in all activities, in line with **lean management**, by reducing the human intervention, which is more prone to errors and to deploy automated based manufacturing to promote higher quality, efficiency and reliability, to minimise any risk of failure. The use of automation to manage various risks associated with the manufacturing process, has been clearly outlined by two participants, by stating that:

'The technology investment in robotics etc, then the cheap labour is challenged since it is still human'; 'With full or semi automation machines (robotics arms etc) the cheap labour is won over, since one has 24/7 shifts and high quality, since human introduce errors which are avoided' (4/INT5); and 'The production efficiency is higher, around 3 times or less, due to automation ...'; '... [automation] promotes better/higher reliability from a quality perspective' (6/INT8).

Logic Diagram of managing risks tentative category

Table A11.12 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Risks tentative	Positive extreme dimensional	Negative extreme dimensional
category properties	variation	variation
Monitoring for possible	Full monitoring and auditing (e.g.	No monitoring at all
shortcomings in supplies	actions against stock-outs through	
	a multiple supplier and effective	
	logistics)	
Remedial actions	Immediate/real-time remedial	Reactive actions
	actions (e.g. delays in supplies	
	deliveries or obsolete stock)	
Assigning Priority	Settlement of high priority issues	No attention to priority at all
Proactive assurance actions	Verification approach (e.g. auditing	No verification at all
	and contingency actions such as	
	multiple supplier engagement)	
Compliance to all quality	Compliance to all requirements	Lack of compliance to all
standards, sustainable and legal		requirements
requirements		
Conditions	Technology deployment in ICT and automation; recession periods effects;	
	suppliers power to enforce minimum limits of supplies; stipulated quality	
	standards together with all environmental, social and legal requirements;	
	and logistics challenges associated with a Small Island State.	

Table A11.12: Managing Risks tentative category (Author)

Conceptual summary on the tentative category

SCI is enabled through managing risks effectively and efficiently, since the focal firm and all SC members operations are safeguarded against failures or bottlenecks by taking all the necessary proactive or contingency measures, through a collective approach, to promote business continuity. Shortcomings may vary since they originate from both internal and external sources. ICT tools integrate all sources of information and enable all processes by providing access to all information updates for the focal firm to deploy an effective risk management approach through immediate recovery actions. Automated activities within manufacturing also promote higher quality, efficiency and reliability to minimise any risk of failure.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The managing risks conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as SC flexibility and visibility, competitive service, win-win outcome, information management, business continuity, customer service, quality, HES and legal compliance, efficiency and effectiveness, business sustainability, business continuity and competitive operations. The moderating variables are lean management, cash-flow management, customer service, performance measurement, auditing of operations, quality standards with environmental, economical and social practices and legal responsibilities, forecasting of inventory requirements, supply management, inventory forecasting and planning, information sharing and technology investment in ICT applications and automation. The relationship of the managing risks tentative category with all other categories is represented by the conceptual framework in Figure A11.12.



Figure A11.12: Managing risks tentative category (Author)

A11.13 Supply Management tentative category

The supply management theme is about taking all the necessary collaborative measures with all suppliers to procure the necessary stock in time, with the right quality and with a competitive price, by managing stock and cash flow effectively. The suppliers need to implement all the necessary sustainable measures in line with the focal firm processes.

Vertical integration is deployed by the focal firm to increase the ownership in the upstream side of the SC to manufacture products in-house so as to promote a lean management approach with less cost attributed to stock held in inventory and with the result of improved cash flow management and logistics management respectively. Such a vertical integrated process, has been described by one participant, by stating that: *'Shift equipment to Malta, to do the components in Malta and not order to store which was leading to idle stock'* (1/INT 2).

The focal firm undertakes innovative initiatives to design new products or else make use of new components with the same products' design, to facilitate supplies availability. Such an innovative approach, has been described by a participant, by stating that: *'The SC may hold the launch of the*

product and if such components are not available then the design may be changed according to the market available components' (2/INT 3).

Improved supplies' delivery is achieved through better collaborative effort in supply management, by the deployment of a consignment stock approach. Such a collaborative approach with suppliers promotes better flexibility to meet new orders or to manage any unforeseen changes in stock requirements, due to the available stock in hand and at the same time manage cash flow effectively, by having less stock tied in capital, considering the relatively high raw materials costs. Such a collaborative scenario, has been described by four participants, by stating that:

'The consignment stock refers to stock supplied before the order is made'; '... consignment agreement to pay until it is used ... this will cater for stock availability in the firm' (5/INT6); '...consignment stock ... assists the cash flow financially' (6/INT8); and 'The business strategy is focused on effective management of materials since 90% of the costs are the material and the rest is labour costs' (2TS/INT3); and '50 % is material [costs]' (1TS/INT5).

Risk management through a multiple supplier arrangement is employed to eliminate the risk of stockout, associated with single supplier situations, due to no alternative sourcing possibility. Such a risk management approach is also achieved through an ongoing stock-take of inventory, to guarantee the required levels of stock for each stock code, since a single missing component may lead to SC failure (e.g. kitting errors in stock issue). Supply management is concerned about timely supplies' deliveries. Such deliveries are monitored to identify at the earliest any possible bottlenecks, by taking all the necessary timely measures to promote business continuity. Such a risk management approach, has been described by four participants, by stating that:

'The latter suppliers are available in more than one, to manage risk of failure in the supply' (1/INT1); '... focus more on strategic issues, say to find new alternative sources of parts due to single source i.e. all the eggs in one basket [situation needs to be avoided]' (2/INT3); '... components are based on weight and may vary [in stock issues]...'; '... the production will be at risk due to no stock and delays from suppliers and also components that are locked to a single supplier. Both the two issues are crucial to the SC operations'; '... every product has a number of components to complete and hence any missing one, will be a bottleneck' (3/INT4); and 'Familiarity breeds contempt, you get the quote for every year budget and you spread the orders among these key suppliers' (13TS/INT 10).

Forecasts are established to guide the inventory levels needed so as to undertake effective supply management in a timely manner to respect the lead time of all orders. With the right inventory requirements in hand, the focal firm is in a position to meet the scheduled manufacturing capacity, in line with the customers' contractual agreements. Though, such a forecasting process may be in error, due to inflated values, given by different downstream SC members, which may drive the firm to end up into over stock situations. Such forecasts represent real values once the SC members employ a certain level of trust. Such an effective supply management becomes more critical in recession periods since stock needs to be pre-booked and procured ahead before time, to guarantee stock availability. Such an approach is against the JIT and pull-SC system, but needs to be deployed to make up for the negative effects brought about by the recession periods so as to act as a contingency measure against stock-out. Such a challenging approach, has been described by three participants, by stating that:

'Two years ago the situation was the opposite situation, since stock was available and just-intime was used'; 'We pre-book, even if there is no need of the material, not to risk losing the orders [in recession periods]' (2/INT3); 'Once the forecast is reliable and show real values'; 'Forecast was made as usual and due to recession the forecasts being requested were not met by the key suppliers (distributors) since they gave us very long lead times' (3/INT4); and 'When a client engages with a manufacturer, both engage in a trustful environment' (4/INT 5). Auditing suppliers is performed both during the suppliers' selection screening process and also in their performance, to ensure that they meet the focal firm requirements, such as in quality, price and lead time criteria. Additionally, control measures are put in place on such suppliers to detect at the earliest any defective products or shortcomings in the delivery performance. The supplies are normally procured from distributors since they are preferred over leading manufacturers sources, since such distributors buy in bulk from such leading manufacturers and at a better price and as a result can supply such components at a more attractive price than the manufacturers themselves. Low order volumes are a key problem to get prospective suppliers engaged within the SC. In fact, effective supply management becomes more critical to achieve if the relatively low volumes required are divided among different suppliers, to deploy a multiple supplier approach, so as to minimise the risk of stock-out, since the orders for each supplier will be much smaller than the normal order levels. Some suppliers may even force a low minimum order level to supply stock, causing excess stock in inventory, low stock turnover and also introduce cash flow problems, due to tied up capital in stock. Such minimum order levels problems and other challenging criteria in supply management, are being encountered from Eastern markets suppliers, since today's trends are to order from such markets, to benefit from attractive price offers to remain competitive. The challenges associated with the managing of supplies, has been outlined by five participants, by stating that:

'Selection of suppliers by audit suppliers with respect to operations, management of inventory of suppliers, the lead times to match our lead time' (1/INT1); 'We buy directly from the supplier (such as Arrow, Abacus) and could be a distributor himself, who in turn they buy from leading manufacturers such as Toshiba, ST etc. These suppliers source the items and reduce price due to the bulk buying principle' (2/INT3); '.... Another supplier may not meet our requirements, due to the low volumes of orders'; 'The supplier force a minimum order quantity per item, this leads to a problem of over stock' (3/INT4); and '... the choice of component will have certain standards and makes a difference to meet the customer in time or not, due to the standards needed' (5/INT6); and 'The cost of supplies needs to cater low cost sources, such as China. Such a stance need special measures which are different from other well established sources of suppliers [such as] ... time, communication problems, material quality, minimum order quantities, lead time variation is greater, credit terms etc' (5TS/INT2).

Local suppliers are preferred over overseas suppliers, whenever they meet the focal firm requirements, since improvement occurs in both lead time and cash flow management respectively, especially for Malta being a sea locked island, as indicated by two participants, by stating that: '*Lead time is eliminated with the use of local manufacture 3rd parties, which are excellent in their operations*'. '... [firm] transferred most of the operations... in Malta, to improve on the lead time' (1/INT 1); and 'Due to the logistics lead time, abroad is 1 day, but here we need 1 week to 1^{1/2} week' (10TS/INT7).

All SC members need to comply with all sustainable measures within both SC and manufacturing operations, to promote uniform processes and improve on the efficiencies in line with lean operations. Such sustainable processes, has been referred by two participants, by stating that:

'[These are the] standards that need to be met'. 'UL needs investment for the approval by implementing these standards, such as raw material specs used, plastic type use'; 'All components need to be RoHS directive for Environmental, segregation of waste, minimize waste' (3/INT4); and 'EU rules, policies and standards are key,... due to standards compliance'; and 'Environment and CSR ... needs traceability on client and supplier waste management processes' (5/INT6).

The use of ICT is pivotal in stock management and its procurement and in all real-time information updates, by integrating the focal firm with all its SC members, so as to manage all stock effectively and efficiently and also ensures that all parties are timely informed in all stock movement and in any deviations from the scheduled plans. Such ICT applications, includes both office and mobile

applications, to ensure that SC visibility is available not only from the workplace but even on the move. The use of technology, has been described by three participants, by stating that:

'The order trend, with suppliers is integrated over internet'; 'The client has access to suppliers, the stock available and the customer is aware of the order and the shipping details...'; 'This monitoring is also done today by smart phones, where managers are mobile and can see the orders, make new orders and raise issues. Thanks to these tools these are pushing more integration across the SC' (4/INT 5); 'Software is being used that manages information per component' (5/INT 6); and 'The effective management of stock uses a common order system for all orders, based on a common database (e.g. ERP) so that all orders are included to promote flexibility to switch from one product to another'; 'This process is very dynamic since such information ... is continuously changing '(2TS/INT3).

Logic Diagram of the supply management tentative category

Table A11.13 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Supply management category	Positive extreme dimensional	Negative extreme dimensional
properties	variation	variation
Supply management set-up	Multiplier supplier for effective	Single supplier arrangement with
	inventory management with vertical integration	risk of stock-outs
Lead time of supplies deliveries	In-time deliveries	Delayed deliveries
Orders' forecasts	Engage in trustful relationship with	Inflated order requirements due to
	all stakeholders to establish the real order requirements	lack of trust
Collaborative approach in supplies	Consignment stock approach	Forced minimum orders of stock
Innovative approach to use	Availability of supplies is a priority	Obsolete or unavailable supplies to
available market components/raw materials	in the products design	meet the products design
SC visibility in stock movement	Information sharing and real-time	Lack of information visibility both
and orders' deliveries	updates within and outside the focal firm	within and outside the focal firm
Conditions	Recession periods; level of trust within the focal firm staff and also	
	between the focal firm and all SC members; innovative products designs; and technology deployment in ICT applications.	

Table A11.13: Supply management tentative category (Author)

Conceptual summary on the tentative category

To achieve SCI the focal firm must select and manage a small suppliers' base, to promote close collaborative initiatives, to be supplied with the **necessary stock** within the scheduled **lead time** and with a **win-win objective** for the two parties, so as to meet all **customers' requirements** effectively. A firm may undergo **vertical** integration, to perform in-house certain operations, so as to improve on its supply management, with improved lean operations, logistics management of all deliveries and cash flow management respectively. Inventory levels required are to be based on **forecasts** or **actual orders** to promote a **JIT** approach and effective cash flow, in line with **lean management**. In recession periods, supply management may need special measures to guarantee stock availability, by implementing **pre-booking** and/or **purchase-to-stock**. Attention shall be given to the customers' forecasts, since the information does not always represent the **real requirements**, unless a trustful relationship exists, but carry inflated figures. A supplier, apart from the **contractual agreements**, may also undergo **collaborative initiatives** with the buyer, to promote improved supply management, by adopting a VMI approach. Suppliers need to be audited, both initially as part of the selection
screening process and also in their operations, to ensure compliance to all focal firm requirements, such as in quality and in all sustainable processes. **Knowledge sharing** is to be exchanged between suppliers and the focal firm on the materials' **quality, specifications and price,** to assist in the right selection of materials. **ICT** assists in the **information sharing** of stock levels across the SC, to promote **flexibility** and **changes** in supply and inventory management in real-time, to ensure no risks associated with stock-outs situations so as to promote business continuity. Stock which becomes obsolete requires **innovative changes** to the product design to use other stock or else procure stock from other alternative sources.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The **supply management** conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as win-win outcome, effective customer service, SC flexibility and visibility, innovative products design, lean operations, business continuity, performance effectiveness and efficiency, and sustainable operations. The moderating variables are lean management, contractual agreements, collaboration, auditing suppliers, knowledge sharing, logistics management, quality and sustainable processes, vertical integration, cash flow management, risk management, and technology deployment in ICT applications. The relationship of the supply management tentative category with all other categories is represented by the conceptual framework in Figure A11.13.



Figure A11.13: Managing the supply management tentative category (Author)

A11.14 Managing Trust tentative category

The managing trust theme determines the type of collaborative relationships between people and organisations across the SC. With a trustful environment, effective information and knowledge sharing takes place, within and across the SC to promote a level of teamwork with all stakeholders.

Trust is function of the relationship build-up between the focal firm and all SC members. Within the focal firm, together with the Corporate, there are a set of common values and objectives which promote a high level of trust between all people to be empowered so as to improve on their performance. The focal firm with all external SC members, need to build a relationship, depending on the nature of operations engaged with the focal firm, although every SC member may have different values, roles and objectives due to globalised nature of the SC. The level of trust varies since a SC member may be a partner or just a one-off supplier of a product or a service. The focal firm needs to build such a trustful environment by auditing all suppliers to establish all work practices so as to ensure common and unified commitment, in line with the overall SC objectives and also to establish any deviations from the scheduled performance to remedy at the earliest and minimise any risk of failure. Such a trustful environment, has been described by four participants, by stating that:

'For inter-company the performance is more easy to be dealt with due to same values, culture and language'; 'But with local suppliers and others external European suppliers to firm, requires the validation and audit parameters/criteria as established with the corporate KPIs and its suppliers'; 'Relationship between US [Corporate] and Malta ... trust between both exist' (1/INT 1);. 'SCI is in depth since the outsourced activities involves part of the core processes'; 'All obstacles are being monitored to ensure such smooth delivery, using risk management approach' (3/INT 4); 'We have suppliers of the compressors, as the core activity ... these suppliers are partners and have an in-depth relationship, based on years of experience' (5/INT6); and 'We have a good relationship with the customers and hence they trust us'; 'The trust is key since we work together and we use empowerment to take decisions' (6/INT 7).

Communication is needed within and outside the focal firm to promote trust building among all people. With trusted SC members, the exchanged information would represent the actual requirements, with no inflated figures, so as to minimise any effect associated with the bull-whip effect. Contractual agreements shall be in place to formalise the SC operations with all SC members, but what counts is the trust between all people to create an open communication environment, with both a formal and an informal organisation approach, to meet all the needed trade-offs for the common good of the overall SC so as to promote effective and efficient performance. Such an open communication based environment, has been described by four participants, by stating that:

'Communication is a priority ... to promote more trust in between the staff' (1/INT 2); 'There are head of sales that communicates with us, some of the requests may have a hidden agenda such as exploded needs and urgency, but one need to achieve a balance to keep the relationship live and kicking'; '....build the trust and collaboration' (3/INT4); 'When a client engages with a manufacturer, both engage in a trustful environment which promotes open communication apart from the formal contracts'; 'This leads to improved operations and the more access given, the more is the trust build-up' (4/INT 5); and 'For example, if a purchasing manager feedback is not trusted, then I will include safety factors in my orders, which may not be needed. So with trust between us, then the orders done are trusted and are kept as given' (6/INT 8).

Intellectual property (IP) protection needs to be established through formal agreements, though such protection assurance depends on the level of trust with the respective SC members. It is not being excluded that a certain level of disclosure occurs independent of the signed non-disclosure agreements. Such a trustful approach, has been described by a participant, by stating that:

'The trust is needed since we share confidential information [with suppliers] and they outline the current technology used. We sign an NDA, non-disclosure agreement, so that the supplier cannot give drawings associated with us and use it to other interests'; and 'Trust exists but leakages are not being excluded' (3/INT 4).

Trust builds with time and engagement, with both internal and external stakeholders. Suppliers may engage with better credit terms and also adopt a consignment stock approach with the focal firm, based on the established trust and close relationships between parties, since '*business is based on trust and not on money*' (9TS/INT 6). Such a trustful relationship, has been described by three participants, by stating that:

'You need to pay before you purchase from such suppliers and as time goes by, the [relationship] depth increase, credit terms improve and trust builds' (5/INT 6); '....we negotiate the material procurement contracts and we use consignment stock' (6/INT7); and 'That the key positions and heads ... are the same people over a period of time, so as to build teamwork, based on experience and trust and the data integrity fed is trusted in the details or contents' (6/INT 8).

Technology investment in ICT applications is pivotal within and across the SC, to promote real-time information sharing, but the trust build-up takes a priority over ICT deployment, since trust between people and the SC members is the initial stepping stone that drives the collaborative environment. The role of ICT remains an important tool to enable and may also drive such a collaborative relationship. The importance of trust over ICT applications, has been described by three participants, by stating that:

'[Factors that] Promote integration, above the information systems, is to build the trust and collaboration' (3/INT4); 'Technology is a big enabler in this case and has contributed to large increase in efficiencies' (4/INT5); and 'The IT is also driving the process of the SC, ... to place the order or to see the orders' (5/INT6).

Logic Diagram of managing the supply management tentative category

Table A11.14 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing Trust category	Positive extreme dimensional	Negative extreme dimensional
properties	variation	variation
Role of SC member vis a vis the	Partner with focal firm	One-off supplier of products or
focal firm		service
Relationship depth between focal	Close relationship	Arm's length relationship
firm and SC members		
Level of trust	Empowered staff of SC members	SC members provide basic services
	to meet focal firm requirements	as required by the focal firm
Monitoring SC members	Little monitoring at all due to high	Full monitoring to ensure meeting
	level of trust in the relationship	the expected performance due to
		arm's length relationship
SC members' behaviour	Joint commitment and teamwork	Opportunistic approach taken by a
	approach with the focal firm with	SC member to the focal firm, such
	real information exchange	as inflated requirements
Conditions	Intellectual property rights protection; contractual agreements; and	
	different roles and values of SC members (e.g. partners or suppliers)	

Conceptual summary on the tentative category

To achieve SCI the focal firm needs to consider that trust remains a key determinant of all collaborative initiatives for relationship building, especially between the focal firm and upstream SC members, with the deployment of a consignment stock arrangement. Trust builds with time of engagement, communication and information sharing between all SC members, so as to tackle all challenges from a **holistic perspective** through a **cross-cultural knowledge** approach for a better working relationship. Such an approach needs to reach a level of flexibility, understanding and joint problem-solving to promote **teamwork** among all SC members, with all the necessary measures to be taken by all entities for the protection of intellectual property rights. Trust enables sharing of real information and minimises opportunistic requirements from any of the SC stakeholders, to minimise the negative effects associated with the bullwhip effect, by exchanging the real orders' requirements, to promotes effective and efficient performance. For multi-national firms, the intercompany trust is more easily achieved, due to the shared strategic objectives, values and culture, but with different SC members, such a trust needs to be built, due to the different strategic objectives, values and cultures of each SC member across the globe. Technology deployment is needed to facilitate and drive all information based processes, but trust among people remains the initial stepping stone before the role of technology becomes effective. Auditing is performed on both the focal firm and externally on all SC members, to build on the trustful relationship by establishing all conformities and especially divergences from the established contractual agreements and scheduled performance, to remedy at the earliest, so as to strengthen the relationships and at the same time minimise any associated risk of failure to promote business continuity.

Tentative relationship between all emerged potential categories in line with the coding paradigm

The **managing trust** conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as effective and efficient performance and business continuity. The moderating variables are collaboration, communication, information sharing, knowledge sharing, SC flexibility, technology deployment in ICT, contractual agreements and risk management. The relationship of managing trust tentative category with all other categories is represented by the conceptual framework in Figure A11.14.



Figure A11.14: Managing trust tentative category (Author)

A11.15 Managing Culture tentative category

The managing culture theme is concerned on the management approach that the focal firm needs to adopt, to create a working attitude that promotes understanding and flexible collaborative efforts between all stakeholders, to achieve a more integrated approach with effective and efficient performance.

The cultural aspect needs to be future looking which shall also include all contingency measures so as to promote efficiencies and resilience in all operations, as one participant outlined, by stating that: 'We use a culture to cater for the future and undergo cost savings (13% reduction say) to make up for future situations that are negative such as recessions. All this promote resilience' (1/INT 2).

Training is needed to meet the innovative changes to be introduced within the focal firm, so as to overcome the challenges offered by the ingrained staff cultural values, associated with the retention of the previous established procedures and legacy systems. Culture differs between different SC members due to the globalised nature of their origins, having different cultural values, which are function of the country of origin and the respective firms' organisational values. SCI requires the focal firm to overcome such cross-cultural differences through learning and with an effective management and leadership approach to promote understanding and flexibility with optimised and innovative processes. Such a need for a training and learning culture, was referred by four participants, by stating that:

'Common values, validation and parameters are used although we are different from other [EU] countries' (1/INT1); 'The USA culture differs from the EU countries in the way of collaboration, since they show a level of superiority and that they take a one-sided view of the communication. The Asian market expects details and full support on a continuous basis. The EU countries are more understandable and flexible' (3/INT 4); and 'One needs to de-train and re-train, to ensure that new processes are being used, due to the previous culture. Not to use the old systems again on the new processes'; 'Other firms have the same product but use different processes due to different management styles and culture' (4/INT5); and 'Maltese workforce is very skilled'; 'When you deal with Japanese and Chinese it is a different story when compared to Malta' (9TS/INT6).

The focal firm needs to keep an open eye for the cultural values of each SC member, since the country of origin has a direct influence on its operations and commitment to all established schedules. This is attributed to the different calendar activities, since any changes in the scheduled performance, due to unforeseen circumstances, may occur during a public holiday of that SC member, which may disrupt the timing schedule of the overall SC, due to the difficulties to take immediate action. Furthermore, more negative cultural values, that may subject bottlenecks in SC operations, may be attributed to SC members that face political instability and corruption. Such different cultural values, were referred by three participants, by stating that:

'Case in point is the Chinese New Year. The supplier had promised to send a number of boards as agreed. We checked for such a delay ... He informed us that being the Chinese New year not even the mail works. This holiday matter is the source for a 15 days delay or more. Israel, Friday, is a holiday but on Sunday, you may communicate with them'; 'All suppliers inform us by giving their calendar to keep us updated [on their holidays]'; 'The customer is not interested on firm lead times but on the delivery of the product in 3 weeks or less' (5/INT6); and '... plant has different work hours per week due to Ramadam, it affects the SC on how we supply them' (6/INT 7); and 'In some areas, where we have no reps, like Eastern Europe, due to heavy corruption. Libya we did not set-up a country manager, due to legal instability and corruption and leads to huge business risk' (13TS/INT10).

The downstream SC members need to be given priority, due to the pull-SC approach, but the customers need to understand that all requests need a lead time dedicated for the manufacturing and

delivery of all orders. Any requests for tighter lead times, requires that customers need to pay for any excess costs allocated for crashing of resources. The downstream SC members' culture is to have all products on demand with zero lead time. Such zero lead time does not represent the current true reality, due to the make-to-order business scenario. Such a customer demanding oriented approach, was referred by a participant, by stating that:

'A customer once had a stock adjustment of 75 large units and the pressure was shifted on us'; 'Say, one of our key customers, expects that when an order is received from his side, we need to give him the unit there and then'; '[Stock adjustment] is a Customer side error, and to meet this, the customer is informed about the lead times and payments needed. This will need special costs to reduce the lead times... since it needs for special air freights to meet such a request' (5/INT 6).

Technology deployment is pivotal in all processes, but the cultural element still plays a crucial role in the introduction and use of new information technology based systems, since employees' culture may offer a certain level of resistance to adapt to changes. Such a resistance to change may drastically obstruct the effective deployment ICT applications to interact with all people, both within and outside the focal firm. Such a cultural element to restrain from the use of technology, was referred by two participants, by stating that: '*At that time this ERP was said to be a 'monster' and most of the workers showed resistance to implement and use'* (1/INT2); and '*Certain cultural barriers are difficult to overcome, since you give them the technology but still they do not use it*' (4/INT5).

Logic Diagram of managing culture tentative category

Table A11.15 outlines the main characteristics of this tentative category in a logic diagram (Strauss and Corbin, 1990).

Managing culture category	Positive extreme dimensional	Negative extreme dimensional
properties	variation	variation
Flexible work practices	Adapt effectively to all changes	Resistance to change
Sensitivity to other SC members'	Dedicated effort to understand each	Rigid approach to others' needs
needs	others' needs	
Oriented approach to business	Long-term oriented (e.g. cost savings	Short-term oriented based on quick
resilience	and training investment)	fix solutions
Change management approach	Innovative approach to change (e.g.	Traditional approach with the
	training investment)	deployment of legacy systems
Conditions	Globalised SC with various SC members across the globe with different	
	cultural values such as different working hours per day; political instability;	
	corruption; and different non-working days across the yearly calendar.	

Table A11.15: Managing culture tentative category (Author)

Conceptual summary on the tentative category

The globalisation of the SC, has created the need that for the focal firm to achieve SCI, both within and outside the focal firm, dedicated measures need to be deployed by the focal firm to manage effectively all the different cultures that characterise all individual stakeholders. Such cultural differences need that the focal firm adopts a flexible behaviour, based on collaborative initiatives and relationship building, so as to be ready to adapt and be sensitive to the different requirements of all stakeholders, by shifting from a local to an international mindset. The culture of any SC member is a set of values and beliefs that determines the way people perform their operations, both within each organisation and how they associate in all interactions with the different stakeholders, both locally and globally. Such a collaborative approach needs to promote a climate of understanding, information sharing and teamwork in all changes, negotiations and interactions, since sometimes the culture of any SC member may be deeply rooted within the people values and is hard to change it (e.g. different selling prices are used for the same products in different countries all over the world even within EU itself; different working times; different understanding approaches; corruption; political instability; and different salaries across the globe). The focal firm needs to build a culture based on sound management values and work practices, through the necessary training, to have an excellent workforce that is ready to adopt the right mindset with all stakeholders and a working attitude that performs effectively and efficiently and at the same time also promotes economic resilience, by taking all the necessary measures for any future economic downturns (e.g. cost cutting approach in all processes).

The globalised SC, has several SC members located in different countries, which is drastically influenced by the each member culture, since apart from the world clock time differences, SC members may have different working times and different holidays and shutdowns along the year, which differ from the local Maltese calendar. Such contextual conditions, influence the SC performance, due to possible bottlenecks in the SC operational time schedules, since this may affect the SC effective collaborative initiatives and logistics management. As a result the focal firm staff needs to adapt to such differences of all suppliers or distributors and other stakeholders, who may have all their services temporarily on hold, due to such non-working hours or days. A lack of understanding and cooperation culture may exist, in the downstream side of the SC for certain manufactured products. Such a stance may be originated by the customers' lack of interest to the lead time requirements of the focal firm to manufacture and deliver the products. Responsibility and pressure may be shifted by the customer on to the focal firm, even for any mistakes originated by the customer, referred as 'stock adjustment'. In such a case the customer, through negotiation with the focal firm, has to make up for the extra costs for a faster response, so that the focal firm is in a position to meet the new order by the necessary dedicated measures (e.g. crash resources and faster delivery mode of transport).

Tentative relationship between all emerged potential categories in line with the coding paradigm

The **managing culture** conceptual element is considered as part of the SCI phenomenon since it is an action category, where with the support of various moderating variables, promotes several consequences, such as sustainable performance, effective and efficient performance, flexibility to adapt to change, optimised and innovative processes. The moderating variables are effective collaborative mindset, sustainable economic measures, effective management and leadership approach and training. The relationship of managing culture tentative category is represented by Figure A11.15.



Figure A11.15: Managing culture tentative category (Author)

Appendix 12: The IMLA supporting strategic actions to achieve competitive capabilities

A12.1 The business strategy

The key scope is to align the business goals at all management levels down to the bottom line to promote unity of actions based on the agreed commitments and a level of collaborative relationship throughout the focal firm together with all stakeholders. Such a strategic approach promotes leverage of the firm core competences with that of the relevant stakeholders, to promote a competitive advantage in all processes from a system wide perspective (e.g. exploit manufacturing and/or SC based processes through outsourcing; sub-contract works which are performed better than the focal firm; and learn from the expertise and build on knowledge from other stakeholders).

A12.2 The Manufacturing Strategy

The key scope is about the 'make' or 'buy' decision, which the focal firm needs to adopt by focussing on its core activities so as to outsource non-essential tasks but still undertake all the necessary collaborative initiatives, so as to exploit each others' resources and capabilities to create value-added operations for an improved and innovative product. The manufacturing strategy needs to employ, apart from the necessary contractual agreements, all the necessary measures to innovate all processes, through cost cutting measures and/or new techniques, on an on-going basis to promote improved manufacturing operations.

A12.3 The Supply Chain Strategy

The key scope is to establish the SC set-up, through either vertical integration (i.e. 'make' manufacturing decision) or virtual integration (i.e. 'buy' manufacturing decision) and to decide on the type of SCI approach to be adopted with all SC actors, depending on the role of each SC actor. The SCI approach may vary from an arm's length to close collaborative approach (i.e. arm's length where information and orders are issued on need to know basis only or else in-depth integration to form a partnership type of approach to work in unison and by adopting dedicated measures, such as VMI).

A12.4 The holistic SCI management approach

The key scope is to integrate all operations both within and outside the focal firm to gain from synergies of each other's performance, so as to achieve a competitive edge. The holistic SCI management approach refers to all the sixteen measures needed to be applied from a holistic perspective, to build a teamwork based environment with all people across the SC, which is enabled by technology, so as to promote visibility of all operations and to take all the necessary proactive actions to ensure that all processes are in line with all quality standards, sustainable objectives and business continuity measures. The sustainable objectives are after the triple bottom line approach, where the overall SC performance considers the economic, environmental and social commitments to promote a sustainable SC from a holistic perspective, which may need to undergo the necessary trade-offs with each SC actor for an overall SC win-win outcome. Such a holistic approach is based on the involvement of all stakeholders in all relevant decisions to promote such a win-win approach for all SC members but with the main priority focused on the common good of the overall SC, so as to build a transparent, responsive and unified SC to achieve competitive capabilities. The sixteen conceptual elements building up the holistic SCI management approach are:

A12.4.1 Auditing Operations

Auditing operations needs to validate all suppliers and contractors together with the focal firm itself, to ensure that all scheduled targets are in place and all materials and processes comply with all quality standards and sustainable measures.

A12.4.2 Supply Management

Supply management needs to ensure that there exist a concentrated number of reliable sources of supplies with a stronger relationship, as partners, to promote all stock deliveries in the right time and at the right price with special dedicated collaborative measures, such as VMI.

A12.4.3 Managing Knowledge

Managing knowledge needs to ensure that the overall SC leverages knowledge from each other core capabilities to create a learning environment based on a holistic approach in view of the end customer to promote value-added and innovative processes and products/services respectively.

A12.4.4 Managing Cash Flow

Managing cash flow needs to promote economical sustainability through the right allocation of stocks, without excess of idle capital in inventory, which may also lead to obsolete stock, but with dedicated provisions of stock to meet unforeseen orders or changes to promote SC flexibility.

A12.4.5 Managing Change/Innovation

Managing change and innovation needs to ensure that the focal firm focuses on change and innovative processes and products development initiatives with the involvement of all SC stakeholders' knowledge, to meet the uncertain business environment with its contextual dynamic conditions (e.g. recessions) to promote 3BL sustainability and produce competitive products/services.

A12.4.6 Managing Collaboration

Managing collaboration needs to ensure, that both within and outside the focal firm, all people contribute to a teamwork organisational climate based on trust to promote synergies across all the SC up to the end customer and also responsive SC (e.g. VMI, CPFR). Such collaborative approaches may vary from an arm's length to a close-collaboration depending on each SC actor role.

A12.4.7 Managing Culture

Managing culture needs to ensure that the focal firm, with all of its stakeholders, understand each other' values and requirements to ensure that together operate in concert and with a shared vision for the common good, by adapting to each other' priorities and needs, to promote a win-win outcome. Such values are enhanced through both formal and informal interactions.

A12.4.8 Managing Customer Service

Managing customer service needs to build on a customer relationship and to adopt a proactive approach to all requirements with customised products/services, so as to promote the right customer service, with all the necessary value-added activities and innovative measures to the end customer.

A12.4.9 Managing Information

Managing information needs to adopt an open dialogue with an effective communication approach consisting from both face to face interactions and electronic based applications, with the necessary depth and frequency of information exchange, to promote SC visibility and an informed mindset with all stakeholders, to promote effective decision-making and an audit trail of all activities.

A12.4.10 Managing Lean Operations

Lean management needs to be adopted to streamline and optimise all SC and manufacturing processes to promote 3BL sustainability (e.g. social sustainability through a teamwork-based motivated workforce with less duplication; economical sustainability through cost efficiencies in less running operating costs; and environmental sustainability through the use of friendly and certified green resources and less waste in all resources and processes).

A12.4.11 Managing Logistics

Managing logistics needs to be adopted to achieve effective and timely movements of all materials within the focal firm and across the SC to promote (e.g. social sustainability by employing the necessary health and safety measures for handling of items in warehouses; economical sustainability through cost efficiencies in delivery of items, through groupage or shortest routes; and environmental sustainability through dedicated route planning to deploy less overall carbon footprint by selecting the right modal mode in all orders' deliveries).

A12.4.12 Managing Performance Measurements

Performance measures need to be adopted, both within the focal firm and across the SC, by implementing a holistic measuring approach of all operations, so as to meet all scheduled targets in a timely and effective manner, so that any variances are detected at the earliest to implement an effective risk management approach to promote business continuity. Such metrics need to include both financial and non-financial indicators to achieve competitive performance.

A12.4.13 Managing Planning and Forecasting

Planning and forecasting is needed to establish all materials requirements and deliveries, with the relevant support of all stakeholders, based on the demand/orders to ensure no stock-out and neither excess stock situation, so as to meet all orders in time and with an effective cash flow to promote economic sustainability.

A12.4.14 Managing Quality and Sustainability

Managing quality and sustainability measures are needed to be adopted to promote compliance to all quality standards and all sustainable measures by all SC stakeholders in all their supplies and processes to promote high quality products/services and also to be in line with all the triple bottom line sustainable measures.

A12.4.15 Managing Risks

Managing risks is needed to monitor and review all operations, both within and outside the focal firm, with the support of the deployment of performance measurements and audits, to ensure immediate bottlenecks identification and to employ timely remedies to promote business continuity of the overall SC.

A12.4.16 Managing Trust

Managing trust is needed to build a solid foundation in all types of interactions, commitments and relationships between people, both within the focal firm and across the SC with all stakeholders. Contractual obligations need to be employed by the focal firm with all SC members, but a level of trust is needed to promote integrity and ethical behaviour by all actors in all type of collaborative initiatives. Such a trustful climate is achieved by sharing the right information and knowledge, so as to enable an effective decision-making approach so as to foster a teamwork environment, with the leverage of each other' capabilities and resources, with a level of flexibility and adaptability, so as to pursue common objectives for a win-win outcome.

A12.4.5 The Technology Deployment

The key scope to deploy technology, in both ICT applications and manufacturing automation, is to enable the holistic SCI management approach to be achieved in a more effective and efficient approach, both within and outside the focal firm, by investing in the right technological applications, in both IT and automation (e.g. ERPs, web portals, CAM and CAD). Such a holistic process needs all management levels to perceive the important role of technology deployment and to understand the dedicated commitment and collaborative initiatives needed by the focal firm with all SC actors, through technology based initiatives, so as to implement the most effective and efficient processes. The ICT applications enable the management of all information in an effective way, by providing SC visibility and transparency, so as that all management levels are informed to enable an effective decision-making process. Such an information platform is highly enhanced with today's applications on the internet, which provides all SC actors the possibility to be updated on a real-time basis across the globe, to assist timely decisions for all actions. IT does not replace people but enable and support all informed decisions effectively and efficiently and sometimes may even drive processes. The SC needs to streamline and unify all the actors' different processes into one seamless holistic process by integrating their IT infrastructure (e.g. interface or replace all disparate legacy IT based systems) to fit and meet all the focal firm requirements with all SC actors, so as to enable real-time information sharing and achieve timely and proactive actions (e.g. web based access provisions through common platforms based on databases/ERPs). Furthermore, the automation based technology within manufacturing is used to perform repetitive tasks with greater process efficiencies to promote a competitive edge.